

Sede amministrativa: Università degli Studi di Padova

Dipartimento di Matematica

SCUOLA DI DOTTORATO DI RICERCA IN SCIENZE MATEMATICHE

INDIRIZZO MATEMATICA

CICLO XXVIII

### On mass distribution and concentration phenomena for linear elliptic partial differential operators

Direttore della Scuola: Ch.mo Prof. Pierpaolo Soravia Coordinatore d'indirizzo: Ch.mo Prof. Franco Cardin

Supervisore: Ch.mo Prof. Pier Domenico Lamberti

Dottorando: Luigi Provenzano

A mio nonno Rocco

## Riassunto

In questa tesi studiamo la dipendenza degli autovalori di operatori differenziali alle derivate parziali di tipo ellittico da perturbazioni della densità di massa su aperti dello spazio euclideo N-dimensionale. In particolare, proviamo risultati di dipendenza continua e analitica degli autovalori di operatori poliarmonici e li applichiamo ad alcuni problemi di ottimizzazione. Per provare i risultati di analiticità, adoperiamo una tecnica generale sviluppata da P.D. Lamberti e M. Lanza de Cristoforis, ottenendo formule per i differenziali di Frechét degli autovalori che ci permettono di caratterizzare le densità critiche sotto il vincolo di massa fissata. Inoltre, enunciamo un 'principio di massimo' per la classe di problemi di ottimizzazione considerata. In seguito, prendiamo in esame una famiglia particolare di densità di massa, ovvero densità che si concentrano al bordo degli aperti dove i problemi differenziali sono definiti. In questo caso, studiamo il comportamento asintotico degli autovalori e delle autofunzioni dei problemi di Neumann per l'operatore di Laplace e l'operatore biarmonico quando la massa si concentra al bordo. Proviamo in entrambi i casi, adattando una tecnica generale sviluppata da J.M. Arrieta, che gli autovalori e le autofunzioni del problema di Neumann convergono agli autovalori e alle autofunzioni di appropriati problemi limite di tipo Steklov. In particolare, il problema di tipo Steklov per l'operatore biarmonico così formulato viene introdotto per la prima volta in questa tesi, dove ne vengono poi studiate alcune proprietà. Nel caso dell'operatore di Laplace, proviamo la validità di un'espansione asintotica degli autovalori e delle autofunzioni del problema di Neumann fino al primo ordine ed otteniamo formule esplicite per i primi termini delle espansioni. Per ottenere questi risultati adattiamo al nostro problema delle tecniche di analisi asintotica utilizzate da M.E. Pérez e S.A. Nazarov per lo studio di sistemi vibranti con masse concentrate in punti o lungo certe curve. Per quanto riguarda il problema di Steklov per l'operatore biarmonico, consideriamo anche il problema della dipendenza degli autovalori dal dominio. Utilizzando sempre la tecnica generale sviluppata da P.D. Lamberti e M. Lanza de Cristoforis, proviamo che le palle sono domini critici per tutti gli autovalori. Inoltre, adattando l'argomento di F. Brock e R.

Weinstock per il problema di Steklov per l'operatore di Laplace, riusciamo a mostrare che la palla massimizza il primo autovalore positivo del problema di Steklov per l'operatore biarmonico tra tutti gli aperti limitati di misura fissata. Proviamo infine una versione quantitativa di questa disuguaglianza isoperimetrica, mostrando poi che l'esponente che compare nella disuguaglianza è ottimale.

La tesi è organizzata come segue. Il Capitolo 1 è dedicato ad alcuni preliminari. Nel Capitolo 2 consideriamo problemi di perturbazione di massa per operatori differenziali ellittici soggetti a diverse condizioni al bordo omogenee. Proviamo risultati di analiticità per gli autovalori e calcoliamo le formule per i differenziali di Frechét che poi verranno usate per caratterizzare le densità di massa critiche sotto il vincolo di massa fissata. Quindi proviamo che per un'ampia famiglia di operatori e condizioni al bordo non esistono densità di massa critiche sotto il solo vincolo di massa. Successivamente proviamo che gli autovalori sono debolmente<sup>\*</sup> continui, il che permette di stabilire una sorta di 'principio di massimo' per la classe di problemi di ottimizzazione considerata nel capitolo. Nel Capitolo 3 consideriamo il problema agli autovalori per l'operatore di Laplace con condizioni al bordo di Neumann e densità di massa che si concentrano al bordo e mostriamo che gli autovalori e le autofunzioni convergono agli autovalori e alle autofunzioni di un opportuno problema di tipo Steklov per l'operatore di Laplace. Il risultato è ottenuto provando la convergenza in norma degli operatori risolventi. Inoltre, studiamo la dipendenza degli autovalori del problema di Steklov dalla densità di massa e mostriamo che sulla palla la densità costante è una densità di massa critica per un'opportuna famiglia di funzioni simmetriche degli autovalori. Nel Capitolo 4 discutiamo il comportamento asintotico degli autovalori del problema di Neumann per l'operatore di Laplace quando la massa si concentra al bordo. In particolare, nel caso della palla troviamo una formula esplicita per le derivate degli autovalori del problema di Neumann nel problema limite e deduciamo che localmente gli autovalori del problema di Steklov minimizzano gli autovalori del problema di Neumann. Inoltre, studiamo il comportamento asintotico degli autovalori del problema di Neumann su aperti limitati del piano e proviamo la validità di un'espansione asintotica per gli autovalori e le autofunzioni del problema di Neumann fino al primo ordine. Otteniamo formule esplicite per i primi termini delle espansioni in termini di soluzioni di opportuni problemi differenziali ausiliari. Nel Capitolo 5 formuliamo il problema agli autovalori per l'operatore biarmonico con condizioni di Neumann e di Steklov. Mostriamo che il problema di Steklov per l'operatore biarmonico può essere ottenuto a partire dal problema di Neumann con densità di massa che si concentra al bordo. In seguito, studiamo la dipendenza degli autovalori del problema di Steklov da perturbazioni del dominio provando formule di tipo Hadamard, e caratterizziamo i domini critici. Quindi, proviamo che per i problemi di Steklov e di Neumann le palle sono critiche. Per quanto riguarda il problema di Steklov, in aggiunta, proviamo che la palla massimizza il primo autovalore positivo tra tutti gli aperti limitati di misura fissata. Infine, all'interno del Capitolo 6 presentiamo alcuni risultati aggiuntivi su problemi di tipo Neumann. In particolare, studiamo la dipendenza degli autovalori dell'operatore biarmonico con condizioni di Neumann dal coefficiente di Poisson. Studiamo anche il comportamento degli autovalori del problema di Neumann per l'operatore di Laplace e per l'operatore biarmonico sulla corona quando la differenza dei due raggi tende a zero.

Riassunto

viii

## Abstract

In this thesis we study the dependence of the eigenvalues of elliptic partial differential operators upon mass density perturbations on open subsets of the N-dimensional euclidean space. We prove continuity and analyticity results for the eigenvalues of poly-harmonic operators and apply them to certain optimization problems. In order to prove analyticity, we use a general technique of P.D. Lamberti and M. Lanza de Cristoforis, and we obtain formulas for the Frechét differentials of the eigenvalues which are used to characterize critical mass densities under the constraint that the total mass is preserved. Then we state a sort of 'maximum principle' in spectral optimization problems for elliptic operators subject to mass density perturbations. Moreover, we consider a special class of densities, namely densities which concentrate near the boundary of open subsets of the N-dimensional euclidean space. We study the asymptotic behavior of the eigenvalues of Neumann-type problems for the Laplace and the biharmonic operator. By adapting a general technique of J.M. Arrieta, we prove that the Neumann eigenvalues converge to the appropriate limiting Steklov eigenvalues. In this way, we formulate a genuine Steklov eigenvalue problem for the biharmonic operator. In the case of the Laplace operator we prove the validity of an asymptotic expansion of the Neumann eigenvalues and eigenfunctions and provide formulas for the first terms in the expansions. We adapt to our case asymptotic analysis techniques used by M.E. Pérez and S.A. Nazarov to describe vibrating systems with masses concentrated at points or along curves. Moreover, we consider the problem of domain perturbations for the biharmonic Steklov problem obtained with this mass concentration procedure and prove that balls are critical domains for all the eigenvalues. Then we adapt the arguments of F. Brock and R. Weinstock to prove that the ball is actually a maximizer for the first positive eigenvalue among bounded domains of given measure. Moreover, we provide a quantitative version of such an isoperimetric inequality, showing also that it is sharp.

This thesis is organized as follows. Chapter 1 is dedicated to some preliminaries. In Chapter 2 we consider mass density perturbation problems for general elliptic operators of higher order subject to various homogeneous boundary conditions. We prove analyticity results for the eigenvalues and compute the Frechét differentials for the symmetric functions of the eigenvalues which are used to provide a characterization of critical mass densities under mass constraint. Then we prove that for a large class of operators and boundary conditions there are no critical mass densities under the constraint of preservation of the total mass. Moreover, we prove weak<sup>\*</sup> continuity of the eigenvalues which allows to state a sort of 'maximum principle' for a class of spectral optimization problems. In Chapter 3 we consider the Neumann eigenvalue problem for the Laplace operator and mass densities which concentrate at the boundary. We prove that the Neumann eigenvalues converge to the appropriate limiting Steklov eigenvalues by proving strong convergence of the resolvent operators. Moreover, we consider the problem of mass density perturbations for the Steklov problem for the Laplace operator and show that in the case of the ball there exist critical mass densities under the sole constraint of preservation of the mass. In Chapter 4 we discuss the asymptotic behavior of the Neumann eigenvalues in the mass concentration phenomenon described in Chapter 3. In particular, in the case of the ball, we prove explicit formulas for the derivatives of the Neumann eigenvalues at the limiting Steklov problem and show that the Steklov eigenvalues locally minimize the Neumann eigenvalues. Moreover, we study the asymptotic behavior of the eigenvalues and the eigenfunctions of the Neumann problem in the case of bounded planar domains. We obtain explicit formulas for the first and second terms of the corresponding asymptotic expansions in terms of solutions to certain auxiliary boundary value problems. In Chapter 5 we introduce the Neumann eigenvalue problem and formulate the Steklov eigenvalue problem for the biharmonic operator. We show that the biharmonic Steklov problem which we introduce can be considered as a limiting Neumann problem for the biharmonic operator in a mass concentration phenomenon. Then we study the dependence of the symmetric functions of the eigenvalues of both Neumann and Steklov problems upon domain perturbations providing Hadamard type formulas, and we give a characterization of critical domains under volume constraint. Then we show that for Neumann and Steklov problems balls are critical domains. Regarding the Steklov problem, we also prove that the ball is a maximizer for the first positive eigenvalue among all bounded open sets of given measure. Finally, in Chapter 6 we include some additional results on Neumann-type problems. In particular, we study the dependence of the Neumann eigenvalues of the biharmonic operator upon the Poisson's ratio. Moreover, we study the behavior of the Neumann eigenvalues of the Laplace and the biharmonic operator on an annulus when the difference between the two radii goes to zero.

# Introduction

Boundary value problems for linear elliptic partial differential equations arise in several models describing various physical phenomena and have been extensively studied for a long time. One of the most famous problems is perhaps the Poisson problem for the Laplace operator

$$\begin{cases} -\Delta u = f, & \text{in } \Omega, \\ u = 0, & \text{on } \partial\Omega, \end{cases}$$
(0.0.1)

which arises for example in the study of the deformation of a fixed membrane of shape  $\Omega \subset \mathbb{R}^2$  subject to an exterior force which is represented by the function f (see e.g., [31, 94, 105] for a detailed discussion and historical information). Problem (0.0.1) is a prototype of second order elliptic problems. The theory of linear second order elliptic equations is well developed and is considered nowadays classical.

However, many other phenomena in the applied sciences are modeled by higher order equations. It was already known at the beginning of the nineteenth century that the study of the bending of a clamped plate leads to the analysis of the following fourth order problem

$$\begin{cases} \Delta^2 u = f, & \text{in } \Omega, \\ u = \frac{\partial u}{\partial \nu} = 0, & \text{on } \partial \Omega, \end{cases}$$

where  $\Omega \subset \mathbb{R}^2$  represents the midplane of the plate and the function f represents the applied load. It was then natural to consider more general equations involving the polyharmonic operators  $(-\Delta)^m$ ,  $m \in \mathbb{N}$ , subject to different types of homogeneous boundary conditions. It is worth mentioning the pioneeristic papers of Almansi [6, 7] and the book of Nicolesco [90] on this subject. These authors were among the first to study the properties of polyharmonic functions and higher order elliptic equations. Since then the interest for polyharmonic operators has grown but the theory of higher order elliptic equations is far less developed than the theory of analogous second order equations. As is well-known, this is also due to the fact that for higher order equations a maximum principle does not hold in general.

A general theory for boundary value problems for linear elliptic operators of order 2m has been developed by Agmon-Douglis-Nirenberg [2, 3, 4]. We also mention the recent book [47] which is devoted to an extensive study of boundary value problems for polyharmonic operators.

In this thesis we are mainly interested in eigenvalue problems for elliptic operators of second and higher order depending on a parameter  $\rho$  which plays the role of a mass density for the underlying physical system. In the case of the Laplace operator subject to Dirichlet boundary conditions such problem can be written as

$$\begin{cases} -\Delta u = \lambda \rho u, & \text{in } \Omega, \\ u = 0, & \text{on } \partial \Omega, \end{cases}$$
(0.0.2)

where  $\rho$  is a measurable positive and bounded function on  $\Omega$ . Here and in the sequel  $\Omega$  is an open subset of  $\mathbb{R}^N$  with sufficiently smooth boundary. We also consider the eigenvalue problem for the Laplace operator subject to Neumann boundary conditions

$$\begin{cases} -\Delta u = \lambda \rho u, & \text{in } \Omega, \\ \frac{\partial u}{\partial \nu} = 0, & \text{on } \partial \Omega. \end{cases}$$
(0.0.3)

As for the biharmonic operator subject to Dirichlet boundary conditions, the eigenvalue problem reads

$$\begin{cases} \Delta^2 u = \lambda \rho u, & \text{in } \Omega, \\ u = \frac{\partial u}{\partial \nu} = 0, & \text{on } \partial \Omega. \end{cases}$$
(0.0.4)

Also for the biharmonic operator, we consider different types of homogeneous boundary conditions such as Neumann boundary conditions

$$\begin{cases} \Delta^2 u = \lambda \rho u, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial \nu^2} = \frac{\partial \Delta u}{\partial \nu} + \operatorname{div}_{\partial \Omega} \left( D^2 u \cdot \nu \right) = 0, & \text{on } \partial \Omega, \end{cases}$$
(0.0.5)

and intermediate boundary conditions

$$\begin{cases} \Delta^2 u = \lambda \rho u, & \text{in } \Omega, \\ u = \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial \Omega. \end{cases}$$
(0.0.6)

As we have said, keeping in mind important problems in linear elasticity (see e.g., [31]), we shall think of the weight  $\rho$  as a mass density. In fact, for N = 2 problems (0.0.2) and (0.0.3) are related to the study of the transverse vibrations of a thin elastic membrane which has a fixed or a free frame respectively, and the mass of which is displaced on  $\Omega \subset \mathbb{R}^2$  with density  $\rho$ . On the other hand, problems (0.0.4), (0.0.5) and (0.0.6) are related to the study of the transverse vibrations of an elastic plate with density  $\rho$  which is clamped, free, and hinged, respectively. We shall refer to the quantity  $\int_{\Omega} \rho dx$  as the total mass of the body.

As for higher order operators, we consider the following eigenvalue problem

$$\begin{cases} (-\Delta)^m u = \lambda \rho u, & \text{in } \Omega, \\ \frac{\partial^j u}{\partial \nu^j} = 0, \ \forall j = 0, ..., k - 1 & \text{on } \partial\Omega, \\ \mathcal{B}_j u = 0, \ \forall j = 1, ..., m - k & \text{on } \partial\Omega, \end{cases}$$
(0.0.7)

where  $\mathcal{B}_j$  are uniquely defined 'complementing' boundary operators (we refer to [88] for details). We observe that when m = 1, k = 1 problem (0.0.7) gives (0.0.2), while for m = 1, k = 0 we have problem (0.0.3). When m = 2 and k = 2, 1 and 0 we have problems (0.0.4), (0.0.6) and (0.0.5), respectively, for suitable operators  $\mathcal{B}_j$ . Problem (0.0.7) motivates the study of a more general class of eigenvalue problems for higher order operators of the form

$$\sum_{0 \le |\alpha|, |\beta| \le m} (-1)^{|\alpha|} D^{\alpha} \left( A_{\alpha\beta} D^{\beta} u \right) = \lambda \rho u,$$

subject to homogeneous boundary conditions on an open subset  $\Omega$  of  $\mathbb{R}^N$ . (We remark that the space dimension does not play any relevant role in our discussion and restriction to the case N = 2 will be done only in Subsection 4.2).

In this thesis we study the dependence of the solutions of the above mentioned problems upon perturbation of the density  $\rho$ , with special attention for the behavior of the eigenvalues.

We note that there is a vast literature on the dependence of the eigenvalues of the Laplace operator subject to Dirichlet and Neumann boundary conditions upon mass density perturbations.

One of the fundamental problems concerns the study of the qualitative behavior of the eigenvalues when the density is perturbed and the corresponding results concern continuity, differentiability and analyticity of the eigenvalues. A related fundamental problem concerns the optimization (maximization or minimization) of the eigenvalues (or suitable functionals of the eigenvalues) with respect to the variable  $\rho$  under suitable constraints, such as  $\int_{\Omega} \rho dx = \text{const.}$  We refer to the monographs [14, 31, 46, 59, 66, 92, 95] and to the papers [43, 44, 69] for an introduction to this subject. We also refer to the recent papers [27, 32, 33, 34, 73] for qualitative results on mass density perturbations problems and for information on the properties of maximizers and minimizers of certain functionals of the eigenvalues of composite membranes.

Another important problem concerns the study of particular classes of mass distributions  $\rho$  which are of order  $\varepsilon^{-1}$  in  $\varepsilon$ -neighborhoods of points or hypersurfaces contained in  $\Omega$  and of order  $\varepsilon$  in the rest of  $\Omega$ , as  $\varepsilon$  goes to zero. There is a vast literature concerning vibrating systems containing concentrated masses along curves or around certain points. We mention the extensive monographs [63, 82, 83, 87, 91, 97, 101] and the survey paper [81] for an introduction to the asymptotic analysis techniques for vibrating systems with concentrated masses. We also refer to the recent papers [49, 50, 51, 52, 53, 54, 55, 56, 80] for more information and results on the asymptotics of the eigenvalues and the eigenfunctions of vibrating systems containing stiff or heavy regions. We also mention the alternative approach of [9, 10, 11, 64, 65, 96], where the authors consider the asymptotic behavior of the solutions of elliptic or parabolic equations with terms concentrating at the boundary. In these cases, the results are obtained by means of functional analysis techniques and resolvent estimates for elliptic operators. It is also worth mentioning another approach for singularly perturbed problems based on potential theory and functional analysis proposed in [35, 79].

In this thesis we face three main problems. First, we study the qualitative behavior of the eigenvalues and, in particular, we prove continuity and analytiticy results for the dependence of suitable functions of the eigenvalues of elliptic operators upon mass density perturbations. These results are applied to certain optimization problems. Second, keeping in mind the above mentioned optimization problems, we consider some classes of mass densities concentrating at the boundary or at points, and study the asymptotic behavior of the eigenelements of elliptic operators under these singular mass density perturbations. Third, we define a genuine Steklov problem for the biharmonic operator by means of a family of Neumann-type problems with mass density concentrating at the boundary, and we study the dependence of its eigenvalues upon perturbation of  $\Omega$ .

Concerning the first type of problems, our study is motivated by wellknown results of Krein [69] and Cox and McLaughlin [32, 33, 34] concerning the description of optimal mass densities for the eigenvalues of the Dirichlet Laplacian under the assumption that the total mass is fixed and the additional condition  $A \leq \rho \leq B$ , where A and B are fixed positive constants. Complete solution to this problem for N = 1 was given by Krein in [69], where explicit formulas for the minimizers and the maximizers of all the eigenvalues were established. In particular, it turns out that optimal mass densities are bang-bang solutions, i.e., they satisfy the condition  $(\rho - A)(B - \rho) = 0$  on  $\Omega$ . The case N > 1 is discussed in [33, 34] and it is proved that maximizers and minimizers of the first eigenvalue of the Dirichlet Laplacian are bang-bang solutions. Moreover, we remark that Friedland [43] proves that the minimizers of suitable functionals of the eigenvalues are bang-bang as well.

Here, we prove the continuity of the eigenvalues not only with respect to the strong topology of  $L^{\infty}(\Omega)$ , but also with respect to the weak<sup>\*</sup> topology, which is more relevant in optimization problems, see Theorem 2.1.5. In this sense we generalize a result of Cox and McLaughlin [32, 33, 34] for the Dirichlet Laplacian. Then we address the problem of the analyticity of the eigenvalues. We use the general technique developed by Lamberti and Lanza de Cristoforis in [75] for compact self-adjoint operators in Hilbert spaces and prove that all simple eigenvalues and the elementary symmetric functions of multiple eigenvalues are real analytic functions of  $\rho$ , see Theorem 2.2.1. We remark that in general one cannot expect to prove analyticity of the eigenvalues with respect to  $\rho$ . This is due to well-known bifurcation phenomena which prevent multiple eigenvalues from being differentiable functions of the parameters involved in the equation. In order to avoid such a situation, in the case of multiple eigenvalues we consider the elementary symmetric functions of the eigenvalues which have the effect of bypassing the splitting phenomenon. Then we compute the appropriate formulas for the Frechét differentials of the symmetric functions of the eigenvalues which are used to address extremum problems. We remark that the general technique of [75] has been used to study domain perturbation problems for different types of operators and boundary conditions, see e.g., [22, 23, 74, 76, 77]. As for mass density perturbation problems, we mention the paper of Lamberti [73] where it is considered the problem of the dependence of the eigenvalues of the Dirichlet Laplacian. In the spirit of [73], here we prove that for a large class of differential operators and boundary conditions, there are no critical mass densities under mass constraint, see Theorem 2.3.2. Moreover, we prove a sort of 'maximum principle' in optimization problems which can be stated as follows: if C is a weakly<sup>\*</sup> compact set of mass densities with prescribed total mass, then 'all simple eigenvalues and the symmetric functions of multiple eigenvalues admit point of maximum and minimum in C with mass constraint and such points of maximum and minimul belong to  $\partial C'$ , see Corollaries 2.3.5 and 2.4.1 for the precise statement.

The above mentioned maximum principle and the corresponding absence of critical mass densities has led us to consider a slightly different kind of problems, where the mass density concentrates at the boundary. In this case, Steklov-type problems arise. Recall that the classical Steklov eigenvalue problem for the Laplace operator reads

$$\begin{cases} \Delta u = 0, & \text{in } \Omega, \\ \frac{\partial u}{\partial \nu} = \lambda \rho u, & \text{on } \partial \Omega, \end{cases}$$
(0.0.8)

where  $\rho$  is a measurable positive and bounded function defined on the boundary  $\partial\Omega$ . This problem has a rather different nature from the eigenvalue problems mentioned above. In fact, for this problem it is possible to find in some cases critical mass densities for the symmetric functions of the eigenvalues under the sole mass constraint  $\int_{\partial\Omega} \rho d\sigma = 0$ . For example, in the case of the ball in  $\mathbb{R}^N$  it is possible to prove that the constant density is a critical point for suitable families of symmetric functions of the eigenvalues, see Corollary 3.2.6. This is not surprising. Indeed, it has been proved by Hersch, Payne and Schiffer in [61] for N = 2, that the constant density is a maximizer for the product of the first two positive eigenvalues (in particular, it is a maximizer for the first positive eigenvalue). We also refer to the book of Bandle [14] for more results and further discussions on the dependence of the eigenvalues of the Steklov Laplacian upon density perturbations on planar domains.

Moreover, for what concerns the minimization problem, we prove that there there are no minimizers for the first positive eigenvalue of the Steklov Laplacian under mass density perturbations preserving the total mass, see Theorems 3.3.3, 3.3.10 and 3.3.19.

We refer to [99] for the physical derivation of problem (0.0.8). We also refer to the recent survey [48] for more information on the Steklov eigenvalues and to [74] for other related problems.

We note that for N = 2 problem (0.0.8) provides the vibration modes of a free elastic membrane the total mass of which is concentrated at the boundary with dentisy  $\rho$ . We provide an explanation of this known concentration phenomenon in terms of spectral convergence of operators. Namely, for any  $\varepsilon > 0$  we define a suitable 'mass density'  $\rho_{\varepsilon}$  in the whole of  $\Omega$  which is of order  $\varepsilon^{-1}$  in a  $\varepsilon$ -neighborhood of the boundary  $\partial\Omega$ , as  $\varepsilon$  goes to zero, while it is of order  $\varepsilon$  in the rest of  $\Omega$ , and such that  $\int_{\Omega} \rho_{\varepsilon} dx = \text{const.}$  Then we consider the eigenvalues of problem (0.0.3) with density  $\rho_{\varepsilon}$  and show that such eigenvalues converge to the eigenvalues of problem (0.0.8) as  $\varepsilon$  goes to zero, see Theorem 3.1.21 and Corollary 3.1.42. This result can be proved by using the notion of compact convergence for the resolvent operators but can also be obtained as a consequence of the more general result proved in [10]. We refer to [9, 10, 11, 64, 65, 96] for a general approach to this kind of problems. Thus the Steklov problem can be considered as a limiting Neumann problem.

Then we address the problem of describing the asymptotic behavior of the Neumann eigenvalues in this mass concentration phenomenon. In particular we show that Neumann eigenvalues are differentiable with respect to  $\varepsilon$  in a small neighborhood of  $\varepsilon = 0$  and provide explicit formulas for their derivatives. First, we consider the case of the unit ball centered at zero in  $\mathbb{R}^N$  and next the case of smooth bounded domains in  $\mathbb{R}^2$ . In the case of the ball we prove that the Neumann eigenvalues have a monotone behavior near their limiting Steklov eigenvalues, and we can conclude that the Steklov eigenvalues locally minimize the Neumann eigenvalues for  $\varepsilon > 0$ small enough, see Theorem 4.1.20 and Corollary 4.1.22. It is interesting to compare these results with those in [89] where authors consider the Neumann problem on the annulus  $1 - \varepsilon < |x| < 1$  and prove that for N = 2 the first positive eigenvalue is a decreasing function of  $\varepsilon$ .

We note that the techniques that we use for the description of the asymptotic behavior of the Neumann eigenvalues in the case of the ball in  $\mathbb{R}^N$  and in the case of general open subsets of  $\mathbb{R}^2$  are completely different. In the case of the unit ball centered at zero we use Bessel functions to recast the eigenvalue problem in the form of an equation  $F(\lambda, \varepsilon) = 0$  in the unknowns  $\lambda, \varepsilon$ . Then after some preparatory work it is possible to apply the Implicit Function Theorem and conclude. We note that, despite the idea of the proof is rather simple and also used in other contexts (see e.g., [78]) this method requires lenghty computations, suitable Taylor's expansions and estimates on the corresponding remainders, as well as recursive formulas for the crossproducts of Bessel functions and their derivatives. Importantly, we remark that the multiplicity of the eigenvalues which is often an obstruction in the application of standard asymptotic analysys, does not affect our method. On the other hand, in the case of a general open and bounded subset  $\Omega$  of  $\mathbb{R}^2$ , we show the validity of an asymptotic expansion of the eigenvalues and of the eigenfunctions of problem (0.0.3) as  $\varepsilon$  goes to zero. In addition we provide explicit formulas for the first two coefficients in the expansions in terms of solutions to suitable auxiliary problems, see Theorems 4.2.10 and 4.2.14. In order to obtain our results, we follow the approach of [52, 53]. The results concerning the asymptotic behavior of Neumann eigenvalues on general domains in  $\mathbb{R}^2$  have been obtained in collaboration with Dr. Matteo Dalla Riva (see also [36]).

The results on the behavior of the eigenvalues of the Neumann Laplacian when the mass concentrates at the boundary motivates the study of an analogous mass concentration problem for the biharmonic operator subject to Neumann-type boundary conditions. Namely, we consider the following Neumann-type problem

$$\begin{cases} \Delta^2 u - \tau \Delta u = \lambda \rho_{\varepsilon} u, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial \nu^2} = \tau \frac{\partial u}{\partial \nu} - \frac{\partial \Delta u}{\partial \nu} - \operatorname{div}_{\partial \Omega} \left( D^2 u \cdot \nu \right) = 0, & \text{on } \partial \Omega, \end{cases}$$
(0.0.9)

where  $\tau \geq 0$  is a fixed non-negative constant and represents the ratio of lateral tension due to flexural rigidity of the plate. Then the eigenvalues of (0.0.9) converge to the eigenvalues of the following Steklov-type problem for the biharmonic operator

$$\begin{cases} \Delta^2 u - \tau \Delta u = 0, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial \Omega, \\ \tau \frac{\partial u}{\partial \nu} - \frac{\partial \Delta u}{\partial \nu} - \operatorname{div}_{\partial \Omega} \left( D^2 u \cdot \nu \right) = \lambda u, & \text{on } \partial \Omega, \end{cases}$$
(0.0.10)

see Theorem 5.3.6 and Corollary 5.3.12. Thus problem (0.0.10) can be considered as a problem modeling the free vibration modes of a plate the mass of which is concentrated at the boundary, and therefore is a natural generalization to the biharmonic operator of the classical Steklov problem (0.0.8).

Moreover, we address the problem of the dependence of the eigenvalues of problem (0.0.10) upon domain perturbations. We note that domain perturbation problems have been widely studied in the case of the Laplace operator subject to different homogeneous boundary conditions (Dirichlet, Neumann, Steklov, etc.), in particular for shape optimization problems. We recall for instance the Faber-Krahn inequality proved in [41, 68], which says that the ball minimizes the first eigenvalue of the Dirichlet Laplacian among all domains with a fixed measure. Similar results have been shown also for other boundary conditions in [19, 104, 106]. As for the biharmonic operator much less is known. The well-known Rayleigh conjecture on the minimization of the first eigenvalue of the clamped plate has been solved by Nadirashvili for N = 2 in [86] and by Ashbaugh and Benguria for N = 3 in [12], while the general case remains an open problem (see also [85, 100]). Regarding Neumann boundary conditions, Chasman [28] proved that the ball is a maximizer for the first positive eigenvalue of problem (0.0.9). We refer to [57, 60] for a general approach to domain perturbation problems and to [59] for a comprehensive discussion on shape optimization problems for the eigenvalues of elliptic operators. We also refer to [22, 23] where the authors prove analiticity properties in the spirit of [75] for Dirichlet and intermediate boundary conditions respectively, and show that balls are critical domains for all elementary symmetric functions of the eigenvalues.

In this thesis, first, we prove analycity results for the symmetric functions of the eigenvalues of problem (0.0.10) in the spirit of [75] (see Theorems 5.4.3 and 5.4.15) and show that balls are critical domains under volume coinstraint (see Theorem 5.4.21). Second, we prove that the ball is actually a maximizer for the first positive eigenvalue of problem (0.0.10) among all bounded open sets of given volume, for any constant  $\tau > 0$ , see Theorem 5.5.27. This is done by following the approach of [19, 28, 104]. We have also considered the problem of the stability of the optimal shape. In fact, we have provided a quantitative version of the isoperimetric inequality for the first positive Steklov eigenvalue and of the analogue inequality for the Neumann problem proved in [28]. Moreover, the two inequalities turn out to be sharp (see Theorems 5.6.22 and 5.6.59). The results concerning problem (0.0.10) have been obtained in collaboration with Dr. Davide Buoso (see also [24, 25]). The results concerning the sharpness of the isoperimetric inequalities have been obtained in collaboration with Dr. Davide Buoso and Dr. Laura M. Chasman (see also [21]).

Note that problem (0.0.10) should not be confused with another important Steklov-type problem already discussed in the literature, namely

$$\begin{cases} \Delta^2 u = 0, & \text{in } \Omega, \\ u = 0, & \text{on } \partial\Omega, \\ \Delta u = \lambda \frac{\partial u}{\partial u}, & \text{on } \partial\Omega, \end{cases}$$
(0.0.11)

which has a rather different nature. In fact, for the first positive eigenvalue of problem (0.0.11) the minimization is an interesting open problem (rather than maximization), and explicit examples show that, surprisingly, the ball is not a minimizer.

At the end of this thesis we include some further results on Neumanntype problems. In particular, we consider the behavior of the eigenvalues of the biharmonic operator with Neumann boundary condition and non-zero Poisson's ratio  $\sigma$ , namely

$$\begin{cases} \Delta^2 u = \lambda u, & \text{in } \Omega, \\ (1 - \sigma) \frac{\partial^2 u}{\partial \nu^2} + \sigma \Delta u = 0, & \text{on } \partial \Omega, \\ \frac{\partial \Delta u}{\partial \nu} + (1 - \sigma) \text{div}_{\partial \Omega} \left( D^2 u \cdot \nu \right)_{\partial \Omega} = 0, & \text{on } \partial \Omega, \end{cases}$$
(0.0.12)

where  $\sigma \in [0, 1[$ . We refer to [39], where the author studies the behavior of the eigenvalues of the biharmonic operator on planar domains upon perturbations of the Poisson's ratio . We prove that all the eigenvalues of problem (0.0.12) go to zero as  $\sigma \to 1^-$ . Moreover, we show that problem (0.0.12)with  $\sigma = 1$  admits an increasing sequence of positive eigenvalues of finite multiplicity diverging to  $+\infty$  and which coincide with the eigenvalues of the Dirichlet problem for the biharmonic operator.

Moreover, we consider the eigenvalue problem for the Laplace and biharmonic operator with Neumann boundary conditions on an annulus of radii 1 and  $1 - \varepsilon$ , with  $\varepsilon \in ] -\infty, 0[\cup]0, 1[$ . In particular, we are interested in the behavior of all the eigenvalues as  $\varepsilon \to 0$ . We mention the paper [89] where the authors consider the first positive eigenvalue of the Laplace operator with Neumann boundary conditions in the case N = 2, and prove that such eigenvalue is continuously differentiable with respect to  $\varepsilon$  in a suitable interval of  $\mathbb{R}$  containing 0, and moreover, that it is an increasing function of  $\varepsilon$ . We prove an analogous result for all the Neumann eigenvalues of the Laplace and biharmonic operator for all  $N \geq 2$ . Namely, we provide an asymptotic expansion of all the Neumann eigenvalues at  $\varepsilon = 0$ . As a bypass product, we prove that all the positive eigenvalues are strictly increasing with respect to  $\varepsilon$  in a suitable neighborhood of 0.

The thesis is organized as follows. Chapter 1 is dedicated to some preliminaries. In Chapter 2 we consider mass density perturbation problems for general elliptic operators of higher order subject to various homogeneous boundary conditions. For all these cases we prove analyticity results for the eigenvalues in the spirit of [75] and compute the Frechét differentials for the symmetric functions of the eigenvalues which are used to provide a characterization of critical mass densities under mass constraint. Then we prove that for a large class of operators and boundary conditions there are no critical mass densities under the sole mass constraint. Moreover, we prove weak\* continuity of the eigenvalues which combined with the results of non-existence of critical points for the eigenvalues, allow to state a sort of 'maximum principle' for a class of optimization problems. In Chapter 3 we consider the Neumann eigenvalue problem for the Laplace operator and mass densities which concentrate at the boundary. We prove that the Neumann eigenvalues converge to the appropriate limiting Steklov eigenvalues by means of strong convergence of the resolvent operators. Moreover, we consider the problem of mass density perturbations for the Steklov problem for the Laplace operator and show that this problem has a rather different nature than that of the problems considered in Chapter 2. In fact we show that in the case of the open unit ball there exist critical mass densities under the sole mass constraint. Moreover, in this chapter we consider other examples of mass concentrations for the Steklov and Dirichlet Laplacian. In Chapter 4 we discuss the asymptotic behavior of the Neumann eigenvalues in the mass concentration phenomenon described in Chapter 3. In particular, in the case of the ball, we prove explicit formulas for the derivatives of the Neumann eigenvalues at the limiting Steklov problem and show that the Steklov eigenvalue locally minimize the Neumann eigenvalues. Moreover, we study the asymptotic behavior of the eigenvalues and eigenfunctions of the Neumann problem in the case of bounded planar domains. We obtain explicit formulas for the first and second terms of the corresponding asymptotic expansions in terms of solutions to certain auxiliary boundary value problems, in the spirit of [52, 53]. In Chapter 5 we consider the biharmonic Neumann eigenvalue problem as described in [28] and problem (0.0.10). We show that problem (0.0.10) can be considered as a limiting Neumann problem for the biharmonic operator in a mass concentration phenomenon. Then we study the dependence of the symmetric functions of the eigenvalues of both Neumann and Steklov problems upon domain perturbations and provide Hadamard type formulas, which allow to give a characterization of critical domains under volume constraint. Then we show that for Neumann and Steklov problems balls are critical domains. Regarding the Steklov problem (0.0.10), we also prove that the ball is a maximizer of the first positive eigenvalue among all bounded open sets of given measure. Finally, in Chapter 6 we collect some results on the Neumann eigenvalues of the Laplace and the biharmonic operators. In particular, we study the dependence of the Neumann eigenvalues of the biharmonic operator upon the Poisson's ratio  $\sigma$ , with particular attention to the behavior of the eigenvalues as  $\sigma \to 1^-$ . Moreover, we study the asymptotic behavior of the Neumann eigenvalues both of the Laplace and the biharmonic operators on an annulus when the difference between the two radii goes to zero.

Part of the results in this thesis have been published or accepted for publication. The results in Chapter 2 on the dependence of the eigenvalues of general elliptic operators of higher order on the mass density and the corresponding 'maximum principle' have been published in [70]. The results in Chapter 3 on the Steklov eigenvalues as limiting Neumann eigenvalues have been partially published in [72]. The results in the first section of Chapter 4 on the asymptotic behavior of the Neumann eigenvalues on the ball are part of the paper [71], which has been accepted for publication. The results in the second section of Chapter 4 on the asymptotic behavior have been published in [24, 25]. The results on the quantitative isoperimetric inequality for the Neumann problem and the sharpness of the Neumann and Steklov inequalities are part of the paper [21] in preparation. The results in the first section of Chapter 6 are part of the paper [93] in preparation.

Introduction

xxii

# List of principal symbols

N	(natural numbers)
$\mathbb{N}_0$	(natural numbers including 0)
$f^{(-1)}$	(inverse function of an invertible function $f$ )
$r^{-1}, f^{-1}$	(reciprocal of a real non-zero number and of a non-vanishing function)
$B,\partial B$	(open unit ball in $\mathbb{R}^N$ and unit sphere in $\mathbb{R}^N$ , centered at zero)
$\omega_N$	(Lebesgue measure of the unit ball)
ν	(outer unit normal to a smooth subset of $\mathbb{R}^N)$
$d\sigma$	(surface measure)
$rac{\partial^k u}{\partial  u^k}$	(normal $k$ -th derivative of $u$ )
$ \alpha ,\alpha!,D^{\alpha}$	(multi-index notation)
Df	(Jacobian matrix of $f$ )
$D^2f$	(Hessian matrix of $f$ )
${\rm div}_{\partial\Omega}F$	(tangential divergence of $F$ defined by $\operatorname{div}_{\partial\Omega} F = \operatorname{div}_{F _{\partial\Omega}} - (DF \cdot \nu) \cdot \nu$ )
$f(z) \in O(g(z))$ as $z \to 0$	(there exists $C > 0$ such that $ f(z)  \le C g(z) $ for any z sufficiently close to zero)
$(r, \theta)$	(standard spherical coordinates in $\mathbb{R}^N$ , see (A))
$\Delta_S$	(Laplace-Beltrami operator on the unit sphere $\partial B$ of $\mathbb{R}^N$ , see (B))

(A) Standard spherical coordinates  $(r, \theta)$  in  $\mathbb{R}^N$ , where  $\theta = (\theta_1, ..., \theta_{N-1})$ . The corresponding change of variables is

$$x_{1} = r \cos(\theta_{1}),$$

$$x_{2} = r \sin(\theta_{1}) \cos(\theta_{2}),$$

$$\vdots$$

$$x_{N-1} = r \sin(\theta_{1}) \sin(\theta_{2}) \cdots \sin(\theta_{N-2}) \cos(\theta_{N-1}),$$

$$x_{N} = r \sin(\theta_{1}) \sin(\theta_{2}) \cdots \sin(\theta_{N-2}) \sin(\theta_{N-1}),$$

with  $\theta_1, ..., \theta_{N-2} \in [0, \pi]$ ,  $\theta_{N-1} \in [0, 2\pi[$  (here it is understood that  $\theta_1 \in [0, 2\pi[$  if N = 2)).

(B) Laplace-Beltrami operator on the unit sphere  $\partial B$  in  $\mathbb{R}^N$ , defined by

$$\Delta_S = \sum_{j=1}^{N-1} \frac{1}{q_j (\sin \theta_j)^{N-j-1}} \frac{\partial}{\partial \theta_j} \left( (\sin \theta_j)^{N-j-1} \frac{\partial}{\partial \theta_j} \right),$$

where

$$q_1 = 1, \quad q_j = (\sin \theta_1 \sin \theta_2 \cdots \sin_{\theta_{j-1}})^2, \quad j = 2, ..., N - 1,$$

(see e.g., [67]).

xxiv

# Contents

Ri	iassu	nto	v
A	bstra	uct	ix
In	trod	uction	xi
Li	st of	principal symbols x:	xiii
1	<b>Pre</b> 1.1 1.2 1.3 1.4	liminaries and notation         Sobolev Spaces         Compact operators and symmetric functions of the eigenvalues         Eigenvalues of elliptic operators         Bessel functions, modified Bessel functions and ultraspherical         Bessel functions         1.4.1         Bessel functions         1.4.2         Modified Bessel function         1.4.3         Ultraspherical Bessel and modified Bessel functions         1.4.4         Spherical harmonics	1 5 8 10 10 13 14 16
<b>2</b>	Elli	ptic operators subject to mass density perturbations and	10
	2.1 2.2 2.3 2.4 2.5	Continuity of the eigenvalues	20 22 25 27 29
3	Ma	ss concentration phenomena for second order operators	33
	3.1	Neumann to Steklov eigenvalues	34
	3.2	The Steklov eigenvalue problem. Mass density perturbations3.2.1Continuity and analyticity of the eigenvalues3.2.2Critical mass densities	49 49 50
	3.3	Minimization of the first positive Steklov eigenvalue $\ldots$ . 3.3.1 The case of the ball in $\mathbb{R}^2$	52 52

		3.3.2	The case of the ball in $\mathbb{R}^N$ with $N \ge 3$	56	
		3.3.3	The case of an arbitrary $\Omega \subset \mathbb{R}^2$	59	
	3.4 On the optimization of the first positive Dirichlet and				
		mann	eigenvalues	62	
		3.4.1	Optimization of the first Dirichlet eigenvalue	63	
		3.4.2	Minimization of the first positive Neumann eigenvalue	69	
4	Neı	ımann	and Steklov problems: an asymptotic analysis	73	
	4.1	The c	ase of the ball in $\mathbb{R}^N$	74	
		4.1.1	Asymptotic behavior of Neumann eigenvalues	75	
		4.1.2	Estimates for the remainders	84	
		4.1.3	The case $N = 1 \dots \dots \dots \dots \dots \dots \dots \dots \dots \dots$	86	
		4.1.4	Behavior of the eigenvalues under dilations	89	
		4.1.5	Cross products of Bessel functions	90	
	4.2	Bound	led domains of class $C^2$ in $\mathbb{R}^2$	93	
		4.2.1	Asymptotic expansions and derivatives of the eigen-		
			values	98	
		4.2.2	First step of the proof of Theorems $4.2.10$ and $4.2.14$ .	99	
		4.2.3	Second Step of the proof of Theorems 4.2.10 and 4.2.14	107	
		4.2.4	Well-posedness of problem $(4.2.31)$	120	
		4.2.5	The case of the unit ball in $\mathbb{R}^2$	122	
		426	Heuristic determination of the expansions	123	
		1.2.0			
5	Ma	ss con	centration phenomena for fourth order operators.		
5	Mas A n	ss cone iew bił	centration phenomena for fourth order operators. narmonic Steklov problem 1	29	
5	<b>Mas</b> <b>A</b> n 5.1	ss cone new bil Formu	centration phenomena for fourth order operators.         narmonic Steklov problem         1         tlating the problem	1 <b>29</b> 130	
5	Mas A n 5.1 5.2	ss cone new bil Formu The S	centration phenomena for fourth order operators.         narmonic Steklov problem         1         teklov spectrum	1 <b>29</b> 130 133	
5	Mas A n 5.1 5.2 5.3	ss cone new bil Formu The S Neum	centration phenomena for fourth order operators.         narmonic Steklov problem         1         tlating the problem	1 <b>29</b> 130 133	
5	Mas A n 5.1 5.2 5.3	ss cone new bil Formu The S Neum der m	centration phenomena for fourth order operators.         narmonic Steklov problem         ilating the problem	1 <b>29</b> 130 133 135	
5	Mas A n 5.1 5.2 5.3 5.4	ss cone new bil Formu The S Neum der m Symm	centration phenomena for fourth order operators.         narmonic Steklov problem         ulating the problem         teklov spectrum         ann problem and behavior of Neumann eigenvalues unass concentration at the boundary         attrictions of the eigenvalues.	130 133 135	
5	Mas A n 5.1 5.2 5.3 5.4	ss cond formu Formu The S Neum der m Symm turbat	centration phenomena for fourth order operators.         narmonic Steklov problem         nalating the problem	130 133 135 140	
5	Mas A n 5.1 5.2 5.3 5.4	ss cond Formu The S Neum der m Symm turbat 5.4.1	centration phenomena for fourth order operators.         narmonic Steklov problem         ulating the problem         teklov spectrum         ann problem and behavior of Neumann eigenvalues unass concentration at the boundary         tetric functions of the eigenvalues. Isovolumetric pertions         The Steklov problem	130 133 135 140 141	
5	<b>Mas</b> <b>A</b> n 5.1 5.2 5.3 5.4	ss cone Formu The S Neum der m Symm turbat 5.4.1 5.4.2	centration phenomena for fourth order operators.         narmonic Steklov problem         ulating the problem         teklov spectrum         ann problem and behavior of Neumann eigenvalues unass concentration at the boundary         ass concentration of the eigenvalues.         Isovolumetric perturbatons	130 133 135 140 141 151	
5	<b>Mas</b> <b>A n</b> 5.1 5.2 5.3 5.4	ss cond Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3	centration phenomena for fourth order operators.         narmonic Steklov problem         nalating the problem	29 130 133 135 140 141 151 154	
5	<b>Mas</b> <b>A</b> n 5.1 5.2 5.3 5.4	ss cond Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3 The fu	centration phenomena for fourth order operators.         narmonic Steklov problem         ulating the problem         ulating the problem         ann problem and behavior of Neumann eigenvalues unass concentration at the boundary         ass concentration of the eigenvalues.         Isovolumetric perturbatons         Isovolumetric perturbatons         The Neumann problem         Indamental tone of the ball.         An isoperimetric inequality	29 130 133 135 140 141 151 154 156	
5	<b>Mas</b> <b>A</b> n 5.1 5.2 5.3 5.4	ss cone Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3 The fu 5.5.1	centration phenomena for fourth order operators.         narmonic Steklov problem         ulating the problem         ulating the problem         teklov spectrum         ann problem and behavior of Neumann eigenvalues unass concentration at the boundary         ass concentration of the eigenvalues. Isovolumetric pertributions         The Steklov problem         Isovolumetric perturbatons         Indamental tone of the ball. An isoperimetric inequality	29 130 133 135 140 141 151 154 156 157	
5	<b>Mas</b> <b>A n</b> 5.1 5.3 5.4	ss cond Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3 The fu 5.5.1 5.5.2	centration phenomena for fourth order operators.         narmonic Steklov problem         nalating the problem         nann problem and behavior of Neumann eigenvalues unass concentration at the boundary         ass concentrations of the eigenvalues. Isovolumetric perturbatons         The Steklov problem         Isovolumetric perturbatons         Indamental tone of the ball. An isoperimetric inequality         Eigenvalues and eigenfunctions on the ball	<b>229</b> 130 133 135 140 141 151 154 156 157 163	
5	<b>Mas</b> <b>A</b> n 5.1 5.2 5.3 5.4	ss cond Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3 The fu 5.5.1 5.5.2 5.5.3	centration phenomena for fourth order operators.         narmonic Steklov problem         alating the problem         teklov spectrum         ann problem and behavior of Neumann eigenvalues unass concentration at the boundary         ass concentration of the eigenvalues.         Isovolumetric perturbatons         The Steklov problem         Isovolumetric perturbatons         The Neumann problem         Indamental tone of the ball.         An isoperimetric inequality         Eigenvalues and eigenfunctions on the ball         The isoperimetric inequality         Some remarks on the case $\tau = 0$	2 <b>29</b> 130 133 135 140 141 151 154 156 157 163 165	
5	Mas A n 5.1 5.2 5.3 5.4 5.5	ss cone Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3 The fu 5.5.1 5.5.2 5.5.3 Neum	centration phenomena for fourth order operators.         narmonic Steklov problem         alating the problem         teklov spectrum         ann problem and behavior of Neumann eigenvalues unass concentration at the boundary         ass concentration of the eigenvalues.         Isovolumetric perturbations         The Steklov problem         Isovolumetric perturbatons         Indamental tone of the ball.         An isoperimetric inequality         Eigenvalues and eigenfunctions on the ball         The isoperimetric inequality         Some remarks on the case $\tau = 0$ Intervalue	<b>29</b> 130 133 135 140 141 151 154 157 163 165	
5	<b>Mas</b> <b>A</b> n 5.1 5.2 5.3 5.4 5.5	ss cond Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3 The fu 5.5.1 5.5.2 5.5.3 Neum ness o	The unsule determination of the expansions $\cdots$ is the expansion of the expansio	<b>29</b> 130           133           135           140           141           151           156           157           163           165           171	
5	<b>Mas</b> <b>A</b> n 5.1 5.2 5.3 5.4 5.5 5.6	ss cond Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3 The fu 5.5.1 5.5.2 5.5.3 Neum ness o 5.6.1	The unstate determination of the expansions $\dots$ is the expansion of the expansi	<b>29</b> 130 133 135 140 141 151 154 156 157 163 165 171	
5	<b>Mas</b> <b>A</b> n 5.1 5.2 5.3 5.4 5.5 5.6	ss cond Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3 The fu 5.5.1 5.5.2 5.5.3 Neum ness o 5.6.1	Centration phenomena for fourth order operators.narmonic Steklov problemnalating the problemnan problem and behavior of Neumann eigenvalues unass concentration at the boundarynan problem and behavior of Neumann eigenvalues unass concentration at the boundarynetric functions of the eigenvalues. Isovolumetric pertionsThe Steklov problemThe Steklov problemIsovolumetric perturbatonsThe Neumann problemIndamental tone of the ball. An isoperimetric inequalityEigenvalues and eigenfunctions on the ballThe isoperimetric inequalitySome remarks on the case $\tau = 0$ Ann isoperimetric inequality in quantitative form. Sharp-f Neumann and Steklov inequalitiesQuantitative isoperimetric inequality for the Neumannproblem	<b>29</b> 130 133 135 140 141 151 154 157 163 165 171 173	
5	<b>Mas</b> <b>A</b> n 5.1 5.2 5.3 5.4 5.5 5.6	ss cone Formu The S Neum der m Symm turbat 5.4.1 5.4.2 5.4.3 The fu 5.5.1 5.5.2 5.5.3 Neum ness o 5.6.1	centration phenomena for fourth order operators.         narmonic Steklov problem         alating the problem         teklov spectrum         ann problem and behavior of Neumann eigenvalues unass concentration at the boundary         ass concentration of the eigenvalues. Isovolumetric pertric functions of the eigenvalues. Isovolumetric pertions         The Steklov problem         The Steklov problem         Isovolumetric perturbatons         Indamental tone of the ball. An isoperimetric inequality         Eigenvalues and eigenfunctions on the ball         The isoperimetric inequality         Some remarks on the case $\tau = 0$ ann isoperimetric inequality in quantitative form. Sharp-         f Neumann and Steklov inequalities         Sharpness of the Neumann inequality	<b>29</b> 130           133           135           140           141           151           157           163           165           171           173           177	

•	٠
XXVI	1

6	A few properties of the eigenvalues of Neumann-type prob- lems 191			
				191
	6.1	Neumann eigenvalues of the biharmonic operator 1		. 191
		6.1.1	Eigenvalues of Neumann and Dirichlet problems	. 193
		6.1.2	Dependence of the Neumann eigenvalues upon the Pois-	
			son's ratio	. 197
		6.1.3	Neumann and Dirichlet eigenvalues in the case of the	
			unit ball	199
	6.2	Neuma	ann eigenvalues on annuli	202
		6.2.1	Eigenvalues of the Laplace operator on the annulus of	
			$\mathbb{R}^2$	203
		6.2.2	Eigenvalues of the Laplace operator on the annulus of	
			$\mathbb{R}^{\bar{N}}$	209
		6.2.3	Eigenvalues of the biharmonic operator on the annulus	
			of $\mathbb{R}^N$	. 211
		6.2.4	Symbolic computations for the eigenvalues of the bi-	
			harmonic operator on the annulus of $\mathbb{R}^2$	213
Bi	bliog	graphy		215
$\mathbf{Ri}$	Ringraziamenti 2			225

xxviii

Contents

### Chapter 1

### **Preliminaries and notation**

In this chapter we set the notation and introduce certain preliminary results which will be used in the sequel.

#### 1.1 Sobolev Spaces

Let  $N \in \mathbb{N}$ . In the sequel we shall always consider  $N \geq 2$  unless otherwise indicated. For any set V in  $\mathbb{R}^N$  and  $\rho > 0$  we denote by  $V_\rho$  the set  $\{x \in V : d(x, \partial\Omega) > \rho\}$ . Here d(x, V) denotes the Euclidean distance from x to the set V. Moreover, by a cuboid we mean any roto-translation of a rectangular parallelepiped in  $\mathbb{R}^N$ .

**Definition 1.1.1.** Let  $\rho > 0$ ,  $s, s' \in \mathbb{N}$ ,  $s' \leq s$  and  $\{V_j\}_{j=1}^s$  be a family of bounded open cuboids and  $\{r_j\}_{j=1}^s$  be a family of isometries in  $\mathbb{R}^N$ . We say that  $\mathcal{A} = (\rho, s, s', \{V_j\}_{j=1}^s, \{r_j\}_{j=1}^s)$  is an atlas in  $\mathbb{R}^N$  with the parameters  $\rho, s, s', \{V_j\}_{j=1}^s, \{r_j\}_{j=1}^s$ , briefly an atlas in  $\mathbb{R}^N$ . We denote by  $C(\mathcal{A})$  the family of all open sets  $\Omega$  in  $\mathbb{R}^N$  satisfying the following properties:

(i) 
$$\Omega \subset \bigcup_{j=1}^{\circ} (V_j)_{\rho} \text{ and } (V_j)_{\rho} \cap \Omega \neq \emptyset;$$
  
(ii)  $V_j \cap \partial \Omega \neq \emptyset \text{ for } j = 1, \dots s', V_j \cap \partial \Omega = \emptyset \text{ for } s' < j \leq s;$   
(iii) for  $j = 1, \dots, s$ 

$$r_j(V_j) = \{ x \in \mathbb{R}^N : a_{ij} < x_i < b_{ij}, i = 1, ..., N \}$$

and

$$r_j(\Omega \cap V_j) = \{ x \in \mathbb{R}^N : a_{Nj} < x_N < g_j(\bar{x}), \ \bar{x} \in W_j \},\$$

where  $\bar{x} = (x_1, ..., x_{N-1}), W_j = \{\bar{x} \in \mathbb{R}^{N-1} : a_{ij} < x_i < b_{ij}, i = 1, ..., N-1\}$ and  $g_j$  is a continuous function defined on  $\overline{W}_j$  (it is meant that if  $s' < j \le s$ then  $g_j(\bar{x}) = b_{Nj}$  for all  $\bar{x} \in \overline{W}_j$ ); moreover for j = 1, ..., s'

$$a_{Nj} + \rho \le g_j(\bar{x}) \le b_{Nj} - \rho,$$

for all  $\bar{x} \in \overline{W}_i$ .

We say that an open set  $\Omega$  in  $\mathbb{R}^N$  is an open set with a continuous boundary if  $\Omega$  is of class  $C(\mathcal{A})$  for some atlas  $\mathcal{A}$ .

Let  $m \in \mathbb{N}, M > 0$ . We say that an open set  $\Omega$  is of class  $C_M^m(\mathcal{A})$  if  $\Omega$  is of class  $C(\mathcal{A})$  and all the functions  $g_j$  in (iii) are of class  $C^m(\overline{W}_j)$  with

$$|g_j|_{c^m(\overline{W}_j)} = \sum_{1 \le |\alpha| \le m} \|D^{\alpha}g_j\|_{L^{\infty}(\overline{W}_j)} \le M$$

We say that an open set  $\Omega$  in  $\mathbb{R}^N$  is an open set of class  $C^m$  if  $\Omega$  is of class  $C_M^m(\mathcal{A})$  for some atlas  $\mathcal{A}, m \in \mathbb{N}$  and M > 0.

Let  $N, m \in \mathbb{N}$ ,  $1 \leq p \leq \infty$  and  $\Omega$  be an open set in  $\mathbb{R}^N$ . Let  $C_c^{\infty}(\Omega)$  be the space of those functions in  $C^{\infty}(\Omega)$  which are compactly supported in  $\Omega$ . We recall the following definitions.

**Definition 1.1.2.** The Sobolev Space  $W^{m,p}(\Omega)$  consists of all (real valued) functions u in  $L^p(\Omega)$  with weak derivatives  $D^{\alpha}u$  in  $L^p(\Omega)$  for all  $|\alpha| \leq m$ .

We consider the space  $W^{m,p}(\Omega)$  endowed with the norm defined by

$$\begin{aligned} \|u\|_{W^{m,p}(\Omega)} &:= \sum_{|\alpha| \le m} \|D^{\alpha}u\|_{L^{p}(\Omega)}, \quad \text{if } \mathbf{p} \neq \infty, \\ \|u\|_{W^{m,\infty}(\Omega)} &:= \sum_{|\alpha| \le m} \|D^{\alpha}u\|_{L^{\infty}(\Omega)}. \end{aligned}$$

We recall the definition of  $W_0^{m,p}(\Omega)$ .

**Definition 1.1.3.** Let  $1 \leq p < \infty$ . We denote by  $W_0^{m,p}(\Omega)$  the closure of  $C_c^{\infty}(\Omega)$  in  $W^{m,p}(\Omega)$ .

For p = 2, we write  $H^m(\Omega) = W^{m,2}(\Omega)$ ,  $H_0^m(\Omega) = W_0^{m,2}(\Omega)$ . We recall the following results on the approximation of functions in  $W^{m,p}(\Omega)$ .

**Theorem 1.1.4.** (Global approximation by smooth functions). Let  $\Omega$  be an open set in  $\mathbb{R}^N$ . Let  $u \in W^{m,p}(\Omega)$  for some  $1 \leq p < \infty$ . Then there exists a sequence  $\{u_k\}_{k\in\mathbb{N}} \subset C^{\infty}(\Omega) \cap W^{m,p}(\Omega)$  converging to u in  $W^{m,p}(\Omega)$ .

**Theorem 1.1.5.** (Global approximation by smooth functions up to the boundary). Let  $\Omega$  be a bounded open set in  $\mathbb{R}^N$  of class  $C^1$ . Let  $u \in W^{m,p}(\Omega)$ for some  $1 \leq p < \infty$ . Then there exists a sequence  $\{u_k\}_{k \in \mathbb{N}} \subset C^{\infty}(\overline{\Omega})$ converging to u in  $W^{m,p}(\Omega)$ .

As a consequence of Theorem 1.1.4 we have the following

**Definition 1.1.6.** Let  $\Omega$  be an open set in  $\mathbb{R}^N$ ,  $1 \leq p \leq \infty$ ,  $u \in L^p(\Omega)$ . Then  $u \in W^{1,p}(\Omega)$  if and only if u coincides almost everywhere with a function  $\tilde{u}$  such that for almost all lines l parallel to the coordinate axis,  $u_{|_l}$  is locally absolutely continuous, and the classic derivatives  $\frac{\partial \tilde{u}}{\partial x_1}, \dots, \frac{\partial \tilde{u}}{\partial x_N}$ , which exist almost everywhere, belong to  $L^p(\Omega)$ .

Under suitable regularity assumptions on the open set  $\Omega$  it makes sense to define the trace of a function  $u \in W^{m,p}(\Omega)$  on  $\partial\Omega$ .

**Theorem 1.1.7.** (Trace). Let  $\Omega$  be a bounded open set in  $\mathbb{R}^N$  of class  $C^1$ . Then there exists a bounded linear operator Tr from  $W^{1,p}(\Omega)$  to  $L^p(\partial\Omega)$  such that:

- i)  $\operatorname{Tr}[u] = u_{|_{\partial\Omega}}$  if  $u \in W^{1,p}(\Omega) \cap C(\overline{\Omega})$ ;
- ii)  $\|\operatorname{Tr}[u]\|_{L^p(\partial\Omega)} \leq C \|u\|_{W^{1,p}(\Omega)}, \forall u \in W^{1,p}(\Omega), \text{ the constant } C \text{ depending only on } p \text{ and } \Omega.$

**Theorem 1.1.8.** Let  $\Omega$  be a bounded open set in  $\mathbb{R}^N$  of class  $C^1$ . Then  $u \in W_0^{1,p}(\Omega)$  if and only in  $\operatorname{Tr}[u] = 0$ .

The next results concern the embeddings of Sobolev Spaces.

**Theorem 1.1.9.** (Gagliardo-Nirenberg-Sobolev inequality). For  $1 \le p < N$  let the Sobolev exponent  $p^*$  be defined by  $p^* := \frac{Np}{N-p}$ . Then there exists C > 0, depending only on p and N, such that

$$||u||_{L^{p^*}(\mathbb{R}^N)} \le C ||\nabla u||_{L^p(\mathbb{R}^N)},$$

for all  $u \in W^{1,p}(\mathbb{R}^N)$ .

**Lemma 1.1.10.** (Poincaré inequality). Let  $\Omega$  be an open set in  $\mathbb{R}^N$  of finite measure,  $1 \leq p < \infty$ . Then there exists C > 0, depending only on p, N and  $\Omega$  such that

$$\left\|u\right\|_{L^{p}(\Omega)} \leq C \left\|\nabla u\right\|_{L^{p}(\Omega)},$$

for all  $u \in W_0^{1,p}(\Omega)$ .

**Theorem 1.1.11.** (Rellich-Kondrakhov). Let  $\Omega$  be a bounded open set in  $\mathbb{R}^N$  of class  $C^1$ ,  $1 \leq p < N$ . Then  $W^{1,p}(\Omega)$  is compactly embedded into  $L^q(\Omega)$  for all  $1 \leq q < p^*$ . If  $p \geq N$ , then  $W^{1,p}(\Omega)$  is compactly embedded into  $L^q(\Omega)$  for all  $q \geq 1$ .

**Corollary 1.1.12.** If  $\Omega$  is an open set of finite measure, then for all  $1 \leq p < \infty$ ,  $W_0^{1,p}(\Omega)$  is compactly embedded into  $L^p(\Omega)$ . If  $\Omega$  is a bounded open set of class  $C^0$ , then for all  $1 \leq p < \infty$ ,  $W^{1,p}(\Omega)$  is compactly embedded into  $L^p(\Omega)$ .

**Theorem 1.1.13.** (Poincaré-Wirtinger inequality). Let  $\Omega$  be a bounded open set in  $\mathbb{R}^N$  of class  $C^1$ ,  $1 \leq p < \infty$ . Then there exists C > 0, depending only on p, N and  $\Omega$  such that

$$\|u - (u)_{\Omega}\|_{L^{p}(\Omega)} \leq C \|\nabla u\|_{L^{p}(\Omega)},$$

where  $(u)_{\Omega} = \frac{\int_{\Omega} u}{|\Omega|}$ .

From Theorem 1.1.11 it follows that the trace operator is compact.

**Theorem 1.1.14.** (Compact trace). Let  $\Omega$  be an open bounded set in  $\mathbb{R}^N$  of class  $C^1$ ,  $1 \leq p < \infty$ . Then the trace operator  $\text{Tr} : W^{1,p}(\Omega) \to L^p(\partial\Omega)$  is compact.

**Remark 1.1.15.** We observe that Theorem 1.1.11 and Theorem 1.1.14 hold even under lower regularity assumptions on the boundary. In fact Theorem 1.1.11 holds e.g., for  $\Omega$  of class  $C^{0,1}$  (see [88, Theorem 6.1]).

For all the proofs of the results contained in this subsection we refer to [26].

# **1.2** Compact operators and symmetric functions of the eigenvalues

In this section we recall some results from [75] which will be used in the sequel. For all the proofs of the results contained in this subsection we refer to [75].

Let  $\mathcal{X}, \mathcal{Y}, \mathcal{Z}$  be real Banach spaces. Let  $\mathcal{L}(\mathcal{X}, \mathcal{Y})$  be the Banach space of bounded linear maps from  $\mathcal{X}$  to  $\mathcal{Y}$  endowed with the usual norm  $\|A\|_{\mathcal{L}(\mathcal{X},\mathcal{V})} :=$  $\sup_{\|x\|_{\mathcal{X}}=1} \|Ax\|_{\mathcal{Y}}$ . Let  $\mathcal{B}(\mathcal{X} \times \mathcal{Y}, \mathcal{Z})$  be the space of bilinear continuous maps from  $\mathcal{X} \times \mathcal{Y}$  to  $\mathcal{Z}$ , endowed with the usual norm of the uniform convergence on the product of the unit ball of  $\mathcal{X}$  and the one of  $\mathcal{Y}$ . Let  $(H, \langle \cdot, \cdot \rangle)$  be a real Hilbert space, and  $\|\cdot\|$  be the norm associated with a scalar product  $\langle \cdot, \cdot \rangle$  of H. We denote by  $H_Q$  the vector space H endowed with the scalar product  $Q = Q(\cdot, \cdot)$ , and by  $\|\cdot\|_Q$  the associated norm. We denote by  $\mathcal{K}(H, H)$  the subspace of  $\mathcal{L}(H, H)$  of compact operators, which is closed in  $\mathcal{L}(H, H)$ . We denote by  $\mathcal{K}_{\mathcal{S}}(H_Q, H_Q)$  the closed subspace of  $\mathcal{K}(H_Q, H_Q)$ of those T such that Q(Tu, u) = Q(u, Tu) for all  $u, v \in H_Q$ . Let T be a compact self-adjoint operator on H, and  $\sigma(T)$  be the spectrum of T, which is well-known to be a finite or countable subset of  $\mathbb{R}$ . The elements of  $\sigma(T) \setminus \{0\}$  are the eigenvalues of T, and 0 is the only possible accumulation point for  $\sigma(T)$ . For the characterization of the spectrum of a compact self-adjoint operator we refer to [18]. We denote by  $j^+(T)$  the number of positive eigenvalues of T, each counted according to its multiplicity, and by  $j^{-}(T)$  the number of negative eigenvalues of T, each counted according to its multiplicity. Following [75] we set

$$J^{+}(T) := \{ j \in \mathbb{Z} : 1 \le j \le j^{+}(T) \},\$$
  
$$J^{-}(T) := \{ j \in \mathbb{Z} : -j^{-}(T) \le j \le -1 \}.$$

Then there exists a unique function  $j \to \mu_j(T)$  of  $J(T) := J^+(T) \cup J^-(T)$ to  $\mathbb{R}$ , which is decreasing on  $J^-(T)$  and on  $J^+(T)$ , with

$$\sigma(T) \setminus \{0\} = \{\mu_j(T) : j \in J(T)\},\$$

and such that each eigenvalue is repeated according to its multiplicity. We set

$$\mathcal{B}_{\mathcal{S}}(H^2,\mathbb{R}) := \{ B \in \mathcal{B}(H^2,\mathbb{R}) : B(u_1, u_2) = B(u_2, u_1) \text{ for all } u_1, u_2 \in H \},\$$

which is a closed subspace of  $\mathcal{B}(H^2,\mathbb{R})$ , and

$$\mathcal{Q}(H^2,\mathbb{R}) := \{ B \in \mathcal{B}_{\mathcal{S}}(H^2,\mathbb{R}) : \eta[B] > 0 \},\$$

where

$$\eta[B] := \inf \left\{ rac{B(u,u)}{\left\| u 
ight\|^2} \, : \, u \in H \setminus \{0\} 
ight\}.$$

Note that the set  $\mathcal{Q}(H)$  is the set of those scalar products on H which are coercive with respect to the fixed scalar product  $\langle \cdot, \cdot \rangle$ . We observe that  $Q \in \mathcal{B}_{\mathcal{S}}(H^2, \mathbb{R})$  is a coercive scalar product if and only if the embedding of  $H_Q$  in H is a homeomorphism. Now we set

$$\mathcal{M} := \{ (Q, T) \in \mathcal{B}_{\mathcal{S}}(H^2, \mathbb{R}) \times \mathcal{K}(H, H) :$$
$$Q(Tu, v) = Q(u, Tv) \text{ for all } u, v \in H \}.$$

The set  $\mathcal{M}$  is closed in  $\mathcal{B}_{\mathcal{S}}(H^2, \mathbb{R}) \times \mathcal{K}(H, H)$ . Moreover, we set

$$\mathcal{O} := \mathcal{M} \cap (\mathcal{Q}(H^2, \mathbb{R}) \times \mathcal{K}(H, H))$$
$$= \{ (Q, T) \in \mathcal{Q}(H^2, \mathbb{R}) \times \mathcal{K}(H, H) : T \in \mathcal{K}_{\mathcal{S}}(H_Q, H_Q) \}.$$

The set  $\mathcal{O}$  is open in  $\mathcal{M}$ . We have the following theorem (see [75]).

**Theorem 1.2.1.** Let H be a real Hilbert space,  $j \in \mathbb{Z} \setminus \{0\}$ . Then the set

$$\mathcal{A}_j := \{ (Q, T) \in \mathcal{O} : j \in J(T) \}$$

is open in  $\mathcal{M}$ , and the function  $\mu_j[\cdot]$  which takes  $(Q,T) \in \mathcal{A}_j$  to  $\mu_j[T]$  is continuous.

We consider a fixed finite subset F of  $\mathbb{Z} \setminus \{0\}$ , and set

$$\mathcal{A}[F] := \{ (Q,T) \in \mathcal{O} : j \in J(T) \,\forall j \in F \,, \, \mu_l[T] \notin \{ \mu_j[T] : j \in F \} \\ \forall l \in J(T) \setminus F \}.$$

By Theorem 1.2.1 it follows that  $\mathcal{A}[F]$  is open in  $\mathcal{M}$  and  $\mu_j[\cdot]$  are continuous on  $\mathcal{A}[F]$ . We consider the orthogonal projection  $P_F[Q, T]$  of  $H_Q$  on the subsapce E[T, F] generated by

$$\{u \in H_Q : Tu = \mu u, \exists \mu \in \{\mu_j[T] : j \in F\}\}.$$

We have the following lemma.

**Lemma 1.2.2.** Let H be a real Hilbert space and F be a finite subset of  $\mathbb{Z} \setminus \{0\}$ . Then E[T, F] has dimension equal to the cardinality of F, and it is an invariant subspace of H for T.

We recall the following result (see [66]).

**Theorem 1.2.3.** Let H be a real Hilbert space, F be a finite subset of  $\mathbb{Z}\setminus\{0\}$ . Then the map  $P_F$  which takes  $(Q,T) \in \mathcal{A}[F]$  to  $P_F[Q,T] \in \mathcal{L}(H,H)$  is continuous.

The projection  $P_F[Q,T]$  depends analytically on (Q,T), in the sense of the following theorem.

**Theorem 1.2.4.** Let H be a real Hilbert space, F be a finite non-empty subset of  $\mathbb{Z}\setminus\{0\}$  and  $(\tilde{Q}, \tilde{T}) \in \mathcal{A}[F]$ . Then there exists an open neighbourhood  $\widetilde{W}$  of  $(\tilde{Q}, \tilde{T})$  in  $\mathcal{Q}(H^2, \mathbb{R}) \times \mathcal{L}(H, H)$ , and a real analytic operator  $P_F^{\sharp}$  of  $\widetilde{W}$ to  $\mathcal{L}(H, H)$  such that  $P_F^{\sharp}[Q, T] = P_F[Q, T]$  for all  $(Q, T) \in \widetilde{W} \cap \mathcal{A}[F]$ .

It is possible to choose a orthonormal basis of E[T, F] which depends analitically on (Q, T), as stated in the following lemma (see [75]).

**Lemma 1.2.5.** Let H be a real Hilbert space, F be a finite subset of  $\mathbb{Z} \setminus \{0\}$ and  $(\tilde{Q}, \tilde{T}) \in \mathcal{A}[F]$ . Let  $\{\tilde{u}_j : j \in F\}$  be an othonormal basis for  $E[\tilde{T}, F]$  in  $H_{\tilde{Q}}$ . Then there exists an open neighbourhood  $\mathcal{W}_0$  of  $(\tilde{Q}, \tilde{T})$  in  $\mathcal{Q}(H^2, \mathbb{R}) \times \mathcal{L}(H, H)$  which is contained in the neighbourhood  $\tilde{\mathcal{W}}$  of Theorem 1.2.4, and |F| real analytic operators  $u_j[\cdot, \cdot], j \in F$ , of  $\mathcal{W}_0$  to H such that:

- i)  $\{u_j[Q,T] : j \in F\}$  is an orthonormal set in  $H_Q$ , for all  $(Q,T) \in \mathcal{W}_0$ ,
- ii)  $\{u_j[Q,T] : j \in F\}$  is an orthonormal basis for the range of  $P_F^{\sharp}[Q,T]$ , which coincide with E[T,F], in  $H_Q$ , for all  $(Q,T) \in \mathcal{W}_0 \cap \mathcal{A}[F]$ ,
- *iii*)  $u_j[\tilde{Q}, \tilde{T}] = \tilde{u}_j$  for all  $j \in F$ .

We need also the following lemma.

**Lemma 1.2.6.** Let H be a real Hilbert space, F be a finite subset of  $\mathbb{Z} \setminus \{0\}$ and  $(\tilde{Q}, \tilde{T}) \in \mathcal{A}[F]$ . Let  $\{\tilde{u_1}, ..., \tilde{u_{|F|}}\}$  be an orthonormal basis of  $E[\tilde{T}, F]$  in  $H_{\tilde{Q}}$ , and  $\{u_j[Q, T] : j = 1, ..., |F|\}$  as in the previous lemma and S the map of  $\mathcal{W}_0$  to the set  $M_{|F|}(\mathbb{R})$  of  $|F| \times |F|$  matrices with real coefficients, defined by

$$\mathcal{S}[Q,T] = (\mathcal{S}_{hk}[Q,T])_{h,k=1,\dots,|F|} := (Q(Tu_k[Q,T],u_h[Q,T]))_{h,k=1,\dots,|F|},$$

for all  $(Q,T) \in W_0$ . Then  $S[\cdot, \cdot]$  is real analytic and S[Q,T] is symmetric for all  $(Q,T) \in W_0 \cap \mathcal{A}[F]$ . Moreover, if  $(Q,T) \in W_0 \cap \mathcal{A}[F]$ , then  $\{\mu_j[T]\}_{j \in F}$ are the eigenvalues of S[Q,T] repeated according to their multiplicity. Finally, if we assume that  $\mu_j[\tilde{T}]$  assume a common value  $\tilde{\mu}_j$  for all  $j \in F$ , then the differential of  $S[\cdot, \cdot]$  in  $(\tilde{Q}, \tilde{T})$  is given by the formula

$$d\mathcal{S}[\tilde{Q},\tilde{T}](\dot{Q},\dot{T}) = \left(\tilde{Q}(\dot{T}\tilde{u}_k,\tilde{u}_h)\right)_{h,k=1,\dots,|F|}, \text{ for all } (\dot{Q},\dot{T}) \in \mathcal{B}_{\mathcal{S}}(H^2,\mathbb{R}) \times \mathcal{L}(H,H).$$

Finally, as a consequence of Theorem 1.2.4, Lemma 1.2.5 and Lemma 1.2.6, we have the following theorem.

**Theorem 1.2.7.** Let H be a real Hilbert space and F be a finite non-empty subset of  $\mathbb{Z} \setminus \{0\}$ . Let

$$M_{F,s}[T] = \sum_{\substack{j_1, \dots, j_s \in F \\ j_1 < \dots < j_s}} \mu_{j_1}[T] \cdots \mu_{j_s}[T], \quad \forall s \in \{1, \dots, |F|\},$$

for all  $(Q,T) \in \mathcal{A}[F]$ , be the elementary symmetric functions of the eigenvalues  $\mu_j[T]$  with indexes  $j \in F$ . Let  $(\tilde{Q}, \tilde{T}) \in \mathcal{A}[F]$ . Then there exists an open neighbourhood  $\widetilde{\mathcal{W}}$  of  $(\tilde{Q}, \tilde{T})$  in  $\mathcal{Q}(H^2, \mathbb{R}) \times \mathcal{L}(H, H)$ , and real analytic functions  $M_{F,s}^{\sharp}[\cdot, \cdot]$ , for s = 1, ..., |F|, of  $\widetilde{\mathcal{W}}$  in  $\mathbb{R}$  such that

$$M_{F,s}^{\sharp}[Q,T] = M_{F,s}[T],$$

for all  $(Q, T) \in \widetilde{\mathcal{W}} \cap \mathcal{A}[F]$ , and for all s = 1, ..., |F|. If we further assume that there exists  $\tilde{\mu} \in \mathbb{R}$  such that  $\tilde{\mu} = \mu_j[T]$  for all  $j \in F$ , and if  $\{\tilde{u}_1, ..., \tilde{u}_{|F|}\}$  is an orthonormal basis for  $E[\tilde{T}, F]$  in  $H_{\tilde{Q}}$ , then the partial derivative of  $M_{F,s}^{\sharp}$ with respect to the variable T at  $(\tilde{Q}, \tilde{T})$  is given by the formula

$$d_T M_{F,s}^{\sharp}[\tilde{Q}, \tilde{T}](\dot{T}) = \binom{|F| - 1}{s - 1} \tilde{\mu}^{s - 1} \sum_{l=1}^{|F|} \tilde{Q}(\dot{T}\tilde{u}_l, \tilde{u}_l),$$

for all  $\dot{T} \in \mathcal{K}_{\mathcal{S}}(H_{\tilde{Q}}, H_{\tilde{Q}})$ , and for all s = 1, ..., |F|.

#### **1.3** Eigenvalues of elliptic operators

Let  $\Omega$  be an open set in  $\mathbb{R}^N$ ,  $m \in \mathbb{N}$  and  $V(\Omega)$  be a closed subspace of  $H^m(\Omega)$  containing  $H_0^m(\Omega)$  and such that the embedding  $V(\Omega) \subset L^2(\Omega)$  is compact. We shall assume that  $A_{\alpha\beta} \in L^{\infty}(\Omega)$  are fixed coefficients such that  $A_{\alpha\beta} = A_{\beta\alpha}$  for all  $\alpha, \beta \in \mathbb{N}_0^N$ , with  $|\alpha|, |\beta| \leq m$ . Let  $\rho \in L^{\infty}(\Omega)$  be such that essinf  $\Omega \rho > 0$ . We consider the following eigenvalue problem

$$\int_{\Omega} \sum_{0 \le |\alpha|, |\beta| \le m} A_{\alpha\beta} D^{\alpha} u D^{\beta} \varphi dx = \lambda \int_{\Omega} \rho u \varphi dx, \quad \forall \varphi \in V(\Omega), \qquad (1.3.1)$$

in the unknowns  $u \in V(\Omega)$  (the eigenfunction) and  $\lambda \in \mathbb{R}$  (the eigenvalue). It is convenient to denote the left-hand side of equation (1.3.1) by  $\mathcal{Q}[u,\varphi]$ . It is also convenient to denote by  $L^2_{\rho}(\Omega)$  the space  $L^2(\Omega)$  endowed with the scalar product defined by

$$\langle u_1, u_2 \rangle_{\rho} := \int_{\Omega} \rho u_1 u_2 dx, \ \forall u_1, u_2 \in L^2(\Omega).$$

Note that the corresponding norm  $||u||_{L^2_{\rho}(\Omega)}$  is equivalent to the standard norm. We assume that the space  $V(\Omega)$  and the coefficients  $A_{\alpha\beta}$  are such that Gårding's inequality holds, i.e., we assume that there exist a, b > 0 such that

$$a\|u\|_{H^m(\Omega)}^2 \le \mathcal{Q}[u, u] + b\|u\|_{L^2(\Omega)}^2, \qquad (1.3.2)$$

for all  $u \in V(\Omega)$ . In many cases it will be convenient to normalize the constants a, b in such a way that

$$a\|u\|_{H^m(\Omega)}^2 \le \mathcal{Q}[u,u] + b\|u\|_{L^2_{\rho}(\Omega)}^2.$$
(1.3.3)
For classical conditions on the coefficients  $A_{\alpha\beta}$  ensuring the validity of (1.3.2) in the case  $V(\Omega) = H_0^m(\Omega)$  we refer to [2, Theorem 7.6]. Moreover, we assume that there exists c > 0 such that

$$Q[u, u] \le c \|u\|_{H^m(\Omega)}^2,$$
 (1.3.4)

for all  $u \in V(\Omega)$ . Note that since the coefficients  $A_{\alpha\beta}$  are bounded, inequality (1.3.4) is always satisfied.

Under assumptions (1.3.3), (1.3.4), it is possible to prove that problem (1.3.1) has a divergent sequence of eigenvalues bounded from below by -b. For the sake of completeness, we recall here the standard procedure to recast problem (1.3.1) into an eigenvalue problem for a compact and self-adjoint operator on a Hilbert space. We consider the bounded linear operator Lfrom  $V(\Omega)$  to its dual  $V(\Omega)'$  which takes any  $u \in V(\Omega)$  to the functional L[u]defined by  $L[u][\varphi] := Q[u, \varphi]$ , for all  $\varphi \in V(\Omega)$ . Moreover, we consider the bounded linear operator  $I_{\rho}$  from  $L^2_{\rho}(\Omega)$  to  $V(\Omega)'$  which takes any  $u \in L^2_{\rho}(\Omega)$ to the functional  $I_{\rho}[u]$  defined by  $I_{\rho}[u][\varphi] := \langle u, \varphi \rangle_{\rho}$ , for all  $\varphi \in V(\Omega)$ . By inequalities (1.3.3), (1.3.4) and by the boundedness of the coefficients  $A_{\alpha\beta}$ , it follows that the quadratic form defined by the right-hand side of (1.3.3) induces in  $V(\Omega)$  a norm equivalent to the standard norm of  $H^m(\Omega)$ . Hence by the Riesz Theorem, it follows that the operator  $L + bI_{\rho}$  is a linear homeomorphism from  $V(\Omega)$  onto  $V(\Omega)'$ , where b is as in (1.3.3). Thus, equation (1.3.1) is equivalent to the equation

$$(L+bI_{\rho})^{(-1)} \circ I_{\rho}[u] = \mu u$$

where

$$\mu = (\lambda + b)^{-1}. \tag{1.3.5}$$

Thus, it is natural to consider the operator  $T_{\rho}$  from  $L^{2}_{\rho}(\Omega)$  to itself defined by

$$T_{\rho} := i \circ (L + bI_{\rho})^{(-1)} \circ I_{\rho}, \qquad (1.3.6)$$

where *i* is the embedding of  $V(\Omega)$  into  $L^2_{\rho}(\Omega)$ . In the sequel, we shall omit *i* and we shall simply write  $T_{\rho} = (L + bI_{\rho})^{(-1)} \circ I_{\rho}$ . Note that

$$< T_{\rho}u_{1}, u_{2} >_{\rho} = I_{\rho}[u_{2}][(L+bI_{\rho})^{(-1)} \circ I_{\rho}[u_{1}]]$$
  
=  $(L+bI_{\rho})[(L+bI_{\rho})^{(-1)} \circ I_{\rho}[u_{1}]][(L+bI_{\rho})^{(-1)} \circ I_{\rho}[u_{2}]],$ 

for all  $u_1, u_2 \in L^2_{\rho}(\Omega)$ . Thus, since the operator  $L+bI_{\rho}$  is symmetric it follows that  $T_{\rho}$  is a self-adjoint operator in  $L^2_{\rho}(\Omega)$ . Moreover, if the embedding  $V(\Omega) \subset L^2(\Omega)$  is compact then the operator  $T_{\rho}$  is compact. By inequality (1.3.3),  $T_{\rho}$  is injective. It follows that the spectrum of  $T_{\rho}$  is discrete and consists of a sequence of positive eigenvalues of finite multiplicity converging to zero. Then by (1.3.5) and standard spectral theory, we easily deduce the validity of the following theorem. **Theorem 1.3.7.** Let  $\rho \in L^{\infty}(\Omega)$  be such that  $\operatorname{ess\,inf}_{\Omega}\rho > 0$ . Assume that inequalities (1.3.3) and (1.3.4) are satisfied for some a, b, c > 0. Then the eigenvalues of equation (1.3.1) have finite multiplicity and can be represented by means of a divergent sequence  $\lambda_j, j \in \mathbb{N}$  as follows

$$\lambda_j = \min_{\substack{E \subset V(\Omega) \\ \dim E = j}} \max_{\substack{u \in E \\ u \neq 0}} \frac{\int_{\Omega} \sum_{|\alpha|, |\beta| \le m} A_{\alpha\beta} D^{\alpha} u D^{\beta} u dx}{\int_{\Omega} \rho u^2 dx} \,. \tag{1.3.8}$$

Each eigenvalue is repeated according to its multiplicity and

$$\lambda_j > -b + \frac{a}{\|\rho\|_{L^{\infty}(\Omega)}},$$

for all  $j \in \mathbb{N}$ . Moreover, the sequence  $\mu_j = (b + \lambda_j)^{-1}$ ,  $j \in \mathbb{N}$ , represents all eigenvalues of the compact self-adjoint operator  $T_{\rho}$ .

## 1.4 Bessel functions, modified Bessel functions and ultraspherical Bessel functions

In this section we recall some facts from the theory of Bessel functions and of spherical harmonics, which will be used in the sequel. For all the proofs of the results concerning the Bessel functions contained in this section and for more information on Bessel functions we refer to [1]. For all the proofs of the results concerning the spherical harmonics and for an introduction to the theory of spherical harmonics we refer to [42].

#### 1.4.1 Bessel functions

Consider the (complex) Bessel equation

$$z^{2}\frac{d^{2}w}{dz^{2}} + z\frac{dw}{dz} + (z^{2} - \nu^{2})w = 0, \ z \in \mathbb{C},$$
(1.4.1)

where  $\nu \in \mathbb{C}$ . As is well-known the solutions of this equation are given by the Bessel functions of the first kind  $J_{\pm\nu}(z)$  and of the second kind  $Y_{\nu}(z)$ . Such functions are holomorphic in the variable  $z \in \mathbb{C} \setminus \mathbb{R}_-$ . Functions  $J_{\nu}(z)$ and  $J_{-\nu}(z)$  are linearly independent except when  $\nu$  is an integer.  $J_{\nu}(z)$  and  $Y_{\nu}(z)$  are linearly independent for all  $\nu$ . We recall some useful relations:

$$Y_{\nu}(z) = \frac{J_{\nu}(z)\cos(\nu\pi) - J_{-\nu}(z)}{\sin(\pi\nu)},$$

$$J_{-n}(z) = (-1)^{n}J_{n}(z),$$

$$Y_{-n}(z) = (-1)^{n}Y_{n}(z),$$
(1.4.2)

where in the last two equations  $n \in \mathbb{N}$ . Note that if  $\nu = 0$  or if  $\nu$  is an integer, the right-hand side of the first equation is replaced by its limiting



value.

We recall some useful recurrence relations between solutions of (1.4.1):

$$C_{\nu-1}(z) + C_{\nu+1}(z) = \frac{2\nu}{z} C_{\nu}(z),$$
  

$$C_{\nu-1}(z) - C_{\nu+1}(z) = 2C'_{\nu}(z),$$
  

$$C'_{\nu}(z) = C_{\nu-1}(z) - \frac{\nu}{z} C_{\nu}(z),$$
  

$$C'_{\nu}(z) = -C_{\nu+1}(z) + \frac{\nu}{z} C_{\nu}(z),$$

where  $C_{\nu}$  denotes  $J_{\nu}$ ,  $Y_{\nu}$  or any linear combination of these functions. We also recall that

$$J'_0(z) = -J_1(z), \quad Y'_0(z) = -Y_1(z).$$

We have the following formulas for the derivatives:

$$\begin{aligned} \mathcal{C}_{\nu}^{(k)}(z) &= \frac{1}{2^{k}} \Big( \mathcal{C}_{\nu-k}(z) - \binom{k}{1} \mathcal{C}_{\nu-k+2}(z) \\ &+ \binom{k}{2} \mathcal{C}_{\nu-k+1}(z) - \ldots + (-1)^{k} \mathcal{C}_{\nu+k}(z) \Big), \end{aligned}$$

for  $k \in \mathbb{N}_0$ . We also recall the following Taylor expansion of  $J_{\nu}(z)$ :

$$J_{\nu}(z) = \left(\frac{z}{2}\right)^{\nu} \sum_{k=0}^{\infty} \frac{\left(-\frac{z^2}{4}\right)^k}{k! \Gamma(\nu+k+1)}.$$
 (1.4.3)

Note that from (1.4.2) and (1.4.3) it follows that  $Y_{\nu}(z)$  has a singularity at z = 0. In particular  $Y_0(z) = \frac{2}{\pi} \ln(z) + o(\ln(z)), \ Y_{\nu}(z) = -\frac{\Gamma(\nu)}{\pi} \left(\frac{z}{2}\right)^{-\nu} + o(z^{-\nu})$  when  $\operatorname{Re}(\nu) > 0$ , as z goes to zero. We have the following formulas for the Wronskians:

$$W(J_{\nu}(z), J_{-\nu}(z)) := J_{\nu+1}(z)J_{-\nu}(z) + J_{\nu}(z)J_{-(\nu+1)}(z) = -\frac{2\sin(\nu\pi)}{\pi z},$$
  
$$W(J_{\nu}(z), Y_{\nu}(z)) := J_{\nu+1}(z)Y_{\nu}(z) - J_{\nu}(z)Y_{\nu+1}(z) = \frac{2}{\pi z}.$$

We recall some useful facts on the zeros of cross products of Bessel functions. Let  $\nu \in \mathbb{R}$  and  $\lambda > 0$ . The zeros of the function which takes  $z \in \mathbb{C}$  to

$$J_{\nu}(z)Y_{\nu}(\lambda z) - J_{\nu}(\lambda z)Y_{\nu}(z)$$

are real and simple, in the sense that the derivative of the cross product with respect to z does not vanish at the zeroes. Moreover, there exist countably many zeroes of  $J_{\nu}(z)Y_{\nu}(\lambda z) - J_{\nu}(\lambda z)Y_{\nu}(z)$ . If  $\lambda > 1$ , the asymptotic expansion of the *n*-th zero with respect to  $\lambda$  near  $\lambda = 1$  is given by

$$\beta + \frac{p}{\beta} + \frac{q - p^2}{\beta^3} + \frac{r - 4pq + 2p^3}{\beta^5} + O(\lambda - 1)^5, \qquad (1.4.4)$$

where

$$\begin{split} \beta &= \frac{n\pi}{\lambda - 1}, \\ p &= \frac{\mu - 1}{8\lambda}, \\ q &= \frac{(\mu - 1)(\mu - 25)(\lambda^3 - 1)}{6(4\lambda)^3(\lambda - 1)}, \\ r &= \frac{(\mu - 1)(\mu^2 - 114\mu + 1073)(\lambda^5 - 1)}{5(4\lambda)^5(\lambda - 1)}, \end{split}$$

with  $\mu = 4\nu^2$ . The asymptotic expansion of the large positive zeros of the function which takes  $z \in \mathbb{C}$  to

$$J_{\nu}'(z)Y_{\nu}'(\lambda z) - J_{\nu}'(\lambda z)Y_{\nu}'(z)$$

is given by (1.4.4) with the same  $\beta$  and

$$p = \frac{\mu + 3}{8\lambda},$$

$$q = \frac{(\mu^2 + 46\mu - 63)(\lambda^3 - 1)}{6(4\lambda)^3(\lambda - 1)},$$

$$r = \frac{(\mu^3 + 185\mu^2 - 2053\mu + 1899)(\lambda^5 - 1)}{5(4\lambda)^5(\lambda - 1)}.$$

The asymptotic expansion of the large positive zeros of the function which takes  $z\in\mathbb{C}$  to

$$J_{\nu}'(z)Y_{\nu}(\lambda z) - J_{\nu}(\lambda z)Y_{\nu}'(z)$$

is given by (1.4.4) with

$$\begin{split} \beta &= \frac{\left(n - \frac{1}{2}\right)\pi}{\lambda - 1}, \\ p &= \frac{\left(\mu + 3\right)\lambda - \left(\mu - 1\right)}{8\lambda(\lambda - 1)}, \\ q &= \frac{\left(\mu^2 + 46\mu - 63\right)\lambda^3 - \left(\mu - 1\right)(\mu - 25)}{6(4\lambda)^3(\lambda - 1)}, \\ r &= \frac{\left(\mu^3 + 185\mu^2 - 2053\mu + 1899\right)\lambda^5 - \left(\mu - 1\right)(\mu^2 - 114\mu + 1073)}{5(4\lambda)^5(\lambda - 1)}. \end{split}$$

#### 1.4.2 Modified Bessel function

Consider the modified Bessel equation

$$z^2 \frac{d^2 w}{dz^2} + z \frac{dw}{dz} - (z^2 + \nu^2)w = 0, \ z \in \mathbb{C}.$$

The solutions of this equation are given by the modified Bessel functions of the first kind  $I_{\pm\nu}(z)$  and of the second kind  $K_{\nu}(z)$ . Such functions are holomorphic in the variable  $z \in \mathbb{C} \setminus \mathbb{R}_-$ .  $I_{\nu}(z)$  and  $I_{-\nu}(z)$  are linearly independent except when  $\nu$  is an integer.  $I_{\nu}(z)$  and  $K_{\nu}(z)$  are linearly independent for all  $\nu$ . The functions  $I_{\nu}(z)$  and  $K_{\nu}(z)$  are real and positive when  $\nu > -1$  and z > 0. We recall some useful relations:

$$K_{\nu}(z) = \frac{\pi}{2} \frac{I_{-\nu}(z) - I_{\nu}(z)}{\sin(\pi\nu)}, \qquad (1.4.5)$$

$$I_{-n}(z) = I_n(z), (1.4.6) K_{-\nu}(z) = K_{\nu}(z),$$

where in (1.4.6) the index n is an integer number.



We recall some other useful recurrence relations:

$$\begin{aligned} \mathcal{C}_{\nu-1}(z) - \mathcal{C}_{\nu+1}(z) &= \frac{2\nu}{z} \mathcal{C}_{\nu}(z), \\ \mathcal{C}_{\nu-1}(z) + \mathcal{C}_{\nu+1}(z) &= 2\mathcal{C}_{\nu}'(z), \\ \mathcal{C}_{\nu}'(z) &= \mathcal{C}_{\nu-1}(z) - \frac{\nu}{z} \mathcal{C}_{\nu}(z), \\ \mathcal{C}_{\nu}'(z) &= -\mathcal{C}_{\nu+1}(z) + \frac{\nu}{z} \mathcal{C}_{\nu}(z), \end{aligned}$$

where  $C_{\nu}$  denotes here any of the functions  $I_{\nu}$ ,  $K_{\nu}$  or any linear combination of them. We also recall that

$$I'_0(z) = I_1(z), \quad K'_0(z) = -K_1(z).$$

We have the following formulas for the derivatives:

$$\begin{aligned} \mathcal{C}_{\nu}^{(k)}(z) &= \frac{1}{2^{k}} \Big( \mathcal{C}_{\nu-k}(z) + \binom{k}{1} \mathcal{C}_{\nu-k+2}(z) \\ &+ \binom{k}{2} \mathcal{C}_{\nu-k+1}(z) + \ldots + \mathcal{C}_{\nu+k}(z) \Big), \end{aligned}$$

for  $k \in \mathbb{N}_0$ . We have the following Taylor expansion of  $I_{\nu}(z)$ :

$$I_{\nu}(z) = \left(\frac{z}{2}\right)^{\nu} \sum_{k=0}^{\infty} \frac{\left(\frac{z^2}{4}\right)^k}{k! \Gamma(\nu+k+1)}.$$
 (1.4.7)

Note that from (1.4.5) and (1.4.7) it follows that  $K_{\nu}(z)$  has a singularity at z = 0, in particular  $K_0(z) = -\ln(z) + o(\ln(z))$ ,  $K_{\nu}(z) = \frac{\Gamma(\nu)}{2} \left(\frac{z}{2}\right)^{-\nu} + o(z^{-\nu})$  when  $\operatorname{Re}(\nu) > 0$ , as z goes to zero. We have the following formulas for the Wronskians:

$$W(I_{\nu}(z), I_{-\nu}(z)) := I_{\nu}(z)I_{-(\nu+1)}(z) - I_{\nu+1}(z)I_{-\nu}(z) = -\frac{2\sin(\nu\pi)}{\pi z},$$
  
$$W(K_{\nu}(z), I_{\nu}(z)) := I_{\nu}(z)K_{\nu+1}(z) + I_{\nu+1}(z)K_{\nu}(z) = \frac{1}{z}.$$

#### 1.4.3 Ultraspherical Bessel and modified Bessel functions

We recall the definitions of ultraspherical Bessel functions and modified ultraspherical Bessel functions. Consider the ultraspherical Bessel equation

$$z^{2}\frac{d^{2}w}{dz^{2}} + (N-1)z\frac{dw}{dz} + \left(z^{2} - l(l+N-2)\right)w = 0, \ z \in \mathbb{C},$$
(1.4.8)

and the modified ultraspherical Bessel equation

$$z^{2}\frac{d^{2}w}{dz^{2}} + (N-1)z\frac{dw}{dz} - \left(z^{2} + l(l+N-2)\right)w = 0, \ z \in \mathbb{C},$$
(1.4.9)

for  $l \in \mathbb{N}_0$ . The ultraspherical Bessel functions of the first and second kind  $j_l(z)$  and  $y_l(z)$  are suitable linearly independent solutions of (1.4.8). The modified ultraspherical Bessel functions of the first and second kind  $i_l(z)$  and  $k_l(z)$  are suitable linearly independent solutions of (1.4.9). Namely, the functions  $j_l(z), y_l(z), i_l(z)$  and  $k_l(z)$  are defined in terms of the Bessel functions  $J_{\nu}(z), Y_{\nu}(z), I_{\nu}(z), K_{\nu}(z)$ , as follows:

$$\begin{aligned} j_l(z) &:= z^{-\frac{N-2}{2}} J_{\frac{N-2}{2}+l}(z), \\ y_l(z) &:= z^{-\frac{N-2}{2}} Y_{\frac{N-2}{2}+l}(z), \\ i_l(z) &:= z^{-\frac{N-2}{2}} I_{\frac{N-2}{2}+l}(z), \\ k_l(z) &:= z^{-\frac{N-2}{2}} K_{\frac{N-2}{2}+l}(z). \end{aligned}$$

The ultraspherical and modified ultraspherical Bessel functions have a number of recurrence relations which are inherited from those of the ordinary Bessel functions. We recall some of these relations for  $j_l(z)$  and  $i_l(z)$ :

$$\begin{aligned} \frac{N-2+2l}{z}j_{l}(z) &= j_{l-1}(z)+j_{l+1}(z),\\ j_{l}'(z) &= \frac{l}{z}j_{l}(z)-j_{l+1}(z)=j_{l-1}(z)-\frac{l+N-2}{z}j_{l}(z),\\ \frac{N-2+2l}{z}i_{l}(z) &= i_{l-1}(z)-i_{l+1}(z),\\ i_{l}'(z) &= \frac{l}{z}i_{l}(z)+j_{l+1}(z). \end{aligned}$$

Note that for N = 2 the expressions above simplify to the corresponding relations for the ordinary Bessel functions. We recall some recurrence relations for the second derivatives:

$$j_l''(z) = \left(\frac{l^2 - l}{z^2} - 1\right) j_l(z) + \frac{N - 1}{z} j_{l+1}(z),$$
  
$$i_l''(z) = \left(\frac{l^2 - l}{z^2} + 1\right) i_l(z) - \frac{N - 1}{z} j_{l+1}(z).$$

Again, when N = 2 each relation simplifies to the analogue for the ordinary Bessel functions. We have the following expansions of  $j_l(z)$  and  $i_l(z)$ :

$$j_{l}(z) = \sum_{k=0}^{\infty} \frac{(-1)^{k} 2^{1-\frac{N}{2}}}{k! \Gamma(k+\frac{N}{2}+l)} \left(\frac{z}{2}\right)^{2k+l},$$
  

$$i_{l}(z) = \sum_{k=0}^{\infty} \frac{2^{1-\frac{N}{2}}}{k! \Gamma(k+\frac{N}{2}+l)} \left(\frac{z}{2}\right)^{2k+l}.$$
(1.4.10)

From (1.4.10) it follows that  $i_l(z)$  and its derivatives are all positive on  $]0, +\infty[$ .

#### 1.4.4 Spherical harmonics

Let *B* be the unit ball in  $\mathbb{R}^N$  centered at zero. We denote by  $\partial B$  the boundary of *B*, i.e., the unit sphere in  $\mathbb{R}^N$ . Let  $k \in \mathbb{N}_0$  and  $\mathcal{P}_k$  be the space of homogeneous polynomials of degree *k* on  $\mathbb{R}^N$  and let

$$\begin{aligned} \mathscr{H}_k &:= & \left\{ P \in \mathcal{P}_k : \Delta P = 0 \right\}, \\ \mathscr{S}_k &:= & \left\{ P_{|_{\partial B}} : P \in \mathscr{H}_k \right\}. \end{aligned}$$

The set  $\mathscr{H}_k$  is the space of homogeneous harmonic polynomials of degree k and  $\mathscr{S}_k$  is the space of their restriction on the unit sphere. The elements of  $\mathscr{S}_k$  are the so-called spherical harmonics of degree k. We denote by  $r^2$  the quantity  $r^2 = |x|^2 = \sum_{j=1}^N x_j^2$ , where  $x = (x_1, x_2, ..., x_N)$  is a point in  $\mathbb{R}^N$ . We have the following lemma.

**Lemma 1.4.11.** Let  $k \in \mathbb{N}_0$ . Then

$$\mathcal{P}_k = \mathscr{H}_k \oplus r^2 \mathcal{P}_{k-2},$$

where  $r^2 \mathcal{P}_{k-2} = \left\{ r^2 P : P \in \mathcal{P}_{k-2} \right\}.$ 

Corollary 1.4.12.

$$\mathcal{P}_k = \bigoplus_{j=0}^k r^j \mathscr{H}_{k-j}$$

**Corollary 1.4.13.** The restriction to the unit sphere of any element of  $\mathcal{P}_k$  is a sum of spherical harmonics of degree at most k.

Let  $L^2(\partial B)$  the Hilbert space of the square integrable functions on  $\partial B$ with respect to the N-1 dimensional Haudorff measure of  $\partial B$ , endowed with the standard scalar product

$$< u, v >_{L^2(\partial B)} := \int_{\partial B} uv d\sigma,$$

for all  $u, v \in L^2(\partial B)$ . Here  $d\sigma$  denotes the surface measure of  $\partial B$ . We have the following theorem on the representation of functions in  $L^2(\partial B)$  in terms of spherical harmonics.

Theorem 1.4.14. We have

$$L^2(\partial B) = \bigoplus_{k=0}^{\infty} \mathscr{S}_k,$$

the expression on the right-hand side being an orthogonal direct sum with respect to the standard scalar product on  $L^2(\partial B)$ . Let

$$d_k := \dim \mathscr{S}_k = \dim \mathscr{H}_k.$$

We have an explicit formula for  $d_k$ .

**Lemma 1.4.15.** We have for  $\mathcal{P}_k$ 

$$\dim \mathcal{P}_k = \frac{(k+N-1)!}{k!(N-1)!}.$$

Corollary 1.4.16. We have for  $\mathscr{S}_k$ 

$$d_k = (2k + N - 2) \frac{(k + N - 3)!}{k!(N - 2)!}.$$

Let us denote by  $\Delta_S$  the Laplace-Beltrami operator on the unit sphere  $\partial B$  in  $\mathbb{R}^N$ . We have the following lemma.

Lemma 1.4.17. The solutions of

$$-\Delta_S \phi = k(k+N-2)\phi$$
, on  $\partial B$ ,

are the spherical harmonics of order k.

Preliminaries and notation

# Chapter 2

# Elliptic operators subject to mass density perturbations and maximum principles

In this chapter we discuss eigenvalue problems for general elliptic operators of arbitrary order subject to different homogeneous boundary conditions on open subsets of  $\mathbb{R}^N$ . The class of operators and boundary conditions which we consider is quite general and contains, for example, all the poly-harmonic operators subject to Dirichlet, intermediate, Neumann or mixed boundary conditions.

Let  $\Omega$  be an open subset of  $\mathbb{R}^N$  and  $m \in \mathbb{N}$ . We consider the elliptic partial differential operator  $\mathcal{L}$  defined by

$$\mathcal{L}u := \sum_{0 \le |\alpha|, |\beta| \le m} (-1)^{\alpha} D^{\alpha} \left( A_{\alpha\beta} D^{\beta} u \right)$$
(2.0.1)

subject to homogeneous boundary conditions. By  $\mathcal{R}$  we denote the subset of  $L^{\infty}(\Omega)$  of those  $\rho \in L^{\infty}(\Omega)$  such that  $\operatorname{ess\,inf}_{\Omega}\rho > 0$ . Let  $\rho \in \mathcal{R}$  be fixed. We consider the eigenvalue problem

$$\mathcal{L}u = \lambda \rho u. \tag{2.0.2}$$

The weak formulation of problem (2.0.2) is

$$\int_{\Omega} \sum_{0 \le |\alpha|, |\beta| \le m} A_{\alpha\beta} D^{\alpha} u D^{\beta} \varphi dx = \lambda \int_{\Omega} \rho u \varphi dx, \quad \forall \varphi \in V(\Omega), \qquad (2.0.3)$$

in the unknowns  $u \in V(\Omega)$  (the eigenfunction) and  $\lambda \in \mathbb{R}$  (the eigenvalue), where  $V(\Omega) \subset H^m(\Omega)$  is the energy space associated with the boundary conditions imposed on  $\mathcal{L}$ . We assume that the coefficients  $A_{\alpha\beta}$  are bounded real-valued functions such that  $A_{\alpha\beta} = A_{\beta\alpha}$ . Moreover, we assume that the space  $V(\Omega)$  and the coefficients  $A_{\alpha\beta}$  are such that inequalities (1.3.3) and (1.3.4) hold. If the embedding  $V(\Omega) \subset L^2(\Omega)$  is compact, from Theorem 1.3.7 it follows that problem (2.0.3) admits a diverging sequence of eigenvalues of finite multiplicity

$$\lambda_1[\rho] \leq \cdots \leq \lambda_j[\rho] \leq \cdots$$
.

We prove a few results concerning the dependence of the eigenvalues  $\lambda_j[\rho]$  upon variation of  $\rho$ .

### 2.1 Continuity of the eigenvalues

By the min-max principle (1.3.8) it follows that  $\lambda_j[\rho]$  is a locally Lipschitz continuous functions of  $\rho \in \mathcal{R}$ . In fact, one can easily prove that

$$|\lambda_j[\rho_1] - \lambda_j[\rho_2]| \le \frac{\min\{\lambda_j[\rho_1], \lambda_j[\rho_2]\} + 2b}{\min\{\operatorname{ess\,inf}\rho_1, \operatorname{ess\,inf}\rho_2\}} \|\rho_1 - \rho_2\|_{L^{\infty}(\Omega)}, \qquad (2.1.1)$$

for all  $\rho_1, \rho_2 \in \mathcal{R}$  satisfying  $\|\rho_1 - \rho_2\|_{L^{\infty}(\Omega)} < \min\{\operatorname{ess\,inf} \rho_1, \operatorname{ess\,inf} \rho_2\}$ . Indeed, for  $u \in V(\Omega)$  and  $\rho_1, \rho_2 \in \mathcal{R}$  we have

$$\left|\frac{\mathcal{Q}[u,u]}{\int_{\Omega}\rho_{1}u^{2}dx} - \frac{\mathcal{Q}[u,u]}{\int_{\Omega}\rho_{2}u^{2}dx}\right| \leq \frac{|\mathcal{Q}[u,u]| \left|\int_{\Omega}(\rho_{2}-\rho_{1})u^{2}dx\right|}{\left(\int_{\Omega}\rho_{1}u^{2}dx\right)\left(\int_{\Omega}\rho_{2}u^{2}dx\right)}$$

$$\leq \frac{|\mathcal{Q}[u,u]| \int_{\Omega}u^{2}dx \|\rho_{2}-\rho_{1}\|_{L^{\infty}(\Omega)}}{\left(\int_{\Omega}\rho_{1}u^{2}dx\right)\left(\int_{\Omega}\rho_{2}u^{2}dx\right)} \leq \frac{|\mathcal{Q}[u,u]| \|\rho_{2}-\rho_{1}\|_{L^{\infty}(\Omega)}}{\left(\int_{\Omega}\rho_{1}u^{2}dx\right)\operatorname{ess\,inf}\rho_{2}}$$

$$= \frac{|\mathcal{Q}[u,u]+b\int_{\Omega}\rho_{1}u^{2}dx-b\int_{\Omega}\rho_{1}u^{2}dx| \|\rho_{2}-\rho_{1}\|_{L^{\infty}(\Omega)}}{\left(\int_{\Omega}\rho_{1}u^{2}dx\right)\operatorname{ess\,inf}\rho_{2}}$$

$$\leq \left(\frac{\mathcal{Q}[u,u]}{\int_{\Omega}\rho_{1}u^{2}dx}+2b\right)\frac{\|\rho_{2}-\rho_{1}\|_{L^{\infty}(\Omega)}}{\operatorname{ess\,inf}\rho_{2}}.$$
 (2.1.2)

From (2.1.2) it follows that

$$\frac{\mathcal{Q}[u,u]}{\int_{\Omega} \rho_1 u^2 dx} \left( 1 - \frac{\|\rho_2 - \rho_1\|_{L^{\infty}(\Omega)}}{\operatorname{ess\,inf} \rho_2} \right) - \frac{2b\|\rho_2 - \rho_1\|_{L^{\infty}(\Omega)}}{\operatorname{ess\,inf} \rho_2} \\
\leq \frac{\mathcal{Q}[u,u]}{\int_{\Omega} \rho_2 u^2 dx} \\
\leq \frac{\mathcal{Q}[u,u]}{\int_{\Omega} \rho_1 u^2 dx} \left( 1 + \frac{\|\rho_2 - \rho_1\|_{L^{\infty}(\Omega)}}{\operatorname{ess\,inf} \rho_2} \right) + \frac{2b\|\rho_2 - \rho_1\|_{L^{\infty}(\Omega)}}{\operatorname{ess\,inf} \rho_2}. \quad (2.1.3)$$

If  $\rho_1, \rho_2$  satisfy  $\|\rho_2 - \rho_1\|_{L^{\infty}(\Omega)} < \text{ess inf } \rho_2$ , then taking the infimum and the supremum in (2.1.3) yields

$$|\lambda_j[\rho_1] - \lambda_j[\rho_2]| \le \frac{\lambda_j[\rho_1] + 2b}{\operatorname{ess\,inf}\,\rho_2} \|\rho_1 - \rho_2\|_{L^{\infty}(\Omega)}.$$
 (2.1.4)

It is easy to see that formula (2.1.4) still holds if in the right-hand side we replace  $\lambda_j[\rho_1]$  with  $\lambda_j[\rho_2]$  and ess inf  $\rho_2$  with ess inf  $\rho_1$ . This proves formula (2.1.1).

Actually,  $\lambda_j[\rho]$  depends with continuity on  $\rho$  not only with respect to the strong topology of  $L^{\infty}(\Omega)$  but also with respect to the weak\* topology, which is clearly more relevant in optimization problems. The following theorem was proved by Cox and McLaughlin [33] in the case of the Dirichlet Laplacian and mass densities uniformly bounded away from zero and infinity. The proof can be easily adapted to the general case. Moreover, it is possible to replace the uniform lower bound for  $\rho$  by a weaker assumption.

**Theorem 2.1.5.** Let  $C \subset \mathcal{R}$  be a bounded set. Assume that there exist a, b, c > 0 such that inequalities (1.3.3) and (1.3.4) are satisfied for all  $\rho \in C$ . Then the functions from C to  $\mathbb{R}$  which take any  $\rho \in C$  to  $\lambda_j[\rho]$  are weakly\* continuous for all  $j \in \mathbb{N}$ .

Proof. Since C is bounded in  $L^{\infty}(\Omega)$ , it suffices to prove that given  $\rho \in C$ and a sequence  $\rho_k \in C$ ,  $k \in \mathbb{N}$  such that  $\rho_k \rightharpoonup^* \rho$  as  $k \to \infty$  then  $\lambda_j[\rho_k] \to \lambda_j[\rho]$ . To do so, we first prove that for each  $j \in \mathbb{N}$  there exists  $L_j > 0$  such that  $\lambda_j[\rho_k] \leq L_j$  for all  $k \in \mathbb{N}$ . This is clearly trivial if we assume that  $0 < \alpha \leq \rho$  for all  $\rho \in C$ , in which case  $\lambda_j[\rho] \leq \lambda_j[\alpha]$ . Let  $j \in \mathbb{N}$  be fixed and  $u_1, \ldots, u_j \in V(\Omega)$  be linearly independent eigenfunctions associated with the eigenvalues  $\lambda_1[\rho], \ldots, \lambda_j[\rho]$ , normalized by  $\langle u_r, u_s \rangle_{\rho} = \delta_{rs}$  for all  $r, s = 1, \ldots, j$ . Note that

$$\lim_{k \to \infty} \int_{\Omega} u_r u_s \rho_k dx = \int_{\Omega} u_r u_s \rho dx,$$

for all  $r, s = 1, \ldots, j$ . Thus

$$\lim_{k \to \infty} \int_{\Omega} \left( \sum_{r=1}^{j} \gamma_r u_r \right)^2 \rho_k dx = \int_{\Omega} \left( \sum_{r=1}^{j} \gamma_r u_r \right)^2 \rho dx, \qquad (2.1.6)$$

uniformly with respect to  $\gamma = (\gamma_1, \ldots, \gamma_j) \in \mathbb{R}^j$  with  $|\gamma| \leq 1$ . Let *E* be the linear space generated by  $u_1, \ldots, u_j$ . By (2.1.6) it follows that for any  $\varepsilon > 0$  there exists  $k_{\varepsilon} \in \mathbb{N}$  such that

$$\frac{\int_{\Omega} \sum_{|\alpha|,|\beta| \le m} A_{\alpha\beta} D^{\alpha} u D^{\beta} u dx}{\int_{\Omega} u^{2} \rho_{k} dx} \le \frac{\int_{\Omega} \sum_{|\alpha|,|\beta| \le m} A_{\alpha\beta} D^{\alpha} u D^{\beta} u dx}{\int_{\Omega} u^{2} \rho dx} + \varepsilon (\lambda_{j}[\rho] + 2b) \le \lambda_{j}[\rho] + \varepsilon (\lambda_{j}[\rho] + 2b)$$
(2.1.7)

for all  $u \in E$ ,  $k \geq k_{\varepsilon}$ . By combining (1.3.8) and (2.1.7) we deduce that  $\lambda_j[\rho_k] \leq \lambda_j[\rho] + \varepsilon(\lambda_j[\rho] + 2b)$  for all  $k \geq k_{\varepsilon}$ , which implies the existence of a uniform bound  $L_j$  as claimed above. The rest of the proof follows the lines of Cox [33]. Let  $u_j[\rho_k]$ ,  $j \in \mathbb{N}$  be a sequence of eigenfunctions associated

with the eigenvalues  $\lambda_j[\rho_k]$  normalized by  $\langle u_j[\rho_k], u_l[\rho_k] \rangle_{\rho_k} = \delta_{jl}$  for all  $j, l \in \mathbb{N}$ . Note that  $\mathcal{Q}[u_j[\rho_k], u_j[\rho_k]] = \lambda_j[\rho_k]$  for all  $k \in \mathbb{N}$ , where we denoted by  $\mathcal{Q}[u, \varphi]$  the left-hand side of (2.0.3). By inequality (1.3.3), the sequence  $u_j[\rho_k], k \in \mathbb{N}$  is bounded in the space  $V(\Omega)$  equipped with the standard norm of  $H^m(\Omega)$ . It follows that possibly passing to subsequences, there exists  $\bar{u}_j \in V(\Omega)$  such that  $u_j[\rho_k]$  weakly converges to  $\bar{u}_j$  as  $k \to \infty$  in  $V(\Omega)$ , and there exists  $\bar{\lambda}_j \in \mathbb{R}$  such  $\lambda_j[\rho_k]$  converges to  $\bar{\lambda}_j$  as  $k \to \infty$ . Moreover, since the embedding  $V(\Omega) \subset L^2(\Omega)$  is compact we can directly assume that  $u_j[\rho_k]$  converges to  $\bar{u}_j$  strongly in  $L^2(\Omega)$  as  $k \to \infty$ . By passing to the limit in the weak equation

$$\mathcal{Q}[u_j[\rho_k], \varphi] = \lambda_j[\rho_k] < u_j[\rho_k], \varphi >_{\rho_k}, \quad \forall \ \varphi \in V(\Omega) \,,$$

it follows that  $\lambda_j$  is an eigenvalue and of problem (2.0.3) and  $\bar{u}_j$  a corresponding eigenfunction. Note that  $\langle \bar{u}_j, \bar{u}_l \rangle_{\rho} = \delta_{jl}$  for all  $j, l \in \mathbb{N}$ , hence  $\lambda_j, j \in \mathbb{N}$  is a divergent sequence. It remains to prove that  $\bar{\lambda}_j = \lambda_j[\rho]$  for all  $j \in \mathbb{N}$ . To do so, assume by contradiction that there exists an eigenfunction  $\bar{u} \in V(\Omega)$  associated with an eigenvalue  $\bar{\lambda}$  of the weak problem (2.0.3) such that  $\langle \bar{u}, \bar{u}_j \rangle_{\rho} = 0$  for all  $j \in \mathbb{N}$ . Assume that  $\bar{u}$  is normalized by  $\|\bar{u}\|_{\rho} = 1/(b + \bar{\lambda})$ . By the Auchmuty principle [13] applied to the operator  $L + bI_{\rho}$ , where  $L + bI_{\rho}$  has been defined in Section 1.3, we have

$$-\frac{1}{2(b+\lambda_j[\rho_k])} \le \frac{\mathcal{Q}[u,u] + b\|u\|_{L^2_{\rho_k}(\Omega)}^2}{2} - \|u - P_{n-1,\rho_k}u\|_{L^2_{\rho_k}(\Omega)} , \quad (2.1.8)$$

for all  $u \in V(\Omega)$  and  $j, k \in \mathbb{N}$ . Here  $P_{j-1,\rho_k}u$  denotes the orthogonal projection in  $L^2_{\rho_k}(\Omega)$  of u onto the space generated by  $u_1[\rho_k], \ldots, u_{j-1}[\rho_k]$  for all  $j \geq 2$  and  $P_{0,\rho_k}u \equiv 0$ . By setting  $u = \bar{u}$  and passing to the limit in (2.1.8) as  $k \to \infty$ , we obtain

$$-\frac{1}{2(b+\bar{\lambda}_j)} \le \frac{\mathcal{Q}[\bar{u},\bar{u}] + b \|\bar{u}\|_{L^2_{\rho}(\Omega)}^2}{2} - \|\bar{u}\|_{L^2_{\rho}(\Omega)} = -\frac{1}{2(b+\bar{\lambda})}$$

for all  $j \in \mathbb{N}$ , which contradicts the fact that  $\bar{\lambda}_j \to \infty$  as  $j \to \infty$ .

### 2.2 Analyticity of the eigenvalues

By classical results in perturbation theory, one can prove that  $\lambda_j[\rho]$  depends real-analytically on  $\rho$  as long as  $\lambda_j[\rho]$  is a simple eigenvalue. This is no longer true if the multiplicity of  $\lambda_j[\rho]$  varies. In the case of multiple eigenvalues, analyticity can be proved for the symmetric functions of the eigenvalues. Namely, given a finite set of indexes  $F \subset \mathbb{N}$ , we set

$$\mathcal{R}[F] := \{ \rho \in \mathcal{R} : \lambda_j[\rho] \neq \lambda_l[\rho], \forall j \in F, l \in \mathbb{N} \setminus F \}$$

and

$$\Lambda_{F,h}[\rho] := \sum_{\substack{j_1, \dots, j_h \in F \\ j_1 < \dots < j_h}} \lambda_{j_1}[\rho] \cdots \lambda_{j_h}[\rho], \quad h = 1, \dots, |F|.$$

Moreover, in order to compute formulas for the Frechét differentials, it is also convenient to set

$$\Theta[F] := \{ \rho \in \mathcal{R}[F] : \lambda_{j_1}[\rho] = \lambda_{j_2}[\rho], \forall j_1, j_2 \in F \}.$$

Then we have the following result:

**Theorem 2.2.1.** Assume that there exist a, b, c > 0 such that inequalities (1.3.3) and (1.3.4) are satisfied. Let F be a finite subset of  $\mathbb{N}$ . Then  $\mathcal{R}[F]$  is an open set in  $L^{\infty}(\Omega)$  and the functions  $\Lambda_{F,h}$  are real-analytic in  $\mathcal{R}[F]$ . Moreover, if  $F = \bigcup_{k=1}^{n} F_k$  and  $\rho \in \bigcap_{k=1}^{n} \Theta[F_k]$  is such that for each  $k = 1, \ldots, n$  the eigenvalues  $\lambda_j[\rho]$  assume the common value  $\lambda_{F_k}[\rho]$  for all  $j \in F_k$ , then the differentials of the functions  $\Lambda_{F,h}$  at the point  $\rho$  are given by the formula

$$d\Lambda_{F,h}[\rho][\dot{\rho}] = -\sum_{k=1}^{n} c_k \sum_{l \in F_k} \int_{\Omega} u_l^2 \dot{\rho} dx , \qquad (2.2.2)$$

for all  $\dot{\rho} \in L^{\infty}(\Omega)$ , where

$$c_{k} = \sum_{\substack{0 \le h_{1} \le |F_{1}| \\ 0 \le h_{n} \le |F_{n}| \\ h_{1} + \dots + h_{n} = h}} \binom{|F_{k}| - 1}{h_{k} - 1} \lambda_{F_{k}}^{h_{k}}[\rho] \prod_{\substack{j=1 \\ j \ne k}}^{n} \binom{|F_{j}|}{h_{j}} \lambda_{F_{j}}^{h_{j}}[\rho],$$

and for each k = 1, ..., n,  $\{u_l\}_{l \in F_k}$  is an orthonormal basis in  $L^2_{\rho}(\Omega)$  of the eigenspace associated with  $\lambda_{F_k}[\rho]$ .

*Proof.* We set

$$\tilde{\Lambda}_{F,h}[\rho] := \sum_{\substack{j_1, \dots, j_h \in F \\ j_1 < \dots < j_h}} (\lambda_{j_1}[\rho] + b) \cdots (\lambda_{j_h}[\rho] + b) \,,$$

for all  $\rho \in \mathcal{R}[F]$ . Note that by elementary combinatorics, we have

$$\Lambda_{F,h}[\rho] = \sum_{k=0}^{h} (-b)^{h-k} {|F|-k \choose h-k} \tilde{\Lambda}_{F,k}[\rho], \qquad (2.2.3)$$

where we have set  $\Lambda_{F,0} = \tilde{\Lambda}_{F,0} = 1$ .

By adapting to the operator  $L + bI_{\rho}$  the same argument used in [76] for the Dirichlet Laplacian, thanks to Theorem 1.3.4 and Theorem 1.2.7, one can prove that  $\mathcal{R}[F]$  is an open set in  $L^{\infty}(\Omega)$  and that  $\Lambda_{F,h}[\rho]$  depends realanalytically on  $\rho \in \mathcal{R}[F]$ . Thus, by (2.2.3) we deduce the real-analyticity of the functions  $\Lambda_{F,h}$ .

We now prove formula (2.2.2). First we assume that n = 1, hence  $F = F_1$  and  $\rho \in \Theta[F]$ . For simplicity, we write  $\lambda_F[\rho]$  rather than  $\lambda_{F_1}[\rho]$ . The same computations used in [76] yield the following formula for the Frechét differential  $d\Lambda_{F,h}[\rho]$  of  $\Lambda_{F,h}$  at the point  $\rho \in \mathcal{R}[F]$ :

$$d\tilde{\Lambda}_{F,h}[\rho][\dot{\rho}] = -(\lambda_F[\rho] + b)^{h+1} \binom{|F| - 1}{h - 1} \sum_{l \in F} \langle dT_{\rho}[\dot{\rho}][u_l], u_l \rangle_{\rho},$$

for all  $\dot{\rho} \in L^{\infty}(\Omega)$ , where  $T_{\rho}$  is defined by (1.3.6) with  $w = \rho$ . By standard calculus and by recalling that  $T_{\rho}u_l = (\lambda_F[\rho] + b)^{-1}u_l$  for all  $l \in F$ , we have

$$< dT_{\rho}[\dot{\rho}][u_{l}], u_{l} >_{\rho} = -b < (L + bI_{\rho})^{-1} dI_{\rho}[\dot{\rho}](L + bI_{\rho})^{-1} I_{\rho} u_{l}, u_{l} >_{\rho} + < (L + bI_{\rho})^{-1} dI_{\rho}[\dot{\rho}]u_{l}, u_{l} >_{\rho} = \frac{\lambda_{F}[\rho]}{\lambda_{F}[\rho] + b} < (L + bI_{\rho})^{-1} dI_{\rho}[\dot{\rho}]u_{l}, u_{l} >_{\rho} = \frac{\lambda_{F}[\rho]}{(\lambda_{F}[\rho] + b)^{2}} \int_{\Omega} u_{l}^{2} \dot{\rho} dx$$

hence

$$d\tilde{\Lambda}_{F,h}[\rho][\dot{\rho}] = -\lambda_F[\rho](\lambda_F[\rho] + b)^{h-1} \binom{|F| - 1}{h - 1} \sum_{l \in F} \int_{\Omega} u_l^2 \dot{\rho} dx , \qquad (2.2.4)$$

for all  $\dot{\rho} \in L^{\infty}(\Omega)$ . By (2.2.3) and (2.2.4) we get

$$d\Lambda_{F,h}[\rho][\dot{\rho}] = -\sum_{k=1}^{h} \lambda_{F}[\rho](\lambda_{F}[\rho] + b)^{k-1}(-b)^{h-k} {|F|-1 \choose k-1} {|F|-k \choose h-k} \sum_{l \in F} \int_{\Omega} u_{l}^{2} \dot{\rho} dx$$
$$= -\lambda_{F}[\rho] {|F|-1 \choose h-1} \sum_{k=0}^{h-1} {h-1 \choose k} (\lambda_{F}[\rho] + b)^{k}(-b)^{h-1-k} \sum_{l \in F} \int_{\Omega} u_{l}^{2} \dot{\rho} dx,$$

which immediately implies (2.2.2) for n = 1. We now consider the case n > 1. By means of a continuity argument, one can easily see that there exists an open neighborhood  $\mathcal{W}$  of  $\rho$  in  $\mathcal{R}[F]$  such that  $\mathcal{W} \subset \bigcap_{k=1}^{n} \mathcal{R}[F_k]$ . Thus,

$$\Lambda_{F,h} = \sum_{\substack{0 \le h_1 \le |F_1|, \dots, 0 \le h_n \le |F_n| \\ h_1 + \dots + h_n = h}} \prod_{k=1}^n \Lambda_{F_k, h_k}$$
(2.2.5)

on  $\mathcal{W}$ . By differentiating equality (2.2.5) at the point  $\rho$  and applying formula (2.2.2) for n = 1 to each function  $\Lambda_{F_k,h_k}$ , we deduce the validity of formula (2.2.2) for arbitrary values of  $n \in \mathbb{N}$ .

#### 2.3 Maximum principle

In this section we consider the case of general intermediate boundary conditions. This means that we assume that  $V(\Omega)$  is a closed subspace of  $H^m(\Omega)$ satisfying the inclusion

$$V(\Omega) \subset H^1_0(\Omega) \,. \tag{2.3.1}$$

Moreover, we assume that  $\Omega$  has finite measure. For all M > 0 we set

$$L_M := \left\{ \rho \in L^{\infty}(\Omega) : \int_{\Omega} \rho dx = M \right\}.$$

The following theorem is a generalization of [76, Theorem 4.4] to the case of intermediate boundary conditions and can be thought as a kind of maximum principle.

**Theorem 2.3.2.** Let all assumptions of Theorem 2.2.1 hold. Assume in addition that  $\Omega$  has finite measure and inclusion (2.3.1) holds. Then for all  $h = 1, \ldots, |F|$  the map  $\Lambda_{F,h}$  of  $\mathcal{R}[F] \cap L_M$  to  $\mathbb{R}$  which takes any  $\rho \in$  $\mathcal{R}[F] \cap L_M$  to  $\Lambda_{F,h}[\rho]$  has no points of local maximum or minimum  $\tilde{\rho}$  such that  $\lambda_j[\tilde{\rho}]$  have the same sign and  $\lambda_j[\tilde{\rho}] \neq 0$  for all  $j \in F$ .

Proof. It is convenient to consider the real-valued function M defined on  $L^{\infty}(\Omega)$  by  $M[\rho] := \int_{\Omega} \rho dx$  for all  $\rho \in L^{\infty}(\Omega)$ . Assume by contradiction the existence of  $\tilde{\rho}$  as in the statement. Then  $\tilde{\rho}$  is a critical point for the function  $\Lambda_{F,h}$  subject to the mass constraint  $M[\rho] = M$ . This implies the existence of a Lagrange multiplier, which means that there exists  $c \in \mathbb{R}$  such that  $d\Lambda_{F,h}[\tilde{\rho}] = cdM[\tilde{\rho}]$  (see e.g., [37, Theorem 26.1]). By formula (2.2.2), it follows that

$$\int_{\Omega} \left( \sum_{k=1}^{n} c_k \sum_{l \in F_k} u_l^2 \right) \dot{\rho} dx = c \int_{\Omega} \dot{\rho} dx,$$

for all  $\dot{\rho} \in L^{\infty}(\Omega)$ . Note that  $c_k$  are non-zero real numbers of the same sign. Since  $\dot{\rho}$  is arbitrary, it follows that

$$\left(\sum_{k=1}^{n} c_k \sum_{l \in F_k} u_l^2\right) = c, \quad \text{a.e. in } \Omega.$$
(2.3.3)

Since  $u_l \in H_0^1(\Omega)$ , then by a standard argument one can prove that the function  $(\sum_{k=1}^n \sum_{l \in F_k} (\sqrt{|c_k|}u_l)^2)^{1/2}$  belongs to the space  $H_0^1(\Omega)$  and equals  $\sqrt{|c|}$  almost everywhere in  $\Omega$ . As is well-known the space  $H_0^1(\Omega)$  does not contain constant functions apart from the function identically equal to zero. Thus c = 0 and accordingly  $u_l = 0$  for all  $l \in F$ , a contradiction.

**Remark 2.3.4.** Theorem 2.3.2 concerns mass densities  $\tilde{\rho}$  such that  $\lambda_j[\tilde{\rho}]$  do not vanish and have the same sign for all  $j \in F$ . This assumption is clearly guaranteed for positively defined operators. Moreover, we note that the sign of the eigenvalues is preserved by small perturbations of  $\rho$ . Hence our assumption is not much restrictive in the analysis of bifurcation phenomena associated with multiple eigenvalues different from zero.

Finally, by Theorems 2.1.5 and 2.3.2 we deduce the following:

**Corollary 2.3.5.** Let all assumptions of Theorem 2.3.2 hold. Let  $C \subset \mathcal{R}[F]$ be a weakly<sup>\*</sup> compact set in  $L^{\infty}(\Omega)$ . Assume that there exist a, b > 0 such that inequality (1.3.3) is satisfied for all  $\rho \in C$ . Let M > 0 be such that  $C \cap L_M$  is not empty. Assume that the eigenvalues  $\lambda_j[\rho]$  have the same sign and do not vanish for all  $j \in F$ ,  $\rho \in C$ . Then for all  $h = 1, \ldots, |F|$ , the map  $\Lambda_{F,h}$  from  $C \cap L_M$  to  $\mathbb{R}$  which takes  $\rho \in C \cap L_M$  to  $\Lambda_{F,h}[\rho]$  admits points of maximum and minimum and all such points belong to  $\partial C \cap L_M$ .

*Proof.* Recall that weakly<sup>\*</sup> compact sets are bounded. Thus, by Theorem 2.1.5 the functions  $\Lambda_{F,h}$  are weakly<sup>\*</sup> continuous on C hence they admit both maximum and minimum on the weakly<sup>\*</sup> compact subset  $C \cap L_M$  of C. By Corollary 2.3.2 the corresponding points of maximum and minimum cannot be interior points of C, hence they belong to  $\partial C \cap L_M$ .

Condition (2.3.1) was used only to guarantee that  $V(\Omega) \setminus \{0\}$  does not contain constant functions. Thus, one may replace condition (2.3.1) by slightly more general conditions. For example one may assume that  $V(\Omega) \subset$  $H_{0,\Gamma}^1(\Omega)$  where  $H_{0,\Gamma}^1(\Omega)$  is the closure in  $H^1(\Omega)$  of  $C^{\infty}$ -functions vanishing in an open neighborhood of a suitable subset of  $\Gamma$  of  $\partial\Omega$ . In this case, one would talk about mixed-intermediate boundary conditions. We can argue as in the proof of Theorem 2.3.2 and get to condition (2.3.3). Then we note that the function  $\left(\sum_{k=1}^n \sum_{l \in F_k} (\sqrt{|c_k|} u_l)^2\right)^{1/2}$  belongs to the space  $H_{0,\Gamma}^1(\Omega)$  and equals  $\sqrt{|c|}$  almost everywhere. The space  $H_{0,\Gamma}^1(\Omega)$  does not contain non-zero constant functions and therefore  $u_l = 0$  for all  $l \in F$ , a contradiction. Therefore Theorem 2.3.2 and Corollary 2.3.5 hold also in the case  $V(\Omega) \subset H_{0,\Gamma}^1(\Omega)$ .

**Remark 2.3.6.** If  $V(\Omega)$  is a closed subspace of  $H^m(\Omega)$  containing constant functions different from zero, then we could argue as in the proof on Theorem 2.3.2 up to condition (2.3.3). Thus, in the general case one could simply characterize the critical mass densities of the functions  $\Lambda_{F,h}$  as those mass densities for which condition (2.3.3) is satisfied. Clearly, in the case of simple eigenvalues condition (2.3.3) reduces to u = const in  $\Omega$  which implies that  $\lambda = 0$ . Thus, we conclude that the maximum principle stated in Theorem 2.3.2 holds for all simple eigenvalues and all homogeneous boundary conditions under consideration. As for multiple eigenvalues we note that the analysis of condition (2.3.3) is not straightforward as it may appear at a first glance.

#### 2.4 Poly-harmonic operators

In this sections we consider the case of poly-harmonic operators. Let  $m \in \mathbb{N}$ . Consider (2.0.1) and (2.0.2) with  $A_{\alpha\beta} = \delta_{\alpha\beta}m!/\alpha!$  for all  $\alpha, \beta \in \mathbb{N}_0^N$  with  $|\alpha| = |\beta| = m$ , where  $\delta_{\alpha\beta} = 1$  if  $\alpha = \beta$  and  $\delta_{\alpha\beta} = 0$  otherwise. Let  $l \in \mathbb{N}_0$ ,  $0 \leq l \leq m$  and  $V(\Omega) = H^m(\Omega) \cap H^l_0(\Omega)$ . Note that (1.3.3) and (1.3.4) are satisfied for any b > 0 where a, c > 0 are suitable constants possibly depending on b. Moreover, if  $1 \leq l \leq m$  and the open set  $\Omega$  has finite Lebesgue measure then the embedding  $V(\Omega) \subset L^2(\Omega)$  is compact. If l = 0 and the open set  $\Omega$  is bounded and has a Lipschitz continuous boundary then the embedding  $V(\Omega) \subset L^2(\Omega)$  is compact (actually it is enough to assume that  $\Omega$  is a bounded open set with a quasi-continuous boundary, see [26, Theorem 8]). Under these assumptions all corresponding eigenvalues  $\lambda_j$  are well-defined and non-negative.

Note that if l = m then  $V(\Omega) = H_0^m(\Omega)$  and by integrating by parts one can easily realize that the bilinear form  $\mathcal{Q}[u, \varphi]$  can be written in the more familiar form

$$\mathcal{Q}[u,\varphi] = \begin{cases} \int_{\Omega} \Delta^{\frac{m}{2}} u \Delta^{\frac{m}{2}} \varphi dx, & \text{if } m \text{ is even}, \\ \\ \int_{\Omega} \nabla \Delta^{\frac{m-1}{2}} u \nabla \Delta^{\frac{m-1}{2}} \varphi dx, & \text{if } m \text{ is odd}, \end{cases}$$

for all  $u, \varphi \in H_0^m(\Omega)$ . In this case we obtain the classical eigenvalue problem for poly-harmonic operators subject to the Dirichlet boundary conditions. The classical formulation of the Dirichlet problem problem is

$$\begin{cases} (-\Delta)^m u = \lambda \rho u, & \text{in } \Omega, \\ u = \frac{\partial u}{\partial \nu} = \dots = \frac{\partial^{m-1} u}{\partial \nu^{m-1}} = 0, & \text{on } \partial \Omega, \end{cases}$$

where we denote by  $\nu$  the outer unit normal to  $\partial\Omega$ . We recall that the Dirichlet problem arises in the study of vibrating strings for N = 1 and m = 1, membranes for N = 2 and m = 1, and clamped plates for N = 2 and m = 2. We refer to [31, 47, 94] for the physical derivation of the problem.

In the general case  $l \leq m$ , the classical formulation of the eigenvalue problem is

$$\begin{cases} (-\Delta)^m u = \lambda \rho u, & \text{in } \Omega, \\ \frac{\partial^j u}{\partial \nu^j} = 0, \ \forall \ j = 0, \dots, l-1, & \text{on } \partial\Omega, \\ \mathcal{B}_j u = 0, \ \forall \ j = 1, \dots, m-l, & \text{on } \partial\Omega, \end{cases}$$

where  $\mathcal{B}_j$  are uniquely defined 'complementing' boundary operators. See [88] for details. For  $N \geq 2$ , m = 2 and l = 1 we obtain the problem

$$\left\{ \begin{array}{ll} \Delta^2 u = \lambda \rho u, & \text{ in } \Omega, \\ u = 0, & \text{ on } \partial \Omega, \\ \Delta u - (N-1) \kappa \frac{\partial u}{\partial \nu} = 0, & \text{ on } \partial \Omega, \end{array} \right.$$

where  $\kappa$  is the mean curvature of the boundary of  $\Omega$ . This problem models a hinged vibrating rod for N = 1 and a hinged plate for N = 2. See [47] for further details.

Finally, we note that if m = 2 and l = 0 then  $V(\Omega) = H^2(\Omega)$  and problem (1.3.1) is the weak formulation of a Neumann-type problem for the biharmonic operator

$$\begin{cases} \Delta^2 u = \lambda \rho u, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial^2 \nu} = 0, & \text{on } \partial \Omega, \\ \operatorname{div}_{\partial \Omega} \left( D^2 u \cdot \nu \right) + \frac{\partial \Delta u}{\partial \nu} = 0, & \text{on } \partial \Omega. \end{cases}$$

Here  $\operatorname{div}_{\partial\Omega}$  is the tangential divergence and  $(D^2 u \cdot \nu)_{\partial\Omega}$  is the orthogonal projection of  $D^2 u \cdot \nu$  onto the tangent hyperplane to  $\partial\Omega$ . This prolem models a free rod for N = 1 and a free plate for N = 2. See also [28].

We consider all the poly-harmonic operators subject to Dirichlet or intermediate boundary conditions, i.e.,  $1 \le l \le m$ . From Theorem 2.1.5 and Theorem 2.3.2 we deduce the following:

**Corollary 2.4.1.** Let  $m, l \in \mathbb{N}$  with  $1 \leq l \leq m$ . Consider problem (2.0.3) with  $A_{\alpha\beta} = \delta_{\alpha\beta}m!/\alpha!$  for all  $\alpha, \beta \in \mathbb{N}_0^N$  with  $|\alpha| = |\beta| = m$  and  $V(\Omega) = H^m(\Omega) \cap H_0^l(\Omega)$ . Assume that  $\Omega$  has finite measure. Then for all h = 1, ..., |F|, the map  $\Lambda_{F,h}$  of  $\mathcal{R}[F] \cap L_M$  to  $\mathbb{R}$  which takes any  $\rho \in \mathcal{R}[F] \cap L_M$  to  $\Lambda_{F,h}[\rho]$  has no points of local maximum or minimum  $\tilde{\rho}$ . Moreover, let  $C \subset \mathcal{R}[F]$  be a weakly\* compact set in  $L^{\infty}(\Omega)$  such that  $\inf_{\rho \in C} \operatorname{essinf}_{x \in \Omega} \rho(x) > 0$ . Let M > 0 be such that  $C \cap L_M$  is not empty. Then for all h = 1, ..., |F|, the map  $\Lambda_{F,h}$  admits points of maximum and minimum and all such points belong to  $\partial C \cap L_M$ .

Proof. First we note that since  $\inf_{\rho \in C} \operatorname{ess} \inf_{x \in \Omega} \rho(x) > 0$ , inequality (1.3.3) is satisfied for suitable constants a, b > 0 not depending on  $\rho \in C$ . Moreover, the embedding  $H^m(\Omega) \cap H^l_0(\Omega) \subset L^2(\Omega)$  is compact. We also note that all the eigenvalues  $\lambda_j[\rho]$  are non-negative. The proof now is the same as that of Theorem 2.3.2 and of Corollary 2.3.5.

We consider now a particular class of weakly<sup>\*</sup> compact sets. Let  $A, B \in L^{\infty}(\Omega)$  be functions satisfying the condition

$$0 < \mathrm{ess} \inf_{x \in \Omega} A(x) < \mathrm{ess} \sup_{x \in \Omega} B(x) < \infty.$$

Let  $C := \{\rho \in L^{\infty}(\Omega) : A \leq \rho \leq B\}$ . Clearly, C is a weakly<sup>\*</sup> compact set. Moreover, since all mass densities  $\rho$  are uniformly bounded away from zero and infinity, inequality (1.3.3) is satisfied for suitable constants a, b > 0not depending on  $\rho \in C$ . Thus Corollary 2.4.1 is applicable to all non-zero eigenvalues. It turns out that point of maximum and minimum  $\tilde{\rho}$  should coincide with A(x) or B(x) in a set of positive measure.

## 2.5 The Laplace operator with Neumann boundary conditions

As we have observed at the end of Section 2.3, the analysis of condition (2.3.3) in the case of multiple eigenvalues and energy spaces  $V(\Omega)$  containing non-zero costant functions is not straightforward. We consider here the prototypical case of the Laplace operator with Neumann boundary conditions

$$\begin{cases} -\Delta u = \lambda \rho u, & \text{in } \Omega, \\ \frac{\partial u}{\partial \nu} = 0, & \text{on } \partial \Omega. \end{cases}$$
(2.5.1)

The weak formulation of problem (2.5.1) is

$$\int_{\Omega} \nabla u \cdot \nabla \varphi dx = \lambda \int_{\Omega} \rho u \varphi dx, \quad \forall \varphi \in H^1(\Omega),$$
 (2.5.2)

which correspond to the choice  $A_{\alpha\beta} = \delta_{\alpha\beta}$ ,  $V(\Omega) = H^1(\Omega)$  in (2.0.3). From Remark 2.3.6 it follows that in the case  $V(\Omega) = H^1(\Omega)$  we have that  $\tilde{\rho}$  is a critical point for  $\Lambda_{F,h}$  provided condition (2.3.3) is satisfied. Under suitable regularity assumptions on the eigenfunctions associated with a double eigenvalue, we may prove that the validity of (2.3.3) implies that the eigenvalue must be zero. This is proved in the following theorem.

**Theorem 2.5.3.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  with a Lipschitz continuous boundary, M > 0 and  $F = \{m, n\}$  with  $m, n \in \mathbb{N}, m \neq n$ . Let  $\tilde{\rho} \in \mathcal{R}[F]$ be continuous and moreover, assume that the solutions to problem (2.5.2) are classical solutions and the nodal domains are such that the Divergence Theorem holds. Then for h = 1, 2,  $\tilde{\rho}$  is not a critical mass density for the function which takes  $\rho \in \mathcal{R}[F]$  to  $\Lambda_{F,h}[\rho]$  under the constraint  $\rho \in \mathcal{R} \cap L_M$ .

*Proof.* Let  $\tilde{\rho} \in \mathcal{R}[F]$  be fixed. Then we have one of the following cases:

Case 1)  $\tilde{\rho} \in \Theta[F]$ . In this case  $\lambda_F = \lambda_m = \lambda_n$  is an eigenvalue of multiplicity 2. Then by (2.2.4) it follows that

$$d\Lambda_{F,1}[\tilde{\rho}][\dot{\rho}] = -\lambda_F^2 \int_{\Omega} \dot{\rho}(u_m^2 + u_n^2) dx$$
$$d\Lambda_{F,2}[\tilde{\rho}][\dot{\rho}] = -\lambda_F^3 \int_{\Omega} \dot{\rho}(u_m^2 + u_n^2) dx,$$

where  $\{u_m, u_n\}$  is a orthogonal basis in  $L^2_{\tilde{\rho}}(\Omega)$  of the eigenspace associated with  $\lambda_F$ .

Case 2)  $\tilde{\rho} \in \bigcap_{k=1}^{2} \Theta[F_k]$ , where  $F_1 = \{m\}$ ,  $F_2 = \{n\}$ . In this case  $\lambda_{F_1} = \lambda_m, \lambda_{F_2} = \lambda_n$  are two simple eigenvalues. Then there exists an open

neighborhood in  $\mathcal{R}$  of  $\tilde{\rho}$  such that  $\mathcal{W} \subseteq \bigcap_{k=1}^{2} \mathcal{R}[F_k]$ . Then

$$\begin{split} d\Lambda_{F,1}[\tilde{\rho}][\dot{\rho}] &= d(\Lambda_{F_2,1} + \Lambda_{F_1,1})[\tilde{\rho}][\dot{\rho}] \\ &= -\int_{\Omega} \dot{\rho}(\lambda_{F_2}^2 u_n^2 + \lambda_{F_1}^2 u_m^2) dx \,, \end{split}$$

$$\begin{split} d\Lambda_{F,2}[\tilde{\rho}][\dot{\rho}] &= d(\Lambda_{F_1,1}\Lambda_{F_2,1})[\tilde{\rho}][\dot{\rho}] \\ &= -\int_{\Omega} \dot{\rho}(\lambda_{F_1}\lambda_{F_2}^2 u_n^2 + \lambda_{F_2}\lambda_{F_1}^2 u_m^2) dx, \end{split}$$

where  $u_m$  is the eigenfunction associated with the eigenvalue  $\lambda_{F_1}$ ,  $u_n$  is the eigenfunction associated with the eigenvalue  $\lambda_{F_2}$ , and  $u_m$ ,  $u_n$  are such that  $\int_{\Omega} \tilde{\rho} u_m^2 dx = 1$ ,  $\int_{\Omega} \tilde{\rho} u_n^2 dx = 1$ .

Suppose now that  $\tilde{\rho}$  is a critical mass density for  $\Lambda_{F,h}$ , h = 1, 2 in  $L_M$ . Then, in both cases, from condition (2.3.3) it follows that there exist  $c_n, c_m > 0$ , c > 0 such that

$$(c_n u_n^2 + c_m u_m^2) = c$$
, a.e. in  $\Omega$ .

Let us consider separately the different cases:

i)  $\tilde{\rho} \in \Theta[F], \ d\Lambda_{F,1}[\tilde{\rho}][\dot{\rho}] = -\lambda_F^2 \int_{\Omega} \dot{\rho}(u_m^2 + u_n^2) dx$  (the case  $d\Lambda_{F,2}[\tilde{\rho}][\dot{\rho}] = -\lambda_F^3 \int_{\Omega} \dot{\rho}(u_m^2 + u_n^2) dx$  is analogous). Then, by differentiating the equality

$$u_m^2 + u_n^2 = C (2.5.4)$$

we obtain

$$u_m \nabla u_m + u_n \nabla u_n = 0 \tag{2.5.5}$$

which implies in particular

$$|\nabla u_m(x)|^2 = \frac{u_n^2(x)}{u_m^2(x)} |\nabla u_n(x)|^2,$$

for all  $x \in \Omega$  such that  $u_m(x) \neq 0$ . Let us differentiate again in (2.5.5) and use the fact that  $-\Delta u_m = \lambda_F \tilde{\rho} u_m$  and  $-\Delta u_n = \lambda_F \tilde{\rho} u_n$ . We obtain

$$|\nabla u_m(x)|^2 + |\nabla u_n(x)|^2 = \lambda_F \tilde{\rho} \left( u_m^2(x) + u_n^2(x) \right).$$

By combining (2.5.4), (2.5.5) and (2.5.6) we get

$$\left(\frac{u_n^2(x)}{u_m^2(x)} + 1\right) |\nabla u_n(x)|^2 = \lambda_F \tilde{\rho} C,$$

hence

$$|\nabla u_n(x)|^2 = \lambda_F \tilde{\rho} u_m(x)^2, \qquad (2.5.6)$$

$$\left|\nabla u_m(x)\right|^2 = \lambda_F \tilde{\rho} u_n^2(x).$$

ii)  $\tilde{\rho} \in \bigcap_{k=1}^{2} \Theta[F_k], d\Lambda_{F,1}[\tilde{\rho}][\dot{\rho}] = -\int_{\Omega} \dot{\rho}(\lambda_{F_2}^2 u_n^2 + \lambda_{F_1}^2 u_m^2) dx$ . In a similar way as for Case 1), since  $\lambda_{F_2}^2 u_n^2 + \lambda_{F_1}^2 u_m^2 = C$  and  $-\Delta u_m = \lambda_{F_1} \tilde{\rho} u_m$ ,  $-\Delta u_n = \lambda_{F_2} \tilde{\rho} u_n$ , we obtain the following relations:

$$|\nabla u_m(x)|^2 = \frac{\lambda_{F_2}^2}{C\lambda_{F_1}^2} \tilde{\rho} \left(\lambda_{F_1}^3 u_m^2(x) + \lambda_{F_2}^3 u_n^2(x)\right) u_n^2(x); \quad (2.5.7)$$
$$|\nabla u_n(x)|^2 = \frac{\lambda_{F_1}^2}{C\lambda_{F_2}^2} \tilde{\rho} \left(\lambda_{F_1}^3 u_m^2(x) + \lambda_{F_2}^3 u_n^2(x)\right) u_m^2(x).$$

iii)  $\tilde{\rho} \in \bigcap_{k=1}^{2} \Theta[F_k], \ d\Lambda_{F,2}[\tilde{\rho}][\dot{\rho}] = -\int_{\Omega} \dot{\rho}(\lambda_{F_1}\lambda_{F_2}^2 u_n^2 + \lambda_{F_2}\lambda_{F_1}^2 u_m^2) dx.$  By imposing  $\lambda_{F_1}\lambda_{F_2}^2 u_n^2 + \lambda_{F_2}\lambda_{F_1}^2 u_m^2 = C$  we obtain

$$|\nabla u_m(x)|^2 = \frac{\lambda_{F_2}^2}{C} \tilde{\rho} \left( \lambda_{F_1}^2 u_m^2(x) + \lambda_{F_2}^2 u_n^2(x) \right) u_n^2(x); \quad (2.5.8)$$
$$|\nabla u_n(x)|^2 = \frac{\lambda_{F_1}^2}{C} \tilde{\rho} \left( \lambda_{F_1}^2 u_m^2(x) + \lambda_{F_2}^2 u_n^2(x) \right) u_m^2(x).$$

By (2.5.6), (2.5.7) and (2.5.8) we observe that in all cases, the nodal set of one of the two eigenfunctions coincides with the set where the gradient of the other vanishes. We also note that there are no points in  $\Omega$  where both  $u_m$  and  $\nabla u_m$  vanish (respectively  $u_n$  and  $\nabla u_n$ ). This implies that nodal sets of  $u_m$  are manifolds and coincide with the sets where  $\nabla u_n$  vanishes. We observe that the nodal sets of the eigenfunctions u of problem (2.5.2) are not empty, since for such functions  $\int_{\Omega} \tilde{\rho} u dx = 0$ , hence u changes its sign on  $\Omega$ .

Let us consider a nodal domain  $\Omega_m$  of  $u_m$ . The function  $u_m$  doesn't change sign on  $\Omega_m$ . The boundary  $\partial\Omega_m$  of  $\Omega_m$  can be written as  $\partial\Omega_m =$  $\Gamma_1 \cup \Gamma_2$ , where  $\Gamma_1 \subset \partial\Omega$  and  $\Gamma_2 \subset \Omega$ . First we show that  $\Gamma_1 \neq \emptyset$ . Assume by contradiction that  $\Gamma_1 = \emptyset$ . The function  $u_n|_{\Omega_m}$  is an eigenfunction of problem (2.5.1) with  $\Omega$  replaced by  $\Omega_m$  corresponding to the eigenvalue  $\lambda_{F_2}$ . Indeed the equation  $-\Delta u_n = \lambda_{F_2} u_n$  is clearly satisfied on  $\Omega_m$  and  $\frac{\partial u_n}{\partial \nu} = 0$ on  $\partial\Omega_m$ , since  $\nabla u_n$  is zero on  $\partial\Omega_m$ . Since  $u_n|_{\Omega_m}$  is not identically zero, it must change sign. Thus, there exist at least two non-empty nodal domains for  $u_n|_{\Omega_m}$  in  $\Omega_m$ . We claim that al least one of them, say  $\Omega_{m_n}$ , is relatively compact in  $\Omega_m$ . If this were false, then there would exist at least a point xof  $\partial\Omega_m$  such that  $u_n(x) = 0$ , hence  $\nabla u_m(x) = 0$ . But we since  $\Gamma_1 = \emptyset$  we have  $u_m(x) = 0$ . Thus  $u_n(x) = u_m(x) = 0$ , hence C = 0, a contradiction. Thus there exists a nodal domain  $\Omega_{m_n}$  of  $u_n|_{\Omega_m}$  such that  $\Omega_{m_n} \subset \Omega_m$ . Now,  $u_m|_{\Omega_{m_n}}$  solves problem (2.5.1) with  $\lambda_{F_1}$ , hence it must change sign  $\Omega_{m_n}$ . But  $\Omega_{m_n}$  is relatively compact in  $\Omega_m$ , and on this set  $u_m$  has constant sign, a contradiction.

Thus we have proved that  $\Gamma_1 \neq \emptyset$ . Recall that  $u_m$  has constant sign on  $\Omega_m$ . Moreover,  $\frac{\partial u_n}{\partial \nu} = 0$  on  $\Gamma_1$ , while  $\nabla u_n = 0$  on  $\Gamma_2$ , since here  $u_m = 0$ . Then  $u_n|_{\Omega_m}$  is solution of problem (2.5.1) with  $\Omega$  replaced by  $\Omega_m$  corresponding to the eigenvalue  $\lambda_{F_2}$  and it changes sign on  $\Omega_m$ . Let  $\Omega_{m_n}$  be a nodal domain of  $u_n|_{\Omega_m}$ . By the arguments above we have that  $\partial\Omega_{m_n} = \Gamma_{1,n} \cup \Gamma_{2,n}$ , where  $\emptyset \neq \Gamma_{1,n} \subset \partial\Omega_m$ , and  $\Gamma_{2,n} \subset \Omega_m$ . We claim that there exists at least one nodal domain  $\Omega_{m_n}$  such that  $\Gamma_{1,n} \subset \partial\Omega$ . If this were false, the boundary  $\partial\Omega_{m_n}$  of each  $\Omega_{m_n}$  would be of the type:  $\partial\Omega_{m_n} = (\Gamma_{1,n} \cap \partial\Omega) \cup (\Gamma_{1,n} \cap (\partial\Omega_m \setminus (\partial\Omega \cap \partial\Omega_m))) \cup \Gamma_{2,n}$ , and each of these partitions of  $\partial\Omega_{m_n}$  would be non-empty. Since  $\Omega_{m_n}$  is connected,  $(\Gamma_{1,n} \cap (\partial\Omega_m \setminus (\partial\Omega \cap \partial\Omega_m))) \cap \Gamma_{2,n} \neq \emptyset$ . On this set  $u_m$  and  $\nabla u_m$  vanish, a contradiction. Thus there exists  $\Omega_{m_n}$  such that  $\Gamma_{1,n} \subset \partial\Omega$ . Then  $u_m|_{\Omega_{m_n}}$  is a nontrivial solution of problem (2.5.1) corresponding to the eigenvalue  $\lambda_{F_2}$ and changes its sign on  $\Omega_{m_n}$ , a contradiction. This concludes the proof.  $\Box$ 

As a consequence of Theorem 2.1.5 and Theorem 2.5.3 we have the following

**Corollary 2.5.9.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  with a Lipschitz continuous boundary,  $F = \{m, n\}$  with  $m, n \in \mathbb{N}$ ,  $m \neq n$ . Let  $C \subseteq \mathcal{R}[F]$  be a weakly\* compact subset of  $L^{\infty}(\Omega)$  such that  $\inf_{\rho \in C} \operatorname{ess} \inf_{\Omega} \rho > 0$ . Let M > 0and  $L_M = \{\rho \in L^{\infty}(\Omega) : \int_{\Omega} \rho = M\}$ . Then for h = 1, 2, the function which takes  $\rho \in C \cap L_M$  to  $\Lambda_{F,h}[\rho]$  has maxima and minima, and if for such points the solutions of problem (2.5.2) are classic solutions, they must belong to  $\partial C \cap L_M$ .

*Proof.* The proof is identical to that of Corollary 2.3.5.

# Chapter 3

# Mass concentration phenomena for second order operators

In this chapter we consider the eigenvalue problem for the Laplace operator  $-\Delta$  subject to Dirichlet and Neumann boundary conditions. In particular we shall consider eigenvalue problems of the type

$$-\Delta u = \lambda \rho_{\varepsilon} u$$

on a bounded open set  $\Omega$  in  $\mathbb{R}^N$ , where  $\rho_{\varepsilon}$  is a measurable and positive function which depends on a small parameter  $\varepsilon > 0$  and which is of order  $\varepsilon^{-1}$  in a  $\varepsilon$ -neighborhood of points or hypersurfaces contained in  $\Omega$  as  $\varepsilon \to 0$ and is of order  $\varepsilon$  in the rest of  $\Omega$ , as  $\varepsilon \to 0$ .

For N = 2 this problem is related to the study of the vibration of a thin membrane which occupies at rest a planar region  $\Omega \subset \mathbb{R}^2$  and the mass of which is displaced on the whole of  $\Omega$  with density  $\rho_{\varepsilon}$ . Roughly speaking, we consider vibrating membranes the mass of which concentrates near points or hypersurfaces contained in  $\Omega$  and we investigate the behavior of the eigenvalues, which represent the squares of the normal modes of vibration, as  $\varepsilon \to 0$ . Since the dimension does not play any relevant role in our discussion, we consider from now on open bounded sets in  $\mathbb{R}^N$ .

In the case of Neumann boundary conditions and mass densities which concentrate near the boundary of  $\Omega$ , we obtain that the Neumann eigenvalues converge to the eigenvalues of the Steklov problem for the Laplace operator, which in this sense can be considered a limiting Neumann problem. Then we shall discuss the dependence of the eigenvalues of the Steklov problem upon mass density perturbations in the same spirit of Chapter 2.

### 3.1 Neumann to Steklov eigenvalues

Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ . Let  $\rho \in \mathcal{R}^S$ , where

$$\mathcal{R}^{\mathcal{S}} := \left\{ \rho \in L^{\infty}(\partial\Omega) : \operatorname{ess\,inf}_{x \in \partial\Omega} \rho(x) > 0 \right\}.$$
(3.1.1)

We consider the classical Steklov eigenvalue problem for the Laplace operator

$$\begin{cases} \Delta u = 0, & \text{in } \Omega, \\ \frac{\partial u}{\partial \nu} = \lambda \rho u, & \text{on } \partial \Omega, \end{cases}$$
(3.1.2)

in the unknowns u (the eigenfunction),  $\lambda$  (the eigenvalue). This problem models a free vibrating membrane whose mass is concentrated at the boundary with surface density  $\rho$  (see [99] for the derivation of the problem). We consider the weak formulation of (3.1.2)

$$\int_{\Omega} \nabla u \cdot \nabla \varphi dx = \lambda \int_{\partial \Omega} \rho u \varphi d\sigma \,, \quad \forall \varphi \in H^1(\Omega) \,, \tag{3.1.3}$$

in the unknowns  $u \in H^1(\Omega)$ ,  $\lambda \in \mathbb{R}$ . Actually, we shall consider a problem in the space  $H^1(\Omega)/\mathbb{R}$  since we need to get rid of the constants, which generate the eigenspace corresponding to the eigenvalue  $\lambda = 0$ .

We denote by Tr the trace operator acting from  $H^1(\Omega)$  to  $L^2(\partial\Omega)$ . We denote by  $\mathcal{J}^{\mathcal{S}}_{\rho}$  the continuous embedding of  $L^2(\partial\Omega)$  into  $H^1(\Omega)'$  defined by

$$\mathcal{J}_{\rho}^{\mathcal{S}}[u][\varphi] := \int_{\partial\Omega} \rho u \varphi d\sigma, \quad \forall u \in L^{2}(\partial\Omega), \varphi \in H^{1}(\Omega).$$

We set

$$H^{1,\mathcal{S}}_{\rho}(\Omega) := \left\{ u \in H^1(\Omega) : \int_{\partial \Omega} \rho u d\sigma = 0 \right\},\,$$

and we consider on  $H^1(\Omega)$  the bilinear form

$$\langle u, v \rangle := \int_{\Omega} \nabla u \cdot \nabla v dx.$$
 (3.1.4)

By the Poincaré-Wirtinger inequality, it turns out that the bilinear form (3.1.4) is a scalar product on  $H^{1,S}_{\rho}(\Omega)$  whose induced norm is equivalent to the standard one. In the sequel we will think of the space  $H^{1,S}_{\rho}$  as endowed with the scalar product (3.1.4). Let  $F(\Omega)$  be defined by  $F(\Omega) := \{G \in H^1(\Omega)' : G[1] = 0\}$ . Then we consider the operator  $\mathcal{M}^S_{\rho}$  acting from  $H^{1,S}_{\rho}(\Omega)$  to  $F(\Omega)$ , defined by

$$\mathcal{M}_{\rho}^{\mathcal{S}}[u][\varphi] := \int_{\Omega} \nabla u \cdot \nabla \varphi dx, \quad \forall u \in H_{\rho}^{1,\mathcal{S}}(\Omega), \varphi \in H^{1}(\Omega).$$

It turns out that  $\mathcal{M}_{\rho}^{\mathcal{S}}$  is a homeomorphism of  $H_{\rho}^{1,\mathcal{S}}(\Omega)$  onto  $F(\Omega)$ . We define the operator  $\pi_{\rho}^{\mathcal{S}}$  from  $H^{1}(\Omega)$  to  $H_{\rho}^{1,\mathcal{S}}(\Omega)$  by

$$\pi_{\rho}^{\mathcal{S}}[u] := u - \frac{\int_{\partial\Omega} \rho u d\sigma}{\int_{\partial\Omega} \rho d\sigma}, \qquad (3.1.5)$$

for all  $u \in H^1(\Omega)$ . We consider the space  $H^1(\Omega)/\mathbb{R}$  endowed with the bilinear form induced by (3.1.4). Such bilinear form renders  $H^1(\Omega)/\mathbb{R}$  a Hilbert space. We denote by  $\pi_{\rho}^{\sharp,S}$  the map from  $H^1(\Omega)/\mathbb{R}$  onto  $H_{\rho}^{1,S}(\Omega)$ defined by the equality  $\pi_{\rho}^{S} = \pi_{\rho}^{\sharp,S} \circ p$ , where p is the canonical projection of  $H^1(\Omega)$  onto  $H^1(\Omega)/\mathbb{R}$ . The map  $\pi_{\rho}^{\sharp,S}$  turns out to be a homeomorphism. Finally, we define the operator  $T_{\rho}^{S}$  acting on  $H^1(\Omega)/\mathbb{R}$  as follows

$$T_{\rho}^{\mathcal{S}} := (\pi_{\rho}^{\sharp,\mathcal{S}})^{-1} \circ (\mathcal{M}_{\rho}^{\mathcal{S}})^{-1} \circ \mathcal{J}_{\rho}^{\mathcal{S}} \circ \operatorname{Tr} \circ \pi_{\rho}^{\sharp,\mathcal{S}}.$$
 (3.1.6)

**Lemma 3.1.7.** The pair  $(\lambda, u)$  of the set  $\mathbb{R} \times (H^{1,S}_{\rho}(\Omega) \setminus \{0\})$  satisfies (3.1.3) if and only if  $\lambda > 0$  and the pair  $(\lambda^{-1}, p[u])$  of the set  $\mathbb{R} \times ((H^1(\Omega)/\mathbb{R}) \setminus \{0\})$  satisfies the equation

$$\lambda^{-1}p[u] = T^{\mathcal{S}}_{\rho}p[u].$$

We have the following theorem.

**Theorem 3.1.8.** The operator  $T_{\rho}^{S}$  is a compact self-adjoint operator in  $H^{1}(\Omega)/\mathbb{R}$ , whose eigenvalues coincide with the reciprocals of the positive eigenvalues of problem (3.1.3). In particular, the set of eigenvalues of problem (3.1.3) is contained in  $[0, +\infty[$  and consists of the image of a sequence increasing to  $+\infty$ . Each eigenvalue has finite multiplicity.

*Proof.* For the self-adjointness, it suffices to observe that

$$\begin{split} < T^{\mathcal{S}}_{\rho}u, v >_{H^{1}(\Omega)/\mathbb{R}} = < (\pi^{\sharp,\mathcal{S}}_{\rho})^{-1} \circ (\mathcal{M}^{\mathcal{S}}_{\rho})^{-1} \circ \mathcal{J}^{\mathcal{S}}_{\rho} \circ \operatorname{Tr} \circ \pi^{\sharp,\mathcal{S}}_{\rho}u, v >_{H^{1}(\Omega)/\mathbb{R}} \\ &= \mathcal{M}^{\mathcal{S}}_{\rho}[(\mathcal{M}^{\mathcal{S}}_{\rho})^{-1} \circ \mathcal{J}^{\mathcal{S}}_{\rho} \circ \operatorname{Tr} \circ \pi^{\sharp,\mathcal{S}}_{\rho}u][\pi^{\sharp,\mathcal{S}}_{\rho}v] \\ &= \mathcal{J}^{\mathcal{S}}_{\rho}[\operatorname{Tr} \circ \pi^{\sharp,\mathcal{S}}_{\rho}u][\pi^{\sharp,\mathcal{S}}_{\rho}v], \quad \forall u, v \in H^{1}(\Omega)/\mathbb{R}, \end{split}$$

and that  $\mathcal{J}_{\rho}^{\mathcal{S}}[\operatorname{Tr}\circ\pi_{\rho}^{\sharp,\mathcal{S}}u][\pi_{\rho}^{\sharp,\mathcal{S}}v]$  is symmetric. As for compactness, just observe that the trace operator acting from  $H^{1}(\Omega)$  to  $L^{2}(\partial\Omega)$  is compact. The remaining statements follow by standard spectral theory.

Therefore the eigenvalues of (3.1.3) can be represented by means of an increasing sequence

$$0 \leq \lambda_1 \leq \lambda_2 \leq \cdots \leq \lambda_j \leq \cdots$$
.

The first positive eigenvalue is  $\lambda_2$  as proved in the following theorem.

**Theorem 3.1.9.** The first eigenvalue  $\lambda_1$  of (3.1.3) is zero and the corresponding eigenfunctions are the constant functions on  $\Omega$ . Moreover,  $\lambda_2 > 0$ .

*Proof.* It is straightforward to prove that the constant functions on  $\Omega$  are eigenfunctions of (3.1.3) with eigenvalue  $\lambda = 0$ . Suppose now that u is an eigenfunction corresponding to the eigenvalue  $\lambda = 0$ . Then we have

$$\int_{\Omega} |\nabla u|^2 dx = 0.$$

Therefore, since  $\Omega$  is connected, u is constant. This concludes the proof.  $\Box$ 

We can characterize the eigenvalues of (3.1.3) by means of the Rayleigh Min-Max Principle:

$$\lambda_j = \min_{\substack{E \subset H^1(\Omega) \\ \dim E = j}} \max_{\substack{u \in E \\ u \neq 0}} \frac{\int_{\Omega} |\nabla u|^2 dx}{\int_{\partial \Omega} \rho u^2 d\sigma}.$$
(3.1.10)

Now we turn our attention to the Neumann eigenvalue problem. Let  $\rho \in \mathcal{R}$ , where

$$\mathcal{R} := \left\{ \rho \in L^{\infty}(\Omega) : \operatorname{ess\,inf}_{x \in \Omega} \rho(x) > 0 \right\}.$$
(3.1.11)

We consider the classical Neumann eigenvalue problem for the Laplace operator

$$\begin{cases} -\Delta u = \lambda \rho u, & \text{in } \Omega, \\ \frac{\partial u}{\partial \nu} = 0, & \text{on } \partial \Omega, \end{cases}$$
(3.1.12)

in the unknowns u (the eigenfunction),  $\lambda$  (the eigenvalue). This problem models a free vibrating membrane of mass density  $\rho$  (see e.g., [31] for the derivation of the problem). We consider the weak formulation of problem (3.1.12)

$$\int_{\Omega} \nabla u \cdot \nabla \varphi dx = \lambda \int_{\Omega} \rho u \varphi dx \,, \quad \forall \varphi \in H^1(\Omega) \,, \tag{3.1.13}$$

in the unknowns  $u \in H^1(\Omega)$ ,  $\lambda \in \mathbb{R}$ . In the sequel we shall recast this problem in  $H^1(\Omega)/\mathbb{R}$  since we need to get rid of the constants, which generate the eigenspace corresponding to the eigenvalue  $\lambda = 0$ . We denote by i the canonical embedding of  $H^1(\Omega)$  into  $L^2(\Omega)$ . We denote by  $\mathcal{J}_{\rho}^{\mathcal{N}}$  the continuous embedding of  $L^2(\Omega)$  into  $H^1(\Omega)'$ , defined by

$$\mathcal{J}^{\mathcal{N}}_{\rho}[u][\varphi] := \int_{\Omega} \rho u \varphi dx \quad \forall u \in L^{2}(\Omega), \varphi \in H^{1}(\Omega).$$

We set

$$H^{1,\mathcal{N}}_{\rho}(\Omega) := \left\{ u \in H^1(\Omega) : \int_{\Omega} u\rho dx = 0 \right\}.$$
 (3.1.14)

In the sequel we shall consider the space  $H^{1,\mathcal{N}}_{\rho}(\Omega)$  as endowed with the scalar product (3.1.4). This scalar product induces on  $H^{1,\mathcal{N}}_{\rho}(\Omega)$  a norm which is equivalent to the standard one. Then we consider the operator  $\mathcal{M}^{\mathcal{N}}_{\rho}$  acting from  $H^{1,\mathcal{N}}_{\rho}(\Omega)$  to  $F(\Omega)$  defined by

$$\mathcal{M}^{\mathcal{N}}_{\rho}[u][\varphi] := \int_{\Omega} \nabla u \cdot \nabla \varphi dx, \quad \forall u \in H^{1,\mathcal{N}}_{\rho}(\Omega), \varphi \in H^{1}(\Omega).$$

The operator  $\mathcal{M}_{\rho}^{\mathcal{N}}$  turns out to be a linear homeomorphism of  $H_{\rho}^{1,\mathcal{N}}(\Omega)$  onto  $F(\Omega)$ . We define the operator  $\pi_{\rho}^{\mathcal{N}}$  from  $H^{1}(\Omega)$  to  $H_{\rho}^{1,\mathcal{N}}(\Omega)$  as

$$\pi_{\rho}^{\mathcal{N}}[u] := u - \frac{\int_{\Omega} u\rho dx}{\int_{\Omega} \rho dx},$$

for all  $u \in H^1(\Omega)$ . We consider the space  $H^1(\Omega)/\mathbb{R}$  endowed with de bilinear form induced by (3.1.4). Such form renders  $H^1(\Omega)/\mathbb{R}$  a Hilbert space. We denote by  $\pi_{\rho}^{\sharp,\mathcal{N}}$  the map from  $H^1(\Omega)/\mathbb{R}$  onto  $H^{1,\mathcal{N}}_{\rho}(\Omega)$  defined by the equality  $\pi_{\rho}^{\mathcal{N}} = \pi_{\rho}^{\sharp,\mathcal{N}} \circ p$ , where p is the canonical projection of  $H^1(\Omega)$  onto  $H^1(\Omega)/\mathbb{R}$ . We define the operator  $T_{\rho}^{\mathcal{N}}$  acting on  $H^1(\Omega)/\mathbb{R}$  as follows

$$T_{\rho}^{\mathcal{N}} := (\pi_{\rho}^{\sharp,\mathcal{N}})^{-1} \circ (\mathcal{M}_{\rho}^{\mathcal{N}})^{-1} \circ \mathcal{J}_{\rho}^{\mathcal{N}} \circ i \circ \pi_{\rho}^{\sharp,\mathcal{N}}.$$
(3.1.15)

**Lemma 3.1.16.** The pair  $(\lambda, u)$  of the set  $\mathbb{R} \times (H^{1,\mathcal{N}}_{\rho}(\Omega) \setminus \{0\})$  satisfies (3.1.13) if and only if  $\lambda > 0$  and the pair  $(\lambda^{-1}, p[u])$  of the set  $\mathbb{R} \times ((H^1(\Omega)/\mathbb{R}) \setminus \{0\})$  satisfies the equation

$$\lambda^{-1}p[u] = T^{\mathcal{N}}_{\rho}p[u].$$

We have the following theorem.

**Theorem 3.1.17.** The operator  $T_{\rho}^{\mathcal{N}}$  is a compact self-adjoint operator in  $H^1(\Omega)/\mathbb{R}$  and its eigenvalues coincide with the reciprocals of the positive eigenvalues of problem (3.1.13). In particular, the set of eigenvalues of problem (3.1.13) is contained in  $[0, +\infty[$  and consists of the image of a sequence increasing to  $+\infty$ . Each eigenvalue has finite multiplicity.

*Proof.* The proof is similar to that of Theorem 3.1.8. Just note that the embedding *i* from  $H^1(\Omega)$  to  $L^2(\Omega)$  is compact.

We have the following theorem on the spectrum of problem (3.1.13).

**Theorem 3.1.18.** The first eigenvalue  $\lambda_1$  of (3.1.13) is zero and the corresponding eigenfunctions are the constants. Moreover, the second eigenvalue  $\lambda_2$  of (3.1.13) is positive.



Now we consider problem (3.1.13) with densities which concentrate in a neighborhood of the boundary of  $\Omega$ . Let us denote by  $\omega_{\varepsilon}$  the set defined by

$$\omega_{\varepsilon} := \{ x \in \Omega : d(x, \partial \Omega) < \varepsilon \}.$$
(3.1.19)

Let us fix a positive number M > 0 and define the family of densities  $\{\rho_{\varepsilon}\}_{\varepsilon \in [0,\varepsilon_0]} \subset \mathcal{R}$  as follows:

$$\rho_{\varepsilon}(x) := \begin{cases} \varepsilon, & \text{if } x \in \Omega \setminus \overline{\omega}_{\varepsilon}, \\ \frac{M - \varepsilon |\Omega \setminus \overline{\omega}_{\varepsilon}|}{|\omega_{\varepsilon}|}, & \text{if } x \in \omega_{\varepsilon}, \end{cases}$$
(3.1.20)

for  $\varepsilon \in ]0, \varepsilon_0[$ , where  $\varepsilon_0$  is sufficiently small. We note that  $\int_{\Omega} \rho_{\varepsilon} dx = M$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . We refer to the quantity M as the total mass of the body.

Problem (3.1.3), and problem (3.1.13) with density  $\rho_{\varepsilon}$  are strictly related. In fact, under the assumption that  $\Omega$  is of class  $C^2$ , it is possible to prove that the eigenvalues of problem (3.1.13) with density  $\rho_{\varepsilon}$  converge to the eigenvalues of problem (3.1.3) with  $\rho \equiv \frac{M}{|\partial \Omega|}$ . This is a consequence of the following theorem.

**Theorem 3.1.21.** Let  $\Omega$  be a bounded domain of class  $C^2$  in  $\mathbb{R}^N$ . Let the operator  $T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}}$  and  $T^{\mathcal{N}}_{\rho_{\varepsilon}}$  be defined as in (3.1.6) and (3.1.15) respectively. Then  $T^{\mathcal{N}}_{\rho_{\varepsilon}}$  converges in norm to  $T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}}$  as  $\varepsilon \to 0$ .

We need some preliminary results in order to prove Theorem 3.1.21. First of all we note that  $\pi_c^{\sharp,S} = \pi_1^{\sharp,S}$  for all  $c \in \mathbb{R}$ , with  $c \neq 0$ . This can be deduced from (3.1.5).

Now we recall some facts from standard calculus which will be used in the sequel. Let M be a parametric hypersurface in  $\mathbb{R}^N$  of class  $C^2$ , i.e., there exists a function  $\phi \in C^2(\overline{D})$  from D to  $\mathbb{R}^N$ , where D is a bounded open subset of  $\mathbb{R}^{N-1}$  such that  $\phi(D) = M$ . We assume that  $\operatorname{rank} D\phi(y_1, \dots, y_{N-1}) = N-1$  for all  $(y_1, \dots, y_{N-1}) \in D$ , where we denoted by  $D\phi$  the Jacobian matrix of

 $\phi$ . We set

$$M(\varepsilon) := \{ \phi(y_1, ..., y_{N-1}) + t\nu(y_1, ..., y_{N-1}) : (y_1, ..., y_{N-1}) \in D, t \in ]0, \varepsilon[\},\$$

where  $\nu(y_1, ..., y_{N-1})$  is the normal vector to  $\phi(D)$  at the point  $\phi(y_1, ..., y_{N-1})$  given by

$$\nu(y_1,...,y_{N-1}) := \frac{\frac{\partial \phi}{\partial y_1} \wedge \cdots \wedge \frac{\partial \phi}{\partial y_{N-1}}}{\left|\frac{\partial \phi}{\partial y_1} \wedge \cdots \wedge \frac{\partial \phi}{\partial y_{N-1}}\right|}.$$

We consider the map  $\psi$  from  $D \times [0, \varepsilon]$  onto  $M(\varepsilon)$  defined by

 $\psi(y_1, \dots, y_{N-1}, t) := \phi(y_1, \dots, y_{N-1}) + t\nu(y_1, \dots, y_{N-1})$ 

for all  $(y_1, ..., y_{N-1}) \in D, t \in ]0, \varepsilon[$ . We need to compute  $\det D\psi$ . We have

$$\det D\psi = \det \left[ \frac{\partial \phi}{\partial y_1} \cdots \frac{\partial \phi}{\partial y_{N-1}} \nu(y_1, ..., y_{N-1}) \right] + tg_1 \left( \frac{\partial \phi}{\partial y_1}, \frac{\partial \nu}{\partial y_1}, ..., \frac{\partial \phi}{\partial y_{N-1}}, \frac{\partial \nu}{\partial y_{N-1}} \right) + \cdots + t^{N-1}g_{N-1} \left( \frac{\partial \phi}{\partial y_1}, \frac{\partial \nu}{\partial y_1}, ..., \frac{\partial \phi}{\partial y_{N-1}}, \frac{\partial \nu}{\partial y_{N-1}} \right),$$

where  $g_i$  are suitable compositions of sums and products of the first partial derivatives of  $\phi$  and  $\nu$ . As is known, the first term in the sum is equal to

$$\left|\frac{\partial\phi}{\partial y_1}\wedge\cdots\wedge\frac{\partial\phi}{\partial y_{N-1}}\right|,\,$$

which is the (N-1)-dimensional measure of the hypersurface. We are ready to prove the following lemma. For the sake of completeness we include also statement ii).

**Lemma 3.1.22.** Let M be a parametric hypersurface and  $(D, \phi)$  a parametrization of M. Assume that  $\inf_D \left| \frac{\partial \phi}{\partial y_1} \wedge \cdots \wedge \frac{\partial \phi}{\partial y_{N-1}} \right| > 0$ . Assume also that  $\varepsilon_0 > 0$  is such that  $\psi$  is a diffeomorphism for all  $\varepsilon \in ]0, \varepsilon_0[$ . Let  $f_{\varepsilon}, f \in H^1(M(\varepsilon_0))$  for all  $\varepsilon > 0$  be such that  $f_{\varepsilon} \to f$  in  $H^1(M(\varepsilon_0))$  as  $\varepsilon \to 0$ . Then we have

i)

$$\lim_{\varepsilon \to 0} \frac{1}{\varepsilon} \int_{M(\varepsilon)} f dx = \int_M f d\sigma; \qquad (3.1.23)$$

ii)

$$\lim_{\varepsilon \to 0} \frac{1}{\varepsilon} \int_{M(\varepsilon)} \left( f_{\varepsilon} - f \right) dx = 0.$$

*Proof.* We write the proof in the case N = 3 for simplicity. The proof for N > 3 is analogous and can be carried out by using the same arguments. We consider

$$\frac{1}{\varepsilon} \int_{M(\varepsilon)} f(x) dx = \frac{1}{\varepsilon} \int_0^{\varepsilon} \int_D (f \circ \psi)(y_1, y_2, t) \left| \det D\psi \right| dy_1 dy_2 dt,$$

and compute the limit as  $\varepsilon \to 0$ . We observe that

$$\det D\psi = \det \left[\frac{\partial\phi}{\partial y_1}, \quad \frac{\partial\phi}{\partial y_2}, \quad \nu(y_1, y_2)\right] + t \det \left[\frac{\partial\nu}{\partial y_1}, \quad \frac{\partial\phi}{\partial y_2}, \quad \nu(y_1, y_2)\right] - t \det \left[\frac{\partial\nu}{\partial y_2}, \quad \frac{\partial\phi}{\partial y_1}, \quad \nu(y_1, y_2)\right] + t^2 \det \left[\frac{\partial\nu}{\partial y_1}, \quad \frac{\partial\nu}{\partial y_2}, \quad \nu(y_1, y_2)\right]. \quad (3.1.24)$$

Moreover

$$\frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} (f \circ \psi)(y_{1}, y_{2}, t) \left| \det D\psi \right| dy_{1} dy_{2} dt$$

$$= \frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} \left( (f \circ \psi)(y_{1}, y_{2}, t) - (f \circ \psi)(y_{1}, y_{2}, 0) \right) \left| \det D\psi \right| dy_{1} dy_{2} dt$$

$$+ \frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} (f \circ \psi)(y_{1}, y_{2}, 0) \left| \det D\psi \right| dy_{1} dy_{2} dt. \quad (3.1.25)$$

For the first summand in the right-hand side of (3.1.25), we observe that for a.e.  $(y_1, y_2) \in D$ , we have  $|(f \circ \psi)(y_1, y_2, t) - (f \circ \psi)(y_1, y_2, 0)| \leq \int_0^{\varepsilon} \left| \frac{\partial (f \circ \psi)}{\partial t'}(y_1, y_2, t') \right| dt'$ . Then, since  $f \in H^1(M(\varepsilon))$ , we have

$$\begin{split} \frac{1}{\varepsilon} \int_0^\varepsilon \int_D \left| (f \circ \psi)(y_1, y_2, t) - (f \circ \psi)(y_1, y_2, 0) \right| \left| \det D\psi \right| dy_1 dy_2 dt \\ & \leq \int_D \int_0^\varepsilon \left| \frac{\partial (f \circ \psi)}{\partial t}(y_1, y_2, t) \right| \left| \det D\psi \right| dt dy_1 dy_2 \\ & \leq |M(\varepsilon)|^{\frac{1}{2}} \left\| \nabla f \right\|_{L^2(M(\varepsilon))}. \end{split}$$

Thus the first summand in the right-hand side of (3.1.25) vanishes as  $\varepsilon \to 0$ . For the second summand, observe that for  $(y_1, y_2) \in D$ 

$$\lim_{\varepsilon \to 0} \frac{1}{\varepsilon} \int_0^\varepsilon \left| \det D\psi(y_1, y_2, t) \right| dt = \left| \det \left[ \frac{\partial \phi}{\partial y_1}, \frac{\partial \phi}{\partial y_2}, \nu(y_1, y_2) \right] \right| = \left| \frac{\partial \phi}{\partial y_1} \wedge \frac{\partial \phi}{\partial y_2} \right|,$$

since the terms in (3.1.24) containing t vanish as  $\varepsilon \to 0$ . The last quantity is exactly the area element of the surface. Then we get

$$\lim_{\varepsilon \to 0} \frac{1}{\varepsilon} \int_{M(\varepsilon)} f dx = \int_M f d\sigma.$$

Now we prove statement ii). We consider

$$\begin{split} \frac{1}{\varepsilon} \int_{M(\varepsilon)} \left( f_{\varepsilon} - f \right) dx \\ &= \frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} \left( (f_{\varepsilon} \circ \psi)(y_{1}, y_{2}, t) - (f \circ \psi)(y_{1}, y_{2}, t) \right) \left| \det D\psi \right| dy_{1} dy_{2} dt \\ &= \frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} \left( (f_{\varepsilon} \circ \psi)(y_{1}, y_{2}, 0) - (f \circ \psi)(y_{1}, y_{2}, 0) \right) \left| \det D\psi \right| dy_{1} dy_{2} dt \\ &+ \frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} \int_{0}^{t} \left( \frac{\partial (f_{\varepsilon} \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') - \frac{\partial (f \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') \right) dt' \left| \det D\psi \right| dy_{1} dy_{2} dt \\ &\leq \frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} \left| (f_{\varepsilon} \circ \psi)(y_{1}, y_{2}, 0) - (f \circ \psi)(y_{1}, y_{2}, 0) \right| \left| \det D\psi \right| dy_{1} dy_{2} dt \\ &+ \frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} \int_{0}^{t} \left| \frac{\partial (f_{\varepsilon} \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') - \frac{\partial (f \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') \right| dt' \left| \det D\psi \right| dy_{1} dy_{2} dt. \end{split}$$

$$(3.1.26)$$

We set

$$G_1(y_1, y_2) := \left| \det \left[ \frac{\partial \nu}{\partial y_1}, \frac{\partial \phi}{\partial y_2}, \nu(y_1, y_2) \right] - \det \left[ \frac{\partial \nu}{\partial y_2}, \frac{\partial \phi}{\partial y_1}, \nu(y_1, y_2) \right] \right|, G_2(y_1, y_2) := \left| \det \left[ \frac{\partial \nu}{\partial y_1}, \frac{\partial \nu}{\partial y_2}, \nu(y_1, y_2) \right] \right|.$$

We have for the first summand of (3.1.26)

$$\begin{split} \frac{1}{\varepsilon} \int_0^\varepsilon \int_D \left| (f_\varepsilon \circ \psi)(y_1, y_2, 0) - (f \circ \psi)(y_1, y_2, 0) \right| \left| \det D\psi \right| dy_1 dy_2 dt \\ &\leq \frac{1}{\varepsilon} \int_0^\varepsilon \int_D \left| (f_\varepsilon \circ \psi)(y_1, y_2, 0) - (f \circ \psi)(y_1, y_2, 0) \right| \left| \frac{\partial \phi}{\partial y_1} \wedge \frac{\partial \phi}{\partial v} \right| dy_1 dy_2 dt \\ &+ \frac{1}{\varepsilon} \int_0^\varepsilon \int_D \left| (f_\varepsilon \circ \psi)(y_1, y_2, 0) - (f \circ \psi)(y_1, y_2, 0) \right| tG_1(y_1, y_2) dy_1 dy_2 dt \\ &+ \frac{1}{\varepsilon} \int_0^\varepsilon \int_D \left| (f_\varepsilon \circ \psi)(y_1, y_2, 0) - (f \circ \psi)(y_1, y_2, 0) \right| t^2 G_2(y_1, y_2) dy_1 dy_2 dt \\ &= \int_M \left| f_\varepsilon - f \right| d\sigma \\ &+ \frac{\varepsilon^2}{3} \int_D \left| (f_\varepsilon \circ \psi)(y_1, y_2, 0) - (f \circ \psi)(y_1, y_2, 0) \right| G_1(y_1, y_2) dy_1 dy_2 \\ &+ \frac{\varepsilon^2}{3} \int_D \left| (f_\varepsilon \circ \psi)(y_1, y_2, 0) - (f \circ \psi)(y_1, y_2, 0) \right| G_2(y_1, y_2) dy_1 dy_2 \\ &\leq C \int_M \left| f_\varepsilon - f \right| d\sigma, \end{split}$$

where C is a positive constant which is bounded, uniformly in  $\varepsilon > 0$ . Thus the first summand in (3.1.26) vanishes as  $\varepsilon \to 0$  because  $f_{\varepsilon} \to f$  in  $L^2(M)$ hence in  $L^1(M)$ . Now we consider the second summand in (3.1.26). We

have

$$\begin{split} &\frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} \int_{0}^{t} \left| \frac{\partial (f_{\varepsilon} \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') - \frac{\partial (f \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') \right| dt' \left| \det D\psi \right| dy_{1} dy_{2} dt \\ &\leq \frac{1}{\varepsilon} \int_{0}^{\varepsilon} \int_{D} t^{\frac{1}{2}} \left( \int_{0}^{t} \left| \frac{\partial (f_{\varepsilon} \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') - \frac{\partial (f \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') \right|^{2} dt' \right)^{\frac{1}{2}} \left| \det D\psi \right| dy_{1} dy_{2} dt \\ &\leq \frac{C}{\varepsilon} \int_{0}^{\varepsilon} t^{\frac{1}{2}} \left( \int_{D} \int_{0}^{t} \left| \frac{\partial (f_{\varepsilon} \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') - \frac{\partial (f \circ \psi)}{\partial t'}(y_{1}, y_{2}, t') \right|^{2} \cdot \left| \det D\psi \right| dt' dy_{1} dy_{2} \right)^{\frac{1}{2}} dt \\ &\leq C \varepsilon^{\frac{1}{2}} \left\| \nabla (f_{\varepsilon} - f) \right\|_{L^{2}(M(\varepsilon))}. \end{split}$$

This concludes the proof.

We define the set  $(\partial \Omega)^{\varepsilon_0}$  by

$$(\partial \Omega)^{\varepsilon_0} := \left\{ x \in \mathbb{R}^N : d(x, \partial \Omega) < \varepsilon_0 \right\}.$$

Moreover, we denote by  $\nu(\bar{x})$  the outer unit normal to  $\partial\Omega$  at a point  $\bar{x} \in \partial\Omega$ . We recall the following theorem.

**Theorem 3.1.27.** (Tubular Neighborhood Theorem). Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^2$ . Then there exists  $\varepsilon_0 > 0$  such that for each  $x \in (\partial \Omega)^{\varepsilon_0}$  there exists a unique couple  $(\bar{x}, s) \in \partial \Omega \times ] - \varepsilon_0, \varepsilon_0[$  such that  $x = \bar{x} + s\nu(\bar{x}),$  where  $\nu(\bar{x})$  is the outer unit normal to  $\partial \Omega$  at the point  $\bar{x}$ . Moreover,  $\bar{x}$  is the (unique) nearest to x point of the boundary and  $s = d(x, \partial \Omega)$ . Finally, possibly reducing the value of  $\varepsilon_0$ , the map  $x \mapsto (\bar{x}, s)$ is a diffeomorphism of class  $C^1$  from  $(\partial \Omega)^{\varepsilon_0}$  onto  $\partial \Omega \times ] - \varepsilon_0, \varepsilon_0[$ .

We are ready to prove the following lemma.

**Lemma 3.1.28.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^2$ . Let  $\varepsilon > 0$ and  $\rho_{\varepsilon} \in \mathcal{R}$  be as in (3.1.20). Then the following statements hold:

- i) For all  $\varphi \in H^1(\Omega)/\mathbb{R}$ ,  $\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi] \to \pi_1^{\sharp,\mathcal{S}}[\varphi]$  in  $L^2(\Omega)$  as  $\varepsilon \to 0$  (hence also in  $H^1(\Omega)$ );
- ii) if  $u_{\varepsilon} \to u$  in  $H^1(\Omega)/\mathbb{R}$  as  $\varepsilon \to 0$  then possibly passing to a subsequence,  $\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u_{\varepsilon}] \to \pi_1^{\sharp,\mathcal{S}}[u]$  in  $L^2(\Omega)$  as  $\varepsilon \to 0$ ;
- iii) assume that  $u_{\varepsilon}, w_{\varepsilon}, u, w \in H^{1}(\Omega)$  are such that  $u_{\varepsilon} \to u, w_{\varepsilon} \to w$  in  $L^{2}(\Omega)$  as  $\varepsilon \to 0$ , and that  $\|\nabla u_{\varepsilon}\|_{L^{2}(\Omega)}, \|\nabla u\|_{L^{2}(\Omega)} \leq C, \|\nabla w_{\varepsilon}\|_{L^{2}(\Omega)}, \|\nabla w\|_{L^{2}(\Omega)} \leq C$ , uniformly in  $\varepsilon > 0$ . Then

$$\int_{\Omega} \rho_{\varepsilon} \left( u_{\varepsilon} - u \right) w_{\varepsilon} dx \to 0,$$

and

$$\int_{\Omega} \rho_{\varepsilon} \left( w_{\varepsilon} - w \right) u dx \to 0,$$

 $as \; \varepsilon \to 0.$ 

42

*Proof.* We start by proving statement i). It is sufficient to show that

$$\lim_{\varepsilon \to 0} \left\| \frac{\int_{\Omega} \rho_{\varepsilon} \tilde{\varphi} dx}{M} - \frac{\int_{\partial \Omega} \tilde{\varphi} d\sigma}{|\partial \Omega|} \right\|_{L^{2}(\Omega)} = 0,$$

where  $\tilde{\varphi} \in H^1(\Omega)$  is such that  $\varphi = p[\tilde{\varphi}]$ . Since the equality

$$\lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \tilde{\varphi} dx = \frac{M}{|\partial \Omega|} \int_{\partial \Omega} \tilde{\varphi} d\sigma$$

holds by Lemma 3.1.22, i), we have the desired result.

Now we prove statement *iii*). We note that since  $u_{\varepsilon} \to u$ ,  $w_{\varepsilon} \to w$  as  $\varepsilon \to 0$ in  $L^2(\Omega)$  and  $u_{\varepsilon}, w_{\varepsilon}$  are uniformly bounded in  $H^1(\Omega)$ , then  $u_{\varepsilon} \to u$ ,  $w_{\varepsilon} \to w$ ,  $\operatorname{Tr}[u_{\varepsilon}] \to \operatorname{Tr}[u]$  and  $\operatorname{Tr}[w_{\varepsilon}] \to \operatorname{Tr}[w]$  as  $\varepsilon \to 0$ . Then, in order to prove *iii*) it is sufficient to prove that

$$\lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} u_{\varepsilon} w_{\varepsilon} dx = 0$$

whenever  $u_{\varepsilon} \to 0$  in  $L^2(\Omega)$  and  $\operatorname{Tr}[u_{\varepsilon}] \to 0$  in  $L^2(\partial \Omega)$ . We have that

$$\int_{\Omega} \rho_{\varepsilon} u_{\varepsilon} w_{\varepsilon} dx = \varepsilon \int_{\Omega \setminus \overline{\omega}_{\varepsilon}} u_{\varepsilon} w_{\varepsilon} dx + C(\varepsilon) \int_{\omega_{\varepsilon}} u_{\varepsilon} w_{\varepsilon} dx, \qquad (3.1.29)$$

where  $C(\varepsilon) = \frac{M-\varepsilon|\Omega\setminus\overline{\omega_{\varepsilon}}|}{|\omega_{\varepsilon}|}$ . The first summand clearly is  $O(\varepsilon)$  as  $\varepsilon \to 0$ . By multiplying and dividing the second summand by  $\varepsilon$  and observing that  $\varepsilon C(\varepsilon) \leq C' < +\infty$  for  $\varepsilon > 0$  small enough, we obtain that the second summand in the right-hand side of (3.1.29) is less than or equal to

$$C' \cdot \frac{1}{\varepsilon} \int_{\omega_{\varepsilon}} |u_{\varepsilon} w_{\varepsilon}| \, dx.$$

We now apply Theorem 3.1.27. Let  $\varepsilon_0 > 0$  be as in Theorem 3.1.27 and let  $\varepsilon \in ]0, \varepsilon_0[$ . Let  $x_0 \in \partial\Omega$  and  $U_0$  be a neighborhood of  $x_0$  in  $\mathbb{R}^N$  such that there exists  $V_0 \subset \mathbb{R}^{N-1}$  and a parametrization  $\phi \in C^2(V_0)$  such that the map  $\psi$  from  $V_0 \times ]0, \varepsilon[$  onto  $M(\varepsilon) = \{x \in \Omega : d(x, \partial\Omega \cap U_0) < \varepsilon\}$  defined by

$$\psi(p,t) := \phi(p) + t\nu(p), \quad \forall (p,t) \in V_0 \times ]0, \varepsilon[$$

is a diffeomorphism from  $V_0 \times ]0, \varepsilon[$  onto  $M(\varepsilon)$ . Here  $p = (p_1, ..., p_{N-1}) \in \mathbb{R}^{N-1}$  and  $\nu(p)$  denotes the unit inner normal to  $\partial\Omega$  at  $\phi(p)$ . Now we consider

$$\int_{M(\varepsilon)} \frac{1}{\varepsilon} |u_{\varepsilon} w_{\varepsilon}| \, dx = \int_{V_0} \int_0^{\varepsilon} \frac{|\det D\psi|}{\varepsilon} |(u_{\varepsilon} \circ \psi)(p, t)| \, |(w_{\varepsilon} \circ \psi)(p, t)| \, dt dp.$$
(3.1.30)

For almost every  $p \in V_0$ ,  $(u_{\varepsilon} \circ \psi)(p, t)$ ,  $(w_{\varepsilon} \circ \psi)(p, t)$  are absolutely continuous on  $[0, \varepsilon]$  and since this set is compact, also their product is absolutely continuous. Let  $p \in V_0$  be fixed. We have

$$\begin{aligned} (u_{\varepsilon} \circ \psi)(p,t)(w_{\varepsilon} \circ \psi)(p,t) \\ &= (u_{\varepsilon} \circ \psi)(p,0)(w_{\varepsilon} \circ \psi)(p,0) \\ + \int_{0}^{t} \frac{\partial(u_{\varepsilon} \circ \psi)}{\partial t'}(p,t')(w_{\varepsilon} \circ \psi)(p,t') + (u_{\varepsilon} \circ \psi)(p,t')\frac{\partial(w_{\varepsilon} \circ \psi)}{\partial t'}(p,t')dt', \\ (u_{\varepsilon} \circ \psi)(p,t') &= (u_{\varepsilon} \circ \psi)(p,0) + \int_{0}^{t'} \frac{\partial(u_{\varepsilon} \circ \psi)}{\partial s}(p,s)ds, \\ (w_{\varepsilon} \circ \psi)(p,t') &= (w_{\varepsilon} \circ \psi)(p,0) + \int_{0}^{t'} \frac{\partial(w_{\varepsilon} \circ \psi)}{\partial s}(p,s)ds. \end{aligned}$$

$$(3.1.31)$$

We observe that, for fixed  $\varepsilon$  and for a.e.  $p \in V_0$ , the quantity  $C_1(t,p) = \left(\int_0^t \left|\frac{\partial(u_{\varepsilon}\circ\psi)}{\partial t'}(t',p)\right|^2 dt'\right)^{\frac{1}{2}}$  is increasing in  $0 \le t \le \varepsilon$  hence  $C_1(t,p) \le C_1(\varepsilon,p)$  for all  $0 \le t \le \varepsilon$ .

The same result holds for  $C_2(t,p) = \left(\int_0^t \left|\frac{\partial(w_{\varepsilon}\circ\psi)}{\partial t'}(t',p)\right|^2 dt'\right)^{\frac{1}{2}}$ . Then, for  $0 \le t' \le t \le \varepsilon$  we have

$$\begin{aligned} \left| (u_{\varepsilon} \circ \psi)(p, t') \right| &\leq \left| (u_{\varepsilon} \circ \psi)(p, 0) \right| + t'^{\frac{1}{2}} C_1(t', p) \\ &\leq \left| (u_{\varepsilon} \circ \psi)(p, 0) \right| + t^{\frac{1}{2}} C_1(\varepsilon, p), \quad (3.1.32) \end{aligned}$$

$$\left| (w_{\varepsilon} \circ \psi)(p, t') \right| \leq \left| (w_{\varepsilon} \circ \psi)(p, 0) \right| + t'^{\frac{1}{2}} C_2(t', p) \\ \leq \left| (w_{\varepsilon} \circ \psi)(p, 0) \right| + t^{\frac{1}{2}} C_2(\varepsilon, p). \quad (3.1.33)$$

Now, let us consider the right hand side in (3.1.30). By using (3.1.31):

$$\begin{split} \int_{V_0} \int_0^{\varepsilon} \frac{|\det D\psi|}{\varepsilon} \left| (u_{\varepsilon} \circ \psi)(p, t) \right| \left| (w_{\varepsilon} \circ \psi)(p, t) \right| dtdp \\ & \leq \int_{V_0} \int_0^{\varepsilon} \frac{1}{\varepsilon} \left| (u_{\varepsilon} \circ \psi)(p, 0) \right| \left| (w_{\varepsilon} \circ \psi)(p, 0) \right| \left| \det D\psi \right| dtdp \\ & + \left\| \det D\psi \right\|_{L^{\infty}(V_0 \times [0,\varepsilon])} \int_{V_0} \int_0^{\varepsilon} \frac{1}{\varepsilon} \int_0^t \left| \frac{\partial (u_{\varepsilon} \circ \psi)}{\partial t'}(p, t') \right| \left| (w_{\varepsilon} \circ \psi)(p, t') \right| dt' dtdp \\ & + \left\| \det D\psi \right\|_{L^{\infty}(V_0 \times [0,\varepsilon])} \int_{V_0} \int_0^{\varepsilon} \frac{1}{\varepsilon} \int_0^t \left| (u_{\varepsilon} \circ \psi)(p, t') \right| \left| \frac{\partial (w_{\varepsilon} \circ \psi)}{\partial t'}(p, t') \right| dt' dtdp. \end{split}$$

$$(3.1.34)$$

Now using (3.1.32) and (3.1.33) and the fact that  $\|\nabla u_{\varepsilon}\|_{L^{2}(\Omega)}, \|\nabla w_{\varepsilon}\|_{L^{2}(\Omega)} \leq C$ , it is easy to prove that the second and third summand in the right-hand
side of (3.1.34) vanish as  $\varepsilon \to 0$ . For the first summand in the right-hand side of (3.1.34), we observe that

$$\det D\psi = \det \left[ \frac{\partial \phi}{\partial p_1} \cdots \frac{\partial \phi}{\partial p_{N-1}} \nu(p_1, ..., p_{N-1}) \right] + tg_1 \left( \frac{\partial \phi}{\partial p_1}, \frac{\partial \nu}{\partial p_1}, ..., \frac{\partial \phi}{\partial p_{N-1}}, \frac{\partial \nu}{\partial p_{N-1}} \right) + \cdots + t^{N-1}g_{N-1} \left( \frac{\partial \phi}{\partial p_1}, \frac{\partial \nu}{\partial p_1}, ..., \frac{\partial \phi}{\partial p_{N-1}}, \frac{\partial \nu}{\partial p_{N-1}} \right),$$

where  $g_i$  are suitable compositions of sums and products of the first partial derivatives of  $\phi$  and  $\nu$ . It is not restrictive to assume that

$$\inf_{V_0} \left| \frac{\partial \phi}{\partial p_1} \wedge \dots \wedge \frac{\partial \phi}{\partial p_{N-1}} \right| > 0.$$

Now, using the same argument as in the proof of statement ii) of Lemma 3.1.22, we obtain

$$\begin{split} \frac{1}{\varepsilon} \int_{V_0} \int_0^\varepsilon |(u_\varepsilon \circ \psi)(p,0)| \, |(w_\varepsilon \circ \psi)(p,0)| \, |\det D\psi| \, dt dp \\ &\leq \frac{1}{\varepsilon} \int_0^\varepsilon \int_{V_0} |(u_\varepsilon \circ \psi)(p,0)| \, |(w_\varepsilon \circ \psi)(p,0)| \, \left| \frac{\partial \phi}{\partial p_1} \wedge \dots \wedge \frac{\partial \phi}{\partial p_{N-1}} \right| \, dp dt \\ &\quad + \sum_{i=1}^{N-1} \frac{1}{\varepsilon} \int_0^\varepsilon \int_{V_0} |(u_\varepsilon \circ \psi)(p,0)| \, |(w_\varepsilon \circ \psi)(p,0)| \, t^i \, |g_i(p)| \, dp dt \\ &\leq \tilde{C} \int_{\partial \Omega \cap U_0} |u_\varepsilon| \, |w_\varepsilon| \, d\sigma, \end{split}$$

where  $\tilde{C}$  is uniformly bounded in  $\varepsilon \in ]0, \varepsilon_0[$ . Since  $\operatorname{Tr}[u_{\varepsilon}] \to 0$  in  $L^2(\partial\Omega)$  as  $\varepsilon \to 0$ , it follows that also the first summand in the right-hand side of (3.1.34) vanishes as  $\varepsilon \to 0$ .

Since  $\omega_{\varepsilon}$  can be covered by a finite number of open sets of the type  $M(\varepsilon)$ , say  $\omega_{\varepsilon} \subset \bigcup_{i=1}^{m} M_i(\varepsilon)$ , we have that

$$\frac{1}{\varepsilon} \int_{\omega_{\varepsilon}} |u_{\varepsilon} w_{\varepsilon}| \, dx \leq \sum_{i=1}^{m} \frac{1}{\varepsilon} \int_{M_i(\varepsilon)} |u_{\varepsilon} w_{\varepsilon}| \, dx.$$

This concludes the proof of statement *iii*). We now prove statement *ii*). Let  $\tilde{u}_{\varepsilon}, \tilde{u} \in H_1^{1,\mathcal{N}}(\Omega)$  (see (3.1.14)) be such that  $u_{\varepsilon} = p[\tilde{u}_{\varepsilon}], u = p[\tilde{u}]$ . We have

$$\begin{aligned} \left\| \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u_{\varepsilon}] - \pi_{1}^{\sharp,\mathcal{S}}[u] \right\|_{L^{2}(\Omega)} \\ &\leq \left\| \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u_{\varepsilon}] - \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u] \right\|_{L^{2}(\Omega)} + \left\| \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u] - \pi_{1}^{\sharp,\mathcal{S}}[u] \right\|_{L^{2}(\Omega)}. \end{aligned}$$
(3.1.35)

By statement i) it follows that the second summand in the right hand side of (3.1.35) goes to zero as  $\varepsilon \to 0$ . For the first summand, we have

$$\begin{split} \left\| \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u_{\varepsilon}] - \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u] \right\|_{L^{2}(\Omega)} &= \left\| \tilde{u}_{\varepsilon} - \frac{\int_{\Omega} \rho_{\varepsilon} \tilde{u}_{\varepsilon} dx}{M} - \tilde{u} + \frac{\int_{\Omega} \rho_{\varepsilon} \tilde{u} dx}{M} \right\|_{L^{2}(\Omega)} \\ &\leq \left\| \tilde{u}_{\varepsilon} - \tilde{u} \right\|_{L^{2}(\Omega)} + \frac{\left\| \int_{\Omega} \rho_{\varepsilon} \left( \tilde{u}_{\varepsilon} - \tilde{u} \right) dx \right\|_{L^{2}(\Omega)}}{M} \\ &\leq \left\| \tilde{u}_{\varepsilon} - \tilde{u} \right\|_{L^{2}(\Omega)} + \left( \frac{|\Omega|^{\frac{1}{2}}}{M} \right) \cdot \left| \int_{\Omega} \rho_{\varepsilon} \left( \tilde{u}_{\varepsilon} - \tilde{u} \right) dx \right| \end{split}$$

Now, if we prove that  $\tilde{u}_{\varepsilon} \to \tilde{u}$  in  $L^2(\Omega)$  we are done, since the result follows by statement *iii*) with  $w_{\varepsilon} \equiv 1$ . Since  $u_{\varepsilon} \rightharpoonup u$  in  $H^1(\Omega)/\mathbb{R}$ , then  $\tilde{u}_{\varepsilon} \rightharpoonup \tilde{u}$ in  $H_1^{1,\mathcal{N}}(\Omega)$  as  $\varepsilon \to 0$ . From the compactness of the embedding of  $H_1^{1,\mathcal{N}}(\Omega)$ into  $L^2(\Omega)$  it follows that  $\tilde{u}_{\varepsilon} \to \tilde{u}$  in  $L^2(\Omega)$ , as  $\varepsilon \to 0$ . This concludes the proof of statement *ii*) and of the theorem.  $\Box$ 

We recall the following definition.

**Definition 3.1.36.** Let H be a real Hilbert space,  $\mathcal{K}(H, H)$  be the Banach subspace of  $\mathcal{L}(H, H)$  of those  $T \in \mathcal{L}(H, H)$  which are compact. A set  $\mathcal{K} \subset \mathcal{K}(H, H)$  is said to be collectively compact if and only if the set

$$\{K[x]: K \in \mathcal{K}, x \in B\},\$$

where B is the open unit ball in H, has compact closure. We say that a sequence of compact operators  $\{K_n\}_{n\in\mathbb{N}}$  compactly converges to the operator K if  $\{K_n\}_{n\in\mathbb{N}}$  is collectively compact and  $K_n[x_n] \to K[x]$  whenever  $x_n \to x$  in H.

We refer to [8, 103] for details. We are now ready to prove Theorem 3.1.21.

Proof of Theorem 3.1.21. We prove that  $T^{\mathcal{N}}_{\rho_{\varepsilon}}$  compactly converges to the compact operator  $T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}}$ . This implies, in fact, that

$$\lim_{\varepsilon \to 0} \left\| \left( T^{\mathcal{N}}_{\rho_{\varepsilon}} - T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}} \right)^2 \right\|_{\mathcal{L}(H^1(\Omega)/\mathbb{R}, H^1(\Omega)/\mathbb{R})} = 0.$$
(3.1.37)

Then, since the operators  $\{T^{\mathcal{N}}_{\rho_{\varepsilon}}\}_{\varepsilon\in]0,\varepsilon_0[}$  and  $T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}}$  are self-adjoint, property (3.1.37) is equivalent to convergence in norm. We refer to [8, 103] for a proof of (3.1.37) and for a more detailed discussion on compact convergence of compact operators on Hilbert spaces. We recall that, by definition,  $T^{\mathcal{N}}_{\rho_{\varepsilon}}$  compactly converges to  $T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}}$  if the following requirements are fulfilled:

i) if  $||u_{\varepsilon}||_{H^{1}(\Omega)/\mathbb{R}} \leq C$  for all  $\varepsilon \in ]0, \varepsilon_{0}[$ , then the family  $\{T_{\rho_{\varepsilon}}^{\mathcal{N}}u_{\varepsilon}\}_{\varepsilon\in]0,\varepsilon_{0}[}$  has compact closure in  $H^{1}(\Omega)/\mathbb{R};$ 

ii) if 
$$u_{\varepsilon} \to u$$
 in  $H^1(\Omega)/\mathbb{R}$ , then  $T^{\mathcal{N}}_{\rho_{\varepsilon}}u_{\varepsilon} \to T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}}u$  in  $H^1(\Omega)/\mathbb{R}$ .

We prove i) first. Let  $u, \varphi \in H^1(\Omega)/\mathbb{R}$ . We have

$$\begin{split} \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[u] \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[\varphi] dx &= \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[u] - \pi_{1}^{\sharp, \mathcal{S}}[u] \right) \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[\varphi] dx \\ &+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}}[u] \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[\varphi] - \pi_{1}^{\sharp, \mathcal{S}}[\varphi] \right) dx \\ &+ \left( \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}}[u] \pi_{1}^{\sharp, \mathcal{S}}[\varphi] dx - \frac{M}{|\partial \Omega|} \int_{\partial \Omega} \pi_{1}^{\sharp, \mathcal{S}}[u] \pi_{1}^{\sharp, \mathcal{S}}[\varphi] d\sigma \right) \\ &+ \frac{M}{|\partial \Omega|} \int_{\partial \Omega} \pi_{1}^{\sharp, \mathcal{S}}[u] \pi_{1}^{\sharp, \mathcal{S}}[\varphi] d\sigma. \quad (3.1.38)$$

By Lemma 3.1.28, *iii*) we have that the first and second summands in the right-hand side of (3.1.38) go to zero as  $\varepsilon \to 0$ . As for the third summand, from Lemma 3.1.22, *i*) applied to the function f = 1 we have that  $|\omega_{\varepsilon}| = \varepsilon |\partial\Omega| + o(\varepsilon)$  as  $\varepsilon \to 0$ . Therefore  $\rho_{\varepsilon} = \frac{M}{\varepsilon |\partial\Omega|} + o(1)$  as  $\varepsilon \to 0$ . Thus, from Lemma 3.1.22 and formula (3.1.23) it follows that also the third summand of (5.3.9) goes to zero as  $\varepsilon \to 0$ . Moreover, the equality  $(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}})^{-1} \circ (\mathcal{M}_{\rho_{\varepsilon}}^{\mathcal{N}})^{-1} = (\pi_{1}^{\sharp,\mathcal{S}})^{-1} \circ (\mathcal{M}_{1}^{\mathcal{S}})^{-1}$  holds. Therefore, from (3.1.38) it follows that  $T_{\rho_{\varepsilon}}^{\mathcal{N}} u$  is bounded for each  $u \in H^{1}(\Omega)/\mathbb{R}$ . Thus, by Banach-Steinhaus Theorem, there exists C' such that  $\|T_{\rho_{\varepsilon}}^{\mathcal{N}}\|_{\mathcal{L}(H^{1}(\Omega)/\mathbb{R}, H^{1}(\Omega)/\mathbb{R})} \leq C'$  for all  $\varepsilon \in ]0, \varepsilon_{0}[$ . Moreover, since  $\|u_{\varepsilon}\|_{H^{1}(\Omega)/\mathbb{R}} \leq C$  for all  $\varepsilon \in ]0, \varepsilon_{0}[$ , possibly passing to a subsequence, we have that  $u_{\varepsilon} \to u$  in  $H^{1}(\Omega)/\mathbb{R}$ , for some  $u \in H^{1}(\Omega)/\mathbb{R}$ . This implies that, possibly passing to a subsequence,  $T_{\rho_{\varepsilon}}^{\mathcal{N}}u_{\varepsilon} \to w$  in  $H^{1}(\Omega)/\mathbb{R}$  as  $\varepsilon \to 0$ , for some  $w \in H^{1}(\Omega)/\mathbb{R}$ . We show that  $w = T_{M}^{\mathcal{S}}u_{\varepsilon}$ . To shorten our notation we set  $w_{\varepsilon} := T_{\rho_{\varepsilon}}^{\mathcal{N}}u_{\varepsilon}$ . By Lemma 3.1.28, *i*) we have

$$\lim_{\varepsilon \to 0} \int_{\Omega} \nabla(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[w_{\varepsilon}]) \cdot \nabla(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi]) dx$$
$$= \int_{\Omega} \nabla(\pi_{1}^{\sharp,\mathcal{S}}[w]) \cdot \nabla(\pi_{1}^{\sharp,\mathcal{S}}[\varphi]) dx, \qquad (3.1.39)$$

for all  $\varphi \in H^1(\Omega)/\mathbb{R}$ .

On the other hand, since  $\left(\mathcal{M}_{\rho_{\varepsilon}}^{\mathcal{N}} \circ \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}\right) w_{\varepsilon} = \left(\mathcal{J}_{\rho_{\varepsilon}}^{\mathcal{N}} \circ i \circ \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}\right) u_{\varepsilon}$ , we have that

$$\int_{\Omega} \nabla(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[w_{\varepsilon}]) \cdot \nabla(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi]) dx = \int_{\Omega} \rho_{\varepsilon} \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u_{\varepsilon}] \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi] dx \qquad (3.1.40)$$

Then, by Lemma 3.1.28, iii, (3.1.39) and (3.1.40) we have

$$< w, \varphi >_{H^{1}(\Omega)/\mathbb{R}}$$

$$= \lim_{\varepsilon \to 0} < w_{\varepsilon}, \varphi >_{H^{1}(\Omega)/\mathbb{R}} = \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [u_{\varepsilon}] \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [\varphi] dx$$

$$= \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [u_{\varepsilon}] - \pi_{1}^{\sharp, \mathcal{S}} [u] \right) \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [\varphi] dx$$

$$+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}} [u] \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [\varphi] - \pi_{1}^{\sharp, \mathcal{S}} [\varphi] \right) dx$$

$$+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}} [u] \pi_{1}^{\sharp, \mathcal{S}} [\varphi] dx$$

$$= \frac{M}{|\partial\Omega|} \int_{\partial\Omega} \pi_{1}^{\sharp, \mathcal{S}} [u] \pi_{1}^{\sharp, \mathcal{S}} [\varphi] d\sigma$$

$$= < T \frac{S}{\frac{M}{|\partial\Omega|}} u, \varphi >_{H^{2}(\Omega)/\mathbb{R}}.$$

hence  $w = T_{\frac{M}{|\partial\Omega|}}^{S} u$ . In a similar way one can prove that  $\|w_{\varepsilon}\|_{H^{1}(\Omega)/\mathbb{R}} \to \|w\|_{H^{1}(\Omega)/\mathbb{R}}$ . In fact

$$\begin{split} \lim_{\varepsilon \to 0} \|w_{\varepsilon}\|_{H^{1}(\Omega)/\mathbb{R}}^{2} &= \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[u_{\varepsilon}] - \pi_{1}^{\sharp, \mathcal{S}}[u] \right) \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[w_{\varepsilon}] dx \\ &+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}}[u] \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[w_{\varepsilon}] - \pi_{1}^{\sharp, \mathcal{S}}[w_{\varepsilon}] \right) dx \\ &+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}}[u] \left( \pi_{1}^{\sharp, \mathcal{S}}[w_{\varepsilon}] - \pi_{1}^{\sharp, \mathcal{S}}[w] \right) dx \\ &+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}}[u] \pi_{1}^{\sharp, \mathcal{S}}[w] dx \\ &= \frac{M}{|\partial\Omega|} \int_{\partial\Omega} \pi_{1}^{\sharp, \mathcal{S}}[u] \pi_{1}^{\sharp, \mathcal{S}}[w] d\sigma = \|w\|_{H^{1}(\Omega)/\mathbb{R}}^{2} .\end{split}$$

This proves *i*). As for point *ii*), let  $u_{\varepsilon} \to u$  in  $H^1(\Omega)/\mathbb{R}$ . Then there exists C'' such that  $||u_{\varepsilon}||_{H^1(\Omega)/\mathbb{R}} \leq C''$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . Then, by the same argument used for point *i*), for each sequence  $\varepsilon_j \to 0$ , possibly passing to a subsequence, we have  $T^{\mathcal{N}}_{\rho_{\varepsilon_j}} u_{\varepsilon_j} \to T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}} u$ . Since this is true for each  $\{\varepsilon_j\}_{j\in\mathbb{N}}$ , we have the convergence for the whole family, i.e.,  $T^{\mathcal{N}}_{\rho_{\varepsilon}} u_{\varepsilon} \to T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}} u$ . This concludes the proof.

We need the following well-known result.

**Theorem 3.1.41.** Let H be a real Hilbert space and  $\{A_{\varepsilon}\}_{\varepsilon \in ]0,\varepsilon_0[}$  be a family of bounded self-adjoint operators converging in norm to the bounded selfadjoint operator A, i.e.,  $\lim_{\varepsilon \to 0} ||A_{\varepsilon} - A||_{\mathcal{L}(H,H)} = 0$ . Then isolated eigenvalues  $\lambda$  of A of finite multiplicity are exactly the limits of the eigenvalues of  $A_{\varepsilon}$ , counting the multiplicity. Moreover, the corresponding eigenprojections converge in norm.

Thanks to Theorem 3.1.41, as an immediate corollary of Theorem 3.1.21 we have

**Corollary 3.1.42.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^2$ . Let  $\lambda_j[\rho_{\varepsilon}]$  denote the eigenvalues of problem (3.1.13) with density  $\rho_{\varepsilon}$  on  $\Omega$  for all  $j \in \mathbb{N}$ . Let  $\lambda_j, j \in \mathbb{N}$  denote the eigenvalues of problem (3.1.3) corresponding to the constant surface density  $\frac{M}{|\partial \Omega|}$ . Then  $\lim_{\varepsilon \to 0} \lambda_j[\rho_{\varepsilon}] = \lambda_j$  for all  $j \in \mathbb{N}$ .

# 3.2 The Steklov eigenvalue problem. Mass density perturbations

In this section we discuss the dependence of the eigenvalues of problem (3.1.2) on the weight  $\rho$ . We shall obtain results of continuity and real analiticity of the eigenvalues in the spirit of Theorem 2.1.5 and Theorem 2.2.1. We note that this problem has a rather different behavior under mass density perturbations with respect to the operators considered in Chapter 2. In fact, in some particular cases we are able to find mass densities which are critical for the symmetric functions of the eigenvalues under mass constraint. Through all this section  $\Omega$  is a bounded domain of class  $C^1$ . Moreover, we shall denote the eigenvalues of problem (3.1.2) by  $\lambda_j[\rho]$  for all  $j \in \mathbb{N}$ .

### 3.2.1 Continuity and analyticity of the eigenvalues

By the min-max principle (3.1.10) it is possible to prove that  $\lambda_j[\rho]$  is a locally Lipschitz continuous function of  $\rho \in \mathcal{R}^S$ . In fact as in Section 2.1 it is possible to prove that

$$|\lambda_j[\rho_1] - \lambda_j[\rho_2]| \le \frac{\min\{\lambda_j[\rho_1], \lambda_j[\rho_2]\}}{\min\{\operatorname{ess\,inf}_{\partial\Omega} \rho_1, \operatorname{ess\,inf}_{\partial\Omega} \rho_2\}} \|\rho_1 - \rho_2\|_{L^{\infty}(\partial\Omega)},$$

for all  $\rho_1, \rho_2 \in \mathcal{R}^S$  satisfying  $\|\rho_1 - \rho_2\|_{L^{\infty}(\partial\Omega)} < \min\{\operatorname{ess\,inf}_{\partial\Omega} \rho_1, \operatorname{ess\,inf}_{\partial\Omega} \rho_2\}$ . The eigenvalues  $\lambda_j[\rho]$  depend with continuity on  $\rho$  also with respect the weak\* topology of  $L^{\infty}(\partial\Omega)$ . We have the following theorem.

**Theorem 3.2.1.** Let  $C \subset \mathcal{R}^{S}$  be a bounded set. Then the function which takes  $\rho \in C$  to  $\lambda_{j}[\rho]$  is continuous in the weak<sup>\*</sup> topology of  $L^{\infty}(\partial\Omega)$ .

*Proof.* The proof is analogous to that of Theorem 2.1.5 and accordingly is omitted.  $\Box$ 

We prove now that all simple eigenvalues and the symmetric functions of the eigenvalues of problem (3.1.2) depend real analitically on  $\rho$  and provide Hadamard-type formulas for the corresponding Frechét differentials. Let F be a finite nonempty subset of  $\mathbb{N}$ . We set

$$\mathcal{R}^{\mathcal{S}}[F] := \{ \rho \in \mathcal{R}^{\mathcal{S}} : \lambda_j[\rho] \neq \lambda_l[\rho], \ \forall j \in F, l \in \mathbb{N} \setminus F \} , \\ \Theta^{\mathcal{S}}[F] := \{ \rho \in \mathcal{R}^{\mathcal{S}}[F] : \lambda_{j_1}[\rho] = \lambda_{j_2}[\rho], \ \forall j_1, j_2 \in F \}.$$

Given  $\rho \in \mathcal{R}^{\mathcal{S}}$ , we denote by  $L^2_{\rho}(\partial \Omega)$  the space  $L^2(\partial \Omega)$  endowed with the bilinear form

$$\langle u, v \rangle_{\rho, \partial \Omega} := \int_{\partial \Omega} \rho u v d\sigma.$$

Such bilinear form is a scalar product on  $L^2(\partial\Omega)$  which induces on  $L^2(\partial\Omega)$ a norm equivalent to the standard one. Then we have the following result.

**Theorem 3.2.2.** Let F be a nonempty finite subset of  $\mathbb{N}$ . Then  $\mathcal{R}^{\mathcal{S}}[F]$  is open in  $L^{\infty}(\partial\Omega)$  and the symmetric functions of the eigenvalues

$$\Lambda_{F,h}[\rho] = \sum_{\substack{j_1,\dots,j_h \in F\\ j_1 < \dots j_h}} \lambda_{j_1}[\rho] \cdots \lambda_{j_h}[\rho], \quad h = 1, \dots, |F|,$$

are real analytic in  $\mathcal{R}^{\mathcal{S}}[F]$ . Moreover, if  $F = \bigcup_{k=1}^{n} F_k$  and  $\rho \in \bigcap_{k=1}^{n} \Theta^{\mathcal{S}}[F_k]$ is such that for each k = 1, ..., n the eigenvalues  $\lambda_j[\rho]$  assume the common value  $\lambda_{F_k}[\rho]$  for all  $j \in F_k$ , then the differentials of the functions  $\Lambda_{F,h}$  at the point  $\rho$  are given by the formula

$$d\Lambda_{F,h}[\rho][\dot{\rho}] = -\sum_{k=1}^{n} c_k \sum_{l \in F_k} \int_{\partial \Omega} u_l^2 \dot{\rho} d\sigma , \qquad (3.2.3)$$

for all  $\dot{\rho} \in L^{\infty}(\partial\Omega)$ , where

$$c_k = \sum_{\substack{0 \le h_1 \le |F_1| \\ \cdots \\ h_n \le |F_n| \\ h_1 + \cdots + h_n = h}} \binom{|F_k| - 1}{h_k - 1} \lambda_{F_k}^{h_k}[\rho] \prod_{\substack{j=1 \\ j \ne k}}^n \binom{|F_j|}{h_j} \lambda_{F_j}^{h_j}[\rho],$$

and for each k = 1, ..., n,  $\{u_l\}_{l \in F_k}$  is an orthonormal basis in  $L^2_{\rho}(\partial \Omega)$  of the eigenspace associated with  $\lambda_{F_k}[\rho]$ .

*Proof.* The proof follows the same lines as that of Theorem 2.2.1 and is accordingly omitted.  $\Box$ 

#### 3.2.2 Critical mass densities

We consider now the problem of finding critical mass densities for the symmetric functions of the eigenvalues under mass constraint, i.e., mass densities  $\rho$  which satisfy  $\operatorname{Ker} dM_{\partial}[\rho] \subset \operatorname{Ker} d\Lambda_{F,h}[\rho]$ , where  $M_{\partial}[\rho] := \int_{\partial\Omega} \rho d\sigma$ . As a consequence of Theorem 3.2.2 and formula (3.2.3) combined with the Lagrange Multipliers Theorem, we can give a characterization of such critical mass densities.

**Corollary 3.2.4.** Let all assumptions of Theorem 3.2.2 hold. Then  $\rho \in \mathbb{R}^{S}$  is a critical mass density for  $\Lambda_{F,h}$  for some h = 1, ..., |F|, subject to mass constraint if and only if there exists  $c \geq 0$  such that

$$\left(\sum_{k=1}^{n}\sum_{l\in F_k} (\operatorname{Tr} u_l)^2\right) = c, \quad \text{a.e. on } \partial\Omega.$$
(3.2.5)

The analysis carried out in Chapter 2 has pointed out that for a large class of non-negative elliptic operators subject to homogeneous boundary conditions there are no critical mass densities for simple eigenvalues and the symmetric functions of multiple eigenvalues. In the case of Steklov boundary conditions the situation is much different. Indeed, if  $\Omega$  is a ball, then a critical mass density exists.

**Corollary 3.2.6.** Let *B* be the unit ball in  $\mathbb{R}^N$ . Let  $k \in \mathbb{N}_0$ . Let us denote by  $n_k$  the number of linearly independent spherical harmonics of degree strictly less than k in  $\mathbb{R}^N$  and by  $d_k$  the number of linearly independent spherical harmonics of degree k (see Corollary 1.4.16). Let  $F = \{n_k + 1, ..., n_k + d_k\}$  and M > 0. Then the constant mass density  $\rho \equiv \frac{M}{|\partial B|}$  is a critical mass density for  $\Lambda_{F,h}$  for  $h = 1, ..., d_k$  under the constraint  $\int_{\partial \Omega} \rho d\sigma = M$ .

Proof. It is well-known that the eigenvalues of problem (3.1.2) on the unit ball with constant density  $\rho \equiv \frac{M}{|\partial B|}$  are of the form  $\lambda_k = \frac{|\partial B|k}{M}$ ,  $k \in \mathbb{N}_0$ . Each eigenvalue  $\lambda_k$  has multiplicity  $d_k$  and the eigenfunctions associated with  $\lambda_k$  are exactly the homogeneous harmonic polynomials of degree k in  $\mathbb{R}^N$ . Therefore, the set  $\{u_{k,j} := |x|^k H_{k,j}\}_{j=1}^{d_k}$ , where  $\{H_{k,j}\}_{j=1}^{d_k}$  is a basis for the spherical harmonics of degree k in  $\mathbb{R}^N$  and  $\{H_{k,j}\}_{j=1}^{d_k}$  are normalized such that  $\frac{M}{|\partial B|} \int_{\partial B} H_{k,j} H_{k,i} d\sigma = \delta_{i,j}$  for all  $i, j = 1, ..., d_k$ , is a basis for the eigenspace associated with the eigenvalue  $\lambda_k$ . Thus condition (3.2.5) is satisfied since it is well-known (see e.g., [40]) that

$$\sum_{j=1}^{d_k} u_{k,j}^2 = c_k \quad \text{on } \partial B,$$

for a suitable constant  $c_k > 0$ . Then the constant density  $\frac{M}{|\partial B|}$  is a critical mass density for  $\Lambda_{F,h}$ .

**Remark 3.2.7.** In the same hypothesis of Corollary 3.2.6, consider the particular case of  $F = \{2, ..., N + 1\}$ . We note that  $d_1 = N$  and the set  $\{u_{1,j} := c_j x_j\}_{j=1}^N$ , where  $c_j = \left(\frac{M}{|\partial B|}\int_{\partial B} x_j^2 d\sigma\right)^{-1}$  for all j = 1, ..., N is a orthonormal basis in  $L^2(\partial B)$  of the eigenspace associated with the first positive eigenvalue  $\frac{M}{|\partial B|}$  of problem (3.1.2) on the unit ball with constant density, which has multiplicity N. Then the constant density  $\frac{M}{|\partial B|}$  is a critical mass density for  $\Lambda_{F,h}$  for h = 1, ..., N under the constraint  $\int_{\partial B} \rho d\sigma = M$ .

It is interesting to compare Corollary 3.2.6 and Remark 3.2.7 with a classical result proved by Hersch, Payne and Schiffer [61] in the case of a class of planar domains. We recall that in the case of the unit ball in  $\mathbb{R}^2$  and constant density  $\frac{M}{2\pi}$ , we have that  $\lambda_2[\frac{M}{2\pi}] = \lambda_3[\frac{M}{2\pi}] = \frac{2\pi}{M}$ . We have the following theorem.

**Theorem 3.2.8** (Hersch, Payne, Schiffer). Let  $\Omega$  be the unit disk in  $\mathbb{R}^2$  centered at zero and M > 0 be fixed. Then

$$\lambda_2[\rho]\lambda_3[\rho] \le \frac{M^2}{4\pi^2}.$$

The equality is attained only at  $\rho \equiv \frac{M}{2\pi}$ .

Thus in the case of a ball in  $\mathbb{R}^2$  the constant mass density is in fact the unique maximizer for the first positive eigenvalue  $\lambda_2[\rho]$  among all mass densities preserving the total mass. We refer to [14] for further discussions on the problem of maximization of Steklov eigenvalues subject to mass density perturbations.

In the next section we consider the problem of minimizing  $\lambda_2[\rho]$  among all mass densities preserving the total mass.

## 3.3 Minimization of the first positive Steklov eigenvalue

Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$  and M > 0 be a fixed number. Let  $\lambda_2[\rho]$  be the first positive eigenvalue of problem (3.1.2) on  $\Omega$ . We shall prove that there exists a sequence  $\rho_{\varepsilon}$  of densities such that  $\int_{\partial\Omega} \rho_{\varepsilon} d\sigma = M$ and  $\lambda_2[\rho_{\varepsilon}] \to 0$  as  $\varepsilon \to 0$ . Therefore, the problem

$$\min_{\substack{\rho \in \mathcal{R}^{\mathcal{S}} \\ \int_{\partial \Omega} \rho d\sigma = M}} \lambda_2[\rho]$$

has no solutions. In Subsection 3.3.1 we prove the result for the unit ball in  $\mathbb{R}^2$ . In Subsection 3.3.2 we extend the result to the case of the unit ball in  $\mathbb{R}^N$  for  $N \geq 3$ . Finally, in Subsection 3.3.3 we consider the case of general bounded domains of class  $C^1$  in  $\mathbb{R}^2$ .

### **3.3.1** The case of the ball in $\mathbb{R}^2$

Through all this subsection, we consider problem (3.1.2) when  $\Omega = B$  is the unit ball in  $\mathbb{R}^2$  centered at zero. We denote by  $\mathcal{R}_{\rho}[u]$  the Rayleigh quotient of a function  $u \in H^1(B)$ :

$$\mathcal{R}_{\rho}[u] = \frac{\int_{B} |\nabla u|^2 dx dy}{\int_{\partial B} \rho u^2 d\sigma},$$

where  $\rho \in \mathcal{R}^{\mathcal{S}}$ . From (3.1.10), we have the following variational representation of the first positive eigenvalue of (3.1.2)

$$\lambda_2[\rho] = \min_{\substack{0 \neq u \in H^1(B) \\ \int_{\partial B} \rho u d\sigma = 0}} \mathcal{R}_{\rho}[u].$$
(3.3.1)

It is convenient to use polar coordinates  $(r, \theta) \in \mathbb{R}_+ \times \partial B$  in  $\mathbb{R}^2$  and the corresponding change of variables  $x = r \cos(\theta), y = r \sin(\theta)$ . Given M > 0, we define the family of densities  $\{\rho_{\varepsilon}\}_{\varepsilon \in ]0, \varepsilon_0[} \subset L^2(\partial B)$  written in polar coordinates as follows:

$$\rho_{\varepsilon}(\theta) := \begin{cases} \varepsilon, & \text{if } \theta \in [\varepsilon, \pi - \varepsilon] \cup [\pi + \varepsilon, 2\pi - \varepsilon], \\ \frac{M - \varepsilon(2\pi - 4\varepsilon)}{4\varepsilon}, & \text{if } \theta \in [0, \varepsilon[\cup]\pi - \varepsilon, \pi + \varepsilon[\cup]2\pi - \varepsilon, 2\pi[, \end{cases} (3.3.2) \end{cases}$$

for all  $\varepsilon \in ]0, \varepsilon_0[$  with  $\varepsilon_0$  sufficiently small. Note that  $\int_{\partial B} \rho_{\varepsilon} d\sigma = M$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . The densities  $\rho_{\varepsilon}$  are piecewise constant and concentrate in a neighborhood of two antipodal points, while they vanish in all the other points, as  $\varepsilon \to 0$ . We have the following theorem.

**Theorem 3.3.3.** Let  $\rho_{\varepsilon}$  be defined by (3.3.2). Let  $\lambda_2[\rho_{\varepsilon}]$  be the first positive eigenvalue of problem (3.1.2) with  $\rho = \rho_{\varepsilon}$  on the unit ball B in  $\mathbb{R}^2$ . Then

$$\lim_{\varepsilon \to 0} \lambda_2[\rho_\varepsilon] = 0. \tag{3.3.4}$$

*Proof.* Let  $B^+$  be the ball of radius 1 centered at (1,0),  $B^-$  the ball of radius 1 centered at (-1,0),  $B_{\varepsilon}^+$  the ball of radius  $\sqrt{2-2\cos(\varepsilon)}$  centered at (1,0) and  $B_{\varepsilon}^-$  the ball of radius  $\sqrt{2-2\cos(\varepsilon)}$  centered at (-1,0). We introduce the family of trial functions  $u_{\varepsilon}$  (see Figure 3.2) given by

$$u_{\varepsilon}(x,y) := \begin{cases} -\ln(\sqrt{(1-x)^2 + y^2}), & \text{if } (x,y) \in (B \cap B^+) \setminus B_{\varepsilon}^+, \\ -\ln(\varepsilon), & \text{if } (x,y) \in B \cap B_{\varepsilon}^+, \\ \ln(\sqrt{(1+x)^2 + y^2}), & \text{if } (x,y) \in (B \cap B^-) \setminus B_{\varepsilon}^-, \\ \ln(\varepsilon), & \text{if } (x,y) \in B \cap B_{\varepsilon}^-, \\ 0, & \text{if } (x,y) \in B \setminus (B^+ \cup B^-). \end{cases}$$

By construction  $u_{\varepsilon} \in H^1(B)$  and by symmetry  $\int_{\partial B} \rho_{\varepsilon} u_{\varepsilon} d\sigma = 0$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . Hence,  $u_{\varepsilon}$  is a suitable trial function for (3.3.1).

We have

$$\lambda_1[\rho_\varepsilon] \le \mathcal{R}_{\rho_\varepsilon}[u_\varepsilon],\tag{3.3.5}$$

for all  $\varepsilon \in ]0, \varepsilon_0[$ . Note that  $|u_{\varepsilon}|, |\nabla u_{\varepsilon}|$  and  $\rho_{\varepsilon}$  are symmetric with respect to the x and the y axes. Then we compute the integrals appearing in  $\mathcal{R}_{\rho_{\varepsilon}}[u_{\varepsilon}]$  restricted to  $B \cap \{(x, y) : x > 0, y > 0\}$ . We consider the numerator of  $\mathcal{R}_{\rho_{\varepsilon}}[u_{\varepsilon}]$ 



Figure 3.2: Trial function  $u_{\varepsilon}$  with  $\varepsilon = 0.05$ .

first. We have

$$\int_{B \cap \{x > 0, y > 0\}} |\nabla u_{\varepsilon}|^2 dx dy = \int_{B \cap B^+ \cap \{y > 0\}} |\nabla u_{\varepsilon}|^2 dx dy$$
$$\leq \int_{B^+ \cap \{x < 1, y > 0\}} |\nabla \tilde{u}_{\varepsilon}|^2 dx dy, \quad (3.3.6)$$

where the first equality holds since  $u_{\varepsilon} = 0$  in  $B \setminus (B^+ \cup B^-)$ , and the last inequality follows from the fact that we are integrating a positive function on a larger set. Here we have denoted by  $\tilde{u}_{\varepsilon}$  the function on  $B^+$  defined by  $\tilde{u}_{\varepsilon}(x,y) = -\ln(\sqrt{(1-x)^2 + y^2})$  for  $2 - 2\cos(\varepsilon) \leq (1-x)^2 + y^2 \leq 1$ and  $\tilde{u}_{\varepsilon}(x,y) = -\ln(\varepsilon)$  for  $(1-x)^2 + y^2 < 2 - 2\cos(\varepsilon)$ . Now we use the polar coordinates  $(r,\theta)$  with respect to the point (1,0) and the corresponding change of variables  $(x,y) = \psi(r,\theta) = (1 + r\cos(\theta), r\sin(\theta))$ , with  $r \in [0, +\infty[, \theta \in [0, 2\pi[$ . In this new coordinates  $\tilde{u}_{\varepsilon}(\psi(r,\theta)) = -\ln(r)$ for  $\sqrt{2 - 2\cos(\varepsilon)} < r < 1$  and  $\tilde{u}_{\varepsilon}(\psi(r,\theta)) = -\ln(\varepsilon)$ , for  $0 < r\sqrt{2 - 2\cos(\varepsilon)}$ . In this new coordinates the right-hand side of (3.3.6) is written as

$$\begin{split} \int_{B^+ \cap \{x < 1, y > 0\}} |\nabla \tilde{u}_{\varepsilon}|^2 dx dy \\ &= \int_{\frac{\pi}{2}}^{\pi} \int_{\sqrt{2-2\cos(\varepsilon)}}^{1} \frac{1}{r} dr d\theta = -\frac{\pi}{2} \ln(\sqrt{2-2\cos(\varepsilon)}). \end{split}$$

We note that

$$\ln(\sqrt{2 - 2\cos(\varepsilon)}) = \ln(\varepsilon) + O(\varepsilon^2), \qquad (3.3.7)$$

as  $\varepsilon \to 0$ . Now we consider the denominator of  $\mathcal{R}_{\rho_{\varepsilon}}[u_{\varepsilon}]$ . We have

$$\begin{split} \int_{\partial B \cap \{x > 0, y > 0\}} \rho_{\varepsilon} u^2 d\sigma &= \int_0^{\varepsilon} \frac{M - \varepsilon (2\pi - 4\varepsilon)}{4\varepsilon} (\ln \varepsilon)^2 d\theta \\ &+ \int_{\varepsilon}^{\frac{\pi}{3}} \varepsilon \left[ \ln \left( 2 \sin \left( \frac{\theta}{2} \right) \right) \right]^2 d\theta \\ &= \left( \frac{M - \varepsilon (2\pi - 4\varepsilon)}{4} \right) (\ln \varepsilon)^2 + \varepsilon \int_{2\sin\left(\frac{\varepsilon}{2}\right)}^1 \frac{(\ln s)^2}{\sqrt{1 - \frac{s^2}{4}}} ds \\ &\geq \left( \frac{M - \varepsilon (2\pi - 4\varepsilon)}{4} \right) (\ln \varepsilon)^2 + \frac{\varepsilon}{\cos\left(\frac{\varepsilon}{2}\right)} \int_{2\sin\left(\frac{\varepsilon}{2}\right)}^1 (\ln s)^2 ds \\ &= \left( \frac{M - \varepsilon (2\pi - 4\varepsilon)}{4} \right) (\ln \varepsilon)^2 \\ &+ \frac{\varepsilon}{\cos\left(\frac{\varepsilon}{2}\right)} \left[ 2 - 4\sin\left(\frac{\varepsilon}{2}\right) + 4\sin\left(\frac{\varepsilon}{2}\right) \ln\left(2\sin\left(\frac{\varepsilon}{2}\right)\right) \right] \\ &- 2\sin\left(\frac{\varepsilon}{2}\right) \left[ \ln\left(2\sin\left(\frac{\varepsilon}{2}\right)\right) \right]^2 \right]. \end{split}$$

We note that

$$\frac{\varepsilon}{\cos\left(\frac{\varepsilon}{2}\right)} \left[ 2 - 4\sin\left(\frac{\varepsilon}{2}\right) + 4\sin\left(\frac{\varepsilon}{2}\right) \ln\left(2\sin\left(\frac{\varepsilon}{2}\right)\right) -2\sin\left(\frac{\varepsilon}{2}\right) \left[\ln\left(2\sin\left(\frac{\varepsilon}{2}\right)\right)\right]^2 \right] = 2\varepsilon + O(\varepsilon^2 (\ln \varepsilon)^2),$$

as  $\varepsilon \to 0$ . From (3.3.7) and (3.3.7), it follows that

$$\frac{\int_{B\cap\{x>0,y>0\}} |\nabla u_{\varepsilon}|^2 dx dy}{\int_{\partial B\cap\{x>0,y>0\}} \rho_{\varepsilon} u^2 d\sigma} \le -\frac{\pi \left(\ln(\varepsilon) + O(\varepsilon^2)\right)}{2 \left(\frac{(M-\varepsilon(2\pi-4\varepsilon))}{4} (\ln\varepsilon)^2 + 2\varepsilon + O(\varepsilon^2(\ln\varepsilon)^2)\right)}$$

as  $\varepsilon \to 0$ . Then, from (3.3.5) and using a Taylor expansion, we have

$$\lambda_1[\rho_{\varepsilon}] \le \frac{2\pi}{M|\ln(\varepsilon)|} + \frac{4\pi\varepsilon \left(4 - \pi(\ln\varepsilon)^2\right)}{M^2(\ln\varepsilon)^3} + O\left(\frac{\varepsilon^2}{\ln(\varepsilon)}\right),$$

as  $\varepsilon \to 0$ , which yields (3.3.4). Moreover we have an upper bound for the rate of convergence of  $\lambda_1[\rho_{\varepsilon}]$  to zero, as  $\varepsilon \to 0$  (see Figure 3.3).

**Remark 3.3.8.** We note that a basis of the harmonic functions on the unit ball B in  $\mathbb{R}^2$  is given (in polar coordinates) by

1, 
$$\ln(r)$$
,  $r^l \cos(l\theta)$ ,  $r^l \sin(l\theta)$ ,  $r^{-l} \cos(l\theta)$ ,  $r^{-l} \sin(l\theta)$ ,



Figure 3.3: Plot of  $\frac{2\pi}{M|\ln(\varepsilon)|} + \frac{4\pi\varepsilon \left(4-\pi(\ln\varepsilon)^2\right)}{M^2(\ln\varepsilon)^3}$  with  $M = 2\pi$  and  $\varepsilon \in ]0, 0.2[$ .

for  $l \in \mathbb{N}, l \geq 1$ . When we consider problem (3.1.2), we require that the solutions are regular in the interior of the domain. Therefore we consider only the functions  $1, r^l \cos(l\theta), r^l \sin(l\theta), l \in \mathbb{N}, l \geq 1$ . In the case of problem (3.1.2) with density given by (3.3.2), the coefficient  $\rho_{\varepsilon}$  vanishes everywhere but in two points, say  $p^+, p^-$ , where it blows up, as  $\varepsilon \to 0$ . Therefore, intuitively the harmonic functions which better mimic the behavior of an eigenfunction near these two points are exactly  $\ln(r)$ ,  $r^{-l}\cos(l\theta)$ ,  $r^{-l}\sin(l\theta)$ , translated in such a way that the singularity occurs at the points where the coefficient blows up. In particular, we note that the harmonic function  $\ln(r)$ corresponds to the value l = 0, i.e., it is candidate to be the "second eigenfunction" of the zero eigenvalue of the limiting problem. Intuitively, the limiting problem (whatever it means), has a zero eigenvalue of multiplicity 2, and the eigenspace is spanned by the constant and by a suitable translations of  $\ln(r)$ . By following this heuristic intuition, one could guess what are the test functions to be used in Rayleigh quotient for the first positive eigenvalue in the case of the unit ball B in  $\mathbb{R}^N$ , for  $N \geq 3$ .

## **3.3.2** The case of the ball in $\mathbb{R}^N$ with $N \geq 3$

Thorugh this subsection we denote by B be the unit ball in  $\mathbb{R}^N$  centered at zero, with  $N \geq 3$ . We introduce the spherical coordinates  $(r, \theta) =$  $(r, \theta_1, ..., \theta_{N-1}) \in \mathbb{R}_+ \times \partial B$  in  $\mathbb{R}^N$  centered at  $x_0 = (x_{1,0}, x_{2,0}, ..., x_{N,0}) \in \mathbb{R}^N$ , and the corresponding change of variables given by

$$x_{1} = x_{1,0} + r \cos(\theta_{1}),$$
  

$$x_{2} = x_{2,0} + r \sin(\theta_{1}) \cos(\theta_{2}),$$
  

$$x_{3} = x_{3,0} + r \sin(\theta_{1}) \sin(\theta_{2}) \cos(\theta_{3}),$$
  

$$\vdots$$
  

$$x_{N-1} = x_{N-1,0} + r \sin(\theta_{1}) \cdots \sin(\theta_{N-2}) \cos(\theta_{N-1})$$

$$x_N = x_{N,0} + r\sin(\theta_1)\cdots\sin(\theta_{N-2})\sin(\theta_{N-1}).$$

The spherical volume element in these coordinates is given by

$$d\sigma = r^{N-1}\sin(\theta_1)^{N-2}\sin(\theta_2)^{N-3}\cdots\sin(\theta_{N-2})drd\theta_1d\theta_2\cdots d\theta_{N-1}.$$

We define the subsets of  $\mathbb{R}^N,\,B^+,B^-,B_{\varepsilon^+},B_{\varepsilon}^-$  as follows

$$\begin{cases} B^+ := B((1,0,...,0),1), \\ B^- := B((-1,0,...,0),1), \\ B^+_\varepsilon := B((1,0,...,0),\varepsilon), \\ B^-_\varepsilon := B((-1,0,...,0),\varepsilon). \end{cases}$$

We introduce the family of densities  $\{\rho_{\varepsilon}\}_{\varepsilon\in]0,\varepsilon_0[} \subset L^2(\partial\Omega)$  defined by

$$\rho_{\varepsilon}(x) := \begin{cases} \varepsilon, & \text{if } x \in \partial B \setminus (B_{\varepsilon}^{+} \cup B_{\varepsilon}^{-}), \\ \frac{M - \varepsilon (N\omega_{N} - 2|C_{\varepsilon}|)}{2|C_{\varepsilon}|}, & \text{if } x \in \partial B \cap (B_{\varepsilon}^{+} \cup B_{\varepsilon}^{-}), \end{cases}$$
(3.3.9)

where M > 0 is fixed,  $N\omega_N$  is the N - 1-dimensional measure of  $\partial B$  and  $C_{\varepsilon} := \partial B \cap B_{\varepsilon}^+$ . Note that  $|\partial B \cap B_{\varepsilon}^+| = |\partial B \cap B_{\varepsilon}^-| = |C_{\varepsilon}|$ . By construction  $\int_{\partial B} \rho_{\varepsilon} d\sigma = M$  for all  $\varepsilon \in ]0, \varepsilon_0[$ , for a suitable  $\varepsilon_0 > 0$ . We have the following theorem.

**Theorem 3.3.10.** Let  $\rho_{\varepsilon}$  be defined by (3.3.9). Let  $\lambda_2[\rho_{\varepsilon}]$  be the first positive eigenvalue of problem (3.1.2) with  $\rho = \rho_{\varepsilon}$  on the unit ball B in  $\mathbb{R}^N$ . Then

$$\lim_{\varepsilon \to 0} \lambda_2[\rho_\varepsilon] = 0. \tag{3.3.11}$$

*Proof.* Let  $u_{\varepsilon}$  be the function on B defined by

$$u_{\varepsilon}(x) := \begin{cases} \frac{1}{\left(\sqrt{(x_1 - 1)^2 + x_2^2 + \dots + x_N^2}\right)^{N-2}} - 1, & \text{if } x \in B \cap (B^+ \setminus B_{\varepsilon}^+), \\ \frac{1}{\varepsilon^{N-2}} - 1, \text{if } x \in B \cap B_{\varepsilon}^+, \\ 1 - \frac{1}{\left(\sqrt{(x_1 + 1)^2 + x_2^2 + \dots + x_N^2}\right)^{N-2}}, & \text{if } x \in B \cap (B^- \setminus B_{\varepsilon}^-), \\ 1 - \frac{1}{\varepsilon^{N-2}}, & \text{if } x \in B \cap B_{\varepsilon}^-, \\ 0, & \text{if } x \in B \setminus (B^+ \cup B^-). \end{cases}$$

By construction,  $u_{\varepsilon} \in H^1(B)$  and  $\int_{\partial B} \rho_{\varepsilon} u_{\varepsilon} d\sigma = 0$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . Hence  $u_{\varepsilon}$  is a suitable trial function for the Rayleigh quotient of  $\lambda_2[\varepsilon]$ . Indeed we have

$$\lambda_2[\rho_\varepsilon] \le \mathcal{R}_{\rho_\varepsilon}[u_\varepsilon],\tag{3.3.12}$$

for all  $\varepsilon \in [0, \varepsilon_0[$ . By the symmetry of  $|u_{\varepsilon}|, |\nabla u_{\varepsilon}|$  and  $\rho_{\varepsilon}$  we can consider the integrals appearing in the Rayleigh quotients restricted on  $B \cap \{x_1 > 0\}$ . We consider the numerator of  $\mathcal{R}_{\rho_{\varepsilon}}[u_{\varepsilon}]$  first and use the spherical coordinates centered in (1, 0, ..., 0). We have

$$\int_{B \cap \{x_1 > 0\}} |\nabla u_{\varepsilon}|^2 dx \leq \frac{N(N-2)^2 \omega_N}{2} \int_{\varepsilon}^1 r^{1-N} dr$$
$$= \frac{N(N-2)\omega_N}{2} \left(\frac{1}{\varepsilon^{N-2}} - 1\right), \quad (3.3.13)$$

where we extended the function  $u_{\varepsilon}$  to the whole of  $B^+$  as we did in the proof of Theorem 3.3.3. Now we consider the denominator of  $\mathcal{R}_{\rho_{\varepsilon}}[u_{\varepsilon}]$ . We have

$$\int_{\partial B \cap \{x_1 > 0\}} \rho_{\varepsilon} u_{\varepsilon}^2 d\sigma = \frac{M - \varepsilon (N\omega_N - 2|C_{\varepsilon}|)}{2} \left(\frac{1}{\varepsilon^{N-2}} - 1\right)^2 + \varepsilon \int_{\partial B \cap (B^+ \setminus B_{\varepsilon}^+)} \left(\frac{1}{\left(\sqrt{(x_1 - 1)^2 + x_2^2 + \dots + x_N^2}\right)^{N-2}} - 1\right)^2 d\sigma. \quad (3.3.14)$$

We need to estimate the second term in the right-hand side of (3.3.14). It is straightforward to see that

$$\varepsilon \int_{\partial B \cap (B^+ \setminus B_{\varepsilon}^+)} \left( \frac{1}{\left( \sqrt{(x_1 - 1)^2 + x_2^2 + \dots + x_N^2} \right)^{N-2}} - 1 \right)^2 d\sigma$$
  

$$\geq \varepsilon \left( \frac{1}{\varepsilon^{N-2}} - 1 \right)^2 |\partial B \cap (B^+ \setminus B_{\varepsilon}^+)|$$
  

$$= |\partial B \cap B^+| \left( \frac{1}{\varepsilon^{N-2}} - 1 \right)^2 \varepsilon - |C_{\varepsilon}| \cdot \left( \frac{1}{\varepsilon^{N-2}} - 1 \right)^2 \varepsilon \quad (3.3.15)$$

We note that  $|C_{\varepsilon}| \in O(\varepsilon^{N-1})$  as  $\varepsilon \to 0$ . From (3.3.12), (3.3.13), (3.3.14) and (3.3.15) we have

$$\lambda_2[\rho_{\varepsilon}] \le \frac{N(N-2)\omega_N}{2\left(\frac{1}{\varepsilon^{N-2}}-1\right)\left(\frac{M-\varepsilon(N\omega_N-2|C_{\varepsilon}|)}{2}+\varepsilon|\partial B\cap B^+|-\varepsilon|C_{\varepsilon}|\right)}.$$
 (3.3.16)

We perform a Taylor expansion of the right-hand side of (3.3.16) and obtain

$$\lambda_{2}[\rho_{\varepsilon}] \leq \varepsilon^{N-2} \left( \frac{N(N-2)\omega_{N}}{M} - \frac{N(N-2)\omega_{N}(2|\partial B \cap B^{+}| - N\omega_{N})\varepsilon}{M^{2}} + O(\varepsilon^{2}) \right),$$

which yields formula (3.3.11). Moreover, we have an upper bound for the rate of convergence of  $\lambda_2[\varepsilon]$  which depends on M and N (see Figure 3.4).  $\Box$ 



Figure 3.4: Plot of  $\varepsilon^{N-2}\left(\frac{N(N-2)\omega_N}{M}\right)$  with  $M = \omega_N, \varepsilon \in ]0, 0.2[$  and N = 3 (Blue), N = 4 (Red), N = 5 (Orange).

## **3.3.3** The case of an arbitrary $\Omega \subset \mathbb{R}^2$

In this subsection we consider a bounded domain  $\Omega$  in  $\mathbb{R}^2$  of class  $C^1$ . We shall consider only the case of  $\mathbb{R}^2$  since the computations are less involved. The result for N > 2 can be obtained by following the same scheme (see also Subsection 3.3.2). Through this subsection we denote by  $x = (x_1, x_2)$  an element of  $\mathbb{R}^2$ .

Let  $x_+, x_- \in \partial \Omega$  be such that

$$d(x_+, x_-) \ge d(x_1, x_2), \quad \forall x_1, x_2 \in \partial\Omega.$$

The distance function d from  $\mathbb{R}^2$  to  $\mathbb{R}$  is continuous and  $\partial \Omega \times \partial \Omega$  is a compact set, therefore there it has a maximum. We write  $x_+ = (x_{+,1}, x_{+,2})$  and  $x_- = (x_{-,1}, x_{-,2})$ . We set

$$d^* := \frac{d(x_+, x_-)}{2}.$$

We introduce the subsets  $B_*^+, B_*^-, B_{\varepsilon}^+, B_{\varepsilon}^-$  of  $\mathbb{R}^2$  defined in the following way (see Figure 3.5):

$$\begin{cases} B_*^+ := B(x_+, d^*), \\ B_*^- := B(x_-, d^*), \\ B_{\varepsilon}^+ := B(x_+, \varepsilon), \\ B_{\varepsilon}^- := B(x_-, \varepsilon). \end{cases}$$

Let M > 0 be fixed. For all  $\varepsilon \in ]0, \varepsilon_0[$  with  $\varepsilon_0$  small enough, we introduce the family  $\{\rho_{\varepsilon}\}_{\varepsilon \in ]0, \varepsilon_0[} \subset L^2(\partial\Omega)$  defined by

$$\rho_{\varepsilon}(x) := \begin{cases} \varepsilon, & \text{if } x \in \partial\Omega \setminus (B_*^+ \cup B_*^-), \\ \rho_+, & \text{if } x \in \partial\Omega \cap B_{\varepsilon}^+, \\ \rho_-, & \text{if } x \in \partial\Omega \cap B_{\varepsilon}^-, \end{cases}$$
(3.3.17)



Figure 3.5: The regions  $\Omega, B_{\varepsilon}^+, B_{\varepsilon}^-, \Omega \cap B_*^+, \Omega \cap B_*^-$ .

where  $\rho_+, \rho_-$  are suitable constants depending on M and  $\varepsilon$  which solve the following linear system

$$\begin{cases} |\partial\Omega \cap B_{\varepsilon}^{+}|\rho_{+} + |\partial\Omega \cap B_{\varepsilon}^{-}|\rho_{-} + |\partial\Omega \setminus (B_{\varepsilon}^{+} \cup B_{\varepsilon}^{-})|\varepsilon = M, \\ |\partial\Omega \cap B_{\varepsilon}^{+}|\rho_{+} - |\partial\Omega \cap B_{\varepsilon}^{-}|\rho_{-} = 0. \end{cases}$$
(3.3.18)

Note that the determinant of the matrix associated with system (3.3.18) is given by  $-(|\partial\Omega \cap B_{\varepsilon}^+||\partial\Omega \cap B_{\varepsilon}^-|)^2$  and is different from zero for all  $\varepsilon \in ]0, \varepsilon_0[$ . Moreover,  $M - |\partial\Omega \setminus (B_{\varepsilon}^+ \cup B_{\varepsilon}^-)| \varepsilon \neq 0$  for all  $\varepsilon \in ]0, \varepsilon_0[$  for  $\varepsilon_0$  small enough. We solve system (3.3.3) and obtain

$$\rho_{+} = \frac{M - |\partial \Omega \setminus (B_{\varepsilon}^{+} \cup B_{\varepsilon}^{-})|\varepsilon}{2|\partial \Omega \cap B_{\varepsilon}^{+}|}$$
$$\rho_{-} = \frac{M - |\partial \Omega \setminus (B_{\varepsilon}^{+} \cup B_{\varepsilon}^{-})|\varepsilon}{2|\partial \Omega \cap B_{\varepsilon}^{-}|}.$$

By construction,  $\int_{\partial\Omega} \rho_{\varepsilon} d\sigma = M$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . We have the following theorem.

**Theorem 3.3.19.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^2$  of class  $C^1$  and let  $\rho_{\varepsilon}$  be defined by (3.3.17). Let  $\lambda_2[\rho_{\varepsilon}]$  be the first positive eigenvalue of problem (3.1.2) with  $\rho = \rho_{\varepsilon}$  on  $\Omega$ . Then

$$\lim_{\varepsilon \to 0} \lambda_2[\rho_\varepsilon] = 0. \tag{3.3.20}$$

*Proof.* Let the function  $u_{\varepsilon}^+$  be defined by

$$u_{\varepsilon}^{+}(x) := \begin{cases} -\ln\left(\frac{\varepsilon}{d^{*}}\right), & \text{if } x \in \Omega \cap B_{\varepsilon}^{+}, \\ -\ln\left(\frac{|x-x_{+}|}{d^{*}}\right), & \text{if } x \in \Omega \cap (B_{*}^{+} \setminus B_{\varepsilon}^{+}). \end{cases}$$

Let  $\alpha(\varepsilon) \in \mathbb{R}$  be defined by

$$\alpha(\varepsilon) := \frac{|\partial\Omega \cap B_{\varepsilon}^{+}|\rho_{+}\ln\left(\frac{\varepsilon}{d^{*}}\right) + \varepsilon \int_{\partial\Omega \cap (B_{*}^{+}\setminus B_{\varepsilon}^{+})}\ln\left(\frac{|x-x_{+}|}{d^{*}}\right)d\sigma}{|\partial\Omega \cap B_{\varepsilon}^{-}|\rho_{-}\ln\left(\frac{\varepsilon}{d^{*}}\right) + \varepsilon \int_{\partial\Omega \cap (B_{*}^{-}\setminus B_{\varepsilon}^{-})}\ln\left(\frac{|x-x_{-}|}{d^{*}}\right)d\sigma},$$

for  $\varepsilon \in ]0, \varepsilon_0[$ . We define the function  $u_{\varepsilon}^-$  by

$$u_{\varepsilon}^{-}(x) := \begin{cases} \alpha(\varepsilon) \ln\left(\frac{\varepsilon}{d^{*}}\right), & \text{if } x \in \Omega \cap B_{\varepsilon}^{-}, \\ \alpha(\varepsilon) \ln\left(\frac{|x-x_{-}|}{d^{*}}\right), & \text{if } x \in \Omega \cap (B_{*}^{-} \setminus B_{\varepsilon}^{-}). \end{cases}$$

Let the function  $u_{\varepsilon}$  from  $\Omega$  to  $\mathbb{R}$  be defined by

$$u_{\varepsilon}(x) := \begin{cases} u_{\varepsilon}^+(x), & \text{if } x \in \Omega \cap (B_*^+), \\ u_{\varepsilon}^-(x), & \text{if } x \in \Omega \cap (B_*^-), \\ 0, & \text{if } x \in \Omega \setminus (B_*^+ \cup B_*^-). \end{cases}$$

By definition  $\int_{\partial\Omega} \rho_{\varepsilon} u_{\varepsilon} = 0$  for all  $\varepsilon \in ]0, \varepsilon_0[$ , where  $\varepsilon_0$  is small enough.



Figure 3.6: Trial function  $u_{\varepsilon}$  with  $\varepsilon = 0.05$ .

We need more information on the coefficient  $\alpha(\varepsilon)$ . We note that  $\alpha(\varepsilon) \ge 0$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . Moreover, we note that

$$\begin{aligned} \alpha(\varepsilon) &\leq \frac{|\partial \Omega \cap B_{\varepsilon}^{+}|\rho_{+}\ln\left(\frac{\varepsilon}{d^{*}}\right)}{|\partial \Omega \cap B_{\varepsilon}^{-}|\rho_{-}\ln\left(\frac{\varepsilon}{d^{*}}\right) + |\partial \Omega \cap (B_{*}^{-} \setminus B_{\varepsilon}^{-})|\varepsilon\ln\left(\frac{\varepsilon}{d^{*}}\right)} \\ &\leq \frac{|\partial \Omega \cap B_{\varepsilon}^{+}|\rho_{+}\ln\left(\frac{\varepsilon}{d^{*}}\right)}{|\partial \Omega \cap B_{\varepsilon}^{-}|\rho_{-}\ln\left(\frac{\varepsilon}{d^{*}}\right)} + O(\varepsilon\ln(\varepsilon)) = 1 + O(\varepsilon\ln(\varepsilon)), \end{aligned}$$

as  $\varepsilon \to 0$ . Hence, if  $\varepsilon_0$  is small enough,  $\alpha(\varepsilon)$  is strictly positive and bounded away from zero and infinity, uniformly in  $\varepsilon \in ]0, \varepsilon_0[$ . Moreover  $\lim_{\varepsilon \to 0} \alpha(\varepsilon) =$ 1, see the second equation in (3.3.18). Then  $u_{\varepsilon}$  is well-defined and belongs to  $H^1(\Omega)$ . Therefore  $u_{\varepsilon}$  is a suitable test function. We have

$$\begin{split} \lambda_{2}[\rho_{\varepsilon}] \\ &\leq \frac{\int_{\Omega \cap B_{\varepsilon}^{+}} |\nabla u_{\varepsilon}|^{2} dx + \int_{\Omega \cap B_{\varepsilon}^{-}} |\nabla u_{\varepsilon}|^{2} dx}{\int_{\partial \Omega \cap B_{\varepsilon}^{+}} \rho_{\varepsilon} u_{\varepsilon}^{2} d\sigma + \int_{\partial \Omega \cap B_{\varepsilon}^{-}} \rho_{\varepsilon} u_{\varepsilon}^{2} d\sigma + \varepsilon \int_{\partial \Omega \cap (B_{\varepsilon}^{+} \setminus B_{\varepsilon}^{-})} u_{\varepsilon}^{2} d\sigma} \\ &\leq \frac{\int_{B_{\varepsilon}^{+}} |\nabla u_{\varepsilon}|^{2} dx + \int_{B_{\varepsilon}^{-}} |\nabla u_{\varepsilon}|^{2} dx}{\left(\rho_{+} |\partial \Omega \cap B_{\varepsilon}^{+}| + \alpha(\varepsilon)^{2} \rho_{-} |\partial \Omega \cap B_{\varepsilon}^{-}|\right) \ln \left(\frac{\varepsilon}{d^{*}}\right)^{2} + \varepsilon \int_{\partial \Omega \cap (B_{\varepsilon}^{+} \setminus B_{\varepsilon}^{+})} u_{\varepsilon}^{2} d\sigma + \varepsilon \int_{\partial \Omega \cap (B_{\varepsilon}^{-} \setminus B_{\varepsilon}^{-})} u_{\varepsilon}^{2} d\sigma} \\ &\leq \frac{(\alpha(\varepsilon)^{2} + 1) \int_{B_{\varepsilon}^{+}} |\nabla u_{\varepsilon}|^{2} dx}{\left(\rho_{+} |\partial \Omega \cap B_{\varepsilon}^{+}| + \rho_{-} \alpha(\varepsilon)^{2} |\partial \Omega \cap B_{\varepsilon}^{-}|\right) \ln \left(\frac{\varepsilon}{d^{*}}\right)^{2}} \\ &= -\frac{2\pi \left(\alpha(\varepsilon)^{2} + 1\right) \ln \left(\frac{\varepsilon}{d^{*}}\right)}{\left(\frac{M - |\partial \Omega \setminus (B_{\varepsilon}^{+} \cup B_{\varepsilon}^{-})|\varepsilon\right)} \left(\alpha(\varepsilon)^{2} + 1\right) \ln \left(\frac{\varepsilon}{d^{*}}\right)^{2}} \\ &= \frac{4\pi}{|\ln \left(\frac{\varepsilon}{d^{*}}\right)| \left(M - |\partial \Omega \setminus (B_{\varepsilon}^{+} \cup B_{\varepsilon}^{-})|\right)}. \end{split}$$
(3.3.21)

We use a suitable Taylor's expansion in (3.3.21) and obtain

$$\lambda_2[\rho_{\varepsilon}] \leq \frac{4\pi}{M|\log\left(\frac{\varepsilon}{d^*}\right)|} + \frac{4\pi\varepsilon|\partial\Omega\setminus\left(B_{\varepsilon}^+\cup B_{\varepsilon}^-\right)|}{M^2|\ln\left(\frac{\varepsilon}{d^*}\right)|} + O\left(\frac{\varepsilon^2}{\ln(\varepsilon)}\right),$$

as  $\varepsilon \to 0$  (see Figure 3.7). This yields formula (3.3.20).





Figure 3.7: Plot of  $\frac{4\pi}{M \log(\frac{\varepsilon}{d^*})}$  with  $M = 2.9, d^* = 1, \varepsilon \in ]0, 0.2[$ .

## 3.4 On the optimization of the first positive Dirichlet and Neumann eigenvalues

In this section we consider the eigenvalue problem for the Laplace operator subject to Dirichlet and Neumann boundary conditions and with density  $\rho \in \mathcal{R}$ , where  $\mathcal{R}$  is given by (3.1.11). We show that there exist densities preserving the total mass such that the first positive Dirichlet eigenvalue is arbitrarily close to zero or arbitrary large. Also, we prove that there exist densities preserving the total mass such that the first positive Neumann eigenvalue is arbitrarily close to zero. The maximization of the first Neumann eigenvalue among all densities preserving the total mass seems to be a more complicated issue. Actually, we have been informed that this problem has been solved by means of techniques from differential geometry. The answer is that there exists a uniform upper bound for the Neumann eigenvalues under the constraint that the mass is fixed. It is still unclear if such an upper bound is attained or not. We refer to the paper [30] for a geometric approach to the study of the spectrum of the Laplacian.

#### 3.4.1 Optimization of the first Dirichlet eigenvalue

Let  $\Omega$  be a domain in  $\mathbb{R}^N$  with finite measure,  $\rho \in \mathcal{R}$  and M > 0 be a fixed number. We consider the Dirichlet eigenvalue problem for the Laplace operator

$$\begin{cases} -\Delta u = \lambda \rho u, & \text{in } \Omega, \\ u = 0, & \text{on } \partial \Omega, \end{cases}$$
(3.4.1)

in the unknowns u (the eigenfunction) and  $\lambda$  (the eigenvalue). It is wellknown that for all  $\rho \in \mathcal{R}$ , problem (3.4.1) admits a diverging sequence of positive eigenvalues of finite multiplicity

$$0 < \lambda_1[\rho] < \lambda_2[\rho] \leq \cdots \leq \lambda_j[\rho] \leq \cdots$$

The first eigenvalue  $\lambda_1[\rho]$  is positive and simple, and an eigenfunction associated with  $\lambda_1[\rho]$  does not change sign in  $\Omega$ .

The aim of this subsection is to prove that there exist sequences  $\hat{\rho}_{\varepsilon}$  and  $\check{\rho}_{\varepsilon}$ in  $\mathcal{R}$  such that  $\int_{\Omega} \hat{\rho}_{\varepsilon} dx = \int_{\Omega} \check{\rho}_{\varepsilon} dx = M$  for all  $\varepsilon \in ]0, \varepsilon_0[$ , and  $\lambda_1[\hat{\rho}_{\varepsilon}] \to +\infty$ ,  $\lambda_1[\check{\rho}_{\varepsilon}] \to 0$ , as  $\varepsilon \to 0$ . Thus, the problems

$$\max_{\substack{\rho \in \mathcal{R} \\ \int_{\Omega} \rho dx = M}} \lambda_1[\rho]$$

and

$$\min_{\substack{\rho \in \mathcal{R} \\ \int_{\Omega} \rho dx = M}} \lambda_1[\rho]$$

have no solutions.

We start with the problem of the maximization. In this case we need the additional assumption that  $\Omega$  is a bounded domain of class  $C^2$ . Let  $\hat{\rho}_{\varepsilon} = \rho_{\varepsilon}$ , where  $\rho_{\varepsilon}$  is defined by (3.1.20). We note that  $\int_{\Omega} \hat{\rho}_{\varepsilon} dx = M$  for all  $\varepsilon \in ]0, \varepsilon_0[$ , for a suitable  $\varepsilon_0 > 0$  small. We have the following theorem.

**Theorem 3.4.2.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^2$ . Let  $\hat{\rho}_{\varepsilon}$  be defined by (3.1.20). Let  $\lambda_1[\hat{\rho}_{\varepsilon}]$  be the first eigenvalue of problem (3.4.1) with  $\rho = \hat{\rho}_{\varepsilon}$ . Then

$$\lim_{\varepsilon \to 0} \lambda_1[\hat{\rho}_{\varepsilon}] = +\infty$$

*Proof.* For all  $\varepsilon \in ]0, \varepsilon_0[$ , with  $\varepsilon_0 > 0$  small enough, let  $u_{\varepsilon} \in H_0^1(\Omega)$  be the unique eigenfunction associated with  $\lambda_1[\hat{\rho}_{\varepsilon}]$  such that  $\int_{\Omega} \hat{\rho}_{\varepsilon} u_{\varepsilon}^2 dx = 1$ . Then it holds

$$\lambda_1[\hat{\rho}_{\varepsilon}] = \int_{\Omega} |\nabla u_{\varepsilon}|^2 dx,$$

for all  $\varepsilon \in ]0, \varepsilon_0[$ . Assume by contradiction that there exists a constant C > 0which does not depend on  $\varepsilon > 0$ , such that  $\lambda_1[\hat{\rho}_{\varepsilon}] \leq C$ , for all  $\varepsilon \in ]0, \varepsilon_0[$ . Thus,  $\int_{\Omega} |\nabla u_{\varepsilon}|^2 dx \leq C$  and from the Poincaré inequality,  $||u_{\varepsilon}||_{L^2(\Omega)} \leq C'$ , for a constant C' > 0 which does not depend on  $\varepsilon > 0$ . Then there exists  $u \in H_0^1(\Omega)$  such that, possibly passing to a subsequence,  $u_{\varepsilon} \rightharpoonup u$  in  $H_0^1(\Omega)$ , as  $\varepsilon \to 0$ , and  $u_{\varepsilon} \to u$  in  $L^2(\Omega)$ , as  $\varepsilon \to 0$ . From Lemma 3.1.22 it follows that  $\lim_{\varepsilon \to 0} \int_{\Omega} \hat{\rho}_{\varepsilon} u_{\varepsilon}^2 dx = \frac{M}{|\partial\Omega|} \int_{\partial\Omega} u^2 d\sigma = 0$ , which is a contradiction with the fact that  $\int_{\Omega} \hat{\rho}_{\varepsilon} u_{\varepsilon}^2 dx = 1$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . Thus  $\lambda_1[\hat{\rho}_{\varepsilon}] \to +\infty$  as  $\varepsilon \to 0$ . This concludes the proof of the theorem.



Figure 3.8: Eigenvalues of problem (3.4.1) on the unit ball in  $\mathbb{R}^2$  with  $\rho = \hat{\rho}_{\varepsilon}, M = \pi, \varepsilon \in ]0, 1[.$ 

Now we consider the problem of the minimization of the first Dirichlet eigenvalue. Let  $\Omega$  be a domain in  $\mathbb{R}^N$  with finite measure. Let  $x_0 \in \Omega$  and  $B_0$  be a ball centered at  $x_0$  such that  $\overline{B}_0 \subset \Omega$ . We denote by  $r_0$  the radius of  $B_0$ . For all  $\varepsilon \in ]0, r_0/2[$ , let  $B_{\varepsilon}$  the ball of radius  $\varepsilon$  centered at  $x_0$ . Let the function  $\check{\rho}_{\varepsilon} \in L^{\infty}(\Omega)$  be defined by

$$\check{\rho}_{\varepsilon}(x) := \begin{cases} \frac{M - \varepsilon(|\Omega| - \varepsilon^N \omega_N)}{\varepsilon^N \omega_N}, & \text{if } x \in B_{\varepsilon}, \\ \varepsilon, & \text{if } x \in \Omega \setminus \overline{B}_{\varepsilon}. \end{cases}$$
(3.4.3)

By definition,  $\int_{\Omega} \check{\rho}_{\varepsilon} dx = M$  for all  $\varepsilon \in ]0, r_0/2[$ .



Figure 3.9: Test function  $u_{\varepsilon}$  on  $\Omega \subset \mathbb{R}^2$ .

We have the following theorem.

**Theorem 3.4.4.** Let  $\Omega$  be a domain in  $\mathbb{R}^N$  with finite measure. Let  $\check{\rho}_{\varepsilon}$  be defined by (3.4.3). Let  $\lambda_1[\check{\rho}_{\varepsilon}]$  be the first eigenvalue of problem (3.4.1) with  $\rho = \check{\rho}_{\varepsilon}$ . Then

$$\lim_{\varepsilon \to 0} \lambda_1[\check{\rho}_{\varepsilon}] = 0. \tag{3.4.5}$$

*Proof.* We distinguish the case N = 2 and  $N \ge 3$ . We start by proving (3.4.5) in the case N = 2. Let  $u_{\varepsilon} \in H_0^1(\Omega)$  be defined by

$$u_{\varepsilon}(x) := \begin{cases} -\ln\left(\frac{\varepsilon}{r_0}\right), & \text{if } x \in B_{\varepsilon}, \\ -\ln\left(\frac{|x-x_0|}{r_0}\right), & \text{if } x \in B_0 \setminus \overline{B}_{\varepsilon}, \\ 0, & \text{if } x \in \Omega \setminus \overline{B}_0. \end{cases}$$

By definition,  $u_{\varepsilon} \in H_0^1(\Omega)$  for all  $\varepsilon \in ]0, r_0/2[$ . From the Min-Max principle we have that

$$\lambda_1[\check{\rho}_{\varepsilon}] \le \frac{\int_{\Omega} |\nabla u_{\varepsilon}|^2 dx}{\int_{\Omega} \check{\rho}_{\varepsilon} u_{\varepsilon}^2 dx}.$$
(3.4.6)

It is convenient to use standard polar coordinates  $(r, \theta) \in \mathbb{R}_+ \times \partial B$  in  $\mathbb{R}^2$ with respect to the point  $x_0$ . We have for the numerator in (3.4.6)

$$\int_{\Omega} |\nabla u_{\varepsilon}|^2 dx = \int_0^{2\pi} \int_{\varepsilon}^{r_0} \frac{1}{r} dr d\theta = 2\pi \ln\left(\frac{r_0}{\varepsilon}\right).$$
(3.4.7)

For the denominator we have

$$\int_{\Omega} \check{\rho}_{\varepsilon} u_{\varepsilon}^{2} dx$$

$$= \varepsilon \int_{0}^{2\pi} \int_{\varepsilon}^{r_{0}} \left( \ln\left(\frac{r}{r_{0}}\right) \right)^{2} r dr d\theta + \frac{M - \varepsilon(|\Omega| - \pi\varepsilon^{2})}{\pi\varepsilon^{2}} \int_{0}^{2\pi} \int_{0}^{\varepsilon} \left( \ln\left(\frac{\varepsilon}{r_{0}}\right) \right)^{2} r dr d\theta$$

$$= \frac{\pi}{2} \left( r_{0}^{2} - \varepsilon^{2} - 2\varepsilon^{2} \ln\left(\frac{r_{0}}{\varepsilon}\right) (1 + \ln\left(\frac{r_{0}}{\varepsilon}\right)) \right) \varepsilon + \left(M - \varepsilon(|\Omega| - \pi\varepsilon^{2})\right) \left( \ln\left(\frac{\varepsilon}{r_{0}}\right) \right)^{2}.$$
(3.4.8)

Therefore, from (3.4.6), (3.4.7) and (3.4.8) we have

$$\lambda_{1}[\tilde{\rho}_{\varepsilon}] \leq \frac{2\pi \ln\left(\frac{r_{0}}{\varepsilon}\right)}{\frac{\pi}{2} \left(r_{0}^{2} - \varepsilon^{2} - 2\varepsilon^{2} \ln\left(\frac{r_{0}}{\varepsilon}\right) \left(1 + \ln\left(\frac{r_{0}}{\varepsilon}\right)\right)\right)\varepsilon + \left(M - \varepsilon(|\Omega| - \pi\varepsilon^{2})\right) \left(\ln\left(\frac{\varepsilon}{r_{0}}\right)\right)^{2}}.$$
(3.4.9)

The right-hand side of (3.4.9) clearly goes to zero as  $\varepsilon \to 0$ . In fact, we can make a Taylor expansion and obtain that the right-hand side of (3.4.9) equals

$$\frac{2\pi}{M\left|\ln\left(\varepsilon/r_{0}\right)\right|}+O\left(\frac{\varepsilon}{\left|\ln(\varepsilon)\right|}\right),$$

as  $\varepsilon \to 0$ . This concludes the proof in the case N = 2.

Consider now the case  $N \geq 3$ . Let  $u_{\varepsilon} \in H_0^1(\Omega)$  be defined by

$$u_{\varepsilon}(x) := \begin{cases} \frac{1}{\varepsilon^{N-2}} - \frac{1}{r_0^{N-2}}, & \text{if } x \in B_{\varepsilon}, \\ \frac{1}{|x-x_0|^{N-2}} - \frac{1}{r_0^{N-2}}, & \text{if } x \in B_0 \setminus \overline{B}_{\varepsilon}, \\ 0, & \text{if } x \in \Omega \setminus \overline{B}_0. \end{cases}$$

By definition,  $u_{\varepsilon} \in H_0^1(\Omega)$  for all  $\varepsilon \in ]0, r_0/2[$ . From the Min-Max principle,  $\lambda_1[\check{\rho}_{\varepsilon}]$  satisfies the inequality (3.4.6).

66

It is convenient to use standard spherical coordinates  $(r, \theta) \in \mathbb{R}_+ \times \partial B$ in  $\mathbb{R}^N$  with respect to the point  $x_0$ . We have for the numerator in (3.4.6)

$$\int_{\Omega} |\nabla u_{\varepsilon}|^2 dx$$
  
= 
$$\int_{\partial B} \int_{\varepsilon}^{r_0} ((2-N)r^{1-N})^2 r^{N-1} dr d\sigma = N\omega_N (N-2)(\varepsilon^{2-N} - r_0^{2-N}).$$
  
(3.4.10)

For the denominator we have

$$\int_{\Omega} \check{\rho}_{\varepsilon} u_{\varepsilon}^{2} dx = \varepsilon \int_{\partial B} \int_{\varepsilon}^{r_{0}} \left( \frac{1}{r^{N-2}} - \frac{1}{r_{0}^{N-2}} \right)^{2} r^{N-1} dr d\sigma$$

$$+ \frac{M - \varepsilon(|\Omega| - \omega_{N} \varepsilon^{N})}{\omega_{N} \varepsilon^{N}} \int_{\partial B} \int_{0}^{\varepsilon} \left( \frac{1}{\varepsilon^{N-2}} - \frac{1}{r_{0}^{N-2}} \right)^{2} r^{N-1} dr d\sigma$$

$$= \omega_{N} \left( Nr_{0}^{2-N} \varepsilon^{2} + \frac{N}{N-4} \varepsilon^{4-N} - r_{0}^{4-N} \varepsilon^{N} - \frac{(N-2)^{2}}{N-4} r_{0}^{4-N} \right) \varepsilon$$

$$+ \left( M - \varepsilon(|\Omega| - \varepsilon^{N} \omega_{N}) \right) (r_{0}^{2-N} - \varepsilon^{2-N}), \quad (3.4.11)$$

if  $N \geq 3$  and  $N \neq 4$ , while

$$\int_{\Omega} \check{\rho}_{\varepsilon} u_{\varepsilon}^{2} dx = \varepsilon \int_{\partial B} \int_{\varepsilon}^{r_{0}} \left( \frac{1}{r^{2}} - \frac{1}{r_{0}^{2}} \right)^{2} r^{3} dr d\sigma + \frac{M - \varepsilon (|\Omega| - \omega_{4} \varepsilon^{4})}{\omega_{4} \varepsilon^{4}} \int_{\partial B} \int_{0}^{\varepsilon} \left( \frac{1}{\varepsilon^{2}} - \frac{1}{r_{0}^{2}} \right)^{2} r^{3} dr d\sigma = \frac{(r_{0}^{2} - \varepsilon^{2})((r_{0}^{2} - \varepsilon^{2})(M - \varepsilon |\Omega|) - 2r_{0}^{2} \varepsilon^{5} \omega_{4})}{r_{0}^{4} \varepsilon^{4}} - 4\varepsilon \omega_{4} \ln \left( \frac{\varepsilon}{r_{0}} \right), \quad (3.4.12)$$

if N = 4. Therefore, from (3.4.6), (3.4.10), (3.4.11) and (3.4.12), we have

$$\lambda_{1}[\tilde{\rho}_{\varepsilon}] \leq \frac{N\omega_{N}(N-2)(\varepsilon^{2-N}-r_{0}^{2-N})}{\omega_{N}\left(Nr_{0}^{2-N}\varepsilon^{2}+\frac{N}{N-4}\varepsilon^{4-N}-r_{0}^{4-N}\varepsilon^{N}-\frac{(N-2)^{2}}{N-4}r_{0}^{4-N}\right)\varepsilon+\left(M-\varepsilon(|\Omega|-\varepsilon^{N}\omega_{N})\right)(r_{0}^{2-N}-\varepsilon^{2-N})}$$

$$(3.4.13)$$

if  $N \geq 3$ ,  $N \neq 4$ , and

$$\lambda_1[\check{\rho}_{\varepsilon}] \le \frac{8\omega_4(\varepsilon^{-2} - r_0^{-2})}{\frac{(r_0^2 - \varepsilon^2)((r_0^2 - \varepsilon^2)(M - \varepsilon|\Omega|) - 2r_0^2 \varepsilon^5 \omega_4)}{r_0^4 \varepsilon^4} - 4\varepsilon \omega_4 \ln\left(\frac{\varepsilon}{r_0}\right)},$$
(3.4.14)

if N = 4.

The right-hand side of both (3.4.13) and (3.4.14) clearly goes to zero as

 $\varepsilon \to 0.$  In fact, we can make a Taylor expansion and obtain that the right-hand side of both (3.4.13) and (3.4.14) equals

$$\frac{N(N-2)}{M}\varepsilon^{N-2} + O(\varepsilon^{N-1}),$$

as  $\varepsilon \to 0$ . This concludes the proof in the case  $N \ge 3$ .



(a)  $\lambda_1[\check{\rho}_{\varepsilon}]$  on the unit ball in  $\mathbb{R}^2$ , (b)  $\lambda_1[\check{\rho}_{\varepsilon}]$  (red) and  $\frac{2}{|\ln(\varepsilon)|}$  (blue),  $M = \pi, \ \varepsilon \in ]0, 1[.$   $\varepsilon \in ]0, 0.1[.$ 

#### 3.4.2 Minimization of the first positive Neumann eigenvalue

Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$  and M > 0 be a fixed number. We consider the eigenvalue problem (3.1.12). We recall that the first eigenvalue  $\lambda_1[\rho]$  of problem (3.1.12) is zero, and the corresponding eigenfunctions are the constant functions on  $\Omega$ . The second eigenvalue  $\lambda_2[\rho]$  is positive. The aim of this subsection is to prove that there exists a sequence  $\rho_{\varepsilon} \in \mathcal{R}$  such that  $\int_{\Omega} \rho_{\varepsilon} dx = M$  for all  $\varepsilon \in ]0, \varepsilon_0[$ , for a suitable  $\varepsilon_0 > 0$ , and such that  $\lambda_2[\rho_{\varepsilon}] \to 0$  as  $\varepsilon \to 0$ . Thus the problem

$$\min_{\substack{\rho \in \mathcal{R} \\ \int_{\Omega} \rho dx = M}} \lambda_2[\rho]$$

has no solutions.

Let  $x_1, x_2 \in \Omega$ ,  $x_1 \neq x_2$ , and let  $r_0 > 0$  be such that  $\overline{B}_1 \subset \Omega$ ,  $\overline{B}_2 \subset \Omega$ and  $B_1 \cap B_2 = \emptyset$ , where  $B_1$  and  $B_2$  are balls of radius  $r_0$  centered at  $x_1$ and  $x_2$ , respectively. Let  $\varepsilon \in ]0, r_0/2[$  and let  $B_{\varepsilon,1}, B_{\varepsilon,2}$  be the balls of radius  $\varepsilon$  centered at  $x_1$  and  $x_2$ , respectively. We introduce the function  $\rho_{\varepsilon} \in \mathcal{R}$ defined by

$$\rho_{\varepsilon}(x) := \begin{cases} \varepsilon, & \text{if } x \in \Omega \setminus \overline{(B_{\varepsilon,1} \cup B_{\varepsilon_2})}, \\ \frac{M - \varepsilon(|\Omega| - 2\varepsilon^N \omega_N)}{2\varepsilon^N \omega_N}, & \text{if } x \in B_{\varepsilon,1} \cup B_{\varepsilon,2}, \end{cases}$$
(3.4.15)

for all  $\varepsilon \in [0, r_0/2[$ . We note that  $\int_{\Omega} \rho_{\varepsilon} dx = M$  for all  $\varepsilon \in [0, r_0/2[$ . We consider problem (3.1.12) with  $\rho = \rho_{\varepsilon}$ . We have the following theorem.

**Theorem 3.4.16.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ . Let  $\rho_{\varepsilon}$  be defined by (3.4.15). Let  $\lambda_2[\rho_{\varepsilon}]$  be the second eigenvalue of problem (3.1.12) with  $\rho = \rho_{\varepsilon}$ . Then

$$\lim_{\varepsilon \to 0} \lambda_2[\rho_\varepsilon] = 0. \tag{3.4.17}$$

*Proof.* Let  $u_{\varepsilon} \in H^1(\Omega)$  be defined by

$$u_{\varepsilon}(x) := \begin{cases} -\ln\left(\frac{\varepsilon}{r_0}\right), & \text{if } x \in B_{\varepsilon,1}, \\ \ln\left(\frac{\varepsilon}{r_0}\right), & \text{if } x \in B_{\varepsilon,2}, \\ -\ln\left(\frac{|x-x_1|}{r_0}\right), & \text{if } x \in B_1 \setminus \overline{B}_{\varepsilon,1}, \\ \ln\left(\frac{|x-x_2|}{r_0}\right), & \text{if } x \in B_2 \setminus \overline{B}_{\varepsilon,2}, \\ 0, & \text{if } x \in \Omega \setminus \overline{(B_1 \cup B_2)}, \end{cases}$$

if N = 2, and by

$$u_{\varepsilon}(x) := \begin{cases} \frac{1}{\varepsilon^{N-2}} - \frac{1}{r_0^{N-2}}, & \text{if } x \in B_{\varepsilon,1}, \\ -\frac{1}{\varepsilon^{N-2}} + \frac{1}{r_0^{N-2}}, & \text{if } x \in B_{\varepsilon,2}, \\ \frac{1}{|x-x_1|^{N-2}} - \frac{1}{r_0^{N-2}}, & \text{if } x \in B_1 \setminus \overline{B}_{\varepsilon,1}, \\ -\frac{1}{|x-x_2|^{N-2}} + \frac{1}{r_0^{N-2}}, & \text{if } x \in B_2 \setminus \overline{B}_{\varepsilon,2}, \\ 0, & \text{if } x \in \Omega \setminus \overline{(B_1 \cup B_2)}, \end{cases}$$

if  $N \geq 3$ . Clearly  $u_{\varepsilon} \in H^1(\Omega)$  for all  $\varepsilon \in ]0, r_0/2[$ . Moreover, by definition,  $\int_{\Omega} \rho_{\varepsilon} u_{\varepsilon} dx = 0$  for all  $\varepsilon \in ]0, r_0/2[$ .



Figure 3.11: Test function  $u_{\varepsilon}$  on  $\Omega \subset \mathbb{R}^2$ .

We recall that by the Min-Max principle we have

$$\lambda_2[\rho_{\varepsilon}] = \min_{\substack{u \in H^1(\Omega) \\ \int_{\Omega} \rho_{\varepsilon} dx = 0}} \frac{\int_{\Omega} |\nabla u_{\varepsilon}|^2 dx}{\int_{\Omega} \rho_{\varepsilon} u_{\varepsilon}^2 dx}.$$

Then it follows that

$$\lambda_2[\rho_{\varepsilon}] \le \frac{\int_{\Omega} |\nabla u_{\varepsilon}|^2}{\int_{\Omega} \rho_{\varepsilon} u_{\varepsilon}^2 dx},$$

for all  $\varepsilon \in ]0, r_0/2[$ . Now the proof of (3.4.17) follows the same lines as the proof of formula (3.4.5) in Theorem (3.4.4), and is accordingly omitted. In particular, we prove that

$$\lim_{\varepsilon \to 0} \frac{\int_{\Omega} |\nabla u_{\varepsilon}|^2 dx}{\int_{\Omega} \rho_{\varepsilon} u_{\varepsilon}^2 dx} = 0,$$

70

which implies the validity of (3.4.17).

Mass concentration phenomena for second order operators

## Chapter 4

## Neumann and Steklov problems: an asymptotic analysis

As we have highlighted in Chapter 3 (Theorem 3.1.21 and Corollary 3.1.42), we can consider the Steklov eigenvalues of the Laplace operator as limiting Neumann eigenvalues in a problem of mass concentration at the boundary of a bounded domain of class  $C^2$  in  $\mathbb{R}^N$ . Through this chapter we shall use a slightly different notation for the eigenvalues and the eigenfunctions of problems (3.1.2) and (3.1.12) with respect to that used in Chapter 3. The notation used in this chapter is more convenient in view of the analysis which we shall carry out. We consider problem (3.1.2) with constant density  $\rho \equiv \frac{M}{|\partial \Omega|}$ , namely

$$\begin{cases} \Delta u = 0, & \text{in } \Omega, \\ \frac{\partial u}{\partial \nu} = \frac{M}{|\partial \Omega|} \mu u, & \text{on } \partial \Omega, \end{cases}$$
(4.0.1)

in the unknowns  $\mu$  (the eigenvalue) and u (the eigenfunction), where M > 0is a fixed constant. We consider also problem (3.1.12) with density  $\rho = \rho_{\varepsilon}$ , where  $\rho_{\varepsilon}$  is given by (3.1.20), namely

$$\begin{cases} -\Delta u = \lambda \rho_{\varepsilon} u, & \text{in } \Omega, \\ \frac{\partial u}{\partial \nu} = 0, & \text{on } \partial \Omega, \end{cases}$$
(4.0.2)

in the unknowns  $\lambda$  (the eigenvalue) and u (the eigenfunction).

In this chapter we discuss the asymptotic behavior of the Neumann eigenvalues and find explicit formulas for their derivatives at the limiting problem in the case of the open unit ball in  $\mathbb{R}^N$  (Section 4.1) and of general bounded domain of class  $C^2$  in  $\mathbb{R}^2$  (Section 4.2). In particular, in the case of the ball in  $\mathbb{R}^N$  we deduce that the Neumann eigenvalues have a monotone behavior in the limit and that Steklov eigenvalues locally minimize the Neumann eigenvalues.

The techniques used in Section 4.1 and Section 4.2 are completely different. In Section 4.1 we use the explicit form of the solutions of problem (4.0.2) on the unit ball in terms of Bessel functions. In this way problem (4.0.2) is recasted in the form of an equation  $F(\lambda, \varepsilon) = 0$ . Then we carry out the analysis of this implicit equation and find a formula for the derivative of the eigenvalues at  $\varepsilon = 0$ . In Section 4.2 we use classical techniques of asymptotic analysis in the spirit of [50, 52, 53]. In particular, we postulate an asymptotic expansion of the eigenelements of problem (4.0.2) and justify the expansion up to the first order.

We remark that the techniques used in Section 4.1 allow to overcome the problem of the multiplicity of the eigenvalues, which is usually an obstruction in the application of the techniques of asymptotic analysis. In fact, the analysis carried out in Section 4.1 involves all multiple eigenvalues. On the other hand, in Section 4.2 the techniques of asymptotic analysis which we used allow to deal only with simple eigenvalues.

## 4.1 The case of the ball in $\mathbb{R}^N$

Let *B* be the open unit ball centered at zero in  $\mathbb{R}^N$ ,  $N \ge 2$ . We denote by  $\omega_N$  the measure of *B*. Therefore, the (N-1)-dimensional measure of  $\partial B$  is given by  $N\omega_N$ . In this section we consider problem (4.0.1) with  $\Omega = B$ . As is well-known (see also Corollary 3.2.6) the eigenvalues of problem (4.0.1) on the unit ball with constant density  $\frac{M}{N\omega_N}$  are given explicitly by the sequence

$$\mu_l = \frac{N\omega_N}{M}l, \quad l \in \mathbb{N}_0, \tag{4.1.1}$$

and the eigenfunctions corresponding to  $\mu_l$  are the homogeneous harmonic polynomials of degree l. In particular, the multiplicity of  $\mu_l$  is (2l+N-2)(l+N-3)!/(l!(N-2)!), and only  $\mu_0$  is simple, the corresponding eigenfunctions being the constant functions.

As in Chapter 3, for any  $\varepsilon \in ]0,1[$ , we consider a mass density  $\rho_{\varepsilon}$  in the whole of *B* defined by (3.1.20). We recall that for any  $x \in B$  we have  $\rho_{\varepsilon}(x) \to 0$  as  $\varepsilon \to 0$ , and  $\int_{B} \rho_{\varepsilon} dx = M$  for all  $\varepsilon > 0$ , which means that the total mass *M* is fixed and concentrates at the boundary of *B* as  $\varepsilon \to 0$ . Then we consider problem (4.0.2) on *B* with  $\rho = \rho_{\varepsilon}$ .

The eigenvalues of (4.0.2) on B with density  $\rho_{\varepsilon}$  have finite multiplicity and form a sequence

$$\lambda_0(\varepsilon) < \lambda_1(\varepsilon) \le \lambda_2(\varepsilon) \le \cdots,$$

depending on  $\varepsilon$ , with  $\lambda_0(\varepsilon) = 0$ . Recall that by Corollary 3.1.42 we have that for any  $l \in \mathbb{N}_0$ 

$$\lambda_l(\varepsilon) \to \mu_l, \quad \text{as } \varepsilon \to 0,$$

$$(4.1.2)$$

(see also [10], [72]).

In this section we study the asymptotic behavior of  $\lambda_l(\varepsilon)$  as  $\varepsilon \to 0$ . Namely, we prove that such eigenvalues are continuously differentiable with respect to  $\varepsilon$  for  $\varepsilon \geq 0$  small enough, and that the following formula holds

$$\lambda_l'(0) = \frac{2l\mu_l}{3} + \frac{2\mu_l^2}{N(2l+N)}.$$
(4.1.3)

In particular, for  $l \neq 0$ ,  $\lambda'_l(0) > 0$  hence  $\lambda_l(\varepsilon)$  is strictly increasing and the Steklov eigenvalues  $\mu_l$  minimize the Neumann eigenvalues  $\lambda_l(\varepsilon)$  for  $\varepsilon$  small enough. We note that our analysis concerns all eigenvalues  $\mu_l$  with arbitrary indexes and multiplicity, and that we do not prove global monotonocity of  $\lambda_l(\varepsilon)$ , which in fact does not hold for any l; see Figures 4.1, 4.2.

In Subsection 4.1.1 we prove formula (4.1.3). The proof of the results relies on the use of Bessel functions which allows to recast problem (4.0.2) in the form of an equation  $F(\lambda, \varepsilon) = 0$  in the unknowns  $\lambda, \varepsilon$ . Then, after some preparatory work, it is possible to apply the Implicit Function Theorem and conclude. The application of this method requires suitable Taylor's expansions and estimates for the corresponding remainders, as well as recursive formulas for the cross-products of Bessel functions and their derivatives. The estimates for the remainders are contained in Subsection 4.1.2. In Subsection 4.1.3 we consider the case N = 1. In Subsection 4.1.4 we study the behavior of the eigenvalues and their derivatives under dilations of the domain. In Subsection 4.1.5 we establish recursive formulas for the cross-products of Bessel functions and their derivatives under dilations of the domain. In Subsection 4.1.5 we establish recursive formulas for the cross-products of Bessel functions and their derivatives used in Subsection 4.1.1.

### 4.1.1 Asymptotic behavior of Neumann eigenvalues

In the sequel we shall use the standard spherical coordinates  $(r, \theta) \in \mathbb{R}_+ \times \partial B$ in  $\mathbb{R}^N$ .

We note that, when  $d(x, \partial B) < \varepsilon$ , we have  $\rho_{\varepsilon}(x) = \frac{M - \varepsilon \omega_N (1 - \varepsilon)^N}{\omega_N (1 - (1 - \varepsilon)^N)}$ . To shorten notation, in what follows we will denote by a and b the quantities defined by

$$a := \sqrt{\lambda \varepsilon} (1 - \varepsilon), \text{ and } b := \sqrt{\lambda \tilde{\rho}_{\varepsilon}} (1 - \varepsilon),$$

where

$$\tilde{\rho}_{\varepsilon} := \frac{M - \varepsilon \omega_N \left(1 - \varepsilon\right)^N}{\omega_N \left(1 - \left(1 - \varepsilon\right)^N\right)}.$$

We have the following lemma.

**Lemma 4.1.4.** Given an eigenvalue  $\lambda$  of problem (4.0.2), a corresponding eigenfunction u is of the form  $u(r, \theta) = S_l(r)H_l(\theta)$  where  $H_l(\theta)$  is a spherical harmonic of some order  $l \in \mathbb{N}_0$  and

$$S_{l}(r) = \begin{cases} r^{1-\frac{N}{2}} J_{\nu_{l}}(\sqrt{\lambda\varepsilon}r), & \text{if } r < 1-\varepsilon, \\ r^{1-\frac{N}{2}} \left(\alpha J_{\nu_{l}}(\sqrt{\lambda\widetilde{\rho}\varepsilon}r) + \beta Y_{\nu_{l}}(\sqrt{\lambda\widetilde{\rho}\varepsilon}r)\right), & \text{if } 1-\varepsilon < r < 1, \end{cases}$$
(4.1.5)

where  $\nu_l := \frac{(N+2l-2)}{2}$  and  $\alpha$ ,  $\beta$  are given by

$$\alpha = \frac{\pi b}{2} \left( J_{\nu_l}(a) Y'_{\nu_l}(b) - \frac{a}{b} J'_{\nu_l}(a) Y_{\nu_l}(b) \right)$$
  
$$\beta = \frac{\pi b}{2} \left( \frac{a}{b} J_{\nu_l}(b) J'_{\nu_l}(a) - J'_{\nu_l}(b) J_{\nu_l}(a) \right).$$

*Proof.* Recall that the Laplace operator can be written in spherical coordinates as

$$\Delta = \partial_{rr} + \frac{N-1}{r}\partial_r + \frac{1}{r^2}\Delta_S,$$

where  $\Delta_S$  is the Laplace-Beltrami operator on the unit sphere  $\partial B$  of  $\mathbb{R}^N$ . In order to solve the equation  $-\Delta u = \lambda \rho_{\varepsilon} u$ , we separate variables so that  $u(r, \theta) = S(r)H(\theta)$ . Then using  $l(l+N-2), l \in \mathbb{N}_0$ , as separation constant, we obtain the equations

$$r^{2}S'' + r(N-1)S' + r^{2}\lambda\rho_{\varepsilon}S - l(l+N-2)S = 0$$
(4.1.6)

and

$$-\Delta_S H = l(l+N-2)H.$$
 (4.1.7)

By setting  $S(r) = r^{1-\frac{N}{2}}\tilde{S}(r)$  into (4.1.6), it follows that  $\tilde{S}(r)$  satisfies the Bessel equation

$$\tilde{S}'' + \frac{1}{r}\tilde{S}' + \left(\lambda\rho_{\varepsilon} - \frac{\nu_l^2}{r^2}\right)\tilde{S} = 0.$$

Since the solutions u of problem (4.0.2) are bounded on  $\Omega$  and  $Y_{\nu_l}(z)$  blows up at z = 0, it follows that for  $r < 1 - \varepsilon$ , S(r) is a multiple of the function  $r^{1-\frac{N}{2}}J_{\nu_l}(\sqrt{\lambda\varepsilon}r)$ . For  $1 - \varepsilon < r < 1$ , S(r) is a linear combination of the functions  $r^{1-\frac{N}{2}}J_{\nu_l}(\sqrt{\lambda\widetilde{\rho}\varepsilon}r)$  and  $r^{1-\frac{N}{2}}Y_{\nu_l}(\sqrt{\lambda\widetilde{\rho}\varepsilon}r)$ . On the other hand, the solutions of equation (4.1.7) are the spherical harmonics of order l. Thus ucan be written as in (4.1.5) for suitable values of  $\alpha, \beta \in \mathbb{R}$ .

Now it remains to compute the coefficient  $\alpha$  and  $\beta$  in (4.1.5). Since the right-hand side of the equation in (4.0.2) is a function in  $L^2(\Omega)$  then by standard regularity theory a solution u of (4.0.2) belongs to the standard Sobolev space  $H^2(\Omega)$ , hence  $\alpha$  and  $\beta$  must be chosen in such a way that u and  $\partial_r u$  are continuous at  $r = 1 - \varepsilon$ , that is

$$\begin{cases} \alpha J_{\nu_l}(\sqrt{\lambda\tilde{\rho}_{\varepsilon}}(1-\varepsilon)) + \beta Y_{\nu_l}(\sqrt{\lambda\tilde{\rho}_{\varepsilon}}(1-\varepsilon)) = J_{\nu_l}(\sqrt{\lambda\varepsilon}(1-\varepsilon)), \\ \alpha J_{\nu_l}'(\sqrt{\lambda\tilde{\rho}_{\varepsilon}}(1-\varepsilon)) + \beta Y_{\nu_l}'(\sqrt{\lambda\tilde{\rho}_{\varepsilon}}(1-\varepsilon)) = \sqrt{\frac{\varepsilon}{\tilde{\rho}_{\varepsilon}}} J_{\nu_l}'(\sqrt{\lambda\varepsilon}(1-\varepsilon)). \end{cases}$$

Solving the system we obtain

$$\alpha = \frac{J_{\nu_l}(a)Y'_{\nu_l}(b) - \frac{a}{b}J'_{\nu_l}(a)Y_{\nu_l}(b)}{J_{\nu_l}(b)Y'_{\nu_l}(b) - J'_{\nu_l}(b)Y_{\nu_l}(b)}, \quad \beta = \frac{\frac{a}{b}J_{\nu_l}(b)J'_{\nu_l}(a) - J'_{\nu_l}(b)J_{\nu_l}(a)}{J_{\nu_l}(b)Y'_{\nu_l}(b) - J'_{\nu_l}(b)Y_{\nu_l}(b)}$$

Note that  $J_{\nu_l}(b)Y'_{\nu_l}(b) - J'_{\nu_l}(b)Y_{\nu_l}(b)$  is the Wronskian in *b*, which is known to be  $\frac{2}{\pi b}$  (see Section 1.4, see also [1, §9]). This concludes the proof.  $\Box$ 

76

We are ready to establish an implicit characterization of the eigenvalues of (4.0.2).

**Lemma 4.1.8.** The nonzero eigenvalues  $\lambda$  of problem (4.0.2) are given implicitly as zeros of the equation

$$\left(1 - \frac{N}{2}\right)P_1(a,b) + \frac{b}{(1-\varepsilon)}P_2(a,b) = 0, \qquad (4.1.9)$$

where

$$P_{1}(a,b) = J_{\nu_{l}}(a) \left( Y_{\nu_{l}}'(b) J_{\nu_{l}}(\frac{b}{1-\varepsilon}) - J_{\nu_{l}}'(b) Y_{\nu_{l}}(\frac{b}{1-\varepsilon}) \right) + \frac{a}{b} J_{\nu_{l}}'(a) \left( J_{\nu_{l}}(b) Y_{\nu_{l}}(\frac{b}{1-\varepsilon}) - Y_{\nu_{l}}(b) J_{\nu_{l}}(\frac{b}{1-\varepsilon}) \right),$$

$$P_{2}(a,b) = J_{\nu_{l}}(a) \left( Y_{\nu_{l}}'(b) J_{\nu_{l}}'(\frac{b}{1-\varepsilon}) - J_{\nu_{l}}'(b) Y_{\nu_{l}}'(\frac{b}{1-\varepsilon}) \right) + \frac{a}{b} J_{\nu_{l}}'(a) \left( J_{\nu_{l}}(b) Y_{\nu_{l}}'(\frac{b}{1-\varepsilon}) - Y_{\nu_{l}}(b) J_{\nu_{l}}'(\frac{b}{1-\varepsilon}) \right).$$

*Proof.* By Lemma 4.1.4, an eigenfunction u associated with an eigenvalue  $\lambda$  is of the form  $u(r, \theta) = S_l(r)H_l(\theta)$ , where for  $r > 1 - \varepsilon$ 

$$\begin{split} S_{l}(r) &= \frac{\pi b}{2} r^{1-\frac{N}{2}} \left[ \left( J_{\nu_{l}}(a) Y_{\nu_{l}}'(b) - \frac{a}{b} J_{\nu_{l}}'(a) Y_{\nu_{l}}(b) \right) J_{\nu_{l}}(\frac{br}{1-\varepsilon}) \right. \\ &+ \left( \frac{a}{b} J_{\nu_{l}}(b) J_{\nu_{l}}'(a) - J_{\nu_{l}}'(b) J_{\nu_{l}}(a) \right) Y_{\nu_{l}}(\frac{br}{1-\varepsilon}) \right]. \end{split}$$

We require that  $\frac{\partial u}{\partial \nu} = \frac{\partial u}{\partial r}|_{r=1} = 0$ , which is true if and only if

$$\begin{split} \frac{\pi b}{2} \left( 1 - \frac{N}{2} \right) & \left[ \left( J_{\nu_l}(a) Y_{\nu_l}'(b) - \frac{a}{b} J_{\nu_l}'(a) Y_{\nu_l}(b) \right) J_{\nu_l}(\frac{b}{1 - \varepsilon}) \right. \\ & \left. + \left( \frac{a}{b} J_{\nu_l}(b) J_{\nu_l}'(a) - J_{\nu_l}'(b) J_{\nu_l}(a) \right) Y_{\nu_l}(\frac{b}{1 - \varepsilon}) \right] \\ & \left. + \frac{\pi b^2}{2(1 - \varepsilon)} \left[ \left( J_{\nu_l}(a) Y_{\nu_l}'(b) - \frac{a}{b} J_{\nu_l}'(a) Y_{\nu_l}(b) \right) J_{\nu_l}'(\frac{b}{1 - \varepsilon}) \right. \\ & \left. + \left( \frac{a}{b} J_{\nu_l}(b) J_{\nu_l}'(a) - J_{\nu_l}'(b) J_{\nu_l}(a) \right) Y_{\nu_l}'(\frac{b}{1 - \varepsilon}) \right] = 0. \end{split}$$

The previous equation can be clearly rewritten in the form (4.1.9).

We now prove the following lemma.

Lemma 4.1.10. Equation (4.1.9) can be written in the form

$$\lambda^{2} \varepsilon \left( \frac{M}{3N\omega_{N}} - \frac{1}{\nu_{l}(1+\nu_{l})} \right) + \lambda \varepsilon \left( \frac{N}{2} - \nu_{l} + \frac{(2-N)N\omega_{N}}{2\nu_{l}(1+\nu_{l})M} \right) - 2\lambda + \frac{2N\omega_{N}l}{M} - \frac{2N\omega_{N}l}{M} \left( \frac{N-1}{2} - \frac{\omega_{N}}{M} - \nu_{l} \right) \varepsilon + \mathcal{R}(\lambda,\varepsilon) = 0 \quad (4.1.11)$$

where  $\mathcal{R}(\lambda,\varepsilon) \in O(\varepsilon\sqrt{\varepsilon})$  as  $\varepsilon \to 0$ .

*Proof.* We plan to divide the left hand-side of (4.1.9) by  $J'_{\nu_l}(a)$  and to analyze the resulting terms using the known Taylor's series for Bessel functions. Note that  $J'_{\nu_l}(a) > 0$  for all  $\varepsilon$  small enough. We split our analysis into three steps.

Step 1. We consider the term  $\frac{P_2(a,b)}{J'_{\nu_l}(a)}$ , that is

$$\frac{J_{\nu_{l}}(a)}{J_{\nu_{l}}'(a)} \left[ Y_{\nu_{l}}'(b) J_{\nu_{l}}'(\frac{b}{1-\varepsilon}) - Y_{\nu_{l}}'(\frac{b}{1-\varepsilon}) J_{\nu_{l}}'(b) \right] \\
+ \frac{a}{b} \left[ Y_{\nu_{l}}'(\frac{b}{1-\varepsilon}) J_{\nu_{l}}(b) - Y_{\nu_{l}}(b) J_{\nu_{l}}'(\frac{b}{1-\varepsilon}) \right]. \quad (4.1.12)$$

Using Taylor's formula, we write the derivatives of the Bessel functions in (4.1.12), call them  $C'_{\nu_l}$ , as follows

$$\mathcal{C}_{\nu_{l}}^{\prime}\left(\frac{b}{1-\varepsilon}\right) = \mathcal{C}_{\nu_{l}}^{\prime}(b) + \mathcal{C}_{\nu_{l}}^{\prime\prime}(b)\frac{\varepsilon b}{1-\varepsilon} + \dots + \frac{\mathcal{C}_{\nu_{l}}^{(n)}(b)}{(n-1)!}\left(\frac{\varepsilon b}{1-\varepsilon}\right)^{n-1} + o\left(\frac{\varepsilon b}{1-\varepsilon}\right)^{n-1}.$$
(4.1.13)

Then, using (4.1.13) with n = 4 for  $J'_{\nu_l}$  and  $Y'_{\nu_l}$  we get

$$\frac{J_{\nu_{l}}(a)}{J_{\nu_{l}}'(a)} \left[ \frac{\varepsilon b}{1-\varepsilon} \left( Y_{\nu_{l}}'(b) J_{\nu_{l}}''(b) - J_{\nu_{l}}'(b) Y_{\nu_{l}}''(b) \right) \\
+ \frac{\varepsilon^{2} b^{2}}{2(1-\varepsilon)^{2}} \left( Y_{\nu_{l}}'(b) J_{\nu_{l}}'''(b) - J_{\nu_{l}}'(b) Y_{\nu_{l}}'''(b) \right) \\
+ \frac{\varepsilon^{3} b^{3}}{6(1-\varepsilon)^{3}} \left( Y_{\nu_{l}}'(b) J_{\nu_{l}}'''(b) - J_{\nu_{l}}'(b) Y_{\nu_{l}}'''(b) \right) + R_{1}(b) \right] \\
+ \frac{a}{b} \left[ \left( J_{\nu_{l}}(b) Y_{\nu_{l}}'(b) - Y_{\nu_{l}}(b) J_{\nu_{l}}'(b) \right) + \frac{\varepsilon b}{1-\varepsilon} \left( J_{\nu_{l}}(b) Y_{\nu_{l}}''(b) - Y_{\nu_{l}}(b) J_{\nu_{l}}''(b) \right) \\
+ \frac{\varepsilon^{2} b^{2}}{2(1-\varepsilon)^{2}} \left( J_{\nu_{l}}(b) Y_{\nu_{l}}''(b) - Y_{\nu_{l}}(b) J_{\nu_{l}}'''(b) - Y_{\nu_{l}}(b) J_{\nu_{l}}''(b) \right) + R_{2}(b) \right], \quad (4.1.14)$$

where  $R_1(b)$ ,  $R_2(b)$  are the appropriate remainders in the Taylor's formulas.

Let  $R_3$  be the remainder defined in Lemma 4.1.23. We set

$$\begin{split} R(\lambda,\varepsilon) &:= R_3(a) \left[ \frac{\varepsilon b}{1-\varepsilon} \left( Y_{\nu_l}'(b) J_{\nu_l}''(b) - J_{\nu_l}'(b) Y_{\nu_l}''(b) \right) \\ &+ \frac{\varepsilon^2 b^2}{2(1-\varepsilon)^2} \left( Y_{\nu_l}'(b) J_{\nu_l}'''(b) - J_{\nu_l}'(b) Y_{\nu_l}'''(b) \right) \\ &+ \frac{\varepsilon^3 b^3}{6(1-\varepsilon)^3} \left( Y_{\nu_l}'(b) J_{\nu_l}'''(b) - J_{\nu_l}'(b) Y_{\nu_l}'''(b) \right) \right] \\ &+ R_1(b) \left[ \frac{a}{\nu_l} + \frac{a^3}{2\nu_l^2(1+\nu_l)} \right] + R_2(b) \frac{a}{b} + R_3(a) R_1(b) . \end{split}$$

By Lemma 4.1.28, it turns out that  $R(\lambda, \varepsilon) \in O(\varepsilon^3)$  as  $\varepsilon \to 0$ . We also set

$$\begin{split} f(\varepsilon) &:= b_1^2(\varepsilon) a_1^3(\varepsilon) f_1(\varepsilon);\\ g(\varepsilon) &:= b_1^2(\varepsilon) a_1(\varepsilon) g_1(\varepsilon) + a_1^3(\varepsilon) g_2(\varepsilon);\\ h(\varepsilon) &:= a_1(\varepsilon) h_1(\varepsilon) + \varepsilon^2 \frac{a_1^3(\varepsilon)}{b_1^2(\varepsilon)} h_2(\varepsilon);\\ k(\varepsilon) &:= \frac{a_1(\varepsilon)}{b_1^2(\varepsilon)} k_1(\varepsilon), \end{split}$$

where

$$\begin{split} a_{1}(\varepsilon) &:= \frac{a}{\sqrt{\lambda\varepsilon}} = (1-\varepsilon); \\ b_{1}(\varepsilon) &:= b\sqrt{\frac{\varepsilon}{\lambda}}; \\ f_{1}(\varepsilon) &:= \frac{1}{6\nu_{l}^{2}(1+\nu_{l})(1-\varepsilon)^{3}}; \\ g_{1}(\varepsilon) &:= \frac{1}{3\nu_{l}(1-\varepsilon)^{3}}; \\ g_{2}(\varepsilon) &:= -\frac{1}{\nu_{l}^{2}(1+\nu_{l})(1-\varepsilon)} + \frac{\varepsilon}{2\nu_{l}^{2}(1+\nu_{l})(1-\varepsilon)^{2}} - \frac{\varepsilon^{2}(3+2\nu_{l}^{2})}{6\nu_{l}^{2}(1+\nu_{l})(1-\varepsilon)^{3}}; \\ h_{1}(\varepsilon) &:= -\frac{2}{\nu_{l}(1-\varepsilon)} + \frac{\varepsilon}{\nu_{l}(1-\varepsilon)^{2}} - \frac{\varepsilon^{2}(3+2\nu_{l}^{2})}{3\nu_{l}(1-\varepsilon)^{3}} - \frac{\varepsilon}{(1-\varepsilon)^{2}}; \\ h_{2}(\varepsilon) &:= \frac{1}{(1+\nu_{l})(1-\varepsilon)} - \frac{3\varepsilon}{2(1+\nu_{l})(1-\varepsilon)^{2}} + \frac{\varepsilon^{2}(\nu_{l}^{4}+11\nu_{l}^{2})}{6\nu_{l}^{2}(1+\nu_{l})(1-\varepsilon)^{3}}; \\ k_{1}(\varepsilon) &:= 2 + \frac{2\varepsilon\nu_{l}}{(1-\varepsilon)} - \frac{3\varepsilon^{2}\nu_{l}}{(1-\varepsilon)^{2}} + \frac{\varepsilon^{3}(\nu_{l}^{4}+11\nu_{l}^{2})}{3\nu_{l}(1-\varepsilon)^{3}} - \frac{2\varepsilon}{(1-\varepsilon)} + \frac{\varepsilon^{2}(2+\nu_{l}^{2})}{(1-\varepsilon)^{2}}. \end{split}$$

Note that functions f, g, h, k are continuous at  $\varepsilon = 0$  and  $f(0), g(0), h(0), k(0) \neq 0$ .

Using in (4.1.14) formula (4.1.24) for  $J_{\nu_l}(a)/J'_{\nu_l}(a)$  and the explicit formulas for the cross products of Bessel functions given by Lemma 4.1.41 and

Corollary 4.1.46, (4.1.12) can be written as

$$\frac{1}{\sqrt{\lambda}\pi}\varepsilon\sqrt{\varepsilon}k(\varepsilon) + \frac{\sqrt{\lambda}}{\pi}\varepsilon\sqrt{\varepsilon}h(\varepsilon) + \frac{\lambda\sqrt{\lambda}}{\pi}\varepsilon^2\sqrt{\varepsilon}g(\varepsilon) + \frac{\lambda^2\sqrt{\lambda}}{\pi}\varepsilon^3\sqrt{\varepsilon}f(\varepsilon) + R(\lambda,\varepsilon).$$
(4.1.15)

Step 2. We consider the quantity  $\frac{P_1(a,b)}{J'_{\nu_l}(a)}$ , that is

$$\frac{J_{\nu_{l}}(a)}{J_{\nu_{l}}'(a)} \left[ Y_{\nu_{l}}'(b) J_{\nu_{l}}(\frac{b}{1-\varepsilon}) - J_{\nu_{l}}'(b) Y_{\nu_{l}}(\frac{b}{1-\varepsilon}) \right] \\
+ \frac{a}{b} \left[ J_{\nu_{l}}(b) Y_{\nu_{l}}(\frac{b}{1-\varepsilon}) - Y_{\nu_{l}}(b) J_{\nu_{l}}(\frac{b}{1-\varepsilon}) \right]. \quad (4.1.16)$$

Proceeding as in Step 1 and setting

$$\begin{split} \tilde{f}(\varepsilon) &:= -\frac{a_1^3(\varepsilon)b_1(\varepsilon)}{2\pi\nu_l^2(1+\nu_l)(1-\varepsilon)^2};\\ \tilde{g}(\varepsilon) &:= \frac{a_1^3(\varepsilon)}{b_1(\varepsilon)} \left(\frac{1}{\pi\nu_l^2(1+\nu_l)} + \frac{\varepsilon^2}{2\pi(1+\nu_l)(1-\varepsilon)^2}\right) - \frac{a_1(\varepsilon)b_1(\varepsilon)}{\nu_l\pi(1-\varepsilon)^2};\\ \tilde{h}(\varepsilon) &:= \frac{a_1(\varepsilon)}{b_1(\varepsilon)} \left(\frac{2}{\nu_l\pi} + \frac{2\varepsilon}{\pi(1-\varepsilon)} + \frac{(\nu_l-1)}{\pi(1-\varepsilon)^2}\varepsilon^2\right), \end{split}$$

one can prove that (4.1.16) can be written as

$$\varepsilon \tilde{h}(\varepsilon) + \lambda \varepsilon^2 \tilde{g}(\varepsilon) + \lambda^2 \varepsilon^3 \tilde{f}(\varepsilon) + \hat{R}(\lambda, \varepsilon), \qquad (4.1.17)$$

where  $\hat{R}(\lambda, \varepsilon) \in O(\varepsilon^2 \sqrt{\varepsilon})$  as  $\varepsilon \to 0$ , see Lemma 4.1.28. Step 3. We combine (4.1.15) and (4.1.17) and rewrite equation (4.1.9) in the form

$$\varepsilon(1-\frac{N}{2})\tilde{h}(\varepsilon) + \varepsilon\frac{b_1(\varepsilon)k(\varepsilon)}{\pi(1-\varepsilon)} + \lambda\varepsilon^2(1-\frac{N}{2})\tilde{g}(\varepsilon) + \lambda\varepsilon\frac{b_1(\varepsilon)h(\varepsilon)}{\pi(1-\varepsilon)} + \lambda^2\varepsilon^3(1-\frac{N}{2})\tilde{f}(\varepsilon) + \lambda^2\varepsilon^2\frac{b_1(\varepsilon)g(\varepsilon)}{\pi(1-\varepsilon)} + \lambda^3\varepsilon^3\frac{b_1(\varepsilon)f(\varepsilon)}{\pi(1-\varepsilon)} + \mathcal{R}_0(\lambda,\varepsilon) = 0, \quad (4.1.18)$$

where

$$\mathcal{R}_0(\lambda,\varepsilon) = \frac{\sqrt{\lambda}b_1(\varepsilon)}{(1-\varepsilon)\sqrt{\varepsilon}}R(\lambda,\varepsilon) + \left(1-\frac{N}{2}\right)\hat{R}(\lambda,\varepsilon).$$

Note that  $\mathcal{R}_0(\lambda,\varepsilon) \in O(\varepsilon^2\sqrt{\varepsilon})$  as  $\varepsilon \to 0$ . Dividing by  $\varepsilon$  in (4.1.18) and setting  $\mathcal{R}_1(\lambda,\varepsilon) := \frac{\mathcal{R}_0(\lambda,\varepsilon)}{\varepsilon}$ , we obtain

$$(1 - \frac{N}{2})\tilde{h}(\varepsilon) + \frac{b_1(\varepsilon)k(\varepsilon)}{\pi(1 - \varepsilon)} + \lambda\varepsilon(1 - \frac{N}{2})\tilde{g}(\varepsilon) + \lambda\frac{b_1(\varepsilon)h(\varepsilon)}{\pi(1 - \varepsilon)} + \lambda^2\varepsilon^2(1 - \frac{N}{2})\tilde{f}(\varepsilon) + \lambda^2\varepsilon\frac{b_1(\varepsilon)g(\varepsilon)}{\pi(1 - \varepsilon)} + \lambda^3\varepsilon^2\frac{b_1(\varepsilon)f(\varepsilon)}{\pi(1 - \varepsilon)} + \mathcal{R}_1(\lambda, \varepsilon) = 0.$$

$$(4.1.19)$$
We now multiply in (4.1.19) by  $\frac{\pi\nu_l(1-\varepsilon)}{b_1(\varepsilon)}$  which is a positive quantity for all  $\varepsilon \in ]0,1[$ . Taking into account the definitions of functions  $g,h,k,\tilde{g},\tilde{h}$ , we can finally rewrite (4.1.19) in the form

$$\begin{split} \lambda^2 \varepsilon \left( \frac{\hat{\rho}(\varepsilon)}{3} - \frac{1}{\nu_l(1+\nu_l)} \right) + \lambda \varepsilon \left( \frac{N}{2} - \nu_l + \frac{2-N}{2\nu_l(1+\nu_l)\hat{\rho}(\varepsilon)} \right) \\ &- 2\lambda + \frac{2l\left(1+\varepsilon\nu_l\right)}{\hat{\rho}(\varepsilon)} + \mathcal{R}(\lambda,\varepsilon) = 0, \end{split}$$

where

$$\hat{\rho}(\varepsilon) := \varepsilon \tilde{\rho}(\varepsilon) = \frac{M - \omega_N \varepsilon (1 - \varepsilon)^N}{\omega_N \left(N - \frac{N(N-1)}{2}\varepsilon - \sum_{k=3}^N {N \choose k} (-1)^k \varepsilon^{k-1}\right)},$$

and  $\mathcal{R}(\lambda,\varepsilon) \in O(\varepsilon\sqrt{\varepsilon})$  as  $\varepsilon \to 0$ . The formulation in (4.1.11) can be easily deduced by observing that

$$\hat{\rho}_{\varepsilon} = \frac{M}{N\omega_N} + 2\frac{M}{N\omega_N} \left(\frac{N-1}{4} - \frac{\omega_N}{2M}\right)\varepsilon + O(\varepsilon^2), \text{ as } \varepsilon \to 0.$$

We are now ready to prove the main result of this section.

**Theorem 4.1.20.** All eigenvalues of problem (4.0.2) have the following asymptotic behavior

$$\lambda_l(\varepsilon) = \mu_l + \left(\frac{2l\mu_l}{3} + \frac{2\mu_l^2}{N(2l+N)}\right)\varepsilon + o(\varepsilon), \quad \text{as } \varepsilon \to 0, \tag{4.1.21}$$

where  $\mu_l$  are the eigenvalues of problem (4.0.1).

Moreover, for all  $l \in \mathbb{N}_0$  the functions defined by  $\lambda_l(\varepsilon)$  for  $\varepsilon > 0$  and  $\lambda_l(0) = \mu_l$ , are continuous in the whole of [0, 1[ and of class  $C^1$  in a neighborhood of  $\varepsilon = 0$ .

*Proof.* By using the Min-Max Principle and related standard arguments, one can easily prove that  $\lambda_l(\varepsilon)$  depends with continuity on  $\varepsilon > 0$  (cfr. formula (2.1.1), see also [70, 75]). Moreover, by using (4.1.2) the maps  $\varepsilon \mapsto \lambda_l(\varepsilon)$  can be extended by continuity at the point  $\varepsilon = 0$  by setting  $\lambda_l(0) = \mu_l$ .

In order to prove differentiability of  $\lambda_l(\varepsilon)$  around zero and the validity of (4.1.21), we consider equation (4.1.11) and apply the Implicit Function Theorem. Note that equation (4.1.11) can be written in the form  $F(\lambda, \varepsilon) = 0$  where F is a function of class  $C^1$  in the variables  $(\lambda, \varepsilon) \in ]0, \infty[\times[0, 1[, with$ 

$$F(\lambda, 0) = -2\lambda + \frac{2N\omega_N l}{M},$$
  

$$F'_{\lambda}(\lambda, 0) = -2,$$
  

$$F'_{\varepsilon}(\lambda, 0) = \lambda^2 \left(\frac{M}{3N\omega_N} - \frac{1}{\nu_l(1+\nu_l)}\right) + \lambda \left(\frac{N}{2} - \nu_l + \frac{(2-N)N\omega_N}{2\nu_l(1+\nu_l)M}\right)$$
  

$$-\frac{2N\omega_N l}{M} \left(\frac{N-1}{2} - \frac{\omega_N}{M} - \nu_l\right)$$

By (4.1.1),  $\mu_l = N\omega_N l/M$  hence  $F(\mu_l, 0) = 0$ . Since  $F'_{\lambda}(\mu_l, 0) \neq 0$ , the Implicit Function Theorem combined with the continuity of the functions  $\lambda_l(\cdot)$  allows to conclude that functions  $\lambda_l(\cdot)$  are of class  $C^1$  around zero.

We now compute the derivative of  $\lambda_l(\cdot)$  at zero. Using the equality  $N\omega_N/M = \mu_l/l$  and recalling that  $\nu_l = l + N/2 - 1$  we get

$$\begin{split} F_{\varepsilon}'(\mu_l,0) &= \mu_l^2 \left( \frac{l}{3\mu_l} - \frac{1}{\nu_l(1+\nu_l)} \right) \\ &+ \mu_l \left( 1 - l + \frac{\mu_l(2-N)}{2l\nu_l(1+\nu_l)} \right) - 2\mu_l \left( \frac{1}{2} - l - \frac{\mu_l}{Nl} \right) \\ &= \mu_l^2 \left( \frac{1}{\nu_l(1+\nu_l)} \left( \frac{2-N}{2l} - 1 \right) + \frac{2}{Nl} \right) + \frac{4}{3}\mu_l l \\ &= \frac{4\mu_l^2}{N^2 + 2Nl} + \frac{4}{3}\mu_l l. \end{split}$$

Finally, formula  $\lambda'_l(0) = -F'_{\varepsilon}(\mu_l, 0)/F'_{\lambda}(\mu_l, 0)$  yields (4.1.3) and the validity of (4.1.21).

**Corollary 4.1.22.** For any  $l \in \mathbb{N}$  there exists  $\delta_l > 0$  such that the function  $\lambda_l(\cdot)$  is strictly increasing in the interval  $[0, \delta_l[$ . In particular,  $\mu_l < \lambda_l(\varepsilon)$  for all  $\varepsilon \in ]0, \delta_l[$ .



Figure 4.1: Solution branches of equation (4.1.9) with N = 2,  $M = \pi$  in the region  $(\varepsilon, \lambda) \in ]0, 1[\times]0, 150[$ . The colors refer to the choice of l in (4.1.9), in particular blue (l = 0), red (l = 1), green (l = 2), purple (l = 3), orange (l = 4).



Figure 4.2: Solution branches of equation (4.1.9) with N = 2,  $M = \pi$  in the region  $(\varepsilon, \lambda) \in ]0, 1[\times]0, 50[$ . The colors refer to the choice of l in (4.1.9), in particular blue (l = 0), red (l = 1), green (l = 2), purple (l = 3), orange (l = 4), cyan (l = 5), pink (l = 6).

### 4.1.2 Estimates for the remainders

This subsection is devoted to the proof of a few technical estimates used in the proof of Lemma 4.1.10.

**Lemma 4.1.23.** The function  $R_3$  defined by

$$\frac{J_{\nu}(z)}{J_{\nu}'(z)} = \frac{z}{\nu} + \frac{z^3}{2\nu^2(1+\nu)} + R_3(z), \qquad (4.1.24)$$

is  $O(z^5)$  as  $z \to 0$ .

*Proof.* Recall the following well-known representation of the Bessel functions of the first species

$$J_{\nu}(z) = \left(\frac{z}{2}\right)^{\nu} \sum_{j=0}^{+\infty} \frac{(-1)^j}{j! \Gamma(j+\nu+1)} \left(\frac{z}{2}\right)^{2j}.$$
 (4.1.25)

For clarity, we simply write

$$J_{\nu}(z) = z^{\nu}(a_0 + a_2 z^2 + a_4 z^4 + O(z^5)), \qquad (4.1.26)$$

hence

$$J'_{\nu}(z) = z^{\nu-1}(\nu a_0 + (\nu+2)a_2z^2 + (\nu+4)a_4z^4 + O(z^5)), \qquad (4.1.27)$$

where the coefficients  $a_0, a_2, a_4$  are defined by (4.1.25). By (4.1.26), (4.1.27) and standard computations it follows that

$$\frac{J_{\nu}(z)}{J_{\nu}'(z)} = \frac{z}{\nu} - \frac{2a_2}{\nu^2 a_0} z^3 + O(z^5),$$

which gives exactly (4.1.24).

**Lemma 4.1.28.** For any  $\lambda > 0$  the remainders  $R(\lambda, \varepsilon)$  and  $\hat{R}(\lambda, \varepsilon)$  defined in the proof of Lemma 4.1.10 are  $O(\varepsilon^3)$ ,  $O(\varepsilon^2 \sqrt{\varepsilon})$ , respectively, as  $\varepsilon \to 0$ . Moreover, the same holds true for the corresponding partial derivatives  $\partial_{\lambda} R(\lambda, \varepsilon)$ ,  $\partial_{\lambda} \hat{R}(\lambda, \varepsilon)$ .

*Proof.* First, we consider  $R_3(a) = R_3(\sqrt{\lambda\varepsilon}(1-\varepsilon))$  where  $R_3$  is defined in Lemma 4.1.23 and we differentiate it with respect to  $\lambda$ . We obtain

$$\frac{\partial R_3(a)}{\partial \lambda} = \frac{a R_3'(a)}{2\lambda},$$

hence by Lemma 4.1.23 we can conclude that  $R_3(a)$  and  $\frac{\partial R_3(a)}{\partial \lambda}$  are  $O(\varepsilon^2 \sqrt{\varepsilon})$  as  $\varepsilon \to 0$ .

Now consider  $R_1(b)$  and  $R_2(b)$  defined in the proof of Lemma 4.1.10. Since  $\lambda > 0$ , we have that b > 0 hence the Bessel functions are analytic in b and we can write

$$R_1(b) = \sum_{k=4}^{+\infty} \frac{\varepsilon^k b^k}{k! (1-\varepsilon)^k} \left( Y'_{\nu}(b) J^{(k+1)}_{\nu}(b) - J'_{\nu}(b) Y^{(k+1)}_{\nu}(b) \right),$$

and

$$2\sqrt{\lambda} \frac{\partial R_1(b)}{\partial \lambda} = \frac{\varepsilon b_1(\varepsilon)}{\sqrt{\varepsilon}(1-\varepsilon)} \sum_{k=4}^{+\infty} \frac{b^{k-1}\varepsilon^{k-1}}{(k-1)!(1-\varepsilon)^{k-1}} \left( Y_{\nu}'(b)J_{\nu}^{(k+1)}(b) - J_{\nu}'(b)Y_{\nu}^{(k+1)}(b) \right) + \frac{b_1(\varepsilon)}{\sqrt{\varepsilon}} \sum_{k=4}^{+\infty} \frac{\varepsilon^k b^k}{k!(1-\varepsilon)^k} \left( Y_{\nu}'(b)J_{\nu}^{(k+1)}(b) - J_{\nu}'(b)Y_{\nu}^{(k+1)}(b) \right)'.$$

Using the fact that  $b = \sqrt{\lambda/\varepsilon} b_1(\varepsilon)$  and Lemma 4.1.41 we conclude that all the cross products of the form  $Y'_{\nu}(b)J^{(k+1)}_{\nu}(b) - J'_{\nu}(b)Y^{(k+1)}_{\nu}(b)$  and their derivatives  $(Y'_{\nu}(b)J^{(k+1)}_{\nu}(b) - J'_{\nu}(b)Y^{(k+1)}_{\nu}(b))'$  are  $O(\sqrt{\varepsilon})$  and  $O(\varepsilon)$  respec-tively, as  $\varepsilon \to 0$ . It follows that  $R_1(b)$  and  $\partial_{\lambda}R_1(b)$  are  $O(\varepsilon^2\sqrt{\varepsilon})$  as  $\varepsilon \to 0$ . Similarly,

$$R_2(b) = \sum_{k=3}^{+\infty} \frac{\varepsilon^k b^k}{k! (1-\varepsilon)^k} \left( J_\nu(b) Y_\nu^{(k+1)}(b) - Y_\nu(b) J_\nu^{(k+1)}(b) \right),$$

and

$$2\sqrt{\lambda} \frac{\partial R_2(b)}{\partial \lambda} = \frac{\varepsilon b_1(\varepsilon)}{\sqrt{\varepsilon}(1-\varepsilon)} \sum_{k=3}^{+\infty} \frac{b^{k-1}\varepsilon^{k-1}}{(k-1)!(1-\varepsilon)^{k-1}} \left( J_\nu(b)Y_\nu^{(k+1)}(b) - Y_\nu(b)J_\nu^{(k+1)}(b) \right) \\ + \frac{b_1(\varepsilon)}{\sqrt{\varepsilon}} \sum_{k=3}^{+\infty} \frac{\varepsilon^k b^k}{k!(1-\varepsilon)^k} \left( J_\nu(b)Y_\nu^{(k+1)}(b) - Y_\nu(b)J_\nu^{(k+1)}(b) \right)',$$

hence  $R_2(b)$  and  $\partial_{\lambda}R_2(b)$  are  $O(\varepsilon^2)$  as  $\varepsilon \to 0$ .

Summing up all the terms, using Lemma 4.1.40 and Corollary 4.1.46, we obtain

$$\begin{aligned} R(\lambda,\varepsilon) &= R_3(a) \left[ \frac{2\varepsilon}{\pi(1-\varepsilon)} \left( \frac{\nu^2}{b^2} - 1 \right) + \frac{\varepsilon^2}{\pi(1-\varepsilon)^2} \left( 1 - \frac{3\nu^2}{b^2} \right) \\ &+ \frac{\varepsilon^3 b^2}{3\pi(1-\varepsilon)^3} \left( \frac{\nu^4 + 11\nu^2}{b^4} - \frac{3 + 2\nu^2}{b^2} + 1 \right) \right] \\ &+ R_1(b) \left[ \frac{a}{\nu} + \frac{a^3}{2\nu^2(1+\nu)} \right] + R_2(b) \frac{a}{b} + R_3(a) R_1(b) \end{aligned}$$

We conclude that  $R(\lambda, \varepsilon)$  is  $O(\varepsilon^3)$  as  $\varepsilon \to 0$ . Moreover, it easily follows that  $\frac{\partial R(\lambda,\varepsilon)}{\partial \lambda} \text{ is also } O(\varepsilon^3) \text{ as } \varepsilon \to 0.$ 

The proof of the estimates for  $\hat{R}$  and its derivatives is similar and we omit it.  $\Box$ 

**Remark 4.1.29.** According to standard Landau's notation, saying that a function f(z) is O(g(z)) as  $z \to 0$  means that there exists C > 0 such that  $|f(z)| \leq C|g(z)|$  for any z sufficiently close to zero. Thus, using Landau's notation in the statements of Lemmas 4.1.10, 4.1.28 understands the existence of such constants C, which in principle may depend on  $\lambda > 0$ . However, a careful analysis of the proofs reveals that given a bounded interval of the type [A, B] with 0 < A < B then the appropriate constants C in the estimates can be taken independent of  $\lambda \in [A, B]$ .

#### **4.1.3** The case N = 1

We include here a description of the case N = 1 for the sake of completeness. Let  $\Omega$  be the open interval ] - 1, 1[. Problem (4.0.1) reads

$$\begin{cases} u''(x) = 0, & \text{if } x \in ]-1, 1[, \\ u'(\pm 1) = \pm \frac{M}{2} \mu u(\pm 1), \end{cases}$$
(4.1.30)

in the unknowns  $\mu$  and u. It is easy to see that the only eigenvalues are  $\mu_0 = 0$  and  $\mu_1 = \frac{2}{M}$  and they are associated with the constant functions and the function u(x) = x, respectively. As in (3.1.20), we define a mass density  $\rho_{\varepsilon}$  on the whole of ] - 1, 1[ by

$$\rho_{\varepsilon}(x) := \begin{cases} \frac{M}{2\varepsilon} - 1 + \varepsilon & \text{if } x \in ]-1, -1 + \varepsilon[\cup]1 - \varepsilon, 1[, \\ \varepsilon & \text{if } x \in ]-1 + \varepsilon, 1 - \varepsilon[. \end{cases}$$

Note that for any  $x \in ]-1, 1[$  we have  $\rho_{\varepsilon}(x) \to 0$  as  $\varepsilon \to 0$ , and  $\int_{-1}^{1} \rho_{\varepsilon} dx = M$  for all  $\varepsilon > 0$ . Problem (4.0.2) for N = 1 reads

$$\begin{cases} -u''(x) = \lambda \rho_{\varepsilon}(x)u(x), & \text{if } x \in ]-1, 1[, \\ u'(-1) = u'(1) = 0. \end{cases}$$
(4.1.31)

It is well-known from Sturm-Liouville theory that problem (4.1.31) has an increasing sequence of non-negative eigenvalues of multiplicity one. We denote the eigenvalues of (4.1.31) by  $\lambda_l(\varepsilon)$  with  $l \in \mathbb{N}_0$ . For any  $\varepsilon \in ]0, 1[$ , the only zero eigenvalue is  $\lambda_0(\varepsilon)$  and the corresponding eigenfunctions are the constant functions.

We establish an implicit characterization of the eigenvalues of (4.1.31).

**Lemma 4.1.32.** The eigenvalues  $\lambda$  of problem (4.1.31) are given implicitly

as zeros of the equation

$$2\sqrt{\varepsilon \left(\frac{M}{2\varepsilon} - 1 + \varepsilon\right)} \cos\left(2\sqrt{\lambda\varepsilon}(1 - \varepsilon)\right) \sin\left(2\varepsilon\sqrt{\lambda \left(\frac{M}{2\varepsilon} - 1 + \varepsilon\right)}\right) + \left[-\frac{M}{2\varepsilon} + 1 + \left(\frac{M}{2\varepsilon} - 1 + 2\varepsilon\right)\cos\left(2\varepsilon\sqrt{\lambda \left(\frac{M}{2\varepsilon} - 1 + \varepsilon\right)}\right)\right] \sin\left(2\sqrt{\lambda\varepsilon}(1 - \varepsilon)\right) = 0. \quad (4.1.33)$$

*Proof.* Given an eigenvalue  $\lambda > 0$  of problem (4.1.31), a corresponding solution u of (4.1.31) is of the form

$$u(x) = \begin{cases} A\cos\left(\sqrt{\lambda\rho_e}x\right) + B\sin\left(\sqrt{\lambda\rho_e}x\right), & \text{if } x \in ]-1, -1 + \varepsilon[, \\ C\cos\left(\sqrt{\lambda\rho_i}x\right) + D\sin\left(\sqrt{\lambda\rho_i}x\right), & \text{if } x \in ]-1 + \varepsilon, 1 - \varepsilon[, \\ E\cos\left(\sqrt{\lambda\rho_e}x\right) + F\sin\left(\sqrt{\lambda\rho_e}x\right), & \text{if } x \in ]1 - \varepsilon, 1[, \end{cases}$$

where  $\rho_i = \varepsilon$ ,  $\rho_e = \frac{M}{2\varepsilon} - 1 + \varepsilon$  and A, B, C, D, E, F are suitable real numbers. By imposing the continuity condition for u and u' at the points  $x = -1 + \varepsilon$ and  $x = 1 - \varepsilon$  and the boundary conditions, we obtain a homogeneous system of six linear equations in six unknowns of the form  $\mathcal{M}v = 0$ , where v = (A, B, C, D, E, F) and  $\mathcal{M}$  is the matrix associated with the system given by

$$\mathcal{M} = \begin{bmatrix} \alpha & -\beta & -\gamma & \delta & 0 & 0\\ 0 & 0 & \gamma & \delta & -\alpha & -\beta\\ \sqrt{\rho_e}\beta & \sqrt{\rho_e}\alpha & -\sqrt{\rho_i}\delta & -\sqrt{\rho_i}\gamma & 0 & 0\\ 0 & 0 & -\sqrt{\rho_i}\delta & \sqrt{\rho_i}\gamma & \sqrt{\rho_e}\beta & -\sqrt{\rho_e}\alpha\\ \eta & \zeta & 0 & 0 & 0\\ 0 & 0 & 0 & 0 & \eta & -\zeta \end{bmatrix},$$

where

$$\begin{aligned} \alpha &:= \cos\left(\sqrt{\lambda\rho_e}(1-\varepsilon)\right)\\ \beta &:= \sin\left(\sqrt{\lambda\rho_e}(1-\varepsilon)\right)\\ \gamma &:= \cos\left(\sqrt{\lambda\rho_i}(1-\varepsilon)\right)\\ \delta &:= \sin\left(\sqrt{\lambda\rho_i}(1-\varepsilon)\right)\\ \zeta &:= \cos\left(\sqrt{\lambda\rho_e}\right)\\ \eta &:= \sin\left(\sqrt{\lambda\rho_e}\right). \end{aligned}$$

We impose the condition  $\det \mathcal{M} = 0$ . Easy computations give

$$2\Big(\beta\delta\eta\sqrt{\rho_i} + \alpha\delta\eta\sqrt{\rho_e} + \alpha\delta\zeta\sqrt{\rho_i} - \beta\gamma\zeta\sqrt{\rho_e}\Big)\Big(\beta\gamma\eta\sqrt{\rho_i} - \alpha\delta\eta\sqrt{\rho_e} + \alpha\gamma\zeta\sqrt{\rho_i} + \beta\delta\zeta\sqrt{\rho_e}\Big) = 0.$$

This yields formula (4.1.33).

Note that  $\lambda = 0$  is a solution of (4.1.33) for all  $\varepsilon > 0$ , then we consider only the case of nonzero eigenvalues. Using standard Taylor's formulas, we easily prove the following lemma.

Lemma 4.1.34. Equation (4.1.33) can be rewritten in the form

$$M - \frac{\lambda M^2}{2} + \frac{\lambda M^2}{6} \left( 1 + \lambda \left( 2 + \frac{M}{2} \right) \right) \varepsilon + R(\lambda, \varepsilon) = 0, \qquad (4.1.35)$$

where  $R(\lambda, \varepsilon) \in O(\varepsilon^2)$  as  $\varepsilon \to 0$ .

*Proof.* It is sufficient to consider a suitable Taylor's expansion in formula (4.1.33). In particular we expand the functions  $\sin\left(2\varepsilon\sqrt{\lambda\left(\frac{M}{2\varepsilon}-1+\varepsilon\right)}\right)$  and  $\sin\left(2\sqrt{\lambda\varepsilon}(1-\varepsilon)\right)$  up to the third order, and the functions  $\cos\left(2\sqrt{\lambda\varepsilon}(1-\varepsilon)\right)$  and  $\cos\left(2\varepsilon\sqrt{\lambda\left(\frac{M}{2\varepsilon}-1+\varepsilon\right)}\right)$  up to the second order. We obtain

$$F(\lambda,\varepsilon) = 4\sqrt{\lambda\varepsilon} \left(\frac{M}{2} - \varepsilon + \varepsilon^2\right) \left(1 - 2\lambda\varepsilon(1-\varepsilon)^2\right) \left(1 - \frac{2}{3}\lambda\varepsilon\left(\frac{M}{2} - \varepsilon + \varepsilon^2\right)\right) + 2\sqrt{\lambda\varepsilon} \left(1 - \varepsilon - \frac{2\lambda\varepsilon(1-\varepsilon)^3}{3}\right) \left[1 - \frac{M}{2\varepsilon} + \left(2\varepsilon - 1 + \frac{M}{2\varepsilon}\right) \left(1 - 2\lambda\varepsilon\left(\frac{M}{2} - \varepsilon + \varepsilon^2\right)\right)\right] + R_0(\varepsilon) = 0, \quad (4.1.36)$$

where  $R_0(\varepsilon) \in O(\varepsilon^2 \sqrt{\varepsilon})$ . We divide the right-hand side of (4.1.36) by  $2\sqrt{\lambda\varepsilon}$ and set  $R(\varepsilon) = \frac{R_0(\varepsilon)}{2\sqrt{\lambda\varepsilon}}$ . From standard computations formula (4.1.35) easily follows.

Finally, we can prove the following theorem. Note that formula (4.1.38) is the same as (4.1.21) with N = 1, l = 1.

**Theorem 4.1.37.** The first nonzero eigenvalue of problem (4.1.31) has the following asymptotic behavior

$$\lambda_1(\varepsilon) = \mu_1 + \frac{2}{3}(\mu_1 + \mu_1^2)\varepsilon + o(\varepsilon) \quad \text{as } \varepsilon \to 0, \qquad (4.1.38)$$

where  $\mu_1 = 2/M$  is the only nonzero eigenvalue of problem (4.1.30). Moreover, for l > 1 we have that  $\lambda_l(\varepsilon) \to +\infty$  as  $\varepsilon \to 0$ .

*Proof.* The proof is similar to that of Theorem 4.1.20. It is possible to prove that the eigenvalues  $\lambda_l(\varepsilon)$  of (4.1.31) depend with continuity on  $\varepsilon > 0$ . We consider equation (4.1.35) and apply the Implicit Function Theorem. Equation (4.1.35) can be written in the form  $F(\lambda, \varepsilon) = 0$ , with F of class  $C^1$  in  $]0, +\infty[\times[0, 1[$  with  $F(\lambda, 0) = M - \frac{\lambda M^2}{2}, F'_{\lambda}(\lambda, 0) = -\frac{M^2}{2}$  and  $F'_{\varepsilon}(\lambda, 0) = \frac{\lambda M^2}{6}(1 + \lambda(2 + \frac{M}{2})).$ 

Since  $\mu_1 = \frac{2}{M}$ ,  $F(\mu_1, 0) = 0$  and  $F'_{\lambda}(\mu_1, 0) \neq 0$ , the zeros of equation (4.1.38) in a neighborhood of  $(\lambda, 0)$  are given by the graph of a  $C^1$ -function  $\varepsilon \mapsto \lambda(\varepsilon)$  with  $\lambda(0) = \mu_1$ . We note that  $\lambda(\varepsilon) = \lambda_1(\varepsilon)$  for all  $\varepsilon$  small enough. Indeed, assuming by contradiction that  $\lambda(\varepsilon) = \lambda_l(\varepsilon)$  with  $l \geq 2$ , we would obtain that, possibly passing to a subsequence,  $\lambda_1(\varepsilon) \to \overline{\lambda}$  as  $\varepsilon \to 0$ , for some  $\overline{\lambda} \in [0, \lambda_1[$ . Then passing to the limit in (4.1.35) as  $\varepsilon \to 0$  we would obtain a contradiction. Thus,  $\lambda_1(\cdot)$  is of class  $C^1$  in a neighborhood of zero and  $\lambda'_1(0) = -F'_{\varepsilon}(\lambda_1, 0)/F'_{\lambda}(\lambda_1, 0)$  which yields formula (4.1.38).

The divergence of the higher eigenvalues  $\lambda_l(\varepsilon)$  with l > 1, as  $\varepsilon \to 0$  is deduced by the fact that the existence of a converging subsequence of the form  $\lambda_l(\varepsilon_n)$ ,  $n \in \mathbb{N}$  would provide the existence of an eigenvalue for the limiting problem (4.1.30) different from  $\mu_0$  and  $\mu_1$ , which is not admissible.

### 4.1.4 Behavior of the eigenvalues under dilations

We consider problems (4.0.1) and (4.0.2) when  $\Omega = B_R$  is the ball centered in zero and of radius R in  $\mathbb{R}^N$ ,  $N \ge 2$ . Let M > 0 be fixed. Let the function  $\rho_{\varepsilon,M,R} \in L^{\infty}(B_R)$  be defined by

$$\rho_{\varepsilon,M,R}(x) := \begin{cases} \varepsilon, & \text{if } x \in B_{R(1-\varepsilon)}, \\ \frac{M - \varepsilon \omega_N R^N (1-\varepsilon)^N}{\omega_N R^N (1-(1-\varepsilon)^N)}, & \text{if } x \in B_R \setminus \overline{B_{R(1-\varepsilon)}}. \end{cases}$$

We consider now the dependence of  $\lambda_l(\varepsilon)$  and  $\mu_l$  and their derivatives on the radius R. We denote by  $\lambda_l(\varepsilon, M, R)$  the eigenvalues of problem (4.0.2) on  $B_R$  with density  $\rho_{\varepsilon,M,R}$ , while we denote by  $\mu_l(M, R)$  the eigenvalues of problem (4.0.1) on  $B_R$ . We introduce the following quantity:

$$\mathcal{R}(u,\varepsilon,M,R) := \frac{\int_{B_R} |\nabla u|^2 \, dx}{\int_{B_R} u^2 \rho_{\varepsilon,M,R} \, dx}.$$
(4.1.39)

From the Min-Max Principle (see (1.3.8)) we have

$$\lambda_l(\varepsilon, M, R) = \min_{\substack{E \subset H^1(B_R) \\ \dim E = l+1}} \max_{\substack{0 \neq u \in E}} \mathcal{R}(u, \varepsilon, M, R), \quad \forall l \in \mathbb{N},$$

We perform the change of variables x = Ry in (4.1.39). We obtain

$$\mathcal{R}(u,\varepsilon,M,R) = R^{-2}\mathcal{R}\left(u(R\cdot),\varepsilon,\frac{M}{R^N},1\right).$$

It follows that

$$\lambda_l(\varepsilon, M, R) = R^{-2} \lambda_l(\varepsilon, \frac{M}{R^N}, 1).$$

Moreover, we have

$$\mu_l(M, R) = \frac{lN\omega_N}{M} R^{N-2},$$
$$\mu_l(\frac{M}{R^N}, 1) = \frac{lN\omega_N}{M} R^N,$$

therefore  $\mu_l\left(\frac{M}{R^N},1\right)=R^{-2}\mu_l(M,R)$ . We rewrite formula (4.1.3) as

$$\partial_{\varepsilon}\lambda_l(\varepsilon, M, 1)|_{\varepsilon=0} = \frac{2l\mu_l(M, 1)}{3} + \frac{2\mu_l^2(M, 1)\omega_N}{N(2l+N)}.$$

We obtain

$$\begin{split} \partial_{\varepsilon}\lambda_{l}(\varepsilon,M,R)|_{\varepsilon=0} &= R^{-2}\partial_{\varepsilon}\lambda_{l}(\varepsilon,\frac{M}{R^{N}},1)|_{\varepsilon=0} \\ &= R^{-2}\left(\frac{2l\mu_{l}(\frac{M}{R^{N}},1)}{3} + \frac{2\mu_{l}^{2}(\frac{M}{R^{N}},1)}{N(2l+N)}\right) \\ &= R^{-4}\left(\frac{2l\mu_{l}(M,R)}{3} + \frac{2R^{-2}\mu_{l}^{2}(M,R)}{N(2l+N)}\right). \end{split}$$

## 4.1.5 Cross products of Bessel functions

We provide here explicit formulas for the cross products of Bessel functions used in this section.

Lemma 4.1.40. The following identities hold:

$$Y_{\nu}(z)J_{\nu}'(z) - J_{\nu}(z)Y_{\nu}'(z) = -\frac{2}{\pi z};$$
  

$$Y_{\nu}(z)J_{\nu}''(z) - J_{\nu}(z)Y_{\nu}''(z) = \frac{2}{\pi z^{2}};$$
  

$$Y_{\nu}'(z)J_{\nu}''(z) - J_{\nu}'(z)Y_{\nu}''(z) = \frac{2}{\pi z}\left(\frac{\nu^{2}}{z^{2}} - 1\right).$$

*Proof.* It is well-known (see Section 1.4, see also  $[1, \S 9]$ ) that

$$J_{\nu}(z)Y_{\nu}'(z) - Y_{\nu}(z)J_{\nu}'(z) = J_{\nu+1}(z)Y_{\nu}(z) - J_{\nu}(z)Y_{\nu+1}(z) = \frac{2}{\pi z},$$

which gives the first identity in the statement. The second identity holds since

$$J_{\nu}(z)Y_{\nu}''(z) - Y_{\nu}(z)J_{\nu}''(z) = (J_{\nu}(z)Y_{\nu}'(z) - Y_{\nu}(z)J_{\nu}'(z))' = \left(\frac{2}{\pi z}\right)' = -\frac{2}{\pi z^2}$$

The third identity holds since

$$\begin{split} Y'_{\nu}(z)J''_{\nu}(z) - J'_{\nu}(z)Y''_{\nu}(z) \\ &= Y'_{\nu}(z)\left(J_{\nu-1}(z) - \frac{\nu}{z}J_{\nu}(z)\right)' - J'_{\nu}(z)\left(Y_{\nu-1}(z) - \frac{\nu}{z}Y_{\nu}(z)\right)' \\ &= Y'_{\nu}(z)J'_{\nu-1}(z) - J'_{\nu}(z)Y'_{\nu-1}(z) + \frac{\nu}{z^2}\left(Y'_{\nu}(z)J_{\nu}(z) - J'_{\nu}(z)Y_{\nu}(z)\right) \\ &= \left(Y'_{\nu}(z)\frac{1}{2}\left(J_{\nu-2}(z) - J_{\nu}(z)\right) - J'_{\nu}(z)\frac{1}{2}\left(Y_{\nu-2}(z) - Y_{\nu}(z)\right)\right) + \frac{2\nu}{\pi z^3} \\ &= \frac{1}{2}\left(Y'_{\nu}(z)J_{\nu-2}(z) - J'_{\nu}(z)Y_{\nu-2}(z)\right) \\ &- \frac{1}{2}\left(Y'_{\nu}(z)J_{\nu}(z) - J'_{\nu}(z)Y_{\nu}(z)\right) + \frac{2\nu}{\pi z^3} \\ &= \frac{1}{2}\left(J'_{\nu}(z)J_{\nu-1}(z) - J'_{\nu}(z)Y_{\nu-1}(z)\right) - \frac{1}{\pi z} + \frac{2\nu}{\pi z^3} \\ &= \frac{\nu - 1}{z}\left(J_{\nu-1}(z)\left(Y_{\nu-1}(z) - \frac{\nu}{z}Y_{\nu}(z)\right) - Y_{\nu-1}(z)\left(J_{\nu-1}(z) - \frac{\nu}{z}J_{\nu}(z)\right)\right) \\ &- \frac{2}{\pi z} + \frac{2\nu}{\pi z^3} \\ &= -\frac{\nu(\nu-1)}{z^2}\left(Y_{\nu}(z)J_{\nu-1}(z) - J_{\nu}(z)Y_{\nu-1}(z)\right) - \frac{2}{\pi z} + \frac{2\nu}{\pi z^3} \\ &= \frac{2}{\pi z}\left(-1 + \frac{\nu^2}{z^2}\right), \end{split}$$

where the first, second and fourth equalities follow respectively from the well-known formulas  $C'_{\nu}(z) = C_{\nu-1}(z) - \frac{\nu}{z}C_{\nu}(z), 2C'_{\nu}(z) = C_{\nu-1}(z) - C_{\nu+1}(z)$  and  $C_{\nu-2}(z) + C_{\nu}(z) = \frac{2(\nu-1)}{z}C_{\nu-1}(z)$ , where  $C_{\nu}(z)$  stands both for  $J_{\nu}(z)$  and  $Y_{\nu}(z)$  (see Section 1.4, see also [1, §9]). This proves the lemma.  $\Box$ 

Lemma 4.1.41. The following identities hold:

$$Y_{\nu}(z)J_{\nu}^{(k)}(z) - J_{\nu}(z)Y_{\nu}^{(k)}(z) = \frac{2}{\pi z} \left( r_k + R_{\nu,k}(z) \right); \quad (4.1.42)$$

$$Y'_{\nu}(z)J^{(k)}_{\nu}(z) - J'_{\nu}(z)Y^{(k)}_{\nu}(z) = \frac{2}{\pi z} \left(q_k + Q_{\nu,k}(z)\right), \quad (4.1.43)$$

for all k > 2 and  $\nu \ge 0$ , where  $r_k, q_k \in \{0, 1, -1\}$ , and  $Q_{\nu,k}(z)$ ,  $R_{\nu,k}(z)$  are finite sums of quotients of the form  $\frac{c_{\nu,k}}{z^m}$ , with  $m \ge 1$  and  $c_{\nu,k}$  a suitable constant, depending on  $\nu, k$ .

*Proof.* We will prove (4.1.42) and (4.1.43) by induction. Identities (4.1.42) and (4.1.43) hold for k = 1 and k = 2 by Lemma 4.1.40. Suppose now that

$$Y_{\nu}(z)J_{\nu}^{(k)}(z) - J_{\nu}(z)Y_{\nu}^{(k)}(z) = \frac{2}{\pi z} \left(r_k + R_{\nu,k}(z)\right),$$
  
$$Y_{\nu}'(z)J_{\nu}^{(k)}(z) - J_{\nu}'(z)Y^{(k)}(z) = \frac{2}{\pi z} \left(q_k + Q_{\nu,k}(z)\right),$$

hold for all  $\nu \geq 0$ . First consider

$$Y'_{\nu}(z)J^{(k+1)}_{\nu}(z) - J'_{\nu}(z)Y^{(k+1)}_{\nu}(z).$$

We use the recurrence relations  $C_{\nu+1}(z) + C_{\nu-1}(z) = \frac{2\nu}{z}C_{\nu}(z)$  and  $2C'(z) = C_{\nu-1}(z) - C_{\nu+1}(z)$ , where  $C_{\nu}(z)$  stands both for  $J_{\nu}(z)$  and  $Y_{\nu}(z)$  (see [1, §9]). We have

$$\begin{aligned} Y_{\nu}'(z) J_{\nu}^{(k+1)}(z) - J_{\nu}'(z) Y_{\nu}^{(k+1)}(z) &= Y_{\nu}'(z) (J_{\nu}')^{(k)}(z) - J_{\nu}'(z) (Y_{\nu}')^{(k)}(z) \\ &= \frac{1}{4} \left[ \left( Y_{\nu-1}(z) - Y_{\nu+1}(z) \right) (J_{\nu-1}(z) - J_{\nu+1}(z))^{(k)} \right] \\ &- (J_{\nu-1}(z) - J_{\nu-1}(z) Y_{\nu-1}^{(k)}(z) \right) + \left( Y_{\nu+1}(z) J_{\nu+1}^{(k)}(z) - J_{\nu+1}(z) Y_{\nu+1}^{(k)}(z) \right) \\ &+ \left( J_{\nu+1}(z) Y_{\nu-1}^{(k)}(z) - Y_{\nu-1}(z) J_{\nu+1}^{(k)}(z) \right) + \left( J_{\nu-1}(z) Y_{\nu+1}^{(k)}(z) - Y_{\nu+1}(z) J_{\nu-1}^{(k)}(z) \right) \right] \\ &= \frac{1}{4} \left[ \frac{2}{\pi z} \left( r_k + R_{\nu-1,k}(z) + r_k + R_{\nu+1,k}(z) \right) \\ &+ \frac{2\nu}{z} \left( J_{\nu}(z) Y_{\nu-1}^{(k)}(z) - Y_{\nu-1}(z) J_{\nu-1}^{(k)}(z) + J_{\nu}(z) Y_{\nu+1}^{(k)}(z) - Y_{\nu}(z) J_{\nu+1}^{(k)}(z) \right) \right] \\ &= \frac{1}{4} \left[ \frac{4}{\pi z} \left( 2r_k + R_{\nu-1,k}(z) + R_{\nu+1,k}(z) \right) \\ &+ \frac{2\nu}{z} \left( J_{\nu}(z) \left( Y_{\nu-1}(z) + Y_{\nu+1}(z) \right)^{(k)} - Y_{\nu}(z) \left( J_{\nu-1}(z) + J_{\nu+1}(z) \right)^{(k)} \right) \right] \right] \\ &= \frac{1}{4} \left[ \frac{4}{\pi z} \left( 2r_k + R_{\nu-1,k}(z) + R_{\nu+1,k}(z) \right) \\ &+ \frac{2\nu}{z} \left( J_{\nu}(z) \left( Y_{\nu-1}(z) + Y_{\nu+1}(z) \right)^{(k)} - Y_{\nu}(z) \left( J_{\nu-1}(z) + J_{\nu+1}(z) \right)^{(k)} \right) \right] \\ &= \frac{1}{\pi z} \left( 2r_k + R_{\nu-1,k}(z) + R_{\nu+1,k}(z) \right) \\ &+ \frac{2\nu}{z} \left( J_{\nu}(z) \left( Y_{\nu-1}(z) + Y_{\nu+1}(z) \right)^{(k)} - Y_{\nu}(z) \left( J_{\nu-1}(z) + J_{\nu+1}(z) \right)^{(k)} \right) \right] \\ &= \frac{1}{\pi z} \left[ r_k + \frac{1}{2} \left( R_{\nu-1,k}(z) + R_{\nu+1,k}(z) \right) \\ &- \frac{\nu^2}{\pi z} \left[ r_k + \frac{1}{2} \left( R_{\nu-1,k}(z) + R_{\nu+1,k}(z) \right) \right]. \quad (4.1.44)$$

We prove now (4.1.43)

$$Y_{\nu}(z)J_{\nu}^{(k+1)}(z) - J_{\nu}(z)Y_{\nu}^{(k+1)}(z) = \left(Y_{\nu}(z)J_{\nu}^{(k)}(z) - J_{\nu}(z)Y_{\nu}^{(k)}(z)\right)' - \left(Y_{\nu}'(z)J_{\nu}^{(k)}(z) - J_{\nu}'(z)Y_{\nu}^{(k)}(z)\right) = \frac{2}{\pi z}\left(-q_{k} - Q_{\nu,k}(z) - \frac{r_{k}}{z} - \frac{R_{\nu,k}(z)}{z} + R_{\nu,k}'(z)\right). \quad (4.1.45)$$

This concludes the proof.

Corollary 4.1.46. The following formulas hold

$$J_{\nu}(z)Y_{\nu}^{\prime\prime\prime\prime}(z) - Y_{\nu}(z)J_{\nu}^{\prime\prime\prime\prime}(z) = \frac{2}{\pi z} \left(\frac{2+\nu^2}{z^2} - 1\right);$$
  

$$Y_{\nu}^{\prime}(z)J_{\nu}^{\prime\prime\prime}(z) - J_{\nu}^{\prime}(z)Y_{\nu}^{\prime\prime\prime\prime}(z) = \frac{2}{\pi z^2} \left(1 - \frac{3\nu^2}{z^2}\right);$$
  

$$Y_{\nu}^{\prime}(z)J_{\nu}^{\prime\prime\prime\prime\prime}(z) - J_{\nu}^{\prime}(z)Y_{\nu}^{\prime\prime\prime\prime\prime}(z) = \frac{2}{\pi z} \left(1 - \frac{3+2\nu^2}{z^2} + \frac{\nu^4 + 11\nu^2}{z^4}\right).$$

*Proof.* From Lemma 4.1.41 (see in particular (4.1.45)) it follows

$$J_{\nu}(z)Y_{\nu}^{\prime\prime\prime\prime}(z) - Y_{\nu}(z)J_{\nu}^{\prime\prime\prime\prime}(z) = -\frac{2}{\pi z} \left[ -q_2 - Q_{\nu,2}(z) - \frac{r_2}{z} - \frac{R_{\nu,2}(z)}{z} + R_{\nu,2}^{\prime}(z) \right]$$
$$= \frac{2}{\pi z} \left( \frac{2 + \nu^2}{z^2} - 1 \right).$$

Next we compute

$$\begin{aligned} Y'_{\nu}(z)J'''_{\nu}(z) &- J'_{\nu}(z)Y'''_{\nu}(z) \\ &= \frac{2}{\pi z} \left[ r_2 + R_{\nu,2}(z) - \frac{\nu^2}{z} \sum_{j=0}^2 \frac{2(-1)^{2-j}}{j! z^{2-j+1}} \left( r_j + R_{\nu,j}(z) \right) \right] \\ &= \frac{2}{\pi z^2} \left( 1 - \frac{3\nu^2}{z^2} \right). \end{aligned}$$

Finally, by (4.1.44) with k = 3, we have

$$Y_{\nu}'(z)J_{\nu}'''(z) - J_{\nu}'(z)Y_{\nu}''''(z) = \frac{2}{\pi z} \left[ r_3 + \frac{1}{2} \left( R_{\nu-1,3}(z) + R_{\nu+1,3}(z) \right) - \frac{\nu^2}{z} \sum_{j=0}^3 \frac{6(-1)^{3-j}}{j! z^{3-j+1}} \left( r_j + R_{\nu,j}(z) \right) \right]$$
$$= \frac{2}{\pi z} \left( 1 - \frac{3 + 2\nu^2}{z^2} + \frac{\nu^4 + 11\nu^2}{z^4} \right).$$

# 4.2 Bounded domains of class $C^2$ in $\mathbb{R}^2$

In this section we consider problems (4.0.1) and (4.0.2) on bounded domains of class  $C^2$  in  $\mathbb{R}^2$ . The aim of this section is to study the asymptotic behavior of the eigenvalues  $\lambda_j(\varepsilon)$  of problem (4.0.2) and of the corresponding eigenfunctions which we shall denote here by  $u_{j,\varepsilon}$ , as  $\varepsilon$  goes to zero. To do so, we show the validity of an asymptotic expansion for  $\lambda_j(\varepsilon)$  and  $u_{j,\varepsilon}$  as  $\varepsilon$  goes to zero. In addition, we provide explicit formulas for the first two coefficients in the expansions in terms of solutions to suitable auxiliary problems. In particular we establish a closed formula for the derivatives of  $\lambda_j(\varepsilon)$  at  $\varepsilon = 0$ . We shall see that such formula in the case of the unit ball centered at zero in  $\mathbb{R}^2$  agrees with formula (4.1.3) when N = 2 (see Subsection 4.2.5). We remark that asymptotic for vibrating systems containing masses concentrated along curves or around certain points have been considered by several authors in the last decades (see e.g., [50, 52, 53, 80, 98]). In particular we shall follow the approach of [50, 52, 53].

For the sake of completeness, in Subsection 4.2.6 we shall also present the formal heuristic computations which allow to postulate the expansions proved in this section.

We denote by  $\mathcal{H}_{\varepsilon}(\Omega)$  the Hilbert space consisting of the functions in the standard Sobolev Space  $H^1(\Omega)$  endowed with the bilinear form

$$\langle u, v \rangle_{\varepsilon} := \int_{\Omega} \nabla u \cdot \nabla v dx + \int_{\Omega} \rho_{\varepsilon} u v dx, \quad \forall u, v \in \mathcal{H}_{\varepsilon}(\Omega).$$
 (4.2.1)

The bilinear form (4.2.1) induces on  $H^1(\Omega)$  a norm which is equivalent to the standard one. We denote such a norm by  $\|\cdot\|_{\varepsilon}$ . We recall that the weak formulation of problem (4.0.2) can be stated as follows: a pair  $(\lambda(\varepsilon), u_{\varepsilon}) \in \mathbb{R} \times H^1(\Omega)$  is a solution of (4.0.2) in the weak sense if and only if

$$\int_{\Omega} \nabla u_{\varepsilon} \cdot \nabla \varphi dx = \lambda(\varepsilon) \int_{\Omega} \rho_{\varepsilon} u_{\varepsilon} \varphi dx, \quad \forall \varphi \in H^1(\Omega)$$

We introduce the operator  $\mathcal{A}_{\varepsilon}$  from  $\mathcal{H}_{\varepsilon}(\Omega)$  to itself which maps a function  $f \in \mathcal{H}_{\varepsilon}(\Omega)$  to the uniquely determined  $u \in \mathcal{H}_{\varepsilon}$  satisfying the equation

$$\int_{\Omega} \nabla u \cdot \nabla \varphi dx + \int_{\Omega} \rho_{\varepsilon} u \varphi dx = \int_{\Omega} \rho_{\varepsilon} f \varphi dx, \quad \forall \varphi \in \mathcal{H}_{\varepsilon}(\Omega).$$
(4.2.2)

In the sequel we shall exploit the following lemma. We refer to  $[91, \S III.1]$  for its proof.

**Lemma 4.2.3.** Let  $A: H \to H$  be a linear, self-adjoint, positive and compact operator from a separable Hilbert space H to itself. Let  $u \in H$ , with  $\|u\|_{H} = 1$ . Let  $\eta, r > 0$  such that  $\|Au - \eta u\|_{H} \leq r$ . Then, there exists an eigenvalue  $\eta^{*}$  of the operator A which satisfy the inequality  $|\eta - \eta^{*}| \leq r$ . Moreover, for any  $r^{*} > r$  there exists  $u^{*} \in H$  with  $\|u^{*}\|_{H} = 1$ ,  $u^{*}$  belonging to the space generated by all the eigenspaces associated with an eigenvalue of the operator A lying on the segment  $[\eta - r^{*}, \eta + r^{*}]$  and such that

$$||u - u^*||_H \le \frac{2r}{r^*}.$$

We also need the following lemma.

**Lemma 4.2.4.** Let  $u \in H^1(\Omega)$  and  $\rho_{\varepsilon}$  be defined as in (3.1.20). Then there exists  $C_{\Omega} > 0$  which does not depend on  $\varepsilon > 0$  and u, such that

$$\left\| u - \frac{1}{M} \int_{\Omega} \rho_{\varepsilon} u dx \right\|_{L^{2}(\Omega)} \le C_{\Omega} \| \nabla u \|_{L^{2}(\Omega)}, \quad \forall u \in H^{1}(\Omega)$$

*Proof.* Suppose by contradiction that for any  $k \in \mathbb{N}$  there exists  $\varepsilon_k$  with  $\varepsilon_k \to 0$  as  $k \to +\infty$ , and  $u_k \in H^1(\Omega)$  such that

$$\left\| u_k - \frac{1}{M} \int_{\Omega} \rho_{\varepsilon_k} u_k dx \right\|_{L^2(\Omega)} > k \| \nabla u_k \|_{L^2(\Omega)}.$$

$$(4.2.5)$$

Set

$$v_k := \left\| u_k - \frac{1}{M} \int_{\Omega} \rho_{\varepsilon_k} u_k dx \right\|_{L^2(\Omega)}^{-1} \left( u_k - \frac{1}{M} \int_{\Omega} \rho_{\varepsilon_k} u_k dx \right)$$

Then  $\int_{\Omega} \rho_{\varepsilon_k} v_k dx = 0$ ,  $\|v_k\|_{L^2(\Omega)} = 1$  for all  $k \in \mathbb{N}$ , and from (4.2.5),  $\|\nabla v_k\| < \frac{1}{k}$ . Then  $v_k$  is bounded in  $H^1(\Omega)$ , hence, possibly passing to a subsequence,  $v_k \rightharpoonup v$  weakly in  $H^1(\Omega)$  and  $v_k \rightarrow v$  strongly in  $L^2(\Omega)$ , for some  $v \in H^1(\Omega)$ . Moreover, since  $\|\nabla v_k\|_{L^2(\Omega)} < \frac{1}{k}$  it follows that  $\nabla v = 0$  a.e. in  $\Omega$ . We have that for all  $k \in \mathbb{N}$ 

$$0 = \int_{\Omega} \rho_{\varepsilon_k} v_k dx = \int_{\Omega} \rho_{\varepsilon_k} (v_k - v) dx + \int_{\Omega} \rho_{\varepsilon_k} v dx.$$

It is standard to prove that

$$\lim_{k \to +\infty} \int_{\Omega} \rho_{\varepsilon_k} v dx = \frac{M}{|\partial \Omega|} \int_{\partial \Omega} v d\sigma,$$

see Lemma 3.1.22. Moreover, we have that

$$\int_{\Omega} \rho_{\varepsilon_k} (v_k - v) dx = \int_{\omega_{\varepsilon_k}} \frac{\tilde{\rho}_{\varepsilon_k}}{\varepsilon_k} (v_k - v) dx + \varepsilon \int_{\Omega \setminus \overline{\omega}_{\varepsilon_k}} (v_k - v) dx.$$
(4.2.6)

Since the sequence  $\{v_k\}$  is bounded in  $H^1(\Omega)$ , the second term in the righthand side of (4.2.6) goes to zero as k tends to  $+\infty$ . Also, from the fact that  $v_k \to v$  strongly in  $L^2(\Omega)$ , and  $\|\nabla v_k\|_{L^2(\Omega)}$  and  $\|\nabla v\|_{L^2(\Omega)}$  are uniformly bounded in  $k \in \mathbb{N}$ , it is possible to prove that the first term in the righthand side of (4.2.6) goes to zero as k tends to  $+\infty$  (see Lemma 3.1.28).

Then we have that  $\int_{\partial\Omega} v d\sigma = 0$  and  $||v||_{L^2(\Omega)} = 1$ . Hence, by the standard Poincaré-Wirtinger inequality for functions in  $H^1(\Omega)$  having zero integral mean on the boundary, we have

$$0 \le \|v\|_{L^2(\Omega)} = \left\|v - \frac{1}{|\partial\Omega|} \int_{\partial\Omega} v d\sigma\right\|_{L^2(\Omega)} \le \tilde{C}_{\Omega} \|\nabla v\|_{L^2(\Omega)} = 0,$$

where the constant  $\tilde{C}_{\Omega}$  depends only on the open set  $\Omega$ . Therefore  $v \equiv 0$  in  $H^1(\Omega)$ , which is a contradiction since  $\|v\|_{L^2(\Omega)} = 1$ .

We also recall that from Corollary 3.1.42 we have

$$\lim_{\varepsilon \to 0} \lambda_j(\varepsilon) = \mu_j$$

for all  $j \in \mathbb{N}_0$ . Therefore, by Corollary 3.1.42, it follows that the function  $\lambda_j(\cdot)$  which takes  $\varepsilon > 0$  to  $\lambda_j(\varepsilon)$  can be extended with continuity at  $\varepsilon = 0$  by setting  $\lambda_j(0) := \mu_j$  for all  $j \in \mathbb{N}_0$ . Suppose that  $\mu_{j-1} \lneq \mu_j \leq \mu_{j+1} \leq \dots \leq \mu_{j+m} \ncong \mu_{j+m+1}$  and  $\lambda_{j-1}(\varepsilon) \nleq \lambda_j(\varepsilon) \leq \lambda_{j+1}(\varepsilon) \leq \dots \leq \lambda_{j+m}(\varepsilon) \gneqq \lambda_{j+m+1}(\varepsilon)$  for all  $\varepsilon > 0$  in a suitable right neighborhood of zero. Let us denote by  $P_{j,m}(\varepsilon)$  the operator from  $L^2(\Omega)$  to itself which maps a function  $f \in L^2(\Omega)$  to its projection on the space generated by all the eigenfunctions associated with the eigenvalues  $\lambda_j(\varepsilon), \dots, \lambda_{j+m}(\varepsilon)$ , and by  $Q_{j,m}$  the operator from  $L^2(\Omega)$  to its projection on the space generated by all the eigenvalues  $\mu_j, \dots, \mu_{j+m}$ . Then, from Corollary 3.1.42 it follows that

$$\lim_{\varepsilon \to 0} \|P_{j,m}(\varepsilon) - Q_{j,m}\|_{\mathcal{L}(L^2(\Omega), L^2(\Omega))} = 0.$$

In particular, if  $\mu_j$  is a simple eigenvalue, then there exist  $\varepsilon_j > 0$  such that  $\lambda_j(\varepsilon)$  is simple for all  $\varepsilon \in [0, \varepsilon_j]$ .

It is well-known that there exists  $\varepsilon_0 > 0$  such that the map  $x \mapsto x - \varepsilon \nu(x)$ is a diffeomorphism from  $\partial \Omega$  to  $\partial \omega_{\varepsilon} \cap \Omega$  for all  $\varepsilon \in ]0, \varepsilon_0[$ , where  $\omega_{\varepsilon}$  is defined by (3.1.19) (see Theorem 3.1.27).

We shall use curvilinear coordinates in the strip  $\omega_{\varepsilon}$ . Let  $\gamma : [0, |\partial\Omega|[ \to \partial\Omega]$ be the arc length parametrization of the boundary  $\partial\Omega$ . Then we consider the map  $\psi : [0, |\partial\Omega|[\times]0, \varepsilon[ \to \omega_{\varepsilon}$  defined by  $\psi(s, t) := \gamma(s) - t\nu(\gamma(s))$  for all  $(s, t) \in [0, |\partial\Omega|[\times]0, \varepsilon[$ , where  $\nu(\gamma(s))$  denotes the outer unit normal to  $\partial\Omega$  at  $\gamma(s)$ . We denote by  $\kappa(s)$  the signed curvature of  $\partial\Omega$ , namely  $\kappa(s) :=$  $\gamma'_1(s)\gamma''_2(s) - \gamma'_2(s)\gamma''_1(s)$  for all  $s \in [0, |\partial\Omega|[$ .

In order to study problem (4.0.2) it is also convenient to introduce a change of variables by setting  $\xi = \frac{t}{\varepsilon}$ . Accordingly, we denote by  $\psi_{\varepsilon}$  the function from  $[0, |\partial\Omega|[\times]0, 1[$  to  $\omega_{\varepsilon}$  defined by  $\psi_{\varepsilon}(s, \xi) := \gamma(s) - \varepsilon \xi \nu(\gamma(s))$  for all  $(s, \xi) \in [0, |\partial\Omega|[\times]0, 1[$ . The variable  $\xi$  is usually called rapid variable. Note that in this new system of coordinates  $(s, \xi)$ , the strip  $\omega_{\varepsilon}$  is transformed into a band of length  $|\partial\Omega|$  and width 1 (see Figures 4.3, 4.4). Moreover, we note that if  $\varepsilon < (\sup_{s \in [0, |\partial\Omega|[} |\kappa(s)|)^{-1})$ , we have  $1 - \varepsilon \xi \kappa(s) > 0$  for all  $\xi \in ]0, 1[$ , so that  $|\det D\psi_{\varepsilon}| = \varepsilon(1 - \varepsilon \xi \kappa(s))$ . From now on we consider  $\varepsilon \in ]0, \varepsilon_0[$  where  $\varepsilon_0$  is sufficiently small and is such that  $\varepsilon_0 < (\sup_{s \in [0, |\partial\Omega|[} |\kappa(s)|)^{-1}$ .



Figure 4.3



Figure 4.4

We also need to write the gradient of a function u with respect to the coordinates  $(s, \xi)$ . We have

$$\left(\nabla u \circ \psi_{\varepsilon}\right)(s,\xi) = \left(\begin{array}{c} \frac{\gamma_1'(s)}{1-\varepsilon\xi\kappa(s)}\partial_s(u\circ\psi_{\varepsilon}(s,\xi)) - \frac{\gamma_2'(s)+\varepsilon\xi\gamma_1''(s)}{\varepsilon(1-\varepsilon\xi\kappa(s))}\partial_\xi(u\circ\psi_{\varepsilon}(s,\xi)) \\ \frac{\gamma_2'(s)}{1-\varepsilon\xi\kappa(s)}\partial_s(u\circ\psi_{\varepsilon}(s,\xi)) + \frac{\gamma_1'(s)-\varepsilon\xi\gamma_2''(s)}{\varepsilon(1-\varepsilon\xi\kappa(s))}\partial_\xi(u\circ\psi_{\varepsilon})(s,\xi) \end{array}\right),$$

and therefore

$$\begin{aligned} (\nabla u \circ \psi_{\varepsilon} \cdot \nabla v \circ \psi_{\varepsilon}) (s,\xi) \\ &= \frac{1}{\varepsilon^2} \partial_{\xi} (u \circ \psi_{\varepsilon}(s,\xi)) \partial_{\xi} (v \circ \psi_{\varepsilon}(s,\xi)) + \frac{\partial_s (u \circ \psi_{\varepsilon}(s,\xi)) \partial_s (v \circ \psi_{\varepsilon}(s,\xi))}{(1 - \varepsilon \xi \kappa(s))^2} \\ &= \frac{1}{\varepsilon^2} \partial_{\xi} (u \circ \psi_{\varepsilon}(s,\xi)) \partial_{\xi} (v \circ \psi_{\varepsilon}(s,\xi)) + \partial_s (u \circ \psi_{\varepsilon}(s,\xi)) \partial_s (v \circ \psi_{\varepsilon}(s,\xi)) \\ &+ \varepsilon \xi \kappa(s) \sum_{j=1}^{+\infty} (j+1) (\varepsilon \xi \kappa(s))^{j-1} \partial_s (u \circ \psi_{\varepsilon}(s,\xi)) \partial_s (v \circ \psi_{\varepsilon}(s,\xi)), \quad (4.2.7) \end{aligned}$$

for all  $(s,\xi) \in [0, |\partial \Omega|[\times]0, 1[.$ 

We observe that for all  $\varepsilon < (\sup_{s \in [0, |\partial \Omega|[} |\kappa(s)|)^{-1}$  the series in the last line of (4.2.7) is convergent.

Finally, we can write  $\rho_{\varepsilon} = \varepsilon + \frac{1}{\varepsilon} \tilde{\rho}_{\varepsilon} \chi_{\omega_{\varepsilon}}$ , where  $\chi_{\omega_{\varepsilon}}$  is the characteristic function of  $\omega_{\varepsilon}$  and

$$\tilde{\rho}_{\varepsilon} := \varepsilon \left( \frac{M - \varepsilon |\Omega \setminus \overline{\omega}_{\varepsilon}|}{|\omega_{\varepsilon}|} \right) - \varepsilon^2, \qquad (4.2.8)$$

for all  $\varepsilon \in ]0, \varepsilon_0[$ . We note that  $\frac{M-\varepsilon|\Omega\setminus\overline{\omega}_{\varepsilon}|}{|\omega_{\varepsilon}|} \in O(\varepsilon^{-1})$  as  $\varepsilon \to 0$ . We note also that there exist  $\varepsilon_0 > 0$  and  $r_2 > r_1 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$  it holds

$$r_2 < \tilde{\rho}_{\varepsilon} < r_1. \tag{4.2.9}$$

# 4.2.1 Asymptotic expansions and derivatives of the eigenvalues

Let  $\mu_j$  be a simple eigenvalue of problem (4.0.1). It is not restrictive to consider simple eigenvalues since the Steklov eigenvalues are generically simple (see e.g., [5, 102]).

We recall that from Corollary 3.1.42 it follows that there exists  $\varepsilon_j \in ]0, \varepsilon_0[$ such that for all  $\varepsilon \in ]0, \varepsilon_j[$  the eigenvalue  $\lambda_j(\varepsilon)$  of problem (4.0.2) is simple and  $\lambda_j(\varepsilon) \to \mu_j$  as  $\varepsilon \to 0$ . We shall prove the following theorem concerning an asymptotic expansion of  $\lambda_j(\varepsilon)$ .

**Theorem 4.2.10.** Let  $j \in \mathbb{N}_0$ . Assume that  $\mu_j$  is a simple eigenvalue of problem (4.0.1). Then

$$\lambda_j(\varepsilon) = \mu_j + \varepsilon \mu_j^1 + O(\varepsilon^2) \tag{4.2.11}$$

as  $\varepsilon \to 0$ , where

$$\mu_j^1 = \frac{|\Omega|\mu_j}{M} - \frac{|\partial\Omega|\mu_j}{M} \int_{\Omega} u_j^2 dx + \frac{2M\mu_j^2}{3|\partial\Omega|} + \frac{\mu_j}{2} \int_{\partial\Omega} u_j^2 \kappa d\sigma - \frac{K\mu_j}{2|\partial\Omega|}, \quad (4.2.12)$$

and  $u_j \in H^1(\Omega)$  is the unique eigenfunction of problem (4.0.1) associated with the eigenvalue  $\mu_j$  such that  $\int_{\partial\Omega} u_j^2 d\sigma = 1$  and  $K \in \mathbb{R}$  is given by

$$K := \int_0^{|\partial\Omega|} \kappa(s) ds. \tag{4.2.13}$$

We also prove an asymptotic expansion for the eigenfunction  $u_{j,\varepsilon}$  associated with  $\lambda_j(\varepsilon)$ . This is contained in the following theorem.

**Theorem 4.2.14.** Let  $j \in \mathbb{N}_0$ . Assume that  $\mu_j$  is a simple eigenvalue of problem (4.0.1) and  $\varepsilon_j > 0$  is such that  $\lambda_j(\varepsilon)$  is a simple eigenvalue of problem (4.0.2) for all  $\varepsilon \in ]0, \varepsilon_j[$ . Let  $u_j$  be the unique eigenfunction of problem (4.0.1) associated with  $\mu_j$  such that  $\int_{\partial\Omega} u_j^2 d\sigma = 1$ . Let  $u_{j,\varepsilon}$  be the unique eigenfunction of problem (4.0.2) corresponding to  $\lambda_j(\varepsilon)$  such that  $\frac{|\partial\Omega|}{M} \int_{\Omega} \rho_{\varepsilon} u_{j,\varepsilon}^2 dx = 1$  for all  $\varepsilon \in ]0, \varepsilon_j[$ . Then there exist  $u_j^1 \in H^1(\Omega)$  and  $w_j, w_j^1 \in H^1([0, |\partial\Omega|[\times]0, 1[)$  such that

$$u_{j,\varepsilon} = u_j + \varepsilon u_j^1 + \varepsilon v_j + \varepsilon^2 v_j^1 + O(\varepsilon^2) \quad \text{in } L^2(\Omega), \tag{4.2.15}$$

as  $\varepsilon \to 0$ , where the functions  $v_j, v_j^1 \in H^1(\Omega)$  are the extensions by 0 of  $w_j \circ \psi_{\varepsilon}^{(-1)}$  and  $w_j^1 \circ \psi_{\varepsilon}^{(-1)}$  respectively to  $\Omega$ .

We shall present explicit formulas for  $w_j$  and  $w_j^1$  (see (4.2.16) and (4.2.32)) and we shall identify  $u_j^1$  as the solution of a certain boundary value problem (see problem (4.2.31)).

The proof of Theorems 4.2.10 and 4.2.14 consists of two steps. In the first step (Subsection 4.2.2) we show that the quantity  $\lambda_j(\varepsilon) - \mu_j$  is of order  $\varepsilon$  as  $\varepsilon$  tends to zero. Moreover, in this step we introduce the function  $w_j$ . We also show that  $||u_{j,\varepsilon} - u_j - \varepsilon v_j||_{L^2(\Omega)}$  is of order  $\varepsilon$  as  $\varepsilon$  tends to zero. In the second step (Subsection 4.2.3) we complete the proof of Theorems 4.2.10 and 4.2.14. Moreover, we introduce the boundary value problem solved by  $u_j^1$  and the function  $w_j^1$ . In Subsection 4.2.4 we recall some technical results on the well-posedness of the auxiliary boundary value problem solved by  $u_j^1$ .

### 4.2.2 First step of the proof of Theorems 4.2.10 and 4.2.14

The aim of this subsection is to prove formulas (4.2.28) and (4.2.30), that is, justify a part of the expansions (4.2.11) and (4.2.15).

Let  $j \in \mathbb{N}_0$ . Assume that  $\mu_j$  is a simple eigenvalue of problem (4.0.1). Let  $u_j$  be the unique eigenfunction of (4.0.1) associated with  $\mu_j$  such that  $\int_{\partial\Omega} u_j^2 d\sigma = 1$ . Let  $\varepsilon_j \in ]0, \varepsilon_0[$  be such that  $\lambda_j(\varepsilon)$  is a simple eigenvalue of (4.0.2) for all  $\varepsilon \in ]0, \varepsilon_j[$ . Let the function  $w_j(s,\xi)$  from  $[0, |\partial\Omega|[\times]0, 1[$  to  $\mathbb{R}$  be defined by

$$w_j(s,\xi) := -\frac{M\mu_j}{2|\partial\Omega|} (u_j \circ \psi_{\varepsilon}(s,0)) \left(\xi - 1\right)^2.$$
(4.2.16)

The function  $w_j$  solves the following problem

$$\begin{cases} -\partial_{\xi}^{2}w_{j}(s,\xi) = \frac{M\mu_{j}}{|\partial\Omega|}(u_{j}\circ\psi_{\varepsilon}(s,0)), & (s,\xi)\in[0,|\partial\Omega|[\times]0,1[,\\ \partial_{\xi}w_{j}(s,0) = \frac{M\mu_{j}}{|\partial\Omega|}(u_{j}\circ\psi_{\varepsilon}(s,0)), & s\in[0,|\partial\Omega|[,\\ w_{j}(s,1) = \partial_{\xi}w_{j}(s,1) = 0, & s\in[0,|\partial\Omega|[. \end{cases}$$

$$(4.2.17)$$

Now let  $v_j \in H^1(\Omega)$  be defined as in Theorem 4.2.14. We note that by construction  $v_j \in H^1(\Omega)$ . We plan to apply Lemma 4.2.3 to the compact and self-adjoint operator  $\mathcal{A}_{\varepsilon}$  acting on the Hilbert space  $\mathcal{H}_{\varepsilon}(\Omega)$ . We note that  $\lambda_j(\varepsilon)$  is an eigenvalue of (4.0.2) if and only if  $\frac{1}{1+\lambda_j(\varepsilon)}$  is an eigenvalue of  $\mathcal{A}_{\varepsilon}$ . Moreover, we note that  $|\mu_j - \mu_{j-1}| > 0$  and  $|\mu_j - \mu_{j+1}| > 0$ , and that  $\lambda_j(\varepsilon) \neq \lambda_{j-1}(\varepsilon)$  and  $\lambda_j(\varepsilon) \neq \lambda_{j+1}(\varepsilon)$  for all  $\varepsilon \in ]0, \varepsilon_j[$ . Then, by the continuity of the eigenvalues, it follows that there exists a constant  $\delta_j > 0$  which does not depend on  $\varepsilon > 0$  and such that  $\left|\frac{1}{1+\mu_j} - \frac{1}{1+\lambda_{j-1}(\varepsilon)}\right|, \left|\frac{1}{1+\mu_j} - \frac{1}{1+\lambda_{j+1}(\varepsilon)}\right| > \delta_j$ , and  $\left|\frac{1}{1+\mu_j} - \frac{1}{1+\lambda_j(\varepsilon)}\right| \leq \delta_j$ , for all  $\varepsilon \in ]0, \varepsilon_j[$  (possibly choosing  $\varepsilon_j > 0$  smaller). We plan to apply Lemma 4.2.3 with  $\eta = \frac{1}{1+\mu_j}, u = \frac{u_j + \varepsilon v_j}{\|u_j + \varepsilon v_j\|_{\varepsilon}}$ 

 $r^* = \delta_j$  and a suitable r > 0. First we prove that

$$\left| \left\langle \mathcal{A}_{\varepsilon}(u_j + \varepsilon v_j) - \frac{1}{1 + \mu_j} (u_j + \varepsilon v_j), \varphi \right\rangle_{\varepsilon} \right| \le C \varepsilon \|\varphi\|_{\varepsilon}, \quad \forall \varphi \in H^1(\Omega),$$
(4.2.18)

where C > 0 is a constant which does not depend on  $\varepsilon$  and  $\varphi$ . Indeed, by (4.2.1) and (4.2.2) we have

$$\begin{split} \left| \left\langle \mathcal{A}_{\varepsilon}(u_{j} + \varepsilon v_{j}) - \frac{1}{1 + \mu_{j}}(u_{j} + \varepsilon v_{j}), \varphi \right\rangle_{\varepsilon} \right| \\ &= \left| \int_{\Omega} \rho_{\varepsilon} u_{j} \varphi dx + \int_{\omega_{\varepsilon}} \varepsilon \rho_{\varepsilon} v_{j} \varphi dx - \frac{1}{1 + \mu_{j}} \left( \int_{\Omega} \nabla u_{j} \cdot \nabla \varphi dx + \int_{\Omega} \rho_{\varepsilon} u_{j} \varphi dx \right) \\ &+ \int_{\omega_{\varepsilon}} \varepsilon \nabla v_{j} \cdot \nabla \varphi dx + \int_{\omega_{\varepsilon}} \varepsilon \rho_{\varepsilon} v_{j} \varphi dx \right) \right| \\ &= \frac{\mu_{j}}{1 + \mu_{j}} \left| \varepsilon \int_{\Omega} u_{j} \varphi dx + \int_{\omega_{\varepsilon}} \frac{1}{\varepsilon} \tilde{\rho}_{\varepsilon} u_{j} \varphi dx - \frac{M}{|\partial\Omega|} \int_{\partial\Omega} u_{j} \varphi d\sigma \\ &+ \varepsilon \int_{\omega_{\varepsilon}} \rho_{\varepsilon} v_{j} \varphi dx - \frac{\varepsilon}{\mu_{j}} \int_{\omega_{\varepsilon}} \nabla v_{j} \cdot \nabla \varphi dx \right|. \quad (4.2.19) \end{split}$$

We introduce the following quantities:

$$J_{1,\varepsilon} := \varepsilon \int_{\Omega} u_{j} \varphi dx,$$
  

$$J_{2,\varepsilon} := \int_{\omega_{\varepsilon}} \frac{1}{\varepsilon} \tilde{\rho}_{\varepsilon} u_{j} \varphi dx,$$
  

$$J_{3,\varepsilon} := \frac{M}{|\partial \Omega|} \int_{\partial \Omega} u_{j} \varphi d\sigma,$$
  

$$J_{4,\varepsilon} := \varepsilon \int_{\omega_{\varepsilon}} \rho_{\varepsilon} v_{j} \varphi dx,$$
  

$$J_{5,\varepsilon} := \varepsilon \int_{\omega_{\varepsilon}} \nabla v_{j} \cdot \nabla \varphi dx.$$

We have that the expression inside the absolute value which appears in the last term of (4.2.19) equals

$$J_{1,\varepsilon} + J_{2,\varepsilon} - J_{3,\varepsilon} + J_{4,\varepsilon} - \frac{1}{\mu_j} J_{5,\varepsilon}.$$

We study the quantities  $J_{1,\varepsilon}$ ,  $J_{2,\varepsilon}$ ,  $J_{3,\varepsilon}$ ,  $J_{4,\varepsilon}$  and  $J_{5,\varepsilon}$  separately. Through all the rest of the section we will denote by C a positive constant which does not depend on  $\varepsilon$  and  $\varphi$  (and which can eventually be re-defined line by line). We consider  $J_{1,\varepsilon}$  first. We have

$$J_{1,\varepsilon} = \varepsilon \int_{\Omega} u_j \varphi dx \le C \varepsilon \|u_j\|_{\varepsilon} \|\varphi\|_{\varepsilon}.$$
(4.2.20)

This follows from the Hölder inequality  $\int_{\Omega} u_j \varphi dx \leq ||u_j||_{L^2(\Omega)} ||\varphi||_{L^2(\Omega)}$  and from Lemma 4.2.4. In fact, given  $\varphi \in H^1(\Omega)$  we have

$$\begin{aligned} \|\varphi\|_{L^{2}(\Omega)} &= \left\|\varphi - \frac{1}{M} \int_{\Omega} \rho_{\varepsilon} \varphi dx + \frac{1}{M} \int_{\Omega} \rho_{\varepsilon} \varphi dx \right\|_{L^{2}(\Omega)} \\ &\leq \left\|\varphi - \frac{1}{M} \int_{\Omega} \rho_{\varepsilon} \varphi dx \right\|_{L^{2}(\Omega)} + \left\|\frac{1}{M} \int_{\Omega} \rho_{\varepsilon} \varphi dx \right\|_{L^{2}(\Omega)} \\ &\leq C_{\Omega} \|\nabla\varphi\|_{L^{2}(\Omega)} + \frac{|\Omega|^{\frac{1}{2}}}{M^{\frac{1}{2}}} \left(\int_{\Omega} \rho_{\varepsilon} \varphi^{2} dx\right)^{\frac{1}{2}} \leq \max\left\{C_{\Omega}, \frac{|\Omega|^{\frac{1}{2}}}{M^{\frac{1}{2}}}\right\} \|\varphi\|_{\varepsilon}, \quad (4.2.21) \end{aligned}$$

where in the second inequality we have used the fact that  $\rho_{\varepsilon} > 0$  in the following way:

$$\int_{\Omega} \rho_{\varepsilon} \varphi dx = \int_{\Omega} \rho_{\varepsilon}^{\frac{1}{2}} \rho_{\varepsilon}^{\frac{1}{2}} \varphi dx \le \left( \int_{\Omega} \rho_{\varepsilon} dx \right)^{\frac{1}{2}} \left( \int_{\Omega} \rho_{\varepsilon} \varphi^{2} dx \right)^{\frac{1}{2}}.$$

Moreover, we observe that  $\|u_j\|_{\varepsilon} \leq C$ . This follows from the fact that  $\lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} u_j^2 dx = \frac{M}{|\partial \Omega|} \int_{\partial \Omega} u_j^2 d\sigma = \frac{M}{|\partial \Omega|}$ . We now consider  $J_{2,\varepsilon}$ .

$$\begin{split} J_{2,\varepsilon} &= \int_{\omega_{\varepsilon}} \frac{1}{\varepsilon} \tilde{\rho}_{\varepsilon} u_{j} \varphi dx \\ &= \int_{0}^{|\partial \Omega|} \int_{0}^{1} \tilde{\rho}_{\varepsilon} (u_{j} \circ \psi_{\varepsilon}(s,\xi)) (\varphi \circ \psi_{\varepsilon}(s,\xi)) (1 - \varepsilon \xi \kappa(s)) d\xi ds \\ &= \int_{0}^{|\partial \Omega|} \int_{0}^{1} \tilde{\rho}_{\varepsilon} (u_{j} \circ \psi_{\varepsilon}(s,\xi)) (\varphi \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds \\ &- \int_{0}^{|\partial \Omega|} \int_{0}^{1} \tilde{\rho}_{\varepsilon} (u_{j} \circ \psi_{\varepsilon}(s,\xi)) (\varphi \circ \psi_{\varepsilon}(s,\xi)) \frac{\xi \kappa(s)}{1 - \varepsilon \xi \kappa(s)} \varepsilon (1 - \varepsilon \xi \kappa(s)) d\xi ds \\ &\leq \int_{0}^{|\partial \Omega|} \int_{0}^{1} \tilde{\rho}_{\varepsilon} (u_{j} \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds + C \varepsilon \int_{\omega_{\varepsilon}} \frac{\tilde{\rho}_{\varepsilon}}{\varepsilon} |u_{j} \varphi| dx \\ &\leq \int_{0}^{|\partial \Omega|} \int_{0}^{1} \tilde{\rho}_{\varepsilon} (u_{j} \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds + C \varepsilon \|u_{j}\|_{\varepsilon} \|\varphi\|_{\varepsilon} \,. \end{split}$$

Then

$$\begin{aligned} J_{1,\varepsilon} + J_{2,\varepsilon} - J_{3,\varepsilon} + \frac{1}{\mu_j} J_{5,\varepsilon} \\ &= \varepsilon \int_{\Omega} u_j \varphi dx + \int_{\omega_{\varepsilon}} \frac{1}{\varepsilon} \tilde{\rho}_{\varepsilon} u_j \varphi dx \\ &- \frac{M}{|\partial \Omega|} \int_0^{|\partial \Omega|} \int_0^1 (u_j \circ \psi_{\varepsilon}(s,0)) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \end{aligned}$$

Neumann vs Steklov: an asymptotic analysis

$$\leq \int_{0}^{|\partial\Omega|} \int_{0}^{1} \tilde{\rho}_{\varepsilon}(u_{j} \circ \psi_{\varepsilon})(\varphi \circ \psi_{\varepsilon})d\xi ds \\ - \frac{M}{|\partial\Omega|} \int_{0}^{|\partial\Omega|} \int_{0}^{1} (u_{j} \circ \psi_{\varepsilon}(s,0))(\varphi \circ \psi_{\varepsilon}(s,\xi))d\xi ds + C\varepsilon \|u_{j}\|_{\varepsilon} \|\varphi\|_{\varepsilon} \,.$$

Consider now

$$\begin{split} \int_{0}^{|\partial\Omega|} \int_{0}^{1} \left( \tilde{\rho}_{\varepsilon}(u_{j} \circ \psi_{\varepsilon}(s,\xi)) - \frac{M}{|\partial\Omega|}(u_{j} \circ \psi_{\varepsilon}(s,0)) \right) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \\ &= \int_{0}^{|\partial\Omega|} \int_{0}^{1} \tilde{\rho}_{\varepsilon} \left( u_{j} \circ \psi_{\varepsilon}(s,\xi) \right) - \left( u_{j} \circ \psi_{\varepsilon}(s,0) \right) \left( \varphi \circ \psi_{\varepsilon}(s,\xi) \right) d\xi ds \\ &+ \int_{0}^{|\partial\Omega|} \int_{0}^{1} \left( u_{j} \circ \psi_{\varepsilon}(s,0) \right) \left( \tilde{\rho}_{\varepsilon} - \frac{M}{|\partial\Omega|} \right) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \end{split}$$

We note that  $|\omega_{\varepsilon}| = \varepsilon |\partial\Omega| - \frac{\varepsilon^2}{2} \int_0^{|\partial\Omega|} \kappa(s) ds = \varepsilon |\partial\Omega| - \frac{\varepsilon^2}{2} K$ , where K is defined by (4.2.13). From standard Taylor's expansions of the right-hand side of (4.2.8) it follows that

$$\tilde{\rho}_{\varepsilon} = \frac{M}{|\partial\Omega|} + \frac{\frac{1}{2}KM - |\Omega||\partial\Omega|}{|\partial\Omega|^2}\varepsilon + F(\varepsilon), \qquad (4.2.22)$$

where  $F(\varepsilon) \in O(\varepsilon^2)$  as  $\varepsilon \to 0$ . It follows that

$$\begin{split} \int_{0}^{|\partial\Omega|} \int_{0}^{1} (u_{j} \circ \psi_{\varepsilon}(s,0)) \left( \tilde{\rho}_{\varepsilon} - \frac{M}{|\partial\Omega|} \right) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \\ &\leq C \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} |(u_{j} \circ \psi_{\varepsilon}(s,0))(\varphi \circ \psi_{\varepsilon}(s,\xi))| d\xi ds \\ &\leq C \varepsilon \|u_{j}\|_{L^{2}(\partial\Omega)} \left( \int_{0}^{|\partial\Omega|} \int_{0}^{1} (\varphi \circ \psi_{\varepsilon}(s,\xi))^{2} \frac{\varepsilon(1 - \varepsilon \xi \kappa(s))}{\varepsilon(1 - \varepsilon \xi \kappa(s))} d\xi ds \right)^{\frac{1}{2}} \\ &\leq C \varepsilon \|u_{j}\|_{\varepsilon} \left( \int_{\omega_{\varepsilon}} \frac{1}{\varepsilon} \varphi^{2} dx \right)^{\frac{1}{2}} \leq C \varepsilon \|u_{j}\|_{\varepsilon} \|\varphi\|_{\varepsilon} \,, \end{split}$$

where in the last line we have used the fact that  $||u_j||_{L^2(\partial\Omega)} \leq C_{\Omega} ||\nabla u_j||_{L^2(\Omega)}$ and (4.2.9). Moreover, since  $u_j$  is a solution of (4.0.1), by standard elliptic regularity (see e.g., [2]), it follows that  $u_j \in C^2(\Omega) \cap C^{1,\alpha}(\overline{\Omega})$  for all  $\alpha \in ]0, 1[$ (note that we only assume  $\Omega$  of class  $C^2$ ). Since  $\psi_{\varepsilon}$  is a diffeomorphism of class  $C^2$ , for all  $(s,\xi) \in [0, |\partial\Omega|[\times]0, 1[$  we have

$$(u_j \circ \psi_{\varepsilon}(s,\xi)) - (u_j \circ \psi_{\varepsilon}(s,0)) = \xi \partial_{\xi}(u_j \circ \psi_{\varepsilon})(s,\xi^*),$$

for some  $\xi^* \in (0,\xi)$ . Then, by recalling the definition of  $\psi$  and setting  $t^* := \varepsilon \xi^*$ , it follows that

$$\begin{split} \int_{0}^{|\partial\Omega|} \int_{0}^{1} \tilde{\rho}_{\varepsilon} \left( (u_{j} \circ \psi_{\varepsilon}(s,\xi)) - (u_{j} \circ \psi_{\varepsilon}(s,0)) \right) \varphi d\xi ds \\ &= \tilde{\rho}_{\varepsilon} \int_{0}^{|\partial\Omega|} \int_{0}^{1} \xi \partial_{\xi} (u_{j} \circ \psi_{\varepsilon}) (s,\xi^{*}) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \\ &= \tilde{\rho}_{\varepsilon} \int_{0}^{|\partial\Omega|} \int_{0}^{\varepsilon} t \partial_{t} (u_{j} \circ \psi) (s,t^{*}) (\varphi \circ \psi(s,t)) \frac{dt}{\varepsilon} ds \\ &\leq C \|u_{j}\|_{C^{1}(\overline{\Omega})} \int_{0}^{|\partial\Omega|} \int_{0}^{\varepsilon} \frac{t}{\varepsilon^{\frac{1}{2}}} \frac{(\varphi \circ \psi)}{\varepsilon^{\frac{1}{2}}} dt ds \\ &\leq \frac{C \|u_{j}\|_{C^{1}(\overline{\Omega})} \varepsilon}{\sqrt{3}} \int_{0}^{|\partial\Omega|} \left( \int_{0}^{\varepsilon} \frac{(\varphi \circ \psi)^{2}}{\varepsilon} dt \right)^{\frac{1}{2}} ds \leq C \varepsilon \|\varphi\|_{\varepsilon}, \quad (4.2.23) \end{split}$$

where we have exploited Hölder inequality to prove the second inequality. We consider now  $J_{4,\varepsilon}$ .

$$J_{4,\varepsilon} = \varepsilon \int_{\omega_{\varepsilon}} \rho_{\varepsilon} v_j \varphi dx \le \varepsilon \|v_j\|_{\varepsilon} \|\varphi\|_{\varepsilon}.$$

From the definition of  $v_j$  (see (4.2.16)) it is standard to prove that  $||v_j||_{\varepsilon} \leq C$ .

We consider  $J_{5,\varepsilon}$  and pass to the coordinates  $(s,\xi)$ . We use formula (4.2.7) and we obtain

$$\begin{split} J_{5,\varepsilon} &= \varepsilon \int_{\omega_{\varepsilon}} \nabla v_{j} \nabla \varphi dx \\ &= \varepsilon^{2} \int_{0}^{|\partial\Omega|} \int_{0}^{1} \left( \frac{1}{\varepsilon^{2}} \partial_{\xi} w_{j}(s,\xi) \partial_{\xi} (\varphi \circ \psi_{\varepsilon})(s,\xi) + \partial_{s} w_{j}(s,\xi) \partial_{s} (\varphi \circ \psi_{\varepsilon})(s,\xi) \right) \\ &+ \varepsilon \xi \kappa(s) \xi \sum_{j=1}^{+\infty} (j+1) (\varepsilon \xi \kappa(s))^{j-1} \partial_{s} w_{j}(s,\xi) \partial_{s} (\varphi \circ \psi_{\varepsilon})(s,\xi) \right) \left( 1 - \varepsilon \xi \kappa(s) \right) d\xi ds \\ &\leq \int_{0}^{|\partial\Omega|} \int_{0}^{1} \partial_{\xi} w_{j}(s,\xi) \partial_{\xi} (\varphi \circ \psi_{\varepsilon})(s,\xi) d\xi ds + C\varepsilon \|\nabla\varphi\|_{L^{2}(\Omega)} \\ &= - \frac{M\mu_{j}}{|\partial\Omega|} \int_{\partial\Omega} u_{j} \varphi d\sigma + \frac{M\mu_{j}}{|\partial\Omega|} \int_{0}^{|\partial\Omega|} \int_{0}^{1} (u_{j} \circ \psi_{\varepsilon}(s,0)) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \\ &+ C\varepsilon \|\nabla\varphi\|_{L^{2}(\Omega)}. \end{split}$$

Thus inequality (4.2.18) is proved.

We prove now that  $||u_j + \varepsilon v_j||_{\varepsilon}$  is uniformly bounded for  $\varepsilon \in ]0, \varepsilon_0[$ . This follows by the following lemma.

**Lemma 4.2.24.** Let  $u_j$  be the unique eigenfunction of (4.0.1) associated with the eigenvalue  $\mu_j$  such that  $\int_{\partial\Omega} u_j^2 d\sigma = 1$ . Then there exists a constant C which does not depend on  $\varepsilon$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$  it holds

$$\|u_j + \varepsilon v_j\|_{\varepsilon}^2 = \frac{M}{|\partial\Omega|}(1+\mu_j) + O(\varepsilon), \qquad (4.2.25)$$

 $as \; \varepsilon \to 0.$ 

*Proof.* We use the explicit formula for  $\|\cdot\|_{\varepsilon}$  and we write

$$||u_j + \varepsilon v_j||_{\varepsilon}^2 = L_{1,\varepsilon} + L_{2,\varepsilon} + L_{3,\varepsilon},$$

where

$$\begin{split} L_{1,\varepsilon} &:= \int_{\Omega} \rho_{\varepsilon} u_{j}^{2} dx, \\ L_{2,\varepsilon} &:= \int_{\Omega} |\nabla u_{j}|^{2} dx, \\ L_{3,\varepsilon} &:= \varepsilon^{2} \int_{\omega_{\varepsilon}} v_{j}^{2} dx - 2\varepsilon \int_{\omega_{\varepsilon}} u_{j} v_{j} dx + \varepsilon^{2} \int_{\omega_{\varepsilon}} |\nabla v_{j}|^{2} dx - 2\varepsilon \int_{\omega_{\varepsilon}} \nabla u_{j} \cdot \nabla v_{j} dx. \end{split}$$

We consider  $L_{1,\varepsilon} - \frac{M}{|\partial\Omega|}$  first. We have

$$\begin{split} L_{1,\varepsilon} &- \frac{M}{|\partial\Omega|} = L_{1,\varepsilon} - \frac{M}{|\partial\Omega|} \int_{\partial\Omega} u_j^2 d\sigma \\ &= \frac{1}{\varepsilon} \int_{\omega_\varepsilon} \tilde{\rho}_\varepsilon u_j^2 dx - \frac{M}{|\partial\Omega|} \int_{\partial\Omega} u_j^2 d\sigma + \varepsilon \int_{\Omega} u_j^2 dx \\ &= \frac{M}{|\partial\Omega|} \frac{1}{\varepsilon} \int_{\omega_\varepsilon} u_j^2 dx - \frac{M}{|\partial\Omega|} \int_{\partial\Omega} u_j^2 d\sigma \\ &+ \int_{\omega_\varepsilon} \left( \frac{\frac{1}{2}KM - |\Omega| |\partial\Omega|}{|\partial\Omega|^2} + \frac{F(\varepsilon)}{\varepsilon} \right) u_j^2 dx + \varepsilon \int_{\Omega} u_j^2 dx \\ &\leq \frac{M}{|\partial\Omega|} \frac{1}{\varepsilon} \int_{\omega_\varepsilon} u_j^2 dx - \frac{M}{|\partial\Omega|} \int_{\partial\Omega} u_j^2 d\sigma + C ||u_j||_{C(\overline{\Omega})}^2 \varepsilon. \end{split}$$

We consider  $\frac{1}{\varepsilon} \int_{\omega_{\varepsilon}} u_j^2 dx - \int_{\partial \Omega} u_j^2 d\sigma$ . We have

$$\begin{split} \frac{1}{\varepsilon} \int_{\omega_{\varepsilon}} u_j^2 dx &- \int_{\partial \Omega} u_j^2 d\sigma \\ &= \int_0^{|\partial \Omega|} \frac{1}{\varepsilon} \left( \int_0^{\varepsilon} (u_j \circ \psi(s,t))^2 (1 - t\kappa(s)) - (u_j \circ \psi(s,0))^2 dt \right) ds \\ &\leq \int_0^{|\partial \Omega|} \frac{1}{\varepsilon} \left( C \|u_j\|_{C^1(\overline{\Omega})}^2 \int_0^{\varepsilon} t dt \right) ds \leq C\varepsilon. \end{split}$$

We consider now  $L_{2,\varepsilon}$ . Since  $u_j$  is an eigenfunction of (4.0.1) and is normalized by  $\int_{\partial\Omega} u_j^2 d\sigma = 1$ , we have

$$L_{2,\varepsilon} = \int_{\Omega} |\nabla u_j|^2 dx = \frac{M\mu_j}{|\partial\Omega|} \int_{\partial\Omega} u_j^2 d\sigma = \frac{M\mu_j}{|\partial\Omega|}.$$

Finally, it is standard to prove that  $|L_{3,\varepsilon}| \leq C\varepsilon$ . We prove the result for the fourth summand in  $L_{3,\varepsilon}$ . The result for the other summands is obtained in a similar way. It is convenient to use the coordinates (s,t) on  $\omega_{\varepsilon}$ . We have

$$\begin{split} \varepsilon \int_{\omega_{\varepsilon}} \nabla u_j \cdot \nabla v_j dx \\ &= \varepsilon \int_0^{|\partial \Omega|} \int_0^{\varepsilon} \left( \partial_t (u_j \circ \psi)(s,t) \partial_t w_j(s,\frac{t}{\varepsilon}) \right. \\ &+ \frac{\partial_s (u_j \circ \psi)(s,t) \partial_s w_j(s,\frac{t}{\varepsilon})}{(1 - t\kappa(s))^2} \right) (1 - t\kappa(s)) dt ds \\ &\leq C \varepsilon \int_0^{|\partial \Omega|} \int_0^{\varepsilon} \frac{\|u_j\|_{C^1(\overline{\Omega})}^2}{\varepsilon} dt ds \leq C \varepsilon. \end{split}$$

This concludes the proof of Lemma 4.2.24.

In order to apply Lemma 4.2.3 it is sufficient to multiply both sides of (4.2.18) by  $||u_j + \varepsilon v_j||_{\varepsilon}^{-1}$ . Thanks to Lemma 4.2.24 we have

$$\left| \left\langle \mathcal{A}_{\varepsilon} \left( \frac{u_j + \varepsilon v_j}{\|u_j + \varepsilon v_j\|_{\varepsilon}} \right) - \frac{1}{1 + \mu_j} \left( \frac{u_j + \varepsilon v_j}{\|u_j + \varepsilon v_j\|_{\varepsilon}} \right), \varphi \right\rangle_{\varepsilon} \right| \\ \leq C \sqrt{\frac{|\partial \Omega|}{M}} (1 + \mu_j)^{-\frac{1}{2}} \varepsilon \|\varphi\|_{\varepsilon}, \quad \forall \varphi \in H^1(\Omega). \quad (4.2.26)$$

Now we set  $\varphi := \mathcal{A}_{\varepsilon} \left( \frac{u_j + \varepsilon v_j}{\|u_j + \varepsilon v_j\|_{\varepsilon}} \right) - \frac{1}{1 + \mu_j} \left( \frac{u_j + \varepsilon v_j}{\|u_j + \varepsilon v_j\|_{\varepsilon}} \right)$  in (4.2.26) and we obtain

$$\left\|\mathcal{A}_{\varepsilon}\left(\frac{u_{j}+\varepsilon v_{j}}{\|u_{j}+\varepsilon v_{j}\|_{\varepsilon}}\right)-\frac{1}{1+\mu_{j}}\left(\frac{u_{j}+\varepsilon v_{j}}{\|u_{j}+\varepsilon v_{j}\|_{\varepsilon}}\right)\right\|_{\varepsilon}\leq C\sqrt{\frac{|\partial\Omega|}{M}}(1+\mu_{j})^{-\frac{1}{2}}\varepsilon.$$

Then we apply Lemma 4.2.3 with  $\eta = \frac{1}{1+\mu_j}$ ,  $u = \frac{u_j + \varepsilon v_j}{\|u_j + \varepsilon v_j\|_{\varepsilon}}$ ,  $r^* = \delta_j$  and  $r = C\sqrt{\frac{|\partial\Omega|}{M}}(1+\mu_j)^{-\frac{1}{2}}\varepsilon$ . It follows that there exists an eigenvalue  $\eta^*$  of  $\mathcal{A}_{\varepsilon}$  such that

$$\left|\frac{1}{1+\mu_j} - \eta^*\right| \le C\sqrt{\frac{|\partial\Omega|}{M}}(1+\mu_j)^{-\frac{1}{2}}\varepsilon.$$
(4.2.27)

When  $\varepsilon \in ]0, \varepsilon_j[$  (possibly choosing  $\varepsilon_j$  smaller), the only eigenvalue of  $\mathcal{A}_{\varepsilon}$  which satisfies (4.2.27) is  $\frac{1}{1+\lambda_j(\varepsilon)}$ . Thus we have that

$$\lambda_j(\varepsilon) = \mu_j + O(\varepsilon), \qquad (4.2.28)$$

as  $\varepsilon \to 0$ . Let  $u_{j,\varepsilon}$  be the unique eigenfunction of (4.0.2) associated with the eigenvalue  $\lambda_j(\varepsilon)$  such that  $\frac{|\partial \Omega|}{M} \int_{\Omega} \rho_{\varepsilon} u_{j,\varepsilon}^2 dx = 1$ . We observe that

$$\|u_{j,\varepsilon}\|_{\varepsilon}^{2} = \int_{\Omega} \rho_{\varepsilon} u_{j,\varepsilon}^{2} dx + \int_{\Omega} |\nabla u_{j,\varepsilon}|^{2} dx = \frac{M}{|\partial \Omega|} (1 + \lambda_{j}(\varepsilon)).$$

From (4.2.25) and (4.2.28) it follows that there exists a constant c > 0 which does not depend on  $\varepsilon \in ]0, \varepsilon_j[$  such that

$$\left| (1+\lambda_j(\varepsilon))^{\frac{1}{2}} - (1+\mu_j)^{\frac{1}{2}} \right| \le c\varepsilon$$

and

$$\left| \|u_j + \varepsilon v_j\|_{\varepsilon} - \frac{M^{\frac{1}{2}}}{|\partial \Omega|^{\frac{1}{2}}} (1 + \mu_j)^{\frac{1}{2}} \right| \le c\varepsilon.$$

From this fact it follows that

$$\begin{split} \|u_{j,\varepsilon} - u_{j} - \varepsilon v_{j}\|_{L^{2}(\Omega)} &= \frac{M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} (1+\mu_{j})^{\frac{1}{2}} \left\| \left( \frac{\|u_{j,\varepsilon}\|_{\varepsilon}}{\frac{M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} (1+\mu_{j})^{\frac{1}{2}}} \right) \frac{u_{j,\varepsilon}}{\|u_{j,\varepsilon}\|_{\varepsilon}} \\ &- \left( \frac{\|u_{j} + \varepsilon v_{j}\|_{\varepsilon}}{\frac{M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} (1+\mu_{j})^{\frac{1}{2}}} \right) \frac{u_{j} + \varepsilon v_{j}}{\|u_{j} + \varepsilon v_{j}\|_{\varepsilon}} \\ &\leq \frac{M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} (1+\mu_{j})^{\frac{1}{2}} \left\| \frac{u_{j,\varepsilon}}{\|u_{j,\varepsilon}\|_{\varepsilon}} + c'\varepsilon \frac{u_{j,\varepsilon}}{\|u_{j,\varepsilon}\|_{\varepsilon}} - \frac{u_{j} + \varepsilon v_{j}}{\|u_{j} + \varepsilon v_{j}\|_{\varepsilon}} - c'\varepsilon \frac{u_{j} + \varepsilon v_{j}}{\|u_{j} + \varepsilon v_{j}\|_{\varepsilon}} \right\|_{L^{2}(\Omega)} \\ &\leq \frac{M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} (1+\mu_{j})^{\frac{1}{2}} \left\| \frac{u_{j,\varepsilon}}{\|u_{j,\varepsilon}\|_{\varepsilon}} - \frac{u_{j} + \varepsilon v_{j}}{\|u_{j} + \varepsilon v_{j}\|_{\varepsilon}} \right\|_{L^{2}(\Omega)} \\ &+ \frac{M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} (1+\mu_{j})^{\frac{1}{2}} \frac{c'\varepsilon}{\|u_{j,\varepsilon}\|_{\varepsilon}} \|u_{j,\varepsilon}\|_{L^{2}(\Omega)} \\ &+ \frac{M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} (1+\mu_{j})^{\frac{1}{2}} \frac{c'\varepsilon}{\|u_{j} + \varepsilon v_{j}\|_{\varepsilon}} \|u_{j} + \varepsilon v_{j}\|_{L^{2}(\Omega)} \\ &\leq \max\left\{ \frac{C_{\Omega}M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}}, \frac{|\Omega|^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} \right\} (1+\mu_{j})^{\frac{1}{2}} \left\| \frac{u_{j,\varepsilon}}{\|u_{j,\varepsilon}\|_{\varepsilon}} - \frac{u_{j} + \varepsilon v_{j}}{\|u_{j} + \varepsilon v_{j}\|_{\varepsilon}} \right\|_{\varepsilon} \\ &+ \max\left\{ \frac{C_{\Omega}M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}}, \frac{|\Omega|^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} \right\} (1+\mu_{j})^{\frac{1}{2}} \left\| (1+\mu_{j})^{\frac{1}{2}} 2c'\varepsilon, \quad (4.2.29) \end{split} \right\} \end{split}$$

for a suitable c' > 0 which does not depend on  $\varepsilon \in ]0, \varepsilon_j[$ . In the last inequality we have used the fact that for all  $f \in H^1(\Omega), ||f||_{L^2(\Omega)} \leq \max\left\{C_{\Omega}, \frac{|\Omega|^{\frac{1}{2}}}{M^{\frac{1}{2}}}\right\} ||f||_{\varepsilon}$  (see also formula (4.2.20)). From Lemma 4.2.3, it follows that there exists a function  $u^* \in \mathcal{H}_{\varepsilon}(\Omega)$  with  $||u^*||_{\varepsilon} = 1$  belonging to the space generated by all the eigenfunctions associated with the eigenvalues lying in the segment  $\left[\frac{1}{1+\mu_j} - \delta_j, \frac{1}{1+\mu_j} + \delta_j\right]$  such that

$$\left\| u^* - \frac{u_j + \varepsilon v_j}{\|u_j + \varepsilon v_j\|_{\varepsilon}} \right\|_{\varepsilon} \le \frac{2C}{\delta_j} \sqrt{\frac{|\partial\Omega|}{M}} (1 + \mu_j)^{\frac{1}{2}} \varepsilon.$$

Since for all  $\varepsilon \in ]0, \varepsilon_j[, \frac{1}{1+\lambda_j(\varepsilon)}$  is the only eigenvalue of  $\mathcal{A}_{\varepsilon}$  lying in the segment  $\left[\frac{1}{1+\mu_j} - \delta_j, \frac{1}{1+\mu_j} + \delta_j\right]$ , and  $\lambda_j(\varepsilon)$  is simple, it follows that necessarily  $u^* = \frac{u_{j,\varepsilon}}{\|u_{j,\varepsilon}\|_{\varepsilon}}$ . Thus, there exists  $c_j > 0$  (possibly depending on j) and which does not depend on  $\varepsilon$ , such that

$$\left\|\frac{u_{j,\varepsilon}}{\|u_{j,\varepsilon}\|_{\varepsilon}} - \frac{u_j + \varepsilon v_j}{\|u_j + \varepsilon v_j\|_{\varepsilon}}\right\|_{\varepsilon} \le c_j \varepsilon,$$

and therefore

$$\|u_{j,\varepsilon} - u_j - \varepsilon v_j\|_{L^2(\Omega)} \le \max\left\{\frac{C_{\Omega}M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}}, \frac{|\Omega|^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}}\right\} (1 + \mu_j)^{\frac{1}{2}} (2c' + c_j)\varepsilon.$$
(4.2.30)

This concludes the first step of the proof of Theorems (4.2.10) and (4.2.14).

### 4.2.3 Second Step of the proof of Theorems 4.2.10 and 4.2.14

The aim of this subsection is to complete the justification of (4.2.11) and (4.2.15) and therefore to complete the proof of Theorems 4.2.10 and 4.2.14. Let  $\mu_j^1 \in \mathbb{R}$  be defined as in (4.2.12). Let  $u_j$  be the unique solution of problem (4.0.1) such that  $\int_{\partial\Omega} u_j^2 d\sigma = 1$ . Then by using Theorem 4.2.66 in Subsection 4.2.4, there exists a unique weak solution  $u_j^1$  of the boundary value problem

$$\begin{cases} -\Delta u_j^1 = \mu_j u_j & \text{in } \Omega, \\ \partial_\nu u_j^1 - \frac{M\mu_j}{|\partial\Omega|} u_j^1 = \left(\frac{M\mu_j}{2|\partial\Omega|^2} (K - |\partial\Omega|\kappa) - \frac{2M^2\mu_j^2}{3|\partial\Omega|^2} - \frac{|\Omega|\mu_j}{|\partial\Omega|}\right) + \frac{M\mu_j^1}{|\partial\Omega|} u_j & \text{on } \partial\Omega, \end{cases}$$

$$(4.2.31)$$

which satisfies the following conditions

$$\int_{\partial\Omega} u_j^1 u_j d\sigma = \left(\frac{\mu_j^1}{2\mu_j} + \frac{M\mu_j}{3|\partial\Omega|}\right)$$

and

$$\int_{\partial\Omega} \left(u_j^1\right)^2 d\sigma = 1 + \left(\frac{\mu_j^1}{2\mu_j} + \frac{M\mu_j}{3|\partial\Omega|}\right)^2.$$

In fact, from Theorem 4.2.66 it follows that the solution of problem (4.2.31) is unique up to multiples of  $u_j$ . Moreover, by standard elliptic regularity (see e.g., [2]),  $u_j^1 \in C^2(\Omega) \cap C^{1,\alpha}(\overline{\Omega})$ , for all  $\alpha \in ]0, 1[$ .

Next, we introduce the function  $w_j^1(s,\xi)$  from  $[0, |\partial \Omega|[\times]0, 1[$  to  $\mathbb{R}$  defined by

$$w_j^1(s,\xi) := -\frac{\kappa(s)M\mu_j}{6|\partial\Omega|} (u_j \circ \psi_{\varepsilon}(s,0))(\xi-1)^3 + \frac{M^2\mu_j^2}{24|\partial\Omega|^2} (u_j \circ \psi_{\varepsilon}(s,0))(\xi^2+2\xi+9)(\xi-1)^2 + \left(\frac{|\Omega|\mu_j}{2|\partial\Omega|} (u_j \circ \psi_{\varepsilon}(s,0)) - \frac{M}{2|\partial\Omega|} (\mu_j(u_j^1 \circ \psi_{\varepsilon}(s,0)) + \mu_j^1 (u_j \circ \psi_{\varepsilon}(s,0))) - \frac{KM\mu_j}{4|\partial\Omega|^2} (u_j \circ \psi_{\varepsilon}(s,0))\right) (\xi-1)^2, \quad (4.2.32)$$

for all  $(s,\xi) \in [0, |\partial\Omega|[\times]0, 1[$ . (See also (4.2.13) for the definition of K.) We note that the function  $w_j^1$  solves the following differential equation

$$-\partial_{\xi}^{2}w_{j}^{1}(s,\xi) = -\kappa(s)\partial_{\xi}w_{j}(s,\xi) + \frac{M}{|\partial\Omega|} \left(\mu_{j}(u_{j}^{1}\circ\psi_{\varepsilon}(s,0)) + \mu_{j}w_{j}(s,\xi) + \mu_{j}^{1}(u_{j}\circ\psi_{\varepsilon}(s,0)) - \xi\frac{M\mu_{j}^{2}}{|\partial\Omega|}(u_{j}\circ\psi_{\varepsilon}(s,0)) - \frac{|\Omega|\mu_{j}}{M}(u_{j}\circ\psi_{\varepsilon}(s,0)) + \frac{K\mu_{j}}{2|\partial\Omega|}(u_{j}\circ\psi_{\varepsilon}(s,0))\right)$$
(4.2.33)

for all  $(s,\xi) \in [0,\partial\Omega) \times [0,1[$ . Moreover, on the boundary we have

$$w_j^1(s,1) = \partial_{\xi} w_j^1(s,1) = 0 \tag{4.2.34}$$

for all  $s \in [0, |\partial \Omega|[$ . Now let  $v_j^1 \in H^1(\Omega)$  be defined as in Theorem 4.2.14. We note that by construction  $v_j^1 \in H^1(\Omega)$ . We plan to apply again Lemma 4.2.3. To do so, we prove that there exists a constant C > 0 which does not depend on  $\varepsilon$  such that

$$\left| \left\langle \mathcal{A}_{\varepsilon}(u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}) - \frac{1}{1 + \mu_{j} + \varepsilon \mu_{j}^{1}} (u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}), \varphi \right\rangle_{\varepsilon} \right| \\ \leq C \varepsilon^{2} \|\varphi\|_{\varepsilon}, \quad (4.2.35)$$

for all  $\varphi \in H^1(\Omega)$  and all  $\varepsilon \in ]0, \varepsilon_0[$ . As usual, through this subsection we denote by C a positive constant which does not depend on  $\varepsilon$  and  $\varphi$ . The constant C may eventually be re-defined line by line.

Consider the left-hand side of (4.2.35). We have

$$\left\langle \mathcal{A}_{\varepsilon}(u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}) - \frac{1}{1 + \mu_{j} + \varepsilon \mu_{j}^{1}} (u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}), \varphi \right\rangle_{\varepsilon}$$
$$= \frac{1}{1 + \mu_{j} + \varepsilon \mu_{j}^{1}} \left( (\mu_{j} + \varepsilon \mu_{j}^{1}) \int_{\Omega} \rho_{\varepsilon}(u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}) \varphi dx - \int_{\Omega} \nabla (u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}) \cdot \nabla \varphi dx \right) \quad (4.2.36)$$

We introduce the following quantities:

$$\begin{split} I_{1,\varepsilon} &:= (\mu_j + \varepsilon \mu_j^1) \varepsilon \int_{\Omega} (u_j + \varepsilon u_j^1) \varphi dx, \\ I_{2,\varepsilon} &:= (\mu_j + \varepsilon \mu_j^1) \varepsilon^2 \int_{\omega_{\varepsilon}} (v_j + \varepsilon v_j^1) \varphi dx, \\ I_{3,\varepsilon} &:= (\mu_j + \varepsilon \mu_j^1) \int_{\omega_{\varepsilon}} \frac{\tilde{\rho}_{\varepsilon}}{\varepsilon} (u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1) \varphi dx, \\ I_{4,\varepsilon} &:= \int_{\Omega} \nabla u_j \cdot \nabla \varphi dx, \\ I_{5,\varepsilon} &:= \varepsilon \int_{\Omega} \nabla u_j^1 \cdot \nabla \varphi dx, \\ I_{6,\varepsilon} &:= \int_{\omega_{\varepsilon}} \nabla (\varepsilon v_j + \varepsilon^2 v_j^1) \cdot \nabla \varphi dx. \end{split}$$

We observe that the expression in brackets in the right-hand side of (4.2.36) equals

$$I_{1,\varepsilon} + I_{2,\varepsilon} + I_{3,\varepsilon} - I_{4,\varepsilon} - I_{5,\varepsilon} - I_{6,\varepsilon}.$$

$$(4.2.37)$$

In what follows we shall recall the definition (4.2.8) of  $\tilde{\rho}_{\varepsilon}$ . We consider each term separately. We start from  $I_{1,\varepsilon}$ . We have

$$I_{1,\varepsilon} = I_{1,1,\varepsilon} + I_{1,2,\varepsilon},$$

where

$$I_{1,1,\varepsilon} := \varepsilon \mu_j \int_{\Omega} u_j \varphi dx, \qquad (4.2.38)$$
$$I_{1,2,\varepsilon} := \varepsilon^2 \mu_j^1 \int_{\Omega} u_j \varphi dx + \varepsilon^2 \mu_j \int_{\Omega} u_j^1 \varphi dx + \varepsilon^3 \mu_j^1 \int_{\Omega} u_j^1 \varphi dx.$$

We note that  $I_{1,2,\varepsilon} \leq C\varepsilon^2 \|\varphi\|_{\varepsilon}$ . Indeed, for the first term of  $I_{1,2,\varepsilon}$  we have

$$\varepsilon^{2} \mu_{j}^{1} \int_{\Omega} u_{j} \varphi dx \leq C \varepsilon^{2} \|u_{j}\|_{L^{2}(\Omega)} \|\varphi\|_{L^{2}(\Omega)} \leq C \varepsilon^{2} \|\varphi\|_{\varepsilon}.$$

The other terms can be treated in a similar way (see also (4.2.21)).

We consider now  $I_{2,\varepsilon}$ . Thanks to (4.2.9) we have

$$I_{2,\varepsilon} = (\mu_j + \varepsilon \mu_j^1) \varepsilon^2 \int_{\omega_{\varepsilon}} (v_j + \varepsilon v_j^1) \varphi dx = (\mu_j + \varepsilon \mu_j^1) \frac{\varepsilon^3}{\tilde{\rho}_{\varepsilon}} \int_{\omega_{\varepsilon}} \frac{\tilde{\rho}_{\varepsilon}}{\varepsilon} (v_j + \varepsilon v_j^1) \varphi dx$$
$$\leq (\mu_j + \varepsilon \mu_j^1) \frac{\varepsilon^3}{\tilde{\rho}_{\varepsilon}} \left( \int_{\omega_{\varepsilon}} \frac{\tilde{\rho}_{\varepsilon}}{\varepsilon} (v_j + \varepsilon v_j^1)^2 dx \right)^{\frac{1}{2}} \|\varphi\|_{\varepsilon} \leq C \varepsilon^{\frac{5}{2}} \|\varphi\|_{\varepsilon}. \quad (4.2.39)$$

We consider  $I_{3,\varepsilon}$ . We have

$$I_{3,\varepsilon} = I_{3,1,\varepsilon} + I_{3,2,\varepsilon} + I_{3,3,\varepsilon} + I_{3,4,\varepsilon} + I_{3,5,\varepsilon},$$

where

$$\begin{split} I_{3,1,\varepsilon} &:= \mu_j \int_{\omega_{\varepsilon}} \frac{\tilde{\rho}_{\varepsilon}}{\varepsilon} u_j \varphi dx, \\ I_{3,2,\varepsilon} &:= \mu_j \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} u_j^1 \varphi dx, \\ I_{3,3,\varepsilon} &:= \mu_j \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} v_j \varphi dx, \\ I_{3,4,\varepsilon} &:= \mu_j^1 \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} u_j \varphi dx, \\ I_{3,5\varepsilon} &:= \mu_j \varepsilon \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} v_j^1 \varphi dx + \mu_j^1 \varepsilon \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} u_j^1 \varphi dx + \mu_j^1 \varepsilon \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} v_j \varphi dx \\ &+ \mu_j^1 \varepsilon^2 \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} v_j^1 \varphi dx. \end{split}$$

We consider first  $I^1_{3,1,\varepsilon}$  and use formula (4.2.22). We have

$$I_{3,1,\varepsilon} = \mu_j \int_{\omega_{\varepsilon}} \frac{M}{\varepsilon |\partial\Omega|} u_j \varphi dx + \mu_j \int_{\omega_{\varepsilon}} \frac{KM}{2|\partial\Omega|^2} u_j \varphi dx - \mu_j \int_{\omega_{\varepsilon}} \frac{|\Omega|}{|\partial\Omega|} u_j \varphi dx + \mu_j \frac{F(\varepsilon)}{\varepsilon} \int_{\omega_{\varepsilon}} u_j \varphi dx. \quad (4.2.40)$$

The last term in (4.2.40) can be bounded from above by  $C\varepsilon^2 \|\varphi\|_{\varepsilon}$ . In fact we have

$$\mu_j F(\varepsilon) \int_{\omega_\varepsilon} \frac{1}{\varepsilon} u_j \varphi dx = \frac{\mu_j F(\varepsilon)}{\tilde{\rho}_\varepsilon} \int_{\omega_\varepsilon} \frac{\tilde{\rho}_\varepsilon}{\varepsilon} u_j \varphi dx \le \frac{\mu_j F(\varepsilon)}{\tilde{\rho}_\varepsilon} \|u_j\|_\varepsilon \|\varphi\|_\varepsilon \le C\varepsilon^2 \|\varphi\|_\varepsilon.$$

For the first term in (4.2.40) we have

$$\mu_{j} \int_{\omega_{\varepsilon}} \frac{M}{\varepsilon |\partial\Omega|} u_{j} \varphi dx$$

$$= \mu_{j} \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{M}{|\partial\Omega|} (u_{j} \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds$$

$$- \mu_{j} \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{M}{|\partial\Omega|} \xi \kappa(s) (u_{j} \circ \psi_{\varepsilon}(s,\xi)) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds, \quad (4.2.41)$$

while for the second term in (4.2.40) we have

$$\mu_{j} \int_{\omega_{\varepsilon}} \frac{KM}{2|\partial\Omega|^{2}} u_{j}\varphi dx$$

$$= \mu_{j}\varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{KM}{2|\partial\Omega|^{2}} (u_{j} \circ \psi_{\varepsilon})(\varphi \circ \psi_{\varepsilon})d\xi ds$$

$$- \mu_{j}\varepsilon^{2} \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{KM}{2|\partial\Omega|^{2}} \xi \kappa(s)(u_{j} \circ \psi_{\varepsilon}(s,\xi))(\varphi \circ \psi_{\varepsilon}(s,\xi))d\xi ds$$

$$\leq \mu_{j}\varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{KM}{2|\partial\Omega|^{2}} (u_{j} \circ \psi_{\varepsilon})(\varphi \circ \psi_{\varepsilon})d\xi ds + C\varepsilon^{2} \|\varphi\|_{\varepsilon}. \quad (4.2.42)$$

The last inequality can be proved by observing that

$$\begin{split} -\mu_{j}\varepsilon^{2}\int_{0}^{|\partial\Omega|}\int_{0}^{1}\frac{KM}{2|\partial\Omega|^{2}}\xi\kappa(s)(u_{j}\circ\psi_{\varepsilon}(s,\xi))(\varphi\circ\psi_{\varepsilon}(s,\xi))d\xi ds\\ &\leq \varepsilon^{2}\frac{C}{\tilde{\rho}_{\varepsilon}}\int_{\omega_{\varepsilon}}\frac{\tilde{\rho}_{\varepsilon}}{\varepsilon}u_{j}\varphi dx\leq C\varepsilon^{2}\|u_{j}\|_{\varepsilon}\|\varphi\|_{\varepsilon}. \end{split}$$

In a similar way, for the third term in (4.2.40), we have

$$-\mu_{j} \int_{\omega_{\varepsilon}} \frac{|\Omega|}{|\partial\Omega|} u_{j} \varphi dx$$
  
$$\leq -\mu_{j} \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{|\Omega|}{|\partial\Omega|} (u_{j} \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds + C \varepsilon^{2} \|\varphi\|_{\varepsilon}. \quad (4.2.43)$$

We collect (4.2.41), (4.2.42) and (4.2.43) and we obtain that

$$\begin{split} I_{3,1,\varepsilon} &\leq \mu_j \int_0^{|\partial\Omega|} \int_0^1 \frac{M}{|\partial\Omega|} (u_j \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds \\ &- \mu_j \varepsilon \int_0^{|\partial\Omega|} \int_0^1 \frac{M}{|\partial\Omega|} \xi \kappa(s) (u_j \circ \psi_{\varepsilon}(s,\xi)) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \\ &+ \mu_j \varepsilon \int_0^{|\partial\Omega|} \int_0^1 \frac{KM}{2|\partial\Omega|^2} (u_j \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds \\ &- \mu_j \varepsilon \int_0^{|\partial\Omega|} \int_0^1 \frac{|\Omega|}{|\partial\Omega|} (u_j \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds + C \varepsilon^2 \|\varphi\|_{\varepsilon}. \end{split}$$

We consider now  $I_{3,2,\varepsilon}$ . Thanks to formula (4.2.22), we have

$$\begin{split} I_{3,2,\varepsilon} &= \mu_j \int_{\omega_{\varepsilon}} \frac{M}{|\partial\Omega|} u_j^1 \varphi dx + \mu_j \varepsilon \int_{\omega_{\varepsilon}} \frac{KM}{2|\partial\Omega|^2} u_j^1 \varphi dx \\ &- \mu_j \varepsilon \int_{\omega_{\varepsilon}} \frac{|\Omega|}{|\partial\Omega|} u_j^1 \varphi dx + F(\varepsilon) \varepsilon^2 \int_{\omega_{\varepsilon}} u_j^1 \varphi dx \\ &\leq \mu_j \varepsilon \int_0^{|\partial\Omega|} \int_0^1 \frac{M}{|\partial\Omega|} (u_j^1 \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds + C \varepsilon^2 \|\varphi\|_{\varepsilon}. \end{split}$$

Similarly, for  $I_{3,3,\varepsilon}$  and  $I_{3,4,\varepsilon}$  we obtain

$$I_{3,3,\varepsilon} = \mu_j \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} v_j \varphi dx \le \varepsilon \mu_j \int_0^{|\partial\Omega|} \int_0^1 \frac{M}{|\partial\Omega|} w_j(\varphi \circ \psi_{\varepsilon}) d\xi ds + C\varepsilon^2 \|\varphi\|_{\varepsilon}$$
$$= -\varepsilon \int_0^{|\partial\Omega|} \int_0^1 \frac{M^2 \mu_j^2}{2|\partial\Omega|^2} (u_j \circ \psi_{\varepsilon}(s,0)) (\xi - 1)^2 (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds + C\varepsilon^2 \|\varphi\|_{\varepsilon}.$$

and

$$I_{3,4,\varepsilon} = \mu_j^1 \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} u_j \varphi dx \le \mu_j^1 \varepsilon \int_0^{|\partial \Omega|} \int_0^1 \frac{M}{|\partial \Omega|} (u_j \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds + C \varepsilon^2 \|\varphi\|_{\varepsilon}.$$

We note that that by the definitions of  $u_j^1$ ,  $v_j$  and  $v_j^1$  (see (4.2.31), (4.2.16) and (4.2.32)) one can prove that the norms  $||u_j^1||_{\varepsilon}$ ,  $||v_j||_{\varepsilon}$  and  $||v_j^1||_{\varepsilon}$  are bounded from above, uniformly in  $\varepsilon \in ]0, \varepsilon_0[$ . Then we verify that  $I_{3,5,\varepsilon} \leq C\varepsilon^2$ . We show this inequality only for the first summand in  $I_{3,5,\varepsilon}$ . The proof for the other summands is similar and is accordingly omitted. By the definition of  $\tilde{\rho}_{\varepsilon}$  and by the Cauchy-Shwartz inequality we have

$$\mu_j \varepsilon \int_{\omega_\varepsilon} \tilde{\rho}_\varepsilon v_j^1 \varphi dx = \mu_j \varepsilon^2 \int_{\omega_\varepsilon} \frac{\tilde{\rho}_\varepsilon}{\varepsilon} v_j^1 \varphi dx \le \mu_j \varepsilon^2 C \|v_j^1\|_\varepsilon \|\varphi\|_\varepsilon \le C \varepsilon^2 \|\varphi\|_\varepsilon.$$

We have proved that

 $I_{3,\varepsilon}$ 

$$\leq \mu_{j} \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{M}{|\partial\Omega|} (u_{j} \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds$$

$$-\mu_{j} \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{M}{|\partial\Omega|} \xi \kappa(s) (u_{j} \circ \psi_{\varepsilon}(s,\xi)) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds$$

$$+\mu_{j} \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{KM}{2|\partial\Omega|^{2}} (u_{j} \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds$$

$$-\mu_{j} \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{|\Omega|}{|\partial\Omega|} (u_{j} \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds$$

$$+\mu_{j} \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{M}{|\partial\Omega|^{2}} (u_{j} \circ \psi_{\varepsilon}(s,0)) (\xi - 1)^{2} (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds$$

$$+\mu_{j}^{1} \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{M}{|\partial\Omega|} (u_{j} \circ \psi_{\varepsilon}) (\varphi \circ \psi_{\varepsilon}) d\xi ds + C \varepsilon^{2} \|\varphi\|_{\varepsilon}. \quad (4.2.44)$$

We consider now  $I_{4,\varepsilon}$ . We have

$$I_{4,\varepsilon} = \int_{\Omega} \nabla u_j \cdot \nabla \varphi dx = \int_{\partial \Omega} \frac{M\mu_j}{|\partial \Omega|} u_j \varphi d\sigma.$$
(4.2.45)

Next we consider  $I_{5,\varepsilon}$ 

$$\begin{split} I_{5,\varepsilon} &= \varepsilon \int_{\Omega} \nabla u_{j}^{1} \cdot \nabla \varphi dx = \varepsilon \int_{\Omega} \mu_{j} u_{j} \varphi dx \\ &+ \varepsilon \int_{\partial \Omega} \left( \frac{M\mu_{j}}{2|\partial \Omega|^{2}} (K - |\partial \Omega| \kappa(s)) - \frac{2M^{2}\mu_{j}^{2}}{3|\partial \Omega|^{2}} + \frac{M\mu_{j}^{1} - |\Omega|\mu_{j}}{|\partial \Omega|} \right) u_{j} \varphi d\sigma \\ &+ \varepsilon \int_{\partial \Omega} \frac{M\mu_{j}}{|\partial \Omega|} u_{j}^{1} \varphi d\sigma. \quad (4.2.46) \end{split}$$

Now we consider the quantity  $I_{6,\varepsilon}$ .

$$I_{6,\varepsilon} = I_{6,1,\varepsilon} + I_{6,2,\varepsilon},$$

where

$$I_{6,1,\varepsilon} := \varepsilon \int_{\omega_{\varepsilon}} \nabla v_j \cdot \nabla \varphi dx,$$
  
$$I_{6,2,\varepsilon} := \varepsilon^2 \int_{\omega_{\varepsilon}} \nabla v_j^1 \cdot \nabla \varphi dx.$$

Thanks to formula (4.2.7) and integrating by parts with respect to the variable  $\xi$  in the interval ]0, 1[, we have

$$\begin{split} I_{6,1,\varepsilon} &= \varepsilon \int_{\omega_{\varepsilon}} \nabla v_{j} \cdot \nabla \varphi dx \\ &= \varepsilon^{2} \int_{0}^{|\partial\Omega|} \int_{0}^{1} \left( \frac{1}{\varepsilon^{2}} \partial_{\xi} w_{j}(s,\xi) \partial_{\xi}(\varphi \circ \psi_{\varepsilon})(s,\xi) \right. \\ &+ \frac{\partial_{s} w_{j}(s,\xi) \partial_{s}(\varphi \circ \psi_{\varepsilon})(s,\xi)}{(1 - \varepsilon \xi \kappa(s))^{2}} \right) (1 - \varepsilon \xi \kappa(s)) d\xi ds \\ &\leq \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{M \mu_{j}}{|\partial\Omega|} (u_{j} \circ \psi_{\varepsilon}(s,0)) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds - \int_{\partial\Omega} \frac{M \mu_{j}}{|\partial\Omega|} u_{j} \varphi d\sigma \\ &- \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{\mu_{j} M}{|\partial\Omega|} \kappa(s) (2\xi - 1) (u_{j} \circ \psi_{\varepsilon}(s,0)) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds + C\varepsilon^{2} \|\varphi\|_{\varepsilon}. \end{split}$$

$$(4.2.47)$$

Moreover, by an analogous argument, we have

$$\begin{split} I_{6,2,\varepsilon} &= \varepsilon^2 \int_{\omega_{\varepsilon}} \nabla v_j^1 \cdot \nabla \varphi dx \\ &\leq -\varepsilon \int_{\partial \Omega} \left( \frac{M\mu_j (K - |\partial \Omega| \kappa)}{2|\partial \Omega|^2} - \frac{2M^2 \mu_j^2}{3|\partial \Omega|^2} + \frac{M\mu_j^1 - |\Omega| \mu_j}{|\partial \Omega|} \right) u_j \varphi d\sigma \\ &\quad -\varepsilon \int_{\partial \Omega} \frac{M\mu_j}{|\partial \Omega|} u_j^1 \varphi d\sigma \end{split}$$

Neumann vs Steklov: an asymptotic analysis

$$+ \varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{M\mu_{j}\kappa(s)}{|\partial\Omega|} (\xi - 1)(u_{j} \circ \psi_{\varepsilon}(s, 0))(\varphi \circ \psi_{\varepsilon}(s, \xi))d\xi ds$$
  
+  $\varepsilon \int_{0}^{|\partial\Omega|} \int_{0}^{1} \frac{M}{|\partial\Omega|} \Big( \mu_{j}(u_{j}^{1} \circ \psi_{\varepsilon}(s, 0)) - \frac{M\mu_{j}^{2}}{2|\partial\Omega|} (\xi - 1)^{2}(u_{j} \circ \psi_{\varepsilon}(s, 0))$   
+  $\mu_{j}^{1}(u_{j} \circ \psi_{\varepsilon}(s, 0)) - \frac{M\mu_{j}^{2}}{|\partial\Omega|} \xi(u_{j} \circ \psi_{\varepsilon}(s, 0))$   
-  $\frac{|\Omega|\mu_{j}}{M} (u_{j} \circ \psi_{\varepsilon}(s, 0)) + \frac{K\mu_{j}}{2|\partial\Omega|} (u_{j} \circ \psi_{\varepsilon}(s, 0)) \Big) (\varphi \circ \psi_{\varepsilon}(s, \xi))d\xi ds$   
+  $C\varepsilon^{2} \|\varphi\|_{\varepsilon}.$  (4.2.48)

We now recall (4.2.37). By (4.2.38), (4.2.39), (4.2.44), (4.2.45), (4.2.46), (4.2.47) and (4.2.48) we compute the following inequality

$$I_{1,\varepsilon} + I_{2,\varepsilon} + I_{3,\varepsilon} - I_{4,\varepsilon} - I_{5,\varepsilon} - I_{6,\varepsilon}$$

$$\leq \frac{M\mu_j}{|\partial\Omega|} \int_0^{|\partial\Omega|} \int_0^1 \left( (u_j \circ \psi_{\varepsilon}(s,\xi)) - (u_j \circ \psi_{\varepsilon}(s,0)) + \varepsilon \frac{M\mu_j}{|\partial\Omega|} \xi(u_j \circ \psi_{\varepsilon}(s,0)) \right) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \qquad (4.2.49)$$

$$M\mu_i \int_0^{|\partial\Omega|} \int_0^1 d\xi d\xi d\xi = 0$$

$$-\varepsilon \frac{M\mu_j}{|\partial\Omega|} \int_0^1 \int_0^1 ((u_j \circ \psi_{\varepsilon}(s,\xi))) - (u_j \circ \psi_{\varepsilon}(s,0))) (\varphi \circ \psi_{\varepsilon}(s,\xi)) \xi \kappa(s) d\xi ds \qquad (4.2.50)$$

$$\mu_i^1 M \int_0^{|\partial\Omega|} \int_0^1$$

$$+\varepsilon \frac{\gamma^{j-1}}{|\partial\Omega|} \int_{0}^{1} \int_{0}^{1} ((u_{j} \circ \psi_{\varepsilon}(s,\xi))) -(u_{j} \circ \psi_{\varepsilon}(s,0))) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \qquad (4.2.51)$$

$$+\varepsilon \frac{M\mu_j}{|\partial\Omega|} \int_0^{|\partial\Omega|} \int_0^1 \left( (u_j^1 \circ \psi_{\varepsilon}(s,\xi)) - (u_j^1 \circ \psi_{\varepsilon}(s,0)) \right) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds$$

$$(4.2.52)$$

$$-\varepsilon \frac{\mu_j |\Omega|}{|\partial \Omega|} \int_0^{|\partial \Omega|} \int_0^1 \left( (u_j \circ \psi_{\varepsilon}(s,\xi)) - (u_j \circ \psi_{\varepsilon}(s,0)) \right) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds$$

$$(4.2.53)$$

$$+\varepsilon \frac{\mu_j KM}{2|\partial \Omega|^2} \int_0^{|\partial \Omega|} \int_0^1 \left( (u_j \circ \psi_{\varepsilon}(s,\xi)) - (u_j \circ \psi_{\varepsilon}(s,0)) \right) (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds.$$

$$+C\varepsilon^2 \|\varphi\|_{\varepsilon}.$$
(4.2.54)

Since both  $u_j$  and  $u_j^1$  are of class  $C^1(\Omega)$ , we can conclude that the terms (4.2.50)-(4.2.54) can be bounded from above by  $C\varepsilon^2 \|\varphi\|_{\varepsilon}$  (see (4.2.23)). It remains to estimate (4.2.49). We recall that  $u_j \in C^2(\Omega)$ . It is more convenient to consider coordinates (s,t) and the corresponding change of variable  $x = \psi(s,t)$ . We note that  $\partial_t u_j(\psi(s,0)) = -\partial_{\nu} u_j(\gamma(s)) = -\frac{\mu_j M}{|\partial \Omega|} u_j(\psi(s,0))$ .

Therefore, for each  $(s,t) \in [0,\partial\Omega) \times ]0,1[$ , there exist  $t^* \in (0,t)$  and  $t^{**} \in (0,t^*)$  such that

$$(u_j \circ \psi)(s,t) - (u_j \circ \psi)(s,0) = t\partial_t (u_j \circ \psi)(s,t^*)$$

and

$$\partial_t (u_j \circ \psi)(s, t^*) - \partial_t (u_j \circ \psi)(s, 0) = t \partial_t^2 (u_j \circ \psi)(s, t^{**}).$$

Then for (4.2.49) we have

$$\begin{split} \frac{M\mu_j}{|\partial\Omega|} \int_0^{|\partial\Omega|} \int_0^1 \left( (u_j \circ \psi_{\varepsilon}(s,\xi)) - (u_j \circ \psi_{\varepsilon}(s,0)) + \varepsilon \frac{M\mu_j}{|\partial\Omega|} \xi(u_j \circ \psi_{\varepsilon}(s,0)) \right) \\ & \cdot (\varphi \circ \psi_{\varepsilon}(s,\xi)) d\xi ds \\ = \frac{M\mu_j}{|\partial\Omega|} \int_0^{|\partial\Omega|} \frac{1}{\varepsilon} \int_0^{\varepsilon} t(\partial_t (u_j \circ \psi)(s,t^*) - \partial_t (u_j \circ \psi)(s,0)) (\varphi \circ \psi(s,t)) dt ds \\ & \leq \frac{M\mu_j C}{|\partial\Omega|} \frac{1}{\varepsilon} \int_0^{|\partial\Omega|} \int_0^{\varepsilon} t^2 \partial_t^2 (u_j \circ \psi)(s,t^{**}) (\varphi \circ \psi(s,t)) dt ds \\ & \leq \frac{M\mu_j C}{|\partial\Omega|\varepsilon^{\frac{1}{2}}} \|u_j\|_{C^2(\Omega)} \int_0^{|\partial\Omega|} \int_0^{\varepsilon} t^2 \frac{(\varphi \circ \psi(s,t))}{\varepsilon^{\frac{1}{2}}} dt ds \\ & \leq \frac{M\mu_j C}{|\partial\Omega|\sqrt{5}} \varepsilon^2 \|\varphi\|_{\varepsilon}. \end{split}$$

This proves (4.2.35).

We note that  $||u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1||_{\varepsilon}^{-1}$  is uniformly bounded for  $\varepsilon \in ]0, \varepsilon_0[$ . We shall need a more precise estimate of  $||u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1||_{\varepsilon}^{-1}$ , which is the aim of the following lemma.

**Lemma 4.2.55.** Let  $u_j$  be the unique eigenfunction of (4.0.1) associated with the eigenvalue  $\mu_j$  such that  $\int_{\partial\Omega} u_j^2 d\sigma = 1$ . Let  $u_j^1$  be the solution of (4.2.64) which satisfies

$$\int_{\partial\Omega} u_j^1 u_j d\sigma = \left(\frac{\mu_j^1}{2\mu_j} + \frac{M\mu}{3|\partial\Omega|}\right).$$

Then there exists a constant C > 0 which does not depend on  $\varepsilon$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$  it holds

$$\|u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1\|_{\varepsilon}^2 = \frac{M}{|\partial\Omega|} \left(1 + \mu_j + \varepsilon \mu_j^1\right) + O(\varepsilon^2), \qquad (4.2.56)$$

 $as \; \varepsilon \to 0.$ 

*Proof.* We have

$$\|u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1\|_{\varepsilon}^2 = N_{1,\varepsilon} + N_{2,\varepsilon} + N_{3,\varepsilon} + N_{4,\varepsilon} + N_{5,\varepsilon},$$

where

$$\begin{split} N_{1,\varepsilon} &:= \varepsilon \int_{\Omega} \left( u_j^2 + 2\varepsilon u_j u_j^1 + \varepsilon^2 \left( u_j^1 \right)^2 \right) dx, \\ N_{2,\varepsilon} &:= \varepsilon \int_{\omega_{\varepsilon}} \left( \varepsilon^2 v_j^2 + \varepsilon^4 \left( v_j^1 \right)^2 + 2\varepsilon u_j v_j + 2\varepsilon^2 u_j^1 v_j + 2\varepsilon^3 u_j^1 v_j^1 + 2\varepsilon^3 v_j v_j^1 \right) dx, \\ N_{3,\varepsilon} &:= \int_{\omega_{\varepsilon}} \frac{\tilde{\rho}_{\varepsilon}}{\varepsilon} \left( u_j^2 + 2\varepsilon u_j u_j^1 + \varepsilon^2 \left( u_j^1 \right)^2 + \varepsilon^2 v_j^2 + \varepsilon^4 \left( v_j^1 \right)^2 + 2\varepsilon u_j v_j + 2\varepsilon^2 u_j^1 v_j \right) dx, \\ N_{4,\varepsilon} &:= \int_{\Omega} |\nabla u_j|^2 + 2\varepsilon \nabla u_j \cdot \nabla u_j^1 + \varepsilon^2 |\nabla u_j^1|^2 dx, \\ N_{5,\varepsilon} &:= \int_{\omega_{\varepsilon}} \varepsilon^2 |\nabla v_j|^2 + \varepsilon^4 |\nabla v_j^1|^2 + 2\varepsilon \nabla u_j \cdot \nabla v_j + 2\varepsilon^2 \nabla u_j \cdot \nabla v_j^1 \\ &\quad + 2\varepsilon^2 \nabla v_j \nabla u_j^1 + 2\varepsilon^3 \nabla v_j \cdot \nabla v_j^1 + 2\varepsilon^3 \nabla u_j^1 \cdot \nabla v_j^1 dx. \end{split}$$

We start from  $N_{1,\varepsilon}$ . We note that from standard elliptic regularity we have that  $\|u_j\|_{C(\overline{\Omega})}, \|u_j^1\|_{C(\overline{\Omega})} \leq C$ . Therefore it holds

$$N_{1,\varepsilon} \leq \varepsilon \int_{\Omega} u_j^2 dx + C\varepsilon^2.$$

Consider now  $N_{2,\varepsilon}$ . From the definition of  $v_j$  and  $v_j^1$  we have that  $||v_j||_{C(\overline{\omega}_{\varepsilon})}$ ,  $||v_j^1||_{C(\overline{\omega}_{\varepsilon})} \leq C$ . Moreover, from the fact that  $|\omega_{\varepsilon}| \leq C\varepsilon$ , it immediately follows that

$$N_{2,\varepsilon} \leq C\varepsilon^2.$$

Now we consider  $N_{3,\varepsilon}$ . By the same arguments above, we have that the third, fourth, fifth, seventh, eighth, ninth and tenth summands of  $N_{3,\varepsilon}$  can be bounded from above by  $C\varepsilon^2$ . Therefore

$$N_{3,\varepsilon} \le N_{3,\varepsilon}^1 + N_{3,\varepsilon}^2 + N_{3,\varepsilon}^3 + C\varepsilon^2,$$

where

$$\begin{split} N_{3,1,\varepsilon} &:= \int_{\omega_{\varepsilon}} \frac{\tilde{\rho}_{\varepsilon}}{\varepsilon} u_j^2 dx, \\ N_{3,2,\varepsilon} &:= 2 \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} u_j u_j^1 dx, \\ N_{3,3,\varepsilon} &:= 2 \int_{\omega_{\varepsilon}} \tilde{\rho}_{\varepsilon} u_j v_j dx. \end{split}$$

Consider  $N_{3,1,\varepsilon}$  first. We use the expansion (4.2.22) for  $\tilde{\rho}_{\varepsilon}$ . Moreover, since  $u_j \in C^2(\Omega)$  we have that  $(u_j \circ \psi(s,\xi))^2 = (u_j \circ \psi(s,0))^2 + 2tu_j(\psi(s,0))\partial_t(u_j \circ \psi)(s,0) + Ct^2$ , which implies

$$(u_j \circ \psi(s,\xi))^2 = (u_j \circ \psi(s,0))^2 + 2t \frac{M\mu_j}{|\partial\Omega|} (u_j \circ \psi(s,0))^2 + Ct^2,$$
since  $u_j$  is a solution of (4.0.1). Then we have

$$N_{3,1,\varepsilon} = \int_0^{|\partial\Omega|} \int_0^{\varepsilon} \left( \frac{M}{\varepsilon |\partial\Omega|} + \frac{\frac{1}{2}KM - |\Omega| |\partial\Omega|}{|\partial\Omega|^2} + \frac{F(\varepsilon)}{\varepsilon} \right) \\ \cdot \left( (u_j \circ \psi(s,0))^2 + 2t \frac{M\mu_j}{|\partial\Omega|} (u_j \circ \psi(s,0))^2 + Ct^2 \right) (1 - t\kappa(s)) dt ds$$

From standard computations and recalling that  $\int_{\partial\Omega} u_j^2 d\sigma = 1$ , it follows that

$$N_{3,1,\varepsilon} \leq \frac{M}{|\partial\Omega|} + \varepsilon \left(\frac{\frac{1}{2}KM - |\Omega||\partial\Omega|}{|\partial\Omega|^2}\right) - \varepsilon \frac{M^2\mu}{|\partial\Omega|^2} - \varepsilon \frac{M}{2|\partial\Omega|} \int_{\partial\Omega} u_j^2 \kappa d\sigma + C\varepsilon^2.$$

Very similar computations for  $N_{3,2,\varepsilon}$  and  $N_{3,3,\varepsilon}$  yield

$$N_{3,2,\varepsilon} \le \varepsilon \frac{2M}{|\partial\Omega|} \left( \frac{\mu_j^1}{2\mu_j} + \frac{M\mu_j}{3|\partial\Omega|} \right) + C\varepsilon^2$$

and

$$N_{3,3,\varepsilon} \leq -\varepsilon \frac{M^2 \mu_j}{3|\partial \Omega|^2} + C\varepsilon^2.$$

Now we pass to the terms involving the gradients. First we recall that from standard elliptic regularity  $||u_j||_{C^1(\Omega)}, ||u_j^1||_{C^1(\Omega)} \leq C$ . Moreover, from the definition of v and  $v_j$ , it follows that  $||v_j||_{C^1(\Omega)}, ||v_j^1||_{C^1(\Omega)} \leq \frac{C}{\varepsilon}$ . We consider  $N_{4,\varepsilon}$ . We have

$$\begin{split} N_{4,\varepsilon} &\leq \frac{M\mu_j}{|\partial\Omega|} \int_{\partial\Omega} u_j^2 d\sigma + 2\varepsilon \frac{M\mu_j}{|\partial\Omega|} \int_{\partial\Omega} u_j u_j^1 d\sigma + C\varepsilon^2 \\ &\leq \frac{M\mu_j}{|\partial\Omega|} + \varepsilon \left( \frac{M\mu_j^1}{|\partial\Omega|} + \frac{2M^2\mu_j^2}{3|\partial\Omega|^2} \right) + C\varepsilon^2. \end{split}$$

Next we consider  $N_{5,\varepsilon}$ . We note that all the summands but the first and the third can be bounded from above by  $C\varepsilon^2$ . Therefore we have

$$N_{5,\varepsilon} = N_{5,1,\varepsilon} + N_{5,2,\varepsilon} + C\varepsilon^2,$$

where

$$N_{5,1,\varepsilon} := \varepsilon^2 \int_{\omega_{\varepsilon}} |\nabla v_j|^2 dx,$$
  
$$N_{5,2,\varepsilon} := 2\varepsilon \int_{\omega_{\varepsilon}} \nabla u_j \cdot \nabla v_j dx.$$

Consider  $N_{5,\varepsilon}^1$ . Passing to coordinates  $(s,\xi)$  and using formula (4.2.7), from standard computations it follows that

$$N_{5,1,\varepsilon} \le \varepsilon \frac{M^2 \mu_j^2}{3|\partial \Omega|^2} + C\varepsilon^2.$$

In a very similar way one can prove that for  $N_{5,2,\varepsilon}$  it holds

$$N_{5,2,\varepsilon} \leq -\varepsilon \frac{M^2 \mu_j^2}{|\partial \Omega|^2} + C\varepsilon^2.$$

We collect all the expression appearing into  $N_{1,\varepsilon}, N_{2,\varepsilon}, N_{3,\varepsilon}, N_{4,\varepsilon}$  and  $N_{5,\varepsilon}$  according to the powers of  $\varepsilon$  appearing. We have

$$\begin{split} \|u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}\|_{\varepsilon}^{2} \\ & \leq \frac{M}{|\partial\Omega|} \left[ 1 + \mu_{j} \right. \\ & \left. + \varepsilon \left( \frac{|\partial\Omega|}{M} \int_{\Omega} u_{j}^{2} dx - \frac{|\Omega|}{M} - \frac{2M\mu_{j}}{3|\partial\Omega|} + \frac{K}{2|\partial\Omega|} - \frac{1}{2} \int_{\partial\Omega} u_{j}^{2} \kappa d\sigma + \frac{\mu_{j}^{1}}{\mu_{j}} + \mu_{j}^{1} \right) \right] \\ & + C\varepsilon^{2}. \end{split}$$

We note that

$$\frac{\mu_j^1}{\mu_j} = -\frac{|\partial\Omega|}{M} \int_{\Omega} u_j^2 dx + \frac{|\Omega|}{M} - \frac{2M\mu_j}{3|\partial\Omega|} - \frac{K}{2|\partial\Omega|} + \frac{1}{2} \int_{\partial\Omega} u_j^2 \kappa d\sigma,$$

therefore

$$\|u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1\|_{\varepsilon}^2 \le \frac{M}{|\partial\Omega|} \left(1 + \mu_j + \varepsilon \mu_j^1\right) + C\varepsilon^2.$$

This concludes the proof.

As we did in Section 4.2.2, in order to apply Lemma 4.2.3 it is sufficient to multiply both sides of (4.2.35) by  $||u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1||_{\varepsilon}^{-1}$ . From Lemma 4.2.55 it follows that

$$\left| \left\langle \mathcal{A}_{\varepsilon} \left( \frac{u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}}{\|u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}\|_{\varepsilon}} \right) - \frac{1}{1 + \mu_{j} + \varepsilon \mu_{j}^{1}} \left( \frac{u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}}{\|u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}\|_{\varepsilon}} \right), \varphi \right\rangle_{\varepsilon} \right| \\ \leq C \sqrt{\frac{|\partial \Omega|}{M}} (1 + \mu_{j})^{-\frac{1}{2}} \left( 1 - \frac{\mu_{j}^{1}}{2(1 + \mu_{j})} \varepsilon \right) \varepsilon^{2} \|\varphi\|_{\varepsilon}, \quad (4.2.57)$$

for all  $\varphi \in H^1(\Omega)$ . Now we set  $\varphi := \frac{u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1}{\|u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1\|_{\varepsilon}}$  in (4.2.57), which

gives

$$\begin{aligned} \left\| \mathcal{A}_{\varepsilon} \left( \frac{u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}}{\|u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}\|_{\varepsilon}} \right) \\ - \frac{1}{1 + \mu_{j} + \varepsilon \mu_{j}^{1}} \left( \frac{u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}}{\|u_{j} + \varepsilon v_{j} + \varepsilon u_{j}^{1} + \varepsilon^{2} v_{j}^{1}\|_{\varepsilon}} \right) \right\|_{\varepsilon} \\ \leq C \sqrt{\frac{|\partial \Omega|}{M}} (1 + \mu_{j})^{-\frac{1}{2}} \left( 1 - \frac{\mu_{j}^{1}}{2(1 + \mu_{j})} \varepsilon \right) \varepsilon^{2} \end{aligned}$$

From Lemma 4.2.3 applied with  $\eta = \frac{1}{1+\mu_j+\varepsilon\mu_j^1}$ ,  $u = \frac{u_j+\varepsilon v_j+\varepsilon u_j^1+\varepsilon^2 v_j^1}{\|u_j+\varepsilon v_j+\varepsilon u_j^1+\varepsilon^2 v_j^1\|_{\varepsilon}}$  and  $r = C\sqrt{\frac{|\partial\Omega|}{M}}(1+\mu_j)^{-\frac{1}{2}}\left(1-\frac{\mu_j^1}{2(1+\mu_j)}\varepsilon\right)\varepsilon^2$ , it follows that there exists an eigenvalue  $\eta^*$  of  $\mathcal{A}_{\varepsilon}$  such that

$$\left|\frac{1}{1+\mu_j+\varepsilon\mu_j^1}-\eta^*\right| \le C\sqrt{\frac{|\partial\Omega|}{M}}(1+\mu_j)^{-\frac{1}{2}}\left(1-\frac{\mu_j^1}{2(1+\mu_j)}\varepsilon\right)\varepsilon^2.$$
 (4.2.58)

When  $\varepsilon \in ]0, \varepsilon_j[$  (possibly choosing  $\varepsilon_j$  smaller), the only eigenvalue of  $\mathcal{A}_{\varepsilon}$  which satisfies (4.2.58) is  $\frac{1}{1+\lambda_j(\varepsilon)}$ . Thus we have that

$$\lambda_j(\varepsilon) = \mu_j + \varepsilon \mu_j^1 + O(\varepsilon^2), \qquad (4.2.59)$$

as  $\varepsilon \to 0$ . Let  $u_{j,\varepsilon}$  be the unique eigenfunction of (4.0.2) associated with the eigenvalue  $\lambda_j(\varepsilon)$  such that  $\frac{M}{|\partial\Omega|} \int_{\Omega} \rho_{\varepsilon} u_{j,\varepsilon}^2 dx = 1$ . We recall that

$$\|u_{j,\varepsilon}\|_{\varepsilon}^{2} = \frac{M}{|\partial\Omega|} \left(1 + \lambda_{j}(\varepsilon)\right)$$

From (4.2.56) and (4.2.59) it follows that there exists a constant c > 0 which does not depend on  $\varepsilon$  such that

$$|(1+\lambda_j(\varepsilon))^{\frac{1}{2}} - (1+\mu_j + \varepsilon\mu_j^1)^{\frac{1}{2}}| \le c\varepsilon^2$$
(4.2.60)

and

$$\left| \|u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1\|_{\varepsilon} - \frac{M^{\frac{1}{2}}}{|\partial\Omega|^{\frac{1}{2}}} (1 + \mu_j + \varepsilon \mu_j^1)^{\frac{1}{2}} \right| \le c\varepsilon^2.$$
(4.2.61)

Moreover, from Lemma 4.2.3, it follows that for all  $\delta > 0$ , there exists a function  $u^* \in \mathcal{H}_{\varepsilon}(\Omega)$  with  $||u^*||_{\varepsilon} = 1$  belonging to the space generated by all the eigenfunctions associated with the eigenvalues of  $\mathcal{A}_{\varepsilon}$  lying in the segment

$$\left\| \frac{1}{1+\mu_j+\varepsilon\mu_j^1} - \delta, \frac{1}{1+\mu_j+\varepsilon\mu_j^1} + \delta \right\| \text{ such that}$$

$$\left\| u^* - \frac{u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1}{\|u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1\|_{\varepsilon}} \right\|_{\varepsilon}$$

$$\leq \frac{2C}{\delta} \sqrt{\frac{|\partial\Omega|}{M}} (1+\mu_j)^{-\frac{1}{2}} \left( 1 - \frac{\mu_j^1}{2(1+\mu_j)} \varepsilon \right) \varepsilon^2. \quad (4.2.62)$$

By the continuity of the eigenvalues  $\lambda_j(\varepsilon)$  and the simplicity of  $\lambda_j(\varepsilon)$  for all  $\varepsilon \in ]0, \varepsilon_j[$ , it follows that (possibly choosing  $\varepsilon_j > 0$  smaller) there exists  $\delta_j > 0$  such that  $\left|\frac{1}{1+\mu_j+\varepsilon\mu_j^{\rm T}}-\frac{1}{1+\lambda_{j-1}(\varepsilon)}\right| > \delta_j, \left|\frac{1}{1+\mu_j+\varepsilon\mu_j^{\rm T}}-\frac{1}{1+\lambda_{j+1}(\varepsilon)}\right| > \delta_j$ and  $\left|\frac{1}{1+\mu_j+\varepsilon\mu_j^{\rm T}}-\frac{1}{1+\lambda_j(\varepsilon)}\right| \leq \delta_j$ , for all  $\varepsilon \in ]0, \varepsilon_j[$ . Then we choose  $\delta = \delta_j$  in (4.2.62). Since for all  $\varepsilon \in ]0, \varepsilon_j[, \frac{1}{1+\lambda_j(\varepsilon)}$  is the only eigenvalue lying in the segment  $\left[\frac{1}{1+\mu_j+\varepsilon_j^{\rm T}}-\delta_j, \frac{1}{1+\mu_j+\varepsilon\mu_j^{\rm T}}+\delta_j\right]$  and  $\lambda_j(\varepsilon)$  is simple, it follows that necessarily  $u^* = \frac{u_{j,\varepsilon}}{\|u_{j,\varepsilon}\|_{\varepsilon}}$ . Thus there exists  $c_j > 0$  (possibly depending on j) which does not depend on  $\varepsilon$ , such that

$$\left\|\frac{u_{j,\varepsilon}}{\|u_{j,\varepsilon}\|_{\varepsilon}} - \frac{u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1}{\|u_j + \varepsilon v_j + \varepsilon u_j^1 + \varepsilon^2 v_j^1\|_{\varepsilon}}\right\|_{\varepsilon} \le c_j \varepsilon^2.$$
(4.2.63)

Finally, by computations similar to those in (4.2.29), using (4.2.63), (4.2.60), (4.2.61), and following the same lines as in the proof of (4.2.30), it is possible to prove that

$$\|u_{j,\varepsilon} - u_j - \varepsilon v_j - \varepsilon u_j^1 - \varepsilon^2 v_j^1\|_{L^2(\Omega)} \le C(1 + \mu_j + \varepsilon \mu_j^1)\varepsilon^2.$$

This concludes the proof of Theorems 4.2.10 and 4.2.14.

#### **4.2.4** Well-posedness of problem (4.2.31)

Let  $u_j$  be the unique eigenfunction associated with a simple eigenvalue  $\mu_j$  of problem (4.0.1) such that  $\int_{\partial\Omega} u_j^2 d\sigma = 1$ . Then we consider the following problem

$$\begin{cases} -\Delta u = f & \text{in } \Omega\\ \partial_{\nu} u - \frac{M\mu_j}{|\partial\Omega|} u = g_1 + \lambda g_2 & \text{on } \partial\Omega, \end{cases}$$
(4.2.64)

where  $f \in L^2(\Omega)$  and  $g_1, g_2 \in L^2(\partial\Omega)$  are given data which satisfy the condition  $\int_{\partial\Omega} g_2 u_j d\sigma \neq 0$ , while the unknowns are the constant  $\lambda$  and the function u. The weak formulation of problem (4.2.64) reads: find  $(u, \lambda) \in H^1(\Omega) \times \mathbb{R}$  such that

$$\int_{\Omega} \nabla u \cdot \nabla \varphi dx - \frac{M\mu_j}{|\partial\Omega|} \int_{\partial\Omega} u\varphi d\sigma = \int_{\Omega} f\varphi dx + \int_{\partial\Omega} g_1 \varphi d\sigma + \lambda \int_{\partial\Omega} g_2 \varphi d\sigma,$$
(4.2.65)

for all  $\varphi \in H^1(\Omega)$ . We have the following theorem.

**Theorem 4.2.66.** Problem (4.2.64) admits a weak solution  $(u, \lambda) \in H^1(\Omega) \times \mathbb{R}$  if and only if

$$\lambda = -\left(\int_{\Omega} f u_j dx + \int_{\partial \Omega} g_1 u_j d\sigma\right) \left(\int_{\partial \Omega} g_2 u_j d\sigma\right)^{-1}.$$
 (4.2.67)

Moreover, given a solution u of (4.2.64), all the solutions of (4.2.64) are given by  $u + Au_j$  with  $A \in \mathbb{R}$ .

*Proof.* Consider the operator  $\mathcal{A}_1$  from  $H^1(\Omega)$  to  $H^1(\Omega)'$  which takes  $u \in H^1(\Omega)$  to the functional  $\mathcal{A}_1[u]$  defined by

$$\mathcal{A}_1[u][\varphi] := \int_{\Omega} \nabla u \cdot \nabla \varphi dx + \int_{\partial \Omega} u\varphi d\sigma, \quad \varphi \in H^1(\Omega).$$

As is well-known, the operator  $\mathcal{A}_1$  is a homeomorphism from  $H^1(\Omega)$  to  $H^1(\Omega)'$ . Then we consider the trace operator Tr from  $H^1(\Omega)$  to  $L^2(\partial\Omega)$  which is compact and the operator  $\mathcal{J}$  from  $L^2(\partial\Omega)$  to  $H^1(\Omega)'$  defined by

$$\mathcal{J}[u][\varphi] := \int_{\partial\Omega} u\varphi d\sigma, \quad \forall \varphi \in H^1(\Omega).$$

We define the operator  $\mathcal{A}_2$  from  $H^1(\Omega)$  to  $H^1(\Omega)'$  as

$$\mathcal{A}_2 := -\left(1 + \frac{M\mu_j}{|\partial\Omega|}\right) \mathcal{J} \circ \mathrm{Tr}.$$

We define the operator  $\mathcal{A}$  from  $H^1(\Omega)$  to  $H^1(\Omega)'$  as  $\mathcal{A} := \mathcal{A}_1 + \mathcal{A}_2$ . This is the sum of an invertible operator and a compact operator, therefore  $\mathcal{A}$  is a Fredholm operator. Finally we denote by  $B(\lambda)$  the element  $B(\lambda) \in H^1(\Omega)'$ defined by

$$B(\lambda)[\varphi] := \int_{\Omega} f\varphi dx + \int_{\partial\Omega} g_1 \varphi d\sigma + \lambda \int_{\partial\Omega} g_2 \varphi d\sigma, \quad \varphi \in H^1(\Omega).$$

Problem (4.2.65) is recasted as follows: find  $(\lambda, u) \in \mathbb{R} \times H^1(\Omega)$  such that

$$\mathcal{A}[u] = B(\lambda).$$

The kernel of  $\mathcal{A}$  is finite dimensional and is the space of those  $u^*$  such that

$$\int_{\Omega} \nabla u^* \cdot \nabla \varphi dx - \frac{M\mu_j}{|\partial \Omega|} \int_{\partial \Omega} u^* \varphi d\sigma = 0, \quad \forall \varphi \in H^1(\Omega).$$

Since we have assumed that  $\mu_j$  is a simple eigenvalue associated with the eigenfunction  $u_j$ , it follows that the kernel of  $\mathcal{A}$  coincides with the one

dimensional subspace of  $H^1(\Omega)$  generated by  $u_j$ . Therefore, problem (4.2.64) has solution if and only if  $B(\lambda)$  satisfies the equality

$$\int_{\Omega} f u_j dx + \int_{\partial \Omega} g_1 u_j d\sigma + \lambda \int_{\partial \Omega} g_2 u_j d\sigma = 0.$$

Accordingly, problem (4.2.65) has solution if and only if  $\lambda$  is given by (4.2.67). Finally we note that the solution u of problem (4.2.65) is defined up to multiples of  $u_j$ .

#### 4.2.5 The case of the unit ball in $\mathbb{R}^2$

In this subsection we consider the case  $\Omega = B$ , where B is the open unit ball centered at zero in  $\mathbb{R}^2$ . We have already established a formula for the derivatives of all the eigenvalues  $\lambda_j(\varepsilon)$  at  $\varepsilon = 0$  in Section 4.1 (see formula (4.1.3)). We show now that formula (4.2.12) in the case of the unit ball in  $\mathbb{R}^2$  formally is the same as formula (4.1.3). We recall that the eigenvalues of problem (4.0.1) on B are given by

$$\mu_{2j-1} = \mu_{2j} = \frac{2\pi j}{M}, \ j \in \mathbb{N},$$

while  $\mu_0 = 0$ . We note that all the positive eigenvalues have multiplicity two. It is convenient to use polar coordinates  $(r, \theta) \in \mathbb{R}_+ \times \partial B$  in  $\mathbb{R}^2$  and the corresponding change of variables  $x = \phi_s(r, \theta) = (r \cos(\theta), r \sin(\theta))$ . The eigenfunctions associated with the eigenvalue  $\mu_{2j-1} = \mu_{2j}$  are the twodimensional harmonic polynomials  $u_{j,1}, u_{j,2}$  of degree j, which can be written in polar coordinates as

$$u_{j,1}(r,\theta) = r^j \cos(j\theta),$$
  
$$u_{j,2}(r,\theta) = r^j \sin(j\theta).$$

Then we consider problem (4.0.2) when  $\Omega = B$ . It is standard to show (see e.g., Lemma 4.1.4, see also [71, 72]) that all the eigenvalues of problem (4.0.2) on *B* have multiplicity which is an integer multiple of two, except the first one which is zero and has multiplicity one. Moreover, for a fixed  $j \in \mathbb{N}$ , there exists  $\varepsilon_j > 0$  such that  $\lambda_j(\varepsilon)$  has multiplicity two for all  $\varepsilon \in ]0, \varepsilon_j[$  (cfr. Corollary 3.1.42 and [71]). Therefore, from Corollary 3.1.42 (see also [71]) it follows that the positive eigenvalues of (4.0.2) on *B* can be labelled with two indexes *k* and *l* and denoted by  $\lambda_{2k-1,l}(\varepsilon) = \lambda_{2k,l}(\varepsilon)$ , for  $k, l \in \mathbb{N}$ . The corresponding eigenfunctions, which we denote by  $u_{0,l,\varepsilon}, u_{k,l,\varepsilon,1}$  and  $u_{k,l,\varepsilon,2}$ can be written in the following form

$$u_{0,l,\varepsilon} = R_{0,l}(r),$$
  

$$u_{k,l,\varepsilon,1} = R_{k,l}(r)\cos(k\theta),$$
  

$$u_{k,l,\varepsilon,2} = R_{k,l}(r)\sin(k\theta),$$

where  $R_{k,l}(r)$  are suitable linear combinations of Bessel Functions of the first and second species and order k (see Lemma 4.1.4). Moreover, it is possible to prove that  $\lambda_{2k-1,1}(\varepsilon) \to \mu_{2k-1}, \lambda_{2k,1}(\varepsilon) \to \mu_{2k}, \lambda_{2k-1,l}(\varepsilon) \to +\infty$ ,  $\lambda_{2k,l}(\varepsilon) \to +\infty$  for  $l \geq 2$ ,  $u_{k,1,\varepsilon,1} \to u_{k,1}$  and  $u_{\varepsilon,k,1,2} \to u_{k,2}$  in the  $L^2(\Omega)$ norm, as  $\varepsilon \to 0$  (see Theorem 4.1.37).

We note that, in principle, Theorem 4.2.10 could not be applied in this case since all the eigenvalues are multiple. From Theorem 4.1.20 we have that

$$\lambda_{2j-1,1}(\varepsilon) = \mu_{2j-1} + \left(\frac{2j\mu_{2j-1}}{3} + \frac{\mu_{2j-1}^2}{2(j+1)}\right)\varepsilon + o(\varepsilon) = \frac{2\pi j}{M} + \frac{2j^2\pi}{M}\left(\frac{2}{3} + \frac{\pi}{M(1+j)}\right)\varepsilon + o(\varepsilon), \quad (4.2.68)$$

as  $\varepsilon \to 0$ . The same formula holds if we substitute  $\lambda_{2j-1,1}(\varepsilon)$  and  $\mu_{2j-1}$  with  $\lambda_{2j,1}(\varepsilon)$  and  $\mu_{2j}$  respectively.

Now, let us consider formally formula (4.2.12) with  $u_j = \pi^{-\frac{1}{2}}(r^j \cos(j\theta)) \circ \phi_s^{(-1)}$  and observe that the mean curvature on  $\partial B$  is constant end equals 1. Standard computations yield formula (4.2.68). This suggests that in some sense in the case of the ball, Theorem 4.2.10 still holds, despite the multiplicity of the eigenvalues is greater than one. This is not surprising. In fact we could have replaced through all this section the space  $H^1(\Omega)$  with the space  $H_j^1(\Omega)$  of those functions u in  $H^1(\Omega)$  which are orthogonal to  $(r^j \cos(j\theta)) \circ \phi_s^{(-1)}$  with respect to the  $H^1(\Omega)$  scalar product. In this way the eigenvalue  $\mu_{2j-1}$  becomes a simple eigenvalue and Theorem 4.2.10 can be applied. However, this is not straightforward.

The method used in this section is more general than the method used in Section 4.1 and allows to find a formula for the derivative of the eigenvalues  $\lambda(\varepsilon)$  of problem (4.0.2) for a quite wide class of domains in  $\mathbb{R}^2$ .

#### 4.2.6 Heuristic determination of the expansions

In this subsection we show how to guess asymptotic expansions for the eigenvalues and eigenfunctions of problem (4.0.2) of the type (4.2.11) and (4.2.15). Let  $\mu_j$  be a simple eigenvalue of problem (4.0.1) and for  $\varepsilon > 0$  small enough, let  $\lambda_j(\varepsilon)$  be a simple eigenvalue of problem (4.0.2) such that  $\lambda_j(\varepsilon) \to \mu_j$  as  $\varepsilon$  goes to zero (see 3.1.42). Then

$$\lambda_j(\varepsilon) = \mu_j + o(1),$$

as  $\varepsilon \to 0$ . The first step in order to postulate an asymptotic expansion is to guess the powers of  $\varepsilon$  of the lower order terms. In analogy with formula

(4.1.21) for the unit ball in  $\mathbb{R}^N$  we postulate an asymptotic expansion for  $\lambda_i(\varepsilon)$  of the following form

$$\lambda_j(\varepsilon) = \mu_j + \varepsilon \mu_j^1 + o(\varepsilon), \qquad (4.2.69)$$

as  $\varepsilon \to 0$ , i.e., we guess that the second term in the asymptotic expansion of  $\lambda_j(\varepsilon)$  is of order  $\varepsilon$  as  $\varepsilon \to 0$ . Since we want to find a formula for the derivative of the eigenvalues at  $\varepsilon = 0$ , we shall consider the expansion (4.2.69) and neglect further terms.

Now we consider an asymptotic expansion for the eigenfunction  $u_{j,\varepsilon}$  associated with the eigenvalue  $\lambda_i(\varepsilon)$  of the form

$$u_{j,\varepsilon} = u_j + o(1),$$
 (4.2.70)

as  $\varepsilon \to 0$ . The equality in (4.2.70) is understood in the sense of the  $L^2(\Omega)$ norm. In formula (4.2.70) we have considered the eigenfunction  $u_{j,\varepsilon}$  normalized by  $\int_{\Omega} \rho_{\varepsilon} u_{j,\varepsilon}^2 dx = M/|\partial \Omega|$ . We denoted by  $u_j$  the unique eigenfunction of (4.0.1) associated with the eigenvalue  $\mu_j$  normalized by  $\int_{\partial \Omega} u_j^2 d\sigma = 1$ . By looking at (4.2.69) we can argue that the second term in the expansion of  $u_{\varepsilon,j}$  is of order  $\varepsilon$  as  $\varepsilon \to 0$ , therefore we consider an expansion of the following type

$$u_{j,\varepsilon} = u_j + \varepsilon U^1_{\varepsilon,j} + o(\varepsilon), \qquad (4.2.71)$$

as  $\varepsilon \to 0$ , for some function  $U_{\varepsilon,j}^1$  which possibly depends explicitly on  $\varepsilon > 0$ . Since the coefficient  $\rho_{\varepsilon}$  is piecewise constant and is of order  $\varepsilon^{-1}$  in  $\omega_{\varepsilon}$ , we need to introduce in the expansion of the eigenfunctions some correcting terms which are supported on  $\omega_{\varepsilon}$  and which are usually called 'boundary layers' (see [52, 53]). Therefore we consider the function  $U_{\varepsilon,i}^1$  of the following form

$$U_{\varepsilon,j}^1 = u_j^1 + v_j + \varepsilon v_j^1,$$

where  $u_j^1 \in H^1(\Omega)$  is supported on the whole of  $\Omega$  and  $v_j, v_j^1 \in H^1(\Omega)$  are extension by zero of  $w_j \circ \psi_{\varepsilon}^{(-1)}$  and  $w_j^1 \circ \psi_{\varepsilon}^{(-1)}$ , where  $w_j$  and  $w_j^1$  are functions in  $H^1([0, |\partial \Omega|[\times]0, 1[))$ . In particular, we will look for functions  $v_j$  and  $v_j^1$ which are uniformly bounded in  $\varepsilon > 0$ , while their gradients are of order  $\varepsilon^{-1}$ . Therefore the postulated expansion (4.2.71) can be rewritten as

$$u_{j,\varepsilon} = u_j + \varepsilon u_j^1 + \varepsilon v_j + \varepsilon^2 v_j^1 + o(\varepsilon), \qquad (4.2.72)$$

as  $\varepsilon \to 0$ . We note that in formula (4.2.72) a term of order  $\varepsilon^2$  appears. This is not surprising. In order to characterize  $u_j^1, v_j$  and  $v_j^1$ , we shall plug the asymptotic expansion (4.2.72) into (4.0.2), and therefore we shall take derivatives of  $v_j$  and  $v_j^1$ , which are of order  $\varepsilon^{-1}$ . Roungly speaking,  $\varepsilon v_j$  is of order O(1) and  $\varepsilon^2 v_j^1$  is of order  $O(\varepsilon)$  as  $\varepsilon \to 0$ , in the sense of the  $H^1(\omega_{\varepsilon})$ norm. We compute now  $|\omega_{\varepsilon}|$ . We pass to the variables  $(s,\xi)$  and obtain

$$|\omega_{\varepsilon}| = \int_{\omega_{\varepsilon}} dx = \int_{0}^{|\partial\Omega|} \int_{0}^{1} \varepsilon (1 - \varepsilon \xi \kappa(s)) d\xi ds = \varepsilon |\partial\Omega| - \frac{\varepsilon^{2}}{2} \int_{0}^{|\partial\Omega|} \kappa(s) ds.$$

We denoted by K the quantity  $\int_{0}^{|\partial\Omega|} \kappa(s) ds$  (see (4.2.13)). Now we expand in Taylor series the quantity  $\frac{M-\varepsilon |\Omega \setminus \overline{\omega}_{\varepsilon}|}{|\omega_{\varepsilon}|}$  around  $\varepsilon = 0$ . We have

$$\frac{M - \varepsilon |\Omega \setminus \overline{\omega}_{\varepsilon}|}{|\omega_{\varepsilon}|} = \frac{M}{\varepsilon |\partial\Omega|} + \frac{\frac{1}{2}KM - |\Omega||\partial\Omega|}{|\partial\Omega|^2} + O(\varepsilon), \qquad (4.2.73)$$

as  $\varepsilon \to 0$ . Note that  $\rho_{\varepsilon}$  can be written in the following equivalent form

$$\rho_{\varepsilon} = \varepsilon + \frac{1}{\varepsilon} \tilde{\rho}_{\varepsilon} \chi_{\omega_{\varepsilon}},$$

where  $\chi_{\omega_{\varepsilon}}$  is the characteristic function of the set  $\omega_{\varepsilon}$  and  $\tilde{\rho}_{\varepsilon}$  is given by formula (4.2.8). Thanks to formula (4.2.73),  $\tilde{\rho}_{\varepsilon}$  can be written in the form (4.2.22).

We also need to write the Laplace operator in the variables  $(s, \xi)$ . From standard calculus it follows that

$$\Delta = \frac{1}{\varepsilon^2} \partial_{\xi}^2 - \frac{1}{\varepsilon} \kappa(s) \partial_{\xi} - \kappa(s)^2 \xi \partial_{\xi} + \partial_s^2 + \cdots,$$

where the remaining terms are of order  $O(\varepsilon)$  as  $\varepsilon \to 0$ . We denoted by dots further asymptotic terms which are not of use in order to postulate the asymptotic expansions (4.2.69) and (4.2.72).

Suppose for the moment that the functions  $u_j$  and  $u_j^1$  are regular enough. For example, suppose that they are of class  $C^2$ . We have

$$(u_j \circ \psi_{\varepsilon})(s,\xi) = (u_j \circ \psi_{\varepsilon})(s,0) + \xi \partial_{\xi}(u_j \circ \psi)(s,0) + O(\varepsilon^2), \quad (4.2.74)$$
  
$$(u_j^1 \circ \psi_{\varepsilon})(s,\xi) = (u_j^1 \circ \psi_{\varepsilon})(s,0) + O(\varepsilon), \quad (4.2.75)$$

as  $\varepsilon \to 0$ . Note that  $\partial_{\xi}(u_j \circ \psi)(s, 0) = -\varepsilon \partial_{\nu} u_j(\gamma(s))$ . We also recall (see Theorem 4.2.14) that

$$\begin{aligned} w_j(s,\xi) &= (v_j \circ \psi_{\varepsilon})(s,\xi), \\ w_j^1(s,\xi) &= (v_j^1 \circ \psi_{\varepsilon})(s,\xi). \end{aligned}$$

Now we plug (4.2.22), (4.2.69), (4.2.70), (4.2.73), (4.2.74) and (4.2.75) in (4.0.2). We have for the left-hand side of the differential equation in (4.0.2)

$$-\Delta(u_j + \varepsilon u_j^1 + \varepsilon v_j + \varepsilon^2 v_j^1)$$
  
=  $-\left(\Delta u_j + \varepsilon \Delta u_j^1\right) - \left(\frac{1}{\varepsilon}\partial_{\xi}^2 w_j - \kappa(s)\partial_{\xi} w_j + \partial_{\xi}^2 w_j^1 + O(\varepsilon)\right), \quad (4.2.76)$ 

as  $\varepsilon \to 0$ , where the first summand of (4.2.76) is defined in the whole of  $\Omega$ , while the second summand is supported in the strip  $\omega_{\varepsilon}$  and is written in the coordinates  $(s,\xi)$ . For the right-hand side of the differential equation in (4.0.2) we have

$$\rho_{\varepsilon}\lambda_{j}(\varepsilon)(u_{j}+\varepsilon u_{j}^{1}+\varepsilon v_{j}+\varepsilon^{2}v_{j}^{1}) = \left(\varepsilon+\frac{1}{\varepsilon}\tilde{\rho}_{\varepsilon}\chi_{\omega_{\varepsilon}}\right)\left(\mu_{j}+\varepsilon\mu_{j}^{1}+O(\varepsilon^{2})\right)\left(u_{j}+\varepsilon u_{j}^{1}+\varepsilon v_{j}+\varepsilon^{2}v_{j}^{1}\right) = P_{1}+P_{2}.$$
 (4.2.77)

For the reader's convenience, we split formula (4.2.77) into two parts. The first is supported on the whole of  $\Omega$ , the second is supported on  $\omega_{\varepsilon}$  and is expressed in the variables  $(s, \xi)$ . We denote these two parts by  $P_1$  and  $P_2$  respectively. We have

$$P_1 = \varepsilon \mu_j u_j + O(\varepsilon^2) \tag{4.2.78}$$

as  $\varepsilon \to 0$ , and

$$P_{2} = \frac{1}{\varepsilon} \frac{M\mu_{j}}{|\partial\Omega|} (u_{j} \circ \psi_{\varepsilon})(s,0) + \left(\frac{\frac{1}{2}KM - |\Omega||\partial\Omega|}{|\partial\Omega|^{2}}\mu_{j} + \frac{M\mu_{j}^{1}}{|\partial\Omega|}\right) (u_{j} \circ \psi_{\varepsilon})(s,0) - \frac{M\mu_{j}\xi\partial_{\nu}u_{j}(\gamma(s))}{|\partial\Omega|} + \frac{M\mu_{j}}{|\partial\Omega|} (u_{j}^{1} \circ \psi_{\varepsilon})(s,0) + \frac{M\mu_{j}}{|\partial\Omega|}w_{j}(s,\xi) + O(\varepsilon), \quad (4.2.79)$$

as  $\varepsilon \to 0$ . Note that in (4.2.79) and in the second term of (4.2.76) it appears a reminder which is of order  $O(\varepsilon)$  as  $\varepsilon \to 0$ . Moreover, we note that these terms are supported in the strip  $\omega_{\varepsilon}$  which has measure of order  $O(\varepsilon)$  as  $\varepsilon \to 0$ . Therefore, roughly speaking, we can think of the terms of order  $O(\varepsilon)$ supported on  $\omega_{\varepsilon}$  as terms of order  $O(\varepsilon^2)$  as  $\varepsilon \to 0$ .

As for boundary conditions we have

$$\partial_{\nu}u_j(\gamma(s)) - \partial_{\xi}w_j(s,0) + \varepsilon \left(\partial_{\nu}u_j^1(\gamma(s)) - \partial_{\xi}w_j^1(s,0)\right) + O(\varepsilon^2) = 0. \quad (4.2.80)$$

The next step is to match the quantities in (4.2.76), (4.2.78), (4.2.79) and (4.2.80) according to the powers of  $\varepsilon$  which appear. We obtain

$$\begin{cases} \Delta u_j = 0, & \text{on } \Omega, \\ -\Delta u_j^1 = \mu_j u_j, & \text{on } \Omega, \\ -\partial_{\xi}^2 w_j(s,\xi) = \frac{M\mu_j}{|\partial\Omega|} (u_j \circ \psi_{\varepsilon})(s,0), & \text{on } [0, |\partial\Omega|[\times]0, 1[ \nabla u_j \otimes \psi_{\varepsilon})(s,0), \end{cases}$$

and

$$-\partial_{\xi}^{2}w_{j}^{1}(s,\xi) = -\kappa(s)\partial_{\xi}w_{j}(s,\xi) + \frac{M}{|\partial\Omega|} \Big(\mu_{j}(u_{j}^{1}\circ\psi_{\varepsilon})(s,0) + \mu_{j}w_{j}(s,\xi) \\ + \mu_{j}^{1}(u_{j}\circ\psi_{\varepsilon})(s,0) - \mu_{j}\xi\partial_{\nu}(u_{j}(\gamma(s))) \\ - \frac{|\Omega|\mu_{j}}{M}(u_{j}\circ\psi_{\varepsilon})(s,0) + \frac{K\mu_{j}}{2|\partial\Omega|}(u_{j}\circ\psi_{\varepsilon})(s,0)\Big) \text{ on } [0,|\partial\Omega|[\times]0,1[, (4.2.81))$$

while for boundary conditions we have

$$\partial_{\nu} u_j(\gamma(s)) = \partial_{\xi} w_j(s,0), \quad s \in [0, |\partial\Omega|[; \\ \partial_{\nu} u_j^1(\gamma(s)) = \partial_{\xi} w_j^1(s,0), \quad s \in [0, |\partial\Omega|[.$$

$$(4.2.82)$$

Now we write the compatibility conditions which must be satisfied by  $u_j$ ,  $u_j^1$ ,  $v_j$ ,  $v_j^1$ . First, note that

$$\partial_{\xi} w_j(s,1) = \partial_{\xi} w_j^1(s,0) = 0.$$

In fact, we integrate  $-\partial_{\xi}^2 w_j(s,\xi)$  with respect to  $\xi \in ]0,1[$ 

$$-\int_0^1 \partial_{\xi}^2 w_j(s,\xi) d\xi = \frac{M\mu_j}{|\partial\Omega|} (u_j \circ \psi_{\varepsilon})(s,0),$$

therefore

$$\partial_{\xi} w_j(s,0) - \partial_{\xi} w_j(s,1) = \frac{M\mu_j}{|\partial\Omega|} (u_j \circ \psi_{\varepsilon})(s,0),$$

which in view of (4.2.82) reads

$$\partial_{\nu} u_j(\gamma(s)) - \partial_{\xi} w_j(s,1) = \frac{M\mu_j}{|\partial\Omega|} (u_j \circ \psi_{\varepsilon})(s,0).$$

This last fact yields

$$\begin{cases} \partial_{\nu} u_j(\gamma(s)) = \frac{M\mu_j}{|\partial\Omega|} (u_j \circ \psi_{\varepsilon})(s, 0), \\ \partial_{\xi} w_j(s, 1) = 0. \end{cases}$$
(4.2.83)

Note that for each  $s \in [0, |\partial\Omega|[$ , the function  $\xi \mapsto w_j(s, \xi)$  is defined up to constants. We choose  $w_j$  such that  $w_j(s, 1) = 0$  for all  $s \in [0, |\partial\Omega|[$ . With this choice  $w_j$  is uniquely determined and solves problem (4.2.17). Moreover, from the compatibility condition (4.2.83) it follows that  $u_j$  solves problem (4.0.1).

Now we repeat the same procedure for  $w_j^1$ . We integrate (4.2.81) with respect to  $\xi \in ]0,1[$  and from the compatibility conditions, as we did for  $w_j$ , it follows that  $w_j^1$  and  $u_j^1$  solve problems (4.2.33)-(4.2.34) and (4.2.64) respectively.

**Remark 4.2.84.** We have chosen  $w_j$  and  $w_j^1$  to satisfy  $w_j(s, 1) = w_j^1(s, 1) = 0$ , so that  $u_j + \varepsilon u_j^1 + \varepsilon v_j + \varepsilon v_j^1 \in H^1(\Omega)$ .

**Remark 4.2.85.** We have chosen the particular powers of  $\varepsilon$  in (4.2.69) and (4.2.72) since a posteriori the matching of all the terms and the compatibility conditions produce auxiliary problems which are well-posed. If we try to postulate an asymptotic expansion with different powers of  $\varepsilon$  (e.g.,  $\mu_j + \varepsilon^{\frac{1}{2}}\mu_j^1 + O(\varepsilon^{\frac{3}{2}})$  and the analogue for  $u_{j,\varepsilon}$ ), this would lead to define problems the solutions of which are trivial, i.e., identically zero).

Neumann vs Steklov: an asymptotic analysis

# Chapter 5

# Mass concentration phenomena for fourth order operators. A new biharmonic Steklov problem

In this chapter we discuss the eigenvalue problem for the biharmonic operator  $\Delta^2$  subject to Steklov-type and Neumann-type boundary conditions. This operator is related do the study of the bending of a plate via the Kirchoff-Love model (see e.g., [31]).

Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ ,  $N \geq 2$ . We introduce the following Steklov-type problem for the biharmonic operator

$$\begin{cases} \Delta^2 u - \tau \Delta u = 0, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial \Omega, \\ \tau \frac{\partial u}{\partial \nu} - \operatorname{div}_{\partial \Omega} \left( D^2 u \cdot \nu \right) - \frac{\partial \Delta u}{\partial \nu} = \lambda \rho u, & \text{on } \partial \Omega, \end{cases}$$
(5.0.1)

in the unknowns u (the eigenfunction),  $\lambda$  (the eigenvalue), where  $\tau > 0$  is a fixed positive constant,  $\rho \in \mathcal{R}^{S}$ , where  $\mathcal{R}^{S}$  is defined by (3.1.1). Here  $\operatorname{div}_{\partial\Omega}$  denotes the tangential divergence operator and  $D^{2}u$  the Hessian matrix of u. We recall that the tangential divergence  $\operatorname{div}_{\partial\Omega}F$  of a vector field F is defined as  $\operatorname{div}_{\partial\Omega}F = \operatorname{div}_{F|_{\partial\Omega}} - (DF.\nu) \cdot \nu$ , where DF is the Jacobian matrix of F. For N = 2, this problem is related to the study of the vibrations of a thin elastic plate with a free frame and mass displaced on the boundary with density  $\rho$ . The spectrum consists of a diverging sequence of eigenvalues of finite multiplicity

$$0 = \lambda_1 < \lambda_2 \le \dots \le \lambda_j \le \dots , \tag{5.0.2}$$

where we agree to repeat the eigenvalues according to their multiplicity. We note that problem (5.0.1) is the analogue for the biharmonic operator of

the classical Steklov problem (3.1.2) for the Laplace operator considered in Chapter 3 and 4. Then we consider the following Neumann-type problem for the biharmonic operator

$$\begin{cases} \Delta^2 u - \tau \Delta u = \lambda \rho u, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial \Omega, \\ \tau \frac{\partial u}{\partial \nu} - \operatorname{div}_{\partial \Omega} \left( D^2 u \cdot \nu \right) - \frac{\partial \Delta u}{\partial \nu} = 0, & \text{on } \partial \Omega, \end{cases}$$
(5.0.3)

where  $\rho \in \mathcal{R}$  and  $\mathcal{R}$  is defined by (3.1.11) (we refer to [28] for the derivation of the boundary conditions). It is well known that this problem arises in the study of a free vibrating plate whose mass is displaced on the whole of  $\Omega$  with density  $\rho$ . Also in this case the spectrum consist of a diverging sequence of eigenvalues of finite multiplicity as in (5.0.2).

In this chapter we consider the dependence of the eigenvalues  $\lambda_j$  of problem (5.0.1) both on the density  $\rho$  and the domain  $\Omega$ . Moreover we shall understand problem (5.0.1) as the limit of Neumann problems (5.0.3) when the mass  $\rho$  is concentrating at the boundary of  $\Omega$ . This behavior is analogous to that of the Laplace operator with Neumann boundary conditions and mass concentrated in a neighborhood of the boundary considered in Section 3.1.

We remark that problem (5.0.1) should not be confused with other Steklov-type problems already discussed in the literature. For example, in [20] the authors consider the following problem

$$\left\{ \begin{array}{ll} \Delta^2 u = 0, & \mbox{in } \Omega, \\ u = 0, & \mbox{on } \partial\Omega, \\ \Delta u = \lambda \frac{\partial u}{\partial \nu}, & \mbox{on } \partial\Omega, \end{array} \right.$$

which has a rather different nature.

We shall think of the Steklov problem (5.0.1) as the natural fourth order version of problem (3.1.2). For this purpose we briefly derive problem (5.0.1) starting from a physical model (see also [99, 105]).

#### 5.1 Formulating the problem

In this section we provide a physical interpretation of problem (5.0.1) for N = 2 as the equation of a thin vibrating plate. As usual, we assume that the mass of the plate is displaced in the middle plane of the plate parallel to its faces. When the body is at its equilibrium it covers a planar domain  $\Omega$  with boundary  $\partial \Omega$  in  $\mathbb{R}^2$ . We describe the vertical deviation from the equilibrium of each point (x, y) of  $\Omega$  at time t by means of a function v = v(x, y, t). We suppose that the whole mass of the plate is concentrated at the boundary with a density which we denote by  $\rho(x, y)$ . Moreover, we assume that  $\rho(x, y)$  is bounded and positive on  $\partial \Omega$ . Under these assumptions, the total kinetic energy of the plate is given by

$$T = \frac{1}{2} \int_{\partial \Omega} \rho \dot{v}^2 d\sigma,$$

where we denote by  $\dot{v}$  the derivative of v with respect to the time t, and by  $d\sigma$  the surface measure on  $\partial\Omega$ . Now we obtain an expression for the potential energy of the plate. By following [105, §10.8], under the assumption that the strain potential energy at each point depends only on the strain configuration at that point and that the Poisson ratio of the material is zero, we have that the strain potential energy is given by

$$V_{s} = \frac{1}{2} \int_{\Omega} \left( v_{xx}^{2} + v_{yy}^{2} + 2v_{xy}^{2} \right) dx dy.$$

Besides  $V_s$ , we have another term of the potential energy due to the lateral tension

$$V_{\tau} = \frac{\tau}{2} \int_{\Omega} \left( v_x^2 + v_y^2 \right) dx dy,$$

where  $\tau > 0$  is the ratio of lateral tension due to flexural rigidity. The Hamilton's integral in the time interval  $[t_1, t_2]$  of the system is given by

$$\mathcal{H} = \int_{t_1}^{t_2} T - V_s - V_\tau \, dt$$
  
=  $\frac{1}{2} \int_{t_1}^{t_2} \int_{\partial\Omega} \rho \dot{v}^2 d\sigma dt - \frac{1}{2} \int_{t_1}^{t_2} \int_{\Omega} \left( v_{xx}^2 + v_{yy}^2 + 2v_{xy}^2 \right) + \tau \left( v_x^2 + v_y^2 \right) dx dy dt.$ 

According to Hamilton's Variational Principle, the actual motion of the system minimizes such integral. Let v(x, y, t) be a minimizer for  $\mathcal{H}$ . Consider the one-parameter family

$$v(x, y, t) + \varepsilon \eta(x, y, t)$$

where we  $\eta$  is a twice continuously differentiable and such that  $\eta(x, y, t_1) = \eta(x, y, t_2) = 0$ . We consider the Hamiltion'n integral  $\mathcal{H}(\varepsilon)$  given by

$$\begin{aligned} \mathcal{H}(\varepsilon) &= \frac{1}{2} \int_{t_1}^{t_2} \int_{\partial\Omega} \rho(\dot{v} + \varepsilon \dot{\eta})^2 d\sigma dt \\ &\quad - \frac{\tau}{2} \int_{t_1}^{t_2} \int_{\Omega} (v_x + \varepsilon \eta_x)^2 + (v_y + \varepsilon \eta_y)^2 dx dy dt \\ &\quad - \frac{1}{2} \int_{t_1}^{t_2} \int_{\Omega} (v_{xx} + \varepsilon \eta_{xx})^2 + (v_{yy} + \varepsilon \eta_{yy})^2 + 2(v_{xy} + \varepsilon \eta_{xy})^2 dx dy dt. \end{aligned}$$

We have that  $\mathcal{H}(\varepsilon)$  has a minimum at  $\varepsilon = 0$ , therefore  $\frac{d\mathcal{H}}{d\varepsilon}|_{\varepsilon=0} = 0$ . We compute  $\frac{d\mathcal{H}}{d\varepsilon}|_{\varepsilon=0}$ .

$$\frac{d\mathcal{H}}{d\varepsilon}\Big|_{\varepsilon=0} = \int_{t_1}^{t_2} \int_{\partial\Omega} \rho \dot{v} \dot{\eta} d\sigma dt \\
- \int_{t_1}^{t_2} \int_{\Omega} \left( v_{xx} \eta_{xx} + v_{yy} \eta_{yy} + 2v_{xy} \eta_{xy} + \tau (v_x \eta_x + v_y \eta_y) \right) dx dy dt = 0.$$
(5.1.1)

We integrate by parts with respect to the variable t the first summand of (5.1.1). We have

$$\int_{t_1}^{t_2} \int_{\partial\Omega} \rho \dot{v} \dot{\eta} d\sigma dt = -\int_{t_1}^{t_2} \int_{\partial\Omega} \rho \ddot{v} \eta d\sigma dt, \qquad (5.1.2)$$

since  $\eta = 0$  at  $t = t_1, t_2$ . Now we integrate by parts separately all the terms in the second summand of (5.1.1). We have

$$\begin{split} \int_{\Omega} v_{xx} \eta_{xx} dx dy &= \int_{\Omega} v_{xxxx} \eta dx dy + \int_{\partial \Omega} \left[ v_{xx} \eta_{x} - v_{xxx} \eta \right] \nu_{(x)} d\sigma, \\ \int_{\Omega} v_{yy} \eta_{yy} dx dy &= \int_{\Omega} u_{yyyy} \eta dx dy + \int_{\partial \Omega} \left[ v_{yy} \eta_{y} - v_{yyy} \eta \right] \nu_{(y)} d\sigma, \\ 2 \int_{\Omega} u_{xy} \eta_{xy} dx dy &= 2 \int_{\Omega} v_{xxyy} \eta dx dy + \int_{\partial \Omega} \left[ v_{xy} \eta_{y} - u_{xyy} \eta \right] \nu_{(x)} d\sigma \\ &+ \int_{\partial \Omega} \left[ v_{xy} \eta_{x} - v_{xxy} \eta \right] \nu_{(y)} d\sigma, \\ \tau \int_{\Omega} v_{x} \eta_{x} &= -\tau \int_{\Omega} v_{xx} \eta dx dy + \tau \int_{\partial \Omega} v_{x} \eta \nu_{(x)} d\sigma, \\ \tau \int_{\Omega} v_{y} \eta_{y} &= -\tau \int_{\Omega} v_{yy} \eta dx dy + \tau \int_{\partial \Omega} v_{y} \eta \nu_{(y)} d\sigma, \end{split}$$

where we have denoted by  $\nu_{(x)}$  and  $\nu_{(y)}$  the components of the unit outer normal  $\nu$  to  $\partial\Omega$ . The terms involving the integrals over the whole of  $\Omega$  sum up to

$$\int_{\Omega} \left( \Delta^2 v - \tau \Delta v \right) \eta dx dy, \tag{5.1.3}$$

while the boundary terms equal

$$\int_{\partial\Omega} \left( \tau \frac{\partial v}{\partial \nu} - \frac{\partial \Delta v}{\partial \nu} \partial \nu \right) \eta + \left( D^2 v \cdot \nu \right) \cdot \nabla \eta d\sigma.$$
 (5.1.4)

We use the Divergence Theorem to find a more suitable expression for the boundary integral  $\int_{\partial\Omega} (D^2 v \cdot \nu) \cdot \nabla \eta d\sigma$ . We have (see [28, 38] for the details)

$$\int_{\partial\Omega} \left( D^2 v \cdot \nu \right) \cdot \nabla \eta d\sigma = \int_{\partial\Omega} \frac{\partial^2 v}{\partial \nu^2} \frac{\partial \eta}{\partial \nu} - \operatorname{div}_{\partial\Omega} \left( D^2 v \cdot \nu \right)_{\partial\Omega} \eta d\sigma, \qquad (5.1.5)$$

where  $(D^2 v \cdot \nu)_{\partial\Omega} = D^2 v \cdot \nu - \frac{\partial^2 v}{\partial \nu^2} \nu$  is the tangential part of  $D^2 v \cdot \nu$ . From (5.1.2), (5.1.3), (5.1.4) and (5.1.5) it follows that v satisfies

$$-\int_{t_1}^{t_2} \int_{\partial\Omega} \rho \ddot{v} \eta d\sigma dt - \int_{t_1}^{t_2} \int_{\Omega} \eta \left( \Delta^2 v - \tau \Delta v \right) dx dy dt -\int_{t_1}^{t_2} \int_{\partial\Omega} \frac{\partial \eta}{\partial \nu} \frac{\partial^2 v}{\partial \nu^2} - \eta \left( \tau \frac{\partial v}{\partial \nu} - \operatorname{div}_{\partial\Omega} \left( D^2 v \cdot \nu \right)_{\partial\Omega} - \frac{\partial \Delta v}{\partial \nu} \right) d\sigma dt = 0,$$

for all  $\eta \in C^2(\Omega \times [t_1, t_2])$  such that  $\eta(x, y, t_1) = \eta(x, y, t_2) = 0$  and  $(x, y) \in \Omega$ . Since  $\eta$  is arbitrary we obtain

$$\begin{cases} \Delta^2 v - \tau \Delta v = 0, & \text{in } \Omega, \\ \frac{\partial^2 v}{\partial \nu^2} = 0, & \text{on } \partial \Omega, \\ \rho \ddot{v} + \tau \frac{\partial v}{\partial \nu} - \operatorname{div}_{\partial \Omega} \left( D^2 v . \nu \right) - \frac{\partial \Delta v}{\partial \nu} = 0, & \text{on } \partial \Omega, \end{cases}$$
(5.1.6)

for all  $t \in [t_1, t_2]$ . We remark that we have written  $\operatorname{div}_{\partial\Omega}(D^2 v.\nu)$  instead of  $\operatorname{div}_{\partial\Omega}(D^2 v.\nu)_{\partial\Omega}$  since  $(D^2 v.\nu)_{\partial\Omega} = D^2 v.\nu - \frac{\partial^2 v}{\partial\nu^2}\nu$  and  $\frac{\partial^2 v}{\partial\nu^2} = 0$  on  $\partial\Omega$ . We separate the variables and, as is customary, we look for solutions

We separate the variables and, as is customary, we look for solutions to problem (5.1.6) of the form v(x, y, t) = u(x, y)w(t). We find that the temporal component w(t) solves the ordinary differential equation  $-\ddot{w}(t) = \lambda w(t)$  for all  $t \in [t_1, t_2]$ , while the spatial component u solves problem (5.0.1).

Note that in the sequel we shall not put any restriction on the space dimension. Thus  $\Omega$  will always denote a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ , with  $N \geq 2$ .

#### 5.2 The Steklov spectrum

We consider the weak formulation of problem (5.0.1),

$$\int_{\Omega} D^2 u : D^2 \varphi + \tau \nabla u \cdot \nabla \varphi dx = \lambda \int_{\partial \Omega} \rho u \varphi d\sigma \,, \quad \forall \varphi \in H^2(\Omega) \,, \qquad (5.2.1)$$

in the unknowns  $u \in H^2(\Omega), \lambda \in \mathbb{R}$ , where

$$D^{2}u: D^{2}\varphi := \sum_{i,j=1}^{N} \frac{\partial^{2}u}{\partial x_{i}\partial x_{j}} \frac{\partial^{2}\varphi}{\partial x_{i}\partial x_{j}}$$

denotes the Frobenius product. Actually, we will consider a problem in the space  $H^2(\Omega)/\mathbb{R}$  since we need to get rid of the constants, which generate the eigenspace corresponding to the eigenvalue  $\lambda = 0$ . We denote by  $\mathcal{J}_{\rho}^{\mathcal{S}}$  the continuous embedding of  $L^2(\partial\Omega)$  into  $H^2(\Omega)'$  defined by

$$\mathcal{J}_{\rho}^{\mathcal{S}}[u][\varphi] := \int_{\partial\Omega} \rho u \varphi d\sigma, \quad \forall u \in L^{2}(\partial\Omega), \varphi \in H^{2}(\Omega).$$

We set

$$H^{2,\mathcal{S}}_{\rho}(\Omega) := \left\{ u \in H^2(\Omega) : \int_{\partial \Omega} \rho u d\sigma = 0 \right\},$$

and we consider in  $H^2(\Omega)$  the bilinear form

$$\langle u, v \rangle := \int_{\Omega} D^2 u : D^2 v + \tau \nabla u \cdot \nabla v dx.$$
 (5.2.2)

By the Poincaré-Wirtinger Inequality, it turns out that this bilinear form is indeed a scalar product on  $H^{2,S}_{\rho}(\Omega)$  whose induced norm is equivalent to the standard one. In the sequel we will think of the space  $H^{2,S}_{\rho}(\Omega)$  as endowed with the scalar product (5.2.2). Let  $F(\Omega)$  be defined by  $F(\Omega) :=$  $\{G \in H^2(\Omega)' : G[1] = 0\}$ . Then, we consider the operator  $\mathcal{P}^S_{\rho}$  as an operator from  $H^{2,S}_{\rho}(\Omega)$  to  $F(\Omega)$ , defined by

$$\mathcal{P}_{\rho}^{\mathcal{S}}[u][\varphi] := \int_{\Omega} D^2 u : D^2 \varphi + \tau \nabla u \cdot \nabla \varphi dx, \ \forall u \in H^{2,\mathcal{S}}_{\rho}(\Omega), \varphi \in H^2(\Omega).$$
(5.2.3)

It turns out that  $\mathcal{P}_{\rho}^{\mathcal{S}}$  is a homeomorphism of  $H_{\rho}^{2,\mathcal{S}}(\Omega)$  onto  $F(\Omega)$ . We define the operator  $\pi_{\rho}^{\mathcal{S}}$  from  $H^{2}(\Omega)$  to  $H_{\rho}^{2,\mathcal{S}}(\Omega)$  by

$$\pi_{\rho}^{\mathcal{S}}[u] := u - \frac{\int_{\partial\Omega} \rho u d\sigma}{\int_{\partial\Omega} \rho d\sigma}.$$
(5.2.4)

We consider the space  $H^2(\Omega)/\mathbb{R}$  endowed with the bilinear form induced by (5.2.2). Such bilinear form renders  $H^2(\Omega)/\mathbb{R}$  a Hilbert space. We denote by  $\pi_{\rho}^{\sharp,S}$  the map from  $H^2(\Omega)/\mathbb{R}$  onto  $H_{\rho}^{2,S}(\Omega)$  defined by the equality  $\pi_{\rho}^{S} = \pi_{\rho}^{\sharp,S} \circ p$ , where p is the canonical projection of  $H^2(\Omega)$  onto  $H^2(\Omega)/\mathbb{R}$ . The map  $\pi_{\rho}^{\sharp,S}$  turns out to be a homeomorphism. Finally, we define the operator  $T_{\rho}^{S}$  acting on  $H^2(\Omega)/\mathbb{R}$  as follows

$$T_{\rho}^{\mathcal{S}} := (\pi_{\rho}^{\sharp,\mathcal{S}})^{-1} \circ (\mathcal{P}_{\rho}^{\mathcal{S}})^{-1} \circ \mathcal{J}_{\rho}^{\mathcal{S}} \circ \operatorname{Tr} \circ \pi_{\rho}^{\sharp,\mathcal{S}},$$
(5.2.5)

where Tr denotes the trace operator acting from  $H^2(\Omega)$  to  $L^2(\partial \Omega)$ .

**Lemma 5.2.6.** The pair  $(\lambda, u)$  of the set  $(\mathbb{R} \setminus \{0\}) \times (H^{2,\mathcal{S}}_{\rho}(\Omega) \setminus \{0\})$  satisfies (5.2.1) if and only if  $\lambda > 0$  and the pair  $(\lambda^{-1}, p[u])$  of the set  $\mathbb{R} \times ((H^2(\Omega)/\mathbb{R}) \setminus \{0\})$  satisfies the equation

$$\lambda^{-1}p[u] = T^{\mathcal{S}}_{\rho}p[u].$$

We have the following theorem.

**Theorem 5.2.7.** The operator  $T_{\rho}^{S}$  is a non-negative compact self-adjoint operator in  $H^{2}(\Omega)/\mathbb{R}$ , whose eigenvalues coincide with the reciprocals of the positive eigenvalues of problem (5.2.1). In particular, the set of eigenvalues of problem (5.2.1) is contained in  $[0, +\infty[$  and consists of the image of a sequence increasing to  $+\infty$ . Each eigenvalue has finite multiplicity.

*Proof.* For the self-adjointness, it suffices to observe that

$$\begin{split} < T^{\mathcal{S}}_{\rho}u, v >_{H^{2}(\Omega)/\mathbb{R}} = < (\pi^{\sharp,\mathcal{S}}_{\rho})^{-1} \circ (\mathcal{P}^{\mathcal{S}}_{\rho})^{-1} \circ \mathcal{J}^{\mathcal{S}}_{\rho} \circ \operatorname{Tr} \circ \pi^{\sharp,\mathcal{S}}_{\rho}u, v >_{H^{2}(\Omega)/\mathbb{R}} \\ &= \mathcal{P}^{\mathcal{S}}_{\rho}[(\mathcal{P}^{\mathcal{S}}_{\rho})^{-1} \circ \mathcal{J}^{\mathcal{S}}_{\rho} \circ \operatorname{Tr} \circ \pi^{\sharp,\mathcal{S}}_{\rho}u][\pi^{\sharp,\mathcal{S}}_{\rho}v] \\ &= \mathcal{J}^{\mathcal{S}}_{\rho}[\operatorname{Tr} \circ \pi^{\sharp,\mathcal{S}}_{\rho}u][\pi^{\sharp,\mathcal{S}}_{\rho}v], \quad \forall u, v \in H^{2}(\Omega)/\mathbb{R}, \end{split}$$

and that  $\mathcal{J}_{\rho}^{\mathcal{S}}[\operatorname{Tr}\circ\pi_{\rho}^{\sharp,\mathcal{S}}u][\pi_{\rho}^{\sharp,\mathcal{S}}v]$  is symmetric. As for compactness, just observe that the trace operator Tr acting from  $H^{1}(\Omega)$  to  $L^{2}(\partial\Omega)$  is compact. The remaining statements follow by standard spectral theory.

As a consequence we have that the spectrum of (5.2.1) consists of an increasing sequence of non-negative eigenvalues of finite multiplicity. Note that the first positive eigenvalue is  $\lambda_2$  as proved by the following theorem.

**Theorem 5.2.8.** The first eigenvalue  $\lambda_1$  of (5.2.1) is zero and the corresponding eigenfunctions are the constants. Moreover,  $\lambda_2 > 0$ .

*Proof.* It is straightforward to see that constant functions are eigenfunctions of (5.2.1) with eigenvalue  $\lambda = 0$ . Suppose now that u is an eigenfunction corresponding to the eigenvalue  $\lambda = 0$ . Then we have

$$\int_{\Omega} |D^2 u|^2 + \tau |\nabla u|^2 dx = 0,$$

where  $|D^2u|^2 = \sum_{i,j=1}^N \left(\frac{\partial^2 u}{\partial x_i \partial x_j}\right)^2$ . Since  $\nabla u = 0$ , it follows that u is constant. Then the eigenvalue  $\lambda = 0$  has multiplicity one.

Thus  $\lambda_2$  is the first positive eigenvalue of (5.2.1) which is usually called the fundamental tone. Note that we can charactrize  $\lambda_2$  by means of the Rayleigh principle

$$\lambda_2 = \min_{\substack{0 \neq u \in H^2(\Omega) \\ \int_{\partial \Omega} \rho u d\sigma = 0}} \frac{\int_{\Omega} |D^2 u|^2 + \tau |\nabla u|^2 dx}{\int_{\partial \Omega} \rho u^2 d\sigma}.$$
 (5.2.9)

## 5.3 Neumann problem and behavior of Neumann eigenvalues under mass concentration at the boundary

We consider now problem (5.0.3). Let  $\omega_{\varepsilon}$  be the set defined by (3.1.19). We fix a positive number M > 0 and choose in (5.0.3)  $\rho = \rho_{\varepsilon}$ , where  $\rho_{\varepsilon}$ is defined by (3.1.20). If in addition we assume that  $\Omega$  is of class  $C^2$ ,  $\varepsilon_0$ can be chosen in such a way that the map  $x \mapsto x - \nu \varepsilon$  is a diffeomorphism between  $\partial \Omega$  and  $\partial \omega_{\varepsilon} \cap \Omega$  for all  $\varepsilon \in ]0, \varepsilon_0[$  (see Theorem 3.1.27). We note that  $\int_{\Omega} \rho_{\varepsilon} dx = M$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . We refer to the quantity M as the total mass of the body.

We prove, under the hypothesis that  $\Omega$  is of class  $C^2$ , convergence of the eigenvalues and eigenfunctions of problem (5.0.3) with density  $\rho_{\varepsilon}$  to the eigenvalues and eigenfunctions of problem (5.2.1) with constant surface density  $\frac{M}{|\partial\Omega|}$  when the parameter  $\varepsilon$  goes to zero (see Corollary 5.3.12). This provides a further interpretation of problem (5.2.1) as the equation of a free vibrating plate whose mass is concentrated at the boundary in the case of domains of class  $C^2$ .

We consider the weak formulation of problem (5.0.3) with density  $\rho_{\varepsilon}$ ,

$$\int_{\Omega} D^2 u : D^2 \varphi + \tau \nabla u \cdot \nabla \varphi dx = \lambda \int_{\Omega} \rho_{\varepsilon} u \varphi dx, \quad \forall \varphi \in H^2(\Omega), \qquad (5.3.1)$$

in the unknowns  $u \in H^2(\Omega)$ ,  $\lambda \in \mathbb{R}$ . In the sequel we shall recast this problem in  $H^2(\Omega)/\mathbb{R}$  since we need to get rid of the constants, which generate the eigenspace corresponding to the eigenvalue  $\lambda = 0$ . We denote by i the canonical embedding of  $H^2(\Omega)$  into  $L^2(\Omega)$ . We denote by  $\mathcal{J}_{\rho_{\varepsilon}}^{\mathcal{N}}$  the continuous embedding of  $L^2(\Omega)$  into  $H^2(\Omega)'$ , defined by

$$\mathcal{J}^{\mathcal{N}}_{\rho_{\varepsilon}}[u][\varphi] := \int_{\Omega} \rho_{\varepsilon} u \varphi dx, \quad \forall u \in L^{2}(\Omega), \varphi \in H^{2}(\Omega).$$

We set

$$H^{2,\mathcal{N}}_{\rho_{\varepsilon}}(\Omega) := \left\{ u \in H^{2}(\Omega) : \int_{\Omega} u \rho_{\varepsilon} dx = 0 \right\}$$

In the sequel we will think of the space  $H^{2,\mathcal{N}}_{\rho_{\varepsilon}}(\Omega)$  as endowed with the scalar product (5.2.2). This scalar product induces on  $H^{2,\mathcal{N}}_{\rho_{\varepsilon}}(\Omega)$  a norm which is equivalent to the standard one. We denote by  $\pi^{\mathcal{N}}_{\rho_{\varepsilon}}$  the map from  $H^2(\Omega)$  to  $H^{2,\mathcal{N}}_{\rho_{\varepsilon}}(\Omega)$  defined by

$$\pi_{\rho_{\varepsilon}}^{\mathcal{N}}[u] := u - \frac{\int_{\Omega} u \rho_{\varepsilon} dx}{\int_{\Omega} \rho_{\varepsilon} dx}$$

for all  $u \in H^2(\Omega)$ . We denote by  $\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}$  the map from  $H^2(\Omega)/\mathbb{R}$  onto  $H^{2,\mathcal{N}}_{\rho_{\varepsilon}}(\Omega)$ defined by the equality  $\pi_{\rho_{\varepsilon}}^{\mathcal{N}} = \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}} \circ p$ , where p is the canonical projection of  $H^2(\Omega)$  onto  $H^2(\Omega)/\mathbb{R}$ . As in (5.2.3), we consider the operator  $\mathcal{P}_{\rho_{\varepsilon}}^{\mathcal{N}}$  as a map from  $H^{2,\mathcal{N}}_{\rho_{\varepsilon}}(\Omega)$  to  $F(\Omega)$  defined by

$$\mathcal{P}_{\rho_{\varepsilon}}^{\mathcal{N}}[u][\varphi] := \int_{\Omega} D^2 u : D^2 \varphi + \tau \nabla u \cdot \nabla \varphi dx, \ \forall u \in H^{2,\mathcal{N}}_{\rho_{\varepsilon}}(\Omega), \varphi \in H^2(\Omega).$$

It turns out that  $\mathcal{P}_{\rho_{\varepsilon}}^{\mathcal{N}}$  is a linear homeomorphism of  $H_{\rho_{\varepsilon}}^{2,\mathcal{N}}(\Omega)$  onto  $F(\Omega)$ . Finally, let the operator  $T_{\rho_{\varepsilon}}^{\mathcal{N}}$  from  $H^{2}(\Omega)/\mathbb{R}$  to itself be defined by

$$T^{\mathcal{N}}_{\rho_{\varepsilon}} := (\pi^{\sharp,\mathcal{N}}_{\rho_{\varepsilon}})^{-1} \circ (\mathcal{P}^{\mathcal{N}}_{\rho_{\varepsilon}})^{-1} \circ \mathcal{J}^{\mathcal{N}}_{\rho_{\varepsilon}} \circ i \circ \pi^{\sharp,\mathcal{N}}_{\rho_{\varepsilon}}.$$
 (5.3.2)

**Lemma 5.3.3.** The pair  $(\lambda, u)$  of the set  $(\mathbb{R} \setminus \{0\}) \times (H^{2,\mathcal{N}}_{\rho_{\varepsilon}}(\Omega) \setminus \{0\})$ satisfies (5.3.1) if and only if  $\lambda > 0$  and the pair  $(\lambda^{-1}, p[u])$  of the set  $\mathbb{R} \times ((H^2(\Omega)/\mathbb{R}) \setminus \{0\})$  satisfies the equation

$$\lambda^{-1}p[u] = T^{\mathcal{N}}_{\rho_{\varepsilon}}p[u].$$

As in Theorem 5.2.7 it is easy to prove the following theorem.

**Theorem 5.3.4.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$  and  $\varepsilon \in [0, \varepsilon_0[$ . The operator  $T_{\rho_{\varepsilon}}^{\mathcal{N}}$  is a compact self-adjoint operator in  $H^2(\Omega)/\mathbb{R}$  and its eigenvalues coincide with the reciprocals of the positive eigenvalues  $\lambda_j(\rho_{\varepsilon})$  of problem (5.3.1) for all  $j \in \mathbb{N}$ . Moreover, the set of eigenvalues of problem (5.3.1) is contained in  $[0, +\infty[$  and consists of the image of a sequence increasing to  $+\infty$ . Each eigenvalue has finite multiplicity.

We have the following theorem on the spectrum of problem (5.3.1) (see also Theorem 5.2.8).

**Theorem 5.3.5.** The first eigenvalue  $\lambda_1$  of (5.3.1) is zero and the corresponding eigenfunctions are the constants. Moreover,  $\lambda_2 > 0$ .

Now we highlight the relations between problems (5.2.1) and (5.3.1) when  $\Omega$  is of class  $C^2$ . In particular we plan to prove the following theorem.

**Theorem 5.3.6.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^2$ . Let the operators  $T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}}$  and  $T^{\mathcal{N}}_{\rho_{\varepsilon}}$  from  $H^2(\Omega)/\mathbb{R}$  to itself be defined as in (5.2.5) and (5.3.2) respectively. Then  $T^{\mathcal{N}}_{\rho_{\varepsilon}}$  converges in norm to  $T^{\mathcal{S}}_{\frac{M}{|\Omega|}}$  as  $\varepsilon \to 0$ .

In order to prove Theorem 5.3.6 we need the following lemma. We remark that by (5.2.4),  $\pi_c^{\sharp,S} = \pi_1^{\sharp,S}$  for all  $c \in \mathbb{R}$ , with  $c \neq 0$ .

**Lemma 5.3.7.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^2$ . Let  $\rho_{\varepsilon} \in \mathcal{R}$  be as in (3.1.20). Then the following statements hold.

- i) For all  $\varphi \in H^2(\Omega)/\mathbb{R}$ ,  $\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi] \to \pi_1^{\sharp,\mathcal{S}}[\varphi]$  in  $L^2(\Omega)$  (hence also in  $H^2(\Omega)$ ) as  $\varepsilon \to 0$ ;
- ii) If  $u_{\varepsilon} \to u$  in  $H^2(\Omega)/\mathbb{R}$ , then possibly passing to a subsequence  $\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u_{\varepsilon}] \to \pi_{1}^{\sharp,\mathcal{S}}[u]$  in  $L^2(\Omega)$  as  $\varepsilon \to 0$ ;
- iii) Assume that  $u_{\varepsilon}, u, w_{\varepsilon}, w \in H^{2}(\Omega)$  are such that  $u_{\varepsilon} \to u, w_{\varepsilon} \to w$  in  $L^{2}(\Omega)$  and  $\operatorname{Tr}[u_{\varepsilon}] \to \operatorname{Tr}[u], \operatorname{Tr}[w_{\varepsilon}] \to \operatorname{Tr}[w]$  in  $L^{2}(\partial\Omega)$  as  $\varepsilon \to 0$ . Moreover assume that there exists a constant C > 0 such that  $\|\nabla u_{\varepsilon}\|_{L^{2}(\Omega)} \leq C$ ,  $\|\nabla w_{\varepsilon}\|_{L^{2}(\Omega)} \leq C$  for all  $\varepsilon \in ]0, \varepsilon_{0}[$ . Then

$$\int_{\Omega} \rho_{\varepsilon} \left( u_{\varepsilon} - u \right) w_{\varepsilon} dx \to 0,$$

and

$$\int_{\Omega} \rho_{\varepsilon} \left( w_{\varepsilon} - w \right) u dx \to 0,$$

 $as \; \varepsilon \to 0$ 

*Proof.* The proof is analogous to that of Lemma 3.1.28 and accordingly is omitted (see also [72] and references therein for details).  $\Box$ 

Proof of Theorem 5.3.6. The proof is similar to that of Theorem 3.1.21. We repeat it here for completeness. It is sufficient to prove that the family  $\{T^{\mathcal{N}}_{\rho_{\varepsilon}}\}_{\varepsilon\in]0,\varepsilon_{0}[}$  of compact operators, compactly converges to the compact operator  $T^{\mathcal{S}}_{\frac{|}{|\partial\Omega|}}$  (see Definition 3.1.36). This implies, in fact, that

$$\lim_{\varepsilon \to 0} \left\| \left( T^{\mathcal{N}}_{\rho_{\varepsilon}} - T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}} \right)^2 \right\|_{\mathcal{L}(H^2(\Omega)/\mathbb{R}, H^2(\Omega)/\mathbb{R})} = 0.$$
(5.3.8)

Then, since the operators  $\{T_{\rho_{\varepsilon}}^{\mathcal{N}}\}_{\varepsilon\in]0,\varepsilon_0[}$  and  $T_{\frac{M}{|\partial\Omega|}}^{\mathcal{S}}$  are self-adjoint, property (5.3.8) is equivalent to convergence in norm (see [8, 103] for a more detailed discussion on compact convergence of compact operators on Hilbert spaces). We recall that, by definition,  $T_{\rho_{\varepsilon}}^{\mathcal{N}}$  compactly converges to  $T_{\frac{M}{|\partial\Omega|}}^{\mathcal{S}}$  if the following requirements are fulfilled:

- i) if  $||u_{\varepsilon}||_{H^{2}(\Omega)/\mathbb{R}} \leq C$  for all  $\varepsilon \in ]0, \varepsilon_{0}[$ , then the family  $\{T_{\rho_{\varepsilon}}^{\mathcal{N}}u_{\varepsilon}\}_{\varepsilon\in ]0,\varepsilon_{0}[}$  is relatively compact in  $H^{2}(\Omega)/\mathbb{R}$ ;
- ii) if  $u_{\varepsilon} \to u$  in  $H^2(\Omega)/\mathbb{R}$ , then  $T^{\mathcal{N}}_{\rho_{\varepsilon}}u_{\varepsilon} \to T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}}u$  in  $H^2(\Omega)/\mathbb{R}$ .

We prove i) first. Let  $u, \varphi \in H^2(\Omega)/\mathbb{R}$  we have

$$\begin{split} \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u] \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi] dx &= \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \left( \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u] - \pi_{1}^{\sharp,\mathcal{S}}[u] \right) \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi] dx \\ &+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp,\mathcal{S}}[u] \left( \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi] - \pi_{1}^{\sharp,\mathcal{S}}[\varphi] \right) dx \\ &+ \left( \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp,\mathcal{S}}[u] \pi_{1}^{\sharp,\mathcal{S}}[\varphi] dx - \frac{M}{|\partial\Omega|} \int_{\partial\Omega} \pi_{1}^{\sharp,\mathcal{S}}[u] \pi_{1}^{\sharp,\mathcal{S}}[\varphi] d\sigma \right) \\ &+ \frac{M}{|\partial\Omega|} \int_{\partial\Omega} \pi_{1}^{\sharp,\mathcal{S}}[u] \pi_{1}^{\sharp,\mathcal{S}}[\varphi] d\sigma. \quad (5.3.9) \end{split}$$

By Lemma 5.3.7, *ii*) we have that the first and second summands in the right-hand side of (5.3.9) go to zero as  $\varepsilon \to 0$ . As for the third summand, from Lemma 3.1.22 i) applied to the function f = 1 we have that  $|\omega_{\varepsilon}| = \varepsilon |\partial \Omega| + o(\varepsilon)$  as  $\varepsilon \to 0$ . Therefore  $\rho_{\varepsilon} = \frac{M}{\varepsilon |\partial \Omega|} + o(1)$  as  $\varepsilon \to 0$ . Thus, from

Lemma 3.1.22 and formula (3.1.23) it follows that also the third summand of (5.3.9) goes to zero as  $\varepsilon \to 0$ . Moreover, the equality  $(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}})^{-1} \circ (\mathcal{P}_{\rho_{\varepsilon}}^{\mathcal{N}})^{-1} = (\pi_{1}^{\sharp,\mathcal{S}})^{-1} \circ (\mathcal{P}_{1}^{\mathcal{S}})^{-1}$  holds. Therefore, from (5.3.9) it follows that  $T_{\rho_{\varepsilon}}^{\mathcal{N}} u$  is bounded for each  $u \in H^{2}(\Omega)/\mathbb{R}$ . Thus, by Banach-Steinhaus Theorem, there exists C' such that  $\|T_{\rho_{\varepsilon}}^{\mathcal{N}}\|_{\mathcal{L}(H^{2}(\Omega)/\mathbb{R}, H^{2}(\Omega)/\mathbb{R})} \leq C'$  for all  $\varepsilon \in ]0, \varepsilon_{0}[$ . Moreover, since  $\|u_{\varepsilon}\|_{H^{2}(\Omega)/\mathbb{R}} \leq C$  for all  $\varepsilon \in ]0, \varepsilon_{0}[$ , possibly passing to a subsequence, we have that  $u_{\varepsilon} \rightharpoonup u$  in  $H^{2}(\Omega)/\mathbb{R}$ , for some  $u \in H^{2}(\Omega)/\mathbb{R}$ . Then, possibly passing to a subsequence,  $T_{\rho_{\varepsilon}}^{\mathcal{N}}u_{\varepsilon} \rightharpoonup w$  in  $H^{2}(\Omega)/\mathbb{R}$  as  $\varepsilon \to 0$ , for some  $w \in H^{2}(\Omega)/\mathbb{R}$ . We show that  $w = T_{\frac{M}{|\partial\Omega|}}^{\mathcal{S}} u$ . To shorten our notation we set  $w_{\varepsilon} := T_{\rho_{\varepsilon}}^{\mathcal{N}}u_{\varepsilon}$ . By Lemma 5.3.7, i) we have

$$\lim_{\varepsilon \to 0} \int_{\Omega} D^2(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[w_{\varepsilon}]) : D^2(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi]) + \tau \nabla(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[w_{\varepsilon}]) \cdot \nabla(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi]) dx$$
$$= \int_{\Omega} D^2(\pi_1^{\sharp,\mathcal{S}}[w]) : D^2(\pi_1^{\sharp,\mathcal{S}}[\varphi]) + \tau \nabla(\pi_1^{\sharp,\mathcal{S}}[w]) \cdot \nabla(\pi_1^{\sharp,\mathcal{S}}[\varphi]) dx, \quad (5.3.10)$$

for all  $\varphi \in H^2(\Omega)/\mathbb{R}$ . On the other hand, since by (5.3.2)  $\left(\mathcal{P}_{\rho_{\varepsilon}}^{\mathcal{N}} \circ \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}\right) w_{\varepsilon} = \left(\mathcal{J}_{\rho_{\varepsilon}}^{\mathcal{N}} \circ i \circ \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}\right) u_{\varepsilon}$ , we have that

$$\int_{\Omega} D^2(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[w_{\varepsilon}]) : D^2(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi]) + \tau \nabla(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[w_{\varepsilon}]) \cdot \nabla(\pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi]) dx$$
$$= \int_{\Omega} \rho_{\varepsilon} \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[u_{\varepsilon}] \pi_{\rho_{\varepsilon}}^{\sharp,\mathcal{N}}[\varphi] dx \quad (5.3.11)$$

Then, by Lemma 5.3.7, iii, (5.3.10) and (5.3.11) we have

$$< w, \varphi >_{H^{2}(\Omega)/\mathbb{R}} = \lim_{\varepsilon \to 0} < w_{\varepsilon}, \varphi >_{H^{2}(\Omega)/\mathbb{R}} = \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [u_{\varepsilon}] \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [\varphi] dx$$

$$= \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [u_{\varepsilon}] - \pi_{1}^{\sharp, \mathcal{S}} [u] \right) \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [\varphi] dx$$

$$+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}} [u] \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}} [\varphi] - \pi_{1}^{\sharp, \mathcal{S}} [\varphi] \right) dx$$

$$+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}} [u] \pi_{1}^{\sharp, \mathcal{S}} [\varphi] dx = \frac{M}{|\partial \Omega|} \int_{\partial \Omega} \pi_{1}^{\sharp, \mathcal{S}} [u] \pi_{1}^{\sharp, \mathcal{S}} [\varphi] d\sigma$$

$$= < T_{\frac{M}{|\partial \Omega|}}^{\mathcal{S}} u, \varphi >_{H^{2}(\Omega)/\mathbb{R}},$$

hence  $w = T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}} u$ . In a similar way one can prove that  $||w_{\varepsilon}||_{H^{2}(\Omega)/\mathbb{R}} \to 0$ 

 $||w||_{H^2(\Omega)/\mathbb{R}}$ . In fact

$$\begin{split} \lim_{\varepsilon \to 0} \|w_{\varepsilon}\|_{H^{2}(\Omega)/\mathbb{R}}^{2} &= \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[u_{\varepsilon}] - \pi_{1}^{\sharp, \mathcal{S}}[u] \right) \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[w_{\varepsilon}] dx \\ &+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}}[u] \left( \pi_{\rho_{\varepsilon}}^{\sharp, \mathcal{N}}[w_{\varepsilon}] - \pi_{1}^{\sharp, \mathcal{S}}[w_{\varepsilon}] \right) dx \\ &+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}}[u] \left( \pi_{1}^{\sharp, \mathcal{S}}[w_{\varepsilon}] - \pi_{1}^{\sharp, \mathcal{S}}[w] \right) dx \\ &+ \lim_{\varepsilon \to 0} \int_{\Omega} \rho_{\varepsilon} \pi_{1}^{\sharp, \mathcal{S}}[u] \pi_{1}^{\sharp, \mathcal{S}}[w] dx \\ &= \frac{M}{|\partial \Omega|} \int_{\partial \Omega} \pi_{1}^{\sharp, \mathcal{S}}[u] \pi_{1}^{\sharp, \mathcal{S}}[w] d\sigma = \|w\|_{H^{2}(\Omega)/\mathbb{R}}^{2} \,. \end{split}$$

This proves *i*). As for point *ii*), let  $u_{\varepsilon} \to u$  in  $H^2(\Omega)/\mathbb{R}$ . Then there exists C'' such that  $||u_{\varepsilon}||_{H^2(\Omega)/\mathbb{R}} \leq C''$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . Then, by the same argument used for point *i*), for each sequence  $\varepsilon_j \to 0$ , possibly passing to a subsequence, we have  $T^{\mathcal{N}}_{\rho_{\varepsilon_j}} u_{\varepsilon_j} \to T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}} u$ . Since this is true for each  $\{\varepsilon_j\}_{j\in\mathbb{N}}$ , we have the convergence for the whole family, i.e.,  $T^{\mathcal{N}}_{\rho_{\varepsilon}} u_{\varepsilon} \to T^{\mathcal{S}}_{\frac{M}{|\partial\Omega|}} u$ . This concludes the proof.

Thanks to Theorem 3.1.41, as an immediate corollary of Theorem 5.3.6 we have

**Corollary 5.3.12.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^2$ . Let  $\lambda_j[\rho_{\varepsilon}]$  denote the eigenvalues of problem (5.3.1) on  $\Omega$  for all  $j \in \mathbb{N}$ . Let  $\lambda_j, j \in \mathbb{N}$  denote the eigenvalues of problem (5.2.1) corresponding to the constant surface density  $\frac{M}{|\partial \Omega|}$ . Then  $\lim_{\varepsilon \to 0} \lambda_j[\rho_{\varepsilon}] = \lambda_j$  for all  $j \in \mathbb{N}$ .

### 5.4 Symmetric functions of the eigenvalues. Isovolumetric perturbations

In this section we address the problem of the dependece of the eigenvalues of problems (5.0.1) and (5.0.3) upon perturbations of the domain  $\Omega$ .

We shall compute Hadamard-type formulas for both the Steklov and the Neumann problems, which will be used to investigate the behavior of the eigenvalues subject to isovolumetric perturbations. To do so, we use the so called transplantation method, see [60] for a general introduction to this approach. We will study problems (5.0.1) and (5.0.3) on  $\phi(\Omega)$ , for a suitable homeomorphism  $\phi$ , where  $\Omega$  has to be thought as a fixed bounded domain of class  $C^1$ . Therefore, we introduce the following class of functions

$$\Phi(\Omega) := \left\{ \phi \in \left( C^2\left(\overline{\Omega}\right) \right)^N : \phi \text{ injective and } \inf_{\Omega} |\det D\phi| > 0 \right\}.$$

We observe that if  $\Omega$  is of class  $C^1$  and  $\phi \in \Phi(\Omega)$ , then also  $\phi(\Omega)$  is of class  $C^1$  and  $\phi^{(-1)} \in \Phi(\phi(\Omega))$ . Therefore, it makes sense to study both problems (5.0.1) and problem (5.0.3) on  $\phi(\Omega)$ . Moreover, we endow the space  $C^2(\overline{\Omega})$  with the standard norm

$$||f||_{C^{2}(\overline{\Omega})} = \sup_{|\alpha| \le 2, \ x \in \overline{\Omega}} |D^{\alpha}f(x)|.$$

Note that  $\Phi(\Omega)$  is open in  $(C^2(\overline{\Omega}))^N$  (see e.g., [75, Lemma 3.11]).

It is known that balls play a relevant role in the study of isovolumetric perturbations of the domain  $\Omega$  for all the eigenvalues of the Dirichlet and Neumann Laplacian. We refer to [75, 77], where the authors prove that the elementary symmetric functions of the eigenvalues depend real analytically on the domain, providing also Hadamard-type formulas for the corresponding derivatives. Then, in [76] they show that balls are critical points for such functions under volume constraint.

From now on we will consider problems (5.0.1) and (5.0.3) with constant mass density  $\rho \equiv 1$ .

#### 5.4.1 The Steklov problem

We plan to study the Steklov problem on the domain  $\phi(\Omega)$  for  $\phi \in \Phi(\Omega)$ , i.e.,

$$\begin{cases} \Delta^2 u - \tau \Delta u = 0, & \text{in } \phi(\Omega), \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial \phi(\Omega), \\ \tau \frac{\partial u}{\partial \nu} - \operatorname{div}_{\partial \phi(\Omega)} (D^2 u \cdot \nu) - \frac{\partial \Delta u}{\partial \nu} = \lambda u, & \text{on } \partial \phi(\Omega). \end{cases}$$
(5.4.1)

To do so, we pull back the resolvent operator of (5.4.1) to  $\Omega$ . Therefore, we are interested in the operator  $\mathcal{P}_{\phi}^{\mathcal{S}}$  from  $H_{\phi}^{2,\mathcal{S}}(\Omega)$  to  $F(\Omega)$ , defined by

$$\mathcal{P}^{\mathcal{S}}_{\phi}[u][\varphi] := \int_{\Omega} (D^2(u \circ \phi^{-1}) \circ \phi) : (D^2(\varphi \circ \phi^{-1}) \circ \phi) |\det D\phi| dx + \tau \int_{\Omega} (\nabla(u \circ \phi^{-1}) \circ \phi) \cdot (\nabla(\varphi \circ \phi^{-1}) \circ \phi) |\det D\phi| dx, \quad (5.4.2)$$

for all  $u \in H^{2,\mathcal{S}}_{\phi}(\Omega), \, \varphi \in H^2(\Omega)$ , where

$$H^{2,\mathcal{S}}_{\phi}(\Omega) := \left\{ u \in H^2(\Omega) : \int_{\partial \Omega} u |\nu(\nabla \phi)^{-1}| |\det D\phi| d\sigma = 0 \right\}.$$

Moreover, for every  $\phi \in \Phi(\Omega)$ , we consider the map  $\mathcal{J}_{\phi}^{\mathcal{S}}$  from  $L^2(\partial\Omega)$  to  $H^2(\Omega)'$  defined by

$$\mathcal{J}_{\phi}^{\mathcal{S}}[u][\varphi] := \int_{\partial\Omega} u\varphi |\nu(\nabla\phi)^{-1}||\det D\phi| d\sigma, \quad \forall u \in L^{2}(\partial\Omega), \varphi \in H^{2}(\Omega).$$

It is easily seen that the form (5.4.2) is a scalar product on  $H^{2,\mathcal{S}}_{\phi}(\Omega)$ . We will think of the space  $H^{2,\mathcal{S}}_{\phi}(\Omega)$  as endowed with the scalar product (5.4.2). We denote by  $\pi^{\mathcal{S}}_{\phi}$  the map from  $H^2(\Omega)$  to  $H^{2,\mathcal{S}}_{\phi}(\Omega)$  defined by

$$\pi_{\phi}^{\mathcal{S}}(u) := u - \frac{\int_{\partial \Omega} u |\nu(\nabla \phi)^{-1}| |\det D\phi| dx}{\int_{\partial \Omega} |\nu(\nabla \phi)^{-1}| |\det D\phi| dx},$$

and by  $\pi_{\phi}^{\sharp,S}$  the map from  $H^2(\Omega)/\mathbb{R}$  onto  $H^{2,S}_{\phi}(\Omega)$  defined by the equality  $\pi_{\phi}^{S} = \pi_{\phi}^{\sharp,S} \circ p$ . Clearly,  $\pi_{\phi}^{\sharp,S}$  is a homeomorphism, and we can recast problem (5.4.1) as

$$\lambda^{-1}u = W^{\mathcal{S}}_{\phi}u,$$

where

$$W_{\phi}^{\mathcal{S}} := (\pi_{\phi}^{\sharp,\mathcal{S}})^{-1} \circ (\mathcal{P}_{\phi}^{\mathcal{S}})^{-1} \circ \mathcal{J}_{\phi}^{\mathcal{S}} \circ \operatorname{Tr} \circ \pi_{\phi}^{\sharp,\mathcal{S}}.$$

The operator  $W_{\phi}^{S}$  can be shown to be compact and self-adjoint, as we have done for the operator  $T_{\rho}^{S}$  defined by (5.2.5) in Theorem 5.2.7 (see also [77, Theorem 2.1]).

In order to avoid bifurcation phenomena, which usually occur when dealing with multiple eigenvalues, we focus our attention on the elementary symmetric functions of the eigenvalues. This is the aim of the following theorem.

**Theorem 5.4.3.** Let  $\Omega$  be a bounded domain of  $\mathbb{R}^N$  of class  $C^1$ . Let F be a finite non-empty subset of  $\mathbb{N}$ . Let

$$\mathcal{A}_{\Omega}[F] := \{ \phi \in \Phi(\Omega) : \lambda_l[\phi] \notin \{\lambda_j[\phi] : j \in F \} \ \forall l \in \mathbb{N} \setminus F \}.$$

Then the following statements hold.

i) The set  $\mathcal{A}_{\Omega}[F]$  is open in  $\Phi(\Omega)$ . The map  $P_F$  of  $\mathcal{A}_{\Omega}[F]$  to the space  $\mathcal{L}\left(H^2(\Omega), H^2(\Omega)\right)$  which takes  $\phi \in \mathcal{A}_{\Omega}[F]$  to the orthogonal projection of  $H^{2,S}_{\phi}(\Omega)$  onto its (finite dimensional) subspace generated by

$$\left\{ u \in H^{2,\mathcal{S}}_{\phi}(\Omega) : \mathcal{P}^{\mathcal{S}}_{\mathrm{Id}}[u \circ \phi^{-1}] = \lambda_j[\phi] \mathcal{J}^{\mathcal{S}}_{\mathrm{Id}} \circ \mathrm{Tr}[u \circ \phi^{-1}] \text{ for some } j \in F \right\}$$

is real analytic.

ii) Let  $s \in \{1, \ldots, |F|\}$ . The function  $\Lambda_{F,s}$  from  $\mathcal{A}_{\Omega}[F]$  to  $\mathbb{R}$  defined by

$$\Lambda_{F,s}[\phi] := \sum_{j_1 < \dots < j_s \in F} \lambda_{j_1}[\phi] \cdots \lambda_{j_s}[\phi]$$

is real analytic.

iii) Let

$$\Theta_{\Omega}[F] := \{ \phi \in \mathcal{A}_{\Omega}[F] : \lambda_{j}[\phi] \text{ have a common value } \lambda_{F}[\phi] \forall j \in F \}.$$

Then the real analytic functions

$$\left(\binom{|F|}{1}^{-1}\Lambda_{F,1}\right)^{\frac{1}{1}},\ldots,\left(\binom{|F|}{|F|}^{-1}\Lambda_{F,|F|}\right)^{\frac{1}{|F|}},$$

of  $\mathcal{A}_{\Omega}[F]$  to  $\mathbb{R}$  coincide on  $\Theta_{\Omega}[F]$  with the function which takes  $\phi$  to  $\lambda_F[\phi]$ .

*Proof.* The proof can be done adapting that of [77, Theorem 2.2 and Corollary 2.3] (see also [75]).  $\Box$ 

In order to compute explicit formulas for the differentials of the functions  $\Lambda_{F,s}$ , we need the following technical lemma.

**Lemma 5.4.4.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ , and let  $\tilde{\phi} \in \Phi(\Omega)$  be such that  $\tilde{\phi}(\Omega)$  is of class  $C^2$ . Let  $u_1, u_2 \in H^2(\Omega)$  be such that  $v_i = u_i \circ \tilde{\phi}^{(-1)} \in H^4(\tilde{\phi}(\Omega))$  for i = 1, 2 and

$$\frac{\partial^2 v_1}{\partial \nu^2} = \frac{\partial^2 v_2}{\partial \nu^2} = 0 \quad \text{on} \quad \partial \tilde{\phi}(\Omega).$$

Then we have

$$d|_{\phi=\tilde{\phi}}\mathcal{P}^{\mathcal{S}}_{\phi}[\psi][u_{1}][u_{2}] = \int_{\partial\tilde{\phi}(\Omega)} (D^{2}v_{1}:D^{2}v_{2}+\tau\nabla v_{1}\cdot\nabla v_{2})\mu\cdot\nu d\sigma + \int_{\partial\tilde{\phi}(\Omega)} \left(\operatorname{div}_{\partial\tilde{\phi}(\Omega)}(D^{2}v_{1}.\nu)\nabla v_{2}+\operatorname{div}_{\partial\tilde{\phi}(\Omega)}(D^{2}v_{2}.\nu)\nabla v_{1}\right)\cdot\mu d\sigma + \int_{\partial\tilde{\phi}(\Omega)} \left(\frac{\partial\Delta v_{1}}{\partial\nu}\nabla v_{2}+\frac{\partial\Delta v_{2}}{\partial\nu}\nabla v_{1}\right)\cdot\mu d\sigma -\tau\int_{\partial\tilde{\phi}(\Omega)} \left(\frac{\partial v_{1}}{\partial\nu}\nabla v_{2}+\frac{\partial v_{2}}{\partial\nu}\nabla v_{1}\right)\cdot\mu d\sigma - \int_{\tilde{\phi}(\Omega)} \left((\Delta^{2}v_{1}-\tau\Delta v_{1})\nabla v_{2}+(\Delta^{2}v_{2}-\tau\Delta v_{2})\nabla v_{1}\right)\cdot\mu d\sigma, \quad (5.4.5)$$

for all  $\psi \in (C^2(\overline{\Omega}))^N$ , where  $\mu = \psi \circ \tilde{\phi}^{-1}$ .

*Proof.* We have

$$\begin{split} d|_{\phi=\tilde{\phi}} \mathcal{P}_{\phi}^{\mathcal{S}}[\psi][u_{1}][u_{2}] \\ &= \int_{\Omega} (d|_{\phi=\tilde{\phi}} D^{2}(u_{1} \circ \phi^{-1}) \circ \phi)[\psi] : (D^{2}(u_{2} \circ \tilde{\phi}^{-1}) \circ \tilde{\phi})|\det D\tilde{\phi}|dx \\ &+ \tau \int_{\Omega} (d|_{\phi=\tilde{\phi}} \nabla(u_{1} \circ \phi^{-1}) \circ \phi)[\psi] \cdot (\nabla(u_{2} \circ \tilde{\phi}^{-1}) \circ \tilde{\phi})|\det D\tilde{\phi}|dx \\ &+ \int_{\Omega} (D^{2}(u_{1} \circ \tilde{\phi}^{-1}) \circ \tilde{\phi}) : (d|_{\phi=\tilde{\phi}} D^{2}(u_{2} \circ \phi^{-1}) \circ \phi)[\psi]|\det D\tilde{\phi}|dx \\ &+ \tau \int_{\Omega} (\nabla(u_{1} \circ \tilde{\phi}^{-1}) \circ \tilde{\phi}) \cdot (d|_{\phi=\tilde{\phi}} \nabla(u_{2} \circ \phi^{-1}) \circ \phi)[\psi]|\det D\tilde{\phi}|dx \\ &+ \int_{\Omega} (D^{2}(u_{1} \circ \tilde{\phi}^{-1}) \circ \tilde{\phi}) : (D^{2}(u_{2} \circ \tilde{\phi}^{-1}) \circ \tilde{\phi})d|_{\phi=\tilde{\phi}}|\det D\phi|[\psi]dx \\ &+ \tau \int_{\Omega} (\nabla(u_{1} \circ \tilde{\phi}^{-1}) \circ \tilde{\phi}) \cdot (\nabla(u_{2} \circ \tilde{\phi}^{-1}) \circ \tilde{\phi})d|_{\phi=\tilde{\phi}}|\det D\phi|[\psi]dx, \quad (5.4.6) \end{split}$$

and we note that the last two summands in (5.4.6) equals

$$\int_{\tilde{\phi}(\Omega)} \left( D^2 v_1 : D^2 v_2 + \tau \nabla v_1 \cdot \nabla v_2 \right) \operatorname{div} \mu dy.$$

(See also Proprosition 5.4.18). By standard calculus we have (see [23, formula (2.15)])

$$D^{2}(u \circ \phi^{-1}) \circ \phi = (\nabla \phi)^{-t} D^{2} u (\nabla \phi)^{-1} + \left( \sum_{k,l=1}^{N} \frac{\partial u}{\partial x_{k}} \frac{\partial \sigma_{k,i}}{\partial x_{l}} \sigma_{l,j} \right)_{i,j},$$

where  $\sigma = (\nabla \phi)^{-1}$ . This yields the following formula

$$d|_{\phi=\tilde{\phi}}(D^2(u\circ\phi^{-1})\circ\phi)[\psi]\circ\tilde{\phi}^{-1} = -D^2v\nabla\mu - \nabla\mu^t D^2v - \sum_{r=1}^N \frac{\partial v}{\partial y_r} D^2\mu_r, \quad (5.4.7)$$

where  $\mu = \psi \circ \tilde{\phi}^{-1}$  and  $v = u \circ \tilde{\phi}^{-1}$ . We rewrite formula (5.4.7) componentwise getting

$$\begin{split} \left(d|_{\phi=\tilde{\phi}}(D^2(u\circ\phi^{-1})\circ\phi)[\psi]\circ\tilde{\phi}^{-1}\right)_{i,j} \\ &=-\sum_{r=1}^N\left(\frac{\partial^2 v}{\partial y_i\partial y_r}\frac{\partial\mu_r}{\partial y_j}+\frac{\partial^2 v}{\partial y_j\partial y_r}\frac{\partial\mu_r}{\partial y_i}+\frac{\partial^2\mu_r}{\partial y_i\partial y_j}\frac{\partial v}{\partial y_r}\right). \end{split}$$

Moreover (see [75, Lemma 3.26])

$$\left(d|_{\phi=\tilde{\phi}}(\nabla(u\circ\phi^{-1})\circ\phi)[\psi]\circ\tilde{\phi}^{-1}\right)_i=-\sum_{r=1}^N\frac{\partial v}{\partial y_r}\frac{\partial \mu_r}{\partial y_i}.$$

Now we use Einstein notation, dropping all the summation symbols. The first summand of the right hand side of (5.4.6) equals

$$-\int_{\tilde{\phi}(\Omega)} \left( \frac{\partial^2 v_1}{\partial y_i \partial y_r} \frac{\partial \mu_r}{\partial y_j} + \frac{\partial^2 v_1}{\partial y_j \partial y_r} \frac{\partial \mu_r}{\partial y_i} + \frac{\partial^2 \mu_r}{\partial y_i \partial y_j} \frac{\partial v_1}{\partial y_r} \right) \frac{\partial^2 v_2}{\partial y_i \partial y_j} dy.$$
(5.4.8)

In order to compute (5.4.8), integrating by parts, we have

$$\begin{split} \int_{\tilde{\phi}(\Omega)} \frac{\partial^2 v_1}{\partial y_i \partial y_r} \frac{\partial \mu_r}{\partial y_j} \frac{\partial^2 v_2}{\partial y_i \partial y_j} dy &= \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_i} \frac{\partial \mu_r}{\partial y_j} \nu_r \frac{\partial^2 v_2}{\partial y_i \partial y_j} d\sigma \\ &- \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_i} \frac{\partial \operatorname{div} \mu}{\partial y_j} \frac{\partial^2 v_2}{\partial y_i \partial y_j} dy - \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_i} \frac{\partial \mu_r}{\partial y_j} \frac{\partial^3 v_2}{\partial y_i \partial y_j \partial y_r} dy \\ &= \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_i} \frac{\partial \mu_r}{\partial y_j} \nu_r \frac{\partial^2 v_2}{\partial y_i \partial y_j} d\sigma - \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_i} \frac{\partial \mu_r}{\partial y_j} \frac{\partial^3 v_2}{\partial y_i \partial y_j \partial y_r} dy \\ &- \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_i} \operatorname{div} \mu \frac{\partial^2 v_2}{\partial y_i \partial y_j} \nu_j d\sigma + \int_{\tilde{\phi}(\Omega)} D^2 v_1 : D^2 v_2 \operatorname{div} \mu dy \\ &+ \int_{\tilde{\phi}(\Omega)} \operatorname{div} \mu \nabla v_1 \cdot \nabla \Delta v_2 dy, \end{split}$$

and

$$\begin{split} \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_r} \frac{\partial^2 \mu_r}{\partial y_i \partial y_j} \frac{\partial^2 v_2}{\partial y_i \partial y_j} dy &= \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_r} \frac{\partial \mu_r}{\partial y_i} \frac{\partial \mu_r}{\partial y_i} \frac{\partial^2 v_2}{\partial y_i \partial y_j} dy \\ &- \int_{\tilde{\phi}(\Omega)} \frac{\partial^2 v_1}{\partial y_r \partial y_j} \frac{\partial \mu_r}{\partial y_i} \frac{\partial^2 v_2}{\partial y_i \partial y_j} dy - \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_r} \frac{\partial \mu_r}{\partial y_i} \frac{\partial \Delta v_2}{\partial y_i} dy \\ &= \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_r} \frac{\partial \mu_r}{\partial y_i} \nu_j \frac{\partial^2 v_2}{\partial y_i \partial y_j} d\sigma - \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_r} \frac{\partial \mu_r}{\partial y_i} \frac{\partial \Delta v_2}{\partial y_i} dy \\ &- \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \nu_r \frac{\partial^2 v_2}{\partial y_i \partial y_j} d\sigma + \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \frac{\partial^2 v_2}{\partial y_i \partial y_j} dy \\ &+ \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \nu_j \frac{\partial^2 v_2}{\partial y_i \partial y_j} d\sigma + \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \frac{\partial \Delta v_2}{\partial y_i \partial y_j} dy \\ &= \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_r} \frac{\partial \mu_r}{\partial y_i} \nu_j \frac{\partial^2 v_2}{\partial y_i \partial y_j} d\sigma - \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_r} \frac{\partial \mu_r}{\partial y_i} \frac{\partial \Delta v_2}{\partial y_i} dy \\ &- \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \nu_j \frac{\partial^2 v_2}{\partial y_i \partial y_j} d\sigma - \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_r} \frac{\partial \mu_r}{\partial y_i} \frac{\partial \Delta v_2}{\partial y_i} dy \\ &+ \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \nu_r \frac{\partial^2 v_2}{\partial y_i \partial y_j} d\sigma + \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \frac{\partial \Delta v_2}{\partial y_i \partial y_r} dy \\ &+ \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \frac{\partial^2 v_2}{\partial y_i \partial y_j} d\sigma + \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \frac{\partial \Delta v_2}{\partial y_i \partial y_r} dy \\ &+ \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_j} \frac{\partial \mu_r}{\partial y_i} \frac{\partial^2 v_2}{\partial y_i \partial y_j} \nu_i d\sigma - \int_{\tilde{\phi}(\Omega)} D^2 v_1 : D^2 v_2 \text{div} \mu dy \\ &- \int_{\tilde{\phi}(\Omega)} \nabla v_1 \cdot \nabla \Delta v_2 \text{div} \mu dy. \end{split}$$

We also have

$$\begin{split} &\int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_r} \frac{\partial \mu_r}{\partial y_i} \frac{\partial v_2}{\partial y_i} dy = \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_2}{\partial \nu} \nabla v_1 \cdot \mu d\sigma - \int_{\tilde{\phi}(\Omega)} \Delta v_2 \nabla v_1 \cdot \mu dy \\ &- \int_{\tilde{\phi}(\Omega)} \frac{\partial v_2}{\partial y_i} \frac{\partial^2 v_1}{\partial y_i \partial y_r} \mu_r dy = \int_{\partial \tilde{\phi}(\Omega)} \frac{\partial v_2}{\partial \nu} \nabla v_1 \cdot \mu d\sigma - \int_{\tilde{\phi}(\Omega)} \Delta v_2 \nabla v_1 \cdot \mu dy \\ &- \int_{\partial \tilde{\phi}(\Omega)} \nabla v_1 \cdot \nabla v_2 \mu \cdot \nu d\sigma + \int_{\tilde{\phi}(\Omega)} \frac{\partial v_1}{\partial y_i} \frac{\partial^2 v_2}{\partial y_i \partial y_r} \mu_r dy + \int_{\tilde{\phi}(\Omega)} \nabla v_1 \cdot \nabla v_2 \mathrm{div} \mu dy. \end{split}$$

It follows that

$$\begin{split} d|_{\phi=\tilde{\phi}} \mathcal{P}^{\mathcal{S}}_{\phi}[\psi][u_{1}][u_{2}] \\ &= -\int_{\tilde{\phi}(\Omega)} D^{2}v_{1}: D^{2}v_{2} \mathrm{div} \mu dy - \int_{\partial \tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial y_{i}} \frac{\partial^{2}v_{2}}{\partial y_{i} \partial y_{j}} + \frac{\partial v_{2}}{\partial y_{i}} \frac{\partial^{2}v_{1}}{\partial y_{i} \partial y_{j}} \right) \frac{\partial \mu_{r}}{\partial y_{j}} \nu_{r} d\sigma \\ &+ \int_{\tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial y_{i}} \frac{\partial^{3}v_{2}}{\partial y_{i} \partial y_{j}} + \frac{\partial v_{2}}{\partial y_{i}} \frac{\partial^{3}v_{1}}{\partial y_{i} \partial y_{j}} \right) \frac{\partial \mu_{r}}{\partial y_{j}} dy \\ &+ \int_{\partial \tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial y_{i}} \frac{\partial^{2}v_{2}}{\partial y_{i} \partial y_{j}} + \frac{\partial v_{2}}{\partial y_{i}} \frac{\partial^{2}v_{1}}{\partial y_{i} \partial y_{j}} \right) \nu_{j} \mathrm{div} \mu d\sigma \\ &- \int_{\tilde{\phi}(\Omega)} \left( \nabla v_{1} \cdot \nabla \Delta v_{2} + \nabla v_{2} \cdot \nabla \Delta v_{1} \right) \mathrm{div} \mu dy \\ &- \int_{\partial \tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial y_{r}} \frac{\partial^{2}v_{2}}{\partial y_{i} \partial y_{j}} + \frac{\partial v_{2}}{\partial y_{r}} \frac{\partial^{2}v_{1}}{\partial y_{i} \partial y_{j}} \right) \nu_{j} \frac{\partial \mu_{r}}{\partial y_{i}} d\sigma \\ &+ \int_{\tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial y_{r}} \frac{\partial \Delta v_{2}}{\partial y_{i} \partial y_{j}} + \frac{\partial v_{2}}{\partial y_{r}} \frac{\partial^{2}v_{1}}{\partial y_{i}} \right) \frac{\partial \mu_{r}}{\partial y_{i}} dy \\ &- \tau \int_{\partial \tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial y_{r}} \frac{\partial \Delta v_{2}}{\partial y_{i}} + \frac{\partial v_{2}}{\partial y_{r}} \frac{\partial \Delta v_{1}}{\partial y_{i}} \right) \cdot \mu d\sigma \\ &+ \tau \int_{\tilde{\phi}(\Omega)} \left( \Delta v_{1} \nabla v_{2} + \Delta v_{2} \nabla v_{1} \right) \cdot \mu dy + \tau \int_{\partial \tilde{\phi}(\Omega)} \nabla v_{1} \cdot \nabla v_{2} \mu \cdot \nu d\sigma. \quad (5.4.9) \end{split}$$

Now we recall that

$$\operatorname{div} \mu = \operatorname{div}_{\partial \tilde{\phi}(\Omega)} \mu + \frac{\partial \mu}{\partial \nu} \cdot \nu \quad \text{on } \partial \tilde{\phi}(\Omega),$$

(see also [38, §8.5]) and that, since  $\nu = \nabla d$ , where *d* is the distance from the boundary defined in an appropriate tubular neighborhood of the boundary, then  $\nabla \nu = (\nabla \nu)^t$  and  $\frac{\partial \nu}{\partial \nu} = 0$ , from which it follows that

$$\nabla_{\partial \tilde{\phi}(\Omega)} \nu = (\nabla_{\partial \tilde{\phi}(\Omega)} \nu)^t$$
 on  $\partial \tilde{\phi}(\Omega)$ .

We will use these identities throughout all the following computations. Using the fact that

$$\frac{\partial^2 v_1}{\partial \nu^2} = \frac{\partial^2 v_2}{\partial \nu^2} = 0 \text{ on } \partial \tilde{\phi}(\Omega),$$

we get that the sixth summand in (5.4.9) equals

$$-\int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_1}{\partial y_r} (D^2 v_2 . \nu)_{\partial\tilde{\phi}(\Omega)} + \frac{\partial v_2}{\partial y_r} (D^2 v_1 . \nu)_{\partial\tilde{\phi}(\Omega)} \right) \cdot \nabla_{\partial\tilde{\phi}(\Omega)} \mu_r d\sigma$$

$$-\int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_1}{\partial y_r} \frac{\partial^2 v_2}{\partial \nu^2} + \frac{\partial v_2}{\partial y_r} \frac{\partial^2 v_1}{\partial \nu^2} \right) \frac{\partial \mu_r}{\partial \nu} d\sigma$$

$$=\int_{\partial\tilde{\phi}(\Omega)} \left( \nabla_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_1}{\partial y_r} \right) (D^2 v_2 . \nu)_{\partial\tilde{\phi}(\Omega)} \right)$$

$$+ \nabla_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_2}{\partial y_r} \right) (D^2 v_1 . \nu)_{\partial\tilde{\phi}(\Omega)} \right) \mu_r d\sigma$$

$$+\int_{\partial\tilde{\phi}(\Omega)} \left( \operatorname{div}_{\partial\tilde{\phi}(\Omega)} (D^2 v_1 . \nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_2 + \operatorname{div}_{\partial\tilde{\phi}(\Omega)} (D^2 v_2 . \nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_1 \right) \cdot \mu d\sigma$$

$$=\int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial^2 v_1}{\partial y_i \partial y_r} \frac{\partial^2 v_2}{\partial y_i \partial y_j} + \frac{\partial^2 v_2}{\partial y_i \partial y_r} \frac{\partial^2 v_1}{\partial y_i \partial y_j} \right) \nu_j \mu_r d\sigma$$

$$+\int_{\partial\tilde{\phi}(\Omega)} \left( \operatorname{div}_{\partial\tilde{\phi}(\Omega)} (D^2 v_1 . \nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_2 + \operatorname{div}_{\partial\tilde{\phi}(\Omega)} (D^2 v_2 . \nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_1 \right) \cdot \mu d\sigma.$$
(5.4.10)

The seventh summand in (5.4.9) equals

$$\begin{split} \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial\Delta v_1}{\partial\nu} \nabla v_2 + \frac{\partial\Delta v_2}{\partial\nu} \nabla v_1 \right) \cdot \mu d\sigma &- \int_{\tilde{\phi}(\Omega)} \left( \Delta^2 v_1 \nabla v_2 + \Delta^2 v_2 \nabla v_1 \right) \cdot \mu d\sigma \\ &- \int_{\tilde{\phi}(\Omega)} \left( \frac{\partial^2 v_1}{\partial y_i \partial y_r} \frac{\partial\Delta v_2}{\partial y_i} + \frac{\partial^2 v_2}{\partial y_i \partial y_r} \frac{\partial\Delta v_1}{\partial y_i} \right) \mu_r dy \\ &= \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial\Delta v_1}{\partial\nu} \nabla v_2 + \frac{\partial\Delta v_2}{\partial\nu} \nabla v_1 \right) \cdot \mu d\sigma - \int_{\tilde{\phi}(\Omega)} \left( \Delta^2 v_1 \nabla v_2 + \Delta^2 v_2 \nabla v_1 \right) \cdot \mu d\sigma \\ &- \int_{\partial\tilde{\phi}(\Omega)} \left( \nabla v_1 \cdot \nabla\Delta v_2 + \nabla v_2 \cdot \nabla\Delta v_1 \right) \mu \cdot \nu d\sigma \\ &+ \int_{\tilde{\phi}(\Omega)} \left( \frac{\partial v_1}{\partial y_i} \frac{\partial^2 \Delta v_2}{\partial y_i \partial y_r} + \frac{\partial v_2}{\partial y_i} \frac{\partial^2 \Delta v_1}{\partial y_i \partial y_r} \right) \mu_r dy \\ &+ \int_{\tilde{\phi}(\Omega)} \left( \nabla v_1 \cdot \nabla\Delta v_2 + \nabla v_2 \cdot \nabla\Delta v_1 \right) \operatorname{div} \mu dy. \quad (5.4.11) \end{split}$$

The second summand in (5.4.9) equals

$$-\int_{\partial\tilde{\phi}(\Omega)} \nabla(\nabla v_1 \cdot \nabla v_2) \nabla(\mu_r) \nu_r d\sigma$$
$$= -\int_{\partial\tilde{\phi}(\Omega)} \nabla_{\partial\tilde{\phi}(\Omega)} (\nabla v_1 \cdot \nabla v_2) \nabla_{\partial\tilde{\phi}(\Omega)} (\mu_r) \nu_r d\sigma$$
$$-\int_{\partial\tilde{\phi}(\Omega)} \frac{\partial}{\partial\nu} (\nabla v_1 \cdot \nabla v_2) \frac{\partial\mu_r}{\partial\nu} \nu_r d\sigma. \quad (5.4.12)$$

The third summand in (5.4.9) equals

$$\begin{split} \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_1}{\partial y_i} \frac{\partial^3 v_2}{\partial y_j \partial y_r} + \frac{\partial v_2}{\partial y_i} \frac{\partial^3 v_1}{\partial y_i \partial y_j \partial y_r} \right) \nu_j \mu_r d\sigma \\ &- \int_{\tilde{\phi}(\Omega)} \left( \frac{\partial v_1}{\partial y_i} \frac{\partial^2 \Delta v_2}{\partial y_i \partial y_r} + \frac{\partial v_2}{\partial y_i} \frac{\partial^2 \Delta v_1}{\partial y_i \partial y_r} \right) \mu_r dy \\ &- \int_{\tilde{\phi}(\Omega)} \left( \frac{\partial^2 v_1}{\partial y_i \partial y_j} \frac{\partial^3 v_2}{\partial y_i \partial y_j \partial y_r} + \frac{\partial^2 v_2}{\partial y_i \partial y_j} \frac{\partial^3 v_1}{\partial y_i \partial y_j \partial y_r} \right) \mu_r dy \\ &= \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_1}{\partial y_i} \frac{\partial^3 v_2}{\partial y_i \partial y_j \partial y_r} + \frac{\partial v_2}{\partial y_i} \frac{\partial^3 v_1}{\partial y_i \partial y_j \partial y_r} \right) \nu_j \mu_r d\sigma \\ &- \int_{\tilde{\phi}(\Omega)} \left( \frac{\partial v_1}{\partial y_i} \frac{\partial^2 \Delta v_2}{\partial y_i \partial y_r} + \frac{\partial v_2}{\partial y_i} \frac{\partial^2 \Delta v_1}{\partial y_i \partial y_j \partial y_r} \right) \mu_r dy \\ &- \int_{\partial\tilde{\phi}(\Omega)} D^2 v_1 : D^2 v_2 \mu \cdot \nu d\sigma + \int_{\tilde{\phi}(\Omega)} D^2 v_1 : D^2 v_2 \mathrm{div} \mu dy. \end{split}$$
(5.4.13)

From (5.4.9)-(5.4.13), it follows that

$$\begin{split} d|_{\phi=\tilde{\phi}} \mathcal{P}^{\mathcal{S}}_{\phi}[\psi][u_{1}][u_{2}] &= -\int_{\partial\tilde{\phi}(\Omega)} \nabla_{\partial\tilde{\phi}(\Omega)} (\nabla v_{1} \cdot \nabla v_{2}) \nabla_{\partial\tilde{\phi}(\Omega)}(\mu_{r})\nu_{r} d\sigma \\ &- \int_{\partial\tilde{\phi}(\Omega)} \frac{\partial}{\partial\nu} (\nabla v_{1} \cdot \nabla v_{2}) \frac{\partial\mu_{r}}{\partial\nu} \nu_{r} d\sigma + \int_{\partial\tilde{\phi}(\Omega)} \frac{\partial}{\partial\nu} (\nabla v_{1} \cdot \nabla v_{2}) \mathrm{div} \mu d\sigma \\ &+ \int_{\partial\tilde{\phi}(\Omega)} \left( \mathrm{div}_{\partial\tilde{\phi}(\Omega)} (D^{2}v_{1}.\nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_{2} + \mathrm{div}_{\partial\tilde{\phi}(\Omega)} (D^{2}v_{2}.\nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_{1} \right) \cdot \mu d\sigma \\ &+ \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial^{2}v_{1}}{\partial y_{i}\partial y_{r}} \frac{\partial^{2}v_{2}}{\partial y_{i}\partial y_{j}} + \frac{\partial^{2}v_{2}}{\partial y_{i}\partial y_{r}} \frac{\partial^{2}v_{1}}{\partial y_{i}\partial y_{j}} \right) \nu_{j}\mu_{r} d\sigma \\ &+ \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial y_{i}} \frac{\partial^{3}v_{2}}{\partial y_{i}\partial y_{j}\partial y_{r}} + \frac{\partial v_{2}}{\partial y_{i}} \frac{\partial^{3}v_{1}}{\partial y_{i}\partial y_{j}\partial y_{r}} \right) \nu_{j}\mu_{r} d\sigma \\ &- \int_{\partial\tilde{\phi}(\Omega)} D^{2}v_{1} : D^{2}v_{2}\mu \cdot \nu d\sigma - \int_{\partial\tilde{\phi}(\Omega)} (\nabla v_{1} \cdot \nabla \Delta v_{2} + \nabla v_{2} \cdot \nabla \Delta v_{1}) \mu \cdot \nu d\sigma \\ &+ \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial\Delta v_{1}}{\partial\nu} \nabla v_{2} + \frac{\partial\Delta v_{2}}{\partial\nu} \nabla v_{1} \right) \cdot \mu d\sigma - \int_{\tilde{\phi}(\Omega)} (\Delta^{2}v_{1} \nabla v_{2} + \Delta^{2}v_{2} \nabla v_{1}) \cdot \mu d\sigma \\ &- \tau \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial\nu} \nabla v_{2} + \frac{\partial v_{2}}{\partial\nu} \nabla v_{1} \right) \cdot \mu d\sigma \\ &+ \tau \int_{\tilde{\phi}(\Omega)} (\Delta v_{1} \nabla v_{2} + \Delta v_{2} \nabla v_{1}) \cdot \mu dy + \tau \int_{\partial\tilde{\phi}(\Omega)} \nabla v_{1} \cdot \nabla v_{2} \mu \cdot \nu d\sigma \\ &= - \int_{\partial\tilde{\phi}(\Omega)} \nabla_{\partial\tilde{\phi}(\Omega)} (\nabla v_{1} \cdot \nabla v_{2}) \nabla_{\partial\tilde{\phi}(\Omega)}(\mu_{r}) \nu_{r} d\sigma \\ &+ \int_{\partial\tilde{\phi}(\Omega)} \frac{\partial}{\partial\nu} (\nabla v_{1} \cdot \nabla v_{2}) \mathrm{div}_{\partial\tilde{\phi}(\Omega)} \mu d\sigma \end{split}$$

$$+ \int_{\partial\tilde{\phi}(\Omega)} \left( \operatorname{div}_{\partial\tilde{\phi}(\Omega)} (D^{2}v_{1}.\nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_{2} + \operatorname{div}_{\partial\tilde{\phi}(\Omega)} (D^{2}v_{2}.\nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_{1} \right) \cdot \mu d\sigma \\ + \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial\Delta v_{1}}{\partial\nu} \nabla v_{2} + \frac{\partial\Delta v_{2}}{\partial\nu} \nabla v_{1} \right) \cdot \mu d\sigma - \tau \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial\nu} \nabla v_{2} + \frac{\partial v_{2}}{\partial\nu} \nabla v_{1} \right) \cdot \mu d\sigma \\ + \int_{\partial\tilde{\phi}(\Omega)} \frac{\partial}{\partial\nu} \left( \frac{\partial}{\partial y_{r}} (\nabla v_{1} \cdot \nabla v_{2}) \right) \mu_{r} d\sigma \\ - \int_{\partial\tilde{\phi}(\Omega)} D^{2}v_{1} : D^{2}v_{2}\mu \cdot \nu d\sigma - \int_{\partial\tilde{\phi}(\Omega)} (\nabla v_{1} \cdot \nabla \Delta v_{2} + \nabla v_{2} \cdot \nabla \Delta v_{1})\mu \cdot \nu d\sigma \\ - \int_{\tilde{\phi}(\Omega)} \left( \Delta^{2}v_{1} \nabla v_{2} + \Delta^{2}v_{2} \nabla v_{1} \right) \cdot \mu d\sigma \\ + \tau \int_{\tilde{\phi}(\Omega)} (\Delta v_{1} \nabla v_{2} + \Delta v_{2} \nabla v_{1}) \cdot \mu dy + \tau \int_{\partial\tilde{\phi}(\Omega)} \nabla v_{1} \cdot \nabla v_{2}\mu \cdot \nu d\sigma. \quad (5.4.14)$$

The first summand on the right hand side of (5.4.14) equals

$$\int_{\partial\tilde{\phi}(\Omega)} \Delta_{\partial\tilde{\phi}(\Omega)} (\nabla v_1 \cdot \nabla v_2) \mu \cdot \nu d\sigma + \int_{\partial\tilde{\phi}(\Omega)} \nabla_{\partial\tilde{\phi}(\Omega)} (\nabla v_1 \cdot \nabla v_2) \cdot (\nabla_{\partial\tilde{\phi}(\Omega)} \nu_r) \mu_r d\sigma,$$

while the sixth one equals

$$\int_{\partial\tilde{\phi}(\Omega)} \frac{\partial^2}{\partial\nu^2} (\nabla v_1 \cdot \nabla v_2) \mu \cdot \nu d\sigma + \int_{\partial\tilde{\phi}(\Omega)} \nabla_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial}{\partial\nu} (\nabla v_1 \cdot \nabla v_2) \right) \cdot \mu d\sigma \\ - \int_{\partial\tilde{\phi}(\Omega)} \nabla_{\partial\tilde{\phi}(\Omega)} (\nabla v_1 \cdot \nabla v_2) \cdot (\nabla_{\partial\tilde{\phi}(\Omega)} \nu_r) \mu_r d\sigma.$$

Using the fact that

$$\int_{\partial \tilde{\phi}(\Omega)} \operatorname{div}_{\partial \tilde{\phi}(\Omega)} \left( \frac{\partial}{\partial \nu} (\nabla v_1 \cdot \nabla v_2) \cdot \mu \right) d\sigma = \int_{\partial \tilde{\phi}(\Omega)} K \frac{\partial}{\partial \nu} (\nabla v_1 \cdot \nabla v_2) \mu \cdot \nu d\sigma,$$

where K denotes the mean curvature of  $\partial \tilde{\phi}(\Omega)$  (see [38, §8.5]), we obtain

$$\begin{split} d|_{\phi=\tilde{\phi}}\mathcal{P}^{\mathcal{S}}_{\phi}[\psi][u_{1}][u_{2}] &= \int_{\partial\tilde{\phi}(\Omega)} \Delta_{\partial\tilde{\phi}(\Omega)} (\nabla v_{1} \cdot \nabla v_{2}) \mu \cdot \nu d\sigma \\ &+ \int_{\partial\tilde{\phi}(\Omega)} K \frac{\partial}{\partial\nu} (\nabla v_{1} \cdot \nabla v_{2}) \mu \cdot \nu d\sigma + \int_{\partial\tilde{\phi}(\Omega)} \frac{\partial^{2}}{\partial\nu^{2}} (\nabla v_{1} \cdot \nabla v_{2}) \mu \cdot \nu d\sigma \\ &- \int_{\partial\tilde{\phi}(\Omega)} D^{2} v_{1} : D^{2} v_{2} \mu \cdot \nu d\sigma - \int_{\partial\tilde{\phi}(\Omega)} (\nabla v_{1} \cdot \nabla \Delta v_{2} + \nabla v_{2} \cdot \nabla \Delta v_{1}) \mu \cdot \nu d\sigma \\ &+ \int_{\partial\tilde{\phi}(\Omega)} \left( \operatorname{div}_{\partial\tilde{\phi}(\Omega)} (D^{2} v_{1} \cdot \nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_{2} + \operatorname{div}_{\partial\tilde{\phi}(\Omega)} (D^{2} v_{2} \cdot \nu)_{\partial\tilde{\phi}(\Omega)} \nabla v_{1} \right) \cdot \mu d\sigma \\ &+ \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial\Delta v_{1}}{\partial\nu} \nabla v_{2} + \frac{\partial\Delta v_{2}}{\partial\nu} \nabla v_{1} \right) \cdot \mu d\sigma - \tau \int_{\partial\tilde{\phi}(\Omega)} \left( \frac{\partial v_{1}}{\partial\nu} \nabla v_{2} + \frac{\partial v_{2}}{\partial\nu} \nabla v_{1} \right) \cdot \mu d\sigma \end{split}$$

$$\begin{split} &-\int_{\tilde{\phi}(\Omega)} \left(\Delta^2 v_1 \nabla v_2 + \Delta^2 v_2 \nabla v_1\right) \cdot \mu d\sigma \\ &+ \tau \int_{\tilde{\phi}(\Omega)} \left(\Delta v_1 \nabla v_2 + \Delta v_2 \nabla v_1\right) \cdot \mu dy + \tau \int_{\partial \tilde{\phi}(\Omega)} \nabla v_1 \cdot \nabla v_2 \mu \cdot \nu d\sigma \\ &= \int_{\partial \tilde{\phi}(\Omega)} \Delta (\nabla v_1 \cdot \nabla v_2) \mu \cdot \nu d\sigma - \int_{\partial \tilde{\phi}(\Omega)} D^2 v_1 : D^2 v_2 \mu \cdot \nu d\sigma \\ &- \int_{\partial \tilde{\phi}(\Omega)} (\nabla v_1 \cdot \nabla \Delta v_2 + \nabla v_2 \cdot \nabla \Delta v_1) \mu \cdot \nu d\sigma \\ &+ \int_{\partial \tilde{\phi}(\Omega)} \left( \operatorname{div}_{\partial \tilde{\phi}(\Omega)} (D^2 v_1 \cdot \nu)_{\partial \tilde{\phi}(\Omega)} \nabla v_2 + \operatorname{div}_{\partial \tilde{\phi}(\Omega)} (D^2 v_2 \cdot \nu)_{\partial \tilde{\phi}(\Omega)} \nabla v_1 \right) \cdot \mu d\sigma \\ &+ \int_{\partial \tilde{\phi}(\Omega)} \left( \frac{\partial \Delta v_1}{\partial \nu} \nabla v_2 + \frac{\partial \Delta v_2}{\partial \nu} \nabla v_1 \right) \cdot \mu d\sigma - \tau \int_{\partial \tilde{\phi}(\Omega)} \left( \frac{\partial v_1}{\partial \nu} \nabla v_2 + \frac{\partial v_2}{\partial \nu} \nabla v_1 \right) \cdot \mu d\sigma \\ &- \int_{\tilde{\phi}(\Omega)} \left( (\Delta^2 v_1 - \tau \Delta v_1) \nabla v_2 + (\Delta^2 v_2 - \tau \Delta v_2) \nabla v_1 \right) \cdot \mu d\sigma \\ &+ \tau \int_{\partial \tilde{\phi}(\Omega)} \nabla v_1 \cdot \nabla v_2 \mu \cdot \nu d\sigma. \end{split}$$

Using the equality

$$\Delta(\nabla v_1 \cdot \nabla v_2) = \nabla \Delta v_1 \cdot \nabla v_2 + \nabla v_1 \cdot \nabla \Delta v_2 + 2D^2 v_1 : D^2 v_2$$

we finally get formula (5.4.5).

Now we can compute Hadamard-type formulas for the eigenvalues of problem (5.4.1).

**Theorem 5.4.15.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ . Let F be a finite non-empty subset of  $\mathbb{N}$ . Let  $\tilde{\phi} \in \Theta_{\Omega}[F]$  be such that  $\partial \tilde{\phi}(\Omega) \in C^4$ . Let  $v_1, \ldots, v_{|F|}$  be an orthonormal basis of the eigenspace associated with the eigenvalue  $\lambda_F[\tilde{\phi}]$  of problem (5.4.1) in  $L^2(\partial \tilde{\phi}(\Omega))$ . Then

$$d|_{\phi=\tilde{\phi}}(\Lambda_{F,s})[\psi] = -\lambda_F^{s-1}[\tilde{\phi}] \binom{|F|-1}{s-1} \sum_{l=1}^{|F|} \int_{\partial \tilde{\phi}(\Omega)} \left(\lambda_F K v_l^2 + \lambda_F \frac{\partial (v_l^2)}{\partial \nu} - \tau |\nabla v_l|^2 - |D^2 v_l|^2\right) \mu \cdot \nu d\sigma,$$

for all  $\psi \in (C^2(\overline{\Omega}))^N$ , where  $\mu = \psi \circ \tilde{\phi}^{(-1)}$ , and K denotes the mean curvature of  $\partial \tilde{\phi}(\Omega)$ .

*Proof.* First of all we note that  $v_1, ..., v_{|F|} \in H^4(\tilde{\phi}(\Omega))$  (see e.g., [47, §2.5]). We set  $u_l = v_l \circ \tilde{\phi}$  for l = 1, ..., |F|. For |F| > 1 (case |F| = 1 is similar),

150

 $s \leq |F|$ , we have

$$d|_{\phi=\tilde{\phi}}(\Lambda_{F,s})[\psi] = -\lambda_{F}^{s+2}[\tilde{\phi}] \binom{|F|-1}{s-1} \sum_{l=1}^{|F|} \mathcal{P}_{\tilde{\phi}}^{\mathcal{S}} \left[ d|_{\phi=\tilde{\phi}} W_{\phi}^{\mathcal{S}}[\psi][p(u_{l})] \right] [p(u_{l})].$$
(5.4.16)

We refer to [75, Theorem 3.38] for a proof of formula (5.4.16).

By standard calculus in normed spaces we have:

$$\begin{aligned} \mathcal{P}_{\tilde{\phi}}^{\mathcal{S}} \left[ \mathrm{d}|_{\phi = \tilde{\phi}} \left( (\pi_{\phi}^{\sharp, \mathcal{S}})^{-1} \circ \left( \mathcal{P}_{\phi}^{\mathcal{S}} \right)^{-1} \circ \mathcal{J}_{\phi}^{\mathcal{S}} \circ \mathrm{Tr} \circ \pi_{\phi}^{\sharp, \mathcal{S}} \right) [\psi][p(u_{l})] \right] [p(u_{l})] \\ &= \mathcal{P}_{\tilde{\phi}}^{\mathcal{S}} \left[ (\pi_{\tilde{\phi}}^{\sharp, \mathcal{S}})^{-1} \circ \left( \mathcal{P}_{\tilde{\phi}}^{\mathcal{S}} \right)^{-1} \circ \mathrm{d}|_{\phi = \tilde{\phi}} \left( \mathcal{J}_{\phi}^{\mathcal{S}} \circ \mathrm{Tr} \circ \pi_{\phi}^{\sharp, \mathcal{S}} \right) [\psi][p(u_{l})] \right] [p(u_{l})] \\ &+ \mathcal{P}_{\tilde{\phi}}^{\mathcal{S}} \left[ \mathrm{d}|_{\phi = \tilde{\phi}} \left( (\pi_{\phi}^{\sharp, \mathcal{S}})^{-1} \circ \left( \mathcal{P}_{\phi}^{\mathcal{S}} \right)^{-1} \right) [\psi] \circ \mathcal{J}_{\tilde{\phi}}^{\mathcal{S}} \circ \mathrm{Tr} \circ \pi_{\tilde{\phi}}^{\sharp, \mathcal{S}} [p(u_{l})] \right] [p(u_{l})] . \end{aligned}$$

Now note that:

$$\mathcal{P}_{\tilde{\phi}}^{\mathcal{S}}\left[(\pi_{\tilde{\phi}}^{\sharp,\mathcal{S}})^{-1} \circ \left(\mathcal{P}_{\tilde{\phi}}^{\mathcal{S}}\right)^{-1} \circ d|_{\phi=\tilde{\phi}} \left(\mathcal{J}_{\phi}^{\mathcal{S}} \circ \operatorname{Tr} \circ \pi_{\phi}^{\sharp,\mathcal{S}}\right) [\psi][p(u_l)]\right] [p(u_l)]$$
$$= \int_{\partial \tilde{\phi}(\Omega)} \left(Kv_l^2 + \frac{\partial(v_l^2)}{\partial \nu}\right) \mu \cdot \nu d\sigma - \int_{\partial \tilde{\phi}(\Omega)} \nabla(v_l^2) \cdot \mu d\sigma,$$

(see also [74, Lemma 3.3]) and

$$\mathcal{P}_{\tilde{\phi}}^{\mathcal{S}} \left[ \mathrm{d}_{\phi = \tilde{\phi}} \left( (\pi_{\phi}^{\sharp, \mathcal{S}})^{-1} \circ \left( \mathcal{P}_{\phi}^{\mathcal{S}} \right)^{-1} \right) [\psi] \circ \mathcal{J}_{\tilde{\phi}}^{\mathcal{S}} \circ \mathrm{Tr} \circ \pi_{\tilde{\phi}}^{\sharp, \mathcal{S}} [p(u_l)] \right] [p(u_l)]$$
$$= -\lambda_F^{-1} d_{\phi = \tilde{\phi}} \left( \mathcal{P}_{\phi}^{\mathcal{S}} \circ \pi_{\phi}^{\mathcal{S}} \right) [\psi] [u_l] [\pi_{\tilde{\phi}}^{\mathcal{S}} (u_l)].$$

(We refer to [77, Lemma 2.4] for more explicit computations). Using formula (5.4.5) we obtain

$$\mathcal{P}^{\mathcal{S}}_{\tilde{\phi}} \left[ \mathrm{d}|_{\phi = \tilde{\phi}} \left( (\pi^{\sharp, \mathcal{S}}_{\phi})^{-1} \circ \left( \mathcal{P}^{\mathcal{S}}_{\phi} \right)^{-1} \right) [\psi] \circ \mathcal{J}^{\mathcal{S}}_{\tilde{\phi}} \circ \mathrm{Tr} \circ \pi^{\sharp, \mathcal{S}}_{\tilde{\phi}} [p(u_l)] \right] [p(u_l)] \\ = -\lambda_F^{-1} \int_{\partial \tilde{\phi}(\Omega)} \left( |D^2 v_l|^2 + \tau |\nabla v_l|^2 \right) \mu \cdot \nu d\sigma + \int_{\partial \tilde{\phi}(\Omega)} \nabla (v_l^2) \cdot \mu d\sigma.$$

This concludes the proof.

#### 5.4.2 Isovolumetric perturbatons

Now we turn our attention to extremum problems of the type

$$\min_{\mathcal{V}(\phi)=\text{const.}} \Lambda_{F,s}[\phi] \text{ or } \max_{\mathcal{V}(\phi)=\text{const.}} \Lambda_{F,s}[\phi],$$

where  $\mathcal{V}(\phi)$  denotes the Lebesgue measure of  $\phi(\Omega)$ , i.e.,

$$\mathcal{V}(\phi) := \int_{\phi(\Omega)} dx = \int_{\Omega} |\det D\phi| dx.$$
 (5.4.17)

Note that all  $\phi$ 's realizing one of the extrema are critical points under measure constraint, i.e., Ker  $d\mathcal{V}(\phi) \subseteq \text{Ker } d\Lambda_{F,s}[\phi]$ . We have the following result (see [76, Proposition 2.10]).

**Proposition 5.4.18.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ . Then the following statements hold.

i) The map  $\mathcal{V}$  from  $\Phi(\Omega)$  to  $\mathbb{R}$  defined in (5.4.17) is real analytic. Moreover, the differential of  $\mathcal{V}$  at  $\tilde{\phi} \in \Phi(\Omega)$  is given by the formula

$$d|_{\phi=\tilde{\phi}}\mathcal{V}(\phi)[\psi] = \int_{\tilde{\phi}(\Omega)} \operatorname{div}(\psi \circ \tilde{\phi}^{-1}) dy = \int_{\partial \tilde{\phi}(\Omega)} (\psi \circ \tilde{\phi}^{-1}) \cdot \nu d\sigma.$$

ii) For  $\mathcal{V}_0 \in ]0, +\infty[$ , let

$$V(\mathcal{V}_0) := \{ \phi \in \Phi(\Omega) : \mathcal{V}(\phi) = \mathcal{V}_0 \}.$$

If  $V(\mathcal{V}_0) \neq \emptyset$ , then  $V(\mathcal{V}_0)$  is a real analytic manifold of  $(C^2(\overline{\Omega}))^N$  of codimension 1.

Using Lagrange Multipliers Theorem, it is easy to prove the following theorem.

**Theorem 5.4.19.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ . Let F be a non-empty finite subset of  $\mathbb{N}$ . Let  $\mathcal{V}_0 \in ]0, +\infty[$ . Let  $\tilde{\phi} \in V(\mathcal{V}_0)$  be such that  $\partial \tilde{\phi}(\Omega) \in C^4$  and  $\lambda_j[\tilde{\phi}]$  have a common value  $\lambda_F[\tilde{\phi}]$  for all  $j \in F$  and  $\lambda_l[\tilde{\phi}] \neq \lambda_F[\tilde{\phi}]$  for all  $l \in \mathbb{N} \setminus F$ . For  $s = 1, \ldots, |F|$ , the function  $\tilde{\phi}$  is a critical point for  $\Lambda_{F,s}$  on  $V(\mathcal{V}_0)$  if and only if there exists an orthonormal basis  $v_1, \ldots, v_{|F|}$  of the eigenspace corresponding to the eigenvalue  $\lambda_F[\tilde{\phi}]$  of problem (5.4.1) in  $L^2(\partial \tilde{\phi}(\Omega))$ , and a constant  $c \in \mathbb{R}$  such that

$$\sum_{l=1}^{|F|} \left( \lambda_F[\tilde{\phi}] \left( K v_l^2 + \frac{\partial \left( v_l^2 \right)}{\partial \nu} \right) - \tau |\nabla v_l|^2 - |D^2 v_l|^2 \right) = c, \text{ a.e. on } \partial \tilde{\phi}(\Omega).$$
(5.4.20)

Now that we have a characterization for the criticality of  $\tilde{\phi}$ , we may wonder whether balls are critical domains. It turns out that indeed balls are criticals for the symmetric functions of the eigenvalues, as proved in the following theorem.

**Theorem 5.4.21.** Let  $\Omega$  be a bounded domain of  $\mathbb{R}^N$  of class  $C^1$ . Let  $\tilde{\phi} \in \Phi(\Omega)$  be such that  $\tilde{\phi}(\Omega)$  is a ball. Let  $\tilde{\lambda}$  be an eigenvalue of problem (5.4.1) in  $\tilde{\phi}(\Omega)$ , and let F be the set of  $j \in \mathbb{N}$  such that  $\lambda_j[\tilde{\phi}] = \tilde{\lambda}$ . Then  $\Lambda_{F,s}$  has a critical point at  $\tilde{\phi}$  on  $V(\mathcal{V}(\tilde{\phi}))$ , for all  $s = 1, \ldots, |F|$ .

*Proof.* Using Lemma 5.4.22 below and the fact that the mean curvature is constant for a ball, condition (5.4.20) is immediately seen to be satisfied.  $\Box$
**Lemma 5.4.22.** Let B be the unit ball in  $\mathbb{R}^N$  centered at zero, and let  $\lambda$  be an eigenvalue of problem (5.4.1) in B. Let F be the subset of  $\mathbb{N}$  of all indexes j such that the j-th eigenvalue of problem (5.4.1) in B coincides with  $\lambda$ . Let  $v_1, \ldots, v_{|F|}$  be an orthonormal basis of the eigenspace associated with the eigenvalue  $\lambda$ , where the orthonormality is taken with respect to the scalar product in  $L^2(\partial B)$ . Then

$$\sum_{j=1}^{|F|} v_j^2, \ \sum_{j=1}^{|F|} |\nabla v_j|^2, \ \sum_{j=1}^{|F|} |D^2 v_j|^2$$

are radial functions.

*Proof.* Let  $O_N(\mathbb{R})$  denote the group of orthogonal linear transformations in  $\mathbb{R}^N$ . Since the Laplace operator is invariant under rotations, then  $v_k \circ A$ , where  $A \in O_N(\mathbb{R})$ , is an eigenfunction with eigenvalue  $\lambda$ ; moreover,  $\{v_j \circ A : j = 1, \ldots, |F|\}$  is a orthonormal basis for the eigenspace associated with  $\lambda$ . Since both  $\{v_j : j = 1, \ldots, |F|\}$  and  $\{v_j \circ A : j = 1, \ldots, |F|\}$  are orthonormal bases, then there exists  $R[A] \in O_N(\mathbb{R})$  with matrix  $(R_{ij}[A])_{i,j=1,\ldots,|F|}$  such that

$$v_j = \sum_{l=1}^{|F|} R_{jl}[A] v_l \circ A.$$

This implies that

$$\sum_{j=1}^{|F|} v_j^2 = \sum_{j=1}^{|F|} (v_j \circ A)^2,$$

from which we get that  $\sum_{j=1}^{|F|} v_j^2$  is radial. Moreover, using standard calculus, we get

$$\sum_{j=1}^{|F|} |\nabla v_j|^2 = \sum_{l_1, l_2=1}^{|F|} R_{jl_1}[A] R_{jl_2}[A] (\nabla v_{l_1} \circ A) \cdot (\nabla v_{l_2} \circ A) = \sum_{l=1}^{|F|} |\nabla v_l \circ A|^2,$$

and

$$D^{2}v_{j} \cdot D^{2}v_{j} = \sum_{l_{1},l_{2}=1}^{|F|} R_{jl_{1}}[A]R_{jl_{2}}[A]A^{t} \cdot (D^{2}v_{l_{1}} \circ A) \cdot A \cdot A^{t} \cdot (D^{2}v_{l_{2}} \circ A) \cdot A$$
$$= \sum_{l_{1},l_{2}=1}^{|F|} R_{jl_{1}}[A]R_{jl_{2}}[A]A^{t} \cdot (D^{2}v_{l_{1}} \circ A) \cdot (D^{2}v_{l_{2}} \circ A) \cdot A,$$

hence

$$|D^{2}v_{j}|^{2} = \operatorname{tr}(D^{2}v_{j} \cdot D^{2}v_{j}) = \sum_{l_{1}, l_{2}=1}^{|F|} R_{jl_{1}}[A]R_{jl_{2}}[A](D^{2}v_{l_{1}} \circ A) : (D^{2}v_{l_{2}} \circ A).$$

from which we get

$$\sum_{j=1}^{|F|} |D^2 v_j|^2 = \sum_{j=1}^{|F|} |D^2 v_j \circ A|^2.$$

### 5.4.3 The Neumann problem

As we have done for the Steklov problem, we study the Neumann problem for the biharmonic operator on  $\phi(\Omega)$ , i.e.,

$$\begin{cases} \Delta^2 u - \tau \Delta u = \lambda u, & \text{in } \phi(\Omega), \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial \phi(\Omega), \\ \tau \frac{\partial u}{\partial \nu} - \operatorname{div}_{\partial \phi(\Omega)} (D^2 u \cdot \nu) - \frac{\partial \Delta u}{\partial \nu} = 0, & \text{on } \partial \phi(\Omega). \end{cases}$$
(5.4.23)

We consider the operator  $\mathcal{P}^{\mathcal{N}}_{\phi}$  from  $H^{2,\mathcal{N}}_{\phi}(\Omega)$  to  $F(\Omega)$ , defined by

$$\mathcal{P}_{\phi}^{\mathcal{N}}[u][\varphi] := \int_{\Omega} (D^2(u \circ \phi^{-1}) \circ \phi) : (D^2(\varphi \circ \phi^{-1}) \circ \phi) |\det D\phi| dx + \tau \int_{\Omega} (\nabla(u \circ \phi^{-1}) \circ \phi) \cdot (\nabla(\varphi \circ \phi^{-1}) \circ \phi) |\det D\phi| dx, \quad (5.4.24)$$

for all  $u \in H^{2,\mathcal{N}}_{\phi}(\Omega), \, \varphi \in H^{2}(\Omega)$ , where

$$H^{2,\mathcal{N}}_{\phi}(\Omega) := \left\{ u \in H^2(\Omega) : \int_{\Omega} u |\det D\phi| dx = 0 \right\},$$

Moreover, for every  $\phi \in \Phi(\Omega)$ , we consider the map  $\mathcal{J}_{\phi}^{\mathcal{N}}$  from  $L^2(\Omega)$  to  $H^2(\Omega)'$  defined by

$$\mathcal{J}_{\phi}^{\mathcal{N}}[u][\varphi] := \int_{\Omega} u\varphi |\det D\phi| d\sigma, \quad \forall u \in L^{2}(\Omega), \varphi \in H^{2}(\Omega).$$

We will think of the space  $H^{2,\mathcal{N}}_{\phi}(\Omega)$  as endowed with the scalar product induced by (5.4.24). We denote by  $\pi^{\mathcal{N}}_{\phi}$  the map from  $H^2(\Omega)$  to  $H^{2,\mathcal{N}}_{\phi}(\Omega)$  defined by

$$\pi_{\phi}^{\mathcal{N}}(u) := u - \frac{\int_{\Omega} u |\det D\phi| dx}{\int_{\Omega} |\det D\phi| dx},$$

and by  $\pi_{\phi}^{\sharp,\mathcal{N}}$  the map from  $H^2(\Omega)/\mathbb{R}$  onto  $H^{2,\mathcal{N}}_{\phi}(\Omega)$  defined by the equality  $\pi_{\phi}^{\mathcal{N}} = \pi_{\phi}^{\sharp,\mathcal{N}} \circ p$ . Clearly,  $\pi_{\phi}^{\sharp,\mathcal{N}}$  is a homeomorphism, and we can recast problem (5.4.23) as

$$\lambda^{-1}u = W_{\phi}^{\mathcal{N}}u,$$

where  $W_{\phi}^{\mathcal{N}} := (\pi_{\phi}^{\sharp,\mathcal{N}})^{-1} \circ (\mathcal{P}_{\phi}^{\mathcal{N}})^{-1} \circ \mathcal{J}_{\phi}^{\mathcal{N}} \circ i \circ \pi_{\phi}^{\sharp,\mathcal{N}}$  and *i* is the canonical embedding of  $H^2(\Omega)$  into  $L^2(\Omega)$ . An analogue of Theorem 5.4.3 can be stated also in this case. Therefore, we can compute Hadamard-type formulas for the Neumann eigenvalues. This is contained in the following theorem.

**Theorem 5.4.25.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ . Let F be a finite non-empty subset of  $\mathbb{N}$ . Let  $\tilde{\phi} \in \Theta_{\Omega}[F]$  be such that  $\partial \tilde{\phi}(\Omega) \in C^4$ . Let  $v_1, \ldots, v_{|F|}$  be an orthonormal basis of the eigenspace associated with the eigenvalue  $\lambda_F[\tilde{\phi}]$  of problem (5.4.23) in  $L^2(\tilde{\phi}(\Omega))$ . Then

$$d|_{\phi=\tilde{\phi}}(\Lambda_{F,s})[\psi]$$
  
=  $-\lambda_F^{s-1}[\tilde{\phi}] \binom{|F|-1}{s-1} \sum_{l=1}^{|F|} \int_{\partial \tilde{\phi}(\Omega)} \left(\lambda_F v_l^2 - \tau |\nabla v_l|^2 - |D^2 v_l|^2\right) \mu \cdot \nu d\sigma,$ 

for all  $\psi \in (C^2(\overline{\Omega}))^N$ , where  $\mu = \psi \circ \tilde{\phi}^{-1}$ .

*Proof.* The proof is similar to that of Theorem 5.4.15.

First of all we note that, by elliptic regularity theory,  $v_1, \ldots, v_{|F|} \in H^4(\tilde{\phi}(\Omega))$  (see [47, §2.5]). We set  $u_l = v_l \circ \tilde{\phi}$  for  $l = 1, \ldots, |F|$ . For |F| > 1 (case |F| = 1 is similar),  $s \leq |F|$ , we have

$$d|_{\phi=\tilde{\phi}}(\Lambda_{F,s})[\psi] = -\lambda_{F}^{s+2}[\tilde{\phi}] \binom{|F|-1}{s-1} \sum_{l=1}^{|F|} \mathcal{P}_{\tilde{\phi}}^{\mathcal{N}} \left[ d|_{\phi=\tilde{\phi}} W_{\phi}^{\mathcal{N}}[\psi][p(u_{l})] \right] [p(u_{l})].$$

By standard calculus in normed spaces we have:

$$\begin{aligned} \mathcal{P}_{\tilde{\phi}}^{\mathcal{N}} \left[ \mathrm{d}|_{\phi = \tilde{\phi}} \left( (\pi_{\phi}^{\sharp, \mathcal{N}})^{-1} \circ \left( \mathcal{P}_{\phi}^{\mathcal{N}} \right)^{-1} \circ \mathcal{J}_{\phi}^{\mathcal{N}} \circ i \circ \pi_{\phi}^{\sharp, \mathcal{N}} \right) [\psi][p(u_l)] \right] [p(u_l)] \\ &= \mathcal{P}_{\tilde{\phi}}^{\mathcal{N}} \left[ (\pi_{\tilde{\phi}}^{\sharp, \mathcal{N}})^{-1} \circ \left( \mathcal{P}_{\tilde{\phi}}^{\mathcal{N}} \right)^{-1} \circ \mathrm{d}|_{\phi = \tilde{\phi}} \left( \mathcal{J}_{\phi}^{\mathcal{N}} \circ i \circ \pi_{\phi}^{\sharp, \mathcal{N}} \right) [\psi][p(u_l)] \right] [p(u_l)] \\ &+ \mathcal{P}_{\tilde{\phi}}^{\mathcal{N}} \left[ \mathrm{d}|_{\phi = \tilde{\phi}} \left( (\pi_{\phi}^{\sharp, \mathcal{N}})^{-1} \circ \left( \mathcal{P}_{\phi}^{\mathcal{N}} \right)^{-1} \right) [\psi] \circ \mathcal{J}_{\tilde{\phi}}^{\mathcal{N}} \circ i \circ \pi_{\tilde{\phi}}^{\sharp, \mathcal{N}} [p(u_l)] \right] [p(u_l)] . \end{aligned}$$

Now note that

$$\mathcal{P}_{\tilde{\phi}}^{\mathcal{N}} \left[ (\pi_{\tilde{\phi}}^{\sharp,\mathcal{N}})^{-1} \circ \left( \mathcal{P}_{\tilde{\phi}}^{\mathcal{N}} \right)^{-1} \circ \mathrm{d}|_{\phi = \tilde{\phi}} \left( \mathcal{J}_{\phi}^{\mathcal{N}} \circ i \circ \pi_{\phi}^{\sharp,\mathcal{N}} \right) [\psi][p(u_l)] \right] [p(u_l)]$$
$$= \int_{\tilde{\phi}(\Omega)} v_l^2 \mathrm{div} \mu dy,$$

(see also Proposition 5.4.18) and

$$\mathcal{P}_{\tilde{\phi}}^{\mathcal{N}}\left[\mathrm{d}|_{\phi=\tilde{\phi}}\left((\pi_{\phi}^{\sharp,\mathcal{N}})^{-1}\circ\left(\mathcal{P}_{\phi}^{\mathcal{N}}\right)^{-1}\right)[\psi]\circ\mathcal{J}_{\tilde{\phi}}^{\mathcal{N}}\circ i\circ\pi_{\tilde{\phi}}^{\sharp,\mathcal{N}}[p(u_{l})]\right][p(u_{l})] \\ = -\lambda_{F}^{-1}d|_{\phi=\tilde{\phi}}\left(\mathcal{P}_{\phi}^{\mathcal{N}}\circ\pi_{\phi}^{\mathcal{N}}\right)[\psi][u_{l}][\pi_{\tilde{\phi}}^{\mathcal{N}}(u_{l})].$$

Using formula (5.4.5) we obtain

$$\mathcal{P}_{\tilde{\phi}}^{\mathcal{N}}\left[\mathrm{d}|_{\phi=\tilde{\phi}}\left((\pi_{\phi}^{\sharp,\mathcal{N}})^{-1}\circ\left(\mathcal{P}_{\phi}^{\mathcal{N}}\right)^{-1}\right)[\psi]\circ\mathcal{J}_{\tilde{\phi}}^{\mathcal{N}}\circ i\circ\pi_{\tilde{\phi}}^{\sharp,\mathcal{N}}[p(u_{l})]\right][p(u_{l})]$$
$$=-\lambda_{F}^{-1}\int_{\partial\tilde{\phi}(\Omega)}\left(|D^{2}v_{l}|^{2}+\tau|\nabla v_{l}|^{2}\right)\mu\cdot\nu d\sigma+\int_{\tilde{\phi}(\Omega)}\nabla(v_{l}^{2})\cdot\mu dy.$$

To conclude, just observe that

$$\int_{\tilde{\phi}(\Omega)} \nabla(v_l^2) \cdot \mu dy = \int_{\partial \tilde{\phi}(\Omega)} v_l^2 \mu \cdot \nu d\sigma - \int_{\tilde{\phi}(\Omega)} (v_l^2) \mathrm{div} \mu dy.$$

Now we can state the analogue of Theorem 5.4.19 for problem (5.4.23).

**Theorem 5.4.26.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ . Let Fbe a non-empty finite subset of  $\mathbb{N}$ . Let  $\mathcal{V}_0 \in ]0, +\infty[$ . Let  $\tilde{\phi} \in V(\mathcal{V}_0)$  be such that  $\partial \tilde{\phi}(\Omega) \in C^4$  and  $\lambda_j[\tilde{\phi}]$  have a common value  $\lambda_F[\tilde{\phi}]$  for all  $j \in F$  and  $\lambda_l[\tilde{\phi}] \neq \lambda_F[\tilde{\phi}]$  for all  $l \in \mathbb{N} \setminus F$ . For  $s = 1, \ldots, |F|$ , the function  $\tilde{\phi}$  is a critical point for  $\Lambda_{F,s}$  on  $V(\mathcal{V}_0)$  if and only if there exists an orthonormal basis  $v_1, \ldots, v_{|F|}$  of the eigenspace corresponding to the eigenvalue  $\lambda_F[\tilde{\phi}]$  of problem (5.4.23) in  $L^2(\tilde{\phi}(\Omega))$ , and a constant  $c \in \mathbb{R}$  such that

$$\sum_{l=1}^{|F|} \left( \lambda_F v_l^2 - \tau |\nabla v_l|^2 - |D^2 v_l|^2 \right) = c, \text{ a.e. on } \partial \tilde{\phi}(\Omega).$$

We observe that Lemma 5.4.22 holds for problem (5.4.23) as well, since in the proof we have only used the rotation invariance of the Laplace operator. Then, we are led to the following theorem.

**Theorem 5.4.27.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ . Let  $\tilde{\phi} \in \Phi(\Omega)$  be such that  $\tilde{\phi}(\Omega)$  is a ball. Let  $\tilde{\lambda}$  be an eigenvalue of problem (5.4.23) on  $\tilde{\phi}(\Omega)$ , and let F be the set of  $j \in \mathbb{N}$  such that  $\lambda_j[\tilde{\phi}] = \tilde{\lambda}$ . Then  $\Lambda_{F,s}$  has a critical point at  $\tilde{\phi}$  on  $V(\mathcal{V}(\tilde{\phi}))$ , for all  $s = 1, \ldots, |F|$ .

## 5.5 The fundamental tone of the ball. An isoperimetric inequality

In the previous section we have shown that the ball is a critical point for all the elementary symmetric functions of the eigenvalues of problem (5.0.1) when  $\rho \equiv 1$ . In this section we prove that the ball is actually a maximizer for the fundamental tone, that is

$$\lambda_2(\Omega) \le \lambda_2(\Omega^*),\tag{5.5.1}$$

where  $\Omega^*$  is a ball such that  $|\Omega| = |\Omega^*|$ . Through all this section we consider problem (5.0.1) which  $\rho \equiv 1$ .

#### 5.5.1 Eigenvalues and eigenfunctions on the ball

We compute the eigenvalues and the eigenfunctions of (5.0.1) when  $\Omega = B$ is the unit ball in  $\mathbb{R}^N$  centered at the origin. It is convenient to use spherical coordinates  $(r, \theta) = (r, \theta_1, ..., \theta_{N-1}) \in \mathbb{R}_+ \times \partial B$  in  $\mathbb{R}^N$ .

The boundary conditions of (5.0.1) in this case are written as

$$\frac{\partial^2 u}{\partial r^2}\Big|_{r=1} = 0, \tag{5.5.2}$$

$$\tau \frac{\partial u}{\partial r} - \frac{1}{r^2} \Delta_S \left( \frac{\partial u}{\partial r} - \frac{u}{r} \right) - \frac{\partial \Delta u}{\partial r}\Big|_{r=1} = \lambda u_{|r=1}, \quad (5.5.3)$$

where  $\Delta_S$  is the angular part of the Laplacian.

It is well known that the eigenfunctions can be written as a product of a radial part and an angular part (see [28] for details). The radial part is given in terms of ultraspherical modified Bessel functions and powertype functions and the angular part is given in terms of spherical harmonics. We have the following theorem.

**Theorem 5.5.4.** Let  $\Omega$  be the unit ball in  $\mathbb{R}^N$  centered at the origin. Any eigenfunction  $u_l$  of problem (5.0.1) is of the form  $u_l(r, \theta) = R_l(r)H_l(\theta)$  where  $H_l(\theta)$  is a spherical harmonic of some order  $l \in \mathbb{N}$  and

$$R_l(r) = A_l r^l + B_l i_l(\sqrt{\tau}r),$$

where  $A_l$  and  $B_l$  are suitable constants such that

$$B_l = \frac{l(1-l)}{\tau i_l''(\sqrt{\tau})} A_l.$$

Moreover, the eigenvalue  $\lambda_{(l)}$  associated with the eigenfunction  $u_l$  is delivered by the formula

$$\lambda_{(l)} = l \Big( (1-l) l i_l(\sqrt{\tau}) + \tau i_l''(\sqrt{\tau}) \Big)^{-1} \Big[ 3(l-1) l (l+N-2) i_l(\sqrt{\tau}) \\ - (l-1) \sqrt{\tau} \big( N-1+2Nl+2l(l-2)l+\tau \big) i_l'(\sqrt{\tau}) \\ + \tau \big( (l-1)(l+2N-3)+\tau \big) i_l''(\sqrt{\tau}) \\ + (l-1)\tau \sqrt{\tau} i_l'''(\sqrt{\tau}) \Big], \quad (5.5.5)$$

for any  $l \in \mathbb{N}_0$ .

*Proof.* Solutions to problem (5.0.1) in the unit ball are smooth (see e.g., [47, Theorem 2.20]). We consider two cases:  $\Delta u = 0$  and  $\Delta u \neq 0$ .

Let u be such that  $\Delta u = 0$ . The Laplacian can be written in spherical coordinates as

$$\Delta = \partial_{rr} + \frac{N-1}{r} \partial_r + \frac{1}{r^2} \Delta_S.$$

Separating variables so that  $u = R(r)Y(\theta)$  we obtain the equations

$$R'' + \frac{N-1}{r}R' - \frac{l(l+N-2)}{r^2}R = 0$$
(5.5.6)

and

$$\Delta_S Y = -l(l+N-2)Y.$$
 (5.5.7)

The solutions to equation (5.5.6) are given by  $R(r) = ar^l + br^{2-N-l}$  if  $l > 0, N \ge 2$ , and by  $R(r) = a + b \log(r)$  if l = 0, N = 2. Since the solutions cannot blow up at r = 0, we must impose b = 0. The solutions of the second equation are the spherical harmonics of order l. Then u can be written as

$$u(r,\theta) = a_l r^l Y_l(\theta)$$

for some  $l \in \mathbb{N}_0$ .

Let us consider now the case  $\Delta u \neq 0$ . We set  $v = \Delta u$  and solve the equation

 $\Delta v = \tau v.$ 

By writing  $v = R(r)Y(\theta)$  we obtain that R solves the equation

$$R'' + \frac{N-1}{r}R' - \frac{l(l+N-2)}{r^2}R = \tau R,$$
(5.5.8)

while Y solves equation (5.5.7). Equation (5.5.8) is the modified ultraspherical Bessel equation that is solved by the modified ultraspherical Bessel functions of first and second kind  $i_l(\sqrt{\tau}r)$  and  $k_l(\sqrt{\tau}r)$ . Since the solutions cannot blow up at r = 0, we must choose only  $i_l(z)$  since  $k_l(z)$  has a singularity at z = 0. Then

$$v(r,\theta) = b_{l_1} i_{l_1}(\sqrt{\tau}r) Y_{l_1}(\theta)$$

for some  $l_1 \in \mathbb{N}_0$ . Now  $v = \frac{\Delta v}{\tau} = \Delta u$ , that is  $\Delta(v/\tau - u) = 0$ . This means that

$$u(r,\theta) = \frac{b_{l_1}}{\tau} i_{l_1}(\sqrt{\tau}r) Y_{l_1}(\theta) - c_{l_2} r^{l_2} Y_{l_2}(\theta)$$
(5.5.9)

for some  $l_2 \in \mathbb{N}_0$ .

Now we prove that the indexes  $l_1$  and  $l_2$  in (5.5.9) must coincide. This can be shown by imposing the boundary condition (5.5.2), which can be written as

$$b_{l_1}i_{l_1}''(\sqrt{\tau})Y_{l_1}(\theta) - c_{l_2}l_2(l_2-1)Y_{l_2}(\theta) = 0.$$

If the two indexes do not agree, the coefficients of  $Y_{l_i}$ , i = 1, 2 must vanish since spherical harmonics with different indexes are linearly independent on  $\partial\Omega$ . Since  $i''_{l_1}(\sqrt{\tau}) > 0$ , this implies  $b_{l_1} = 0$  and therefore  $l_2 = 0$  or  $l_2 = 1$ . Thus we have

$$u_l(r,\theta) = \left(A_l r^l + B_l i_l(\sqrt{\tau}r)\right) H_l(\theta), \qquad (5.5.10)$$

with suitable constants  $A_l, B_l$ . In the case  $l \neq 0, 1$ , again from the boundary condition (5.5.2) we have

$$l(l-1)A_l + \tau i_l''(\sqrt{\tau})B_l = 0, \qquad (5.5.11)$$

hence  $B_l = \frac{l(1-l)}{\tau i_l''(\sqrt{\tau})} A_l$ . Note that the formula holds also in the case l = 0, 1 since these indexes correspond to  $B_l = 0$ .

Finally, let us consider the boundary condition (5.5.3). Using in (5.5.3) the representation of  $u_l$  provided by formula (5.5.10), we get

$$\begin{bmatrix} \left(-\lambda + l\left((l-1)(l+N-2)+\tau\right)\right)A_l + \left(-\left(3l(l+N-2)+\lambda\right)i_l(\sqrt{\tau}\right) \\ -\sqrt{\tau}\left((N-1-2Nl-2(l-2)l-\tau\right)i_l'(\sqrt{\tau}) + (N-1)\sqrt{\tau}i_l''(\sqrt{\tau}) \\ +\tau i_l'''(\sqrt{\tau})\right)B_l \end{bmatrix}H_l(\theta) = \lambda \left(A_l + B_l i_l(\sqrt{\tau})\right)H_l(\theta).$$

Using equality (5.5.11) we get that the function  $u_l$  given by (5.5.10) is an eigenfunction of (5.0.1) on the unit ball. Moreover, as a consequence, we also get formula (5.5.5) for the associated eigenvalue. This concludes the proof.

We are ready to state and prove the following theorem concerning the first positive eigenvalue.

**Theorem 5.5.12.** Let  $\Omega$  be the unit ball in  $\mathbb{R}^N$  centered at the origin. The first positive eigenvalue of (5.0.1) is  $\lambda_2 = \lambda_{(1)} = \tau$ . The corresponding eigenspace is generated by  $\{x_1, x_2, ..., x_N\}$ .

*Proof.* By Theorem 5.5.4,  $0 = \lambda_{(0)} < \tau = \lambda_{(1)}$ . We consider formula (5.5.5) with l = 2. We have

$$\lambda_{(2)} = 2 \Big( \tau i_2''(\sqrt{\tau}) - 2i_2(\sqrt{\tau}) \Big)^{-1} \Big[ 6Ni_2(\sqrt{\tau}) - \sqrt{\tau}(5N - 1 + \tau)i_2'(\sqrt{\tau}) \\ + \tau (2N - 1 + \tau)i_2''(\sqrt{\tau}) + \tau \sqrt{\tau}i_2'''(\sqrt{\tau}) \Big]. \quad (5.5.13)$$

In order to prove that  $\lambda_{(2)} > \tau$ , we use some well-known recurrence relations between ultraspherical Bessel functions (see [1, p. 376]),

$$\begin{aligned} i'_{l}(\sqrt{\tau}) &= \frac{l}{\sqrt{\tau}}i_{l}(\sqrt{\tau}) + i_{l+1}(\sqrt{\tau}), \\ i''_{l}(\sqrt{\tau}) &= \frac{l(l-1)}{\tau}i_{l}(\sqrt{\tau}) + \frac{l+2}{\tau}i_{l+1}(\sqrt{\tau}) + i_{l+2}(\sqrt{\tau}), \\ i'''_{l}(\sqrt{\tau}) &= \frac{l(l-1)(l-2)}{\tau\sqrt{\tau}}i_{l}(\sqrt{\tau}) + \frac{l(2l+1)}{\tau}i_{l+1}(\sqrt{\tau}) \\ &+ \frac{2(l+2)}{\sqrt{\tau}}i_{l+2}(\sqrt{\tau}) + i_{l+3}(\sqrt{\tau}). \end{aligned}$$

Using these relations in (5.5.13), we obtain an equivalent formula for  $\lambda_{(2)}$ ,

$$\lambda_{(2)} = 2 \Big( 5\sqrt{\tau} i_3(\sqrt{\tau}) + \tau i_4(\sqrt{\tau}) \Big)^{-1} \Big[ (10N - 2 + 2\tau) i_2(\sqrt{\tau}) \\ + \Big( 2 - 10N + (7 + 10N)\sqrt{\tau} - 2\tau + 5\tau\sqrt{\tau} \Big) i_3(\sqrt{\tau}) \\ + \tau (8 + 2N + \tau) i_4(\sqrt{\tau}) + \tau\sqrt{\tau} i_5(\sqrt{\tau}) \Big].$$

By well-known properties of the functions  $I_{\nu}$  (see [1, §9]), it follows that  $i_l \geq i_{l+1}$  for all  $l \in \mathbb{N}_0$ . This implies

$$(10N - 2 + 2\tau)i_2(\sqrt{\tau}) + (2 - 10N + (7 + 10N)\sqrt{\tau} - 2\tau + 5\tau\sqrt{\tau})i_3(\sqrt{\tau}) + \tau(8 + 2N + \tau)i_4(\sqrt{\tau}) + \tau\sqrt{\tau}i_5(\sqrt{\tau}) \ge (5\tau\sqrt{\tau}i_3(\sqrt{\tau}) + \tau^2i_4(\sqrt{\tau})),$$

then

$$\lambda_{(2)} \ge 2\tau > \tau = \lambda_{(1)}.$$

Now it remains to prove that  $\lambda_{(l)}$  is an increasing function of l for  $l \geq 2$ . We adapt the method used in [28, Theorem 3]. We claim that for any smooth radial function R(r) the Rayleigh quotient

$$\mathcal{Q}(R(r)H_l(\theta)) = \frac{\int_{\partial B} \int_0^1 \left( |D^2(R(r)H_l(\theta))|^2 + \tau |\nabla(R(r)H_l(\theta))|^2 \right) r^{N-1} dr d\sigma(\theta)}{\int_{\partial B} R(r)^2 H_l(\theta)^2 d\sigma(\theta)}$$

is an increasing function of l for  $l \ge 2$ . Here and in the sequel we shall denote by  $\sigma(\theta)$  the (N-1)-dimensional measure element of the unit sphere, which is given by

$$d\sigma(\theta) = \sin(\theta_1)^{N-2} \sin(\theta_2)^{N-3} \cdots \sin(\theta_{N-2}) d\theta_1 \cdots d\theta_{N-1}$$

We consider the spherical harmonics to be normalized with respect to the  $L^2(\partial B)$  scalar product. In particular, we have that the denominator of  $\mathcal{Q}(R(r)H_l(\theta)), D[R(r)H_l(\theta)]$  is  $R^2(1)$ . Now we need to write the numerator of the Rayleigh quotient in a suitable way. We recall that the numerator of the Rayleigh quotient of a function  $u \in H^2(B)$  is given by

$$N[u] = \int_B |D^2 u|^2 + \tau |\nabla u|^2 dx,$$

We use the following pointwise identity to re-write the Hessian term:

$$|D^2 u|^2 = \frac{1}{2} \Delta \left( |\nabla u|^2 \right) - \nabla u \cdot \nabla (\Delta u).$$
(5.5.14)

We also need to write the gradient in spherical coordinates

$$\nabla u = \frac{\partial u}{\partial r}\vec{r} + \frac{1}{r}\nabla_S u, \qquad (5.5.15)$$

where  $\frac{1}{r}\nabla_S u$  is the tangential gradient of u on  $\partial B$  and  $\vec{r} = \frac{x}{r}$  is the unit normal. We note that the two vectors in the right-hand side of (5.5.15) are orthogonal. We recall that, since we considered the spherical harmonics to be normalized with respect to the  $L^2(\partial B)$  scalar product, we have

$$\int_{\partial B} |\nabla_S H_l(\theta)|^2 d\sigma(\theta) = l(l+N-2).$$
(5.5.16)

To simplify the notation, we set k := l(l + N - 2). Moreover, from the Tangential Divergence Theorem, it follows that

$$\int_{\partial B} \frac{1}{r^2} \Delta_S u d\sigma(\theta) = \int_{\partial B} \operatorname{div}_{\partial B} \left(\frac{1}{r} \nabla_S u\right) d\sigma(\theta) = 0$$
 (5.5.17)

We use (5.5.14) to re-write the integral of the Hessian as follows:

$$\int_{B} |D^{2}u|^{2} dx = \int_{B} \frac{1}{2} \Delta \left( |\nabla u|^{2} \right) - \nabla u \cdot \nabla (\Delta u) dx.$$
 (5.5.18)

Now we choose in (5.5.18)  $u = R(r)H_l(\theta)$ . We have

$$\begin{split} &\int_{B} |D^{2}u|^{2} dx \\ &= \frac{1}{2} \int_{\partial B} \int_{0}^{1} \left( \frac{\partial^{2}}{\partial r^{2}} + \frac{N-1}{r} \frac{\partial}{\partial r} + \frac{1}{r^{2}} \Delta_{S} \right) \\ &\left( (R'(r))^{2} |H_{l}(\theta)|^{2} + \frac{R(r)^{2}}{r^{2}} |\nabla_{S}H_{l}(\theta)|^{2} \right) r^{N-1} dr d\sigma(\theta) \\ &- \int_{\partial B} \int_{0}^{1} \left( R'(r)H_{l}(\theta)\vec{r} + \frac{R(r)}{r} \nabla_{S}H_{l}(\theta) \right) \\ &\cdot \nabla \left( (R''(r) + \frac{N-1}{r}R'(r) - \frac{k}{r^{2}}R(r))H_{l}(\theta) \right) r^{N-1} dr d\sigma(\theta) \\ &= \int_{\partial B} \int_{0}^{1} \left( (R''(r))^{2} + R'(r)R'''(r) + \frac{N-1}{r}R'(r)R''(r) \right) |H_{l}(\theta)|^{2}r^{N-1} dr d\sigma(\theta) \\ &+ \int_{\partial B} \int_{0}^{1} \left( \frac{(R'(r))^{2}}{r^{2}} + \frac{R(r)R''(r)}{r^{2}} + \frac{R(r)R''(r)}{r^{2}} \right) \\ &+ \frac{N-5}{r^{3}}R(r)R'(r) + \frac{4-N}{r^{4}}R(r)^{2} \right) |\nabla_{S}H_{l}(\theta)|^{2}r^{N-1} dr d\sigma(\theta) \\ &- \int_{\partial B} \int_{0}^{1} \left( R'(r)R'''(r) + \frac{N-1}{r^{4}}R'(r)R''(r) \right) |H_{l}(\theta)|^{2}r^{N-1} dr d\sigma(\theta) \\ &- \int_{\partial B} \int_{0}^{1} \left( \frac{R(r)R''(r)}{r^{2}} + \frac{N-1}{r^{3}}R(r)R'(r) - \frac{k}{r^{4}}R(r)^{2} \right) |\nabla_{S}H_{l}(\theta)|^{2}r^{N-1} dr d\sigma(\theta) \\ &= \int_{\partial B} \int_{0}^{1} \left( (R''(r))^{2} + \frac{N-1}{r^{2}}(R'(r))^{2} - \frac{2k}{r^{3}}R(r)R'(r) \right) |H_{l}(\theta)|^{2}r^{N-1} dr d\sigma(\theta) \end{split}$$

Mass concentration for fourth order operators

$$+\int_{\partial_B} \int_0^1 \left(\frac{(R'(r))^2}{r^2} - \frac{4}{r^3}R(r)R'(r) + \frac{4-N+k}{r^4}R(r)^2\right) |\nabla H_l(\theta)|^2 r^{N-1} dr d\sigma(\theta),$$
(5.5.19)

where in the first equality we have used the fact that  $-\Delta_S H_l = kH_l$  and in the second equality we have used the fact that

$$\int_{\partial B} \int_0^1 \frac{1}{r^2} \Delta_S \left( (R'(r))^2 |H_l(\theta)|^2 + \frac{R(r)^2}{r^2} |\nabla_S H_l(\theta)|^2 \right) r^{N-1} dr d\sigma(\theta) = 0$$

which is a consequence of (5.5.17). Now, expanding the integrands in (5.5.19) and using (5.5.16) and the orthonormality of  $H_l(\theta)$  with respect to the  $L^2(\partial B)$  scalar product, we obtain that

$$\int_{B} |D^{2}u|^{2} dx = \int_{0}^{1} \left( (R''(r))^{2} + \frac{N-1}{r^{2}} (R'(r))^{2} + \frac{2k}{r^{4}} \left( rR'(r) - \frac{3}{2}R(r) \right)^{2} + \frac{k(k+N-1/2)}{r^{4}} R(r)^{2} \right) r^{N-1} dr. \quad (5.5.20)$$

As for the gradient term we have

$$\int_{B} |\nabla u|^{2} dx$$

$$= \int_{\partial B} \int_{0}^{1} \left( (R'(r))^{2} |H_{l}(\theta)|^{2} + \frac{R(r)^{2}}{r^{2}} |\nabla_{S} H_{l}(\theta)|^{2} \right) r^{N-1} dr d\sigma(\theta)$$

$$= \int_{0}^{1} \left( (R'(r))^{2} + \frac{k}{r^{2}} R(r)^{2} \right) r^{N-1} dr \quad (5.5.21)$$

Combining (5.5.20) and (5.5.21) we have that the numerator of the Rayleigh quotient  $N[R(r)H_l(\theta)]$  can be written in the following form

$$N[R(r)H_{l}(\theta)] = \int_{0}^{1} \left(\frac{2k}{r^{4}} \left(rR' - \frac{3}{2}R\right)^{2} + \frac{k(k-N-1/2)}{r^{4}}R^{2} + \tau \frac{kR^{2}}{r^{2}}\right) r^{N-1}dr + \int_{0}^{1} \left((R''^{2}) + \frac{N-1}{r^{2}}(R')^{2} + \tau (R')^{2}\right) r^{N-1}dr$$

The above expression is increasing in k for  $k \ge N + 1/2$  and since k is an increasing function of l, we easily get that each term involving l is an increasing function of l for  $l \ge 2$ . Thus the claim above is proved.

For each  $l \in \mathbb{N}_0$ ,

$$\lambda_{(l)} = \inf \mathcal{Q}(u) = \inf \frac{\int_B |D^2 u|^2 + \tau |\nabla u|^2 dx}{\int_{\partial B} u^2 d\sigma}, \qquad (5.5.22)$$

162

where the infimum is taken among all functions u that are  $L^2(\partial B)$ -orthogonal to the first m-1 eigenfunctions  $u_i$  and  $m \in \mathbb{N}$  is such that  $\lambda_{(l)} = \lambda_m$  is the m-th eigenvalue of problem (5.0.1). The eigenfunctions  $u_l$  are of the form  $u_l = R_l(r)H_l(\theta)$ , and  $u_l$  realizes the infimum in (5.5.22). Then

$$\lambda_{(l)} = \mathcal{Q}(R_l(r)H_l(\theta)) \le \mathcal{Q}(R_{l+1}(r)H_l(\theta)) \le \mathcal{Q}(R_{l+1}(r)Y_{l+1}(\theta)) = \lambda_{(l+1)},$$

where the first inequality follows from the fact that  $R_{l+1}(r)H_l(\theta)$  is also orthogonal with respect to the  $L^2(\partial B)$  scalar product to the first m-1eigenfunctions  $R_i(r)Y_i(\theta)$  for i = 1, ..., m-1, and then it is a suitable trial function in (5.5.22). The second inequality follows from the fact that the quotient  $\mathcal{Q}(R(r)H_l(\theta))$  is an increasing function of l, for  $l \geq 2$ . This concludes the proof.

#### 5.5.2 The isoperimetric inequality

In this subsection we prove the isoperimetric inequality (5.5.1). Actually, we prove a stronger result, that is a quantitative version of (5.5.1). We adapt to our case a result of [16], where the authors prove a quantitative version of the Brock-Weinstock inequality for the Steklov Laplacian. We also refer to [58, 84] where these kind of questions have been considered for the first time (see also [17, 45]).

Throughout this section  $\Omega$  is a bounded domain of class  $C^1$ . We recall the following lemma from [16].

**Lemma 5.5.23.** Let  $\Omega$  be an open set with Lipschitz boundary and p > 1. Then

$$\int_{\partial\Omega} |x|^p d\sigma \ge \int_{\partial\Omega^*} |x|^p d\sigma \left(1 + c_{N,p} \left(\frac{|\Omega \triangle \Omega^*|}{|\Omega|}\right)^2\right),$$

where  $\Omega^*$  is the ball centered at zero with the same measure as  $\Omega$ ,  $\Omega \triangle \Omega^*$  is the symmetric difference of  $\Omega$  and  $\Omega^*$ , and  $c_{N,p}$  is a constant depending only on N and p given by

$$c_{N,p} := \frac{(N+p-1)(p-1)}{4} \frac{\sqrt[N]{2}-1}{N} \left(\min_{t \in [1, \sqrt[N]{2}]} t^{p-2}\right).$$

We also recall the following characterization of the inverses of the eigenvalues of (5.0.1) from [62] (see also [14]).

**Lemma 5.5.24.** Let  $\Omega$  be a bounded domain of class  $C^1$  in  $\mathbb{R}^N$ . Then the eigenvalues of problem (5.0.1) on  $\Omega$  satisfy,

$$\sum_{l=k+1}^{k+N} \frac{1}{\lambda_l(\Omega)} = \max\left\{\sum_{l=k+1}^{k+N} \int_{\partial\Omega} v_l^2 d\sigma\right\},\tag{5.5.25}$$

where the maximum is taken over the families  $\{v_l\}_{l=k+1}^{k+N}$  in  $H^2(\Omega)$  satisfying  $\int_{\Omega} D^2 v_i : D^2 v_j + \tau \nabla v_i \cdot \nabla v_j dx = \delta_{ij}$ , and  $\int_{\partial\Omega} v_i u_j d\sigma = 0$  for all i = k + 1, ..., k + N and j = 1, 2, ..., k, where  $u_1, u_2, ..., u_k$  are the first k eigenfunctions of problem (5.0.1).

For every open set  $\Omega \in \mathbb{R}^N$  with finite measure, we recall the definition of Fraenkel asymmetry

$$\mathcal{A}(\Omega) := \inf\left\{\frac{\|\chi_{\Omega} - \chi_B\|_{L^1(\mathbb{R}^N)}}{|\Omega|} : B \text{ ball with } |B| = |\Omega|\right\}.$$
 (5.5.26)

The quantity  $\mathcal{A}(\Omega)$  is the distance in the  $L^1(\mathbb{R}^N)$  norm of a set  $\Omega$  from the set of all balls of the same measure as  $\Omega$ . This quantity turns out to be a suitable distance between sets for the purposes of stability estimates of eigenvalues. Note that  $\mathcal{A}(\Omega)$  is scaling invariant and  $0 \leq \mathcal{A}(\Omega) < 2$ .

We are ready to prove the following theorem.

**Theorem 5.5.27.** For every domain  $\Omega$  in  $\mathbb{R}^N$  of class  $C^1$  the following estimate holds

$$\lambda_2(\Omega) \le \lambda_2(\Omega^*) \left( 1 - \delta_N \mathcal{A}(\Omega)^2 \right), \qquad (5.5.28)$$

where  $\delta_N$  is given by

$$\delta_N := \frac{c_{N,2}}{2} = \frac{N+1}{8N} \left( \sqrt[N]{2} - 1 \right),$$

and  $\Omega^*$  is a ball with the same measure as  $\Omega$ .

Proof. Let  $\Omega$  be a bounded domain of class  $C^1$  in  $\mathbb{R}^N$  with the same measure as the unit ball B. We consider in (5.5.25) l = 2, ..., N + 1 and  $v_l = (\tau |\Omega|)^{-1/2} x_l$  as trial functions. The trial functions must have zero integral mean over  $\partial \Omega$ . This can be obtained by a change of coordinates  $x = y - \frac{1}{|\partial \Omega|} \int_{\partial \Omega} y d\sigma$ . Moreover, the functions  $v_l$  satisfy the normalization condition of Lemma 5.5.24. Then  $v_l$  can be used as test functions in (5.5.25). We get

$$\sum_{l=2}^{N+1} \frac{1}{\lambda_l(\Omega)} \ge \frac{1}{\tau |\Omega|} \int_{\partial \Omega} |x|^2 d\sigma.$$

We use Lemma 5.5.23 with p = 2. This yields

$$\sum_{l=2}^{N+1} \frac{1}{\lambda_l(\Omega)} \ge \frac{1}{\tau |\Omega|} \int_{\partial B} |x|^2 d\sigma \left( 1 + c_{N,2} \left( \frac{|\Omega \triangle B|}{|\Omega|} \right)^2 \right)$$
$$= \frac{N|B|}{\tau |B|} \left( 1 + c_{N,2} \left( \frac{|\Omega \triangle B|}{|\Omega|} \right)^2 \right) = \frac{N}{\tau} \left( 1 + c_{N,2} \left( \frac{|\Omega \triangle B|}{|\Omega|} \right)^2 \right)$$
$$= \sum_{l=2}^{N+1} \frac{1}{\lambda_l(B)} \left( 1 + c_{N,2} \left( \frac{|\Omega \triangle B|}{|\Omega|} \right)^2 \right). \quad (5.5.29)$$

Suppose now that  $\lambda_2(\Omega) \geq \frac{\tau}{2}$ , otherwise estimate (5.5.28) is trivially true, since  $0 \leq \mathcal{A}(\Omega) < 2$ . Since  $\lambda_2(\Omega) \leq \lambda_l(\Omega)$  for all  $l \geq 3$ , inequality (5.5.29) and the definition of  $\mathcal{A}(\Omega)$  yield

$$\lambda_2(\Omega) \left( 1 + c_{N,2} \mathcal{A}(\Omega)^2 \right) \le \lambda_2(B).$$
(5.5.30)

Therefore, since  $\lambda_2(\Omega) \geq \frac{\tau}{2}$  and  $\lambda_2(B) = \tau$ , from (5.5.30) we have

$$\lambda_2(\Omega) \le \tau - \lambda_2(\Omega) c_{N,2} \mathcal{A}(\Omega)^2 \le \tau \left(1 - \frac{c_{N,2} \mathcal{A}(\Omega)^2}{2}\right),$$

which implies (5.5.28) with  $\delta_N = \frac{1}{8} \min\{1, \frac{N+1}{N}(\sqrt[N]{2}-1)\}$ . We note that  $\min\{1, \frac{N+1}{N}(\sqrt[N]{2}-1)\} = \frac{N+1}{N}(\sqrt[N]{2}-1)$ . This concludes the proof in the case  $\Omega$  has the same measure as the unit ball.

The proof for general finite values of  $|\Omega|$  relies on the well-known scaling properties of the eigenvalues. Namely, for all s > 0, if we write an eigenvalue of problem (5.0.1) as  $\lambda(\tau, \Omega)$ , we have

$$\lambda(\tau, \Omega) = s^3 \lambda(s^{-2}\tau, s\Omega).$$

This is easy to prove by looking at the variational characterization of  $\lambda(\tau, \Omega)$ and  $\lambda(s^{-2}\tau, s\Omega)$  and performing a change of variable  $x \mapsto x/s$  in the Rayleigh quotient (5.2.9). This last observation concludes the proof of the theorem.

The isoperimetric inequality (5.5.1) is an immediate consequence of Theorem 5.5.27.

**Corollary 5.5.31.** Among all bounded domains of class  $C^1$  with fixed measure, the ball maximizes the first non-negative eigenvalue of problem (5.0.1), that is  $\lambda_2(\Omega) \leq \lambda_2(\Omega^*)$ , where  $\lambda_2(\Omega)$  has been defined in (5.2.9) and  $\Omega^*$  is a ball with the same measure as  $\Omega$ .

#### **5.5.3** Some remarks on the case $\tau = 0$

Throughout this chapter we have only considered problems (5.0.1) and (5.0.3) with  $\tau > 0$ . If we set  $\tau = 0$ , problem (5.0.1) reads

$$\begin{cases} \Delta^2 u = 0, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial\Omega, \\ -\text{div}_{\partial\Omega} (D^2 u \cdot \nu) - \frac{\partial \Delta u}{\partial \nu} = \lambda u, & \text{on } \partial\Omega, \end{cases}$$
(5.5.32)

while problem (5.0.3) reads

$$\begin{cases} \Delta^2 u = \lambda u, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial\Omega, \\ \operatorname{div}_{\partial\Omega} \left( D^2 u \cdot \nu \right) + \frac{\partial \Delta u}{\partial \nu} = 0, & \text{on } \partial\Omega. \end{cases}$$
(5.5.33)

Problems (5.5.32) and (5.5.33) model free vibrating plates which are not subject to lateral tension. These problems have a sequence of non-negative eigenvalues of finite multiplicity and the corresponding eigenfunctions form a orthonormal basis of  $H^2(\Omega)$ . The coordinate functions  $x_1, ..., x_N$  and the constant functions are eigenfunctions of both problems (5.5.32) and (5.5.33) corresponding to the eigenvalue  $\lambda = 0$ , which has multiplicity N + 1. Therefore, the first non-zero eigenvalue is the (N + 2)-th eigenvalue.

As we did in Theorem 5.3.6, we can define the family of problems

$$\begin{cases} \Delta^2 u = \lambda \rho_{\varepsilon} u, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial\Omega, \\ \operatorname{div}_{\partial\Omega} (D^2 u \cdot \nu) + \frac{\partial \Delta u}{\partial \nu} = 0, & \text{on } \partial\Omega, \end{cases}$$
(5.5.34)

where  $\rho_{\varepsilon}$  is defined as in (3.1.20). We have the following theorem, whose proof can be easily done adapting that of Theorem 5.3.6.

**Theorem 5.5.35.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^2$ . Let  $\rho_{\varepsilon}$  be defined by (3.1.20). Let  $\lambda_j[\rho_{\varepsilon}]$  be the eigenvalues of problem (5.5.34) on  $\Omega$  for all  $j \in \mathbb{N}$ . Let  $\lambda_j$ ,  $j \in \mathbb{N}$  denote the eigenvalues of problem (5.5.32) corresponding to the constant surface density  $\frac{M}{|\partial\Omega|}$ . Then we have  $\lim_{\varepsilon \to 0} \lambda_j[\rho_{\varepsilon}] = \lambda_j$  for all  $j \in \mathbb{N}$ .

It is clear that a discussion similar to that of Section 5.4 can be carried out for problems (5.5.32) and (5.5.33) as well, by means of a change of the projections  $\pi_{\phi}^{S}, \pi_{\phi}^{N}$  according to the kernel. In particular, all the formulas in Section 5.4 remain true, by setting  $\tau = 0$ . Then we have the following theorem.

**Theorem 5.5.36.** Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$ . Let  $\tilde{\phi} \in \Phi(\Omega)$  be such that  $\tilde{\phi}(\Omega)$  is a ball. Let  $\tilde{\lambda}$  be an eigenvalue of problem (5.5.32) (problem (5.5.33) respectively) in  $\tilde{\phi}(\Omega)$ , and let F be the set of  $j \in \mathbb{N}$  such that  $\lambda_j[\tilde{\phi}] =$  $\tilde{\lambda}$ . Then  $\Lambda_{F,s}$  has a critical point at  $\tilde{\phi}$  on  $V(\mathcal{V}(\tilde{\phi}))$ , for all  $s = 1, \ldots, |F|$ .

Moreover, for problem (5.5.32), it is possible to identify the fundamental modes and the fundamental tone on the ball. We have the following

**Theorem 5.5.37.** Let  $\Omega = B$  be the unit ball in  $\mathbb{R}^N$ . The eigenfunctions of problem (5.5.32) are of the form

$$u_l(r,\theta) = \left(A_l r^l + B_l r^{2+l}\right) H_l(\theta),$$

for  $l \in \mathbb{N}_0$ , where  $A_l$  and  $B_l$  are suitable constants such that

$$B_l = -\frac{l(l-1)}{(l+2)(l+1)}A_l.$$

The eigenvalues  $\lambda_{(l)}$  of problem (5.5.32) corresponding to the eigenfunctions  $u_l(r, \theta)$  are delivered by the formula

$$\lambda_{(l)} = \frac{l(l-1)\left(N+2Nl+(l-1)(2+3l)\right)}{1+2l}.$$
(5.5.38)

The first positive eigenvalue is

$$\lambda_{N+2} = \lambda_{(2)} = 2\left(N + \frac{8}{5}\right), \qquad (5.5.39)$$

and the corresponding eigenfunctions are

$$u_2(r,\theta) = (6r^2 - r^4)H_2(\theta).$$
(5.5.40)

Proof. The proof is similar to that of Theorem 5.5.4, from which it differs only for the use of biharmonic functions on the ball as solutions of the differential equation  $\Delta^2 u = 0$ . Any smooth and bounded function u on B which satisfies  $\Delta^2 u = 0$  can be written in spherical coordinates  $(r, \theta)$ as a linear combination of functions of the form  $(ar^l + br^{2+l}) H_l(\theta)$  with  $l \in \mathbb{N}_0$  (see also [6, 7, 90] for a complete characterization of biharmonic and poly-harmonic functions on the unit ball). The rest of the proof follows the same lines as that of Theorem 5.5.4. Then, once formula (5.5.38) has been established, it is straightforward to identify the fundamental tone (5.5.39) and the corresponding modes (5.5.40).

Now we have an explicit form for the fundamental tone and for the corresponding eigenfunctions in the case of the unit ball which suggests how to construct trial functions for the Rayleigh quotient of  $\lambda_{N+2}$ . If we want to use a function of the form  $R(r)H_2(\theta)$  as a test function as we did in Theorem 5.5.27 we must impose that  $R(r)H_2(\theta)$  is othogonal to the constants and to the coordinate functions with respect to the  $L^2(\partial\Omega)$  scalar product and we can no more obtain this just by translating the domain  $\Omega$ .

We note that functions of the form  $R(r)H_2(\theta)$  are suitable trial functions for the annuli centered at zero. We recall from formula (5.5.40) that the radial part of an eigenfunction associated with the first positive eigenvalue  $\lambda_{N+2}$  of th unit ball is of the form  $6r^2 - r^4$ . We want to construct a test function for the annulus centered at zero of the form  $R(r)H_2(\theta)$  in such a way that the radial part R(r) equals  $6r^2 - r^4$  whenever  $r \leq 1$ . Moreover we want that the radial part is an increasing function of r, for  $r \in [0, +\infty[$ . We note that  $6r^2 - r^4$  is increasing for  $r \in [0, 1]$ . Note that test functions must belong to  $H^2(\Omega)$ . Therefore we choose R(r) = 8r - 3 for r > 1 which is increasing and moreover, R(r) and R'(r) are continuous at r = 1. With this choice of the test function it possible to compare the Rayleigh quotients of the annulus and of the ball and prove that the first positive eigenvalue of the ball centered at zero is bigger than the first positive eigenvalue of an annulus centered at zero with the same measure. We need the following preliminary lemma. **Lemma 5.5.41.** Let  $\Omega$  be an open set in  $\mathbb{R}^N$  and  $\Omega^*$  be the ball centered at zero with the same measure as  $\Omega$ . Let F be a measurable and radial function such that

$$F(x) > F(y) \quad \forall x \in \Omega^*, y \notin \Omega^*.$$
(5.5.42)

Then

$$\int_{\Omega} F dx \le \int_{\Omega^*} F dx,$$

with equality if and only if  $\Omega = \Omega^*$ .

*Proof.* Since  $|\Omega| = |\Omega^*|$  we have that  $|\Omega \setminus \Omega^*| = |\Omega^* \setminus \Omega|$ . Let F be a measurable function satisfying condition (5.5.42). We have

$$\begin{split} \int_{\Omega} Fdx &= \int_{\Omega \cap \Omega^*} Fdx + \int_{\Omega \setminus \Omega^*} Fdx \\ &\leq \int_{\Omega \cap \Omega^*} Fdx + \sup_{x \in \Omega \cap \Omega^*} |F(x)| |\Omega \setminus \Omega^*| \\ &\leq \int_{\Omega \cap \Omega^*} Fdx + \inf_{x \in \Omega^* \setminus \Omega} |F(x)| |\Omega^* \setminus \Omega| \\ &\leq \int_{\Omega \cap \Omega^*} Fdx + \int_{\Omega^* \setminus \Omega} Fdx = \int_{\Omega^*} Fdx. \end{split}$$

The second inequality follows from (5.5.42). Note that if  $|\Omega \setminus \Omega^*| > 0$ , either the second inequality or the third is strict by the strict inequality in (5.5.42).

We have the following theorem (see also Figure 5.1).

**Theorem 5.5.43.** Let B be the unit ball in  $\mathbb{R}^N$  centered at zero. Let  $\rho > 0$ and  $A_{\rho}$  be the subset of  $\mathbb{R}^N$  be defined by

$$A_{\rho} := \left\{ x \in \mathbb{R}^{N} : \rho < |x| < (1 + \rho^{N})^{\frac{1}{N}} \right\}.$$

Then

$$\lambda_{N+2}(B) \ge \lambda_{N+2}(A_{\rho})$$

for all  $\rho \in ]0, +\infty[$ .

*Proof.* Let the function R(r) from  $[0, +\infty]$  to  $\mathbb{R}$  be defined by

$$R(r) = \begin{cases} 6r^2 - r^4, & \text{if } r \in [0, 1], \\ 8r - 3, & \text{if } r \in ]1, +\infty[ \end{cases}$$

We note that by construction,  $R \in C^2([0, +\infty[)$ . We set  $v(r, \theta) := R(r)H_2(\theta)$ , where  $H_2(\theta)$  is a spherical harmonic of degree 2 normalized by  $\int_{\partial B} H_2^2 d\sigma =$  1. With this choice we also have that  $\int_{\partial B} |\nabla_S H_2|^2 d\sigma = 2N$ . Moreover, from well-known properties of spherical harmonics, we have that it holds

$$\int_{\partial A_{\rho}} H_2 d\sigma = 0$$

and

$$\int_{\partial A_{\rho}} \frac{x_l}{|x_l|} H_2 d\sigma = 0, \quad \forall l = 1, \dots N.$$

We recall the following characterization of  $\lambda_{N+2}(A_{\rho})$ 

$$\lambda_{N+2}(A_{\rho}) = \min_{\substack{0 \neq u \in H^2(\Omega) \\ \int_{\partial A_{\rho}} u d\sigma = \int_{\partial A_{\rho}} x_i u d\sigma = 0, \forall i=1,\dots,N}} \frac{\int_{A_{\rho}} |D^2 u|^2 dx}{\int_{\partial A_{\rho}} u^2 d\sigma}.$$
 (5.5.44)

Therefore, since v is a suitable test function for the Rayleigh quotient (5.5.44) we have

$$\lambda_{N+2}(A_{\rho}) \leq \frac{\int_{A_{\rho}} |D^2 v|^2 dx}{\int_{\partial A_{\rho}} v^2 d\sigma}.$$
(5.5.45)

By following the same lines of the proof of formula (5.5.20) we compute the integral of  $|D^2v|^2$  on  $A_{\rho}$ . We have

$$\begin{split} \int_{A_{\rho}} |D^{2}v|^{2} dx \\ &= \int_{\rho}^{(1+\rho^{N})^{\frac{1}{N}}} \left( R''(r)^{2} + \frac{5N-1}{r^{2}} R'(r)^{2} \right. \\ &\left. - \frac{12N}{r^{3}} R'(r) R(r) + \frac{2N(3N+4)}{r^{4}} R(r)^{2} \right) r^{N-1} dr \\ &= \frac{1}{N\omega_{N}} \int_{A_{\rho}} F(|x|) dx, \quad (5.5.46) \end{split}$$

where in the last equality we have multiplied and divided the integral by  $\int_{\partial B} d\sigma$  which equals  $N\omega_N$ , and we have re-written the integral in cartesian coordinates. The function F(r) is given by

$$F(r) := R''(r)^2 + \frac{5N-1}{r^2}R'(r)^2 - \frac{12N}{r^3}R'(r)R(r) + \frac{2N(3N+4)}{r^4}R(r)^2.$$

As for the denominator of the Rayleigh quotient (5.5.45), from the normalization of  $H_2(\theta)$  we have

$$\int_{\partial A_{\rho}} v^2 d\sigma = R(\rho)^2 + R\left((1+\rho^N)^{\frac{1}{N}}\right)^2.$$
 (5.5.47)

From (5.5.46) and (5.5.47) we have

$$\lambda_{N+2}(A_{\rho}) \leq \frac{\int_{A_{\rho}} F(|x|) dx}{N\omega_N \left( R(\rho)^2 + R\left( (1+\rho^N)^{\frac{1}{N}} \right)^2 \right)},$$

From standard computations it follows that  $R(\rho)^2 + R\left(\left(1+\rho^N\right)^{\frac{1}{N}}\right)^2$  is an increasing function of  $\rho$ , for  $\rho \in [0, +\infty[$ . Therefore it attains its minimum at  $\rho = 0$  and such a minimum is 25. As for the numerator, it is possible to prove that the function F(r) is a decreasing function of r, for  $r \in [0, 1]$ . In fact, when  $r \in [0, 1]$  we have

$$F(r) = 72N(3N+2) - 24(8+3N(N+2))r^2 + 2(64+N(3N+20)),$$

and

$$F'(r) = 8(64 + N(3N + 20))r^3 - 48(8 + 3N(N + 2))r.$$

Since  $r \in [0, 1], r^3 \leq r$ . Then

$$F'(r) \le 8(64 + N(3N + 20))r - 48(8 + 3N(N + 2)) = -8r(16(N - 1) + 15N^2) < 0.$$
  
Now, let  $r \in ]1, +\infty[$ . We set  $\tilde{F}(r) := r^4(F(1) - F(r))$  for  $r \ge 1$ . We have

$$\begin{split} \tilde{F}(r) \\ &= 2(5N+4)(15N-8)r^4 - 64(6N^2+N-1)r^2 + 96N(3N+1)r - 18N(3N+4). \end{split}$$

We show that  $\tilde{F}(r) > 0$  for all  $r \in ]1, +\infty[$ . Clearly  $\tilde{F}(1) = 0$ . We compute  $\tilde{F}'(1) = 8(16(N-1) + 15N^2) > 0$ . Moreover we have  $\tilde{F}''(r) = 24(5N + 4)(15N - 8)r^2 - 128(6N^2 + N - 1) > 8(N(129N + 44) - 80) > 0$  since r > 1. We have shown that  $F(|x_1|) > F(|x_2|)$  for all  $x_1, x_2$  such that  $x_1 \in B$ ,  $x_2 \notin B$ . Therefore, from Lemma 5.5.41 it follows that

$$\int_{A_{\rho}} F(|x|) dx \le \int_{B} F(|x|) dx = 10(8+5N)N\omega_{N}.$$

From this it follows that

$$\lambda_{N+2}(A_{\rho}) \leq \frac{10(8+5N)N\omega_N}{25N\omega_N} = 2\left(N+\frac{8}{5}\right) = \lambda_{N+2}(B).$$

The results contained in this subsection suggest that the ball should be a maximizer also for problems (5.5.32) and (5.5.33). For what concerns problem (5.5.33), unfortunately a characterization of the fundamental tone is still unavailable.



## 5.6 Neumann isoperimetric inequality in quantitative form. Sharpness of Neumann and Steklov inequalities

In this section, we consider problem (5.0.3) with  $\rho \equiv 1$ , namely problem

$$\begin{cases} \Delta^2 u - \tau \Delta u = \lambda u, & \text{in } \Omega, \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial \Omega, \\ \tau \frac{\partial u}{\partial \nu} - \operatorname{div}_{\partial \Omega} \left( D^2 u \cdot \nu \right) - \frac{\partial \Delta u}{\partial \nu} = 0, & \text{on } \partial \Omega. \end{cases}$$
(5.6.1)

We recall the following isoperimetric inequality for the fundamental tone of problem (5.6.1) proved in [28]

$$\lambda_2(\Omega) \le \lambda_2(\Omega^*),\tag{5.6.2}$$

where  $\Omega$  is a bounded domain in  $\mathbb{R}^N$  of class  $C^1$ ,  $\lambda_2(\Omega)$  is the first positive eigenvalue of problem (5.6.1) on  $\Omega$ , and  $\Omega^*$  is a ball such that  $|\Omega| = |\Omega^*|$ . The aim of this section is to improve inequality (5.6.2) in the following quantitative form

$$\lambda_2(\Omega) \le \lambda_2(\Omega^*) \left( 1 - C\mathcal{A}(\Omega)^2 \right), \qquad (5.6.3)$$

where  $\mathcal{A}(\Omega)$  is the Fraenkel asymmetry, defined in (5.5.26). In this section, we also prove the sharpness of both inequality (5.6.3) and inequality (5.5.28) corresponding to the biharmonic Steklov problem (5.5.28).

We introduce now some preliminaries which are used throughout the section, and we recall some results proved in [28]. Let B be the unit ball in  $\mathbb{R}^N$  centered at zero. For a fixed  $\tau > 0$ , we take positive constants a, b satisfying  $a^2b^2 = \lambda_2(B)$  and  $b^2 - a^2 = \tau$ . We then define the function

$$R(r) = j_1(ar) + \gamma i_1(br),$$

where

$$\gamma = -\frac{a^2 j_1''(a)}{b^2 i_1''(b)},$$

and  $j_1(z)$  and  $i_1(z)$  are the ultraspherical and modified ultraspherical Bessel functions of the first species and order 1 respectively. We then define the function  $\rho: [0, +\infty[ \rightarrow [0, +\infty[$  by

$$\rho(r) := \begin{cases} R(r), & \text{if } r \in [0, 1[\\ R(1) + (r-1)R'(1), & \text{if } r \in [1, +\infty[. \end{cases}) \end{cases}$$

Let  $u_k : \mathbb{R}^N \to \mathbb{R}$  be the functions defined by

$$u_k(x) := \rho(|x|) \frac{x_k}{|x_k|},\tag{5.6.4}$$

for k = 1, ..., N. The functions  $u_{k|B}$  are in fact the eigenfunctions associated with the eigenvalue  $\lambda_2(B)$  of the Neumann problem (5.6.1) on the unit ball B, which has multiplicity N (see [29, Theorem 3]). Moreover, we have (see [28, p. 437])

$$\sum_{k=1}^{N} |u_k|^2 = \rho(|x|)^2;$$
  

$$\sum_{k=1}^{N} |Du_k|^2 = \frac{N-1}{r^2} \rho(|x|)^2 + (\rho'(|x|))^2;$$
  

$$\sum_{k=1}^{N} |D^2u_k|^2 = (\rho''(|x|))^2 + \frac{3(N-1)}{|x|^4} (\rho(|x|) - |x|\rho'(|x|))^2.$$

We denote by  $N[\rho]$  the quantity

$$N[\rho] := \sum_{k=1}^{N} |D^2 u_k|^2 + \tau |Du_k|^2.$$

Finally, we recall some properties enjoyed by the functions  $\rho$  and  $N[\rho]$  which were proved in [28].

**Lemma 5.6.5.** The function  $\rho$  satisfies the following properties.

- i)  $\rho''(r) \leq 0$  for all  $r \geq 0$ , therefore  $\rho'$  is non-increasing.
- ii)  $\rho(r) r\rho'(r) \ge 0$  for all  $r \ge 0$ , equality holding only for r = 0.
- iii) The function  $\rho(r)^2$  is strictly increasing.
- iv) The function  $\rho(r)^2/r^2$  is decreasing.

172

- v) The function  $3(\rho(r) r\rho'(r))^2/r^4 + \tau \rho^2(r)/r^2$  is decreasing.
- vi)  $N[\rho(r_1)] > N[\rho(r_2)]$  for all  $r_1 \in [0, 1[, r_2 \in [1, +\infty[.$
- vii) For all  $r \ge 0$  we have

$$N[\rho] = (\rho'')^2 + \frac{3(N-1)(\rho(r) - r\rho'(r))^2}{r^4} + \tau \frac{(N-1)\rho^2(r)}{r^2} + \tau (\rho'(r))^2.$$

viii) For all  $r \geq 1$ ,  $N[\rho]$  is decreasing.

# 5.6.1 Quantitative isoperimetric inequality for the Neumann problem

In this subsection we state and prove the Neumann quantitative inequality:

**Theorem 5.6.6.** For every bounded domain  $\Omega$  in  $\mathbb{R}^N$  of class  $C^1$  the following estimate holds

$$\lambda_2(\Omega) \le \lambda_2(\Omega^*) \left( 1 - \eta_{N,\tau,|\Omega|} \mathcal{A}(\Omega)^2 \right), \qquad (5.6.7)$$

where

 $\eta_{N,\tau,|\Omega|} > 0$ 

is a positive constant, and  $\Omega^*$  is a ball such that  $|\Omega^*| = |\Omega|$ .

*Proof.* Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^1$  with the same measure as the unit ball B. We recall the variational characterization of the second eigenvalue  $\lambda_2(\Omega)$  of (5.6.1) on  $\Omega$ :

$$\lambda_2(\Omega) = \inf_{\substack{0 \neq u \in H^2(\Omega) \\ \int_{\Omega} u dx = 0}} \frac{\int_{\Omega} |D^2 u|^2 + \tau |Du|^2 dx}{\int_{\Omega} u^2 dx}.$$
 (5.6.8)

Let  $u_k(x)$ , for k = 1, ..., N be the eigenfunctions corresponding to  $\lambda_2(B)$  defined in (5.6.4). Clearly  $u_{k|_{\Omega}} \in H^2(\Omega)$  by construction. It is possible to choose the origin of the coordinate axes in  $\mathbb{R}^N$  in such a way that  $\int_{\Omega} u_k dx = 0$  for all k = 1, ..., N. With this choice, the functions  $u_k$  are suitable trial functions for the Rayleigh quotient (5.6.8). Once we have fixed the origin, let

$$\alpha := \frac{|\Omega \triangle B|}{|\Omega|}.$$

By definition of Fraenkel asymmetry we have

$$\mathcal{A}(\Omega) \le \alpha. \tag{5.6.9}$$

From the variational characterization (5.6.8), it follows that for each  $k = 1, \ldots, N$ ,

$$\lambda_2(\Omega) \le \frac{\int_{\Omega} |D^2 u_k|^2 + \tau |D u_k|^2 dx}{\int_{\Omega} u_k^2 dx}.$$
(5.6.10)

We multiply both sides of (5.6.10) by  $\int_{\Omega} u_k^2 dx$  and sum over  $k = 1, \ldots, N$ , obtaining

$$\lambda_2(\Omega) \le \frac{\int_{\Omega} N[\rho] dx}{\int_{\Omega} \rho^2 dx}.$$
(5.6.11)

The same procedure for  $\lambda_2(B)$  clearly yields

$$\lambda_2(B) = \frac{\int_B N[\rho] dx}{\int_B \rho^2 dx}.$$
(5.6.12)

From (5.6.11) and (5.6.12) it follows that

$$\lambda_2(B) \int_B \rho^2 dx - \lambda_2(\Omega) \int_\Omega \rho^2 dx \ge \int_B N[\rho] dx - \int_\Omega N[\rho] dx \ge 0, \quad (5.6.13)$$

where the last inequality follows from Lemma 5.6.5, vi) and [28, Lemma 14].

Now we consider the two balls  $B_1$  and  $B_2$  centered at the origin with radii  $r_1, r_2$  taken such that  $|\Omega \cap B| = |B_1| = \omega_N r_1^N$  and  $|\Omega \setminus B| = |B_2 \setminus B| = \omega_N (r_2^N - 1)$ . Then  $|B_2| = \omega_N r_2^N$ , and by construction

$$1 - r_1^N = \frac{\alpha}{2} = r_2^N - 1.$$

This follows since  $|\Omega| + |B| = |\Omega \triangle B| + 2|\Omega \cap B|$  and then  $1 - r_1^N = \alpha/2$ . Similarly,  $|\Omega \setminus B| + |\Omega \cap B| = |\Omega|$ , hence  $r_1^N = 2 - r_2^N$  and then  $r_2^N - 1 = \alpha/2$ . Now we observe, again by Lemma 5.6.5, vi) and viii) that

$$\int_{\Omega} N[\rho] dx \le \int_{B_1} N[\rho] dx + \int_{B_2 \setminus B} N[\rho] dx.$$

From this and (5.6.13), we obtain

$$\lambda_{2}(B) \int_{B} \rho^{2} dx - \lambda_{2}(\Omega) \int_{\Omega} \rho^{2} dx \ge \int_{B} N[\rho] dx - \int_{\Omega} N[\rho] dx \qquad (5.6.14)$$
$$\ge \int_{B \setminus B_{1}} N[\rho] dx - \int_{B_{2} \setminus B} N[\rho] dx.$$

Since the function  $\rho(r)^2$  is strictly increasing by Lemma 5.6.5, iii), we have

$$\int_{\Omega} \rho^2 dx \ge \int_{B} \rho^2 dx = N\omega_N \int_0^1 \rho^2(r) r^{N-1} dr =: C_{N,\tau}^{(1)}.$$

Therefore

$$\lambda_{2}(B) \int_{B} \rho^{2} dx - \lambda_{2}(\Omega) \int_{\Omega} \rho^{2} dx$$
  
=  $(\lambda_{2}(B) - \lambda_{2}(\Omega)) \int_{B} \rho^{2} dx + \lambda_{2}(\Omega) \left( \int_{B} \rho^{2} dx - \int_{\Omega} \rho^{2} dx \right)$   
 $\leq C_{N,\tau}^{(1)} (\lambda_{2}(B) - \lambda_{2}(\Omega)).$  (5.6.15)

174

Now we consider the right-hand side of (5.6.14). We write  $N[\rho]$  more explicitly in terms of  $\rho$ , obtaining:

$$\int_{B\setminus B_1} N[\rho] dx = N\omega_N \int_{r_1}^1 \left( (\rho''(r))^2 + \frac{3(N-1)(\rho(r) - r\rho'(r))^2}{r^4} + \tau(\rho'(r))^2 + \frac{\tau(N-1)}{r^2}\rho(r)^2 \right) r^{N-1} dr$$

$$\geq N\omega_N \int_{r_1}^1 \left( \frac{3(N-1)(\rho(r) - r\rho'(r))^2}{r^4} + \tau(\rho'(r))^2 + \frac{\tau(N-1)}{r^2}\rho(r)^2 \right) r^{N-1} dr$$

$$\geq \omega_N \left( 3(N-1)(R(1) - R'(1))^2 + \tau R'(1)^2 + \tau(N-1)R(1)^2 \right) (1 - r_1^N),$$
(5.6.16)

where in the last inequality, we use the fact that the integrand is non-increasing in r by Lemma 5.6.5, i) and v). Moreover,

$$\int_{B_2 \setminus B} N[\rho] dx$$

$$= N\omega_N \int_1^{r_2} \left( \frac{3(N-1)}{r^4} (R(1) - R'(1))^2 + \tau R'(1)^2 + \frac{\tau(N-1)}{r^2} (R(1) - R'(1))^2 + 2rR'(1)(R(1) - R'(1))) \right) \\
+ \frac{\tau(N-1)}{r^2} (r^2 R'(1)^2) r^{N-1} dr \\
\leq N\omega_N \int_1^{r_2} \left( N\tau R'(1)^2 + \frac{N-1}{r} ((3+\tau)(R(1) - R'(1))^2 + 2\tau R'(1)(R(1) - R'(1))) \right) r^{N-1} dr \\
= N\omega_N \tau R'(1)^2 (r_2^N - 1) + N\omega_N ((3+\tau)(R(1) - R'(1))^2 + 2\tau R'(1)(R(1) - R'(1))) (r_2^{N-1} - 1),$$
(5.6.17)

where we have estimated the quantities  $1/r^2$  and  $1/r^4$  by 1/r. We note that  $r_2 = (1 + \alpha/2)^{1/N}$  and  $0 \le \alpha \le 2$ . Using the Taylor expansion up to order 1 and remainder in Lagrange form, we obtain

$$r_2^{N-1} = 1 + \frac{N-1}{N}\frac{\alpha}{2} - \frac{(N-1)\left(1+\frac{\xi}{2}\right)^{\frac{N-1}{N}-2}}{8N^2}\alpha^2$$
(5.6.18)  
$$\leq 1 + \frac{N-1}{N}\frac{\alpha}{2} - \frac{(N-1)2^{\frac{N-1}{N}-2}}{8N^2}\alpha^2 = 1 + \frac{N-1}{N}\frac{\alpha}{2} - c_N\alpha^2,$$

for some  $\xi \in (0, \alpha)$ , where  $c_N$  is a positive constant which depends only on N. Using (5.6.16), (5.6.17), (5.6.18) and the fact that  $1 - r_1^N = r_2^N - 1 = \alpha/2$ 

in the right-hand side of (5.6.14), we obtain:

$$\begin{split} \int_{B \setminus B_1} N[\rho] dx &- \int_{B_2 \setminus B} N[\rho] dx \qquad (5.6.19) \\ &\geq -N \omega_N \Big( (3+\tau) (R(1) - R'(1))^2 \\ &+ 2\tau R'(1) (R(1) - R'(1)) \Big) \left( \frac{N-1}{N} \frac{\alpha}{2} - c_N \alpha^2 \right) \\ &+ \omega_N \left( 3(N-1) (R(1) - R'(1))^2 + \tau R'(1)^2 + \tau (N-1) R(1)^2 \right) \frac{\alpha}{2} \\ &- N \omega_N \tau R'(1)^2 \frac{\alpha}{2} \\ &=: C_{N,\tau}^{(2)} \alpha^2, \end{split}$$

where the constant  $C_{N,\tau}^{(2)} > 0$  is given by

$$C_{N,\tau}^{(2)} = N\omega_N \left( (3+\tau)(R(1)-R'(1))^2 + 2\tau R'(1)(R(1)-R'(1)) \right) c_N.$$

From (5.6.9), (5.6.15) and (5.6.19) it follows

$$\lambda_2(B) - \lambda_2(\Omega) \ge \frac{C_{N,\tau}^{(2)}}{C_{N,\tau}^{(1)}} \mathcal{A}(\Omega)^2,$$

and therefore,

$$\lambda_2(\Omega) \le \lambda_2(B) \left( 1 - \frac{C_{N,\tau}^{(2)}}{\lambda_2(B)C_{N,\tau}^{(1)}} \mathcal{A}(\Omega)^2 \right).$$
 (5.6.20)

The isoperimetric inequality is thus proved in the case of  $\Omega$  with the same measure as the unit ball. The inequality for a generic domain  $\Omega$  follows from the well-known scaling properties of the eigenvalues of problem (5.6.1). Writing our eigenvalues as  $\lambda_2(\tau, \Omega)$  to make explicit the dependence on the parameter  $\tau$ , we have

$$\lambda_2(\tau, \Omega) = s^4 \lambda_2(s^{-2}\tau, s\Omega), \qquad (5.6.21)$$

for all s > 0. From (5.6.20) and taking  $s = (\omega_N / |\Omega|)^{1/N}$  in (5.6.21), it follows that for every  $\Omega$  in  $\mathbb{R}^N$  of class  $C^1$  we have

$$\lambda_{2}(\tau,\Omega) = s^{4}\lambda_{2}(s^{-2}\tau,s\Omega)$$

$$\leq s^{4}\lambda_{2}(s^{-2}\tau,B) \left(1 - \frac{C_{N,s^{-2}\tau}^{(2)}}{\lambda_{2}(s^{-2}\tau,B)C_{N,s^{-2}\tau}^{(1)}}\mathcal{A}(s\Omega)\right)$$

$$= \lambda_{2}(\tau,\Omega^{*}) \left(1 - \frac{C_{N,s^{-2}\tau}^{(2)}}{\lambda_{2}(s^{-2}\tau,B)C_{N,s^{-2}\tau}^{(1)}}\mathcal{A}(\Omega)\right).$$

176

We set  $\eta_{N,\tau,|\Omega|} := \frac{C_{N,s^{-2}\tau}^{(2)}}{\lambda_2(s^{-2}\tau,B)C_{N,s^{-2}\tau}^{(1)}}$ . This concludes the proof of the theorem.

#### 5.6.2 Sharpness of the Neumann inequality

In this subsection we prove that inequality (5.6.7) is sharp, that is, the exponent 2 for the Fraenkel asymmetry is optimal in the decay rate of  $\lambda_2(\Omega^*) - \lambda_2(\Omega)$ . To do so, we exhibit a family of sets  $\{\Omega_{\varepsilon}\}_{\varepsilon>0}$  approaching the unit ball centered at zero such that  $\mathcal{A}(\Omega_{\varepsilon}) \in O(\varepsilon)$  and  $\lambda_2(\Omega^*) - \lambda_2(\Omega) \in O(\varepsilon^2)$ , as  $\varepsilon \to 0$ . We have the following.

**Theorem 5.6.22.** Let B be the unit ball in  $\mathbb{R}^N$  centered at zero. There exist a family  $\{\Omega_{\varepsilon}\}_{\varepsilon>0}$  of smooth domains and positive constants  $c_1, c_2, c_3, c_4$  and  $r_1, r_2, r_3, r_4$  which do not depend on  $\varepsilon > 0$  such that

$$r_1 \varepsilon^2 \le \left| |\Omega_{\varepsilon}| - |B| \right| \le r_2 \varepsilon^2,$$
 (5.6.23)

$$c_1 \varepsilon \le c_2 \mathcal{A}(\Omega_{\varepsilon}) \le \frac{|\Omega_{\varepsilon} \triangle B|}{|\Omega_{\varepsilon}|} \le c_3 \mathcal{A}(\Omega_{\varepsilon}) \le c_4 \varepsilon,$$
 (5.6.24)

and

$$r_3\varepsilon^2 \le |\lambda_2(\Omega_\varepsilon) - \lambda_2(B)| \le r_4\varepsilon^2, \tag{5.6.25}$$

for all  $\varepsilon \in [0, \varepsilon_0[$ , where  $\varepsilon_0 > 0$  is sufficiently small.

In order to prove Theorem 5.6.22, we follow the same lines of [16], where the authors consider the same issue in the case of the Steklov eigenvalue problem for the Laplace operator.

First, we consider a class of domains satisfying suitable geometrical assumptions (see (5.6.26) and (5.6.27)). Under such assumptions, it is standard to prove that (5.6.23) and (5.6.24) hold. In order to prove (5.6.25), we construct suitable test functions for the Rayleigh quotient of  $\lambda_2(\Omega_{\varepsilon})$  starting from the eigenfunctions associated with  $\lambda_2(B)$ . We obtain an estimate for  $\lambda_2(B) - \lambda_2(\Omega)$  in terms of some error functions  $R_1(\varepsilon), R_2(\varepsilon)$  (see (5.6.38)). Then we prove that a better estimates of  $R_1(\varepsilon)$  and  $R_2(\varepsilon)$  produces a better estimates on  $\lambda_2(B) - \lambda_2(\Omega)$ , see Lemma 5.6.39. Then we prove Lemmas 5.6.40 and 5.6.52, which give us suitable estimates on  $R_1(\varepsilon)$  and  $R_2(\varepsilon)$ , finally providing (5.6.25).

Let us define the family of domains  $\{\Omega_{\varepsilon}\}_{\varepsilon>0}$  as follows:

$$\Omega_{\varepsilon} := \left\{ x \in \mathbb{R}^N : x = 0 \text{ or } |x| < 1 + \varepsilon \psi \left( \frac{x}{|x|} \right) \right\}, \qquad (5.6.26)$$

where  $\psi$  is a function belonging to the following class

$$\mathcal{P} := \left\{ \psi \in C^{\infty}(\partial B) : \int_{\partial B} \psi d\sigma = \int_{\partial B} (a \cdot x) \psi d\sigma = \int_{\partial B} (a \cdot x)^2 \psi d\sigma = 0 \right\},$$
(5.6.27)



Figure 5.2: Domains  $\Omega_{\varepsilon}$  defined by (5.6.26) with  $\psi \in \mathcal{P}$ .

for all  $a \in \mathbb{R}^N$ . (Note that the class  $\mathcal{P}$  is non-empty, in fact all spherical harmonics  $H_l$  of degree  $l \geq 3$  belong to this class, see Figure 5.2).

Under this choice of  $\Omega_{\varepsilon}$ , the existence of constants  $r_1, r_2, c_1, \ldots, c_4$  satisfying inequalities (5.6.23) and (5.6.24) follow immediately from [16, Lemma 6.2]. Thus, we need only to prove (5.6.25).

Let  $\lambda_2(\Omega_{\varepsilon})$  be the first positive eigenvalue of the Neumann problem (5.6.1) on  $\Omega_{\varepsilon}$ , and let  $u_{\varepsilon}$  be an associated eigenfunction normalized by  $||u_{\varepsilon}||_{L^2(\Omega_{\varepsilon})} = 1$ , so that

$$\int_{\Omega_{\varepsilon}} |D^2 u_{\varepsilon}|^2 + \tau |D u_{\varepsilon}|^2 \, dx = \lambda_2(\Omega_{\varepsilon}).$$
(5.6.28)

By standard elliptic regularity (see e.g., [47, §2.4.3]), since  $\Omega_{\varepsilon}$  is of class  $C^{\infty}$  by construction, we may take a sufficiently small  $\varepsilon_0 > 0$  so that  $u_{\varepsilon} \in C^{\infty}(\overline{\Omega}_{\varepsilon})$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . Moreover, for all  $k \in \mathbb{N}$ , the sets  $\Omega_{\varepsilon}$  are of class  $C^k$  uniformly in  $\varepsilon \in ]0, \varepsilon_0[$ , which means that there exist constants  $H_k > 0$  which do not depend on  $\varepsilon$  such that

$$\|u_{\varepsilon}\|_{C^{k}(\overline{\Omega}_{\varepsilon})} \le H_{k}.$$
(5.6.29)

Now let  $\tilde{u}_{\varepsilon}$  be a  $C^5$  extension of  $u_{\varepsilon}$  to some open neighborhood A of  $B \cup \Omega_{\varepsilon}$ . Then, there exists  $K_A > 0$  which does not depend on  $\varepsilon > 0$  such that

$$\|\tilde{u}_{\varepsilon}\|_{C^{5}(\overline{A})} \le K_{A} \|u_{\varepsilon}\|_{C^{5}(\overline{\Omega}_{\varepsilon})} \le K_{A} H_{5}.$$

$$(5.6.30)$$

From the fact that  $\int_{\Omega_{\varepsilon}} u_{\varepsilon} dx = 0$  and  $|B \setminus \Omega_{\varepsilon}|, |\Omega_{\varepsilon} \setminus B| \in O(\varepsilon)$  as  $\varepsilon \to 0$ , it follows that the quantity  $\delta := \frac{1}{|B|} \int_{B} \tilde{u}_{\varepsilon} dx$  satisfies

$$\delta = \frac{1}{|B|} \int_{B} \tilde{u}_{\varepsilon} \, dx = \frac{1}{|B|} \left( \int_{B \setminus \Omega_{\varepsilon}} \tilde{u}_{\varepsilon} \, dx - \int_{\Omega_{\varepsilon} \setminus B} u_{\varepsilon} \, dx \right) \le c\varepsilon, \qquad (5.6.31)$$

where c > 0 does not depend on  $\varepsilon \in ]0, \varepsilon_0[$ . Now we set

$$v_{\varepsilon} := \tilde{u_{\varepsilon|_B}} - \delta. \tag{5.6.32}$$

The function  $v_{\varepsilon}$  is of class  $C^{5}(\overline{B})$ ,  $\int_{B} v_{\varepsilon} dx = 0$  and satisfies

$$\|v_{\varepsilon}\|_{C^{5}(\overline{B})} \le K_{1} \tag{5.6.33}$$

for some constant  $K_1 > 0$  independent of  $\varepsilon \in ]0, \varepsilon_0[$ . Therefore,  $v_{\varepsilon}$  is a suitable trial function for the Rayleigh quotient of  $\lambda_2(B)$  (see formula (6.1.13)). Thus,

$$\lambda_2(B) \le \frac{\int_B |D^2 v_\varepsilon|^2 + \tau |D v_\varepsilon|^2 \, dx}{\int_B v_\varepsilon^2 \, dx}.$$
(5.6.34)

We now consider the quantity  $\left|\int_B v_{\varepsilon}^2 - \tilde{u}_{\varepsilon}^2 dx\right|$ . We have

$$\left| \int_{B} v_{\varepsilon}^{2} - \tilde{u}_{\varepsilon}^{2} dx \right| = \left| \int_{B} \delta^{2} - 2\delta \tilde{u}_{\varepsilon} dx \right| = \left| \int_{B} \delta(-v_{\varepsilon} - \tilde{u}_{\varepsilon}) dx \right| \qquad (5.6.35)$$
$$= \frac{1}{|B|} \left( \int_{B} \tilde{u}_{\varepsilon} dx \right)^{2} \le K_{2} \varepsilon^{2},$$

where  $K_2 > 0$  is a positive constant independent of  $\varepsilon \in ]0, \varepsilon_0[$ . Moreover, we have that by (5.6.30) and (5.6.33),

$$\left| \int_{B \setminus \Omega_{\varepsilon}} v_{\varepsilon}^{2} - \tilde{u}_{\varepsilon}^{2} dx \right| \leq \int_{B \setminus \Omega_{\varepsilon}} |v_{\varepsilon}^{2} - \tilde{u}_{\varepsilon}^{2}| dx \leq K_{3} \int_{B \setminus \Omega_{\varepsilon}} |v_{\varepsilon} - \tilde{u}_{\varepsilon}| dx \quad (5.6.36)$$
$$= K_{3} \frac{|B \setminus \Omega_{\varepsilon}|}{|B|} \left| \int_{B} \tilde{u}_{\varepsilon} dx \right| \leq k\varepsilon^{2},$$

where  $K_3, k > 0$  are positive constants independent of  $\varepsilon \in ]0, \varepsilon_0[$ . Therefore, from (5.6.28), (5.6.34), (5.6.35), and (5.6.36), it follows that

$$\lambda_{2}(B) \leq \frac{\int_{B\cap\Omega_{\varepsilon}} |D^{2}u_{\varepsilon}|^{2} + \tau |Du_{\varepsilon}|^{2} dx + \int_{B\setminus\Omega_{\varepsilon}} |D^{2}v_{\varepsilon}|^{2} + \tau |Dv_{\varepsilon}|^{2} dx}{\int_{B} \tilde{u}_{\varepsilon}^{2} dx - K_{2}\varepsilon^{2}}$$

$$\leq \frac{\lambda_{2}(\Omega_{\varepsilon}) + \int_{B\setminus\Omega_{\varepsilon}} |D^{2}v_{\varepsilon}|^{2} + \tau |Dv_{\varepsilon}|^{2} dx - \int_{\Omega_{\varepsilon}\setminus B} |D^{2}u_{\varepsilon}|^{2} + \tau |Du_{\varepsilon}|^{2} dx}{1 + \int_{B\setminus\Omega_{\varepsilon}} v_{\varepsilon}^{2} dx - \int_{\Omega_{\varepsilon}\setminus B} u_{\varepsilon}^{2} dx - (k + K_{2})\varepsilon^{2}}.$$
(5.6.37)

We introduce now the two error terms  $R_1(\varepsilon)$  and  $R_2(\varepsilon)$  defined by

$$R_1(\varepsilon) := \int_{B \setminus \Omega_{\varepsilon}} |D^2 v_{\varepsilon}|^2 + \tau |Dv_{\varepsilon}|^2 dx - \int_{\Omega_{\varepsilon} \setminus B} |D^2 u_{\varepsilon}|^2 + \tau |Du_{\varepsilon}|^2 dx$$

and

$$R_2(\varepsilon) := \int_{B \setminus \Omega_{\varepsilon}} v_{\varepsilon}^2 dx - \int_{\Omega_{\varepsilon} \setminus B} u_{\varepsilon}^2 dx.$$

Then inequality (5.6.37) can be rewritten as

$$\lambda_2(B) \le \frac{\lambda_2(\Omega_{\varepsilon}) + R_1(\varepsilon)}{1 + R_2(\varepsilon) - K_4\varepsilon^2},\tag{5.6.38}$$

where  $K_4 = k + K_2$ . From the uniform estimates (5.6.29) and (5.6.33) on  $u_{\varepsilon}$ and  $v_{\varepsilon}$ , it easily follows that  $R_1, R_2 \in O(\varepsilon)$  as  $\varepsilon \to 0$ , which together with (5.6.38) immediately yields  $\lambda_2(B) \leq \lambda_2(\Omega_{\varepsilon}) + C\varepsilon$  for some constant C > 0which does not depend on  $\varepsilon \in ]0, \varepsilon_0[$  (taking  $\varepsilon_0 > 0$  smaller if necessary).

We observe that, due to the close relation of  $R_1(\varepsilon)$  and  $R_2(\varepsilon)$  with the difference  $\lambda_2(B) - \lambda_2(\Omega_{\varepsilon})$ , a better estimate for  $R_1(\varepsilon)$  and  $R_2(\varepsilon)$  provides a better estimate for  $\lambda_2(B) - \lambda_2(\Omega_{\varepsilon})$ . More precisely, we have the following lemma.

**Lemma 5.6.39.** Let  $K_4$  be as in (5.6.38). Let  $\omega : [0,1] \to [0,+\infty[$  be a continuous function such that  $\frac{t^2}{K_4} \leq \omega(t) \leq K_4 t$ . If there exists a constant C > 0 such that  $|R_1(\varepsilon)| \leq C\omega(\varepsilon)$  and  $|R_2(\varepsilon)| \leq C\omega(\varepsilon)$ , then there exists a constant C' > 0 such that

$$\lambda_2(B) \le \lambda_2(\Omega_{\varepsilon}) + C'\omega(\varepsilon)$$

for every sufficient small  $\varepsilon > 0$ .

*Proof.* We refer to [17, Lemma 6.2] for the proof (see also [16, Lemma 6.7]).  $\Box$ 

We also need the following:

**Lemma 5.6.40.** Let  $\omega$  be a function as in Lemma 5.6.39, and let  $v_{\varepsilon}$  be as in (5.6.32). Suppose that there exists C > 0 such that for all  $\varepsilon > 0$  sufficiently small we have  $|R_1(\varepsilon)| \leq C\omega(\varepsilon)$  and  $|R_2(\varepsilon)| \leq C\omega(\varepsilon)$ . Then there exists an eigenfunction  $\xi_{\varepsilon}$  associated with  $\lambda_2(B)$  such that

$$\|v_{\varepsilon} - \xi_{\varepsilon}\|_{C^{3}(\overline{B})} \leq \tilde{C}\sqrt{\omega(\varepsilon)},$$

for some  $\tilde{C} > 0$  which does not depend on  $\varepsilon > 0$ .

*Proof.* Let  $\{\xi_n\}_{n\geq 1}$  be an orthonormal basis of  $L^2(B)$  consisting of eigenfunctions of the Neumann biharmonic problem (5.6.1) for  $\Omega = B$ , with  $\xi_1$  constant. Note then that by our normalization, we have

$$\int_{B} |D^{2}\xi_{n}|^{2} + \tau |D\xi_{n}|^{2} dx = \lambda_{n}(B) \quad \forall n \in \mathbb{N}.$$

We may now write  $v_{\varepsilon} = \sum_{n=1}^{+\infty} a_n(\varepsilon)\xi_n$ . Note that  $a_1(\varepsilon) \equiv 0$  since  $v_{\varepsilon}$  has zero integral mean over B and  $\xi_1$  is a constant. We have

$$\sum_{n=2}^{+\infty} a_n(\varepsilon)^2 - 1 = \|v_\varepsilon\|_{L^2(B)}^2 - 1 = \int_B v_\varepsilon^2 dx - \int_{\Omega_\varepsilon} u_\varepsilon^2 dx$$
$$= \int_B (v_\varepsilon^2 - \tilde{u}_\varepsilon^2) dx - \int_{B \setminus \Omega_\varepsilon} (v_\varepsilon^2 - \tilde{u}_\varepsilon^2) dx + R_2(\varepsilon),$$

and then by using (5.6.35), (5.6.36) we obtain

$$\left|\sum_{n=2}^{+\infty} a_n(\varepsilon)^2 - 1\right| \le K_4 \varepsilon^2 + C\omega(\varepsilon) \le C_1 \omega(\varepsilon).$$
 (5.6.41)

We may now write

$$\begin{split} \lambda_2(\Omega_{\varepsilon}) &= \int_{\Omega_{\varepsilon}} |D^2 u_{\varepsilon}|^2 + \tau |D u_{\varepsilon}|^2 dx \\ &= \int_B |D^2 v_{\varepsilon}|^2 + \tau |D v_{\varepsilon}|^2 dx + \int_{\Omega_{\varepsilon} \setminus B} |D^2 u_{\varepsilon}|^2 + \tau |D u_{\varepsilon}|^2 dx \\ &- \int_{B \setminus \Omega_{\varepsilon}} |D^2 v_{\varepsilon}|^2 + \tau |D v_{\varepsilon}|^2 dx \\ &= \sum_{n=2}^{+\infty} a_n(\varepsilon)^2 \lambda_n(B) - R_1(\varepsilon). \end{split}$$

From Lemma 5.6.39 it follows that

$$|\lambda_2(B) - \lambda_2(\Omega_{\varepsilon})| \le C'\omega(\varepsilon),$$

therefore

$$\left|\sum_{n=2}^{+\infty} a_n(\varepsilon)^2 \lambda_n(B) - \lambda_2(B)\right| = \left|\lambda_2(\Omega_{\varepsilon}) + R_1(\varepsilon) - \lambda_2(B)\right| \le C_2 \omega(\varepsilon).$$
 (5.6.42)

By the symmetry of the ball, the first nonzero eigenvalue  $\lambda_2(B)$  has multiplicity N, and so  $\lambda_2(B) = \lambda_3(B) = \cdots = \lambda_{N+1}(B) < \lambda_{N+2}(B)$ . Therefore

$$C_{2}\omega(\varepsilon) \geq \left|\sum_{n=2}^{N+1} a_{n}(\varepsilon)^{2}\lambda_{2}(B) + \sum_{n=N+2}^{+\infty} a_{n}(\varepsilon)^{2}\lambda_{n}(B) - \lambda_{2}(B)\right|$$
$$= \left|\lambda_{2}(B)\left(\sum_{n=2}^{+\infty} a_{n}(\varepsilon)^{2} - 1\right) + \sum_{n=N+2}^{+\infty} a_{n}(\varepsilon)^{2}\left(\lambda_{n}(B) - \lambda_{2}(B)\right)\right|$$
$$\geq \left(\lambda_{N+2}(B) - \lambda_{2}(B)\right)\sum_{n=N+2}^{+\infty} a_{n}(\varepsilon)^{2} - \lambda_{2}(B)C_{1}\omega(\varepsilon),$$

which yields

$$\sum_{n=N+2}^{+\infty} a_n(\varepsilon)^2 \le C_3 \omega(\varepsilon), \qquad (5.6.43)$$

hence by (5.6.41)

$$\left|\sum_{n=2}^{N+1} a_n(\varepsilon)^2 - 1\right| \le C_4 \omega(\varepsilon).$$
(5.6.44)

Revisiting (5.6.42), we see

$$C_{2}\omega(\varepsilon) \geq \left| \sum_{n=2}^{N+1} a_{n}(\varepsilon)^{2}\lambda_{2}(B) + \sum_{n=N+2}^{+\infty} a_{n}(\varepsilon)^{2}\lambda_{n}(B) - \lambda_{2}(B) \right|$$
$$= \left| \lambda_{2}(B) \left( \sum_{n=2}^{N+1} a_{n}(\varepsilon)^{2} - 1 \right) + \sum_{n=N+2}^{+\infty} a_{n}(\varepsilon)^{2}\lambda_{n}(B) \right|$$
$$\geq \lambda_{2}(B) \left( \sum_{n=2}^{N+1} a_{n}(\varepsilon)^{2} - 1 \right) + \sum_{n=N+2}^{+\infty} a_{n}(\varepsilon)^{2}\lambda_{n}(B),$$

which, together with (5.6.43) and (5.6.44) yields

$$\sum_{n=N+2}^{+\infty} a_n(\varepsilon)^2 \lambda_n(B) \le C_2 \omega(\varepsilon) - \lambda_2(B) \left(\sum_{n=2}^{N+1} a_n(\varepsilon)^2 - 1\right) \le C_5 \omega(\varepsilon).$$
(5.6.45)

Now set  $\varphi := \sum_{n=2}^{N+1} a_n(\varepsilon) \xi_n = v_{\varepsilon} - \sum_{n=N+2}^{\infty} a_n(\varepsilon) \xi_n$  and define the norm  $\|\cdot\|_{H^2_{\tau}(B)}$  by (5.6.45)

$$\|h\|_{H^2_{\tau}(B)}^2 := \int_B |D^2h|^2 + \tau |Dh|^2 + h^2 \, dx, \qquad h \in H^2(B).$$

This norm is equivalent to the standard  $H^2(B)$  norm by coercivity of the bilinear form.

We now estimate the quantity  $||v_{\varepsilon} - \varphi||_{H^2_{\tau}(B)}$ . We have

$$\|v_{\varepsilon} - \varphi\|^{2}_{H^{2}_{\tau}(B)} = \int_{B} |D^{2}(v_{\varepsilon} - \varphi)|^{2} + \tau |D(v_{\varepsilon} - \varphi)|^{2} + (v_{\varepsilon} - \varphi)^{2} dx$$
$$= \int_{B} \sum_{n=N+2}^{+\infty} a_{n}(\varepsilon)^{2} (|D^{2}\xi_{n}|^{2} + \tau |D\xi_{n}|^{2} + \xi_{n}^{2}) dx$$
$$= \sum_{n=N+2}^{+\infty} a_{n}(\varepsilon)^{2} (1 + \lambda_{n}(B)) \leq C_{6}\omega(\varepsilon), \qquad (5.6.46)$$

where the last inequality follows from (5.6.43) and (5.6.45). Thus the function  $v_{\varepsilon}$  is  $\sqrt{\omega(\varepsilon)}$ -close to  $\varphi$  in the  $H^2_{\tau}(B)$  norm.

Now we pass from the bound on the  $H^2_{\tau}(B)$  norm to the bound on the  $C^3(\overline{B})$  norm. To do so, we use standard elliptic regularity estimates for the biharmonic operator. We have that, in  $B \cap \Omega_{\varepsilon}$ ,

$$\Delta^2 v_{\varepsilon} - \tau \Delta v_{\varepsilon} = \Delta^2 u_{\varepsilon} - \tau \Delta u_{\varepsilon} = \lambda_2(\Omega_{\varepsilon}) u_{\varepsilon} = \lambda_2(\Omega_{\varepsilon}) (v_{\varepsilon} + \delta).$$

Recall that  $\delta \in O(\varepsilon)$  as  $\varepsilon \to 0$  by (5.6.31). We set

$$f_{\varepsilon} := \Delta^2 v_{\varepsilon} - \tau \Delta v_{\varepsilon}. \tag{5.6.47}$$

182

Note that in particular,  $f_{\varepsilon} = \lambda_2(\Omega_{\varepsilon})(v_{\varepsilon} + \delta)$  on  $B \cap \Omega_{\varepsilon}$ . Then defining the functions  $g_{\varepsilon}^{(1)}$  and  $g_{\varepsilon}^{(2)}$  on  $\partial B$  by  $g_{\varepsilon}^{(1)} := \frac{\partial^2 v_{\varepsilon}}{\partial \nu^2}$  and  $g_{\varepsilon}^{(2)} := \tau \frac{\partial v_{\varepsilon}}{\partial \nu} - \operatorname{div}_{\partial B}(D^2 v_{\varepsilon} \cdot \nu) - \frac{\partial \Delta v_{\varepsilon}}{\partial \nu}$ , we see that the function  $v_{\varepsilon}$  uniquely solves the following problem:

$$\begin{cases} \Delta^2 u - \tau \Delta u = f_{\varepsilon}, & \text{in } B, \\ \frac{\partial^2 u}{\partial \nu^2} = g_{\varepsilon}^{(1)}, & \text{on } \partial B, \\ \tau \frac{\partial u}{\partial \nu} - \operatorname{div}_{\partial B} (D^2 u \cdot \nu) - \frac{\partial \Delta u}{\partial \nu} = g_{\varepsilon}^{(2)}, & \text{on } \partial B, \\ \int_B u dx = 0. \end{cases}$$

Now let  $f := \lambda_2(B)\varphi$ . Then by definition the function  $\varphi$  is the unique solution of

$$\begin{cases} \Delta^2 u - \tau \Delta u = f, & \text{in } B, \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial B, \\ \tau \frac{\partial u}{\partial \nu} - \operatorname{div}_{\partial B} (D^2 u \cdot \nu) - \frac{\partial \Delta u}{\partial \nu} = 0, & \text{on } \partial B, \\ \int_B u \, dx = 0. \end{cases}$$

Finally, define the function  $w := v_{\varepsilon} - \varphi = \sum_{n=N+2}^{\infty} a_n(\varepsilon)\xi_n$ , which is the unique solution of

$$\begin{cases} \Delta^2 w - \tau \Delta w = f_{\varepsilon} - f, & \text{in } B, \\ \frac{\partial^2 w}{\partial \nu^2} = g_{\varepsilon}^{(1)}, & \text{on } \partial B, \\ \tau \frac{\partial w}{\partial \nu} - \operatorname{div}_{\partial B} (D^2 w \cdot \nu) - \frac{\partial \Delta w}{\partial \nu} = g_{\varepsilon}^{(2)}, & \text{on } \partial B, \\ \int_B w \, dx = 0. \end{cases}$$

For every p > N we have (see e.g., [47, Theorem 2.20])

$$\|w\|_{W^{4,p}(B)} \le C\left(\|f_{\varepsilon} - f\|_{L^{p}(B)} + \|g_{\varepsilon}^{(1)}\|_{W^{2-\frac{1}{p},p}(\partial B)} + \|g_{\varepsilon}^{(2)}\|_{W^{1-\frac{1}{p},p}(\partial B)}\right).$$
(5.6.48)

We consider separately the three summands in the right-hand side of (5.6.48). We start from the first summand. Recall that for any  $x \in B \cap \Omega_{\varepsilon}$  we have (see (5.6.47))  $f_{\varepsilon}(x) = \lambda_2(\Omega_{\varepsilon})(v_{\varepsilon}(x) + \delta)$ . Moreover, for any  $x \in B$  there exists  $\tilde{x} \in B \cap \Omega_{\varepsilon}$  such that  $|x - \tilde{x}| \leq C\varepsilon$ . Therefore, from the regularity of  $v_{\varepsilon}$  (see (5.6.33)), for  $x \in B$  we have, as  $\varepsilon \to 0$ ,

$$f_{\varepsilon}(x) = \Delta^2 v_{\varepsilon}(x) - \tau \Delta v_{\varepsilon}(x) = \Delta^2 v_{\varepsilon}(\tilde{x}) - \tau \Delta v_{\varepsilon}(\tilde{x}) + O(\varepsilon)$$
  
=  $\lambda_2(\Omega_{\varepsilon})(v_{\varepsilon}(\tilde{x}) + \delta) + O(\varepsilon) = \lambda_2(\Omega_{\varepsilon})v_{\varepsilon}(x) + O(\varepsilon).$ 

Thus, as  $\varepsilon \to 0$ , for all p > N, we have by Lemma 5.6.39 and by (5.6.46)

$$\|f_{\varepsilon} - f\|_{L^{p}(B)} = \|\lambda_{2}(\Omega_{\varepsilon})v_{\varepsilon} - \lambda_{2}(B)\varphi\|_{L^{p}(B)} + O(\varepsilon)$$

$$\leq |\lambda_{2}(\Omega_{\varepsilon}) - \lambda_{2}(B)|\|v_{\varepsilon}\|_{L^{p}(B)} + |\lambda_{2}(B)|\|v_{\varepsilon} - \varphi\|_{L^{p}(B)} + O(\varepsilon)$$

$$\leq C_{7}\omega(\varepsilon) + C_{8}\sqrt{\omega(\varepsilon)} + O(\varepsilon) \leq C_{9}\sqrt{\omega(\varepsilon)}.$$
(5.6.49)

Now we consider the second summand in the right-hand side of (5.6.48). From the fact that  $g_{\varepsilon}^{(1)} = \frac{\partial^2 v_{\varepsilon}}{\partial \nu^2}$ , the fact that  $v_{\varepsilon}$  is an extension of  $u_{\varepsilon}$ , the regularity of both  $u_{\varepsilon}$  and  $v_{\varepsilon}$  (5.6.29), (5.6.33) and from the fact that  $\frac{\partial^2 u_{\varepsilon}}{\partial \nu^2} = 0$  on  $\partial \Omega_{\varepsilon}$ , we may conclude

$$\|g_{\varepsilon}^{(1)}\|_{W^{2-\frac{1}{p},p}(\partial B)} \le C\varepsilon.$$
(5.6.50)

For the same reason, for the third summand in the right-hand side of (5.6.48) we have

$$\|g_{\varepsilon}^2\|_{W^{2-\frac{1}{p},p}(\partial B)} \le C\varepsilon.$$
(5.6.51)

From (5.6.48) and the bounds (5.6.49), (5.6.50), and (5.6.51) on each summand, it follows that for all p > N,

$$\|v_{\varepsilon} - \varphi\|_{W^{4,p}(B)} \le C_{10}\sqrt{\omega(\varepsilon)},$$

and thus, from the Sobolev embedding theorem,

$$\|v_{\varepsilon} - \varphi\|_{C^3(\overline{B})} \le \tilde{C}\sqrt{\omega(\varepsilon)}.$$

This concludes the proof of the lemma.

The next lemma gives us refined bounds on  $|R_1(\varepsilon)|$  and  $|R_2(\varepsilon)|$ .

**Lemma 5.6.52.** Let  $\omega(t), v_{\varepsilon}$  be as in Lemma 5.6.39. Suppose that for all  $\varepsilon > 0$  small enough there exists an eigenfunction  $\xi_{\varepsilon}$  associated with  $\lambda_2(B)$  such that

$$\|v_{\varepsilon} - \xi_{\varepsilon}\|_{C^3(\overline{B})} \le C\sqrt{\omega(\varepsilon)},\tag{5.6.53}$$

for some C > 0 which does not depend on  $\varepsilon > 0$ . Then there exists  $\tilde{C} > 0$ which does not depend on  $\varepsilon$  such that  $|R_1(\varepsilon)|, |R_2(\varepsilon)| \leq \tilde{C}\varepsilon \sqrt{\omega(\varepsilon)}$ .

Proof. It is convenient to use spherical coordinates  $(r, \theta) \in \mathbb{R}_+ \times \partial B$  in  $\mathbb{R}^N$ and the corresponding change of variables  $x = \phi(r, \theta)$ . We denote by  $\mathcal{D}$  and  $\tilde{\mathcal{D}}$  the sets  $\mathcal{D} := \partial(\Omega_{\varepsilon} \setminus B) \cap \partial B$  and  $\tilde{\mathcal{D}} = \partial(B \setminus \Omega_{\varepsilon}) \cap \partial B$ . Observe  $\psi \ge 0$ on  $\mathcal{D}$  and  $\psi \le 0$  on  $\tilde{\mathcal{D}}$ .

Thanks to the regularity of  $u_{\varepsilon}$  and  $\tilde{u}_{\varepsilon}$  (see (5.6.30)), on  $\Omega_{\varepsilon} \setminus B$  we have

$$D^{2}u_{\varepsilon} \circ \phi(1 + \varepsilon\psi, \theta) = D^{2}u_{\varepsilon} \circ \phi(1, \theta) + O(\varepsilon),$$
  

$$Du_{\varepsilon} \circ \phi(1 + \varepsilon\psi, \theta) = Du_{\varepsilon} \circ \phi(1, \theta) + O(\varepsilon),$$

as  $\varepsilon \to 0$ . Therefore, integrating with respect to the radius r and applying our definition of  $v_{\varepsilon}$  (5.6.32), we see

$$\int_{\Omega_{\varepsilon}\setminus B} |D^{2}u_{\varepsilon}|^{2} + \tau |Du_{\varepsilon}|^{2} dx = \varepsilon \int_{\mathcal{D}} \left( |D^{2}u_{\varepsilon}|^{2} + \tau |Du_{\varepsilon}|^{2} \right) \psi d\sigma + O(\varepsilon^{2})$$
$$= \varepsilon \int_{\mathcal{D}} \left( |D^{2}v_{\varepsilon}|^{2} + \tau |Dv_{\varepsilon}|^{2} \right) \psi d\sigma + O(\varepsilon^{2}),$$

as  $\varepsilon \to 0$ . Similarly,

$$\int_{B\setminus\Omega_{\varepsilon}} |D^2 v_{\varepsilon}|^2 + \tau |Dv_{\varepsilon}|^2 dx = -\varepsilon \int_{\tilde{\mathcal{D}}} \left( \left| D^2 v_{\varepsilon} \right|^2 + \tau |Dv_{\varepsilon}|^2 \right) \psi d\sigma + O(\varepsilon^2),$$

as  $\varepsilon \to 0$ . From these and from hypothesis (5.6.53), we see

$$\begin{aligned} |R_{1}(\varepsilon)| &\leq \varepsilon \left| \int_{\partial B} \psi \left( \left| D^{2} v_{\varepsilon} \right|^{2} + \tau \left| D v_{\varepsilon} \right|^{2} \right) d\sigma \right| + O(\varepsilon^{2}) \\ &\leq \varepsilon \left| \int_{\partial B} \psi \left( \left| D^{2} \xi_{\varepsilon} \right|^{2} + \tau \left| D \xi_{\varepsilon} \right|^{2} \right) d\sigma \right| + C \varepsilon \sqrt{\omega(\varepsilon)} + O(\varepsilon^{2}) \\ &\leq \tilde{C} \varepsilon \sqrt{\omega(\varepsilon)}, \quad (5.6.54) \end{aligned}$$

as  $\varepsilon \to 0$ , where in the last inequality we have used the following fact of eigenfunctions of  $\lambda_2(B)$ :

$$\left(\left|D^{2}\xi_{\varepsilon}\right|^{2}+\tau\left|D\xi_{\varepsilon}\right|^{2}\right)\Big|_{\partial B}=(a\cdot x)^{2},$$
(5.6.55)

for some  $a \in \mathbb{R}^N$  (cf. (5.6.4)).

By following the very same scheme we can prove the analogue of (5.6.54) for  $R_2(\varepsilon)$ . This concludes the proof of the lemma.

We can now proceed to complete the proof of Theorem 5.6.2.

Let  $\omega_0(\varepsilon) := |R_1(\varepsilon)| + |R_2(\varepsilon)|$ . This function is continuous in  $\varepsilon$  and, moreover, has the property

$$\frac{\varepsilon^2}{K} \le \omega_0(\varepsilon) \le K\varepsilon.$$

The first inequality follows from Theorem 5.6.6, while the second follows from the fact that  $R_1, R_2 \in O(\varepsilon)$ . By Lemma 5.6.40, it follows that there exists an eigenfunction  $\xi_{\varepsilon}$  of the Neumann problem (5.6.1) on *B* associated with eigenvalue  $\lambda_2(B)$  such that

$$\|v_{\varepsilon} - \xi_{\varepsilon}\|_{C^{3}(\overline{B})} \le C\sqrt{\omega_{0}(\varepsilon)}.$$

Now we apply Lemma 5.6.52, obtaining

$$\omega_0(\varepsilon) \le 2\tilde{C}\varepsilon\sqrt{\omega_0(\varepsilon)},$$

and therefore

$$\sqrt{\omega_0(\varepsilon)} = \frac{|R_1(\varepsilon)| + |R_2(\varepsilon)|}{\sqrt{\omega_0(\varepsilon)}} \le 2\tilde{C}\varepsilon.$$

From this it follows that  $\omega_0(\varepsilon) \leq 4\tilde{C}^2\varepsilon^2$ , and hence both  $|R_1(\varepsilon)|, |R_2(\varepsilon)| \leq 4\tilde{C}^2\varepsilon^2$ . Finally, we apply Lemma 5.6.39 and obtain that

$$\lambda_2(B) \le \lambda_2(\Omega_\varepsilon) + \mathcal{C}\varepsilon^2$$

for a constant C > 0 which does not depend on  $\varepsilon \in ]0, \varepsilon_0[$ . This concludes the proof of Theorem 5.6.22.



Figure 5.3: The domain  $\Omega_{\varepsilon}$  introduced in [17].

**Remark 5.6.56.** In [17], the authors provided an explicit construction of a family  $\{\Omega_{\varepsilon}\}_{\varepsilon}$  in  $\mathbb{R}^2$  suitable to show the sharpness of their inequality (see Figure 5.3). This family turns out to be suitable to show also the sharpness of inequalities (5.5.28) and (5.6.7). On the other hand, in [16] the authors gave only sufficient conditions to generate a suitable family  $\{\Omega_{\varepsilon}\}_{\varepsilon}$ , which are exactly those we apply in (5.6.27). We observe that the first two conditions, namely

$$\int_{\partial B} \psi d\sigma = \int_{\partial B} a \cdot x \psi d\sigma = 0, \qquad (5.6.57)$$

have a purely geometrical meaning, and are used to prove inequalities (5.6.23) and (5.6.24) (cf. [16, Lemma 6.2]). In particular, the first one says that the measure of  $\Omega_{\varepsilon}$  is the same as the measure of B up to an error of order  $\varepsilon^2$ . The second says that the baricenter of  $\Omega_{\varepsilon}$  is the origin up to an error of order  $\varepsilon^2$ . This implies in particular that  $\mathcal{A}(\Omega_{\varepsilon})$  is of order  $\varepsilon$ . The third condition has instead a stricter relation with the problem, since any function  $\xi$ belonging to the eigenspace associated with  $\lambda_2(B)$  satisfies equality (5.6.55), which is crucial in the proof of (5.6.25). This is due to the fact that it can be factorized as a radial part times a spherical harmonic polynomial of degree 1. This also tells us that the correct conditions to impose are still (5.6.27) when considering the Steklov problem. In particular, as pointed out in [16, Remark 6.9], ellipsoids satisfy conditions (5.6.57), and hence inequalities (5.6.23) and (5.6.24) hold, but miss the final condition, and therefore are not a suitable family for this problem.

**Remark 5.6.58.** We note that  $\lambda_2(B)$  has multiplicity N. Thus  $\lambda_2(\Omega)$  is not differentiable at  $\Omega = B$ . This implies that along some directions  $\lambda_2(\Omega)$  could have a non-trivial super-differential. This is exactly what happens when we consider the eigenvalues of nearly spherical ellipsoids (see Figure 5.4a). In order to show that this does not happen for every direction, we build a



family  $\Omega_{\varepsilon}$  apporaching the ball such that the multiplicity of the eigenvalue is preserved in a neighborhood of the ball. This is sufficient to ensure that along such direction the eigenvalue is differentiable and therefore it converges with the sharp exponent 2 (see Figure 5.4b). As already discussed, the conditions given in (5.6.27) are sufficient to achieve the exponent 2 for the Fraenkel asymmetry. The family of domains  $\Omega_{\varepsilon}$  introduced in [17] (see Figure 5.3) does not satisfy the second condition in (5.6.27) for all  $a \in \mathbb{R}^N$ , but the eigenvalues converge with the sharp exponent 2. We also note that along this direction the eigenvalue  $\lambda_2(\Omega)$  is not differentiable (see Figure 5.4c).

#### 5.6.3 Sharpness of the Steklov inequality

In this subsection we prove the following result, that tells us that inequality (5.5.28) is sharp. Due to the strong similarities between the two problems, we shall maintain the same notation as in the previous subsection.

**Theorem 5.6.59.** Let B be the unit ball in  $\mathbb{R}^N$  centered at zero. There exist a family  $\{\Omega_{\varepsilon}\}_{\varepsilon>0}$  of smooth domains and positive constants  $c_1, c_2, c_3, c_4$  and  $r_1, r_2, r_3, r_4$ , which do not depend on  $\varepsilon > 0$  such that (5.6.23), (5.6.24) and (5.6.25) hold for all  $\varepsilon \in ]0, \varepsilon_0[$ , where  $\lambda_2(\Omega)$  denotes the first positive eigenvalue of problem (5.0.1) with  $\rho \equiv 1$ .

To prove this theorem, we begin by defining the family  $\{\Omega_{\varepsilon}\}_{\varepsilon>0}$  as in (5.6.26). Thus it remains only to prove (5.6.25).

We recall the variational characterization of the first positive eigenvalue of the Steklov problem (5.0.1) with  $\rho \equiv 1$  on a domain  $\Omega$ :

$$\lambda_2(\Omega) = \inf_{\substack{0 \neq u \in H^2(\Omega) \\ \int_{\partial \Omega} u \, d\sigma = 0}} \frac{\int_{\Omega} |D^2 u|^2 + \tau |Du|^2 \, dx}{\int_{\partial \Omega} u^2 \, d\sigma}.$$
 (5.6.60)

Let  $\lambda_2(\Omega_{\varepsilon})$  be the first positive eigenvalue of the Steklov problem (5.0.1) on  $\Omega_{\varepsilon}$ , and let  $u_{\varepsilon}$  be an associated eigenfunction, normalized by

$$\int_{\partial\Omega_{\varepsilon}} u_{\varepsilon}^2 dx = 1.$$

Then by (5.6.60),

$$\int_{\Omega_{\varepsilon}} |D^2 u_{\varepsilon}|^2 + \tau |\nabla u_{\varepsilon}|^2 dx = \lambda_2(\Omega_{\varepsilon}).$$

By standard elliptic regularity (see e.g., [47, §2.4.3]), since  $\Omega_{\varepsilon}$  is of class  $C^{\infty}$ by construction, we have that  $u_{\varepsilon} \in C^{\infty}(\overline{\Omega_{\varepsilon}})$  for all  $\varepsilon \in ]0, \varepsilon_0[$ . Moreover, for all  $k \in \mathbb{N}$ , the sets  $\Omega_{\varepsilon}$  are of class  $C^k$  uniformly in  $\varepsilon \in ]0, \varepsilon_0[$ , which means that there exists a constant  $H_k > 0$  independent of  $\varepsilon$  such that

$$\|u_{\varepsilon}\|_{C^k(\overline{\Omega_{\varepsilon}})} \leq H_k.$$

Let now  $\tilde{u}_{\varepsilon}$  be a  $C^5$  extension of  $u_{\varepsilon}$  to an open neighborhood A of  $B \cup \Omega_{\varepsilon}$ . Then, there exists  $K_A > 0$  which does not depend on  $\varepsilon > 0$  such that

$$\|\tilde{u}_{\varepsilon}\|_{C^{5}(\overline{A})} \leq K_{A} \|u_{\varepsilon}\|_{C^{5}(\overline{\Omega_{\varepsilon}})} \leq K_{A} H_{5}$$

As in the Neumann case, take  $\delta := \frac{1}{|\partial B|} \int_{\partial B} \tilde{u}_{\varepsilon} d\sigma$  to be the mean of  $\tilde{u}_{\varepsilon}$  over  $\partial B$ . From the fact that  $\int_{\partial \Omega_{\varepsilon}} u_{\varepsilon} dx = 0$  and  $|B \setminus \Omega_{\varepsilon}|, |\Omega_{\varepsilon} \setminus B| \in O(\varepsilon)$  as  $\varepsilon \to 0$ , it follows that as as  $\varepsilon \to 0$ ,

$$\delta = \frac{1}{|\partial B|} \int_{\partial B} \tilde{u}_{\varepsilon} \, d\sigma \in O(\varepsilon).$$

(See also [16, formula (6.15)]). Now let us set

$$v_{\varepsilon} := \tilde{u_{\varepsilon|_B}} - \delta \tag{5.6.61}$$

This function is of class  $C^5(\overline{B})$  and satisfies  $\int_{\partial B} v_{\varepsilon} d\sigma = 0$  and

$$\|v_{\varepsilon}\|_{C^{5}(B)} \le K_{1}$$

for a constant  $K_1 > 0$  independent of  $\varepsilon \in ]0, \varepsilon_0[$ . Therefore,  $v_{\varepsilon}$  is a suitable trial function for the Rayleigh quotient of  $\lambda_2(B)$ , and so

$$\lambda_2(B) \le \frac{\int_B |D^2 v_\varepsilon|^2 + \tau |\nabla v_\varepsilon|^2 dx}{\int_{\partial B} v_\varepsilon^2 d\sigma}.$$

On the other hand,

$$\left|\int_{\partial B} v_{\varepsilon}^2 - \tilde{u}_{\varepsilon}^2 \, d\sigma\right| = \left|\int_{\partial B} \delta^2 - 2\delta \tilde{u}_{\varepsilon} \, d\sigma\right| \le K_2 \varepsilon^2,$$

where  $K_2 > 0$  is a positive constant independent of  $\varepsilon \in ]0, \varepsilon_0[$ . Therefore, we may write

$$\lambda_2(B) \le \frac{\lambda_2(\Omega_{\varepsilon}) + R_1(\varepsilon)}{1 + R_2(\varepsilon) - K_3 \varepsilon^2},$$

188
where we have once again defined the error terms

$$R_1(\varepsilon) := \int_{B \setminus \Omega_{\varepsilon}} |D^2 v_{\varepsilon}|^2 + \tau |\nabla v_{\varepsilon}|^2 dx - \int_{\Omega_{\varepsilon} \setminus B} |D^2 u_{\varepsilon}|^2 + \tau |\nabla u_{\varepsilon}|^2 dx,$$

and

$$R_2(\varepsilon) := \int_{\partial B} v_{\varepsilon}^2 \, d\sigma - \int_{\partial \Omega_{\varepsilon}} u_{\varepsilon}^2 \, d\sigma.$$

At this point, we note that the observations made in Subsection 5.6.2 remain valid here. Therefore, in order to conclude the proof of (5.6.25), we need only the following lemmas, which are the analogues of Lemmas 5.6.39, 5.6.40 and 5.6.52 and which we recall here for the reader's convenience.

**Lemma 5.6.62.** Let  $K_3$  be as in (5.6.3). Let  $\omega : [0,1] \to [0,+\infty[$  be a continuous function such that  $\frac{t^2}{K_3} \leq \omega(t) \leq K_3 t$ . If there exists a constant C > 0 such that  $|R_1(\varepsilon)| \leq C\omega(\varepsilon)$  and  $|R_2(\varepsilon)| \leq C\omega(\varepsilon)$ , then there exists a constant C' > 0 such that

$$\lambda_2(B) \le \lambda_2(\Omega_{\varepsilon}) + C'\omega(\varepsilon)$$

for every sufficient small  $\varepsilon > 0$ .

*Proof.* See [16, Lemma 6.7].

**Lemma 5.6.63.** Let  $\omega$  be a function as in Lemma 5.6.62, and let  $v_{\varepsilon}$  be as in (5.6.61). Suppose that there exists C > 0 such that for all  $\varepsilon > 0$  sufficiently small we have  $|R_1(\varepsilon)| \leq C\omega(\varepsilon)$  and  $|R_2(\varepsilon)| \leq C\omega(\varepsilon)$ . Then there exists an eigenfunction  $\xi_{\varepsilon}$  associated with  $\lambda_2(B)$  such that

$$\|v_{\varepsilon} - \xi_{\varepsilon}\|_{C^{3}(\overline{B})} \leq \tilde{C}\sqrt{\omega(\varepsilon)},$$

for some  $\tilde{C} > 0$  which does not depend on  $\varepsilon > 0$ .

*Proof.* The proof is essentially identical to that of Lemma 5.6.40 and hence the details are omitted.  $\Box$ 

**Lemma 5.6.64.** Let  $\omega(t), v_{\varepsilon}$  be as in Lemma 5.6.62. Suppose that for all  $\varepsilon > 0$  small enough there exists an eigenfunction  $\xi_{\varepsilon}$  associated with  $\lambda_2(B)$  such that

$$\|v_{\varepsilon} - \xi_{\varepsilon}\|_{C^{3}(\overline{B})} \le C\sqrt{\omega(\varepsilon)},$$

for some C > 0 which does not depend on  $\varepsilon > 0$ . Then there exists  $\tilde{C} > 0$ which does not depend on  $\varepsilon$  such that  $|R_1(\varepsilon)|, |R_2(\varepsilon)| \leq \tilde{C}\varepsilon\sqrt{\omega(\varepsilon)}$ .

*Proof.* Regarding the bound on  $R_1$ , we refer to the proof of Lemma 5.6.52. For  $R_2$ , we refer to [16, Lemma 6.8], observing that if  $\xi_{\varepsilon}$  is an eigenfunction associated with  $\lambda_2(B)$ , then

$$\operatorname{div}_{\partial B}(D^2\xi_{\varepsilon}\cdot\nu) + \frac{\partial\Delta\xi_{\varepsilon}}{\partial\nu} = 0$$

on  $\partial B$ , and therefore the second boundary condition in (5.0.1) reads  $\tau \frac{\partial \xi_{\varepsilon}}{\partial \nu} = \lambda_2(B)\xi_{\varepsilon}$ .

### Chapter 6

# A few properties of the eigenvalues of Neumann-type problems

In this chapter we collect a number of results concerning the eigenvalues of Neumann-type problems for the Laplace and biharmonic operators. This chapter is divided in two sections where we consider two different problems. In Section 6.1 we discuss the eigenvalue problem for the biharmonic operator subject to Neumann-type boundary conditions, with particular attention to the dependence of the eigenvalues upon variations of the Poisson's ratio. In Section 6.2 we consider the eigenvalue problem for the Laplace and biharmonic operators subject to Neumann boundary conditions on annuli, and describe the behavior of the eigenvalues as the difference of the two radii goes to zero. In particular, we obtain a counterexample to the domain monotonicity of all the Neumann eigenvalues.

### 6.1 Neumann eigenvalues of the biharmonic operator

In this section we study the dependence of the eigenvalues of the biharmonic operator subject to Neumann and Dirichlet boundary conditions upon perturbations of the Poisson's ratio  $\sigma$ . In particular, we prove that the Neumann eigenvalues are Lipschitz continuous with respect to  $\sigma \in [0, 1]$  and that all the Neumann eigenvalues go to zero as  $\sigma \to 1^-$ . Moreover, we show that the Neumann problem defined by setting  $\sigma = 1$  admits a sequence of positive eigenvalues of finite multiplicity which are not limiting points of Neumann eigenvalues with  $\sigma \in [0, 1]$  as  $\sigma \to 1^-$ , and which coincide with the Dirichlet eigenvalues of the Biharmonic operator.

Let  $\Omega$  be a bounded domain in  $\mathbb{R}^N$  of class  $C^{4,\alpha}$  for some  $\alpha \in ]0,1[$ . Let

 $\sigma \in [0, 1[$ . We consider the following Neumann problem for the biharmonic operator with Poisson's ratio  $\sigma \in [0, 1[$ 

$$\begin{cases} \Delta^2 u = \lambda u, & \text{in } \Omega, \\ (1 - \sigma) \frac{\partial^2 u}{\partial \nu^2} + \sigma \Delta u = 0, & \text{on } \partial \Omega, \\ \frac{\partial \Delta u}{\partial \nu} + (1 - \sigma) \text{div}_{\partial \Omega} \left( D^2 u \cdot \nu \right)_{\partial \Omega} = 0, & \text{on } \partial \Omega, \end{cases}$$
(6.1.1)

in the unknowns u (the eigenfunction) and  $\lambda$  (the eigenvalue). For N = 2this problem is related to the study of the transverse vibrations of a thin plate with a free edge and which occupies at rest a planar region of shape  $\Omega \subset \mathbb{R}^2$ . The coefficient  $\sigma$  represents the Poisson's ratio of the material the plate is made of. We refer e.g., to [31] for more details on the physical interpretation of problem (6.1.1) and on the Poisson's ratio  $\sigma$ . We also mention the paper [39], where the author studies the dependence of the vibrational modes of a plate subject to homogeneous boundary conditions upon the Poisson's ratio  $\sigma \in \left]0, \frac{1}{2}\right[$ , providing also a perturbation formula for the frequencies as functions of the Poisson's coefficient.

We recall that problem (6.1.1) admits an infinite sequence of non-negative eigenvalues of finite multiplicity which depend on  $\sigma \in [0, 1]$  and which we denote here by

$$0 = \lambda_1(\sigma) = \lambda_2(\sigma) = \dots = \lambda_{N+1}(\sigma) < \lambda_{N+2}(\sigma) \le \dots \le \lambda_j(\sigma) \le \dots$$

We note that  $\lambda = 0$  is an eigenvalue of (6.1.1) of multiplicity N + 1, and a set of linearly independent eigenfunctions associated with  $\lambda = 0$  is given by  $\{1, x_1, ..., x_N\}$ .

If we set  $\sigma = 1$ , problem (6.1.1) reads

$$\begin{cases} \Delta^2 u = \lambda u, & \text{in } \Omega, \\ \Delta u = 0, & \text{on } \partial\Omega, \\ \frac{\partial \Delta u}{\partial \nu} = 0, & \text{on } \partial\Omega. \end{cases}$$
(6.1.2)

We note that the differential operator associated with problem (6.1.2) is not a Fredholm operator. Indeed all the harmonic functions in  $\Omega$  are eigenfunctions corresponding to the eigenvalue  $\lambda = 0$ , hence the kernel of the associated operator is infinite dimensional. We also note that the boundary conditions in (6.1.2) do not satisfy the so-called 'complementing conditions' (see [2, §10] and [47] for details), which are necessary conditions for the wellposedness of a differential problem. Nevertheless, problem (6.1.2) admits a countable number of positive eigenvalues of finite multiplicity diverging to  $+\infty$ , which we denote here by

$$0 < \lambda_1 \leq \lambda_2 \leq \cdots \leq \lambda_j \leq \cdots,$$

see Theorem 6.1.11 below.

We show that  $\lambda_j(\sigma) \to 0$  as  $\sigma \to 1^-$  for all  $j \in \mathbb{N}$ . Thus, the positive

eigenvalues of problem (6.1.2) are not limiting points for the eigenvalues of problem (6.1.1) as  $\sigma \to 1^-$ . Moreover, we show that the positive eigenvalues  $\lambda_j$  of problem (6.1.2) coincide with the eigenvalues of the Dirichlet problem for the biharmonic operator, namely problem

$$\begin{cases} \Delta^2 w = \mu w, & \text{in } \Omega, \\ w = 0, & \text{on } \partial \Omega, \\ \frac{\partial w}{\partial \nu} = 0, & \text{on } \partial \Omega. \end{cases}$$
(6.1.3)

We recall that the eigenvalues of (6.1.3) form an increasing sequence of positive eigenvalues of finite multiplicity diverging to  $+\infty$ , which we denote here by

$$0 < \mu_1 \le \mu_2 \le \dots \le \mu_j \le \dots . \tag{6.1.4}$$

#### 6.1.1 Eigenvalues of Neumann and Dirichlet problems

We consider problems (6.1.1), (6.1.2) and (6.1.3) in their weak formulation. The weak formulation of problem (6.1.1) when  $\sigma \in [0, 1]$  is

$$\int_{\Omega} (1-\sigma) D^2 u : D^2 \varphi + \sigma \Delta u \Delta \varphi dx = \lambda \int_{\Omega} u \varphi dx , \qquad (6.1.5)$$

for all  $\varphi \in H^2(\Omega)$ , in the unknowns  $u \in H^2(\Omega)$ ,  $\lambda \in \mathbb{R}$ , where we recall that  $D^2 u : D^2 \varphi = \sum_{i,j=1}^N \frac{\partial^2 u}{\partial x_i \partial x_j} \frac{\partial^2 \varphi}{\partial x_i \partial x_j}$  denotes the Frobenius product. Actually we will recast problem (6.1.5) in  $H^2(\Omega)/N$ , where  $N \subset H^2(\Omega)$  is

Actually we will recast problem (6.1.5) in  $H^2(\Omega)/N$ , where  $N \subset H^2(\Omega)$  is the subspace of  $H^2(\Omega)$  generated by the functions  $\{1, x_1, ..., x_N\}$ . To do so, we set

$$H_N^2(\Omega) := \left\{ u \in H^2(\Omega) : \int_{\Omega} u dx = \int_{\Omega} \frac{\partial u}{\partial x_i} dx = 0, \forall i = 1, ..., N \right\}.$$

In the sequel we will think of the space  $H_N^2(\Omega)$  as endowed with the bilinear form given by the left-hand side of (6.1.5). From the fact that  $|D^2u|^2 \geq \frac{1}{N}(\Delta u)^2$  for all  $u \in H^2(\Omega)$  and from the Poincaré-Wirtinger inequality, it follows that such bilinear form defines on  $H_N^2(\Omega)$  a scalar product whose induced norm is equivalent to the standard one. We denote by  $\pi_N$  the map from  $H^2(\Omega)$  to  $H_N^2(\Omega)$  defined by

$$\pi_{N}[u] := u - \frac{1}{|\Omega|} \int_{\Omega} u + \frac{1}{|\Omega|^{2}} \sum_{i=1}^{N} \left( \int_{\Omega} \frac{\partial u}{\partial x_{i}} dx \right) \left( \int_{\Omega} x_{i} dx \right) - \frac{1}{|\Omega|} \sum_{i=1}^{N} \left( \int_{\Omega} \frac{\partial u}{\partial x_{i}} dx \right) x_{i},$$

for all  $u \in H^2(\Omega)$ . We denote by  $\pi_N^{\sharp}$  the map from  $H^2(\Omega)/N$  onto  $H^2_N(\Omega)$  defined by the equality  $\pi_N = \pi_N^{\sharp} \circ p$ , where p is the canonical projection of

 $H^2(\Omega)$  onto  $H^2(\Omega)/N$ . The map  $\pi_N^{\sharp}$  turns out to be a homeomorphism. Let  $F(\Omega)$  be defined by  $F(\Omega) := \{G \in H^2(\Omega)' : G[1] = G[x_i] = 0, \forall i = 1, ..., N\}$ . Then we consider the operator  $\mathcal{P}_{\sigma}$  as an operator from  $H^2_N(\Omega)$  to  $F(\Omega)$  defined by

$$\mathcal{P}_{\sigma}[u][\varphi] := \int_{\Omega} (1-\sigma) D^2 u : D^2 \varphi + \sigma \Delta u \Delta \varphi dx \,, \quad \forall u \in H^2_N(\Omega), \varphi \in H^2(\Omega).$$

It turns out that  $\mathcal{P}_{\sigma}$  is a homeomorphism of  $H^2_N(\Omega)$  onto  $F(\Omega)$ . We denote by  $\mathcal{J}$  the continuous embedding of  $L^2(\Omega)$  into  $H^2(\Omega)'$  defined by

$$\mathcal{J}[u][arphi] := \int_{\Omega} u arphi dx \,, \quad \forall u \in L^2(\Omega), arphi \in H^2(\Omega).$$

Finally, we define the operator  $T_{\sigma}$  acting on  $H^2(\Omega)/N$  as follows:

$$T_{\sigma} = (\pi_N^{\sharp})^{(-1)} \circ \mathcal{P}_{\sigma}^{(-1)} \circ \mathcal{J} \circ i \circ \pi_N^{\sharp},$$

where *i* denotes the embedding of  $H^2(\Omega)$  into  $L^2(\Omega)$ .

**Lemma 6.1.6.** The pair  $(\lambda, u)$  of the set  $(\mathbb{R} \setminus \{0\}) \times (H_N^2(\Omega) \setminus \{0\})$  satisfies (6.1.5) if and only if  $\lambda > 0$  and the pair  $(\lambda^{-1}, p[u])$  of the set  $\mathbb{R} \times (H^2(\Omega)/N \setminus \{0\})$  satisfies the equation

$$\lambda^{-1}p[u] = T_{\sigma}p[u].$$

We have the following theorem.

**Theorem 6.1.7.** The operator  $T_{\sigma}$  is a non-negative compact self-adjoint operator in  $H^2(\Omega)/N$ , whose eigenvalues coincide with the reciprocals of the positive eigenvalues of problem (6.1.5). In particular, the set of eigenvalues of problem (6.1.5) is contained in  $[0, +\infty[$  and consists of the image of a sequence increasing to  $+\infty$  and each eigenvalue has finite multiplicity. Moreover, the first eigenvalue is  $\lambda = 0$  and has multiplicity N + 1, and a set of linearly independent eigenfunctions corresponding to  $\lambda = 0$  is given by  $\{1, x_1, ..., x_N\}$ .

*Proof.* It is easy to prove that the operator  $T_{\sigma}$  is self-adjoint. The compactness of the operator  $T_{\sigma}$  follows from the compactness of the embedding *i*. The last statement is straightforward.

In an analogous way it is possible to show that the eigenvalues of (6.1.3) are positive and of finite multiplicity. In fact, the weak formulation of problem (6.1.3) reads: find  $(u, \lambda) \in H_0^2(\Omega) \times \mathbb{R}$  such that u solves equation  $\int_{\Omega} \Delta u \Delta \varphi dx = \lambda \int_{\Omega} u \varphi dx$  for all  $\varphi \in H_0^2(\Omega)$ . We note that this this is equivalent to find  $(u, \lambda) \in H_0^2(\Omega) \times \mathbb{R}$  such that equation (6.1.5) holds for all  $\varphi \in H_0^2(\Omega)$ . From the Poincaré inequality it follows that the bilinear form given by the right-hand side of (6.1.5) defines on  $H_0^2(\Omega)$  a scalar product whose induced norm is equivalent to the standard one. Therefore the analogue of Theorem 6.1.7 holds. Thus, the eigenvalues of problem (6.1.3) are positive and can be represented by means of an infinite sequence diverging to  $+\infty$  of the form (6.1.4), and the corresponding eigenfunctions form a orthonormal basis of  $H_0^2(\Omega)$ .

Finally, we show that problem (6.1.2) admits an infinite sequence of positive eigenvalues. We have already observed that all harmonic functions in  $H^2(\Omega)$  are eigenfunctions corresponding to the eigenvalue  $\lambda = 0$ . We start by recalling the following direct decomposition of the space  $H^2(\Omega)$  via the biharmonic operator (see [15, Theorem 4.7] for details):

$$H^{2}(\Omega) = H^{2}_{h}(\Omega) \oplus \Delta(H^{4}(\Omega) \cap H^{2}_{0}(\Omega)), \qquad (6.1.8)$$

where  $H_h^2(\Omega) := \{ u \in H^2(\Omega) : \Delta u = 0 \}$  is the space of harmonic functions in  $H^2(\Omega)$ . The fact that the sum is direct follows since if  $f \in H_h^2(\Omega)$  and  $u \in H^4(\Omega) \cap H_0^2(\Omega)$  with  $\Delta u = f$ , then

$$\int_{\Omega} (\Delta u)^2 dx = \int_{\Omega} \Delta u f dx$$
$$= \int_{\Omega} \Delta f u dx + \int_{\partial \Omega} f \frac{\partial u}{\partial \nu} d\sigma - \int_{\partial \Omega} \frac{\partial f}{\partial \nu} u d\sigma$$
$$= \int_{\Omega} \Delta f u dx = 0,$$

thus  $\Delta u = 0$  and then f = 0.

In order to characterize the positive eigenvalues of problem (6.1.2) and to get rid of the harmonic functions which are the eigenfunctions associated with the eigenvalue  $\lambda = 0$ , we will obtain a problem in  $\Delta(H^4(\Omega) \cap H_0^2(\Omega))$ . Thus, we consider the following weak formulation of problem (6.1.2) for  $\lambda \neq 0$ . (In the case  $\lambda = 0$ , the solution of (6.1.2) are exactly the harmonic functions in  $H^2(\Omega)$ .)

$$\int_{\Omega} \Delta^2 u \Delta^2 \varphi dx = \lambda \int_{\Omega} \Delta u \Delta \varphi, \quad \forall u, \varphi \in H^4(\Omega) \cap H^2_0(\Omega), \tag{6.1.9}$$

in the unknowns  $u \in H^4(\Omega) \cap H^2_0(\Omega)$ ,  $\lambda \in \mathbb{R}$ . In fact, assume we have a solution  $u \in H^4(\Omega)$  to (6.1.2). For any  $\varphi \in H^2(\Omega)$  we have

$$\begin{split} \lambda \int_{\Omega} u\varphi dx &= \int_{\Omega} \Delta^2 u\varphi dx \\ &= \int_{\Omega} \Delta u \Delta \varphi dx + \int_{\partial \Omega} \frac{\partial \Delta u}{\partial \nu} \varphi d\sigma - \int_{\partial \Omega} \Delta u \frac{\partial \varphi}{\partial \nu} d\sigma \\ &= \int_{\Omega} \Delta u \Delta \varphi dx. \end{split}$$

Thus

$$\int_{\Omega} \Delta u \Delta \varphi dx = \lambda \int_{\Omega} u \varphi dx.$$
(6.1.10)

By (6.1.8) we have that

$$\varphi = \varphi_h + \Delta \psi,$$

where  $\varphi_h \in H^2(\Omega)$  is harmonic and  $\psi \in H^4(\Omega) \cap H^2_0(\Omega)$ . Thus

$$\int_{\Omega} \Delta u \left( \Delta \varphi_h + \Delta^2 \psi \right) dx = \lambda \int_{\Omega} u \left( \varphi_h + \Delta \psi \right).$$

Now if  $\lambda \neq 0$ , then by (6.1.10) we have

$$0 = \int_{\Omega} \Delta u \Delta \varphi_h dx = \lambda \int_{\Omega} u \varphi_h,$$

then  $\int_{\Omega} u\varphi_h dx = 0$ . Hence

$$\int_{\Omega} \Delta u \Delta^2 \psi dx = \lambda \int_{\Omega} u \Delta \psi dx.$$

Similarly,  $u = u_h + \Delta v$ , where  $u_h \in H^2(\Omega)$  is harmonic and  $v \in H^4(\Omega) \cap H^2_0(\Omega)$ . Thus

$$\begin{split} \int_{\Omega} \Delta \left( u_h + \Delta v \right) \Delta^2 \psi dx &= \lambda \int_{\Omega} \left( u_h + \Delta v \right) \Delta \psi dx \\ &= \lambda \int_{\Omega} \Delta v \Delta \psi dx + \lambda \int_{\Omega} \Delta u_h \psi dx + \int_{\partial \Omega} u_h \frac{\partial \psi}{\partial \nu} d\sigma - \int_{\partial \Omega} \frac{\partial u_h}{\partial \nu} \psi d\sigma \\ &= \lambda \int_{\Omega} \Delta v \Delta \psi dx. \end{split}$$

We note that there exists a constant C > 0 such that  $\int_{\Omega} \Delta^2 u \Delta^2 \varphi dx \leq C \|u\|_{H^4(\Omega)} \|\varphi\|_{H^4(\Omega)}$  and  $\|u\|_{H^4(\Omega)} \leq C \|\Delta^2 u\|_{L^2(\Omega)}$  for all  $u, \varphi \in H^4(\Omega) \cap H_0^2(\Omega)$  (the second inequality follows from standard elliptic regularity for the Dirichlet problem for the biharmonic operator and from the regularity assumptions of  $\Omega$ , see [47, Theorem 2.20] for details). Therefore, the bilinear form given by the left-hand side of (6.1.9) defines on  $H^4(\Omega) \cap H_0^2(\Omega)$  a scalar product whose induced norm is equivalent to the standard norm of  $H^4(\Omega)$ . Thus, the analogue of Theorem 6.1.7 holds.

**Theorem 6.1.11.** The set of eigenvalues of problem (6.1.2) is contained in  $[0, +\infty[$ . The eigenspace corresponding to the eigenvalue  $\lambda = 0$  has infinite dimension and all harmonic functions in  $H^2(\Omega)$  are eigenfunctions associated with  $\lambda = 0$ . Moreover, the set of positive eigenvalues consists of the image of a sequence increasing to  $+\infty$ . Each positive eigenvalue has finite multiplicity and the corresponding eigenfunctions form a orthonormal basis of  $H^4(\Omega) \cap H^2_0(\Omega)$ .

196

### 6.1.2 Dependence of the Neumann eigenvalues upon the Poisson's ratio

In the first part of this subsection we consider the behavior of the eigenvalues of problem (6.1.1) as  $\sigma \to 1^-$ . In the second part, we show that the positive eigenvalues of problem (6.1.2) and the eigenvalues of problem (6.1.3) coincide. We start by proving the following lemma.

**Lemma 6.1.12.** The function which maps  $\sigma \in [0, 1[$  to  $\lambda_j(\sigma)$  is Lipschitz continuous.

*Proof.* We recall that for each  $\sigma \in [0, 1]$  we have the following formula for  $\lambda_j(\sigma)$ 

$$\lambda_j(\sigma) = \inf_{\substack{E \le H^2(\Omega) \\ \dim E = j}} \sup_{0 \ne u \in E} \frac{\int_{\Omega} (1 - \sigma) |D^2 u|^2 + \sigma(\Delta u)^2 dx}{\int_{\Omega} u^2 dx}.$$
 (6.1.13)

Therefore, for each  $\sigma_1, \sigma_2 \in [0, 1]$  and  $u \in H^2(\Omega)$  we have

$$\left| \frac{\int_{\Omega} (1-\sigma_{1}) |D^{2}u|^{2} + \sigma_{1}(\Delta u)^{2} dx}{\int_{\Omega} u^{2} dx} - \frac{\int_{\Omega} (1-\sigma_{2}) |D^{2}u|^{2} + \sigma_{2}(\Delta u)^{2} dx}{\int_{\Omega} u^{2} dx} \right| \\
\leq |\sigma_{1} - \sigma_{2}| \frac{\int_{\Omega} |D^{2}u|^{2} + (\Delta u)^{2} dx}{\int_{\Omega} u^{2} dx} \\
\leq (1+N) |\sigma_{1} - \sigma_{2}| \frac{\int_{\Omega} |D^{2}u|^{2} dx}{\int_{\Omega} u^{2} dx}, \quad (6.1.14)$$

then taking the infimum and the supremum in (6.1.14) yields

$$|\lambda_j(\sigma_1) - \lambda_j(\sigma_2)| \le (1+N)\lambda_j(0)|\sigma_1 - \sigma_2|.$$

Then the function  $\lambda_i(\sigma)$  is Lipschitz in [0, 1].

**Remark 6.1.15.** From Lemma 6.1.12 it follows that, since 
$$\lambda_j(\sigma)$$
 is Lipschitz on  $[0, 1[$ , then it is uniformly continuous in the whole of  $[0, 1]$ . Hence, for all  $j \in \mathbb{N}$ , there exist  $\overline{\lambda}_j \in \mathbb{R}$  such that  $\lim_{\sigma \to 1^-} \lambda_j(\sigma) = \overline{\lambda}_j$ .

In the next theorem we show that  $\lim_{\sigma \to 1^{-}} \lambda_j(\sigma) = 0$  for all  $j \in \mathbb{N}$ .

**Theorem 6.1.16.** For all  $j \in \mathbb{N}$  it holds  $\lim_{\sigma \to 1^{-}} \lambda_j(\sigma) = 0$ .

*Proof.* We recall that the space  $H_h^2(\Omega)$  is closed in  $H^2(\Omega)$  and therefore it is a Hilbert space, endowed with the standard scalar product of  $H^2(\Omega)$ . Let  $\{u_i\}_{i=1}^{\infty}$  be a set of linearly independent functions in  $H_h^2(\Omega)$  such that  $\int_{\Omega} u_i u_k = \delta_{ik}$  for all  $i, k \in \mathbb{N}$ . Then, from (6.1.13) we have that for all  $j \in \mathbb{N}$ it holds

$$\lambda_j(\sigma) \le \max_{c_1,\dots,c_j \in \mathbb{R}} \frac{(1-\sigma) \int_{\Omega} \left| \sum_{i=1}^j c_i D^2 u_i \right|^2 dx}{\int_{\Omega} \left( \sum_{i=1}^j c_i u_i \right)^2 dx},$$

where we have chosen as *j*-dimensional space E in (6.1.13) the space generated by  $\{u_1, ..., u_j\}$ . Then we have

$$\max_{c_1,\dots,c_j \in \mathbb{R}} \frac{(1-\sigma)\int_{\Omega} \left|\sum_{i=1}^{j} c_i D^2 u_i\right|^2 dx}{\int_{\Omega} \left(\sum_{i=1}^{j} c_i u_i\right)^2 dx}$$
  
$$\leq \max_{c_1,\dots,c_j \in \mathbb{R}} j(1-\sigma) \frac{\sum_{i=1}^{j} c_i^2 \int_{\Omega} |D^2 u_i|^2 dx}{\sum_{i=1}^{j} c_i^2}$$
  
$$\leq j(1-\sigma) \max_{i=1,\dots,j} \int_{\Omega} |D^2 u_i|^2 dx,$$

and therefore  $\lim_{\sigma \to 1^{-}} \lambda_j(\sigma) = 0$  for all  $j \in \mathbb{N}$ .

Thus, the positive eigenvalues of problem (6.1.2) are not limiting points for the eigenvalues of problem (6.1.1) as  $\sigma \to 1^-$ .

Now we consider problems (6.1.2) and (6.1.3). We note that, under the assumptions that  $\Omega$  is of class  $C^{4,\alpha}$ , we have that the eigenfunctions w of problem (6.1.3) are of class  $C^{4,\alpha}(\overline{\Omega})$  (see [47, Theorem 2.20]). We have the following theorem.

**Theorem 6.1.17.** All the positive eigenvalues of problem (6.1.2) coincide with the eigenvalues of problem (6.1.3).

*Proof.* Let  $\mu$  be an eigenvalue of problem (6.1.3) and let  $w \in H^2_0(\Omega)$  be an eigenfunction associated with  $\mu$ . Let  $v_0 \in H^2(\Omega) \cap H^1_0(\Omega)$  be the unique solution of

$$\begin{cases} \Delta v_0 = w, & \text{in } \Omega, \\ v_0 = 0, & \text{on } \partial \Omega \end{cases}$$

We set  $v_h = v_0 + h$  for some harmonic function  $h \in H^2(\Omega)$ . Now we consider the following problem: find a harmonic function h such that

$$\begin{cases} \Delta^2 v_h = \mu v_h, & \text{in } \Omega, \\ \Delta v_h = 0, & \text{on } \partial \Omega, \\ \frac{\partial \Delta v_h}{\partial \nu} = 0, & \text{on } \partial \Omega. \end{cases}$$

Clearly  $\Delta v_{h|_{\partial\Omega}} = \frac{\partial \Delta v_h}{\partial \nu}|_{\partial\Omega} = 0$  for all harmonic functions h. As for the differential equation, we have  $\Delta^2(v_0 + h) = \mu(v_0 + h)$  if and only if  $\Delta(\Delta v_0 + \Delta h) = \mu(v_0 + h)$ , that is  $\Delta w = \mu(v_0 + h)$  and therefore  $h = \frac{\Delta w}{\mu} - v_0$ , which is clearly harmonic and belongs to  $H^2(\Omega)$ . Therefore each eigenvalue  $\mu$  of problem (6.1.3) is an eigenvalue of problem (6.1.2) and a corresponding eigenfunction is given by  $v = \frac{\Delta w}{\mu}$ . On the other hand, suppose that  $\lambda > 0$  is an eigenvalue of problem (6.1.2) and let  $u \in H^4(\Omega) \cap H^2_0(\Omega)$  be a

corresponding eigenfunction. Then, the function  $w = \Delta u$  is in  $H_0^2(\Omega)$  and solves

$$\begin{cases} \Delta^2 w = \lambda w, & \text{in } \Omega, \\ w = 0, & \text{on } \partial\Omega, \\ \frac{\partial w}{\partial \nu} = 0, & \text{on } \partial\Omega, \end{cases}$$

therefore,  $\lambda$  is an eigenvalue of problem (6.1.3) with corresponding eigenfunction  $\Delta v$ .

### 6.1.3 Neumann and Dirichlet eigenvalues in the case of the unit ball

In this section we consider problems (6.1.1), (6.1.2) and (6.1.3) when  $\Omega = B$  is the unit ball in  $\mathbb{R}^N$  centered at zero. In this case it is possible to perform explicit computations which allow to recast the eigenvalue problems (6.1.1), (6.1.2) and (6.1.3) into suitable equations of the form  $F(\lambda) = 0$  and then gather informations on the behavior of the eigenvalues.

It is convenient to use the standard spherical coordinates  $(r, \theta) \in \mathbb{R}_+ \times \partial B$ . Recall that by  $H_l(\theta)$  we denote a spherical harmonic of order  $l \in \mathbb{N}_0$ . We also recall that for each  $l \in \mathbb{N}_0$ ,  $H_l$  is a solution of the equation  $-\Delta_S H_l = l(l + N - 2)H_l$ .

As customary, for  $l \in \mathbb{N}_0$ , we denote by  $j_l$  and  $i_l$  the ultraspherical and modified ultraspherical Bessel functions of the first species and order l respectively.

We consider first problem (6.1.3) on B. We have the following lemma.

**Lemma 6.1.18.** Given an eigenvalue  $\mu$  of problem (6.1.3) on B, a corresponding eigenfunction w is of the form  $w(r,\theta) = W_l(r)H_l(\theta)$ , for some  $l \in \mathbb{N}_0$ , where

$$W_l(r) = \alpha j_l(\sqrt[4]{\mu}r) + \beta i_l(\sqrt[4]{\mu}r), \qquad (6.1.19)$$

for suitable  $\alpha, \beta \in \mathbb{R}$ .

We refer e.g., to [29] for the proof of Lemma 6.1.18. We establish now an implicit characterization of the eigenvalues of problem (6.1.3) on B.

**Lemma 6.1.20.** The eigenvalues  $\mu$  of problem (6.1.3) on B are given implicitly as zeroes of the equation

$$j_l(\sqrt[4]{\mu})i'_l(\sqrt[4]{\mu}) - i_l(\sqrt[4]{\mu})j'_l(\sqrt[4]{\mu}) = 0.$$
(6.1.21)

*Proof.* By Lemma 6.1.18, an eigenfunction w associated with an eigenvalue  $\mu$  is of the form  $w(r, \theta) = W_l(r)H_l(\theta)$ , where  $W_l(r)$  is given by (6.1.19). We recall that in spherical coordinates the Dirichlet boundary conditions are written as

$$w_{|_{r=1}} = \partial_r w_{|_{r=1}} = 0.$$

By imposing such boundary conditions to  $w(r, \theta)$  we obtain the following homogeneous system of two equations in two unknowns  $\alpha$  and  $\beta$ 

$$\begin{cases} \alpha j_l(\sqrt[4]{\mu}) + \beta i_l(\sqrt[4]{\mu}) = 0, \\ \alpha j'_l(\sqrt[4]{\mu}) + \beta i'_l(\sqrt[4]{\mu}) = 0, \end{cases}$$

which has solutions if and only if its determinant vanishes. This yields formula (6.1.21).

Now we consider problem (6.1.1) on B. We have the following lemma (see e.g., [28] for the proof).

**Lemma 6.1.22.** Given an eigenvalue  $\lambda$  of problem (6.1.1) with  $\sigma \in [0, 1]$ on B, a corresponding eigenfunction u is of the form  $u(r, \theta) = U_l(r)H_l(\theta)$ , for some  $l \in \mathbb{N}_0$ , where

$$U_l(r) = \alpha j_l(\sqrt[4]{\lambda}r) + \beta i_l(\sqrt[4]{\lambda}r), \qquad (6.1.23)$$

for suitable  $\alpha, \beta \in \mathbb{R}$ .

We have the following lemma on the eigenvalues of problem (6.1.1) on B.

**Lemma 6.1.24.** The eigenvalues  $\lambda$  of problem (6.1.1) with  $\sigma \in [0, 1]$  on B are given implicitly as zeroes of the equation

$$\det M(\lambda, \sigma) = 0, \tag{6.1.25}$$

where  $M(\lambda, \sigma)$  is the 2 × 2 matrix defined by

$$\begin{bmatrix} \sqrt{\lambda}j_{l}''(\sqrt[4]{\lambda}) & \sqrt{\lambda}i_{l}''(\sqrt[4]{\lambda}) \\ +(N-1)\sqrt[4]{\lambda}\sigma j_{l}'(\sqrt[4]{\lambda}) & +(N-1)\sqrt[4]{\lambda}\sigma i_{l}'(\sqrt[4]{\lambda}) \\ -l(l+N-2)\sigma j_{l}(\sqrt[4]{\lambda}) & -l(l+N-2)\sigma i_{l}(\sqrt[4]{\lambda}) \\ \frac{\sqrt[4]{\lambda}3}{j_{l}'''(\sqrt[4]{\lambda})} +(N-1)\sqrt{\lambda}j_{l}''(\sqrt[4]{\lambda}) & \sqrt[4]{\lambda}i_{l}'''(\sqrt[4]{\lambda}) +(N-1)\sqrt{\lambda}i_{l}''(\sqrt[4]{\lambda}) \\ +\sqrt[4]{\lambda}(1-N+l(\sigma-2)(N+l-2))j_{l}'(\sqrt[4]{\lambda}) & +\sqrt[4]{\lambda}(1-N+l(\sigma-2)(N+l-2))i_{l}'(\sqrt[4]{\lambda}) \\ -l(l+N-2)(\sigma-3)j_{l}(\sqrt[4]{\lambda}) & -l(l+N-2)(\sigma-3)i_{l}(\sqrt[4]{\lambda}) \end{bmatrix}$$
(6.1.26)

*Proof.* By Lemma 6.1.22, an eigenfunction u associated with an eigenvalue  $\lambda$  is of the form  $u(r,\theta) = U_l(r)H_l(\theta)$ , where  $U_l(r)$  is given by (6.1.23). We recall that in spherical coordinates the Neumann boundary conditions are written as

$$\begin{cases} (1-\sigma)\partial_{rr}^2 u + \sigma \Delta u_{|_{r=1}} = 0, \\ \partial_r(\Delta u) + (1-\sigma)\frac{1}{r^2}\Delta_S \left(\partial_r u - \frac{u}{r}\right)_{|_{r=1}} = 0, \end{cases}$$

see [28] for details. By imposing boundary conditions to the function u we obtain a system of two equations in two unknowns  $\alpha$  and  $\beta$ , and the associated matrix is given by (6.1.26). Thus the eigenvalues must solve equation (6.1.25).

We give now an alternative proof of Theorem 6.1.17 when  $\Omega = B$  is the unit ball in  $\mathbb{R}^N$  centered at zero based on the explicit representations of the eigenvalues discussed in this section. We have the following theorem.

**Theorem 6.1.27.** Equations det $M(\lambda, 1) = 0$  and (6.1.21) admit the same non-zero solutions.

*Proof.* We consider (6.1.26) with  $\sigma = 1$ . Let  $\lambda > 0$  be a solution of  $\det M(\lambda, 1) = 0$ . We compute  $F(\lambda) = \det M(\lambda, 1)$ . We have

$$F(\lambda) = -\sqrt[4]{\lambda}l(l-1)(N+l-2)(N+l-1)\left(j_l(\sqrt[4]{\lambda})i_l'(\sqrt[4]{\lambda}) - i_l(\sqrt[4]{\lambda})j_l'(\sqrt[4]{\lambda})\right) + \sqrt{\lambda}l(N+1)(l+N-2)\left(j_l(\sqrt[4]{\lambda})i_l''(\sqrt[4]{\lambda}) - i_l(\sqrt[4]{\lambda})j_l''(\sqrt[4]{\lambda})\right) - \lambda^{3/4}(N(N-1)+l(N+l-2))\left(j_l'(\sqrt[4]{\lambda})i_l'''(\sqrt[4]{\lambda}) - i_l'(\sqrt[4]{\lambda})j_l'''(\sqrt[4]{\lambda})\right) + \lambda^{3/4}l(l+N-2)(j_l(\sqrt[4]{\lambda})i_l'''(\sqrt[4]{\lambda}) - i_l(\sqrt[4]{\lambda})j_l'''(\sqrt[4]{\lambda})) - \lambda(N-1)\left(j_l'(\sqrt[4]{\lambda})i_l'''(\sqrt[4]{\lambda}) - i_l'(\sqrt[4]{\lambda})j_l'''(\sqrt[4]{\lambda})\right) + \lambda^{5/4}\left(j_l''(\sqrt[4]{\lambda})i_l'''(\sqrt[4]{\lambda}) - i_l''(\sqrt[4]{\lambda})j_l'''(\sqrt[4]{\lambda})\right). \quad (6.1.28)$$

We recall the well-known recurrence formulas for Bessel functions and their derivatives (see [1, 9.1.27 and 9.6.26])

$$J_{\nu-1}(z) + J_{\nu+1}(z) = \frac{2\nu}{z} J_{\nu}(z), \quad I_{\nu-1}(z) - I_{\nu+1}(z) = \frac{2\nu}{z} J_{\nu}(z), J_{\nu-1}(z) - J_{\nu+1}(z) = 2J_{\nu}'(z), \quad I_{\nu-1}(z) + I_{\nu+1}(z) = 2I_{\nu}'(z), J_{\nu}'(z) = J_{\nu-1}(z) - \frac{\nu}{z} J_{\nu}(z), \quad I_{\nu}'(z) = I_{\nu-1}(z) - \frac{\nu}{z} I_{\nu}(z), J_{\nu}'(z) = -J_{\nu+1}(z) + \frac{\nu}{z} J_{\nu}(z), \quad I_{\nu}'(z) = I_{\nu+1}(z) + \frac{\nu}{z} I_{\nu}(z).$$

$$(6.1.29)$$

We set  $C_l^{\pm}(z)=I_{\frac{N}{2}+l}(z)J_{\frac{N}{2}-1+l}(z)\pm I_{\frac{N}{2}-1+l}(z)J_{\frac{N}{2}+l}(z).$  We use (6.1.29) to get

$$j_{l}(z)i'_{l}(z) - i_{l}(z)j'_{l}(z) = z^{2-N}C^{+}_{l}(z),$$

$$j_{l}(z)i''_{l}(z) - i_{l}(z)j''_{l}(z) = z^{1-N} \left(2zI_{\frac{N}{2}-1+l}(z)J_{\frac{N}{2}-1+l}(z)\right)$$
(6.1.30)

$$-(N-1)C_{l}^{+}(z)), \qquad (6.1.31)$$

$$j'_{l}(z)i''_{l}(z) - i'_{l}(z)j''_{l}(z) = z^{-N} \left( z^{2}C_{l}^{-}(z) + 2lzI_{\frac{N}{2}-1+l}(z)J_{\frac{N}{2}-1+l}(z) - l(l+N-2)C_{l}^{+}(z) \right),$$

$$(6.1.32)$$

$$j_{l}(z)i_{l}^{\prime\prime\prime}(z) - i_{l}(z)j_{l}^{\prime\prime\prime}(z) = z^{-N} \left( z^{2}C_{l}^{-}(z) + 2(1-N+l)zI_{\frac{N}{2}-1+l}(z)J_{\frac{N}{2}-1+l}(z) + (N(N-1)+l(l+N-2))C_{l}^{+}(z) \right),$$
(6.1.33)

$$j'_{l}(z)i'''_{l}(z) - i'_{l}(z)j'''_{l}(z) = z^{-1-N} \left(-2z^{3}I_{\frac{N}{2}+l}(z)J_{\frac{N}{2}+l}(z) + (1-N+2l)z^{2}C_{l}^{-}(z) + 2l(1-N+l)zI_{\frac{N}{2}-1+l}(z)J_{\frac{N}{2}-1+l}(z) + l(l+N-2)(N+1)C_{l}^{+}(z)\right).$$

$$(6.1.34)$$

$$+i(l+N-2)(N+1)C_{l}(z)),$$

$$(0.1.34)$$

$$j_{l}''(z)i_{l}'''(z) - i_{l}''(z)j_{l}'''(z) = z^{-2-N} \left(-z^{4}C_{l}^{+}(z) + 2(N-1)z^{3}I_{\frac{N}{2}+l}(z)J_{\frac{N}{2}+l}(z) -(N+1)(2l+1)z^{2}C_{l}^{-}(z) -(N+1)(2l+1)zI_{\frac{N}{2}-1+l}(z)J_{\frac{N}{2}-1+l}(z) +l(l-1)(l+N-2)(l+N-1)C_{l}^{+}(z)\right).$$

$$(6.1.35)$$

Thanks to (6.1.30)-(6.1.35), expression (6.1.28) simplifies to

$$F(\lambda) = \lambda^{5/4} \left( j_l(\sqrt[4]{\lambda}) i'_l(\sqrt[4]{\lambda}) - i_l(\sqrt[4]{\lambda}) j'_l(\sqrt[4]{\lambda}) \right).$$
(6.1.36)

Therefore by comparing (6.1.36) with (6.1.21) we see that the non-zero eigenvalues of problem (6.1.2) and the eigenvalues of problem (6.1.3) on the unit ball coincide.

**Remark 6.1.37.** From Theorem 6.1.16 it follows that all the eigenvalues  $\lambda_j(\sigma) \to 0$  as  $\sigma \to 1^-$ . This means that there are infinitely many branches of solutions  $\sigma \mapsto \lambda(\sigma)$  of equation (6.1.25) such that  $\lambda(\sigma) \to 0$  as  $\sigma \to 1^-$ . Theorem 6.1.27 shows that there are also infinitely many branches  $\sigma \mapsto \lambda(\sigma)$  such that  $\lambda(\sigma) \to \mu$  as  $\sigma \to 1^-$ , for some solution  $\mu > 0$  to equation (6.1.21) (see Figure 6.1).



Figure 6.1: Solution branches of equation (6.1.25) with N = 2 for  $(\sigma, \lambda) \in [0, 1[\times]0, 500[$ . The color refers to the choice of l in (6.1.25): blue (l = 0), red (l = 1), green (l = 2), orange (l = 3), purple (l = 4), pink (l = 5), cyan (l = 6), yellow (l = 7), magenta (l = 8), brown (l = 9).

#### 6.2 Neumann eigenvalues on annuli

In this section we consider the behavior of the Neumann eigenvalues both for the Laplace and the biharmonic operator on the annulus of radii  $1 - \varepsilon$ and 1, for  $\varepsilon \in ] -\varepsilon_0, 0[\cup]0, \varepsilon_0[$ , for a suitable  $\varepsilon_0 > 0$  small. We mention the paper [89], where the authors consider the Neumann Laplacian on the annulus of radii  $1 - \varepsilon$  and 1 and prove that for N = 2 the first positive eigenvalue converges to 1 as  $\varepsilon \to 0$  and it is an increasing function of  $\varepsilon$  for  $\varepsilon \in ] -\varepsilon_0, \varepsilon_0[$ , for a suitable  $\varepsilon_0 > 0$ . As a bypass product, they obtain a counterexaple of the domain monotonicity for the first Neumann eigenvalue of the Laplace operator. The proof of the results in [89] relies on the use of the variational characterization of the first Neumann eigenvalue and on estimates for the derivatives of the eigenvalues of suitable Sturm-Liouville problems. In this section we consider the explicit characterization of all the eigenvalues in terms of zeros of suitable combinations of Bessel's functions, as well as suitable Taylor's expansions of such functions and estimates for the correspondig remainders. This technique allows to consider all eigenvalues with arbitrary indexes and multiplicity, and can be used also for the corresponding problem for the biharmonic operator.

Let  $\varepsilon \in ]-\infty, 0[\cup]0, 1[$  and let  $A_{\varepsilon} \subset \mathbb{R}^N$  be the set defined by

$$A_{\varepsilon} = \begin{cases} x \in \mathbb{R}^N : 1 - \varepsilon < |x| < 1, & \text{if } \varepsilon \in ]0, 1[, \\ x \in \mathbb{R}^N : 1 < |x| < 1 - \varepsilon, & \text{if } \varepsilon \in ] - \infty, 0[. \end{cases}$$
(6.2.1)

We consider the Neumann eigenvalue problem for the Laplace operator in  $A_{\varepsilon}$ , namely

$$\begin{cases} -\Delta u = \lambda u, & \text{in } A_{\varepsilon}, \\ \frac{\partial u}{\partial \nu} = 0, & \text{on } \partial A_{\varepsilon}. \end{cases}$$
(6.2.2)

It is convenient to use standard spherical coordinates  $(r, \theta) \in \mathbb{R}_+ \times \partial B$  in  $\mathbb{R}^N$ . We recall that the eigenvalues of problem (6.2.2) have finite multiplicity and can be represented by means of an increasing sequence diverging to  $+\infty$  of the form

$$0 = \lambda_0(\varepsilon) < \lambda_1(\varepsilon) \le \cdots \le \lambda_l(\varepsilon) \le \cdots .$$

We are interested in the behavior of the map  $\varepsilon \mapsto \lambda_i(\varepsilon)$  as  $\varepsilon \to 0$ .

### 6.2.1 Eigenvalues of the Laplace operator on the annulus of $\mathbb{R}^2$

In this subsection we consider the case N = 2. We have the following theorem on the asymptotic behavior of the eigenvalues of (6.2.2) as  $\varepsilon \to 0$ .

**Theorem 6.2.3.** For N = 2, all the eigenvalues of problem (6.2.2) have the following asymptotic behavior

$$\lambda_l(\varepsilon) = l^2 + \varepsilon l^2 + O(\varepsilon^2), \qquad (6.2.4)$$

as  $\varepsilon \to 0$ . In particular,  $\lambda'_l(0) > 0$  for all  $l \ge 1$ .

*Proof.* In polar coordinates, the first equation in (6.2.2) reads

$$-\partial_{rr}^2 u - \frac{1}{r}\partial_r u - \frac{1}{r^2}\partial_{\theta\theta}^2 u = \lambda u, \qquad (6.2.5)$$

for  $\theta \in [0, 2\pi[$  and  $r \in ]1 - \varepsilon, 1[$  (if  $\varepsilon > 0$ ), or  $r \in ]1, 1 - \varepsilon[$  (if  $\varepsilon < 0$ ). The boundary conditions read

$$\partial_r u(1,\theta) = \partial_r u(1-\varepsilon,\theta) = 0, \quad \forall \theta \in [0,2\pi[.$$
 (6.2.6)

Moreover  $u(r, 0) = u(r, 2\pi)$  for all  $r \in [1 - \varepsilon, 1[$  (if  $\varepsilon > 0$ ) or  $r \in [1, 1 - \varepsilon[$  (if  $\varepsilon < 0$ ). As customary, by separating variables so that  $u(r, \theta) = w(r)v(\theta)$  and using  $l^2$  as separation constant, with  $l \in \mathbb{N}_0$ , we have that  $v(\theta)$  is a solution of

$$-v''(\theta) = l^2 v(\theta), \qquad (6.2.7)$$

that is  $v(\theta) = A\cos(l\theta) + B\sin(l\theta)$  for  $l \in \mathbb{N}_0$  and  $A, B \in \mathbb{R}$ . Then  $u(r, \theta) = w(r) (A\cos(l\theta) + B\sin(l\theta))$ , for some  $A, B \in \mathbb{R}$ . By using formula (6.2.7) into (6.2.5), we obtain that w solves

$$w''(r) + \frac{1}{r}w'(r) + \left(\lambda - \frac{l^2}{r^2}\right)w(r) = 0.$$
 (6.2.8)

This implies that u is of the form  $u(r, \theta) = w_l(r) (A \cos(l\theta) + B \sin(l\theta))$ , where

$$w_l(r) = \alpha J_l(\sqrt{\lambda r}) + \beta Y_l(\sqrt{\lambda r}), \qquad (6.2.9)$$

where  $J_l$  and  $Y_l$  are the Bessel functions of the first and second species and order l respectively, and  $\alpha, \beta \in \mathbb{R}$  are suitable constants. By imposing boundary conditions (6.2.6) we obtain that  $\lambda$  must satisfy the condition det  $\mathcal{B} = 0$ , where

$$\mathcal{B} = \begin{bmatrix} J'_l(\sqrt{\lambda}) & Y'_l(\sqrt{\lambda}) \\ J'_l(\sqrt{\lambda}(1-\varepsilon)) & Y'_l(\sqrt{\lambda}(1-\varepsilon)) \end{bmatrix}.$$

Then  $\lambda$  is such that

$$J'_{l}(\sqrt{\lambda})Y'_{l}(\sqrt{\lambda}(1-\varepsilon)) - J'_{l}(\sqrt{\lambda}(1-\varepsilon))Y'_{l}(\sqrt{\lambda}) = 0$$
(6.2.10)

Now we expand the left-hand side of (6.2.10) in Taylor series around  $\varepsilon = 0$  up to the second order, obtaining

$$-\sqrt{\lambda}\varepsilon \left[J_l'(\sqrt{\lambda})Y_l''(\sqrt{\lambda}) - J_l''(\sqrt{\lambda})Y_l'(\sqrt{\lambda})\right] + \frac{\lambda\varepsilon^2}{2} \left[J_l'(\sqrt{\lambda})Y_l'''(\sqrt{\lambda}) - J_l'''(\sqrt{\lambda})Y_l'(\sqrt{\lambda})\right] + R(\lambda,\varepsilon) = 0, \quad (6.2.11)$$

where  $R(\lambda, \varepsilon)$  is given by

$$R(\lambda,\varepsilon) = \sum_{k=3}^{\infty} \frac{(-1)^k \varepsilon^k \sqrt{\lambda^k}}{k!} \left( Y_l'(\sqrt{\lambda}) J_l^{(k+1)}(\sqrt{\lambda}) - J_l'(\sqrt{\lambda}) Y_l^{(k+1)}(\sqrt{\lambda}) \right),$$

and is  $O(\varepsilon^3)$  as  $\varepsilon \to 0$ . The fact that the remainder  $R(\lambda, \varepsilon)$  in (6.2.11) is of order  $\varepsilon^3$  follows from the fact that for all  $m \in \mathbb{N}_0$ , the eigenvalues of (6.2.2) which we denote here by  $\lambda_m(\varepsilon)$ , are bounded away from zero and infinity uniformly in  $\varepsilon$ , as  $\varepsilon \to 0$ . Indeed, by the Rayleigh Min-Max Principle, we have that

$$\lambda_m(\varepsilon) \le \max_{c_0, \dots, c_m \in \mathbb{R}^{m+1} \setminus \{0\}} \frac{\int_{A_\varepsilon} |\sum_{k=0}^m c_k \nabla \varphi_k|^2 dx}{\int_{A_\varepsilon} (\sum_{k=0}^m c_k \varphi_k)^2 dx},$$

where  $\varphi_k(r,\theta) = r^k \cos(k\theta)$ . We consider the case  $\varepsilon > 0$  (the case  $\varepsilon < 0$  is analogous). From standard computations it follows that

$$\begin{split} \lambda_{m}(\varepsilon) \\ &\leq \max_{c_{0},...,c_{m}\in\mathbb{R}^{m+1}\setminus\{0\}} \frac{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} \left(\sum_{k=0}^{m} c_{k}kr^{k-1}\cos(k\theta)\right)^{2} r dr d\theta}{\pi \sum_{k=0}^{m} c_{k}^{2} \frac{\left(1-(1-\varepsilon)^{2(1+k)}\right)}{2(1+k)}} \\ &+ \frac{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} \frac{1}{r} \left(\sum_{k=0}^{m} c_{k}kr^{k}\sin(k\theta)\right)^{2} dr d\theta}{\pi \sum_{k=0}^{m} c_{k}^{2} \frac{\left(1-(1-\varepsilon)^{2(1+k)}\right)}{2(1+k)}} \\ &\leq (m+1) \max_{c_{0},...,c_{m}\in\mathbb{R}^{m+1}\setminus\{0\}} \frac{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} \left(\sum_{k=0}^{m} c_{k}^{2}k^{2}r^{2(k-1)}\cos(k\theta)^{2}\right) r dr d\theta}{\pi \sum_{k=0}^{m} c_{k}^{2} \frac{\left(1-(1-\varepsilon)^{2(1+k)}\right)}{2(1+k)}} \\ &+ \frac{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} \left(\sum_{k=0}^{m} c_{k}^{2}k^{2}r^{2(k-1)}\sin(k\theta)^{2}\right) r dr d\theta}{\pi \sum_{k=0}^{m} c_{k}^{2} \frac{\left(1-(1-\varepsilon)^{2(1+k)}\right)}{2(1+k)}} \\ &= (m+1) \max_{c_{0},...,c_{m}\in\mathbb{R}^{m+1}\setminus\{0\}} \frac{\sum_{k=0}^{m} c_{k}^{2}k \left(1-(1-\varepsilon)^{2(1+k)}\right)}{\sum_{k=0}^{m} c_{k}^{2} \frac{\left(1-(1-\varepsilon)^{2(1+k)}\right)}{2(1+k)}}. \end{split}$$

We note that, for all choices of  $(c_0, ..., c_m) \in \mathbb{R}^{m+1} \setminus \{0\}$ ,

$$\lim_{\varepsilon \to 0} \frac{\sum_{k=0}^{m} c_k^2 k \left( 1 - (1 - \varepsilon)^{2k} \right)}{\sum_{k=0}^{m} c_k^2 \frac{\left( 1 - (1 - \varepsilon)^{2(1+k)} \right)}{2(1+k)}} = 2 \frac{\sum_{k=0}^{m} k^2 c_k^2}{\sum_{k=1}^{m} c_k^2} \le 2(m+1)^2.$$

Thus, for  $\varepsilon > 0$  sufficiently small,  $\lambda_m(\varepsilon)$  is uniformly bounded from above, as  $\varepsilon \to 0$ . We now prove that all the positive eigenvalues are bounded away from zero. In fact, given a positive eigenvalue  $\lambda$  of problem (6.2.2), a corresponding eigenfunction is given by  $w_l(r)(A\cos(l\theta) + B\sin(l\theta))$ , for some  $l \in \mathbb{N}_0$ , and  $w_l(r)$  is as in (6.2.9). Then we have

$$\lambda = \frac{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} rw_{l}'(r)^{2} (A\cos(l\theta) + B\sin(l\theta))^{2} drd\theta}{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} w_{l}(r)^{2} (A\cos(l\theta) + B\sin(l\theta))^{2} rdrd\theta} + \frac{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} \frac{1}{r} \left(w_{l}(r)^{2} l^{2} (-A\sin(l\theta) + B\cos(l\theta))^{2}\right) drd\theta}{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} w_{l}(r)^{2} (A\cos(l\theta) + B\sin(l\theta))^{2} rdrd\theta} \geq \frac{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} \frac{1}{r} \left(w_{l}(r)^{2} l^{2} (-A\sin(l\theta) + B\cos(l\theta))^{2}\right) drd\theta}{\int_{0}^{2\pi} \int_{1-\varepsilon}^{1} w_{l}(r)^{2} (A\cos(l\theta) + B\sin(l\theta))^{2} rdrd\theta} = l^{2} \frac{\int_{1-\varepsilon}^{1} \frac{w_{l}(r)^{2}}{r} dr}{\int_{1-\varepsilon}^{1} w_{l}(r)^{2} rdr} \geq 1, \quad (6.2.12)$$

if  $\varepsilon \in ]0, 1[$ , while

$$\begin{split} \lambda &= \frac{\int_{0}^{2\pi} \int_{1}^{1-\varepsilon} rw_{l}'(r)^{2} (A\cos(l\theta) + B\sin(l\theta))^{2} drd\theta}{\int_{0}^{2\pi} \int_{1}^{1-\varepsilon} w_{l}(r)^{2} (A\cos(l\theta) + B\sin(l\theta))^{2} r drd\theta} \\ &+ \frac{\int_{0}^{2\pi} \int_{1}^{1-\varepsilon} \frac{1}{r} \left( w_{l}(r)^{2} l^{2} (-A\sin(l\theta) + B\cos(l\theta))^{2} \right) drd\theta}{\int_{0}^{2\pi} \int_{1}^{1-\varepsilon} w_{l}(r)^{2} (A\cos(l\theta) + B\sin(l\theta))^{2} r drd\theta} \\ &\geq \frac{\int_{0}^{2\pi} \int_{1}^{1-\varepsilon} \frac{1}{r} \left( w_{l}(r)^{2} l^{2} (-A\sin(l\theta) + B\cos(l\theta))^{2} \right) drd\theta}{\int_{0}^{2\pi} \int_{1}^{1-\varepsilon} w_{l}(r)^{2} (A\cos(l\theta) + B\sin(l\theta))^{2} r drd\theta} \\ &= l^{2} \frac{\int_{1}^{1-\varepsilon} \frac{w_{l}(r)^{2}}{r} dr}{\int_{1}^{1-\varepsilon} w_{l}(r)^{2} r dr} \geq \frac{1}{4}, \quad (6.2.13) \end{split}$$

if  $\varepsilon \in ]-1,0[$ .

We note now that  $\lambda = 0$  is an eigenvalue of problem (6.2.2) and the associated eigenfunctions are the constant functions on  $A_{\varepsilon}$ . In particular,  $\lambda = 0$ is the first eigenvalue of (6.2.8) with l = 0 and with Neumann boundary conditions (6.2.6). Moreover, all the non-zero eigenvalues of problem (6.2.2) are bounded away from zero. Then in equation (6.2.11), we can assume  $\lambda \neq 0$ . Dividing by  $\varepsilon$  and multiplying by  $\pi\lambda$  in (6.2.11), and recalling that (see Lemmas 4.1.40 and 4.1.46)

$$J_l'(\sqrt{\lambda})Y_l''(\sqrt{\lambda}) - J_l''(\sqrt{\lambda})Y_l'(\sqrt{\lambda}) = -\frac{2}{\pi\sqrt{\lambda}}\left(\frac{l^2}{\lambda} - 1\right) \quad (6.2.14)$$

$$J_l'(\sqrt{\lambda})Y_l'''(\sqrt{\lambda}) - J_l'''(\sqrt{\lambda})Y_l'(\sqrt{\lambda}) = -\frac{2}{\pi\lambda}\left(1 - \frac{3l^2}{\lambda}\right), \quad (6.2.15)$$

equation (6.2.11) can be rewritten as

$$2(l^{2} - \lambda) - (\lambda - 3l^{2})\varepsilon + \tilde{R}(\lambda, \varepsilon), \qquad (6.2.16)$$

where  $\tilde{R}(\lambda, \varepsilon) = \pi \lambda R(\lambda, \varepsilon) / \varepsilon \in O(\varepsilon^2)$  as  $\varepsilon \to 0$ .

We consider equation (6.2.16) and apply the Implicit Function Theorem. Equation (6.2.16) can be written in the form  $F(\lambda, \varepsilon) = 0$ , with Fof class  $C^1$  in  $]0, +\infty[\times[0,1[, \text{ and } F(\lambda,0) = 2(l^2 - \lambda), \partial_{\lambda}F(\lambda,0) = -2,$ and  $\partial_{\varepsilon}F(\lambda,0) = 3l^2 - \lambda$ . Since  $F(l^2,0) = 0$  and  $\partial_{\lambda}F(l^2,0) = -2 \neq 0$  for all  $l \in \mathbb{N}_0$ , the zeros of equation (6.2.11) in a neighborhood of  $(\lambda,0)$  are given by the graph of a  $C^1$ -function  $\varepsilon \mapsto \lambda(\varepsilon)$  with  $\lambda(0) = l^2$ . Moreover  $\lambda'(0) = -\partial_{\varepsilon}F(l^2,0)/\partial_{\lambda}F(l^2,0) = l^2$  for  $l \in \mathbb{N}_0$ . Then  $\lambda(\varepsilon) = l^2 + \varepsilon l^2 + O(\varepsilon^2)$ as  $\varepsilon \to 0$ , for all  $l \in \mathbb{N}$ . Moreover  $\lambda_0(\varepsilon) = 0$  for all  $\varepsilon \in ] -\infty, 0[\cup]0, 1[$ , thus formula (6.2.4) holds also for l = 0. Moreover  $\lambda'_l(0) > 0$  for all  $l \geq 1$ . This concludes the proof of the theorem.  $\Box$ 

**Remark 6.2.17.** The limiting eigenvalues are the eigenvalues  $\mu$  of  $-v''(s) = \mu v(s)$  on  $]0, 2\pi[$  with periodic conditions  $v(0) = v(2\pi)$ .

**Remark 6.2.18.** We note that, by standard Sturm-Liouville theory, for all  $l \in \mathbb{N}_0$ , equation (6.2.8) with Neumann boundary conditions (6.2.6) admits a diverging sequence of non-negative and simple eigenvalues, which we denote by  $\lambda_{l,k}(\varepsilon)$ ,  $k \in \mathbb{N}_0$ , with  $\lambda_{l,k_1}(\varepsilon) < \lambda_{l,k_2}(\varepsilon)$  if  $k_1 < k_2$ . In particular,  $\lambda_{0,0}(\varepsilon) = 0$  for all  $\varepsilon \in ] -\infty, 0[\cup]0, 1[$ , and  $\lambda_{l,k}(\varepsilon) \geq \frac{1}{4}$  for all  $\varepsilon \in ] -1, 0[\cup]0, 1[$  (this follows from (6.2.12) and (6.2.13)) and for all  $l, k \in \mathbb{N}_0$ ,  $(l, k) \neq (0, 0)$ . Moreover, as a consequence of Theorem 6.2.3, we have that  $\lambda_{l,0} \to l^2$  for all  $l \in \mathbb{N}_0$  and  $\lambda_{l,k} \to +\infty$  for all  $l \in \mathbb{N}_0$  and for all  $k \geq 1$ , as  $\varepsilon \to 0$ .

Such behavior of the eigenvalues can be also explained as follows. Let us recall that the k-th zero of the cross product

$$J'_{l}(z)Y'_{l}(\alpha z) - Y'_{l}(z)J'_{l}(\alpha z)$$
(6.2.19)

has a well-known asymptotic expansion. We set  $\eta = 4l^2$ ,  $\beta = \frac{k\pi}{\alpha - 1}$ ,  $p = \frac{\eta + 3}{8\alpha}$ ,  $q = \frac{(\eta^2 + 46\eta - 63)(\alpha^3 - 1)}{6(4\alpha)^3(\alpha - 1)}$ ,  $r = \frac{(\eta^3 + 185\eta^2 - 2053\eta + 1899)}{5(4\eta)^5(\alpha - 1)}$ . Then the k-th zero of (6.2.19) is given by

$$\beta + \frac{p}{\beta} + \frac{q-p^2}{\beta^3} + \frac{r-4pq+2p^3}{\beta^5} + \cdots$$

We exploit all the computations by setting  $\alpha = 1 - \varepsilon$  and  $z = \sqrt{\lambda}$  in (6.2.19) and collect the terms according to the powers of  $\varepsilon$ , obtaining

$$\lambda_{l,k}(\varepsilon) = \frac{\pi^2 k^2}{\varepsilon^2} + l^2 + \varepsilon l^2 + O(\varepsilon^2).$$
(6.2.20)

Note that  $\lambda_{l,0} = l^2 + \varepsilon l^2 + O(\varepsilon^2)$  which converges to  $\lambda_l(0) = l^2$  as  $\varepsilon \to 0$ , while for k > 1,  $\lambda_{l,k}(\varepsilon) \to +\infty$  as  $\varepsilon \to 0$ . See also Figure 6.2.

**Remark 6.2.21.** Problem (6.2.2) presents some analogies with the eigenvalue problem for the Laplace operator  $-\Delta u = \lambda u$  on  $]0, 2\pi[\times]0, \varepsilon[$  with

Neumann conditions at  $x_2 = 0$  and  $x_2 = \varepsilon$  and periodic condition at  $x_1 = 0$ and  $x_1 = 2\pi$ . In this case, for each fixed  $\varepsilon$  the eigenvalues are given by  $\frac{\pi^2 k^2}{\varepsilon^2} + l^2$ , with  $l, k \in \mathbb{N}_0$ , which are exactly the first two terms appearing in formula (6.2.20).



Figure 6.2: Solution branches of equation (6.2.10) for  $(\varepsilon, \lambda) \in ]-1, 1[\times]0, 40[$ . The color refers to the choice of l in (6.2.10). In particular l = 0 (blue), l = 1 (red), l = 2 (purple), l = 3 (green), l = 4 (orange).

## 6.2.2 Eigenvalues of the Laplace operator on the annulus of $\mathbb{R}^N$

In this subsection we consider the case  $N \geq 3$ . We have the following theorem on the behavior of the eigenvalues of (6.2.2) as  $\varepsilon \to 0$ .

**Theorem 6.2.22.** All the eigenvalues of problem (6.2.2) have the following asymptotic behavior

$$\lambda_l(\varepsilon) = l(l+N-2) + \varepsilon l(l+N-2) + O(\varepsilon^2),$$

as  $\varepsilon \to 0$ . In particular,  $\lambda'_l(0) > 0$  for all  $l \ge 1$ .

*Proof.* The proof follows the same lines as that of Theorem 6.2.3. We repeat briefly it here for the reader's convenience. As is customary, we separate variables and look for solutions to problem (6.2.2) of the form  $u(r, \theta) =$  $w(r)H(\theta)$ . We use l(l + N - 2) with  $l \in \mathbb{N}_0$  as separation constant. It turns out that H solves the equation

$$-\Delta_S H = l(l+N-2)H$$
, on  $\partial B$ ,

for some  $l \in \mathbb{N}_0$ . Therefore  $H(\theta) = H_l(\theta)$ , where  $H_l(\theta)$  is a spherical harmonic of some order  $l \in \mathbb{N}_0$ . The radial part w(r) solves the equation

$$w''(r) + \frac{N-1}{r}w'(r) + \left(\lambda - \frac{l(l+N-2)}{r^2}\right)w(r) = 0.$$

This implies that, given an eigenvalue  $\lambda$  of problem (6.2.2) on  $A_{\varepsilon}$ , a corresponding eigenfunction is of the form

$$u(r,\theta) = (\alpha j_l(\sqrt{\lambda}r) + \beta y_l(\sqrt{\lambda}r))H_l(\theta),$$

where  $\alpha, \beta \in \mathbb{R}$  are suitable constants and  $j_l$  and  $y_l$  are ultraspherical Bessel functions of the first and second species and order l respectively. Now we impose Neumann boundary conditions. We obtain

$$\begin{aligned} \alpha j'_l(\lambda) + \beta y'_l(\lambda) &= 0, \\ \alpha j'_l(\lambda(1-\varepsilon)) + \beta y'_l(\lambda(1-\varepsilon)) &= 0 \end{aligned}$$

Thus, the eigenvalues are implicitly given by the equation  $\det \mathcal{B} = 0$ , where

$$\mathcal{B} = \begin{bmatrix} j'_l(\lambda) & y'_l(\lambda) \\ j'_l(\lambda(1-\varepsilon)) & y'_l(\lambda(1-\varepsilon))0 \end{bmatrix}.$$

From standard computations it follows that the eigenvalues of problem (6.2.2) are solutions to the following implicit equation

$$(N-2)^{2} \left( J_{\frac{N}{2}-1+l}(\sqrt{\lambda})Y_{\frac{N}{2}-1+l}(\sqrt{\lambda}(1-\varepsilon)) - Y_{\frac{N}{2}-1+l}(\sqrt{\lambda})J_{\frac{N}{2}-1+l}(\sqrt{\lambda}(1-\varepsilon)) \right) + 2(N-2)\sqrt{\lambda} \left( Y_{\frac{N}{2}-1+l}(\sqrt{\lambda})J_{\frac{N}{2}-1+l}(\sqrt{\lambda}(1-\varepsilon)) - J_{\frac{N}{2}-1+l}(\sqrt{\lambda})Y_{\frac{N}{2}-1+l}(\sqrt{\lambda}(1-\varepsilon)) \right) + 2(N-2)(1-\varepsilon)\sqrt{\lambda} \left( Y_{\frac{N}{2}-1+l}(\sqrt{\lambda})J_{\frac{N}{2}-1+l}(\sqrt{\lambda}(1-\varepsilon)) - J_{\frac{N}{2}-1+l}(\sqrt{\lambda})Y_{\frac{N}{2}-1+l}(\sqrt{\lambda}(1-\varepsilon)) \right) + 4(1-\varepsilon)\lambda \left( J_{\frac{N}{2}-1+l}(\sqrt{\lambda})Y_{\frac{N}{2}-1+l}(\sqrt{\lambda}(1-\varepsilon)) - Y_{\frac{N}{2}-1+l}(\sqrt{\lambda})J_{\frac{N}{2}-1+l}(\sqrt{\lambda}(1-\varepsilon)) \right) = 0.$$

$$(6.2.23)$$

We expand the left-hand side of (6.2.23) in  $\varepsilon$  near  $\varepsilon = 0$  up to the second order. Then we divide both sides of the expansion of (6.2.23) by  $\varepsilon$  and use formulas (6.2.14) and (6.2.15). From standard computations we obtain that equation (6.2.23) can be re-written in the following equivalent form

$$8(l(l+N-2)-\lambda) + 4(l(l+N-2)+\lambda)\varepsilon + R(\lambda,\varepsilon) = 0,$$

where  $R(\lambda, \varepsilon) \in O(\varepsilon^2)$ , uniformly in  $\lambda > 0$ , as  $\varepsilon \to 0$ . Thus, by letting  $\varepsilon \to 0$ we obtain that the following asymptotic expansion of the eigenvalues  $\lambda_l(\varepsilon)$ of (6.2.2) holds

$$\lambda(\varepsilon) = l(l+N-2) + l(l+N-2)\varepsilon + O(\varepsilon^2), \qquad (6.2.24)$$

as  $\varepsilon \to 0$ . This concludes the proof of the theorem.

**Remark 6.2.25.** We note that in a neighborhood of  $\varepsilon = 0$ , the eigenvalues of problem (6.2.2) behave like (6.2.24). In particular, we deduce that all the eigenvalues of (6.2.2) are not monotonic with respect to the inclusion of sets (this is shown in the case N = 2 for the first eigenvalue of problem (6.2.2) in [89]). In fact, if  $0 < \varepsilon_1 < \varepsilon_2$  are sufficiently small (possibly depending on l), then  $A_{\varepsilon_2} \supseteq A_{\varepsilon_1}$  and  $\lambda_l(\varepsilon_2) > \lambda_l(\varepsilon_1)$ . Conversely, if  $\varepsilon_2 < \varepsilon_1 < 0$ , then  $A_{\varepsilon_2} \supseteq A_{\varepsilon_1}$  and  $\lambda_l(\varepsilon_2) < \lambda_l(\varepsilon_1)$ .

### 6.2.3 Eigenvalues of the biharmonic operator on the annulus of $\mathbb{R}^N$

Let  $\varepsilon \in ]-1, 0[\cup]0, 1[$  and let  $A_{\varepsilon} \subset \mathbb{R}^N$  be the set defined by (6.2.1). We consider the following eigenvalue problem for the biharmonic operator subject to Neumann boundary conditions on  $A_{\varepsilon}$ 

$$\begin{cases} \Delta^2 u = \lambda u, & \text{in } A_{\varepsilon}, \\ \frac{\partial^2 u}{\partial \nu^2} = 0, & \text{on } \partial A_{\varepsilon}, \\ \operatorname{div}_{\partial \Omega} \left( D^2 u. \nu \right) + \frac{\partial \Delta u}{\partial \nu} = 0, & \text{on } \partial A_{\varepsilon}. \end{cases}$$
(6.2.26)

We recall that in spherical coordinates we have  $\Delta u = \partial_{rr}^2 u + \frac{N-1}{r} \partial_r u + \frac{1}{r^2} \Delta_S u$ . The boundary conditions in (6.2.26) are written in spherical coordinates as

$$\begin{cases}
\frac{\partial_{rr} u_{|r=1}}{\partial_{rr} u_{|r=1-\varepsilon}} = 0, \\
\frac{\partial_{rr} u_{|r=1-\varepsilon}}{\partial_{rr} \Delta_{S}} \left( \frac{\partial_{r} u - \frac{u}{r}}{r} \right) + \partial_{r} \left( \Delta u \right)_{|r=1}} = 0, \\
\frac{1}{r^{2}} \Delta_{S} \left( \partial_{r} u - \frac{u}{r} \right) + \partial_{r} \left( \Delta u \right)_{|r=1-\varepsilon}} = 0.
\end{cases}$$
(6.2.27)

By following [28], in order to find a solution to the differential equation  $\Delta^2 u = \lambda u$ , we factor the eigenvalue equation equation as

$$(\Delta + a^2)(\Delta - a^2)u = 0,$$

where a > 0 is such that  $\lambda = a^4$ . The eigenfunctions turn out to be linear combinations of  $v_1$  and  $v_2$  where  $-\Delta v_1 = a^2 v_1$  and  $\Delta v_2 = a^2 v_2$ . It follows that

$$v_1 = (j_l(ar) + y_l(ar)) H_l(\theta), \quad v_2 = (i_l(ar) + k_l(ar)) H_l(\theta),$$

where  $j_l, y_l$  are the ultraspherical Bessel functions of the first and second kind respectively, and  $i_l, k_l$  are the ultraspherical modified Bessel functions of the first and second kind respectively, and  $H_l$  is a spherical harmonic of some order  $l \in \mathbb{N}_0$ . Then, it is standard to prove that, given an eigenvalue  $\lambda$ of problem (6.2.26), a corresponding eigenfunction u has the following form (see [28] for more details)

$$u(r,\theta) = (Aj_l(ar) + By_l(ar) + Ci_l(ar) + Dk_l(ar)) H_l(\theta),$$

for  $l \in \mathbb{N}_0$ , where a > 0 is such that  $a^4 = \lambda$  and  $A, B, C, D \in \mathbb{R}$  are suitable constants. Now we impose the boundary conditions (6.2.27). We obtain

i)

$$a^{2} \left( A j_{l}''(a) + B y_{l}''(a) + C i_{l}''(a) + D k_{l}''(a) \right) = 0,$$

ii)

$$a^{2}\left(Aj_{l}''(a-a\varepsilon)+By_{l}''(a-a\varepsilon)+Ci_{l}''(a-a\varepsilon)+Dk_{l}''(a-a\varepsilon)\right)=0,$$

iii)

$$\begin{aligned} A\left(3l(l+N-2)j_{l}(a)+a(1-N)j_{l}'(a)\right.\\ &\left.-2al(l+N-2)j_{l}'(a)+a^{2}(N-1)j_{l}''(a)+a^{3}j_{l}'''(a)\right)\\ &+B\left(3l(l+N-2)y_{l}(a)+a(1-N)y_{l}'(a)\right.\\ &\left.-2al(l+N-2)y_{l}'(a)+a^{2}(N-1)y_{l}''(a)+a^{3}y_{l}'''(a)\right)\\ &+C\left(3l(l+N-2)i_{l}(a)+a(1-N)i_{l}'(a)\right.\\ &\left.-2al(l+N-2)i_{l}'(a)+a^{2}(N-1)i_{l}''(a)+a^{3}i_{l}'''(a)\right)\\ &+D\left(3l(l+N-2)k_{l}(a)+a(1-N)k_{l}'(a)\right.\\ &\left.-2al(l+N-2)k_{l}'(a)+a^{2}(N-1)k_{l}''(a)+a^{3}k_{l}'''(a)\right)=0,\end{aligned}$$

iv)

$$\begin{split} A\left(\frac{3l(l+N-2)}{(1-\varepsilon)^3}j_l(a-a\varepsilon) + \frac{a(1-N)}{(1-\varepsilon)^2}j_l'(a-a\varepsilon) \\ -\frac{2al(l+N-2)}{(1-\varepsilon)^2}j_l'(a-a\varepsilon) + \frac{a^2(N-1)}{1-\varepsilon}j_l''(a-a\varepsilon) + a^3j_l'''(a-a\varepsilon)\right) \\ + B\left(\frac{3l(l+N-2)}{(1-\varepsilon)^3}y_l(a-a\varepsilon) + \frac{a(1-N)}{(1-\varepsilon)^2}y_l'(a-a\varepsilon) \\ -\frac{2al(l+N-2)}{(1-\varepsilon)^2}y_l'(a-a\varepsilon) + \frac{a^2(N-1)}{1-\varepsilon}y_l''(a-a\varepsilon) + a^3y_l'''(a-a\varepsilon)\right) \\ + C\left(\frac{3l(l+N-2)}{(1-\varepsilon)^3}i_l(a-a\varepsilon) + \frac{a(1-N)}{(1-\varepsilon)^2}i_l'(a-a\varepsilon) \\ -\frac{2al(l+N-2)}{(1-\varepsilon)^2}i_l'(a-a\varepsilon) + \frac{a^2(N-1)}{1-\varepsilon}i_l''(a-a\varepsilon) + a^3i_l'''(a-a\varepsilon)\right) \\ + D\left(\frac{3l(l+N-2)}{(1-\varepsilon)^3}k_l(a-a\varepsilon) + \frac{a(1-N)}{(1-\varepsilon)^2}k_l'(a-a\varepsilon) \\ -\frac{2al(l+N-2)}{(1-\varepsilon)^2}k_l'(a-a\varepsilon) + \frac{a^2(N-1)}{1-\varepsilon}k_l''(a-a\varepsilon) + a^3k_l'''(a-a\varepsilon)\right) \\ = 0. \end{split}$$

This is a system of four equations in four unknowns,  $A, B, C, D \in \mathbb{R}$ , which has solution if and only if the determinant of the associated matrix is zero.

For a fixed  $l \in \mathbb{N}_0$ , let  $\mathcal{B}_l(\lambda, \varepsilon)$  be the matrix associated with the linear system. The eigenvalues are implicitly characterized by the equation  $\det \mathcal{B}_l(\lambda, \varepsilon) = 0$ . In principle, the equation can be handled in the same way as for the second order case, by using a suitable Taylor's expansion of  $\det \mathcal{B}_l(\lambda, \varepsilon)$  in  $\varepsilon = 0$  and then simplifying by using suitable recurrence formulas for cross products of Bessel functions and their derivatives. Actually, the computations become very long and involved, and it seems quite difficult to obtain a simple closed formulas for the limiting eigenvalues and their derivatives. We have used the software *Mathematica* to perform symbolic computations in the case N = 2. The software was not able to handle the case  $N \geq 3$ , which has a larger number of terms and an additional symbolic quantity N.

#### 6.2.4 Symbolic computations for the eigenvalues of the biharmonic operator on the annulus of $\mathbb{R}^2$

We have used the software *Mathematica* for symbolic computations in order to expand and simplify the expression det  $\mathcal{B}_l(\lambda, \varepsilon)$  of Subsection 6.2.3.

As in the case of the Laplace operator, we have expanded in Taylor series the function det $\mathcal{B}_l(\lambda, \varepsilon)$  up to the third order around  $\varepsilon = 0$ . We considered only the case N = 2. After a long computation time we obtained the following expression

$$\frac{1}{\pi} \left( -8 \left( \lambda + 2(\lambda - 1)l^2 + 4l^4 - 2l^6 \right) \varepsilon^2 + 8 \left( \lambda + 2(1 + \lambda)l^2 - 4l^4 + 2l^6 \right) \varepsilon^3 \right) + O(\varepsilon^4) = 0.$$

Using the same arguments as in the case of the eigenvalues of the Laplace operator on the annulus, we find that near  $\varepsilon = 0$  the behavior of the eigenvalue  $\lambda_l(\varepsilon)$  is

$$\lambda_l(\varepsilon) = \frac{2l^2(l^2-1)^2}{1+2l^2} + \frac{4l^2(l^2-1)^2}{1+2l^2}\varepsilon + O(\varepsilon^2),$$

as  $\varepsilon \to 0$ , for all  $l \in \mathbb{N}_0$ . See Figure 6.3

This shows that also in the case of the biharmonic operator and N = 2there is no monotonicity of all the eigenvalues under inclusion of domains.



Figure 6.3: Solution branches of equation det  $\mathcal{B}(\lambda, \varepsilon)$  with N = 2 and  $(\varepsilon, \lambda) \in ] -1, 1[\times]0, 500[$ . The colors refers to the choice of l in the equation. In particular, l = 0 (blue), l = 1 (red), l = 2 (purple), l = 3 (green), l = 4 (orange). We also plotted the line  $\frac{2l^2(l^2-1)^2}{1+2l^2} + \frac{4l^2(l^2-1)^2}{1+2l^2}\varepsilon$  with l = 2 (purple), l = 3 (green) and l = 4 (orange).

### Bibliography

- M. Abramowitz and I. A. Stegun. Handbook of mathematical functions with formulas, graphs, and mathematical tables, volume 55 of National Bureau of Standards Applied Mathematics Series. For sale by the Superintendent of Documents, U.S. Government Printing Office, Washington, D.C., 1964.
- [2] S. Agmon. Lectures on elliptic boundary value problems. Prepared for publication by B. Frank Jones, Jr. with the assistance of George W. Batten, Jr. Van Nostrand Mathematical Studies, No. 2. D. Van Nostrand Co., Inc., Princeton, N.J.-Toronto-London, 1965.
- [3] S. Agmon, A. Douglis, and L. Nirenberg. Estimates near the boundary for solutions of elliptic partial differential equations satisfying general boundary conditions. I. *Comm. Pure Appl. Math.*, 12:623–727, 1959.
- [4] S. Agmon, A. Douglis, and L. Nirenberg. Estimates near the boundary for solutions of elliptic partial differential equations satisfying general boundary conditions. II. Comm. Pure Appl. Math., 17:35–92, 1964.
- [5] J. H. Albert. Generic properties of eigenfunctions of elliptic partial differential operators. *Trans. Amer. Math. Soc.*, 238:341–354, 1978.
- [6] E. Almansi. Sull'integrazione dell'equazione differenziale  $\Delta^2 \Delta^2 = 0$ . Annali di Matematica Pura ed Applicata, 31(14):881–888, 1896.
- [7] E. Almansi. Sull'integrazione dell'equazione differenziale  $\Delta^{2n} = 0$ . Annali di Matematica Pura ed Applicata, 2(1):1–51, 1899.
- [8] P. M. Anselone. Collectively compact operator approximation theory and applications to integral equations. Prentice-Hall, Inc., Englewood Cliffs, N. J., 1971. With an appendix by Joel Davis, Prentice-Hall Series in Automatic Computation.
- [9] J. Arrieta, A. Jiménez-Casas, and A. Rodriguez-Bernal. Nonhomogeneous flux condition as limit of concentrated reactions. *Revista Iberoamericana de Matematicas*, 24(1):183–211, 2008.

- [10] J. M. Arrieta, A. Jiménez-Casas, and A. Rodríguez-Bernal. Flux terms and Robin boundary conditions as limit of reactions and potentials concentrating at the boundary. *Rev. Mat. Iberoam.*, 24(1):183–211, 2008.
- [11] J. M. Arrieta, A. Rodríguez-Bernal, and J. D. Rossi. The best Sobolev trace constant as limit of the usual Sobolev constant for small strips near the boundary. *Proc. Roy. Soc. Edinburgh Sect. A*, 138(2):223–237, 2008.
- [12] M. S. Ashbaugh and R. D. Benguria. On Rayleigh's conjecture for the clamped plate and its generalization to three dimensions. *Duke Math.* J., 78(1):1–17, 1995.
- [13] G. Auchmuty. Dual variational principles for eigenvalue problems. In Nonlinear functional analysis and its applications, Part 1 (Berkeley, Calif., 1983), volume 45 of Proc. Sympos. Pure Math., pages 55–71. Amer. Math. Soc., Providence, RI, 1986.
- [14] C. Bandle. Isoperimetric inequalities and applications, volume 7 of Monographs and Studies in Mathematics. Pitman (Advanced Publishing Program), Boston, Mass.-London, 1980.
- [15] I. A. Borovikov. Normally resolvable operators and direct decompositions of Sobolev spaces. J. Math. Sci. (N. Y.), 186(2):153–178, 2012. Problems in mathematical analysis. No. 66.
- [16] L. Brasco, G. De Philippis, and B. Ruffini. Spectral optimization for the Stekloff-Laplacian: the stability issue. J. Funct. Anal., 262(11):4675–4710, 2012.
- [17] L. Brasco and A. Pratelli. Sharp stability of some spectral inequalities. Geom. Funct. Anal., 22(1):107–135, 2012.
- [18] H. Brezis. Analyse fonctionnelle. Collection Mathématiques Appliquées pour la Maîtrise. [Collection of Applied Mathematics for the Master's Degree]. Masson, Paris, 1983. Théorie et applications. [Theory and applications].
- [19] F. Brock. An isoperimetric inequality for eigenvalues of the Stekloff problem. ZAMM Z. Angew. Math. Mech., 81(1):69–71, 2001.
- [20] D. Bucur, A. Ferrero, and F. Gazzola. On the first eigenvalue of a fourth order Steklov problem. *Calc. Var. Partial Differential Equations*, 35(1):103–131, 2009.
- [21] D. Buoso, L. M. Chasman, and L. Provenzano. On the stability of some isoperimetric inequalities for the fundamental tones of free plates. *In* preparation, 2016.

- [22] D. Buoso and P. D. Lamberti. Eigenvalues of polyharmonic operators on variable domains. ESAIM Control Optim. Calc. Var., 19(4):1225– 1235, 2013.
- [23] D. Buoso and P. D. Lamberti. Shape deformation for vibrating hinged plates. Math. Methods Appl. Sci., 37(2):237–244, 2014.
- [24] D. Buoso and L. Provenzano. A few shape optimization results for a biharmonic Steklov problem. J. Differential Equations, 259(5):1778– 1818, 2015.
- [25] D. Buoso and L. Provenzano. On the Eigenvalues of a Biharmonic Steklov Problem. In Integral Methods in Science and Engineering, Theoretical and Computational Advances, pages 81–89, 2015.
- [26] V. I. Burenkov. Sobolev spaces on domains, volume 137 of Teubner-Texte zur Mathematik [Teubner Texts in Mathematics]. B. G. Teubner Verlagsgesellschaft mbH, Stuttgart, 1998.
- [27] S. Chanillo, D. Grieser, M. Imai, K. Kurata, and I. Ohnishi. Symmetry breaking and other phenomena in the optimization of eigenvalues for composite membranes. *Comm. Math. Phys.*, 214(2):315–337, 2000.
- [28] L. M. Chasman. An isoperimetric inequality for fundamental tones of free plates. *Comm. Math. Phys.*, 303(2):421–449, 2011.
- [29] L. M. Chasman. Vibrational modes of circular free plates under tension. Appl. Anal., 90(12):1877–1895, 2011.
- [30] B. Colbois, A. El Soufi, and A. Savo. Eigenvalues of the Laplacian on a compact manifold with density. *Comm. Anal. Geom.*, 23(3):639–670, 2015.
- [31] R. Courant and D. Hilbert. Methods of mathematical physics. Vol. I. Interscience Publishers, Inc., New York, N.Y., 1953.
- [32] S. J. Cox. Extremal eigenvalue problems for the Laplacian. In Recent advances in partial differential equations (El Escorial, 1992), volume 30 of RAM Res. Appl. Math., pages 37–53. Masson, Paris, 1994.
- [33] S. J. Cox and J. R. McLaughlin. Extremal eigenvalue problems for composite membranes. I, Appl. Math. Optim., 22(2):153–167, 1990.
- [34] S. J. Cox and J. R. McLaughlin. Extremal eigenvalue problems for composite membranes, II. Appl. Math. Optim., 22(1):169–187, 1990.
- [35] M. Dalla Riva and P. Musolino. Real analytic families of harmonic functions in a domain with a small hole. J. Differential Equations, 252(12):6337–6355, 2012.

- [36] M. Dalla Riva and L. Provenzano. On vibrating thin membranes with mass concentrated near the boundary. *In preparation*, 2016.
- [37] K. Deimling. Nonlinear functional analysis. Springer-Verlag, Berlin, 1985.
- [38] M. C. Delfour and J.-P. Zolésio. Shapes and geometries, volume 22 of Advances in Design and Control. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, second edition, 2011. Metrics, analysis, differential calculus, and optimization.
- [39] R. Duffin. the influence of Poisson's ratio on the vibrational spectrum. SIAM J. Appl. Math., 17:179–191, 1969.
- [40] C. Efthimiou and C. Frye. Spherical harmonics in p dimensions. World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2014.
- [41] G. Faber. Beweis, daß unter allen homogenen Membranen von gleicher Fläche und gleicher Spannung die kreisförmige den tiefsten Grundton gibt. Münch. Ber. 1923, 169-172 (1923)., 1923.
- [42] G. B. Folland. Introduction to partial differential equations. Princeton University Press, Princeton, NJ, second edition, 1995.
- [43] S. Friedland. Extremal eigenvalue problems defined for certain classes of functions. Arch. Rational Mech. Anal., 67(1):73–81, 1977.
- [44] S. Friedland. Extremal eigenvalue problems. Bol. Soc. Brasil. Mat., 9(1):13–40, 1978.
- [45] N. Fusco, F. Maggi, and A. Pratelli. Stability estimates for certain Faber-Krahn, isocapacitary and Cheeger inequalities. Ann. Sc. Norm. Super. Pisa Cl. Sci. (5), 8(1):51–71, 2009.
- [46] A. Gajewski, N. Olhoff, J. Rondal, A. P. Seyranian, and M. Życzkowski. Structural optimization under stability and vibration constraints, volume 308 of CISM Courses and Lectures. Springer-Verlag, Vienna, 1989.
- [47] F. Gazzola, H.-C. Grunau, and G. Sweers. Polyharmonic boundary value problems, volume 1991 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 2010. Positivity preserving and nonlinear higher order elliptic equations in bounded domains.
- [48] A. Girouard and I. Polterovich. Spectral geometry of the steklov problem, 2009.

- [49] Y. D. Golovaty. Spectral properties of oscillatory systems with attached masses. Tr. Mosk. Mat. Obshch. (Trans. Moscow Math. Soc., 1993), 54:29–72, 1992.
- [50] Y. D. Golovaty, D. Gómez, M. Lobo, and E. Pérez. On vibrating membranes with very heavy thin inclusions. *Math. Models Methods Appl. Sci.*, 14(7):987–1034, 2004.
- [51] Y. D. Golovaty and A. Lavrenyuk. Asymptotic expansions of local eigenvibrations for plate with density perturbed in neighbourhood of one-dimensional manifold. *Mat. Stud*, 13(1):51–62, 2000.
- [52] D. Gómez, M. Lobo, S. A. Nazarov, and E. Pérez. Asymptotics for the spectrum of the Wentzell problem with a small parameter and other related stiff problems. J. Math. Pures Appl. (9), 86(5):369–402, 2006.
- [53] D. Gómez, M. Lobo, S. A. Nazarov, and E. Pérez. Spectral stiff problems in domains surrounded by thin bands: asymptotic and uniform estimates for eigenvalues. J. Math. Pures Appl. (9), 85(4):598–632, 2006.
- [54] D. Gómez, M. Lobo, and E. Pérez. On the eigenfunctions associated with the high frequencies in systems with a concentrated mass. J. Math. Pures Appl. (9), 78(8):841–865, 1999.
- [55] D. Gómez, M. Lobo, and E. Pérez. Vibrating plates with concentrated masses and very small thickness: low frequencies. In *EQUADIFF* 2003, pages 473–475. World Sci. Publ., Hackensack, NJ, 2005.
- [56] D. Gómez, M. Lobo, and E. Pérez. On the structure of the eigenfunctions of a vibrating plate with a concentrated mass and very small thickness. In *Integral methods in science and engineering*, pages 47– 59. Birkhäuser Boston, Boston, MA, 2006.
- [57] J. K. Hale. Eigenvalues and perturbed domains. In *Ten mathematical essays on approximation in analysis and topology*, pages 95–123. Elsevier B. V., Amsterdam, 2005.
- [58] W. Hansen and N. Nadirashvili. Isoperimetric inequalities in potential theory. In Proceedings from the International Conference on Potential Theory (Amersfoort, 1991), volume 3 (1), pages 1–14, 1994.
- [59] A. Henrot. *Extremum problems for eigenvalues of elliptic operators*. Frontiers in Mathematics. Birkhäuser Verlag, Basel, 2006.
- [60] D. Henry. Perturbation of the boundary in boundary-value problems of partial differential equations, volume 318 of London Mathematical Society Lecture Note Series. Cambridge University Press, Cambridge,

2005. With editorial assistance from Jack Hale and Antônio Luiz Pereira.

- [61] J. Hersch, L. E. Payne, and M. M. Schiffer. Some inequalities for Stekloff eigenvalues. Arch. Rational Mech. Anal., 57:99–114, 1975.
- [62] G. N. Hile and Z. Y. Xu. Inequalities for sums of reciprocals of eigenvalues. J. Math. Anal. Appl., 180(2):412–430, 1993.
- [63] A. M. Il'in. Matching of asymptotic expansions of solutions of boundary value problems, volume 102 of Translations of Mathematical Monographs. American Mathematical Society, Providence, RI, 1992. Translated from the Russian by V. Minachin [V. V. Minakhin].
- [64] Å. Jiménez-Casas and A. Rodríguez-Bernal. Asymptotic behavior of a parabolic problem with terms concentrated in the boundary. *Nonlinear Anal.*, 71(12):e2377–e2383, 2009.
- [65] A. Jiménez-Casas and A. Rodríguez-Bernal. Singular limit for a nonlinear parabolic equation with terms concentrating on the boundary. J. Math. Anal. Appl., 379(2):567–588, 2011.
- [66] T. Kato. Perturbation theory for linear operators. Springer-Verlag, Berlin-New York, second edition, 1976. Grundlehren der Mathematischen Wissenschaften, Band 132.
- [67] V. A. Kozlov, V. G. Maz'ya, and J. Rossmann. Spectral problems associated with corner singularities of solutions to elliptic equations, volume 85 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2001.
- [68] E. Krahn. Über eine von Rayleigh formulierte Minimaleigenschaft des Kreises. Math. Ann., 94(1):97–100, 1925.
- [69] M. G. Krein. On certain problems on the maximum and minimum of characteristic values and on the Lyapunov zones of stability. *Amer. Math. Soc. Transl.* (2), 1:163–187, 1955.
- [70] P. Lamberti and L. Provenzano. A maximum principle in spectral optimization problems for elliptic operators subject to mass density perturbations. *Eurasian Math. J.*, 4(3):70–83, 2013.
- [71] P. Lamberti and L. Provenzano. Neumann to Steklov eigenvalues: asymptotic and monotonicity results. Accepted for publication in the journal *Proceedings of the Royal Sciety of Edinburgh: A*, 2015.

- [72] P. Lamberti and L. Provenzano. Viewing the Steklov eigenvalues of the Laplace operator as critical Neumann eigenvalues. In V. V. Mityushev and M. V. Ruzhansky, editors, *Current Trends in Analysis* and Its Applications, Trends in Mathematics, pages 171–178. Springer International Publishing, 2015.
- [73] P. D. Lamberti. Absence of critical mass densities for a vibrating membrane. Appl. Math. Optim., 59(3):319–327, 2009.
- [74] P. D. Lamberti. Steklov-type eigenvalues associated with best Sobolev trace constants: domain perturbation and overdetermined systems. *Complex Var. Elliptic Equ.*, 59(3):309–323, 2014.
- [75] P. D. Lamberti and M. Lanza de Cristoforis. A real analyticity result for symmetric functions of the eigenvalues of a domain dependent Dirichlet problem for the Laplace operator. J. Nonlinear Convex Anal., 5(1):19–42, 2004.
- [76] P. D. Lamberti and M. Lanza de Cristoforis. Critical points of the symmetric functions of the eigenvalues of the Laplace operator and overdetermined problems. J. Math. Soc. Japan, 58(1):231-245, 2006.
- [77] P. D. Lamberti and M. Lanza de Cristoforis. A real analyticity result for symmetric functions of the eigenvalues of a domain-dependent Neumann problem for the Laplace operator. *Mediterr. J. Math.*, 4(4):435– 449, 2007.
- [78] P. D. Lamberti and M. Perin. On the sharpness of a certain spectral stability estimate for the Dirichlet Laplacian. *Eurasian Math. J.*, 1(1):111–122, 2010.
- [79] M. Lanza de Cristoforis. Simple Neumann eigenvalues for the Laplace operator in a domain with a small hole. A functional analytic approach. *Rev. Mat. Complut.*, 25(2):369–412, 2012.
- [80] M. Lobo and E. Pérez. On vibrations of a body with many concentrated masses near the boundary. *Math. Models Methods Appl. Sci.*, 3(2):249–273, 1993.
- [81] M. Lobo and E. Pérez. Local problems for vibrating systems with concentrated masses: a review. Comptes Rendus Mécanique, 331(4):303 – 317, 2003.
- [82] V. Maz'ya, S. Nazarov, and B. Plamenevskij. Asymptotic theory of elliptic boundary value problems in singularly perturbed domains. Vol. I, volume 111 of Operator Theory: Advances and Applications. Birkhäuser Verlag, Basel, 2000. Translated from the German by Georg Heinig and Christian Posthoff.

- [83] V. Maz'ya, S. Nazarov, and B. Plamenevskij. Asymptotic theory of elliptic boundary value problems in singularly perturbed domains. Vol. II, volume 112 of Operator Theory: Advances and Applications. Birkhäuser Verlag, Basel, 2000. Translated from the German by Plamenevskij.
- [84] A. D. Melas. The stability of some eigenvalue estimates. J. Differential Geom., 36(1):19–33, 1992.
- [85] E. Mohr. Uber die Rayleighsche Vermutung: unter allen Platten von gegebener Fläche und konstanter Dichte und Elastizität hat die kreisförmige den tiefsten Grundton. Ann. Mat. Pura Appl. (4), 104:85– 122, 1975.
- [86] N. S. Nadirashvili. Rayleigh's conjecture on the principal frequency of the clamped plate. Arch. Rational Mech. Anal., 129(1):1–10, 1995.
- [87] S. Nazarov. Asymptotic Theory of Thin Plates and Rods, Vol. I,Dimension Reduction and Integral Estimates. Nauchnaya Kniga, Novosibirsk, 2002.
- [88] J. Nečas. Les méthodes directes en théorie des équations elliptiques. Masson et Cie, Éditeurs, Paris; Academia, Éditeurs, Prague, 1967.
- [89] W.-M. Ni and X. Wang. On the first positive Neumann eigenvalue. Discrete Contin. Dyn. Syst., 17(1):1–19, 2007.
- [90] M. Nicolesco. Recherches sur les fonctions polyharmoniques. Ann. Sci. École Norm. Sup. (3), 52:183–220, 1935.
- [91] O. A. Oleĭnik, A. S. Shamaev, and G. A. Yosifian. Mathematical problems in elasticity and homogenization, volume 26 of Studies in Mathematics and its Applications. North-Holland Publishing Co., Amsterdam, 1992.
- [92] N. Olhoff. Optimal design with respect to structural eigenvalues. Theoretical and applied mechanics, Proc. 15th int. Congr., Toronto 1980, 133-149 (1980)., 1980.
- [93] L. Provenzano. A note on the Neumann eigenvalues of the biharmonic operator. *In preparation*, 2016.
- [94] J. W. S. Rayleigh, Baron. The Theory of Sound. Dover Publications, New York, N. Y., 1945. 2d ed.
- [95] F. Rellich. Perturbation theory of eigenvalue problems. Assisted by J. Berkowitz. With a preface by Jacob T. Schwartz. Gordon and Breach Science Publishers, New York-London-Paris, 1969.

- [96] A. Rodríguez-Bernal. A singular perturbation in a linear parabolic equation with terms concentrating on the boundary. *Rev. Mat. Complut.*, 25(1):165–197, 2012.
- [97] J. Sanchez Hubert and E. Sanchez Palencia. Vibration and coupling of continuous systems. Asymptotic methods. Berlin etc.: Springer-Verlag, 1989.
- [98] E. Sánchez-Palencia and H. Tchatat. Vibration de systèmes élastiques avec des masses concentrées. *Rend. Sem. Mat. Univ. Politec. Torino*, 42(3):43–63, 1984.
- [99] W. Stekloff. Sur les problèmes fondamentaux de la physique mathématique (suite et fin). Ann. Sci. École Norm. Sup. (3), 19:455– 490, 1902.
- [100] G. Szegö. On membranes and plates. Proc. Nat. Acad. Sci. U. S. A., 36:210–216, 1950.
- [101] H. Tchatat. Perturbations spectrales pour des systèmes avec masses concentrées. Université Pierre et Marie Curie, Paris VI, Paris, 1984.
- [102] K. Uhlenbeck. Generic properties of eigenfunctions. Amer. J. Math., 98(4):1059–1078, 1976.
- [103] J. Weidmann. Strong operator convergence and spectral theory of ordinary differential operators. Univ. Iagel. Acta Math., 1(34):153– 163, 1997.
- [104] H. F. Weinberger. An isoperimetric inequality for the N-dimensional free membrane problem. J. Rational Mech. Anal., 5:633–636, 1956.
- [105] R. Weinstock. Calculus of variations with applications to physics and engineering. McGraw-Hill Book Company Inc., New York-Toronto-London, 1952.
- [106] R. Weinstock. Inequalities for a classical eigenvalue problem. J. Rational Mech. Anal., 3:745–753, 1954.

Bibliography
## Ringraziamenti

Il mio più profondo e sincero ringraziamento va al mio supervisore, Professor Pier Domenico Lamberti per la dedizione, l'impegno e la professionalità che ha dimostrato nel seguirmi durante la preparazone di questa tesi. Il Professor Lamberti si è dimostrato un ottimo mentore, sempre presente e pronto a dare consigli, sia dal punto di vista matematico che umano, non solo durante gli anni del dottorato, ma fin dal 2008, quando accettò di essere mio relatore di laurea triennale, ed in seguito anche di laurea magistrale. Senza dubbio è stata una figura fondamentale per la mia crescita professionale e personale e per questo gli sarò profondamente grato, e spero vivamente che la collaborazione scientifica e il rapporto umano che ho con lui possano perdurare a lungo in futuro.

Desidero inoltre ringraziare il mio carissimo amico Davide, che ho avuto il piacere di conoscere già durante la laurea triennale e con il quale ho avuto la gioia di fare della Matematica insieme. Parte dei risultati presenti in questa tesi sono stati ottenuti insieme a lui e confido che ce ne saranno sempre di più e sempre più importanti nel futuro. La sua collaborazione è senz'altro uno dei risultati più preziosi che ho ottenuto durante questi tre anni.

Ringrazio profondamente il mio amico e co-supervisore Matteo, che si è preso l'incarico di seguirmi per alcuni mesi durante il dottorato presso di lui, ad Aveiro. Ho potuto apprezzare e conoscere un modo differente da quello a cui ero abituato di fare Matematica, ed in più ho avuto il piacere di lavorare con una persona eccezionale con cui manterrò e rafforzerò un rapporto di amicizia che ritengo importante.

Un ringraziamento speciale va a Laura. È stata la cosa più bella, incredibile ed importante che mi sia capitata. Ho passato momenti difficili negli ultimi periodi, ma lei mi è sempre stata accanto con forza e decisione, senza mai esitazioni. Non esagero affermando che molto di ciò che sono arrivato ad essere oggi lo devo a lei. Nessuna frase che possa essere scritta qui potrà mai esprimere quello che ha rappresentato, rappresenta e rappresenterà Laura per me.

Voglio esprimere la mia più profonda gratitudine nei confronti della mia famiglia: mio padre Carmelo, mia madre Isabella, mia sorella Alessia e mia nonna Lucia. Durante questi anni il loro sostegno che sentivo forte in ogni momento è stato per me come un faro, la sicurezza di poter avere un punto di riferimento quando anche qualsiasi altro fosse venuto meno.

Ringrazio vivamente i miei coinquilini Giulia, Davide e Mauro. Con loro ho condiviso non solo la casa per quasi tutti gli ultimi nove anni della mia permanenza a Padova, ma anche gioie, dispiaceri, affanni, emozioni. Siamo cresciuti insieme, e loro si sono dimostrati ben più di amici: dei veri compagni di vita.

Un grazie va ai miei compagni di dottorato, Daria, Luisa, Marta, Valentina, Daniele, Federico, Francesco, Gabriele, Michele, ai miei compagni della laurea, Valeria, Gianluigi, Giovanni, e al mio grande amico Michele, per avermi fatto vivere esperienze che per sempre resteranno nel mio cuore e nella mia memoria.

Un ringaziamento è dovuto al Professor Massimo Lanza de Cristoforis, al Professor Carlo Mariconda e al Dottor Paolo Musolino, i cui fondamentali consigli mi sono stati d'aiuto in ben più di un'occasione. Spero di poter continuare ad avere con entrambi un rapporto personale e professionale costruttivo e duraturo.

Ultimi, ma non da ultimi, ringrazio i miei amici di una vita Emanuele, Giordano, Lorenzo, Matteo, e il mio amico fraterno Francesco, che conosco da ormai più di vent'anni e che sono stati sempre presenti per me, seppur da un po' di tempo lontani, anch'essi vero punto di riferimento.

## Ancknowledgments

I wish to express my deepest and sincere gratitude to my supervisor, Professor Pier Domenico Lamberti, for the dedication and the professionalism he demonstrated while following me in the preparation of this thesis. Professor Lamberti has been an excellent mentor, always being available whenever I had doubts, and always ready to give useful advices, both from the mathematical and the human point of view. He cared about my formation as a mathematician since 2008 when he accepted me as a bachelor's student. Without any doubt, he has been a crucial figure for my professional and personal growth and for these reasons I will be always grateful to him. I sincerely hope that our scientific collaboration and our human relationship will last long in the future.

I also wish to thank my friend Davide, which I met during my bachelor's program and with whom I had the greatest pleasure of doing Mathematics together. Part of the results contained in this thesis have been obtained together with him and I am confident that many others are still to come in the future. The collaboration with him is one of the greatest achievements of these years.

I am deeply grateful to my friend and co-supervisor Matteo, who followed my mathematical formation for some months during my Ph.D, when I was his guest in Aveiro. With him, I had the occasion to come in contact and appreciate a different way of doing Mathematics, and I had the pleasure of working with an exceptional person, with whom I will keep and strengthen a friendship which I consider important.

Special thanks go to Laura. She was the most beautiful, incredible and important thing that happened to me. I had some difficult moments in the last times, but she was always by my side with strength and with no hesitations. I am not exaggerating by saying that the most of what I am now is due to her. There is no sentence to be written here that can express what Laura represents and will represent in the future for me.

I wish to express my deepest gratitude to my family: my father Carmelo, my mother Isabella, my sister Alessia and my grandmother Lucia. During these years I could feel their support at any moment, and this was for me like a guiding light, the certainty to have always a reference point.

Many thanks go to my flatmates Giulia, Davide and Mauro. I shared with them not only the flat for almost all the last nine years I spent in Padova, but also sufferings, anxieties, feelings. We have grown together and I can say that they are more than friends to me: they are life companions.

Particular thanks go to my fellow Ph.D students Daria, Luisa, Marta, Valentina, Daniele, Federico, Francesco, Gabriele, Michele, to my fellow bachelor's students Valeria, Gianluigi Giovanni, and to my great friend Michele, with whom I spent unforgettable moments. I will keep all these memories in my heart and in my mind.

Sincere thanks are due to Professor Massimo Lanza de Cristoforis, to Professor Carlo Mariconda and to Doctor Paolo Musolino, whose precious advices have been really important in more than one occasion. I hope that the professional and personal relationship I have with them will last in the years to come.

Last but not least, I thank my friends of a lifetime Emanuele, Giordano, Lorenzo, Matteo and my brotherly friend Francesco. I know them since more than twenty years and they have always been part of my life, being for me a reference point.